Technical University of Denmark



Very High Frequency Switch-Mode Power Supplies.

Miniaturization of Power Electronics.

Madsen, Mickey Pierre; Andersen, Michael A. E.; Knott, Arnold

Publication date: 2015

Document Version Publisher's PDF, also known as Version of record

Link back to DTU Orbit

Citation (APA): Madsen, M. P., Andersen, M. A. E., & Knott, A. (2015). Very High Frequency Switch-Mode Power Supplies.: Miniaturization of Power Electronics. . Technical University of Denmark, Department of Electrical Engineering.

DTU Library Technical Information Center of Denmark

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain
- You may freely distribute the URL identifying the publication in the public portal

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Technical University of Denmark



Mickey Madsen

Very High Frequency Switch-Mode Power Supplies

Miniaturization of Power Electronics

PhD Thesis, May 2015

DTU Electrical Engineering Department of Electrical Engineering

Mickey Madsen

Very High Frequency Switch-Mode Power Supplies

Miniaturization of Power Electronics

PhD Thesis, May 2015

Very High Frequency Switch-Mode Power Supplies, Miniaturization of Power Electronics

This report was prepared by Mickey Madsen

Supervisors Michael A.E. Andersen Arnold Knott

Release date: Category:	June 12th 2015 1 (public)
Edition:	Final
Comments:	This thesis is submitted in partial fulfillment of the requirements for obtaining the PhD degree at the Technical University of Denmark.
Rights:	©Mickey Madsen, 2015

Department of Electrical Engineering Electronics Group (ELE) Technical University of Denmark Oersteds Plads bygning 349 DK-2800 Kgs. Lyngby Denmark

www.ele.elektro.dtu.dk Tel: (+45) 45 25 25 25 Fax: (+45) 45 88 01 17

Preface and Acknowledgment

This thesis was conducted with support from the Danish Innovation Foundation as a part of the research collaboration "008-2011-4 Intelligent and Efficient Power Electronics (IEPE)". The project is carried out in collaboration between Technical University of Denmark, University of Southern Denmark, Aalborg University, Grundfos, Danfoss, Vestas and KK-Electronics, the leading Danish research institutions and companies within power electronics.

The three years of my PhD studies have been an incredible journey. I have learned a lot, gone through a huge personal development and have numerous great memories to look back at. Through the three years I have engaged with many people, whom have all helped and supported me and without those people I would not be where I am today.

First of all I would like to give my sincerest thank to my girlfriend, Marlene. Your patience, compassion, support and endless love throughout the years, have made it possible for me to work hard and devoted on this project.

I also owe a big thank to my Dad, Hans, and my two sisters, Michelle and Mathilde. You have all supported me, accepted that my time to be with you have been scarce and helped me to get my mind of the project from time to time.

This great journey would not have been possible without my supervisors, Michael A.E. Andersen and Arnold Knott. You have both supported me to an extent that is above and beyond what I expected. Time and time again you have supported me in ways I had not imagined.

My closest colleagues through these years have been Jakob Mønster, Thomas Andersen and Jeppe Pedersen. A special acknowledgement goes to Jakob for your help with the PCB embedded magnetics and to all of you for your great work on the prototypes described in chapter 6 and 7. You are all a big part of the reason why these years have been such a great journey - and luckily our venture has just started.

A special thanks goes to all my colleagues at the electronics group, especially to Henriette Wolff, Dorte Svejstrup, Ole Thomsen, Bertil Morelli, Milovan Kovacevic, Lars Petersen and Ivan Jørgensen. You have all been a great support, some of you just by making my day easier and more joyful and some of you also with great technical discussions. For a period I had the special privilege to have a group of new colleagues, namely the Research Laboratory of Electronics at MIT. I would like to thank the entire group for welcoming me, especially David Perreault and Juan Santiago-Gonzalez.

Finally I would like to thank all the people with whom I have collaborated throughout my thesis. The list is long so I will not mention everyone, but a special thanks goes to Fredrik Jeppsson, Niels Petersen, Christian Wolf and Anders Mynster.

To all the people mentioned above and all the people who have supported, I sincerely thank you for all your help, love and support.

Abstract

The importance of technology and electronics in our daily life is constantly increasing. At the same time portability and energy efficiency are currently some of the hottest topics. This creates a huge need for power converters in a compact form factor and with high efficiency, which can supply these electronic devices.

This calls for new technologies in order to miniaturize the power electronics of today. One way to do this is by increasing the switching frequency dramatically and develop very high frequency switch mode power supplies. If these converters can be designed to operate efficiently, a huge size, weight and cost reduction can be achieved due to the smaller energy storing elements needed at these frequencies. The research presented in this thesis focuses on exactly this.

First various technologies for miniaturization of power supplies are studied, e.g. piezo electric transformers, wide band gap semiconductors and integrated power supplies. Afterwards a wide range of topologies suited for operation at very high frequencies is investigated and the most promising ones are tested experimentally. Through a comparison of these topologies the class DE inverter is found to be superior to the other alternatives, at least for converters with hundreds of volts as input and a few tens of watts output power.

A class DE inverter does however require a high side gate drive, which have never been presented before for these frequencies and voltages. This thesis presents the worlds first high side gate drive capable of operating at these frequencies and voltage levels. With this gate drive the worlds first class DE inverter operating at very high frequencies with more than 100 V input is also developed and presented. These achievements are considered huge breakthroughs in the development of technologies for very high frequency switch mode power supplies.

At these highly elevated frequencies normal bulky magnetics with heavy cores consisting of rare earth materials, can be replaced by air core inductors embedded in the printed circuit board. This is investigated thoroughly and both spirals, solenoids and toroids are considered, both for use as inductors and transformers.

Two control methods are also investigated, namely burst mode control and outphasing. It is shown that a very flat efficiency curve can be achieved with burst mode. A 89.5% efficient converter is implemented and the efficiency only drops 5% at 10% load. This is some of the highest efficiencies presented for converters operating at these frequencies. Burst mode control does however have two major drawbacks, introductions of low frequency harmonics and decreased control bandwidth. Outphasing is therefore investigated as an alternative, which does not introduce these drawbacks.

In the last chapter the conducted and radiated electromagnetic interference from two prototypes are investigated, one running with constant output and one with burst mode control implemented.

By the end of the thesis it is shown, that a size reduction of 70%, weight reduction of 81%, cost reduction of 56% and efficiency gain of 4.5%-points can be achieved with a very high frequency class DE converter, compared to a commercial product.

Resumé

Vigtigheden af teknologi og elektronik i vores hverdag er konstant stigende. Samtidig er portabilitet og energi effektivitet nogle af de mest populære emner. Det skaber et enormt behov for strømforsyninger i en kompakt form faktor og med høj effektivitet, som kan forsyne disse elektroniske enheder.

Dette kræver nye teknologier for at miniaturisere nutidens strømforsyninger. En made hvorpå dette kan gøres er at øge skifte frekvensen dramatisk og udvikle meget høj frekvente strømforsyninger. Hvis disse strømforsyninger kan designes til at fungere effektivt, kan en enorm størrelse, vægt og kost reduktion opnås på grund af de mindre energi lagrer der skal bruges ved disse frekvenser. Forskningen der er præsenteret i denne afhandling fokuserer netop på dette.

Først bliver en række teknologier til miniaturisering af strømforsyninger studeret, f.eks. piezoelektriske transformere, wide band gap halvledere og integrerede strømforsyninger. Herefter undersøges et bredt udvalg af topologier egnet til at arbejde ved meget høje frekvenser og de mest lovende testes eksperimentielt. Gennem en sammenligning af disse topologier bliver klasse DE vekselretteren fundet bedre end de andre alternativer, i hvert fald til strømforsyninger med hundrede af volt som indgangsspænding og et to cifret antal watt som udgangseffekt.

En klasse DE vekselretter kræver imidlertid en *high side gate driver*, hvilket aldrig før er blevet præsenteret til disse frekvenser og spændinger. Denne afhandling præsenterer verdens første *high side gate driver* i stand til at arbejde ved disse frekvenser og spændings niveauer. Med denne gate driver er verdens første klasse DE vekselretter der arbejder ved meget høje frekvenser med mere end 100 V som indgangsspænding udviklet og præsenteret. Disse bedrifter betragtes som meget store gennembrud i udviklingen af teknologier til meget høj frekvente strømforsyninger.

Ved disse meget hævede frekvenser kan normale klodsede magnetiske komponenter med tunge kerner bestående af sjældne jordarter erstattes af luft spoler indlagt i printpladen. Dette er undersøgt grundigt og både spiraler, solenoider og toroider er betragtet, både til anvendelse som spoler og transformatorer.

To kontrol metoder er også undersøgt, nemlig tænd-sluk kontrol og udfasning. Det er vidst at en meget flad effektivitets kurve kan opnås med tænd-sluk kontrol. En 89,5% effektiv strømforsyning er implementeret og effektiviteten falder kun 5% ved 10% belastning. Det er nogle af de højeste effektiviteter der er er præsenteret for strømforsyninger ved disse frekvenser. Tænd-sluk kontrol har dog to store ulemper, introduktion af lav frekvente harmoniske og lavere kontrol båndbrede. Udfasning er derfor undersøgt som et alternativ, som ikke introducerer disse ulemper.

I det sidste kapitel er den ledningsbårne og udstrålede elektromagnetiske støj fra to prototyper undersøgt, en der kører med konstant effekt og en med tænd-sluk kontrol implementeret.

I slutningen af afhandlingen vises det, at en størrelses reduktion på 70%, vægt reduktion på 81%, kost reduktion på 56% og effektivitets stigning på 4.5%-point kan opnås med en meget høj frekvent klasse DE strømforsyning, sammenlignet med et kommercielt produkt.

Contents

Pı	refac	e	i
\mathbf{A}	bstra	ict i	ii
R	esum	é	v
Co	onter	nts v	ii
Li	st of	Figures	xi
Li	st of	Tables x	v
1	Intr	oduction	1
	1.1	Background and Motivation	1
	1.2	Objective	2
	1.3	Structure and Content	2
2	\mathbf{Swi}	tch-Mode Power Supplies	5
	2.1	Miniaturization	6
		2.1.1 Resonant Converters	6
		2.1.2 Piezoelectric Transformers	7
		2.1.3 Switched Capacitor	8
		2.1.4 Wide Band Gap Semiconductors	9
		2.1.5 Power Supply in Package and on Chip	12
	2.2	Summary of Trends	4
3	Тор	ologies 1	.5

6	$\begin{array}{c} 6.1 \\ 6.2 \end{array}$	Outphasing	66 67
6	6.1		66
6		Durat Mada	~ ~
~	Cor	atrol	65
		0.2.0 SOIEIIOIQ	01
		5.2.2 Solonoid	0U 61
		 5.2.1 Spiral	59 69
	5.2	Iransformers	59
	F 9	5.1.5 Comparison	55
		5.1.4 Design and Optimization	53
		5.1.3 Toroid	52
		5.1.2 Solenoid	51
		5.1.1 Spiral	50
	5.1	Inductors	49
5		B Embedded Magnetics	49
_	D		
		4.2.2 High Side Gate Drive	45
		4.2.1 Synchronous Rectification	44
	4.2	Self-Oscillating	41
	4.1	Externally Driven	40
4	Gat	e Drive	39
	3.4	Summary	35
		3.3.1 Input-Output Rearrangement	31
	3.3	Stacking	30
		3.2.2 LLC Converter	30
		3.2.1 Class DE	28
	3.2	Half Bridge	28
		3.1.4 Resonant Boost	27
		3.1.3 Single-Ended Primary-Inductor Converter	26
		3.1.2 Class ϕ_2	25
		3.1.1 Class E	20
	3.1	Single Switch	19

8	Con	Conclusion and Future Perspectives		
	8.1	Summary	79	
	8.2	Conclusion	81	
	8.3	Future Perspectives	82	
		8.3.1 Research within VHF SMPS	82	
		8.3.2 Technologies for Miniaturized Power Electronics	84	
Bi	ibliog	raphy	87	
\mathbf{A}_{j}	ppen	dix	109	
\mathbf{A}	Pub	lications and Patents	109	
	A.1	On the Ongoing Evolution of Very High Frequency Power Supplies $% \mathcal{A}^{(n)}$.	112	
	A.2	Evolution of Very High Frequency Power Supplies	119	
	A.3	Low Power Very High Frequency Resonant Converter with High Step Down Ratio	129	
	A.4	Low Power Very High Frequency Switch-Mode Power Supply with 50 V Input and 5 V Output	136	
	A.5	Very High Frequency Resonant DC/DC Converters for LED Lighting	149	
	A.6	Very High Frequency Half Bridge DC/DC Converter	155	
	A.7	Input-Output Rearrangement of Isolated Converters	162	
	A.8	Step-up dc-dc power converter	169	
	A.9	Step-down dc-dc power converter	207	
	A.10	Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rec- tifiers Composed Solely of Passive Components	250	
	A.11	Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter	258	
	A.12	Self-oscillating resonant power converter	264	
	A.13	Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies	308	
	A.14	Design Optimization of Printed Circuit Board Embedded Inductors through Genetic Algorithms with verification by COMSOL	317	
	A.15	Investigation of a Hybrid Winding Concept for Toroidal Inductors using 3D Finite Element Modeling	324	
	A.16	Optimizing Inductor Winding Geometry for Lowest DC-Resistance using LiveLink between COMSOL and MATLAB	329	

A.17	Investigation, development and verification of printed circuit board embedded air-core solenoid transformers	335
A.18	Embedded solenoid transformer for power conversion	343
A.19	Outphasing Control of Gallium Nitride based Very High Frequency Resonant Converters	379
A.20	Burst Mode Control	386
A.21	On and Off Controlled Resonant dc-dc Power Converter \ldots	430
A.22	Confidential	470

List of Figures

2.1	Apples 60 W sugar cube laptop charger	5
2.2	Schematic of a serial-parallel resonant LLC converter	7
2.3	Picture and drawing of a rosen type piezoelectric transformer $\ . \ . \ .$	8
2.4	Schematic of a switch capacitor circuit with a step-up conversion ratio of 5	8
2.5	Bounderies of semiconductor materials and switch technologies	10
2.6	Material chioce depending on power level and operation frequency $% {\displaystyle \sum} {\displaystyle $	12
2.7	Three levels of integration	12
2.8	Material and technology selection depending on voltage and power level	14
3.1	Block diagram of a VHF converter	15
3.2	Relation between $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ and η for the converters in table 3.2	18
3.3	Schematic of the class E inverter	20
3.4	Normalized values of L_R and C_R as function of loaded Q factor	23
3.5	Deviation from the decired reactance depending on loaded Q factor.	23
3.6	Schematic of the class E rectifier	24
3.7	The $\frac{P_{OUT}}{V_{IN}\cdot f_S}\text{-}\text{factor}$ and η achieved in A.4 next to previous results	25
3.8	Schematic of the class ϕ_2 inverter	26
3.9	Schematic of the SEPIC converter	27
3.10	Schematic of the resonant boost converter	27
3.11	Schematic of the class DE inverter	28
3.12	Schematic of the class DE rectifier	29
3.13	Conventional isolated dc/dc converter topology. Dashed line repre- sents galvanic isolation.	31

3.14	Step-up configuration: two converter cells with inputs in parallel and outputs in series (top) and a single cell equivalent (bottom)	32
3.15	Step-down configuration: two converter cells with inputs in series and outputs in parallel (top) and a single cell equivalent (bottom).	32
3.16	Power processed by the converter and efficiency improvement by re- arranging	34
3.17	Two stacked converters in step-down configuration	34
3.18	Measured efficiencies on two rearanged isolated converters	35
4.1	Examples of a resonant gate drive	40
4.2	Example of a self-oscillating VHF inverter	41
4.3	Schematic of the basic self-oscillating gate drive. The gate resistance and the body diode has been left out for simplicity	41
4.4	Schematic of the self-oscillating gate drive with the 2nd and 4th harmonic to source and drain.	43
4.5	Examples of transfer functions from drain-source to gate-source for implementations of the resonant gate drive with added capacitance or 2nd harmonic LC filter	43
4.6	The 1st, 3rd and 5th harmonics in and out of phase with the drain signal.	43
4.7	Gate signals with 1st, 3rd and 5th order harmonics for 25% and 50% duty cycle (*=signal in phase with V_{DS}).	44
4.8	Schematic of a class E dc/dc converter with synchronous rectification.	44
4.9	Schematic of the class DE inverter with self oscillating gate drive	45
4.10	Picture of the worlds first descrite VHF class DE converter	46
4.11	Measured waveforms on the class DE converter	46
4.12	Efficiency and output power as function of input voltage for a VHF class DE converter.	47
5.1	A two layer PCB embedded spiral inductor	50
5.2	A PCB embedded solenoid inductor	51
5.3	A two layer PCB embedded toroidal inductor	52
5.4	MATLAB GUI for design and optimization of PCB embedded in- ductors	54
5.5	Measurements of the inductance and DC and AC resistance of some of the experimental prototypes	55
5.6	Inductance and series resistance of prototypes with a low number of turns (solid) and high number of turns (dashed).	56

5.7	Three types of PCB embedded inductors	57
5.8	Finite element simulations of the magnetic flux density (B-field [T]) inside the three types of PCB embedded inductors.	57
5.9	Finite element simulations of the magnetic field (H-field $[A/m]$) 2 mm above the three types of PCB embedded inductors	57
5.10	Near-field scans of the field strength from the three types of PCB embedded inductors (z component not included as in simulation)	57
5.11	Resonant converters with the different structures for PCB embedded inductors	58
5.12	Example of a two layer PCB embedded spiral transformer	60
5.13	Example of a four layer PCB embedded toriodal transformer $\ . \ . \ .$	60
5.14	Different winding configurations of rectangular PCB embedded air- core solenoid transformers	61
6.1	Measured efficiency curve for a DE converter switching at 27 MHz with fixed frequency burst mode control at 20 kHz	67
6.2	System view of a converter using outphasing modulation	67
6.3	Schematic of the lossless combiner for outphasing modulation $\ . \ . \ .$	68
7.1	Pictures of 20 W converters operating at 37 MHz	73
7.2	EMI measurements on a 20 W converter operating at 37 MHz	75
7.3	A 20 W prototype based on a 27 MHz class DE converter	76
7.4	Conducted EMI measurements on a converter with burst mode control.	76
7.5	Radiated EMI measurements on a converter with burst mode control.	77
8.1	A 20 W prototype based on a 27 MHz DE converter compared with an equivalent commercial product.	81

List of Tables

1.1	Structure and content of the thesis	4
2.1	Material proporties related to performance in power electronics	10
3.1	Two sets of converter specifications	17
3.2	Results from previous research	19
3.3	Pros and cons of the investigated inverter topologies	35
5.1	The weight factors used for the cost function	54
5.2	Parameters for three inductors with approx. 82 nH inductance in 8x8mm	55
5.3	Pros and Cons of the three structures	58
5.4	Inductances of different configurations of PCB embedded solenoid transformers.	62
7.1	EN55022 limits for conducted disturbance at the mains ports of class B ITE	71
7.2	EN55022 limits for radiated disturbance of class B ITE at a measuring distance of 10 M	72
8.1	Performance improvements compared with a commercial product	81

CHAPTER 1

Introduction

1.1 Background and Motivation

Electronics has forever changed the way we live and interact with each other. It is the basis for the information age and the importance of electronics in our daily life is constantly increasing. Few people can imagine even a single day without at least a few electronic devices such as artificial lighting, television, radio, kitchen appliances, computers and cell phones.

All of these devices require power conversion and power electronics has hence become one of the key enablers of modern civilization. In most cases power conversion is handled by a switch-mode power supply (SMPS) due to its high efficiency and power density. Traditional hard-switched SMPSs have, however, reached a point where only incremental improvements are achieved.

In order to fulfill the ever-increasing demand for miniaturization and portability, new ways of designing power converters is hence needed. The power converter is also the weakest link in regards to lifespan in many electronic products. A good example is the rapidly increasing market for LED lighting, where more than 50% of the products fail due to the LED driver [1].

Global warming and reduction of the global energy consumption is one of the main focus areas of the 20th century. Hence no new technology can come at the expense of increased energy consumption. High efficiency is therefore a must have for any new power conversion technology.

At the same time the market for power converters is, as most markets, driven by cost. Any new technology with the above-mentioned benefits should, therefore, come at the same price, preferably lower, as the technology of today.

The volume, weight, and cost of SMPSs are mainly governed by passive energy storing elements (inductors and capacitors). The value, size, weight, and price of these scales with the switching frequency and a dramatic increase in switching frequency will hence lead to highly increased power density and reduced cost. Furthermore, decreased need for energy storage leads air core magnetics to be a feasible alternative to cored magnetics, which utilize rare earth materials that are primarily mined in an environmentally damaging process in China [2]. The reduced need for capacitance eliminates (or significantly reduce) the need for electrolytic capacitors, which are the main cause of failure in power converters [3, 4, 5].

SMPSs with highly increased switching frequencies thereby posses the ability to enable all the above-mentioned benefits. In order to make a real difference, the frequency has to be increased a decade or more. If the switching frequency is increased a decade from 1-2 MHz, which is state-of-art today [6, 7], a switching frequency of 10-20 MHz will be achieved. At these frequencies, the core loss in the core materials available today start to become significant and lead to decreased efficiency [8]. Using air core magnetics is as mentioned a feasible alternative, but less inductance per volume can be achieved without the core. The frequency, therefore, needs to be increased even further, to get the desired size reduction without the magnetic core.

1.2 Objective

The objective of this PhD thesis is to investigate, whether it is possible to increase the switching frequency of power converters into the Very High Frequency range (VHF, 30-300 MHz) [9].

The converters should achieve efficiencies at least on par with commercial products and use standard components in order to keep cost low. The aim is to show that a size and weight reduction of 80% can be achieved without compromising cost, efficiency or reliability.

1.3 Structure and Content

The structure and content of the PhD thesis are visualized in Table 1.1.

The first chapter covers an introduction to the thesis with background and motivation and the scope and objectives of the project. Chapter two covers a state-of-art analysis with special focus on technologies for miniaturization of power converters.

The circuit topologies suitable for use at these frequencies are analyzed in chapter three. Furthermore the option of stacking multiple converters with serial or parallel input and output are analyzed and a new arrangement of the input and output terminals is described.

Chapter four analyses the gate drives suitable for operation at these frequencies and especially centers on a new self-oscillating resonant gate drive. Chapter five covers the design of PCB embedded magnetics based on spirals, solenoids and toroids, both inductors and transformers are analyzed.

The control is covered in chapter six and in chapter seven the electromagnetic interference of the converters are analyzed and measured.

Finally chapter eight summarizes and concludes the thesis. The main results are

summerized and the work to be done in the future is described.

The chapters and sections are linked with the associated peer reviewed conference papers, journal paper and patent applications; these are all located in appendix A. These papers and patents are an integrated part of the thesis and the thesis mainly covers complementary information. This is to present a coherent and complete overview of the research work that has been done throughout the project.

Chapters	Sections	Appendicies			
	Background and motivation				
1 Introduction	Objective				
	Structure and content				
2 Switch Mode Power Supplies	Miniaturization	A.1 "On the Ongoing Evolution of Very High Frequency Power Supplies"A.2 "Evolution of Very High Frequency Power Supplies"			
	Summary of trends				
3 Topologies	Single switch	 A.3 "Low Power Very High Frequency Resonant Converter with High Step Down Ratio" A.4 "Low Power Very High Frequency Switch-Mode Power Supply with 50 V Input and 5 V Output" A.5 "Very High Frequency Resonant DC/DC Converters for LED Lighting" 			
	Half bridge	A.6 "Very High Frequency Half Bridge DC/DC Converter"			
	Stacking	A.7 "Input-Output Rearrangement of Isolated Converters"A.8 "Step-up dc-dc Power Converter"A.9 "Step-down dc-dc Power Converter"			
	Summary				
	Externally driven				
4 Gate drive	Self-oscillating	 A.10 "Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components" A.11 "Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter" A.12 "Self-oscillating Resonant Power Converter" 			
5 PCB embedded magnetics	Inductors	A.13 "Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies" A.14 "Design Optimization of Printed Circuit Board Embedded Inductors through Genetic Algorithms with verification by COMSOL" A.15 "Investigation of a Hybrid Winding Concept for Toroidal Inductors using 3D Finite Element Modeling" A.16 "Optimizing Inductor Winding Geometry for Lowest DC-Resistance using Live Link between COMSOL and MATLAB"			
	Transformers	A.17 Investigation, development and verification of printed circuit board embedded air-core solenoid transformers"A.18 "Embedded solenoid transformer for power conversion"			
6 Control	Burst Mode				
U CONTION	Outphasing	A.19 "Outphasing Control of Gallium Nitride based Very High Frequency Resonant Converters"			
7 Electromagnetic Interference					
8 Conclusion	Summary				
& Future	Conclusion				
Perspectives	Future work				

 Table 1.1: Structure and content of the thesis.

Chapter 2

Switch-Mode Power Supplies

The first switch-mode power supplies were developed in the early 70's [10] and have since become the market standard, due to amongst other the high power density, high efficiency and lower cost compared to iron core transformers.

The first SMPSs operated at 20 kHz with efficiencies around 75% and power densities of approximately 0.15 W/cm^3 [11]. Since then, the technology has been improved, matured, and components optimized for use specifically in power supplies developed.

This combined with more than 40 years of research and development has moved the technology far. Today efficiencies up to 99.7% [10] and power densities of more than 10 W/cm³ [12] can be achieved. Through the past decade a doubling in power density have been seen and this is expected to continue for the next decade [13].

These efficiencies and power densities are, however, for kilo and mega watt power converters. The main focus of this thesis is converters with an output power in the range from a few to around hundred watts. At these power levels, the performance that can be achieved is lower, some of the best results published are efficiencies around 95% [14] and power densities of 0.88 W/cm³ [15].



Figure 2.1: Apples 60 W sugar cube laptop charger.

These results are, however, achieved in a laboratory with a controlled environment and without a strong focus on cost. For commercial products, cost is one of the main concerns and reduced efficiency and power density is accepted in order to keep the cost low.

Apple is known for their sugar cube laptop adaptors, which are some of the smallest on the market, see Figure 2.1. The power density of the 60 watt version is 0.59 W/cm^3 (60 $W/(6.4 cm \cdot 6.4 cm \cdot 2.5 cm)$, including casing and plug) and the efficiency is 90% [16]. For a USB charger, the efficiency and power density is even worse, with an efficiency of 75% and power density of 0.31 W/cm^3 (5 $W/(2.54 cm)^3$, including casing and plug) [17]. The same trend can be seen for LED drivers, where a 60 W driver with a power density of 0.15 W/cm^3 (60 $W/(18.15 cm \cdot 6.2 cm \cdot 3.5 cm)$, including casing and plug) and efficiency of 89% is commercially available [18]. The drop in efficiency and power density is partly due to the fact that casing, plugs, control, start-up, protection and other housekeeping circuitry is needed independent of the power level and partly due to a tradeoff with price. As the power level increases, the efficiency becomes more important and it becomes more acceptable to increase the price slightly, if a higher efficiency can be achieved due to the larger amount of energy that can be saved.

2.1 Miniaturization

As mentioned in Chapter 1, the traditional SMPSs have reached a point where the technology is matured and only incremental improvements are achieved year-on-year.

The market for power supplies is huge (\$21.6 billion in 2014 [19]) and the technology still posses some big drawbacks. It has become one of the major bottle necks for further size reduction of modern electronic products. Hence, there are large market forces driving the research and development of new technologies, which can improve or change the way power supplies are made, some of these are described in the following subsections.

2.1.1 Resonant Converters

As stated in Chapter 1, the size of modern SMPSs is mainly governed by the passive energy storing elements. The direct way to reduce the size of these is to increase the frequency. If this is done without paying special attention to the switching losses, the efficiency will be poor and most likely cause overheating and failure of the power semiconductors. With resonant converters, zero voltage switching (ZVS) can be achieved and losses due to parasitic switch capacitance can thereby be avoided [20].

Several types of resonant converters exist, but they can all be split into three groups, series resonant, parallel resonant and series-parallel resonant converters. The series resonant converter has the best efficiency and lowest complexity, but has fundamental challenges with output regulation, especially for light and no-load situations. Parallel resonant converter has better load regulation, but a major

drawback because the resonating currents do not scale with the output power. Hence it will have the full load losses even at light loads, causing very low light load efficiencies [21].



Figure 2.2: Schematic of a serial-parallel resonant LLC converter [22].

Series-parallel resonant converters have both a series resonant and a parallel resonant element. These elements can be balanced to get the advantages of both the series resonant and the parallel resonant topologies while reducing the drawbacks significantly. The LLC converter shown in Figure 2.2 is the most commonly used topology for resonant converters. It can be designed with zero voltage swithcing (ZVS) to reduce switching losses and increase frequency [23]. LLC converters are often used in step down application from several hundreds volts to a few tens of volts and commonly in the power range of 400-4000 W [24].

Several 1 kW LLC converters operating at 1 MHz with efficiencies over 90% have been made [22, 6, 24, 7]. The converter presented in [6] achieves a peak efficiency above 96% and a power density of 5.86 W/cm^3 . The one in [7] has a power density of more than 50 W/cm³ and in [24] a GaN based LLC converter operating at 350 kHz with 98% efficiency is presented.

For the last decade there has been an increasing focus on and research in resonant power converters operating at very high frequencies [25, 26, 27, 28, 29, 30]. Moving into this frequency range dramaticly reduces the need for passive energy storage and cored magnetics and electrolytic capacitors can be replaced by air-core magnetics and ceramic capacitors, hence minimizing size and price while extending the lifespan [31, 32] and A.2. This will be covered more in-depth in the following chapters.

2.1.2 Piezoelectric Transformers

The magnetics, and especially the transformers, are often the largest components in an isolated converter. Hence, reducing the size of the transformer would be a great step towards miniaturization of power electronics. One way to do this is by using a piezoelectric transformer, which utilizes two pieces of ceramic material with a piezoelectric property. When a material with a piezoelectric property is subjected to a mechanical stress it becomes electrically charged. Likewise, an electrical field will cause a deformation of the material. These effects are also known as the direct and converse piezoelectric effect, respectively [33, 34]. In the 1950's C. A. Rosen [35] utilized these properties to create the rosen type piezoelectric transformer, see Figure 2.3.



Figure 2.3: Picture and drawing of a rosen type piezoelectric transformer [36].

These transformers are especially suitable for applications where a high step ratio is required [37, 38]. The material exhibits a very high Q factor (>1000 [39]), high efficiencies (>98% [37]) and power density of more than 100 W/cm³ [33] can be achieved. This is, however, for the transformer alone. For a complete converter, the efficiencies are closer to 80% and the power densities around 0.5 W/cm³ [40, 41]. The piezo based converters operate around the resonance frequency of the transformer and for most available transformers this frequency is around 50 kHz [38, 34]. This limits the power density that can be achieved, as the converters will still require big input and output filters. Piezo based transformers have, however, been tested at more than 1 MHz [33], but here the efficiency of the transformer alone drops below 90%. Piezo based converters, therefore, seem to be best suited for high step up or down applications, where the limited efficiency and power density can be justified by the big conversion ratio.

2.1.3 Switched Capacitor

Another way to get rid of the bulky magnetics is to use switch capacitor (SC) circuits. These circuits do not rely on energy storage in magnetics, but use a network of switches and capacitors for energy conversion [42, 43]. An example of a switch capacitor circuit is shown in Figure 2.4.



Figure 2.4: Schematic of a switch capacitor circuit with a step-up conversion ratio of 5 [42].

As the energy density of capacitors is over 1000 times higher than that of inductors, it is possible to achieve a large increase in power density with these circuits [44, 45]. Further fully monolithic integration in silicon becomes much easier as switches and capacitors are much easier to integrate than inductors [44, 46, 43]. Efficiencies up to 93% [47] and power densities of several watts per square millimeter [48, 49, 50] has been achieved in fully integrated circuits.

One of the main challenges of this type of circuits is control and especially efficiency when the converter is used to regulate the voltage [46, 43]. These converters are designed to operate with a fixed integer conversion ratio from input to output voltage where transition between parallel and serial connection of a finite number of capacitors are used to achieve this conversion ratio. When operating at this optimum point the efficiency is ideally 100% [43], however the efficiency quickly rolls of when the point of operation moves away from this optimum. Lately a lot of research has been done within so called gear shifting switch capacitor circuits [46, 51]. These switched capacitor circuits have different modes of operation with different optimum conversion ratios, for instance 2:1 and 3:2. This increases the range in which the circuits can operate efficiently, but increases the complexity.

The number of switches and capacitor increases rapidly with the conversion ratio [42]. With a moderate conversion ratio of 5, 4 capacitors and 10 switches are required, see Figure 2.4. Further all of these switches requires a gate drive and 7 of them even a high side gate drive. This makes the complete circuit quite complex and a lot of components are required. This is not a big challenge in integrated circuits where the components can be scaled according to the application and placed very close to each other. For discrete circuits the size would however quickly increase as soon as the step ratio becomes more than a simple 2:1 or 1:2 converter.

For the above mentioned reasons, switched capacitor circuits are mainly used in fully integrated circuits and mainly with conversion ratios up to 8:1 [44, 43]. The highest power level presented so far is 10 W [51] and the voltage levels are typically from a few volts up to around 12 V [48, 45].

2.1.4 Wide Band Gap Semiconductors

Since the development of the switch mode power supply in the early 70s, one of the main drivers towards miniaturization of power electronics has been increased switching frequencies [13]. The ability to increase the switching frequency while still achieving the same, or even higher efficiencies, can greatly be attributed to the development of better power semiconductors. Silicon based power semiconductors have however reached the fundamental limit of the material properties. For now constant innovation has enabled semiconductor manufactures to push the boundaries, see Figure 2.5, but the pace is slowing down [52].

Lately there has been an increasing focus on so-called wide band gap power semiconductors. These power semiconductors are based on new materials, which require larger energy for an electron to jump from the top of the valence band to the bottom of the conduction band (the band gap) [54]. The band gap and other important parameters of these new materials are summarized in Table 2.1. Some of the materials have a range on some of the properties; this is not due to variations in these parameters, but due to in consistence in the literature. The references [55, 56, 57, 58] all list some or all of these parameters, but there is a significant variation in the values they list. Most of the values are showing the same trend, but for the break down field the literature is not consistent in which of the materials GaN and 4H-SiC that has the largest value ([56] states SiC > GaN whereas [57, 58] states GaN > SiC).

The literature is however consistent in stating that these materials are superior to silicon in terms of higher operating temperatures, higher power densities, higher voltages and higher frequencies [61, 62, 54]. Several different figures of merits (FOM) have been made in order to compare the different materials. The JFOM derived by Johnson in 1965 [59] and the BFOM derived by Baliga in 1982 [60] is



Figure 2.5: Bounderies of semiconductor materials and switch technologies [53].

	Si	GaAs	4H-SiC	GaN	Diamond
Bandgap E_g (eV)	1.12	1.42	3.26	3.39	5.45
Dielectric constant $\epsilon \ (\mathrm{cm}^{-3})$	11.8-11.9	12.5-13.1	10	9.0-9.5	5.5
Mobility $\mu_n \; (\mathrm{cm}^2/\mathrm{V} \; \mathrm{s})$	1300-1350	5000-8500	260-720	900-2000	1900
Saturation velocity v_{sat} (MV/cm)	1.0	1.0-2.0	2.0	2.5	2.7
Break down field E_{br} (MV/cm)	0.3	0.4	2.0-3.5	2-3.3	5.6
Thermal conducti. λ (W/cm K)	1.5	0.5	4.5	1.3	20
Johnson FOM [59] $E_{br} \cdot v_{sat}/2\pi$	1	2.7	20	27.5	50
Baliga FOM [60] $\epsilon \cdot \mu_n \cdot E_g^3$	1	9.6-17	3.1	24.6-134	5.45

Table 2.1: Material proporties related to performance in power electronics.

the most widely adopted FOMs, for both high values are good. JFOM is based on the power-frequency product for low voltage transistors and BFOM is based on the conduction losses in power FETs [63]. The two FOMs are defined as:

$$JFOM = E_{br} \cdot v_{sat}/2\pi \tag{2.1}$$

$$BFOM = \epsilon \cdot \mu_n \cdot E_g^3 \tag{2.2}$$

GaAs Gallium Arsenide is without comparison the compound with the highest mobility. This makes it excellent for circuits were very fast switching speed is the main parameter. For this reason it is widely adopted in communication circuitry for cellphones, radars, satellites and other microwave applications. However the low break down field makes it unsuited for applications were more than a few volts are required.

4H-SiC Silicon Carbide is, contrary to GaAs, very well suited for high voltage applications. The 4H refers to the crystal structure of the material [64]. 4H-SiC has the highest electron mobility and is therefore considered to be the best structure [65]. Due to the high break down field it has been possible to fabricate a MOSFET with a break down voltage of 10 kV, the highest ever [62]. The combination of high break down voltage, the high thermal conductivity and the fact that SiC can operate at up to 350 °C [66] makes it extremely well suited for high power applications. Several implementations have shown the high efficiency and power densities which can be achieved with these new devices, two examples are a converter with an efficiency of 99% and power density of 3.3 W/cm^3 [67] and another with an efficiency of 98% and a power density of 62 W/cm³ [68].

GaN Gallium Nitride is the least mature of the three wide band gap materials [61], at least for power electronics. The material is widely used for blue LED (which are then turned white by adding a layer of phosphor) [55]. The RF industry is also starting to adopt GaN, as it can be used for higher power and voltage levels than GaAs due to the higher break down field and thermal conductivity. With the high mobility and high break down voltage, GaN is also very well suited for switch-mode power supplies and class D amplifiers. Due to the lower thermal conductivity compared to SiC it is however most suited for power levels up to a few kW. Several examples of GaN based point of load converters are shown in [69], all operating at 1-5 MHz with 1.2 V output and achieve power densities around 60 W/m³ and efficiencies between 85% and 95%.

The transition from silicon to wide band gap semiconductors will definitely help to move towards miniaturization of power electronics. Great results have already been achieved and the technologies, especially GaN, are just starting to become mature enough to be used in commercial products. Further each of the three materials seems to have found their own market niche, see Figure 2.6, and are starting to get traction.



Figure 2.6: Material chioce depending on power level and operation frequency [70].

2.1.5 Power Supply in Package and on Chip

The ultimate goal for miniaturization of power electronics is to integrate the entire power supply on a chip (PwrSoC). However taking a 10 kW converter and just going for full integration in one step would be a very big and risky step. Therefore the road from complete discrete solutions to fully integrated converters has been split into different levels of integration. Figure 2.7 illustrates this with the three levels 2.7a Power modules, 2.7b Power Supply in a Package (PSiP) and 2.7c PwrSoC.



Figure 2.7: Three levels of integration.

Modules Power modules are the first step towards integration. Here the entire converter is tightly packed in a single module. The converter design inside the modules are highly optimized and the interconnections between the components made much more advanced, than what can be done on a standard PCB with discrete components. Further the availability of complete modules with entire power converters ease the design for system engineers, as they can leave the converter design to the module manufactures. Vicor is one of the leading companies in this area, with a wide selection of modules and some of the best specifications. They have just released a new 400 W AC-DC module with PFC, 92 % efficiency and a

power density of more than 7 W/cm^3 . The modules are however still to expensive for most applications, with a price point of more than \$150 for the Vicor module just mentioned.

PSiP The next step towards full integration is PSiP. Here all the active components are integrated on the same die, but at least one of the passives are still discrete and co-packaged in the same package as the die. Here the advances in packaging and interconnection gained through the modules are taken one step further. The active die and the passives are now packed to a level, where it can be difficult to distinguish a PSiP and a PwrSoC just by looking at the size. The number of PSiP products available have greatly increased through the last years with more than 200 products on the market already in 2012 [74].

A good example of a PSiP is the 480 MHz buck presented in [75]. It has one air core inductor that is not on the chip, but the high switching frequency is only possible due to the tight interconnections between the die and the passives. The converter steps from 1.8 V to 0.9 V with 72% efficiency and the efficiency can be increase to 76% by decreasing the switching frequency to 250 MHz. In [76] a GaN buck with external filter, a switching frequency of 10-200 MHz and an output power of 10 W is presented. This is very close to a PSiP, but due to the fact that the control is not on the same die as the converter, it does not meet the definition of a PSiP [74]. Combinations of wide band gap devices and the tightly packed PSiPs are however expected to bring very interesting products on the market within the near future [77].

PwrSoC The final step towards monolithic integration, is to get the passives on the same die as the active components. It is a great challenge to integrate passives and due to the lack of core materials, the switching frequency should generally be above 50 MHz for PwrSoC to achieve higher power density than PSiP [78]. One solution to this is of course to use the switch capacitor circuits described in section 2.1.3, in this way no inductors are needed and only capacitors need to be integrated. Switch capacitors will however not be covered further in this section, as they have already been described.

The bond wires can be used as inductors as in [79], were a fully integrated 200 MHz converter with a peak efficiency of 89% is presented. This is efficient as bond wires have high Q factor compared to what can be achieved on chip, but it is not a true PwrSoC as the inductors are still of chip. In [73] the inductors are moved a bit closer to the chip by implementing them in the silicon interposer. In this way a converter with a switching frequency of up to 300 MHz, optimum at 200 MHz, and peak efficiency of 75% is designed. The next step toward true monolithic integration is to move the inductor to the bottom layer of a standard flip chip package, this is done in [80] were a 140 MHz 90% efficient converter is presented. Figure 2.7c shows the fully integrated buck converter presented in [73]. Though some of the other examples almost meet the definition of a PwrSoC, this is the only true PwrSoC. It has a switching frequency up to 225 MHz and converts from 2.6 V to 1.2 V with an efficiency of 58%, the die shown handles up to 800 mW.

It is very early days for PwrSoC with just a few experimental results and the
first commercial product presented in the fall 2014 [80]. Constant improvements in packaging and especially thermals are however needed to get more and better PwrSoC product on the market [74]. Finally all the PSiPs and PwrSoCs shown are working at very low voltages, the range of 0.5-3 V.

2.2 Summary of Trends

Several existing and new emerging technologies for miniaturisation of power electronics have been presented in this chapter. To give an overview the technologies and the voltage and power levels were they are most suited, they are mapped and shown in Figure 2.8.



Figure 2.8: Material and technology selection depending on voltage and power level.

The general trend is to push for higher operating frequencies in order to minimize the passive elements. This is done both through new circuit topologies and new components/materials. The only technology not following this trend is the piezo transformer, they are best suited for operation at relatively low frequencies. With this technology the bulky transformers can be replaced by small ceramic elements, but big input and output filters are still needed.

The rest of the technologies support each other quite well and the different materials each have their own areas where they are best suited. For the voltage and power levels of interest in this thesis, the technologies from the modules/PSiP and GaN could go very well hand in hand with resonant VHF converters. SiC are generally better suited for higher power and voltage levels and GaAs, SC and PwrSoC are still to far from the power and voltage levels of interest. Chapter 3

Topologies

Traditional SMPS topologies like Buck and Boost are hard switching, this means the MOSFET is switching while there is voltage across it and/or current running through it. The result is that energy is dissipated in the MOSFET every time it turns on, i.e. switching losses. In traditional converters the switching frequency is chosen as a tradeoff between switching losses, size and price. In most commercial product a switching frequency in the range 50-400 kHz is chosen as this gives a good tradeoff. When the frequency is increased to the Very High Frequency (VHF) range (30-300 MHz) the switching losses get almost 1000 times larger. This amount of energy would ruin the efficiency and require extreme cooling of the MOSFET.

In order to avoid switching losses and be able to increase the frequency while keeping the efficiency high, new topologies have to be used. Through three decades (since the 1980's [89, 90]) research has been done in order to enable the use of resonant RF amplifiers (inverters) combined with a rectifier for dc/dc converters, see Figure 3.1. With this type of converters it is possible to achieve Zero Voltage Switching (ZVS) and/or Zero Current Switching (ZCS). In this case the MOSFET turns on when the voltage and/or current across/through it is zero. Theoretically this should eliminate switching losses, if the switching is done instantaneously and at exactly the right time. This is not practically achievable, but even with slight deviations from the ideal case very high efficiencies can be achieved.



Figure 3.1: Block diagram of a VHF converter.

As already mentioned the value of the passive components depends on the switching frequency. Hence an increase in frequency will lead to a reduction in size, as long as the size of the passives scales with the value. This assumption generally holds, but magnetic materials and packaging introduce some challenges. When the frequency is pushed far into the MHz range, magnetic core loss increases rapidly and becomes unacceptably high for most core materials [91]. At this point air core and PCB embedded inductors become a viable solutions, as the inductances needed at these frequencies can be made in a small physical size and the core losses avoided [92, 93].

Increasing the switching frequency also leads to capacitors with lower values. Electrolytic capacitors, which often limits the overall lifetime, can hence be avoided [94, 95]. The reduction in component values also leads to a cost reduction, as smaller components are generally cheaper. If the frequency is increased enough, some of the components can even be left out, as they can be constituted by the parasitic parts of other components. The capacitor in parallel with the MOSFET will for instance often be, at least partly, constituted by the ouput capacitance of the MOSFET. An increase in switching frequency will also make it easier to comply with EMI requirements, as small and cheap filters can easily filter out switching harmonics.

Since the development of the first SMPSs, there has been a general push towards higher frequencies for the above-mentioned reasons. This and the recent development in wide band gap semiconductors has lead to switching frequencies at a few MHz in todays state-of-the-art SMPSs [96]. Following this trend with incremental improvements and increase in frequency, the next step would be to make SMPSs with switching frequency of 5-10 MHz. However at frequencies above a few MHz the core materials start to exhibit huge core losses and air core inductors becomes the only efficient solution. Without the core the structure needed to achieve the desired inductance at 5-10 MHz would however be even bigger, than the cored inductors that could be used at 1-2 MHz. Therefore the frequency needs to be pushed even further to achieve a size reduction without core materials. At frequencies around 20-25 MHz similar sizes can be achieved and beyond that size reduction can be achieved even with the air core inductors [97].

With a switching frequency between 30 and 300 MHz, the main concern when selecting topology is switching losses. The switching loss in a MOSFET due to the parasitic output capacitance increases linearly with the switching frequency and becomes the dominating loss mechanism at these frequencies, if the topology does not take this into account.

To illustrate the losses at these frequencies, the simplified equations for conduction and switching losses in a synchronous buck converter will be used [98, 99]:

$$P_{CON,LOSS} = I_{OUT}^2 \cdot R_{DS,ON} \tag{3.1}$$

$$P_{SW,LOSS} = 2 \cdot \frac{1}{2} \cdot C_{OSS} \cdot V_{DS}^2 \cdot f_S \tag{3.2}$$

Here V_{DS} is the voltage between the drain and the source of the MOSFET, C_{OSS} is the parasitic output capacitance of the MOSFET, $R_{DS,ON}$ is the on resistance of the MOSFET and I_{OUT} is the output current of the converter.

The two sets of converter specifications given in Table 3.1 will be used as exam-

	\mathbf{V}_{IN}	\mathbf{V}_{OUT}	\mathbf{P}_{OUT}
$SMPS_1$	$50 \mathrm{V}$	5 V	$5 \mathrm{W}$
$SMPS_2$	330 V	20 V	$50 \mathrm{W}$

Table 3.1: Two sets of converter specifications

ples throughout this chapter. MOSFETs with high on resistance and low output capacitance will be used for both, in order to reduce the switching losses. The on resistance and output capacitance is linked due to the FOM of the material and the device structure, hence one have to be traded off to improve the other. The output capacitance domminates the switching losses, so for fast switching conveters it is essential to keep this low. For the 50 V converter, this gives $R_{DS,ON} = 1.6 \Omega$ and $C_{OSS} = 10 \text{ pF}$ [100] and for the 330 V converter $R_{DS,ON} = 1.2 \Omega$ and $C_{OSS} = 20 \text{ pF}$ [101]. With these values the losses for $SMPS_1$ becomes:

$$P_{CON,LOSS} = 1 \ A^2 \cdot 1.6 \ \Omega = 1.6 \ W \tag{3.3}$$

$$P_{SW,LOSS} = 2 \cdot \frac{1}{2} \cdot 10 \ pF \cdot 50 \ V^2 \cdot 30 \ MHz = 0.7 \ 5W \tag{3.4}$$

and for $SMPS_2$ it becomes:

$$P_{CON,LOSS} = 2.5 \ A^2 \cdot 1.2 \ \Omega = 6.25 \ W \tag{3.5}$$

$$P_{SW,LOSS} = 2 \cdot \frac{1}{2} \cdot 20 \ pF \cdot 330 \ V^2 \cdot 30 \ MHz = 65.3 \ W \tag{3.6}$$

These losses are clearly not acceptable. For the 5 W converter an efficiency of 68% could be achieved if all other components where ideal, but for the 50 W converter the maximum efficiency would be 41%.

Therefore all topologies investigated in this chapter will be so called zero voltage switching (ZVS) topologies. All the topologies are derived from the class E inverter, which utilizes the output capacitance in the design and insures that the capacitance is fully discharged before the MOSFET is turned on.

Some topologies can achieve zero current switching (ZCS) as well. This removes the loss caused by parasitic inductances in for instance the package of the MOSFET. This is generally not a big loss mechanism in power converters [99]. This is however very desirable due to the fact, that it leads the derivative of the voltage to be zero at the switching instance (ZdVS or ZDS). This reduces the impact if the MOSFET is not turned on exactly at the right time, as the voltage across it will be close to zero for an amount of time.

When designing resonant DC/DC converters it is very common to split the converter into two parts; 1) a resonant inverter converting the DC input voltage to an AC output current 2) a resonant rectifier rectifying the AC current to a DC output [102, 103, 104], see Figure 3.1.

It is possible to design the inverter to a load, which give optimal operating conditions for the inverter, and then use different resistance compression networks to make the impedance of the rectifier match [105, 106, 107]. Though this is a solution often used, it will increase the complexity of the converter unnecessarily and



Figure 3.2: Relation between $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ and η for the converters in table 3.2.

possibly reduce the achievable efficiency, size, and price. Therefore the best design procedure is firstly to design a rectifier for the given load and then design the inverter for the given input and rectifier impedance.

The most commonly used inverter is the class E, however several other topologies exist. Some of the research results are summed up in Table 3.2. From Table 3.2 it is seen that very high efficiencies are achievable for the inverters, up to 97%. However the efficiency drops around 10% for the complete DC/DC converters, i.e. when a rectifier is added. By further inspection of Figure 3.2, it can be seen that it is problematic to have a high input voltage and switching frequency while having a low output power and still keep the efficiency high. This is mainly due to the output capacitance, C_{OSS} , and the energy stored herein. The energy stored in C_{OSS} is set by the capacitance and the peak voltage across the MOSFET. The peak voltage varies between V_{IN} for a half bridge [108] and $3.56 \cdot V_{IN}$ for a class E [102], but is for all topologies a function of V_{IN} .

The energy stored in C_{OSS} has to be charged and discharged every switching cycle in order to achieve ZVS. Combining C_{OSS} , V_{IN} and f_S hence gives a measure for the minimum resonating currents needed inside the converter to achieve ZVS. If this current is much larger than what is needed to deliver the desired output power, losses will apear in the ESR of the components in the power stage without giving more power out. Hence the converter will exhibit higher loss without increased output power, causing the efficiency to drop. This relation can be described by eq. 3.7:

$$\frac{V_{IN}^2 \cdot C_{OSS} \cdot f_S}{P_{OUT}} \propto \frac{1}{\eta} \tag{3.7}$$

The next sections will describe and compare the various topologies suitable for operation in the VHF range.

Inverters						
Topology	\mathbf{f}_s	\mathbf{V}_{IN}	\mathbf{V}_{OUT}	\mathbf{P}_{OUT}	η	Year
	[MHz]	[V]	[V]	[W]	[%]	
Class DE	5.3	330	N/A	1154	89	1999 [109]
Class E	1	128	N/A	366	96.6	2006 [110]
Class ϕ_2	1	129	N/A	526	97.1	2006 [110]
Class ϕ_2	30	160	N/A	330	93.7	2007 [111]
Class E	1	129	N/A	322.7	97	2007 [112]
Class E	100	9	N/A	6.8	82	2011 [113]

Table 3.2: Results from previous research

Converters

Topology	\mathbf{f}_s	\mathbf{V}_{IN}	\mathbf{V}_{OUT}	\mathbf{P}_{OUT}	η	Year
	[MHz]	[V]	[V]	[W]	[%]	
Class E	1	20	25	8.9	89	1989 [104]
Class ϕ_2	30	165	33	265	87	2006 [99]
Class E	100	11	12	10	75	2006 [32]
Class ϕ_2	30	150	33	180	84	2008 [103]
Class ϕ_2	10	170	75	250	91	2009 [31]
Class ϕ_2	30	165	33	225	87	2009 [31]
Class ϕ_2	30	330	50	900	79	2009 [114]
Class E	100	12	23.7	1.7	55	2010 [115]
Push-Pull ϕ_2	30	140	65.4	471.9	83.4	2010 [116]

3.1 Single Switch

Single switch inverters and rectifiers are by far the most common choice for VHF or even HF converters. Only one of the examples given in Table 3.2 is not a single switch topology.

The class E inverter is the fundamental ZVS single switch topology and the most commonly used [110, 112]. Several other topologies derived from the class E inverter exist, such as the class $\text{EF}_2(\phi_2)$ [117, 103], the resonant SEPIC [118, 119] or the resonant boost converter [120, 121].

On the rectifier side it is more or less the same situation, with the class E being the most commonly used rectifier [104, 32, 114] and a few other alternatives derived hereof [102, 90, 108].

All the single switch topologies have a switch and an inductor coupled in series across the input or output, for the inverter and the rectifier respectively. In order to keep the volt-second balance across the inductor, the average drain-source voltage (V_{DS}) of the switch has to be equal to the input voltage. Even if V_{DS} is assumed constant when the switch is closed, the peak voltage has to be two times V_{IN} for a duty cycle of 50%. In reality the voltage across the switch is more like a half wave rectified sine wave in order to achieve ZVS, which result in a peak voltage of 3.56 times V_{IN} for the class E inverter [110]. This high voltage stress on the semiconductors is the biggest challenge with single switch topologies, but they are often chosen due to the complexity and losses connected with a high side gate drive for operation in the VHF range.

This section will first introduce the class E inverter and its equivalent rectifier. Then three modifications of this will be covered, namely the class ϕ_2 , the resonant single-ended primary-inductor converter (SEPIC) and the resonant boost.

3.1.1 Class E

The most commonly used resonant inverter is the class E, a schematic of it is shown in Figure 3.3. It consists of a single MOSFET, two inductors, and two capacitors. In optimum operation L_{IN} is an infinite choke providing a pure dc input current. The resonant circuit (L_R and C_R) is inductive at the switching frequency and the inverter is designed to have both ZVS and ZDS.



Figure 3.3: Schematic of the class E inverter.

The inverter can only achieve both ZVS and ZDS switching in very specific situations. According to [122, 123] this can only be achieved if:

$$R_L = \frac{8}{\pi^2 + 4} \cdot \frac{V_{IN}^2}{P_{OUT}}$$
(3.8)

$$f_{S,max} = \frac{P_{OUT}}{2 \cdot \pi \cdot C_S \cdot V_{IN}^2} \tag{3.9}$$

If the values from the two converter examples are put into these equations we get: For the 50 V input and 5 W:

$$R_L = \frac{8}{\pi^2 + 4} \cdot \frac{50 \ V^2}{5 \ W} = 288 \ \Omega$$
$$f_{S,max} = \frac{5 \ W}{2 \cdot \pi \cdot 10 \ pF \cdot 50 \ V^2} = 31.8 \ MHz$$

For the 330 V input and 50 W:

$$R_L = \frac{8}{\pi^2 + 4} \cdot \frac{330 \ V^2}{50 \ W} = 1.26 \ k\Omega$$
$$f_{S,max} = \frac{50 \ W}{2 \cdot \pi \cdot 20 \ pF \cdot 330 \ V^2} = 3.7 \ MHz$$

It is possible to drive the 5 W converter at 31 MHz, i.e., in the VHF range, and achieve both ZVS and ZCS, but it requires a load impedance of 288 Ω . The

impedance of the load in this example is 25 Ω , but impedance matching networks can be used to adjust the impedance seen by the inverter [105, 106, 107]. These matching networks do however add to the complexity of the circuit and there will be losses associated with these components as well. Hence it might be more desirable to sacrifice the ZCS in order to keep the number of components low.

For the 50 W converter the maximum operating frequency is 3.7 MHz, high compared to commercial products, but way below the VHF range. The impedance of the load is 8 Ω , but the load impedance of the inverter should be 1.26 k Ω . Again impedance matching networks could be used to compensate for some of the misalignment, but the desired specifications are clearly too far from the specifications needed to achieve both ZVS and ZCS at VHF.

By sacrificing the ZCS for more design flexibility the control of the inverter becomes more important. The inverter will be running in a subnominal condition, as described further in [124, 113], and the derivative of voltage across the MOSFET will not be zero as it crosses zero volt. The result is that there will be one very specific point in time were ZVS can be achieved, if the MOSFET is turned on just a little earlier or later the voltage will have changed. The amount it has changed depends on how far the design is from the optimum case, i.e., how much current is running through the switch when the voltage crosses zero. If the MOSFET is turned on to early, the capacitor will not be fully discharged and switching losses will occur. If the MOSFET, on the other hand, is turned on too late, the body diode will start to conduct and prevent the voltage from dropping significantly below zero. Body diodes are generally lossy and this is hence not desirable [125].

Not designing for the optimal operation point of the class E inverter also makes it possible to change the duty cycle and remove the choke inductor at the input, hence increasing both power density and response. Doing so the standard equation found in RF amplifier text books [122, 123] cannot be used and a new set of design equations has to be derived.

If the drain source voltage of the MOSFET is assumed to be a half sine wave when it is off and zero when it is on, the peak voltage across the MOSFET will be:

$$V_{IN} = \int V_{DS} = V_{DS,peak} \frac{2 \cdot (1 - D)}{\pi}$$

$$(3.10)$$

$$V_{DS,peak} = V_{IN} \frac{\pi}{2 \cdot (1-D)}$$
(3.11)

The rms value of a half wave rectified sine wave is:

$$V_{DS,rms} = V_{DS,peak} \sqrt{\frac{D}{2}} \tag{3.12}$$

And the rms value of the output voltage is:

$$V_{OUT,rms} = \sqrt{P_{OUT} \cdot R_L} \tag{3.13}$$

According to [99] the reactance of the resonance circuit can now be determined by:

$$X_{RC} = R_L \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1}$$
(3.14)

By combining equation 3.11, 3.12, 3.13, and 3.14, an expression for the needed reactance as function of input voltage, duty cycle, output power, and load is obtained:

$$X_{RC} = R_L \cdot \sqrt{\frac{V_{IN}^2 \cdot \pi^2 \cdot D}{2 \cdot (2 \cdot D - 2)^2 \cdot P_{OUT} \cdot R_L} - 1}$$
(3.15)

It can be desirable to keep the duty cycle low in order to reduce the peak voltage across the MOSFET, however due to turn on and off times and delays it is desirable to keep it in the range 30-50%.

The values of the inductor and capacitor for the resonance tank can be calculated based on the reactance derived in equation 3.15 and the desired loaded Q factor of the resonant tank:

$$Q_R = \frac{1}{R_L} \sqrt{\frac{L_R}{C_R}} \tag{3.16}$$

$$X_{RC} = \omega_S \cdot L_R - \frac{1}{\omega_S \cdot C_R} \tag{3.17}$$

From these two equations expressions for L_R and C_R can be derived:

$$C_{R} = \frac{X_{RC} + \sqrt{4 \cdot Q_{R}^{2} \cdot R_{L}^{2} + X_{RC}^{2}}}{2 \cdot Q_{R}^{2} \cdot R_{L}^{2} \cdot \omega_{S}}$$
(3.18)

$$L_R = \frac{2 \cdot Q_R^2 \cdot R_L^2 + X_{RC} \cdot \sqrt{4 \cdot Q_R^2 \cdot R_L^2 + X_{RC}^2} + X_{RC}^2}{(X_{RC} + \sqrt{4 \cdot Q_R^2 \cdot R_L^2 + X_{RC}^2}) \cdot \omega_S}$$
(3.19)

In class E inverters for RF amplifiers, it is desirable to keep the loaded Q of the resonant tank high (>10) in order to insure a pure sinusoid at the output [126, 127]. For power conversion where a DC output is the target, this is however not necessary. In fact, there are a few good reasons not to have a high Q resonant tank. The resonant tank is inductive at the switching frequency in order to achieve ZVS, hence a high Q would require a large inductor, see equation 3.16. This is both the largest component in the circuit; it slows down the transient response and introduces a larger series resistance. It is generally good to keep the Q between 0.3 and 2. As seen in Figure 3.4 the size of the inductor increases rapidly beyond this range and the size of the capacitor start to increase significantly below this range.

Another important parameter when selecting the components for the resonant tank is component tolerances. Most components come with a given tolerance, which can be reduced for a premium on the price. Ceramic capacitors with 5% tolerances [128] and air core inductors with 2% tolerances [129] can generally be bought at fair prices. As the output power of the inverter is set directly by the reactance of the resonant tank, see equation 3.15, these tolerances will directly impact the output power. Further the reactance of the resonance tank contributes to the resonance of the entire converter and thereby to achieving the crucial ZVS. Tolerances in the components will therefore affect the frequency at which ZVS can be achieved. As the resonant tank is inductive at the switching frequency, the inductor dominates the reactance of the tank and the tolerance of the inductor is therefore the most important.



Figure 3.4: Normalized values of L_R and C_R as function of loaded Q factor.



Figure 3.5: Deviation from the decired reactance depending on loaded Q factor.

Figure 3.5 shows the deviation in the reactance of the resonant tank that can appear with the mentioned tolerances depending on the selected Q. For a Q below 1 the deviation is directly dependent on the inductor, but for Q values above 1 the tolerances of the capacitor starts to add to the deviation. The plot in Figure 3.5

just shows the tolerances on the capacitors when the DC voltage across is zero. In the converter the capacitor in the resonant tank will be the part that blocks a DC current from running between the input and output, hence the capacitor will have a DC bias equal to the difference between the input and output. This DC bias can cause up to 50% [130] deviation form the rated value. For this reason it can be desirable to keep the Q below 1 if the converter is designed to have a big step ratio.

The two remaining components to be calculated are L_{IN} and C_S . From equation 3.9 it was seen that C_{OSS} will often set the upper limit for the switching frequency. As explained it is possible to design around this, but at the expense of increased switching currents. Therefore C_S will be constituted solely by the parasitic capacitance of the MOSFET in VHF converter operating with high input voltages and low power levels (compared to what is possible according to equation 3.9 and 3.8).

For class E inverters used for RF amplifiers the input inductor is normally a choke, but this limits the transient response and will be a bulky component. Further the inductor needs to be small enough to resonate with the resonant tank and C_S in order to achieve ZVS. The frequency at which these components have to resonate is equal to two times the period where the switch is open, i.e.:

$$T_R = 2 \cdot (1 - D) \cdot T_S \Leftrightarrow f_R = \frac{f_S}{2 \cdot (1 - D)}$$
(3.20)

In order to calculate the value of L_{IN} , it is necessary to calculate the effective value of C_S . The output capacitance of the MOSFET is only contributing to the resonance in the part of the period where the MOSFET is off, hence it has to be scaled by 1-D in order to find the effective capacitance. The effective capacitance of the output capacitor is $C_{S,eff} = \frac{C_S}{1-D}$. Knowing the values of X_{RC} , the input inductance can be calculated according to:

$$L_{IN} = \frac{1}{\omega_R^2 \cdot C_{S,eff} - \frac{\omega_R}{X_{RC}}}$$
(3.21)

The design equations for all the components in the class E inverter have now been given. In order to make a complete VHF converter a rectifier is needed. The class E rectifier shown in fig. 3.6 is equivalent to the inverter in many ways. The waveforms of the two circuits are shown and described in A.3.



Figure 3.6: Schematic of the class E rectifier.

The rectifier is designed to appear resistive at the switching frequency, by insuring that L_R and C_R resonate at this frequency. This simplify the design of the inverter as the design equations are for a resistive load.

The duty cycle of the diode can be adjusted by these components. A long duty cycle will reduce the peak currents but increase the peak voltage across the diode and vice versa for a short duty cycle. With a diode duty cycle, D_D , of 50% the value of C_R should be [102]:

$$C_R = \frac{1}{2 \cdot \pi^2 \cdot f_S \cdot R_L} \tag{3.22}$$

With this capacitance the value of L_R can be calculated according to [102]:

$$L_R = \frac{1}{(2 \cdot \pi \cdot f_S)^2 \cdot C_R} \tag{3.23}$$

The complete design of a class E inverter with a class E rectifier is covered in A.4. The paper describes the design of a converter with 50 V input and an output of 5 V and 1 W. This places this design far from any other designs that had been made at that point, see Figure 3.7. Further this inverter was designed with a standard silicon MOSFET and achieved efficiencies up to 82.9%.

The low output voltage compared to the forward voltage drop of the diode, gave rise to a significant loss in the rectifier diode. Therefore synchronous rectification were investigated in A.11 and it was found that synchronous rectification of VHF converters can indeed give a significant efficiency boost for applications were the output voltage is low (<5 V) and the current high (>2 A).



Figure 3.7: The $\frac{P_{OUT}}{V_{IN} \cdot f_S}$ -factor and η achieved in A.4 next to previous results.

3.1.2 Class ϕ_2

As written in section 3.1, the large voltage peak across the MOSFET is the major problem for single switch inverters in application with high input voltage.

Voltage stress in these converters can be reduced by employing multi-resonant networks, either transmission line [131, 132, 133] or discrete with limited number of harmonics [111, 134, 103]. By implementing these techniques, switch voltage stress in VHF converters can be reduced to 2-2.5 times the input voltage for the same duty cycle, but this technique requires extra resonant elements. The class ϕ_2 (or EF_2) inverter, which is a hybrid between the class E and F_2 inverters, was developed in order to make the voltage across the MOSFET closer to a square wave. The voltage across the MOSFET should thereby become significantly smaller (ideally $2 \cdot V_{IN}$ for D = 50%). This is done by inserting an LC circuit in parallel with the MOSFET as shown in Figure 3.8. This circuit is designed to have a resonance frequency at the second harmonic, which causes the voltage across the MOSFET to become a trapezoidal wave consisting of the 1st and 3rd harmonic. The same benefits can be achieved with the flat-top class-E amplifier described in [135].



Figure 3.8: Schematic of the class ϕ_2 inverter.

According to [103] the rms voltage across the MOSFET can be estimated by $V_{DS,rms} = V_{IN} \frac{4}{\pi \cdot \sqrt{2}}$, thus the needed reactance of the resonant circuit is different (see equation 3.14). No exact equations for the calculations of the added LC circuit or the input inductance are given in literature, however the following gives results which are close [111]:

$$L_{IN} = \frac{1}{9 \cdot \pi^2 \cdot f_S^2 \cdot C_S} \tag{3.24}$$

$$L_{MR} = \frac{1}{15 \cdot \pi^2 \cdot f_S^2 \cdot C_S}$$
(3.25)

$$C_{MR} = \frac{15}{16} \cdot C_S \tag{3.26}$$

This topology was investigated in A.4, but it was found to have higher losses and a larger component count than the class E inverter due to the introduced third harmonics. The voltage across the MOSFET is indeed lowered, which can be an advantage in applications were a small reduction in required break down voltage makes new semiconductors available. The resonant currents at the switching frequency are however the same as for the class E, but with added currents at the third harmonics. This makes it even more difficult to achieve high efficiency at low power levels, due to the relation described in equation 3.7.

3.1.3 Single-Ended Primary-Inductor Converter

The Single-Ended Primary-Inductor Converter (SEPIC) has several semilarities to the class E inverter with a class E rectifier. As shown in Figure 3.9 the only difference is that the inductor in the resonant tank is removed. As explained in section 3.1.1 this inductor is however quite important and used both to regulate the output power and to insure that ZVS is achieved.



Figure 3.9: Schematic of the SEPIC converter.

Hence the design and behaviour of the SEPIC is different than that of the class E. No fixed equations are available for this topology, but one design methodology is given in [118]. Here the converter is designed by firstly designing the rectifier part as a normal class E rectifier, were the rectifier is made to apear resistive at the switching frequency. However for the SEPIC described in A.5, the rectifier is designed to apear slightly inductive at the resonance frequency, as this was found to give higher efficiency. Once the rectifier part is designed the capacitor, C_T , is set to get the desired output power and finally the input inductor, L_{IN} , is set to insure ZVS.

The design process of the SEPIC is more complicated than for the class E, due to the lack of equations to calculate the component values. However once the converter is designed it achieves better total specifications than seen for the class E. It has one inductor less than the class E and both of the two remaining inductors are smaller. This does not only increase the power density and reduce the bill of material, it also enables higher efficiency as the resonating current runs through less series resistance.

3.1.4 Resonant Boost

The resonant boost converter shown in Figure 3.10 is another class E derived topology suited for operation at VHF. During the design the converter is as for the other topologies split into a inverter and a rectifier; the inverter is constituted by the input inductor, L_{IN} , the MOSFET and the capacitor, C_S , and the rectifier is made up by the inductor, L_R , the capacitor, C_R , and the diode [136]. In practical designs the two capacitors may be constituted by the parasitic capacitances of the MOSFET and diode, respectively. Hence this converter can be made with only four components, a MOSFET, a diode and two inductors.



Figure 3.10: Schematic of the resonant boost converter.

The converter has a DC path from input to output, which makes it useless for buck

(step down) applications, but it is quite attractive for boost applications due to the low component count. As the focus of this thesis has been on buck applications this topology has not been tested, but good results have been shown with efficiencies of up to 87% [137, 138, 139].

3.2 Half Bridge

The research in VHF converters has been very focused on the single switch topologies. This is due to the fact that no high side gate drives have been available for operation above a few volts. A 200 MHz integrated buck converter with a high side gate drive was presented in [140], but it only had a 3.6 V input and peak efficiency of 77%. If a high side gate drive can be designed for operation at VHF and at higher voltages, it opens up a new range of topologies, which all deal with the major drawbacks of the single switch topologies, namely high voltage stress and large resonating currents.

Some of the topologies most suited for operation at VHF are investigated further in the following subsections.

3.2.1 Class DE

The class DE inverter is a merger of the class D half bridge converter and the class E zero voltage switching inverter. Hence this topology offers the low semiconductor stress of the half bridge with zero voltage switching. In the class DE inverter the switches are directly connected to the input and the voltage across them is therefore limited to the input voltage (see Figure 3.11). The class DE inverter has two other great advantages over the single switch topologies, it has only a single inductor and due to the lower peak voltage across the MOSFET, the stored energy is more than ten times lower than for the class E (E = $\frac{1}{2} \cdot C_{OSS} \cdot V_{IN}^2$ compared to E = $\frac{1}{2} \cdot C_{OSS} \cdot (3.56 \cdot V_{IN})^2$), leading to much smaller resonating currents in a buck type converter, see A.3.



Figure 3.11: Schematic of the class DE inverter.

The component values can be derived with the same approach as used for the other topologies. If the voltages on both sides of the resonant circuit $(C_T \text{ and } L_T)$

are assumed to be trapezoidal, the rms voltages in those nodes can be calculated according to equation 3.27 and 3.28.

$$V_{DS,rms} = V_{IN} \cdot \sqrt{\frac{D_M + 1}{3}}$$
 (3.27)

$$V_{REC,rms} = V_{OUT} \cdot \sqrt{\frac{D_D + 1}{3}} \tag{3.28}$$

Where D_M and D_D is the duty cycle of the MOSFET and the diode, respectively. The reactance of the resonance circuit is calculated according to equation 3.29 [99].

$$X_{RC} = R_{REC} \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1}$$
(3.29)

Once the reactance of the resonant circuit is calculated, a suitable capacitance can be chosen and the inductance calculated. The capacitors C_{S1} and C_{S2} need to resonate with the resonant circuit at a frequency given by:

$$\omega_R = \frac{\omega}{2 \cdot (1/2 - D_M)} \tag{3.30}$$

Hence:

$$\omega_R^2 = \frac{1}{2 \cdot C_S \cdot \frac{X_{RC}}{\omega_R}} \tag{3.31}$$

Combining equation 3.30 and 3.31 leads to the following expression for the value of C_{S1} and C_{S2} :

$$C_S = \frac{1/2 - D_M}{\omega \cdot X_{RC}} \tag{3.32}$$



Figure 3.12: Schematic of the class DE rectifier.

For high output voltages, a class DE rectifier [141] also offers some advantages over the class E rectifier. As for the inverter the diodes are connected directly to the output and the voltage across them is hence limited to the output voltage (see Figure 3.12). It is a challenge to find schottky diodes suited for operation at VHF with break down voltages above 60-80 V, which limits the class E rectifier to output voltages around 20 V ($\approx \frac{60-80V}{3.56}$). With a DE rectifier output voltages up to 60 V can be achieved with good design margin. This comes at the expense of twice the forward conduction loss, so this rectifier is not suited for low voltages. Furthermore the class DE rectifier is inductor less and hence offers a much higher power density, as the inductor is generally the biggest component.

3.2.2 LLC Converter

The LLC converter were presented in section 2.1.1 and is one of, if not the most, used topologies for resonant converters switching up to a few MHz. It is less load dependent than the purely series loaded resonant topologies, which the class DE with a class E or DE rectifier would be. At the same time it keeps the high efficiency, in contrast to the purely parallel loaded topologies. This topology is therefore very likely to be well suited for operation in the VHF range as well.

The first VHF half bridge was however demonstrated in A.6 and no other have been presented so far. Hence no experimental results are available for a LLC converter operating at VHF. With the results from A.6, this is however a very interesting next step.

Another more or less equivalent option is the LCC, which would be equivalent to add more capacitance across the rectifier diodes, so in some way it is already there. However a converter with more capacitance, or with variable capacitance, need to be build to test the load dependence and efficiency variation to tell if this is an efficient solution.

3.3 Stacking

Another approach to reduce the voltage stress across the switches is stacking cells, i.e. coupling multiple small converters with serial input [119, 142]. This reduces the switch voltage stress by N compared to a single cell case, where N is the number of converter cells. This is a normal approach for low frequency converters when a large step down ratio is required [143]. By arranging several cells with serial input and parallel output (SIPO), a large overall step ratio can be achieved while keeping the stress on the individual cells low [114].

For resonant converters this does not only reduce the peak voltage across the MOS-FET, it also makes it possible to get closer to the optimum point of operation. In section 3.1.1 it was found that the maximum operating frequency for the inverter with high input voltage, $SMPS_1$, was 3.7 MHz and the impedance was more than 100 times to high. If the converter where split into seven smaller cells in a SIPO configuration, it would be possible to use the 50 V MOSFET and run the inverter in the optimum point at more than 45 MHz. Further the load impedance required would be divided by 7, thus it would still be to high, but making the final adjustments with impedance matching circuits would be much easier.

Stacking has two main drawbacks, the component count increases significantly and control becomes much more complicated since it requires balancing of the voltages across the cells.

3.3.1 Input-Output Rearrangement

A way to achieve the benefits of stacking without the drawbacks, is to rearrange the input and output as proposed in A.8 and A.9. The converter needs to be isolated to apply this methodology and the isolation will be lost, but a technical isolation can easily be achieved with small ceramic capacitors in VHF converters.

Figure 3.13 shows the conventional input-output configuration of an isolated DC/DC converter. If two isolated converters have their inputs connected in parallel and outputs connected in series, we obtain the structure in Figure 3.14 (top). Similarly, if the inputs are connected in series and outputs in parallel, the structure in Figure 3.15 (top) is obtained.

If one of the converters in each configuration has a 1:1 voltage transformation ratio, it is possible to connect its input to its output directly and remove the 1:1 converter from the circuit, without affecting the rest of the system. In doing so, part of the delivered power flows directly from the input to the output, and is not processed by the converter. As a consequence, higher converter efficiency is achieved. Moreover, compared to the single cell design, voltage and/or current stresses on the switches are reduced, just as if stacking were applied but without the increased component count and complexity. The only difference is that the input is not galvanic isolated from the load. Figure 3.14 and 3.15 (bottom) demonstrate the final connection rearrangement to obtain step-up and step-down configuration, respectively.

To quantify the benefits from the proposed configurations, the case where input voltage V_{in} and output voltage V_{out} are held constant across all configurations is analyzed. Moreover, output current I_{out} and output power P_{out} are held constant as well. A distinction is made between the power processed by the converter P_C , and P_{out} .

The voltage transformation ratio of the conventional converter is

$$M = \frac{V_{out}}{V_{in}}.$$
(3.33)

In step-up and step-down configurations, transformation ratio of the converter cell is reduced to

$$M_{up} = \frac{V_{out} - V_{in}}{V_{in}} \tag{3.34}$$



Figure 3.13: Conventional isolated dc/dc converter topology. Dashed line represents galvanic isolation.



Figure 3.14: Step-up configuration: two converter cells with inputs in parallel and outputs in series (top) and a single cell equivalent (bottom).



Figure 3.15: Step-down configuration: two converter cells with inputs in series and outputs in parallel (top) and a single cell equivalent (bottom).

and

$$M_{down} = \frac{V_{out}}{V_{in} - V_{out}}.$$
(3.35)

In general, switch voltage stress is a function of both input and output voltages. In step-up configuration, output voltage stress contribution is reduced by a factor of $(1 - V_{in}/V_{out})$. In step-down configuration, input voltage stress contribution is reduced by $(1 - V_{out}/V_{in})$.

Output power of all three configurations is the same:

$$P_{out} = V_{out} I_{out} \tag{3.36}$$

The power processed by the converter P_C in the conventional structure is equal to P_{out} . In the step-up and step-down cases, this power is given as:

$$P_{C,up} = P_{out} \left(1 - \frac{V_{in}}{V_{out}} \right)$$
(3.37)

$$P_{C,down} = P_{out} \left(1 - \frac{V_{out}}{V_{in}} \right)$$
(3.38)

 $P_{C,up}$ and $P_{C,down}$ are smaller than P_{out} , and the remaining power is delivered through the DC path. Efficiency of that power transfer is very close to 100%. Assuming that P_C , $P_{C,up}$, and $P_{C,down}$ are transferred with the same efficiency η , effective efficiencies of the step-up and step-down configurations can be expressed in terms of η , P_{out} , and $P_{C,up/down}$:

$$\eta_{up/down} = \eta \; \frac{P_{C,up/down}}{P_{out}} + \frac{P_{out} - P_{C,up/down}}{P_{out}} \tag{3.39}$$

Equations 3.37-3.39 are illustrated in Figure 3.16. The plot clearly shows that for limited step-up or step-down ratios, the percentage of the output power processed by the converter is small and hence the efficiency improvement becomes significant. For a step-up converter with 330V input and 400V output, the converter would process only 17.5% of the output power. If the converter efficiency were 80%, the rearranged efficiency would be 96.5%. The converter would in that case have to be designed as a technically (i.e. not according to safety standards) galvanic isolated converter with 330V input, 70V output and a peak output power equal to 17.5% of the required power. Hence a low power flyback or SEPIC converter with moderate efficiency could replace a high power boost converter with high efficiency, while achieving the same overall system performance.

Adaptations shown in Figure 3.14 and 3.15 can be used for any type of isolated converter, regardless of the isolation type (inductive or capacitive). In A.7, a flyback converter is used in a step-up configuration and a class DE converter (DE inverter and DE rectifier) in step-down configuration.

The proposed technique may be used together with stacking as shown in Figure 3.17. That will introduce the drawbacks and advantages of stacking, but the drawbacks will still be reduced as one cell less might be needed.



Figure 3.16: Power processed by the converter and efficiency improvement by rearranging.



Figure 3.17: Two stacked converters in step-down configuration.



Figure 3.18: Measured efficiencies on two rearanged isolated converters.

The presented technique can be used to reduce voltage and/or current stress for isolated dc-dc converters, in applications where isolation is not a requirement. The technique provides higher efficiency compared to a conventional single cell design, and it can be combined with other methods of voltage stress reduction, as long as the initial converter provides capacitive or inductive isolation. The obtained benefits are very appealing for non-isolated applications. The experimental results shown in A.7 are in good agreement with the presented theory, see Figure 3.18. Twice the power handling capabilities and 5-10% higher efficiencies are shown for the tested converters.

3.4 Summary

Several topologies have been analyzed throughout this chapter. Experimental prototypes of the most suited have been presented in A.5, A.4 and A.6, the class E, SEPIC and class DE respectively. All the inverters had some pros and cons, thus the same inverter will not be best for all applications. The most important pros and cons of the class E, class ϕ_2 and class DE are listed in Table 3.3.

	Class E	Class ϕ_2	Class DE
Pros	Low side switchEasy tuningWell documented	 Low side switch Reduced stress	One inductorLow lossLow stress
Cons	 Large stress Large inductors	 Largest loss Complex	\bullet High side switch

Table 3.3: Pros and cons of the investigated inverter topologies.

The class E inverter is by far the least complex of the topologies. It is well described in textbooks and straightforward design process are available where the individual components do not severely affect each other. The inverter consists of only one MOSFET, two inductors, and a capacitor (if C_S is composed by the output capacitance of the MOSFET). The voltage stress is the main drawback of this topology for applications with high input voltage, but it is very well suited for applications with low input voltage. If designed to operate in the optimum situation the inductors are the largest for any of the topologies, this limits the transient response and the power density that can be achieved. The inverter can however be designed to operate in a subnominal situation with smaller inductors and faster transient response. The design process for this is not described as thoroughly as the optimal operation, but several publications with different descriptions are available [124, 144, 145].

The SEPIC converter can be seen as a slightly modified version of a class E inverter with a class E rectifier, the only difference in the schematic is that the inductor in the resonant tank is removed. This does not only reduce the number of inductors, but the two remaining inductors will also be smaller than those seen for the class E (if it is designed to operate close to the optimum). The design of the SEPIC is however more complex as the inverter and rectifier cannot be designed separately and all components thereby influence each other. Hence better performance in terms of efficiency, transient response, size and cost can be achieved with the SEPIC, but the design is more complex.

The class ϕ_2 inverter is also a modified version of the class E, the only difference being the added LC circuit put in to reduce the voltage across the MOSFET. While this is a good way to reduce the voltage stress, the steep voltage curves require larger currents, making the loss larger than seen for the class E inverter. Though it has 2 extra components, compared to the class E inverter, the physical size can be more or less the same as the inductors are smaller. The total loss is larger than for the class E inverter due to the higher resonant currents. This might be acceptable if it makes it possible to chose from another class of MOSFET, e.g. 100 V devices instead of 150 V devices, but if that is not the case the class E or SEPIC are better choices.

The class DE inverter is the only half bridge inverter that has been designed and tested experimentally. This is the first VHF half bridge converter operating at more than a few volts presented and thereby one of the major breakthroughs in the work presented.

As C_{S1} and C_{S2} can be composed by the parasitic capacitance of the MOSFETs, the total component count for the class DE inverter is the same as for the class E inverter. The peak voltage across the MOSFETs is by far the lowest seen in any of the inverters and the currents are also the lowest. For class E inverters the nonlinear output capacitance of the MOSFET can also be a big challenge, as it changes the peak voltage across the MOSFET [146, 147, 148], this problem is eliminated with the class DE due to the clamping across the input voltage. Further this topology only has one inductor, which at the same time is smaller than the once in any of the other circuits. The half bridge solution is therefore superior to all the single switch topologies if an efficient high side gate drive can be designed.

The class ϕ_2 inverter was the single switch inverter with the lowest voltage stress.

The voltage stress is approximately $2.5 \cdot V_{IN}$ which is still 2.5 times more than for the half bridge. This result in more than 6 times more energy stored in the output capacitance of the MOSFET, this is the minimum energy that needs to resonate in order to get ZVS. For low power applications with high input voltages this therefore sets the amount of resonant currents. Further the class ϕ_2 specifically has even more resonating current due to the 3rd harmonic introduced to reduce the peak voltage.

From this analysis the class DE inverter is by far the best solution and the SEPIC comes in second. However during this analysis the gate drive has not been considered. This will be done in the next chapter.

The development of each of the prototypes are described thoroughly in the appendices, but some common and important challenges were experienced. First of all it has been a big challenge to find MOSFET suitable for operation in these converters. Not because Si MOSFETs cannot be used for operation in VHF SMPSs, but because they are optimized for hard switched converters. In low frequency hard switching SMPSs the main optimization criterias for MOSFETs are on resistance and gate charge. These parameters are also important in VHF SMPSs, but the main design parameters are output capacitance and gate resistance, secondly on resistance and series resistance in the output capacitance. The series resistance of the output capacitance is normally not specified in the datasheet, but in VHF application it causes losses similar to those caused by the on resistance.

Further inductors has become a problem, due to the high resonating current caused by the high output capacitance in the MOSFETs available. With large AC currents the AC resistance or Q factor of the inductors and capacitors becomes very important. Ceramic capacitors are almost lossless due to Q factors as high as 1000 [149, 150], but inductor are hard to find with Q factors above 100-150 [129]. This causes the inductors to be one of the most lossy components in the converters.

Measurements has also been a great challenge due to the low impedance of even a small capacitor at these frequencies. Measuring with a normal probe introduces 10 pF capacitance from the measurement node to ground, this can cause the converter to fail almost instantly. If the probe is added to measure drain-source voltage, the 10 pF ruins the ZVS and causes switching losses so severe that the MOSFET fails due to overheating. Capacitive voltage dividers with small capacitances (1 pF) have been used to get some waveform measurements. The most efficient way have however been to use a thermal camera to measure the hot spots and combine that with simulations to optimize the circuits.

Finally heat and thermal management have been a big challenge. When the power density is increased as dramatically as done in this work, it causes a large power loss in a small area. Even with the same or slightly higher efficiencies than traditional converters, this causes hot spots which has to be dealt with to insure reliable operation. Ceramic and aluminium PCBs have been tested to reduce the hot spots, but the ceramic do not improve the thermal proporties enough and the aluminium causes to high parasitic capacitances. Ceramic thermal bridges placed close to the components causing the hot spots and transfering the heat to large copper areas on the PCB, has been the most efficient way to reduce hot spots so far. 3.4. Summary

CHAPTER 4

Gate Drive

One of the most challenging parts in a reliable and efficient design of a VHF converter is the gate drive. This is especially the case if high side switches or synchronous rectifiers are used, but even the gate drive for a low side switch can be quite challenging. There are two important choices which need to be made early in the design process, should the gate drive be resonant or hard gating and should it be externally driven or self-oscillating.

The first choice is mainly a matter of power level and frequency. The losses associated with driving a MOSFET with a square wave and a sine wave is given in [32]:

$$P_{Gate,Square} = C_{ISS} \cdot V_G^2 \cdot f_S \tag{4.1}$$

$$P_{Gate,Sine} = 2 \cdot \pi^2 \cdot f_S^2 \cdot C_{ISS}^2 \cdot R_G \cdot V_{G,ac}^2 \tag{4.2}$$

In the same paper a comparison is made between the two drive schemes in a 100 MHz converter. The square wave results in gating losses of 1.17 W and the sine wave in 173 mW. The sine wave is the most efficient with more than 6 times lower losses, but from equation 4.2 it is seen that the loss increase with the frequency squared whereas the square wave increase linearly with the frequency. Hence the square wave will become more efficient at some point. Depending on the wave form of the current through the MOSFET, the square wave might also result in lower conduction losses in the MOSFET as it is fully on for the entire conduction period [151]. For the sine wave the MOSFET is turned on slowly as the sine wave rises, which results in higher conduction losses in the beginning and end of the conduction period [152].

For a low to medium power converter switching in the VHF range, the choice is however relatively simple as hard gating leads to gating losses which are unacceptably high, at least for the semiconductors available today and for low to medium power levels (0-200 W) [32, 153]. Resonant gating is therefore also the most used way to drive VHF converters [154, 155, 156] and A.3. Hard gating will hence not be considered further in this thesis and the two following sections will describe externally driven and self-oscillating resonant gate drives.

4.1 Externally Driven

Several ways of designing externally driven resonant gate drives can be found in the literature [157, 158]. The most basic example is the gate drive shown in Figure 4.1 [140].



Figure 4.1: Examples of a resonant gate drive [140].

The MOSFET in the power stage, M_{Power} , is turned on by switching on M1 allowing C_{Gate} to be charged through the inductor, L. Once the gate has been charged to the desired voltage M1 is turned off and M3 turned on and the energy stored in the inductor is sent back to the supply through M3 and the body diode of M2. To turn of M_{Power} , M2 is switch on and the energy stored in C_{Gate} is transferred to the inductor. When the capacitor is discharged M2 is turned of and the inductor is discharged through M1 and M4.

Another common way to drive VHF converters, is to use a small inverter to drive the MOSFET, or even a small hard switched inverter to drive a resonant inverter which then drives the main switch as in [103]. This is a quite complex method with many components, but for high power application it can be the most efficient way of driving a large main switch. For the low to medium (0-200 W) power applications of interest in this thesis this would however be an overkill.

A very efficient way to drive the MOSFET is with a single multi resonant circuit [159, 118]. These circuits utilize the same method as shown in [160] and in the class EF_2 inverter explained in the previous section. In this way the gate drive can be made fully resonant, but achieve a trapezoidal waveform instead of a sine wave at the gate. In this way the power loss in the gate resistance is minimized as the applied current during the falling and rising edges of the gate voltage is constant [32].

Many other options for driving VHF converters with and without external signal exist [99, 161]. However all of these options add several components to the circuit, require an auxiliary supply and require close control between the timings of external signal and the resonances of the power stage. Self-oscillating circuits do not need this and are therefore considered more suitable for low to medium power application.

4.2 Self-Oscillating

Self-oscillating converters are very attractive for low to mid power applications were size and cost are main concerns. One example of a self-oscillating converter is described in [162, 115] and the schematic of the inverter is shown in Figure 4.2. In this way a complete converter can be designed with very few components besides the components in the actual power stage. This minimizes the cost and size and removes the losses, which would normally exist in the auxiliary circuits and the supply for this. Further the power stage and gate drive are linked closely together and a change in the power stage due to component tolerances will hence also affect the gate drive, making it less sensitive to component tolerances than an externally driven.



Figure 4.2: Example of a self-oscillating VHF inverter [115].

For externally driven inverters a small inverter is some times used as a resonant gate drive for a larger inverter [103]. In the same way [155] utilizes one inverter to drive another, but the second inverter is also used to drive the first inverter. In this way two equal inverters are used to drive each other and an interleaved operation and power sharing of the two inverters is achieved. This is particularly interesting for applications, where it is neccesary to parallel multiple converters to handle the power.

The gate drive presented in [163] and A.12 was invented just prior to the start of this thesis. It has hence been investigated thoroughly in the beginning of the project and been found to be superior for applications were size and cost are the main concerns. The new resonant gate drive is solely constituted by passive components around the main semiconductor. This leads to a robust and very simple and low cost gate drive.



Figure 4.3: Schematic of the basic self-oscillating gate drive. The gate resistance and the body diode has been left out for simplicity.

The simplest implementation of the gate drive is shown in Figure 4.3, the only component added is an inductor at the gate of the semiconductor and a bias voltage. If V_D is seen as input and V_G as output, the inductor and the parasitic capacitances composes a high pass filter with a capacitive load.

If this high pass filter is designed to have a gain, $G = V_{GS,pp}/V_{DS,pp}$, and has a phase shift close to 180° at the resonance frequency of the power stage, it will create a sine wave at the gate with a peak to peak voltage swing of $V_{GS,pp}$ and a DC offset equal to V_{Bias} .

This DC offset can be used to control the switching frequency and the duty cycle of the power stage and thereby to regulate the output power. In some situations the parasitic capacitances of the MOSFET will lead to too high or too low gain at the desired frequency and additional capacitors, C_{GDext} and C_{GSext} , can be added to adjust this gain.

Most resonant circuits behave as voltage controlled current sources in open-loop situations; hence the output power increases with the input voltage. As the amplitude of the gate signal is a fixed ratio of the input voltage, a change in input voltage will lead to a change in amplitude of the gate signal. Hence the gate signal will be small at low input voltages, giving low gating losses at low power levels, and be large at high power, levels leading to lower conduction losses in the MOSFET.

In order to improve the turn on speed of the MOSFET, and thereby lower the drain to source resistance, higher order harmonics can be added to the fundamental sine wave leading to a more trapezoidal gate signal. This can be achieved by adding small LC circuits between the gate and drain or source of the MOSFET (see Figure 4.4). LC circuits connected to the drain will cause the higher harmonics to be in phase with V_{DS} and LCs circuit connected to the source will cause the harmonics to be out of phase with V_{DS} (see Figure 4.5).

The number of harmonics to include in a given design will depend on several parameters as price, complexity, efficiency etc. Adding higher order harmonics will in general increase the performance of the converter, but it is important to consider which harmonics to include and the magnitude of those harmonics compared to the fundamental. Figure 4.6 shows the fundamental and the 3rd and 5th harmonics in and out of phase with the drain signal. It is clear that it is desirable to have the fundamental out of phase with the drain signal, but for the 3rd and 5th harmonic it depends on the duty cycle and the current waveform. From the figure it can be seen that it is desirable to have the 3rd harmonic out of phase for a duty cycle of 50%, but for a duty cycle of 25% it has to be in phase and will only add to the center part of the conducting period where the current usually is the smallest [102, 164]. The gate signals that can be achieved by adding harmonics are shown in Figure 4.7.

As the resonant gate drive is only relying on the resonance of the power stage and is floating between the drain and source of the MOSFET, it can be used in all resonant circuits. It can for instance be used in the class E inverter [110, 112], a class EF₂ (ϕ_2) inverter [117, 103], a resonant SEPIC [118, 119], a resonant boost converter [121, 120] or a class DE inverter (including high side gate drive) [165, 108], but it can also be used for synchronous rectification and bidirectional power flow.



Figure 4.4: Schematic of the self-oscillating gate drive with the 2nd and 4th harmonic to source and drain.



Figure 4.5: Examples of transfer functions from drain-source to gate-source for implementations of the resonant gate drive with added capacitance or 2nd harmonic LC filter.



Figure 4.6: The 1st, 3rd and 5th harmonics in and out of phase with the drain signal.



Figure 4.7: Gate signals with 1st, 3rd and 5th order harmonics for 25% and 50% duty cycle (*=signal in phase with V_{DS}).

A specific design example of the gate drive is given in A.10.

4.2.1 Synchronous Rectification

The self-oscillating gate drive is very well suited for synchronous rectification, as it does not require a control signal to control the phase between the two gate signals. The two gate signals will automatically oscillate out of phase with the drain-source voltage of the MOSFET they are controlling, hence the gate drive on the inverter side will act as a master drive and the rectifier drive act as a slave drive following the frequency set by the master drive. This principle automatically takes component tolerances and temperature variations of critical design parameters into account. If used in an isolated converter this will further more benefit from not having a control signal and hence no need for communication across the isolation barrier. Figure 4.8 shows how a dc/dc converter with synchronous rectification can be made with this gate drive. As it can be seen the converter is completely symmetric across the resonant tank, C_R and L_R , hence operation in both directions is possible allowing for bidirectional power flow.



Figure 4.8: Schematic of a class E dc/dc converter with synchronous rectification.

Several resonant converters with bidirectional power flow have been published previously, e.g. a bidirectional full-bridge class-E converter in [166] and LLC converters in [167, 168, 169]. The use of synchronous rectification in VHF converters is however a bigger challenge due to the timing of the gate signals. The gate signals amplitude and duty cycle needs to be tightly controlled to ensure ZVS on the MOSFET in the rectifier and to keep the gate losses to a minimum. This gate drive has the advantage of being controlled by the drain voltage, hence the oscillation will start when the inverter starts sending power through the resonant tank and the voltage across the MOSFET in the rectifier rises.

A specific design examples of the gate drive used for synchronous rectification and bidirectional power flow is given in A.11. This was proof of principle showing that the gate drive can indeed be used for synchronous rectification. Further it was found that the converter had almost exactly the same open loop gain in both directions when used for bidirectional power flow.

4.2.2 High Side Gate Drive

The biggest advantage of this gate drive is that it can be used to drive high side switches, hence enable the use of half and full bridge topologies like the DE inverter shown in Figure 3.2. These topologies offer several important advantages over the single switch topologies and the only drawback is the need for a high side gate drive. This gate drive is the first gate drive presented suited for high side gate drive at VHF. This hence enables a whole new set of topologies for VHF SMPS, which has major advantages over the single switch topologies.



Figure 4.9: Schematic of the class DE inverter with self oscillating gate drive.

An implementation of the self-oscillating resonant gate drive in a half bridge is shown in Figure 4.9. Here the two MOSFETs have equally sized inductors at the gates, the only difference is that the other node of L_{G1} is connected to V_{bias} through L_H and to the switch node through C_{G1} . In this way the voltage at this node will follow the switch node but have a dc offset set by V_{Bias1} . As the phase of gate signals of the MOSFETs are directly coupled to drain source voltages, which are 180° shifted, the gate signals will automatically be phase shifted and shoot through is avoided.

In A.6 the worlds first half bridge VHF converter working with more than a few volts at the converters input is presented. The prototype is shown in Figure 4.12 and some of the measured waveforms are shown in Figure 4.11. The circuits is highly affected by the measurements (the capacitance added by the probe) and the efficiency drops significantly during the measurements. The efficiency and output power of the converter is shown as function of input voltage in Figure 4.12.



Figure 4.10: Picture of the worlds first descrite VHF class DE converter.



Figure 4.11: Measured waveforms on the class DE converter.



Figure 4.12: Efficiency and output power as function of input voltage for a VHF class DE converter.

4.2. Self-Oscillating

Chapter 5

PCB Embedded Magnetics

Implementing the magnetics as part of the PCB is widely used in traditional converters [170, 171]. For VHF converters, the selection of suited magnetic materials is limited, due to extensive core losses in most materials at these frequencies, and air core inductors are therefore used [172]. In this chapter spiral, solenoid and toroid structures will be analyzed and compared for use as PCB embedded inductors and transformers in converters operating at VHF.

5.1 Inductors

One of the challenges to achieve high efficiency at VHF is to design high-Q inductors, as most magnetic materials are limited to operation up to a few megahertz. Even materials, which theoretically can be used with good performance up to 100 MHz, show poor performance in the VHF range [173]. At the high frequencies, the inductance needed is however also lowered significantly and it is hence possible to use relatively small air core inductors. So far, the type of inductor most commonly used are air core solenoids.

Given the small inductance, the inductors can, however, also be designed as Printed Circuit Board (PCB) embedded inductors. Several structures for the design of these inductors exist and many of them have been described several times in previous publications [174, 175, 176]. Spiral inductors are commonly used in Integrated Circuits (ICs) and hence several publications with design guides for these exists [177, 178, 179]. Though discrete air core solenoids are commonly used, solenoids are not as common for embedded inductors. Some publications are however available showing inductors with Q values of up to more than 300 [180, 181]. Toroid inductors are of specific interest due to their natural encapsulation of the field due to the shape of the structure. With no magnetic core to guide the flux, a structure which on its own encapsulates the field is necessary to avoid problems with electromagnetic interference (EMI) [174, 182, 183]. In the following subsections these structures will be investigated individually and then compared.
5.1.1 Spiral

Spiral inductors can be designed in many ways and the shape is often chosen due to limitations in production. Round shapes are some times not possible in IC processes and therefore square or octagonal spirals are often used in ICs [184, 185]. When core materials are used, it is desirable to utilize a shape were the tracks are shaped in a way that ease the design of the core. Racetracks are often used for this as it allows for the use of a straight and simple core shape along the length of the structure [186, 187].



Figure 5.1: A two layer PCB embedded spiral inductor

In this section, air core spirals on a PCB will be investigated, hence no core will be used and circles and curves can easily be made in the production process. Therefore, the spirals are designed as archimedean spirals, i.e., with constant trace width and a constant distance between the traces, see Figure 5.1. The length of the trace in an archimedean spiral can be calculated according to equation 5.1 with the assumption that $1 + 2 + 3 + ... + N = N \cdot \frac{N+1}{2}$

$$l_{TRACE} = \pi \cdot N_{TURNS} \frac{D+d}{2} = \pi \frac{D^2 - d^2}{4 \cdot T_{TRACE}}$$
(5.1)

Here N_{TURNS} is the number of turns, T_{TRACE} is the thickness of the copper and D and d is the outer and inner diameter, respectively. MATLAB has been used to design the inductors; here the build in MATLAB function pdist2 can be used to get a more accurate length when the coordinates for the trace are known.

Now that trace length is known, the DC resistance is simply calculated from equation 5.2. If a two layer spiral is made, the resistance of course has to be doubled and the resistance of the via has to be added. The ac resistance, caused by the skin effect, of a planar inductor can, according to [188], be approximated by equation 5.3.

$$R_{DC} = \frac{\rho \cdot l_{TRACE}}{W_{TRACE} \cdot T_{TRACE}}$$
(5.2)

$$R_{AC} = \frac{\rho \cdot l_{TRACE}}{W_{TRACE} \cdot \delta(1 - e^{T_{TRACE}/\delta})}$$
(5.3)

Here ρ is the electrical resistivity of copper, W_{TRACE} is the width of a trace and δ is the skin depth.

The inductance of a spiral can be calculated by equation 5.4 [189]. Here c_1 and c_2 are constants of 2.46 and 0.20, respectively, and p is a fill ratio defined as $p = \frac{D-d}{D+d}$.

$$L = \frac{\mu \cdot N_{TURNS}^2 \frac{D+d}{2}}{2} \left(ln \left(\frac{c_1}{p} \right) + c_2 \cdot p^2 \right)$$
(5.4)

5.1.2 Solenoid



Figure 5.2: A PCB embedded solenoid inductor

In order to calculate the resistance of a solenoid, the structure is split in two; the vias and the traces. The dc resistance of a single via and a single trace, can now be calculated according to equation 5.5 and 5.6, respectively [113].

$$R_{DC,VIA} = \frac{\rho \cdot T_{PCB}}{A_{VIA}} = \frac{\rho \cdot T_{PCB}}{\pi \cdot T_{VIA}(D_{VIA} - T_{VIA})}$$
(5.5)

$$R_{DC,TRACE} = \frac{\rho \cdot W_L}{W_{TRACE} \cdot T_{TRACE}}$$
(5.6)

Here T_{PCB} is the thickness of the PCB, A_{VIA} is the area of a cross section of a via, D_{VIA} is the via diameter, T_{VIA} is the thickness of the copper in a via and W_L is the width of the inductor.

The total DC resistance can then be calculated as the sum of all the vias and traces. Using multiple vias in parallel for each turn will lower the total resistance. If this is done the total resistance can be calculated according to equation 5.7. Here N_{VIAS} is the number of parallel vias in a single turn.

$$R_{DC} = 2 \cdot N_{TURNS} \left(\frac{R_{DC,VIA}}{N_{VIAS}} + R_{DC,TRACE} \right)$$
(5.7)

The AC resistance is calculated as a first order approximation of the skin effect, hence neglecting the proximity effect. This will not give exact results, but it ease the comparison of the three structures and is sufficient for evaluation of pros and cons of each structure.

It is assumed that the ac current will run along the inner edge of the solenoid, hence along the inner edge of the traces and the inner edge of the inner half of the vias. This assumption has been verified through finite element (FEM) simulations in COMSOL in A.14. Using this assumption and the skin depth, the ac resistance is calculated according to equation 5.8.

$$R_{AC} = 2 \cdot N_{TURNS} \left(\frac{R_{AC,VIA}}{N_{VIAS}} + R_{AC,TRACE} \right)$$
(5.8)

Where:

$$R_{AC,VIA} = \frac{\rho \cdot T_{PCB}}{A_{VIA}} = \frac{2 \cdot \rho \cdot T_{PCB}}{\pi \cdot \delta(D_{VIA} - \delta)}$$
$$R_{AC,TRACE} = \frac{\rho \cdot W_L}{W_{TRACE} \cdot \delta}$$

The inductance of the solenoid can be calculated according to Niwas formula [190]. In equation 5.9, L_L is the length of the inductor and g is the square of the hypotenuse of a cross section of the inductor given by; $g^2 = W_L^2 + T_{PCB}^2$

$$\begin{split} L &= 8e^{-9} \cdot N_{TURNS}^2 \frac{W_L \cdot T_{PCB}}{L_L} \left[\frac{1}{2} \frac{L_L}{T_{PCB}} sinh^{-1} \frac{T_{PCB}}{L_L} \right. \\ &+ \frac{1}{2} \frac{L_L}{W_L} sinh^{-1} \frac{T_{PCB}}{L_L} \\ &- \frac{1}{2} \left(1 - \frac{T_{PCB}^2}{L_L^2} \right) \frac{L_L}{T_{PCB}} sinh^{-1} \frac{W_L}{L_L \sqrt{1 + T_{PCB}^2 / L_L^2}} \\ &- \frac{1}{2} \left(1 - \frac{W_L^2}{L_L^2} \right) \frac{L_L}{W_L} sinh^{-1} \frac{T_{PCB}}{L_L \sqrt{1 + W_L^2 / L_L^2}} \\ &- \frac{1}{2} \frac{T_{PCB}}{L_L} sinh^{-1} \frac{W_L}{T_{PCB}} - \frac{1}{2} \frac{W_L}{L_L} sinh^{-1} \frac{T_{PCB}}{W_L} \\ &+ \frac{\pi}{2} - tan^{-1} \frac{W_L \cdot T_{PCB}}{L_L^2 \sqrt{1 + g^2 / L_L^2}} \\ &+ \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{g^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{g^2}{L_L^2} \right) + \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \\ &- \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{W_L^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{W_L^2}{L_L^2} \right) \\ &- \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{T_{PCB}^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{T_{PCB}^2}{L_L^2} \right) \\ &+ \frac{1}{6} \frac{L_L^2}{W_L \cdot T_{PCB}} \left(\frac{g^2 - W_L^2 - T_{PCB}^2}{L_L^2} \right) \right] \end{split}$$

5.1.3 Toroid



Figure 5.3: A two layer PCB embedded toroidal inductor

A toroid is a solenoid bent in a circle; hence the same procedure and some of the assumption can be used again. It is thus assumed that the current run along the inner edges and the toroid is split into slabs and vias to get the resistance. The dc resistance of a slab is determined by integration from the inner radius to the outer radius [182].

$$R_{DC,SLAB} = \int_{r_I}^{r_O} \frac{\rho}{T_{TRACE} \left(\frac{2 \cdot \pi \cdot r}{N_{TURNS}} - C_{TRACE}\right)}$$
$$= \frac{\rho \cdot N_{TURNS}}{2 \cdot \pi \cdot T_{TRACE}} ln \left(\frac{2 \cdot \pi \cdot r_O - C_{TRACE} \cdot N_{TURNS}}{2 \cdot \pi \cdot r_I - C_{TRACE} \cdot N_{TURNS}}\right)$$
(5.10)

Here C_{TRACE} is the clearance between two slabs and r_I and r_O are the inner and outer radius, respectively. As for the solenoid a first order approximation of the skin effect is used for the ac resistance, hence this becomes:

$$R_{AC,SLAB} = \int_{r_I}^{r_O} \frac{\rho}{\delta \left(\frac{2 \cdot \pi \cdot r}{N_{TURNS}} - C_{TRACE}\right)}$$
$$= \frac{\rho \cdot N_{TURNS}}{2 \cdot \pi \cdot \delta} ln \left(\frac{2 \cdot \pi \cdot r_O - C_{TRACE} \cdot N_{TURNS}}{2 \cdot \pi \cdot r_I - C_{TRACE} \cdot N_{TURNS}}\right)$$
(5.11)

The resistance of the vias are the same as for the solenoid and the total resistance becomes:

$$R_{DC} = N_{TURNS} \left(2 \cdot R_{DC,SLAB} + \frac{R_{DC,VIA}}{N_{INNER}} + \frac{R_{DC,VIA}}{N_{OUTER}} \right)$$
(5.12)

$$R_{AC} = N_{TURNS} \left(2 \cdot R_{AC,SLAB} + \frac{R_{AC,VIA}}{N_{INNER}} + \frac{R_{AC,VIA}}{N_{OUTER}} \right)$$
(5.13)

Here N_{INNER} and N_{OUTER} are the number of inner and outer vias, respectively. The inductance of a toroid is derived in [183] and repeated in equation 5.14.

$$L = \frac{\mu \cdot N_{TURNS}^2 \cdot T_{PCB}}{2 \cdot \pi} ln\left(\frac{r_O}{r_I}\right) \dots + \frac{r_O + r_I}{2} \mu \left[ln\left(8\frac{r_O + r_I}{r_O - r_I}\right) - 2 \right]$$
(5.14)

5.1.4 Design and Optimization

In order to design and produce the inductors, a MATLAB program with a GUI has been designed, see Figure 5.4. The program takes the production parameters and the desired inductance and size as input and uses a genetic algorithm and formulas from the previous subsections to find parameters for the initial design.

Then COMSOL is used to investigate the current distribution and magnetic fields and to verify the inductance and resistance. Once the design has been verified in COMSOL, a PCB file is generated directly from the same interface.

The optimization routine is used to find the best inductor, within a given area. Best is defined as the required inductance, with the highest Q factor and with the



Figure 5.4: MATLAB GUI for design and optimization of PCB embedded inductors.

lowest area, in that preferred order. The optimization routine is based on a genetic algorithm already implemented in MATLAB and has a maximum area set by the user as a boundary condition.

The inductor should have the highest possible Q factor and the desired inductance. This is achieved by having a cost function that the genetic algorithm minimizes. The genetic algorithm takes any number of inputs, with possibility to define the variable as an integer. Each variable is setup so a closer value to the desired will lower the cost function, and each variable is weighted according to the importance of that specific parameter. The cost function used is:

$$Cost = a \cdot \left(\frac{L}{L_{spec}} - 1\right) + b \cdot \frac{1}{Q} + c \cdot A \tag{5.15}$$

where L_{spec} is the specified inductance and A is the area of the inductor. The weight factors used are shown in Table 5.1.

Table 5.1: The weight factors used for the cost function.

а	b	с
100	10.000	1

To verify that the calculations used to derive the optimum structure, and evaluate that the formulas in overall, are usable the MATLAB program interfaces to COM-SOL. The structure is then simulated in COMSOL which returns the resistance and inductance as results, this is described further in A.14. Also more than 50 different test structures of each type of inductor has been made and measured, see Figure 5.5, these are described further in A.13.



Figure 5.5: Measurements of the inductance and DC and AC resistance of some of the experimental prototypes.

5.1.5 Comparison

An example of the calculated impedances of the three structures optimized for 82 nH in 8x8 mm is given in Table 5.2. The table shows that the solenoid has almost twice the AC resistance of the spiral and 20% more than the toroid. It should however be noted that a quadratic solenoid is far from optimal and higher Q could be obtained, if a more narrow solenoid was designed. A 6mm wide solenoid with 8 turns can have 86nH and a Q factor of 92 in $48mm^2$.

Structure	Spiral	Solenoid	Toroid
N_{TURNS}	3.2	6	16
L	81 nH	81 nH	$85 \ \mathrm{nH}$
R_{DC}	$65 \ \mathrm{m}\Omega$	$72 \text{ m}\Omega$	$100~\mathrm{m}\Omega$
$R_{AC} @ 50 \text{ MHz}$	$253~\mathrm{m}\Omega$	$495 \text{ m}\Omega$	$392~\mathrm{m}\Omega$
Q @ 50 MHz	100	51	68

Table 5.2: Parameters for three inductors with approx. 82 nH inductance in
8x8mm

Figure 5.6 shows a frequency sweep of the three structures, both of prototypes with a low number of turns and of ones with a high number of turns. The inductance is very close to constant across the entire frequency range, except for the big spiral which is close to the resonance at the higher frequencies. From Figure 5.6 it can be seen that the resistance of the spiral, starts to increase due to the skin effect much earlier than the two other structures. This is because the resistance of the spiral is almost entirely due to the resistance of the trace, here the skin effect starts at 1-5 MHz ($\delta = 65 - 30 \ \mu m$). The resistance of the solenoid and the toroid only start to increase at approximately 10 MHz where the skin depth equals the thickness of the vias (20 μm). At 50 MHz the skin depth is approximately 10 μm and all the structures are clearly affected by the skin depth.



Figure 5.6: Inductance and series resistance of prototypes with a low number of turns (solid) and high number of turns (dashed).

From the formulas for the inductance of each of the three structures, it can be seen that each of the three structures has a limited number of parameters, which can be adjusted if the PCB parameters are given. The number of turns is the only common design parameter for the three structures.

The only parameter left for the spiral, is the width of the trace (and the number of layers or inductors in series).

For the solenoid both the length and width can be adjusted and though it is generally best to avoid the extremes, i.e. long and narrow or short and wide, this gives a larger amount of design flexibility.

For the toroid it is possible to adjust both the inner and outer radius. For a small structure, the minimum inner radius will however be limited by the size of a via (incl. annular ring) and the number of turns. If the inner radius is increased, it is possible to ad more vias in parallel to reduce the resistance. The core of the inductor is then decreased and more turns are needed to keep the same inductance, thus the gain of doing so is limited.

Both spirals and solenoids have strong external magnetics fields, see Figure 5.7a-5.10a and 5.7b-5.10b, respectively. This make them likely to cause Electromagnetic Interference (EMI) problems, either by direct radiation or by coupling to nearby metallic structures such as cables and other circuit boards. The fields have been described for spirals in [174] and for solenoids and toroids in [175]. The external field from the toroid is much weaker than for the two other structures, see Figure 5.7c-5.10c.



Figure 5.7: Three types of PCB embedded inductors.



(a) Spiral

(b) Solenoid

(c) Toroid

Figure 5.8: Finite element simulations of the magnetic flux density (B-field [T]) inside the three types of PCB embedded inductors.







Figure 5.10: Near-field scans of the field strength from the three types of PCB embedded inductors (z component not included as in simulation).

Another very important parameter is the flexibility or design freedom of the inductor types, both when scaling the inductor and laying out the complete PCB.

The spiral is by far superior in this perspective. It can be made in a single layer, the width of the trace can be scaled to get the desired relation between size and Q factor and finally it does not need and integer number of turns, which makes it possible to get the terminals placed in a way that fits into the rest of the circuit.

The solenoid also has some benefits over the two other structures. It has a rectangular shape and it has both the terminals on the same layer. The rectangular shape is much easier to align together with other rectangular components on a board, which most often also is rectangular. The ability to adjust the width and length and thereby design it to fit perfectly into the circuit is also very convenient.

The toroid is clearly the least flexible as the connectors are fixed to be right above each other. Furthermore it is the structure that can achieve the lowest inductance per area and both of the two other structures can be design with lower DC and AC resistances for a given inductance in the same area. The high resistance is partly caused by the poor fill factor of the vias used on the inner and outer edges of the toroid. A way of handling this and adding a core to the design has been investigated in A.16 and A.15. With the fully PCB embedded structure the only real advantage of the toroid is however the small external field.

Several aspects of the three PCB inductors have now been evaluated and the pros and cons has been summed up in Table 5.3.

Structure	Pros	Cons
Toroid	Smallest external field	Least flexible
		Lowest Q and L pr area
Solenoid	Rectangular shape	Wide external field
	$High \ Q$	
Spiral	Highest Q	Strong external field
	Most flexible	
	Single layer	

 Table 5.3: Pros and Cons of the three structures.



Figure 5.11: Resonant converters with the different structures for PCB embedded inductors.

The three converters in Figure 5.11 have been designed to give and idea of how the complete circuit with PCB embedded inductors will behave. Embedding the inductors in the PCB clearly gives some benefits, i.e. a much lower profile and a higher power density, this is described further and illustrated in a circuit in A.5. However the design still has to be optimized in order to compete with discrete air core solenoids which can have Q factors of more than 200 [129].

The three inductor types have now been compared with regard to resistance, inductance, magnetic field and design flexibility, both theoretically and experimentally.

The results show that there are huge differences in what can be achieved with the different structure and that a single structure cannot be selected as the superior for all applications. The spiral can achieve the highest inductance per area and is the only structure that can be made in a single layer; the solenoid is easy to scale and can be designed to fit perfectly into a complete design and the toroid is by far superior with regard to the external magnetic field.

5.2 Transformers

Galvanic isolation in traditional high-frequency SMPS is usually implemented with an isolation transformer, which utilizes a magnetic core. This approach is not suitable for VHF-SMPS as the core-losses, as previously described, will be significant at VHF [91]. Air-core magnetics with the lack of magnetic core removes the corelosses and the possibility of saturation, making them a suitable solution for VHF operation with the converters low inductance requirements.

Transformers in general exist in a vast number of designs, most focused on versions with a core to guide the flux to achieve a good coupling between the primary and secondary side. Air-core transformers can also be made in many designs, but are for practical purposes limited to structures that in itself will guide the flux through a common winding area, since there is no core to help guiding the flux.

In the following subsections it will be investigated how the three inductor structures can be used to design PCB embedded transformers suited for use at VHF. The spirals and toroids have been investigated thoroughly in previous literature [191, 192] and will therefore only be covered briefly. For the solenoids on the other hand several new ways of designing PCB embedded transformers will be described and investigated.

5.2.1 Spiral

Due to the strong vertical field in the center of the spiral inductors, see Figure 5.9b, a strong coupling between two inductors can be achieved, by placing them vertically aligned on two subsequent layers in a PCB [193, 194]. Figure 5.12 shows a example of a two layer structure, but the structure can easily be scaled two include more layers in a multilayer PCB [195].

As the primary and secondary winding are placed on seperate layers of the PCB, this structure is very well suited in applications were a safety isolation is required.



Figure 5.12: Example of a two layer PCB embedded spiral transformer [193].

The structure is however also well suited for signals and can be combined with PCB embedded spiral inductor to get a complete converter with all magnetic components embedded in the PCB [196, 197]. In [198] a 1 kW LLC converter with a power density of 12.2 W/cm^3 and an efficiency of 95.5% is demonstrated, this shows that very high power densities can be achived by embedding the magnetics in the PCB while keeping the efficiency high. Spiral structures, or arrays of these, is also the most commen way to get coupling between two units in a wireless charging system [199, 200].

5.2.2 Toroid

The PCB embedded toroidal transformer was first presented in [172]. The structure is made in a minimum 4 layer PCB were one of the windings is made in the inner layers with burried viases and the other winding is made in the outer layers completely surounding the first winding, see Figure 5.13.



Figure 5.13: Example of a four layer PCB embedded toriodal transformer [172].

The structure were investigated further in [191] were analytic models were derived and verified through FEM simulations and measurements. The coupling which can be achieved with this structure is highly dependent on the PCB processes, but coupling in the range 0.4-0.65 has been shown [191, 201]. Higher copuling can be achieved if a core is embedded in the PCB and utilized to guide the flux inside the center structure [202]. This is however a complex and costly process and the concept in A.15 and A.16 seems more apealing from a process/cost point of view, if a core is to be used.

A coupling of 0.65 is fully acceptable for resonant converters as the leakage inductance can be utilized in the resonant tank. So with the core materials available today and the complexity involved in embedding the core, the air core version is the most attractive for VHF converters. The structure was tested in a converter in [203] were the isolation voltage was also tested and shown to comply with the required safety standards.

5.2.3 Solenoid

The PCB embedded rectangular air-core solenoid transformer can be designed in different configurations, but only the two main configurations will be investigated.

The solenoid transformer can be made by one solenoid structure nested inside another slightly larger solenoid structure, giving the transformer a rectangular cross section as seen in Figure 5.14a. This requires minimum a four layer PCB with buried vias. The second configuration is two or more windings in an interleaved pattern on the same layers. The interleaved configurations ranges from the bifilar configuration to the end-to-end configuration as seen in Figure 5.14b, 5.14c and 5.14d. The interleaved configurations can be implemented in a two layer PCB without buried vias.



Figure 5.14: Different winding configurations of rectangular PCB embedded aircore solenoid transformers

The inductive model of a transformer can be defined as a two-port passive device and is described by the inductance matrix in equation 5.16 [98]. The inductance matrix consists of the self-inductance for each winding, L_{pp} and L_{ss} , and the mutual inductances, L_{ps} and L_{sp} , which is determined by the flux that flows through the mutual area of the transformer. Note that L_{ps} and L_{sp} are equal since the transformer is a passive device.

$$\begin{bmatrix} v_p(t) \\ v_s(t) \end{bmatrix} = \begin{bmatrix} L_{pp} & L_{ps} \\ L_{sp} & L_{ss} \end{bmatrix} \frac{d}{dt} \begin{bmatrix} i_p(t) \\ i_s(t) \end{bmatrix}$$
(5.16)

The coupling ratio is defined as equation 5.17, and is a ratio of how much flux couples between the primary and secondary windings. In regular transformers wound around a core, the coupling is in the range of k = 0.99 for the best [98].

$$k = \frac{L_{ps}}{\sqrt{L_{pp}L_{ss}}} \tag{5.17}$$

To find the inductance matrix, an Agilent 4396B was set up as a vector network analyzer and the S-parameters measured. The S-parameter measurements can then be converted to Z-parameters [204] and the inductances calculated by equation 5.18.

$$L = \frac{\Im(Z)}{\omega} \tag{5.18}$$

Prototypes of each of the configurations shown in Figure 5.14 were produced and measured. The resulting inductances are listed in Table 5.4 for comparison. To remove any small measurement noise the values are mean values from 1 MHz to 10 MHz.

Nested				
Variable	L_{pp}	L_{ps}	L_{ss}	k
Measured [nH]	67.398	17.616	9.5757	69.8%
	Bifil	lar		
Variable	L_{pp}	L_{ps}	L_{ss}	k
Measured [nH]	42.247	18.945	41.866	45.5%
	Sections	s(2:2)		
Variable	L_{pp}	L_{ps}	L_{ss}	k
Measured [nH]	40.949	10.520	38.814	26.5%
End-to-end				
Variable	L_{pp}	L_{ps}	L_{ss}	k
Measured [nH]	40.999	32.203	39.453	8.0%

 Table 5.4: Inductances of different configurations of PCB embedded solenoid transformers.

The prototypes of the interleaved configurations have coupling factors in the range from 0.08 to 0.45. The more intertwined the primary and secondary side is, the higher the coupling. The coupling is in general limited by the clearance between the traces on the PCB, leaving the end-to-end configuration with the lowest coupling. The configuration with sections shows better properties and the bifilar configuration is the optimum of the interleaved, with a coupling of 0.45. The nested configuration is superior to all the interleaved configurations, as the coupling for this configuration is 0.7.

Though these coupling factors are lower than what can be achieved with a core, they are still suitable for VHF converters as the leakage inductance can be utilized in the design as part of the resonant tank. The nested configuration has the best performance, but is also the most expensive to produce due to the buried viases. This configuration has been evaluated in a practical design which is described in A.17.

5.2. Transformers

Chapter 6

Control

With a switching frequency in the VHF range, it will be possible to achieve very fast transient responses [121] which are highly demanded, e.g., for envelope tracking [205]. However, in order to fully benefit from this, an efficient and fast control loop has to be implemented. This is a big challenge and while some ways of achieving continuous regulation has been found [206, 207], the best results are still achieved using burst mode (or cell modulation) as in [99, 208, 118]. Due to the high switching frequency, the converter will reach steady state after just a few microseconds, this makes it possible to use an array of small converters and switch them on and off as needed. In this way, each converter is designed to operate with a defined load/output. This makes the design much easier as resonant inverters are generally very load dependent.

Control of these converters is also a challenge due to their resonant behavior and burst mode control has been used to overcome this challenge [99, 209], but it introduces spectral components at the bursting frequency and its harmonics.

Pulse width modulation (PWM) is the most commonly used control scheme for power converters, but if a resonant converter is controlled in this way, ZVS is lost as soon as the duty cycle moves away from one optimum value.

Pulse frequency modulation (PFM) is another option, but it requires a very wide frequency operation range to ensure even a limited controllability of the converter. Combinations of PWM and PFM have shown good results [210], but the control circuit becomes quite complex and still requires a wide frequency operation range.

Outphasing is another control option not as commonly used for VHF converters [211, 212]. Compared to burst mode control it has the advantage of not introducing low frequency harmonics.

Burst mode and outphasing control will be investigated in the following sections.

6.1 Burst Mode

So far the most commonly used control method for VHF converters has been burst mode control (also called cell modulation, on-off or bang-bang control) [103, 134, 213]. Here the entire converter is switched on and off in order to control the output. The benefit of controlling the converter in this way is that the converter will either be on and working in its optimal operating point or off with only small standby losses. The result is a wide control range with an almost flat efficiency curve [103, 134]. Further, the converter can be optimized for high peak efficiency in this specific operating point, instead of compromising the peak efficiency for higher average efficiency.

The major downside of burst mode control is that low frequency harmonics are introduced, which increases the size of the needed input and output filters. Another disadvantage is added circuitry that adds both to complexity and to the losses, which reduces the efficiency at full load. Further burst mode requires the converter to be able to start and reach steady state quickly, most implementations use a burst frequency approximately two decades below the switching frequency in order to insure that the converter has time to reach steady state and operate efficiently for some time before it is shut down again [103, 134, 213].

Several control schemes for burst mode control are available with hysteresis [103, 32] and PWM [118] being the most commonly used, phase shift [213] and constant ontime are two other alternatives.

Hysteresis control has the best transient response, as the transitions between the on- and off-states are defined by the hysteresis window, instead of a fixed frequency or times as in the other schemes. Another advantage is very good efficiency at light loads due to a very low burst frequency. These advantages come at the expense of widely varying burst frequencies, which increases the complexity and potentially size of the needed input/EMI filter.

Phase-shift control has the same pros and cons as hysteresis control, but the delay in the control circuitry is utilized in the design, which makes it possible to use slower components than would have been needed with hysteresis control.

The design of the input filter is more straightforward with PWM, where a fixed burst frequency is used, but the transient response and light load efficiency degrades. The EMI performance of a VHF converter controlled by PWM burst mode, is in many aspects similar to a conventional hard switched converter operating at the burst frequency.

A 27 MHz class DE converter with a 21 kHz PWM burst mode control has been implemented, see Figure 7.3. The converter converts from the US mains to a 60 V LED load, hence transient respons is not important and the control bandwidth can be very low. Therefore, 21 kHz was chosen in order to be out of the audible range, but keep the burst frequency as low as possible in order to reduce the number of start-ups and the losses introduces by these. With this control, a very flat efficiency curve was achieved. The peak efficiency was 89.5% at full load and only dropped 5% at 10% load, see Figure 7.3. This was possible due to a very fast start-up circuit that made it possible to turn the converter fully on within 100 μ s, hence start up



Figure 6.1: Measured efficiency curve for a DE converter switching at 27 MHz with fixed frequency burst mode control at 20 kHz.

losses were very low and only affected the efficiency when the ouput power became low.

Constant on-time control can be compared to PWM control, but with the on-time fixed instead of the frequency. This gives a more stable frequency than seen for hysteresis and phase-shift control, but still with good light load efficiency.

6.2 Outphasing

Outphasing, also referred to as Linear Amplification with Nonlinear Components (LINC), was introduced for RF amplifiers in the 1930s [214]. This control method has however only been used a few times in previous publication [211, 212] and posses several advantages compared to burst mode.



Figure 6.2: System view of a converter using outphasing modulation.

Outphasing control utilizes a phase shift between two or more inverters to control the combined output through a common rectifier, as illustrated with two inverters in Figure 6.2. In this way, the individual inverters run continuously at a fixed frequency while the total output to the load is regulated.

With this control method, all the benefits of the high switching frequency can be achieved, both fast transient response, wide control range and small input and output filters. The main drawback of outphasing is that the losses in the inverters are almost constant at light and full load, which decreases the light load efficiency significantly. Further, the varying load and the inverters impact on each other is a challenge.



Figure 6.3: Schematic of the lossless combiner for outphasing modulation [215].

Selecting the right circuit to combine the output of the inverters (the combiner) is crucial to achieve a good result with outphasing. Several approaches have been suggested [216, 217, 218, 219], but the lossless combiner proposed in [215] (shown in Figure 6.3) posses several benefits for use in resonant converters. The combiner only consists of an inductor and capacitor, if these are assumed ideal the combiner is lossless. Further, the admittance seen by the two inverters can be made zero, ensuring purely resistive loading of the inverters. As resonant converters are very load dependent and quickly looses ZVS when loaded reactively, this is crucial in order to ensure high efficiency.

The input impedance of the rectifier, R_{REC} , can be assumed purely resistive at the switching frequency. Further, the output voltage of the two inverters, V_{INV1} and V_{INV2} , can be assumed equal to:

$$V_{INV1} = V_{IN} \cdot G_{INV} \cdot e^{-I \cdot \phi} \tag{6.1}$$

$$V_{INV2} = V_{IN} \cdot G_{INV} \cdot e^{I \cdot \phi} \tag{6.2}$$

Where V_{IN} is the input voltage amplitude, G_{INV} is a fixed gain between the input voltage to the converters and the peak output voltage.

The admittance seen by each inverter can now be written as:

$$Y_1 = \frac{1}{X} - \frac{I \cdot e^{I \cdot \phi} \cdot V_{OUT}}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6.3)

$$Y_2 = \frac{1}{X} - \frac{I \cdot V_{OUT}}{X \cdot e^{I \cdot \phi} \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6.4)

Where G_{REC} is a fixed voltage gain from the peak input voltage of the rectifier to

the output voltage, V_{OUT} . The imaginary part of the admittance is:

$$\Im(Y_1) = \frac{1}{X} - \frac{V_{OUT} \cdot \cos(\phi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6.5)

$$\Im(Y_2) = \frac{1}{X} + \frac{V_{OUT} \cdot \cos(\phi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6.6)

Setting this to zero and solving for the phase, ϕ , gives:

$$\phi = \arccos\left(\frac{V_{IN} \cdot G_{INV} \cdot G_{REC}}{V_{OUT}}\right) \tag{6.7}$$

If the phase is controlled according to this equation, the loading of the inverters will be purely resistive. The real part of the admittance is:

$$\Re(Y_1) = \Re(Y_2) = \frac{V_{OUT} \cdot sin(\phi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6.8)

From this the output power, P_{OUT} , can be calculated for a given value of the combiner reactance, X, and input and output voltage by:

$$P_{OUT} = \frac{G_{INV} \cdot \sqrt{V_{OUT}^2 - V_{IN}^2 \cdot G_{INV}^2 \cdot G_{REC}^2}}{X \cdot G_{REC}}$$
(6.9)

From equation 6.8, it is seen that it is not possible to control the power, if the loading of the converters needs to be purely resistive. If the output voltage is kept fixed, the output power will, however, be a quadratic function of the input voltage that can be offset by the chosen reactance for the combiner.

A design example is given in A.19. The designed converter achieves very good performance with equally and purely resistive loading of the two inverters across the entire input voltage range. This combined with a design of the inverters that ensures that they can operate with ZVS across a wide load range, ensures efficient operation of the converter across the entire input voltage range. The aim was to achieve efficient line regulation with less than $\pm 20\%$ output power variation for an input variation of $\pm 30\%$. This was achieved as the converters keept within $\pm 16.5\%$ output power variation. Furthermore, the design exhibited very high efficiency across the entire input voltage range and only drops below 90% at the maximum input voltage.

6.2. Outphasing

Chapter 7

Electromagnetic Interference

Increasing the switching frequency into the VHF range reduces the value as well as the physical size of passive components significantly. This does not only reduce the size of the energy storing components inside the converter, but also the EMI filter can be greatly reduced [220, 221].

EMI problems arising from SMPS are well known and well described [222, 223, 224]. The vast majority of SMPS operate at switching frequencies well below 1 MHz and the size of the EMI-filter is governed by the limits regarding conducted emissions in the frequency range of 150 kHz to 30 MHz. Radiated emission in the range of 30 MHz to 1 GHz is typically a matter of proper PCB layout and design, not so much related to filter size. When the switching frequency is moved into the VHF range, above the range were conducted emissions are measured and into the range for radiated emissions, this changes and new challenges arise.

Frequency range	Limits		
MHz	$dB(\mu V/m)$		
	Quasi-peak	Average	
0.15 to 0.50	79	66	
0.50 to 30	73	60	

 Table 7.1: EN55022 limits for conducted disturbance at the mains ports of class
 B ITE [225].

In terms of EMI emission test, there are two standard tests; conducted emission in the frequency range of 150 kHz - 30 MHz, see Table 7.1, and radiated emission in the frequency range of 30 MHz - 1 GHz, see Table 7.2. The separation of conducted and radiated at 30 MHz is made because Electro Magnetic (EM) disturbance is more likely to use near-field coupling paths such as inductive or capacitive at lower frequencies since impedance levels are below that of free space. At higher frequencies the characteristic impedance of traces and such increases and exceeds that of free space, making it more likely for the EM disturbance to radiate [226].

Frequency range	Quasi-peak limits	
MHz	$dB(\mu V/m)$	
30 to 230	30	
230 to 1000	37	

 Table 7.2: EN55022 limits for radiated disturbance of class B ITE at a measuring distance of 10 m [225]

For non-VHF converters, the EMI-filter is mainly designed to attenuate the switching frequency and/or harmonics of this in the conducted emission range. The radiated disturbance encountered is usually caused by second-order effects such as oscillations caused by parasitics and diode reverse recovery issues [227]. Thus the radiated emissions from the converter are more related to layout issues and parasitics than the actual switching frequency.

For VHF converters, this is a lot different, since the fundamental switching frequency and not a secondary effect now generates the radiated emission. The major source is not an oscillation with small amplitude compared to the main switching waveform, it is main switching waveform.

If the physical size of the noise generator is much less than the wavelength, λ , of the disturbing frequency, it is normally assumed that there is no or little radiation from the source, since the generator does not constitute an effective antenna. The wavelength is given by:

$$\lambda = \frac{c}{f} \tag{7.1}$$

Where λ is the wavelength, c is the velocity, and f is the frequency. In the VHF range, the wavelength is 1-10 m, which is much longer than any dimensions in the converter. To be an effective antenna, the dimensions should at least be in the area of quarter wavelength.

For VHF converters, the noise source is much stronger than in traditional converters, so the assumption of no or little radiation from the converter does not apply. If the impedance to the outside world is larger than the intrinsic impedance of free air $(Z_0 = \mu_0 \cdot c_0 \approx 377 \ \Omega)$, we can estimate the maximum radiation from this source as [228]:

$$|E| = \frac{V \cdot l \cdot b \cdot (2 \cdot \pi/\lambda)^2}{4 \cdot \pi \cdot r}$$
(7.2)

Here V is the voltage, l is the length of the circuit, b is the width of the circuit, λ is the wavelength and r is the distance to the source.

As an example, the VHF converter in Figure 7.1a will be used. This converter consists of three class E inverters with class DE rectifiers coupled with serial input and parallel output (SIPO). The converter operates at 37 MHz, converts a 120 V_{AC} input to a 60 V_{DC} output and delivers up to 20 W.

The main driver for the radiated emission is the drain node of the converter. The voltage swing at this node is approximately 120 V. The maximum noise from the circuit at a distance of 3 meters (assuming positive interference of the 3 VHF converters) will be approximately 79 dB μ V. This is well above the limits of IEC55022B which is 40 dB μ V at 37 MHz. Filters on the cables will hence not be enough to get the radiation below the limits. As a consequence, a shield has been mounted on the prototype in Figure 7.1b to reduce the radiation. Small filters are also needed to avoid radiation from the cables, these are however very small and can barely be seen under the electrolytic capacitor.

Figure 7.2a shows the conducted emissions measured on the prototype. As expected almost no emissions can be measured below 30 MHz. A small peak around 10 MHz is observed, the reason for this is unknown and due to the small amplitude this has not been investigated further. The radiated emissions on the other hand are very large for the unshielded prototype. At the switching frequency, the peak is almost 50 dB μ V above the limit and all the harmonics are also way above the limits, see Figure 7.2b. Adding the shield and a small filter efficiently reduces these peaks well below the limits, see Figure 7.2c. This clearly shows that shielding is necessary, but also that it is a very effective way of reducing the radiated emissions.

As mentioned in chapter 6, one of the major drawbacks of burst mode control is the introduction of low frequency harmonics, which needs to be filtered to meet the EMI standards. For application were high control bandwidth is not required, but wide load range is, it is however very appealing and it is therefore interesting to investigate how big the drawback is. A class DE converter with specifications semilar to the one just described has been developed, but with a 21 kHz burst mode control added.

The prototype is shown in Figure 7.3 and the EMI measurements made on it is shown in Figure 7.4 and 7.5. The conducted emmisions were expected to be the biggest challenge due to the low frequencies introduced, but as seen in Figure 7.4b and 7.4c the prototype complies with EN 55022 both at full and 50% load. At full load the burst mode control is not active as the duty cycle is 100%, but at 50% load the full effect is seen as a lot of low frequency harmonics. It was however possible to reduce these harmonics to a level well below the limits with small filters. The electrolytic capacitor at the input was needed to reduce flicker on the output and



(a) Three stacked converters with serial input and parallel output(b) With EMI filter and shield

Figure 7.1: Pictures of 20 W converters operating at 37 MHz.

reduce the input voltage range for the converter, hence the only component added to attenuate the low frequency harmonics was a small inductor, which can hardly be seen under the capacitor.

The radiated emmissions are seen in Figure 7.5a and 7.5b, without and with filter and shield, respectively. A lot of the radiated emmissions were removed by adding the shield and a small filter, but three of the peaks are still above the limits. It is expected that these peaks can be removed by optimization of the layout and use of three legged capacitors, but this has not been implemented yet.











(c) Radiated emmisions from the prototype with filter and shield

Figure 7.2: EMI measurements on a 20 W converter operating at 37 MHz.



Figure 7.3: A 20 W prototype based on a 27 MHz class DE converter.



(c) Conducted emmisions with filter at 50% load

Figure 7.4: Conducted EMI measurements on a converter with burst mode control.



Figure 7.5: Radiated EMI measurements on a converter with burst mode control.

Electromagnetic Interference

Chapter 8

Conclusion and Future Perspectives

8.1 Summary

Several interesting topics have been researched throughout this thesis, so before making the final conclusion a small summary will be made.

The single most innovative part of this thesis is the self-oscillating resonat gate drive, it is the basis for most of the other results that have been achieved. The gate drive described in section 4.2 was first described in [163], but has been investigated and applied much further during this work. The gate drive is simple, cost effective and reliable due to its low component count and its self-oscillating behavior, which automatically incorporates component tolerances. The gate drive has been used for low side switches as many other gate drives shown in the literature, but it has also been shown in a synchronous rectifier were the self-oscillating behavior and independency of signals from the primary side is great advantages. Further it has been shown working as a high side gate drive, this is the first VHF high side gate drive presented in the world.

The fact that there is now a gate drive that can drive high side switches in VHF converters opens up a whole new range of topologies. Half and full bridge topologies are commonly used at lower frequencies, especially for step down converters, but this is the first time that a VHF version has been shown. So far only a simple DE converter has been made, but with the high side gate drive new topologies as the LLC, LCC, full bridge etc. are to be explored.

Further, the gate drive has only been used in the simple version with just a sine wave as the gate signal. Adding more harmonics to get a more trapezoidal signal and thereby faster switching of the MOSFET is expected to make it possible to achieve even higher efficiencies.

The class DE converter lowers the voltage stress on the MOSFET, from 3.56 times the input voltage for the class E to just the input voltage. This is a great improvement on one of the biggest challenges for VHF converters, but with the introduced input-output rearrangement presented in Section 3.3, this can be improved even further. For a non-isolated 2:1 converter the voltage stress on the MOSFET can now be reduced from 3.56 times the input voltage to just half the input voltage, hence 7.12 ($3.56 \cdot 2$) times. Further, half of the power is delivered to the load without going through the converter, hence almost 100% efficiently and without causing heat in the converter. This not only decreases the demand for break down voltage of the MOSFET, but also makes the resonating energy inside the converter more than 50 times lower (($3.56 \cdot 2$)²). This makes it much easier to make a VHF converter with a high input voltage, such as rectified mains, which delivers a low to medium level output power (0 to 200 W), for instance for charging a phone or laptop or for LED lighting.

With the very high frequencies air core inductors, and especially PCB embedded air core inductors, become a feasible alternative to cored magnetics. A MATLAB program with an intuitive and easy to use graphical user interface was made to design these. This program takes the PCB production parameters and the desired inductor specifications as input, uses a genetic algorithm to optimize the inductor structure and gives a PCB file with the inductor ready for production as output. This makes it very simple to design converters utilizing PCB embedded inductors and hence removes a great barrier for circuit designers.

Two widely different control algorithms have been investigated, namely burst mode and outphasing. Outphasing is great for applications were control bandwidth is the main concern and were the converter operates close to full load most of the time. For applications were light load efficiency is important, outphasing fails due to more or less constant loss independent on output power. Here burst mode is much more efficient and if control bandwidth is not a concern, particularly PWM burst mode with a low switching frequency. A prototype of a DE converter with PWM burst mode control was presented where the efficiency only dropped 5%-point from full load to 10% of full load.

Finally, the new electromagnetic challenges arising from the highly increased switching frequencies have been investigated. It has been shown how a shield around the power stage efficiently encapsulates the magnetic fields and makes it possible to pass EN 55022 [225]. Further, the EMI performance of a burst mode controlled VHF converter has also been presented for the first time. The measurements showed a lot of harmonics at low frequencies, but they could be filtered out with small and inexpensive filters in order to make the prototype pass the requirements for conducted emissions.

To sum up, the main contributions from this work are:

- Further investigation of the gate drive invented in [163]
- Proof of principle that the gate drive also work for synchronous rectification and high side gate drives
- The worlds first VHF half-bridge converter operating at more than a few volts
- Development of a rearranging method that significantly reduces the voltage stress and improves the efficiency

- The first direct comparison of toroid, spiral, and solenoid PCB embedded inductors
- Presentation of conducted and radiated EMI measurements with and without shield and burst mode control
- Development of prototypes that clearly shows the potential of VHF converters

8.2 Conclusion

This thesis started out with the objective:

"The objective of this PhD thesis is to investigate, whether it is possible to increase the switching frequency of power converters into the Very High Frequency range (VHF, 30-300 MHz).

The converters should achieve efficiencies at least on par with commercial products and use standard components in order to keep cost low. The aim is to show that a size and weight reduction of 80% can be achieved without compromising cost, efficiency or reliability."



Figure 8.1: A 20 W prototype based on a 27 MHz DE converter compared with an equivalent commercial product.

Table 8.1: Performance improvements compared with a commercial product.

Parameter	Commercial	Thesis result	Improvement
Power	20 W	20 W	0%
Efficiency	$85 \ \%$	89.5%	4.5%
Size	$33 \ {\rm cm}^3$	$9.9~{\rm cm}^3$	70%
Wheight	$53~{ m g}$	10 g	81%
Cost	≈ 3.2 \$	≈1.4 \$	56%

To verify if this has been achieved, the 20 W prototype with burst mode control is compared to commercial product in Figure 8.1 and Table 8.1. The picture clearly shows a dramatic size reduction, not the full 80%, but 70% size reduction was

achieved. This is more than has been achieved the last decade and more than predicted for the comming decade [13]. To be fair, the frequency is strictly speaking slightly outside the VHF range, but the technology and principles has been shown to work at VHF as well.

A weight reduction of 81% was achieved, hence more than 80% aimed for. This is largely due to the removal of magnetic core materials. The cost is not only kept low, but actually decreased more than 50%. The prices are calculated based on bill-of-materials and production of 100,000 units. These numbers are not exact, but clearly indicate that the cost is not compromised but reduced.

Early in the project, it became clear that heat would become the main challenge in order to achieve the desired size reduction. Focus has therefore been on improving the efficiency, as well. This has been done with an improvement of 4.5% or a 30% reduction of the losses generated in the converter.

These results clearly show a big potential for this technology. It is not only possible to get a converter that is significantly smaller and lighter, it is also more cost and power efficient. This project started out as an ambitious research project, but the results have been so promising that a company has been founded on this basis and commercial products are expected on the market by the end of the year.

8.3 Future Perspectives

Several aspects of VHF converters have been researched throughout this thesis, but there is still more to be done. The great achievements gotten so far leaves a good insentive to continue the research and develop the technology further and reach even higher level of miniaturization, higher efficiencies and wider control bandwidth. The future perspectives are split into two parts in the following, one focusing specificly on VHF converters and one on technologies for miniaturization.

8.3.1 Research within VHF SMPS

Some of the aspects that should be covered in the future research within VHF converters include:

• Multi resonant gate drives

The simple version of the resonant self-oscillating gate drive has been used in most curcuits so far, but significant improvements are expected when the more advanced implementations are used.

• LLC, LCC and full bridge

With a high side gate drive available the path is clear for research in a whole new range of toplogies suited for operation at VHF. All these topologies have yet to be investigated and evaluated at VHF for the first time.

• Power factor correction

Power factor correction is already a requirement in many applications above 75 W, but the requirements are expected to be even more strict in the future.

The power level for LED lighting is for instance already lowered to 25 W. It is therefore of prime importance to investigate how VHF converters that can be compatible with the existing and comming requirements for power factor correction.

• Control

Some control schemes have already been explored, but they are all a trade off between control bandwidth and efficiency. The high switching frequencies posses the opportunity to achieve very high control bandwidth and rapid transient respons, but so far the only control schemes achieving this either sacrifice light load efficiency or introduce low frequency harmonics.

• Magnetics

Future research within PCB embedded inductors should include derivation of formulas to calculate the parasitic capacitance and thereby also the resonance frequency, as these will become important for designs where large structures are needed to get higher Q factors.

Though there are many benefits of avoiding core materials, they also give some great advantages, such as smaller inductor sizes, better control of the magnetic fields and stronger coupling in transformers. Future research should therefore also include investigation in material suitable for use at these frequencies.

• Semiconductors

Today semiconductors, especially MOSFETs, are mainly optimized for use in low frequency hard switching converters where on-resistance and gate charge are the two most important design parameters. For MOSFETs used in VHF converters these parameters are still important, but output capacitance and gate resistance are more important. Hence, MOSFETs which are optimized for this could increase the efficiency of VHF converters even further.

• Gallium Nitride

From Figure 2.8, it is clear that there is a big area where both VHF converters and GaN FETs are well suited, hence GaN based VHF converters should be investigated further. With GaN, greater efficiencies can be achieved due to the better charge/resistance FOM, but GaN have also higher electron mobility enabling devices that can switch even faster and hence the opportunity to push the switching frequency even further.

• Increased frequencies

The possibility of pushing the switching frequency even further should definitely be investigated further, with or without GaN, the limits of Si have not been reached.

• Packaging

Packaging of semiconductors have to be researched further to push the frequency and insure thermal performance. In order to get MOSFETs with a trade off between on resistance and output capacitance more suited for VHF converter a smaller device is needed, hence the package has to handle the same heat from a smaller device. Further the parasitcs of the package, especially inductances, become extremely important when the frequency is pushed even

8.3. Future Perspectives

further, hence low inductance packages with great thermal performance will be needed for future VHF converters.

• Integration

Several integrated VHF converters were presented in Section 2.1.3 and 2.1.5, but all for low voltage levels. Integration of medium voltage and power VHF converters have yet to be fully explored. The best way to do this would properly be to follow the three steps desribed in Section 2.1.5, by first intregating the discrete versions into modules, then to PSiP, and finally aim for fully monolithic integration in a PwrSoC.

• Thermal management

Already during this project thermal management started to become the biggest challenge and this challenge will only increase as power density is pushed even further. Therefore, thermal management will be of prime importance in future research, heat sinks, thermal bridges, potting materials and PCB technologies have to be investigated to find the most efficienct ways of getting the heat away from the converter.

• Electromagnetic Interference

Detailed investigation of the EMI from VHF SMPS is also needed. If it turns out that this is a problem, structures surrounding the inductor to encapsulate the field could be a solution, magnetic shielding or a secondary short-circuited winding might be a solution.

8.3.2 Technologies for Miniaturized Power Electronics

The VHF technology alone will not be enough to meet the constant requirement for miniaturization. As already mentioned in the previous section other technologies have to support this development. This section below gives the authors thoughts about how the technologies for power electronics suited for mid-range voltages (10-500 V), mid power (1-500 W) and consumer/mid-range products will develop during the next decade:

This year:

- Modules and PwrSiP They are already here and the range of products increase fast, Vicor for instance released their newest module just last month [71]. The prices are however still high and need to go down to get mass adoption.
- GaN The first semiconductor products have been on the market for a few years, but the first products are just about to reach the market [229].
- VHF SMPS Several publications about VHF SMPS have been made during the last decade, but the first commercial products will reach the market this year [230].

Within five years:

- PwrSoC The first commercial products operating at a few volts were presented in [80], but products will slowly start to move in to the lowest range of power electronics (10-50 V and 1-5 W).
- VHF GaN SMPS Once both technologies are matured and adopted by the market, the volume will increase and price go down due to the economics of scale. Hence it will be possible to make very attractive products at a fair price by combining these technologies.
- VHF Modules/PwrSiP As the power density continue to increase, this needs to happen to get full control of parasitics, interconnections and thermals.
- SC+VHF PwrSoC Combining switch capacitor circuits and resonant converters in a fully integrated converter posses some great opportunities for exploring more advanced topologies and control methods. This can hence become a powerful combination again starting with the range 10-50 V and 1-5 W.
- GaN based ICs With the first GaN semiconductors ready on the market, GaN is expected to follow the same roadmap as Si based semiconductors and ICs. The next step will hence be a few devices on the same die and later full ICs [77].

Within ten years:

- SC+VHF GaN PwrSoC When both PwrSoCs with switch capacitor and VHF converters and GaN ICs are ready it will be a natural next step to combine these technologies to get even better performance.
- UHF Ultra high frequency (0.3-3 GHz [9]) converters have already been published, for instance a 770 MHz converter with an efficiency above 70% [231]. Commercial products are, however, still a bit far out in the future, but might become available within the next decade due to advances in both VHF technologies, wide band gap devices and packaging.
Bibliography

- U.S. Department of Energy. Solid-state lighting fact sheet lifetime and reliability. Technical report, Energy Efficiency & Renewable Energy, August 2013.
- [2] Simon Moores. China admits 40 Minning.com, October 2014. URL http://www.mining.com/web/ china-admits-40-of-magnetic-rare-earths-supply-is-illegal-2/.
- [3] Gab-Su Seo, Bo-Hyung Cho, and Kyu-Chan Lee. Electrolytic capacitor-less pv converter for full lifetime guarantee interfaced with dc distribution. In *Power Electronics and Motion Control Conference (IPEMC), 2012 7th International*, volume 2, pages 1235–1240, June 2012. doi: 10.1109/IPEMC.2012. 6259017.
- [4] Kun Zhao, P. Ciufo, and S. Perera. Lifetime analysis of aluminum electrolytic capacitor subject to voltage fluctuations. In *Harmonics and Quality of Power* (*ICHQP*), 2010 14th International Conference on, pages 1–5, Sept 2010. doi: 10.1109/ICHQP.2010.5625486.
- [5] A. Lahyani, P. Venet, G. Grellet, and P.-J. Viverge. Failure prediction of electrolytic capacitors during operation of a switchmode power supply. *Power Electronics, IEEE Transactions on*, 13(6):1199–1207, Nov 1998. ISSN 0885-8993. doi: 10.1109/63.728347.
- [6] Dianbo Fu, Bing Lu, and F.C. Lee. 1mhz high efficiency llc resonant converters with synchronous rectifier. In *Power Electronics Specialists Conference*, 2007. PESC 2007. IEEE, pages 2404–2410, June 2007. doi: 10.1109/PESC. 2007.4342388.
- [7] Daocheng Huang, Shu Ji, and F.C. Lee. Llc resonant converter with matrix transformer. *Power Electronics, IEEE Transactions on*, 29(8):4339–4347, Aug 2014. ISSN 0885-8993. doi: 10.1109/TPEL.2013.2292676.
- [8] Yehui Han and D.J. Perreault. Inductor design methods with lowpermeability rf core materials. *Industry Applications, IEEE Transactions* on, 48(5):1616–1627, Sept 2012. ISSN 0093-9994. doi: 10.1109/TIA.2012. 2209192.

- [9] 521-2002 ieee standard letter designations for radar-frequency bands. IEEE Standard, 2009.
- [10] J. Rabkowski, D. Peftitsis, and H.-P. Nee. Design steps towards a 40-kva sic inverter with an efficiency exceeding 99.5 In Applied Power Electronics Conference and Exposition (APEC), 2012 Twenty-Seventh Annual IEEE, pages 1536–1543, Feb 2012. doi: 10.1109/APEC.2012.6166024.
- [11] R.B. Peterson. Switch mode power supplies a new approach for consumer audio systems. Broadcast and Television Receivers, IEEE Transactions on, BTR-19(4):263–268, Nov 1973. ISSN 0018-9308. doi: 10.1109/TBTR1.1973. 299774.
- [12] Daocheng Huang, D. Gilham, Weiyi Feng, Pengju Kong, Dianbo Fu, and F.C. Lee. High power density high efficiency dc/dc converter. In *Energy Conversion Congress and Exposition (ECCE), 2011 IEEE*, pages 1392–1399, Sept 2011. doi: 10.1109/ECCE.2011.6063942.
- [13] J.W. Kolar, U. Drofenik, J. Biela, M.L. Heldwein, H. Ertl, T. Friedli, and S.D. Round. Pwm converter power density barriers. In *Power Conversion Conference - Nagoya, 2007. PCC '07*, pages P–9–P–29, April 2007. doi: 10.1109/PCCON.2007.372914.
- [14] Jae-Won Yang and Hyun-Lark Do. High-efficiency zvs ac-dc led driver using a self-driven synchronous rectifier. *Circuits and Systems I: Regular Papers*, *IEEE Transactions on*, 61(8):2505–2512, Aug 2014. ISSN 1549-8328. doi: 10.1109/TCSI.2014.2309837.
- [15] Jun-Ho Kim, Moon-Young Kim, Cheol-O Yeon, and Gun-Woo Moon. Analysis and design of boost-llc converter for high power density ac-dc adapter. In *ECCE Asia Downunder (ECCE Asia), 2013 IEEE*, pages 6–11, June 2013. doi: 10.1109/ECCE-Asia.2013.6579066.
- [16] Apple Inc. 13-inch macbook pro with retina display. Environmental Report, March 2015.
- [17] Apple Inc. iphone 6. Environmental Report, September 2014.
- [18] MeanWell. Plc-60 series 60w single output led power supply. Datasheet, March 2014.
- [19] IHS. Power supply market to grow by \$3.5 billion in four years. Online article, July 2014.
- [20] Kwang-Hwa Liu and F.C.Y. Lee. Zero-voltage switching technique in dc/dc converters. *Power Electronics, IEEE Transactions on*, 5(3):293–304, Jul 1990. ISSN 0885-8993. doi: 10.1109/63.56520.
- [21] R.L. Steigerwald. A comparison of half-bridge resonant converter topologies. *Power Electronics, IEEE Transactions on*, 3(2):174–182, Apr 1988. ISSN 0885-8993. doi: 10.1109/63.4347.

- [22] Bing Lu, Wenduo Liu, Yan Liang, F.C. Lee, and J.D. Van Wyk. Optimal design methodology for llc resonant converter. In *Applied Power Electronics Conference and Exposition*, 2006. APEC '06. Twenty-First Annual IEEE, pages 6 pp.-, March 2006. doi: 10.1109/APEC.2006.1620590.
- [23] B. Yang, F.C. Lee, A.J. Zhang, and Guisong Huang. Llc resonant converter for front end dc/dc conversion. In *Applied Power Electronics Conference and Exposition*, 2002. APEC 2002. Seventeenth Annual IEEE, volume 2, pages 1108–1112 vol.2, 2002. doi: 10.1109/APEC.2002.989382.
- [24] M.D. Seeman. Gan devices in resonant llc converters: System-level considerations. *Power Electronics Magazine*, *IEEE*, 2(1):36–41, March 2015. ISSN 2329-9207. doi: 10.1109/MPEL.2014.2381456.
- [25] D. A. Jackson. Design and characterization of a radio-frequency dc/dc powerconverter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [26] J. R. Warren III. Cell modulated dc/dc converter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [27] R. C. N. Pilawa-Podgurski. Design and evaluation of a very high frequency dc/dc converter. Master's thesis, Massachusetts Institute of Technology, 2007.
- [28] O. Leitermann. Radio frequency dc-dc converters: Device characterization. Master's thesis, Massachusetts Institute of Technology, 2008.
- [29] J. W. Phinney. Multi-resonant Passive Components for Power Conversion. PhD thesis, Massachusetts Institute of Technology, 2005.
- [30] Y. Han. Circuits and Passive Components for Radio-Frequency Power Conversion. PhD thesis, Massachusetts Institute of Technology, 2010.
- [31] D.J. Perreault, Jingying Hu, J.M. Rivas, Yehui Han, O. Leitermann, R.C.N. Pilawa-Podgurski, A. Sagneri, and C.R. Sullivan. Opportunities and challenges in very high frequency power conversion. In *Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE*, pages 1–14, Feb 2009. doi: 10.1109/APEC.2009.4802625.
- [32] J.M. Rivas, D. Jackson, O. Leitermann, A.D. Sagneri, Yehui Han, and D.J. Perreault. Design considerations for very high frequency dc-dc converters. In *Power Electronics Specialists Conference, 2006. PESC '06. 37th IEEE*, pages 1–11, June 2006. doi: 10.1109/PESC.2006.1712114.
- [33] Wei Wei Shao, Li Juan Chen, Cheng Liang Pan, Yong Bin Liu, and Zhi Hua Feng. Power density of piezoelectric transformers improved using a contact heat transfer structure. Ultrasonics, Ferroelectrics, and Frequency Control, IEEE Transactions on, 59(1):73–81, January 2012. ISSN 0885-3010. doi: 10.1109/TUFFC.2012.2157.
- [34] M.S. Rodgaard, T. Andersen, M.A.E. Andersen, and K.S. Meyer. Design of interleaved interdigitated electrode multilayer piezoelectric transformer utilizing longitudinal and thickness mode vibrations. In *Power and Energy* (*PECon*), 2012 IEEE International Conference on, pages 38–43, Dec 2012. doi: 10.1109/PECon.2012.6450243.

- [35] K.A. Fish, C.A. Rosen, and H.C. Rothenberg. Electromechanical transducer, April 1958. URL http://www.google.com/patents/US2830274. US Patent 2,830,274.
- [36] T. Andersen. Piezoelectric transformer based power supply for dielectric electro active polymers. PhD thesis, Technical University of Denmark, 2012.
- [37] M.S. Rodgaard, T. Andersen, K.S. Meyer, and M.A.E. Andersen. Design of interleaved multi layer rosen type piezoelectric transformer for high voltage dc/dc applications. In *Power Electronics, Machines and Drives (PEMD* 2012), 6th IET International Conference on, pages 1–6, March 2012. doi: 10.1049/cp.2012.0206.
- [38] M.S. Rodgaard, T. Andersen, and M.A.E. Andersen. Empiric analysis of zero voltage switching in piezoelectric transformer based resonant converters. In *Power Electronics, Machines and Drives (PEMD 2012), 6th IET International Conference on*, pages 1–6, March 2012. doi: 10.1049/cp.2012.0217.
- [39] T. Andersen, M.S. Rodgaard, and M.A.E. Andersen. Active match load circuit intended for testing piezoelectric transformers. In *Power Electronics, Machines and Drives (PEMD 2012), 6th IET International Conference on*, pages 1–4, March 2012. doi: 10.1049/cp.2012.0169.
- [40] Thomas Andersen. Piezoelectric transformer based power supply for dielectric electro active polymers. PhD thesis, Technical University of Denmark, Department of Electrical Engineering, 2012.
- [41] T. Bove, W. Wolny, E. Ringgaard, and K. Breboel. New type of piezoelectric transformer with very high power density. In *Applications of Ferroelectrics*, 2000. ISAF 2000. Proceedings of the 2000 12th IEEE International Symposium on, volume 1, pages 321–324 vol. 1, 2000. doi: 10.1109/ISAF.2000. 941564.
- M.S. Makowski and D. Maksimovic. Performance limits of switched-capacitor dc-dc converters. In *Power Electronics Specialists Conference*, 1995. PESC '95 Record., 26th Annual IEEE, volume 2, pages 1215–1221 vol.2, Jun 1995. doi: 10.1109/PESC.1995.474969.
- [43] Michael Douglas Seeman. A Design Methodology for Switched-Capacitor DC-DC Converters. PhD thesis, University of California at Berkeley, 2009.
- [44] V.W. Ng and S.R. Sanders. A high-efficiency wide-input-voltage range switched capacitor point-of-load dc-dc converter. *Power Electronics, IEEE Transactions on*, 28(9):4335–4341, Sept 2013. ISSN 0885-8993. doi: 10.1109/ TPEL.2012.2224887.
- [45] M.D. Seeman, V.W. Ng, Hanh-Phuc Le, M. John, E. Alon, and S.R. Sanders. A comparative analysis of switched-capacitor and inductor-based dc-dc conversion technologies. In *Control and Modeling for Power Electronics (COMPEL), 2010 IEEE 12th Workshop on*, pages 1–7, June 2010. doi: 10.1109/COMPEL.2010.5562407.

- [46] Toke Meyer Andersen. On-Chip Switched Capacitor Voltage Regulators for Granular Microprocessor Power Delivery. PhD thesis, Eidgenossische Technische Hochschule (ETH) Zurich, 2015.
- [47] D. El-Damak, S. Bandyopadhyay, and A.P. Chandrakasan. A 93using on-chip ferroelectric capacitors. In Solid-State Circuits Conference Digest of Technical Papers (ISSCC), 2013 IEEE International, pages 374–375, Feb 2013. doi: 10.1109/ISSCC.2013.6487776.
- [48] T.M. Andersen, F. Krismer, J.W. Kolar, T. Toifl, C. Menolfi, L. Kull, T. Morf, M. Kossel, M. Brandli, P. Buchmann, and P.A. Francese. A 4.6w/mm2 power density 86dc-dc converter in 32 nm soi cmos. In *Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE*, pages 692–699, March 2013. doi: 10.1109/APEC.2013.6520285.
- [49] T.M. Andersen, F. Krismer, J.W. Kolar, T. Toifl, C. Menolfi, L. Kull, T. Morf, M. Kossel, M. Brandli, P. Buchmann, and P.A. Francese. A sub-ns response on-chip switched-capacitor dc-dc voltage regulator delivering 3.7w/mm2 at 90soi cmos. In Solid-State Circuits Conference Digest of Technical Papers (ISSCC), 2014 IEEE International, pages 90–91, Feb 2014. doi: 10.1109/ ISSCC.2014.6757351.
- [50] L. Chang, R.K. Montoye, B.L. Ji, A.J. Weger, K.G. Stawiasz, and R.H. Dennard. A fully-integrated switched-capacitor 2:1 voltage converter with regulation capability and 90 In VLSI Circuits (VLSIC), 2010 IEEE Symposium on, pages 55–56, June 2010. doi: 10.1109/VLSIC.2010.5560267.
- [51] Toke Meyer Andersen, Florian Krismer, Johann Walter Kolar, Thomas Toifl, Christian Menolfi, Lukas Kuli, Thomas Morf, Marcel Kossel, Matthias Brandii, and Pier Andrea Francese. A feedforward controlled on-chip switched-capacitor voltage regulator delivering 10w in 32nm soi cmos. In *Solid- State Circuits Conference - (ISSCC), 2015 IEEE International*, pages 1–3, Feb 2015. doi: 10.1109/ISSCC.2015.7063076.
- [52] S.V. Araujo, M. Kazanbas, and P. Zacharias. Breaking the theoretical limits of silicon with innovative switch technologies. In *Industrial Electronics (ISIE)*, 2010 IEEE International Symposium on, pages 676–681, July 2010. doi: 10.1109/ISIE.2010.5637608.
- [53] Michael A. Briere. Gan-based power devices offer game-changing potential in power-conversion electronics. online article, 12 2008. URL http://www. eetimes.com/document.asp?doc_id=1272514.
- [54] Microsemi PPG. Gallium nitride (gan) versus silicon carbide (sic) in the high frequency (rf) and power switching applications, 2014. URL http:// www.digikey.com/Web%20Export/Supplier%20Content/Microsemi_278/ PDF/Microsemi_GalliumNitride_VS_SiliconCarbide.pdf?redirected=1.
- [55] N. Ikeda, Y. Niiyama, H. Kambayashi, Y. Sato, T. Nomura, S. Kato, and S. Yoshida. Gan power transistors on si substrates for switching applications. *Proceedings of the IEEE*, 98(7):1151–1161, July 2010. ISSN 0018-9219. doi: 10.1109/JPROC.2009.2034397.

- [56] R.S. Pengelly, S.M. Wood, J.W. Milligan, S.T. Sheppard, and W.L. Pribble. A review of gan on sic high electron-mobility power transistors and mmics. *Microwave Theory and Techniques, IEEE Transactions on*, 60(6):1764–1783, June 2012. ISSN 0018-9480. doi: 10.1109/TMTT.2012.2187535.
- [57] Umesh K. Mishra, L. Shen, T.E. Kazior, and Yi-Feng Wu. Gan-based rf power devices and amplifiers. *Proceedings of the IEEE*, 96(2):287–305, Feb 2008. ISSN 0018-9219. doi: 10.1109/JPROC.2007.911060.
- [58] A. Elasser and T.P. Chow. Silicon carbide benefits and advantages for power electronics circuits and systems. *Proceedings of the IEEE*, 90(6):969–986, Jun 2002. ISSN 0018-9219. doi: 10.1109/JPROC.2002.1021562.
- [59] E. Johnson. Physical limitations on frequency and power parameters of transistors. In 1958 IRE International Convention Record, volume 13, pages 27–34, March 1965. doi: 10.1109/IRECON.1965.1147520.
- [60] B. J. Baliga. Semiconductors for high voltage, vertical channel field effect transistors. Journal of Applied Physics, 53(3):1759–1764, 1982. doi: http:// dx.doi.org/10.1063/1.331646. URL http://scitation.aip.org/content/ aip/journal/jap/53/3/10.1063/1.331646.
- [61] J. Millan, P. Godignon, X. Perpina, A. Perez-Tomas, and J. Rebollo. A survey of wide bandgap power semiconductor devices. *Power Electronics*, *IEEE Transactions on*, 29(5):2155–2163, May 2014. ISSN 0885-8993. doi: 10.1109/TPEL.2013.2268900.
- [62] S. Dimitrijev, J. Han, D. Haasmann, H.A. Moghadam, and A. Aminbeidokhti. Power-switching applications beyond silicon: The status and future prospects of sic and gan devices. In *Microelectronics Proceedings - MIEL 2014, 2014* 29th International Conference on, pages 43–46, May 2014. doi: 10.1109/ MIEL.2014.6842083.
- [63] B.Jayant Baliga. Power semiconductor device figure of merit for high-frequency applications. *Electron Device Letters*, *IEEE*, 10(10):455–457, Oct 1989. ISSN 0741-3106. doi: 10.1109/55.43098.
- [64] H. Morkoç, S. Strite, G. B. Gao, M. E. Lin, B. Sverdlov, and M. Burns. Large-band-gap sic, iii-v nitride, and ii-vi znse-based semiconductor device technologies. *Journal of Applied Physics*, 76(3):1363–1398, 1994. doi: http: //dx.doi.org/10.1063/1.358463.
- [65] M. Roschke and F. Schwierz. Electron mobility models for 4h, 6h, and 3c sic [mesfets]. *Electron Devices, IEEE Transactions on*, 48(7):1442–1447, Jul 2001. ISSN 0018-9383. doi: 10.1109/16.930664.
- [66] J.W. Palmour, R. Singh, R.C. Glass, O. Kordina, and Jr. Carter, C.H. Silicon carbide for power devices. In *Power Semiconductor Devices and IC's*, 1997. *ISPSD '97.*, 1997 IEEE International Symposium on, pages 25–32, May 1997. doi: 10.1109/ISPSD.1997.601423.
- [67] M. Schweizer and J.W. Kolar. Design and implementation of a highly efficient three-level t-type converter for low-voltage applications. *Power Electronics*,

IEEE Transactions on, 28(2):899–907, Feb 2013. ISSN 0885-8993. doi: 10. 1109/TPEL.2012.2203151.

- [68] O. Kreutzer, B. Eckardt, and M. Marz. Unidirectional fast switching nonisolated 100 kw fuel cell boost converter. In *Power Electronics and Applications (EPE'14-ECCE Europe), 2014 16th European Conference on*, pages 1–10, Aug 2014. doi: 10.1109/EPE.2014.6910797.
- [69] Shu Ji, D. Reusch, and F.C. Lee. High-frequency high power density 3-d integrated gallium-nitride-based point of load module design. *Power Electronics*, *IEEE Transactions on*, 28(9):4216–4226, Sept 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2235859.
- [70] Sanken. Next generation power semiconductors: Sanken's commitment to gan/sic development. online article, 2015. URL http://www.semicon. sanken-ele.co.jp/en/guide/GaNSiC.html/.
- [71] Vicor. Via pfm family isolated ac-dc converter with pfc in 125 mm package. Datasheet, April 2015.
- [72] Ipdia expands its high temperature (250c) silicon capacitor lines, dedicated to mcm assembly modules, with new voltage ranges from 11 v to 450 v. Company webpage, February 2015.
- [73] M. Wens and M. Steyaert. A fully integrated cmos 800-mw four-phase semiconstant on/off-time step-down converter. *Power Electronics, IEEE Transactions on*, 26(2):326–333, Feb 2011. ISSN 0885-8993. doi: 10.1109/TPEL. 2010.2057445.
- [74] R. Foley, F. Waldron, J. Slowey, A. Alderman, B. Narveson, and S.C. O'Mathuna. Technology roadmapping for power supply in package (psip) and power supply on chip (pwrsoc). In Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE, pages 525–532, Feb 2010. doi: 10.1109/APEC.2010.5433622.
- [75] G. Schrom, P. Hazucha, J. Hahn, D.S. Gardner, B.A. Bloechel, G. Dermer, S.G. Narendra, T. Karnik, and V. De. A 480-mhz, multi-phase interleaved buck dc-dc converter with hysteretic control. In *Power Electronics Specialists Conference, 2004. PESC 04. 2004 IEEE 35th Annual*, volume 6, pages 4702– 4707 Vol.6, June 2004. doi: 10.1109/PESC.2004.1354830.
- [76] Dragan Maksimovic. Integration of gan supply modulators and rf power amplifiers. Invited presentation at IEEE PwrSoC 2014, 2014.
- [77] Dean Takahashi. Move over, silicon. gallium nitride chips are taking over. online article, 2015. URL http://venturebeat.com/2015/04/02/ move-over-silicon-gallium-nitride-chips-are-taking-over/.
- [78] C.O. Mathuna, Ningning Wang, S. Kulkarni, and Saibal Roy. Review of integrated magnetics for power supply on chip (pwrsoc). *Power Electronics, IEEE Transactions on*, 27(11):4799–4816, Nov 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2198891.

- [79] Min Kyu Song, M.F. Dehghanpour, J. Sankman, and Dongsheng Ma. A vhf-level fully integrated multi-phase switching converter using bond-wire inductors, on-chip decoupling capacitors and dll phase synchronization. In Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, pages 1422–1425, March 2014. doi: 10.1109/APEC. 2014.6803493.
- [80] E.A. Burton, G. Schrom, F. Paillet, J. Douglas, W.J. Lambert, K. Radhakrishnan, and M.J. Hill. Fivr - fully integrated voltage regulators on 4th generation intel core socs. In *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, pages 432–439, March 2014. doi: 10.1109/APEC.2014.6803344.
- [81] Intersil. Isl8272m 50a digital dc/dc pmbus power module. Datasheet, January 2015.
- [82] Texas Instruments. Miniature, 1w isolated regulated dc/dc converters. Datasheet, October 2001.
- [83] Altera. Em1130p01qi 30 a powersoc featuring digital control with pmbus 1.2 compliant bus. Datasheet, February 2015.
- [84] Liyu Yang, Tiefu Zhao, Jun Wang, and A.Q. Huang. Design and analysis of a 270kw five-level dc/dc converter for solid state transformer using 10kv sic power devices. In *Power Electronics Specialists Conference*, 2007. *PESC* 2007. *IEEE*, pages 245–251, June 2007. doi: 10.1109/PESC.2007.4341996.
- [85] V. Pala, Han Peng, P. Wright, M.M. Hella, and T.Paul Chow. Integrated high-frequency power converters based on gaas phemt: Technology characterization and design examples. *Power Electronics, IEEE Transactions on*, 27 (5):2644–2656, May 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2011.2174803.
- [86] V.W. Ng, M.D. Seeman, and S.R. Sanders. Minimum pcb footprint point-ofload dc-dc converter realized with switched-capacitor architecture. In *Energy Conversion Congress and Exposition*, 2009. ECCE 2009. IEEE, pages 1575– 1581, Sept 2009. doi: 10.1109/ECCE.2009.5316488.
- [87] Zhemin Zhang, K.D.T. Ngo, and J.L. Nilles. A 30-w flyback converter operating at 5 mhz. In Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, pages 1415–1421, March 2014. doi: 10.1109/APEC.2014.6803492.
- [88] Sungjin Choi, Taeil Kim, and B.H. Cho. Design of half-bridge piezotransformer converters in the ac adapter applications. In Applied Power Electronics Conference and Exposition, 2005. APEC 2005. Twentieth Annual IEEE, volume 1, pages 244–248 Vol. 1, March 2005. doi: 10.1109/APEC.2005. 1452927.
- [89] R.J. Gutmann. Application of rf circuit design principles to distributed power converters. *Industrial Electronics and Control Instrumentation, IEEE Transactions on*, IECI-27(3):156–164, Aug 1980. ISSN 0018-9421. doi: 10.1109/TIECI.1980.351669.

- [90] W.C. Bowman, F.T. Balicki, F.T. Dickens, R.M. Honeycutt, W.A. Nitz, W. Strauss, W.B. Suiter, and N.G. Ziesse. A resonant dc-to-dc converter operating at 22 megahertz. In Applied Power Electronics Conference and Exposition, 1988. APEC '88. Conference Proceedings 1988., Third Annual IEEE, pages 3–11, Feb 1988. doi: 10.1109/APEC.1988.10545.
- [91] Yehui Han, G. Cheung, An Li, C.R. Sullivan, and D.J. Perreault. Evaluation of magnetic materials for very high frequency power applications. *Power Electronics, IEEE Transactions on*, 27(1):425–435, Jan 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2011.2159995.
- [92] Jizheng Qiu and C.R. Sullivan. Design and fabrication of vhf tapped power inductors using nanogranular magnetic films. *Power Electronics*, *IEEE Transactions on*, 27(12):4965–4975, Dec 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2193143.
- [93] H.B. Kotte, R. Ambatipudi, and K. Bertilsson. High-speed (mhz) series resonant converter (src) using multilayered coreless printed circuit board (pcb) step-down power transformer. *Power Electronics, IEEE Transactions on*, 28(3):1253–1264, March 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2012. 2208123.
- [94] Wu Chen and S.Y.R. Hui. Elimination of an electrolytic capacitor in ac/dc light-emitting diode (led) driver with high input power factor and constant output current. *Power Electronics, IEEE Transactions on*, 27(3):1598–1607, March 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2010.2103959.
- [95] Hongbo Ma, Jih-Sheng Lai, Quanyuan Feng, Wensong Yu, Cong Zheng, and Zheng Zhao. A novel valley-fill sepic-derived power supply without electrolytic capacitor for led lighting application. *Power Electronics, IEEE Transactions* on, 27(6):3057–3071, June 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2011. 2174446.
- [96] D. Reusch and J. Strydom. Evaluation of gallium nitride transistors in high frequency resonant and soft-switching dc-dc converters. *Power Electronics*, *IEEE Transactions on*, PP(99):1–1, 2014. ISSN 0885-8993. doi: 10.1109/ TPEL.2014.2364799.
- [97] M. Kovacevic. Radio frequency switch-mode power supplies. PhD thesis, Technical University of Denmark, 2015.
- [98] R. W. Erickson and D. Maksimic. *Fundamentals of Power Electronics*. Springer, 2001.
- [99] J. Rivas. Radio Frequency dc-dc Power Conversion. PhD thesis, Massachusetts Institute of Technology, 2006.
- [100] Fairchild Semiconductors. Bss138k n-channel logic level enhancement mode field effect transistor. Datasheet, May 2013. URL https://www. fairchildsemi.com/datasheets/BS/BSS138K.pdf.
- [101] STMicroelectronics. Stb6n52k3 n-channel 525 v, 1 Ω, 5 a supermesh3 power mosfet. Datasheet, March 2011. URL http://www.st.com/web/en/ resource/technical/document/datasheet/CD00208985.pdf.

- [102] Marian K Kazimierczuk and Dariusz Czarkowski. Resonant power converters. John Wiley & Sons, 2012.
- [103] J.M. Rivas, O. Leitermann, Yehui Han, and D.J. Perreault. A very high frequency dc-dc converter based on a class ϕ_2 resonant inverter. *Power Electronics, IEEE Transactions on*, 26(10):2980–2992, Oct 2011. ISSN 0885-8993. doi: 10.1109/TPEL.2011.2108669.
- [104] M.K. Kazimierczuk and J. Jozwik. Resonant dc/dc converter with class-e inverter and class-e rectifier. *Industrial Electronics, IEEE Transactions on*, 36(4):468–478, Nov 1989. ISSN 0278-0046. doi: 10.1109/41.43017.
- [105] Yehui Han, O. Leitermann, D.A. Jackson, J.M. Rivas, and D.J. Perreault. Resistance compression networks for radio-frequency power conversion. *Power Electronics, IEEE Transactions on*, 22(1):41–53, Jan 2007. ISSN 0885-8993. doi: 10.1109/TPEL.2006.886601.
- [106] R.S. Wahby. Radio frequency rectifiers for dc-dc power conversion. Master's thesis, Massachusetts Institute of Technology, 2004.
- [107] Wardah Inam, Khurram K. Afridi, and David J. Perreault. High efficiency resonant dc/dc converter utilizing a resistance compression network. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 1399–1405, March 2013. doi: 10.1109/APEC. 2013.6520482.
- [108] D.C. Hamill. Class de inverters and rectifiers for dc-dc conversion. In Power Electronics Specialists Conference, 1996. PESC '96 Record., 27th Annual IEEE, volume 1, pages 854–860 vol.1, Jun 1996. doi: 10.1109/PESC.1996. 548681.
- [109] I. Douglas de Vries. High Power and High Frequency Class-DE Inverters. PhD thesis, University of Cape Town, 1999.
- [110] Z. Kaczmarczyk. High-efficiency class e, ef_2 , and e/f_3 inverters. Industrial Electronics, IEEE Transactions on, 53(5):1584–1593, Oct 2006. ISSN 0278-0046. doi: 10.1109/TIE.2006.882011.
- [111] J.M. Rivas, Yehui Han, O. Leitermann, A.D. Sagneri, and D.J. Perreault. A high-frequency resonant inverter topology with low-voltage stress. *Power Electronics, IEEE Transactions on*, 23(4):1759–1771, July 2008. ISSN 0885-8993. doi: 10.1109/TPEL.2008.924616.
- [112] Z. Kaczmarczyk. A high-efficiency class-e inverter computer model, laboratory measurements and spice simulation. Bulletin of the Polish Academy of Science, Technical Sciences, 55(4), 2007.
- [113] P. Kamby. High efficiency radio frequency switch-mode power supply for led applications. Master's thesis, Technical University of Denmark, 2011.
- [114] J.S. Glaser, J. Nasadoski, and R. Heinrich. A 900w, 300v to 50v dc-dc power converter with a 30 mhz switching frequency. In *Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE*, pages 1121–1128, Feb 2009. doi: 10.1109/APEC.2009.4802804.

- [115] T. M. Andersen. Radio frequency switch-mode power supplies. Master's thesis, Technical University of Denmark, 2010.
- [116] J.S. Glaser and J.M. Rivas. A 500 w push-pull dc-dc power converter with a 30 mhz switching frequency. In Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE, pages 654–661, Feb 2010. doi: 10.1109/APEC.2010.5433602.
- [117] A.D. Sagneri, D.I. Anderson, and D.J. Perreault. Optimization of integrated transistors for very high frequency dc-dc converters. *Power Electronics, IEEE Transactions on*, 28(7):3614–3626, July 2013. ISSN 0885-8993. doi: 10.1109/ TPEL.2012.2222048.
- [118] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, and D.J. Perreault. High-frequency resonant sepic converter with wide input and output voltage ranges. *Power Electronics, IEEE Transactions on*, 27(1):189–200, Jan 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2011.2149543.
- [119] M. Kovacevic, A. Knott, and M.A.E. Andersen. Vhf series-input paralleloutput interleaved self-oscillating resonant sepic converter. In *Energy Con*version Congress and Exposition (ECCE), 2013 IEEE, pages 2052–2056, Sept 2013. doi: 10.1109/ECCE.2013.6646959.
- [120] J. M. Burkhart, R. Korsunsky, and D. J. Perreault. Design methodology for a very high frequency resonant boost converter. *Power Electronics, IEEE Transactions on*, 28(4):1929–1937, April 2013. ISSN 0885-8993. doi: 10.1109/ TPEL.2012.2202128.
- P. Shamsi and B. Fahimi. Design and development of very high frequency resonant dc-dc boost converters. *Power Electronics, IEEE Transactions on*, 27 (8):3725–3733, Aug 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2185518.
- [122] Andrei Grebennikov and Nathan O. Sokal. Switchmode RF Power Amplifiers. Newnes, 2007.
- [123] Marian K Kazimierczuk. RF power amplifier. John Wiley & Sons, 2014.
- [124] M. Hayati, A. Lotfi, M.K. Kazimierczuk, and H. Sekiya. Performance study of class-e power amplifier with a shunt inductor at subnominal condition. *Power Electronics, IEEE Transactions on*, 28(8):3834–3844, Aug 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2227814.
- [125] G. Maderbacher, T. Jackum, W. Pribyl, and C. Sandner. A sensor concept for minimizing body diode conduction losses in dc/dc converters. In *ESSCIRC*, 2010 Proceedings of the, pages 442–445, Sept 2010. doi: 10.1109/ESSCIRC. 2010.5619738.
- [126] N.O. Sokal and A.D. Sokal. Class e-a new class of high-efficiency tuned singleended switching power amplifiers. *Solid-State Circuits, IEEE Journal of*, 10 (3):168–176, Jun 1975. ISSN 0018-9200. doi: 10.1109/JSSC.1975.1050582.
- [127] R. Redl, Bela Molnar, and N.O. Sokal. Class e resonant regulated dc/dc power converters: Analysis of operations, and experimental results at 1.5

mhz. Power Electronics, IEEE Transactions on, PE-1(2):111–120, April 1986. ISSN 0885-8993. doi: 10.1109/TPEL.1986.4766289.

- [128] Murata. Chip monolithic ceramic capacitors. Datasheet, August 2014. URL http://www.murata.com/~/media/webrenewal/support/ library/catalog/products/capacitor/mlcc/c02e.ashx.
- [129] Coilcraft. Maxi spring air core inductors. Datasheet, November 2011. URL http://www.coilcraft.com/pdfs/maxi.pdf.
- [130] Mark Fortunato. Temperature and voltage variation of ceramic capacitors, or why your 4.7uf capacitor becomes a 0.33uf capacitor. Online Tutorial, November 2012. URL http://www.maximintegrated.com/en/app-notes/ index.mvp/id/5527.
- [131] J.W. Phinney, D.J. Perreault, and Jeffrey H. Lang. Radio-frequency inverters with transmission-line input networks. *Power Electronics, IEEE Transactions* on, 22(4):1154–1161, July 2007. ISSN 0885-8993. doi: 10.1109/TPEL.2007. 900465.
- [132] M.C. Smit, J.A. Ferreira, and J.D. van Wyk. Application of transmission line principles to high frequency power converters. In *Power Electronics Specialists Conference*, 1992. PESC '92 Record., 23rd Annual IEEE, pages 1423–1430 vol.2, Jun 1992. doi: 10.1109/PESC.1992.254746.
- S. Sander. Buck and boost converters with transmission lines. Power Electronics, IEEE Transactions on, 27(9):4013-4020, Sept 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2188044.
- [134] R.C.N. Pilawa-Podgurski, A.D. Sagneri, J.M. Rivas, D.I. Anderson, and D.J. Perreault. Very-high-frequency resonant boost converters. *Power Electronics*, *IEEE Transactions on*, 24(6):1654–1665, June 2009. ISSN 0885-8993. doi: 10.1109/TPEL.2009.2016098.
- [135] A. Mediano and N.O. Sokal. Class-e rf power amplifier with a flat-top transistor-voltage waveform. In *Microwave Symposium Digest (MTT)*, 2012 *IEEE MTT-S International*, pages 1–3, June 2012. doi: 10.1109/MWSYM. 2012.6259441.
- [136] J.M. Burkhart, R. Korsunsky, and D.J. Perreault. Design methodology for a very high frequency resonant boost converter. In *Power Electronics Conference (IPEC), 2010 International*, pages 1902–1909, June 2010. doi: 10.1109/IPEC.2010.5542054.
- [137] Wei Cai and Zhiliang Zhang. Analysis and design of a 30 mhz resonant boost converter. In *Power Electronics and Motion Control Conference (IPEMC)*, 2012 7th International, volume 3, pages 1905–1909, June 2012. doi: 10.1109/ IPEMC.2012.6259129.
- [138] R.C.N. Pilawa-Podgurski, A.D. Sagneri, J.M. Rivas, D.I. Anderson, and D.J. Perreault. Very-high-frequency resonant boost converters. *Power Electronics*, *IEEE Transactions on*, 24(6):1654–1665, June 2009. ISSN 0885-8993. doi: 10.1109/TPEL.2009.2016098.

- [139] P. Shamsi and B. Fahimi. Design and development of very high frequency resonant dc-dc boost converters. *Power Electronics, IEEE Transactions on*, 27 (8):3725–3733, Aug 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2185518.
- [140] M. Bathily, B. Allard, and F. Hasbani. A 200-mhz integrated buck converter with resonant gate drivers for an rf power amplifier. *Power Electronics, IEEE Transactions on*, 27(2):610–613, Feb 2012. ISSN 0885-8993. doi: 10.1109/ TPEL.2011.2119380.
- [141] D.C. Hamill. Half bridge class de rectifier. *Electronics Letters*, 31(22):1885–1886, Oct 1995. ISSN 0013-5194. doi: 10.1049/el:19951318.
- [142] N. Golbon, F. Ghodousipour, and G. Moschopoulos. A dc-dc converter with stacked flyback converters. In *Energy Conversion Congress and Exposition* (ECCE), 2013 IEEE, pages 4894–4899, Sept 2013. doi: 10.1109/ECCE.2013. 6647360.
- [143] K. Siri, M. Willhoff, K.A. Conner, and D.Q. Tran. High-voltage-input, low-voltage-output, series-connected converters with uniform voltage distribution. In Aerospace conference, 2009 IEEE, pages 1–9, March 2009. doi: 10.1109/ AERO.2009.4839519.
- [144] L. Roslaniec and D.J. Perreault. Design of variable-resistance class e inverters for load modulation. In *Energy Conversion Congress and Exposition (ECCE)*, 2012 IEEE, pages 3226–3232, Sept 2012. doi: 10.1109/ECCE.2012.6342344.
- [145] M.K. Kazimierczuk and K. Puczko. Exact analysis of class e tuned power amplifier at any q and switch duty cycle. *Circuits and Systems, IEEE Transactions on*, 34(2):149–159, Feb 1987. ISSN 0098-4094. doi: 10.1109/TCS. 1987.1086114.
- [146] T. Suetsugu and M.K. Kazimierczuk. Analysis and design of class e amplifier with shunt capacitance composed of nonlinear and linear capacitances. *Circuits and Systems I: Regular Papers, IEEE Transactions on*, 51(7):1261–1268, July 2004. ISSN 1549-8328. doi: 10.1109/TCSI.2004.830695.
- [147] Michael J. Chudobiak. The use of parasitic nonlinear capacitors in class e amplifiers. Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on, 41(12):941–944, Dec 1994. ISSN 1057-7122. doi: 10.1109/81.340867.
- [148] T. Suetsugu and M.K. Kazimierczuk. Comparison of class-e amplifier with nonlinear and linear shunt capacitance. *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on*, 50(8):1089–1097, Aug 2003. ISSN 1057-7122. doi: 10.1109/TCSI.2003.815208.
- [149] T. Roy, L. Smith, and J. Prymak. Esr and esl of ceramic capacitor applied to decoupling applications. In *Electrical Performance of Electronic Packaging*, 1998. IEEE 7th Topical Meeting on, pages 213–216, Oct 1998. doi: 10.1109/ EPEP.1998.733985.
- [150] M. Ingalls and Gordon Kent. Measurement of the characteristics of high-q ceramic capacitors. *Components, Hybrids, and Manufacturing Technology,*

IEEE Transactions on, 10(4):487–495, Dec 1987. ISSN 0148-6411. doi: 10. 1109/TCHMT.1987.1134797.

- [151] T. Tatsuta, Y. Ishitani, and T. Suetsugu. Gate power loss of class e amplifier with rectangular wave gate drive. In *TENCON 2010 - 2010 IEEE Region* 10 Conference, pages 1784–1787, Nov 2010. doi: 10.1109/TENCON.2010. 5686054.
- [152] Wei Liang, J. Glaser, and J. Rivas. 13.56 mhz high density dc-dc converter with pcb inductors. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 633–640, March 2013. doi: 10.1109/APEC.2013.6520277.
- [153] I.D. de Vries, J.H. van Nierop, and J.R. Greene. Solid state class de rf power source. In *Industrial Electronics*, 1998. Proceedings. ISIE '98. IEEE International Symposium on, volume 2, pages 524–529 vol.2, Jul 1998. doi: 10.1109/ISIE.1998.711593.
- [154] T.M. Andersen, S.K. Christensen, A. Knott, and M.A.E. Andersen. A vhf class e dc-dc converter with self-oscillating gate driver. In *Applied Power Electronics Conference and Exposition (APEC), 2011 Twenty-Sixth Annual IEEE*, pages 885–891, March 2011. doi: 10.1109/APEC.2011.5744699.
- [155] M. Kovacevic, A. Knott, and M. A. E. Andersen. Interleaved self-oscillating class e derived resonant dc/dc converters. *International Conference on Electrical and Computer Systems*, 2012.
- [156] H. Fujita. A resonant gate-drive circuit with optically isolated control signal and power supply for fast-switching and high-voltage power semiconductor devices. *Power Electronics, IEEE Transactions on*, 28(11):5423–5430, Nov 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2013.2247423.
- [157] Runruo Chen and Fang Zheng Peng. A high-performance resonant gate-drive circuit for mosfets and igbts. *Power Electronics, IEEE Transactions on*, 29 (8):4366–4373, Aug 2014. ISSN 0885-8993. doi: 10.1109/TPEL.2013.2284836.
- [158] H. Fujita. A resonant gate-drive circuit capable of high-frequency and highefficiency operation. *Power Electronics, IEEE Transactions on*, 25(4):962– 969, April 2010. ISSN 0885-8993. doi: 10.1109/TPEL.2009.2030201.
- [159] Wei Cai, Zhiliang Zhang, Xiaoyong Ren, and Yan fei Liu. A 30-mhz isolated push-pull vhf resonant converter. In Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, pages 1456–1460, March 2014. doi: 10.1109/APEC.2014.6803498.
- [160] F.H. Raab. Class-f power amplifiers with maximally flat waveforms. Microwave Theory and Techniques, IEEE Transactions on, 45(11):2007–2012, Nov 1997. ISSN 0018-9480. doi: 10.1109/22.644215.
- [161] A. Sagneri. Design of a very high frequency dc-dc boost converter. Master's thesis, Massachusetts Institute of Technology, 2007.

- [162] M.K. Kazimierczuk, V.G. Krizhanovski, J.V. Rassokhina, and D.V. Chernov. Class-e mosfet tuned power oscillator design procedure. *Circuits and Systems I: Regular Papers, IEEE Transactions on*, 52(6):1138–1147, June 2005. ISSN 1549-8328. doi: 10.1109/TCSI.2005.849127.
- [163] M. Madsen. Offline very high frequency power converters. Master's thesis, Technical University of Denmark, 2012.
- [164] T. Suetsugu and M.K. Kazimierczuk. Integrated class de synchronized dcdc converter for on-chip power supplies. In *Power Electronics Specialists Conference, 2006. PESC '06. 37th IEEE*, pages 1–5, June 2006. doi: 10. 1109/PESC.2006.1711803.
- [165] D.C. Hamill. Impedance plane analysis of class de amplifier. *Electronics Letters*, 30(23):1905–1906, Nov 1994. ISSN 0013-5194. doi: 10.1049/el:19941361.
- [166] S. Jalbrzykowski, A. Bogdan, and T. Citko. A dual full-bridge resonant classe bidirectional dc-dc converter. *Industrial Electronics, IEEE Transactions* on, 58(9):3879–3883, Sept 2011. ISSN 0278-0046. doi: 10.1109/TIE.2010. 2100335.
- [167] Eun-Soo Kim, Jun-Hyoung Park, Yong-Seog Jeon, Young-Su Kong, Seung-Min Lee, and Kwangseob Kim. Bidirectional secondary llc resonant converter using auxiliary switches and inductor. In Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, pages 1941–1947, March 2014. doi: 10.1109/APEC.2014.6803572.
- [168] Tianyang Jiang, Junming Zhang, Xinke Wu, Kuang Sheng, and Yousheng Wang. A bidirectional llc resonant converter with automatic forward and backward mode transition. *Power Electronics, IEEE Transactions on*, 30(2): 757–770, Feb 2015. ISSN 0885-8993. doi: 10.1109/TPEL.2014.2307329.
- [169] T. LaBella, Wensong Yu, Jih-Sheng Lai, M. Senesky, and D. Anderson. A bidirectional-switch-based wide-input range high-efficiency isolated resonant converter for photovoltaic applications. *Power Electronics, IEEE Transactions on*, 29(7):3473–3484, July 2014. ISSN 0885-8993. doi: 10.1109/TPEL. 2013.2282258.
- [170] H. Schneider, T. Andersen, A. Knott, and M.A.E. Andersen. Hybrid winding concept for toroids. In *ECCE Asia Downunder (ECCE Asia)*, 2013 IEEE, pages 936–940, June 2013. doi: 10.1109/ECCE-Asia.2013.6579218.
- [171] R. Pittini, Zhe Zhang, and M.A.E. Andersen. High current planar magnetics for high efficiency bidirectional dc-dc converters for fuel cell applications. In *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, pages 2641–2648, March 2014. doi: 10.1109/APEC.2014. 6803677.
- [172] Jens Pejtersen and A. Knott. Design and measurement of planar toroidal transformers for very high frequency power applications. In *Power Electronics* and Motion Control Conference (IPEMC), 2012 7th International, volume 1, pages 688–692, June 2012. doi: 10.1109/IPEMC.2012.6258779.

- [173] Shanshan Lu, Yuqin Sun, M. Goldbeck, D.R. Zimmanck, and C.R. Sullivan. 30-mhz power inductor using nano-granular magnetic material. In *Power Electronics Specialists Conference, 2007. PESC 2007. IEEE*, pages 1773–1776, June 2007. doi: 10.1109/PESC.2007.4342268.
- [174] S. Orlandi, B. Allongue, G. Blanchot, S. Buso, F. Faccio, C. Fuentes, M. Kayal, S. Michelis, and G. Spiazzi. Optimization of shielded pcb aircore toroids for high-efficiency dc-dc converters. *Power Electronics, IEEE Transactions on*, 26(7):1837–1846, July 2011. ISSN 0885-8993. doi: 10.1109/ TPEL.2010.2090902.
- [175] Dong Hui, Zhu Yisheng, and Zhao Baishan. Research on the electromagnetic radiation of a pcb planar inductor. In *Microwave Conference Proceedings*, 2005. APMC 2005. Asia-Pacific Conference Proceedings, volume 1, pages 3 pp.-, Dec 2005. doi: 10.1109/APMC.2005.1606313.
- [176] Liangliang Li, Dok Won Lee, Kyu-Pyung Hwang, Yongki Min, T. Hizume, M. Tanaka, Ming Mao, Thomas Schneider, Randhir Bubber, and Shan X. Wang. Small-resistance and high-quality-factor magnetic integrated inductors on pcb. Advanced Packaging, IEEE Transactions on, 32(4):780–787, Nov 2009. ISSN 1521-3323. doi: 10.1109/TADVP.2009.2019845.
- [177] Dong Hui, Zhu Yi-sheng, and Zhao Bai-shan. Analysis of electromagnetic scattering from a pcb inductor based on wavelet-mom. In *Microwave and Millimeter Wave Technology*, 2008. ICMMT 2008. International Conference on, volume 2, pages 544–547, April 2008. doi: 10.1109/ICMMT.2008.4540449.
- [178] M. Zolog, D. Pitica, and O. Pop. Characterization of spiral planar inductors built on printed circuit boards. In *Electronics Technology*, 30th International Spring Seminar on, pages 308–313, May 2007. doi: 10.1109/ISSE. 2007.4432869.
- [179] S.J. Cheon, W.C. Park, and J.Y. Park. Highly performed pcb embedded inductors using inverted-patterned ground structure. In *Microwave Conference* (*EuMC*), 2010 European, pages 312–315, Sept 2010.
- [180] R. Kamali-Sarvestani and J.D. Williams. Fabrication of high quality factor rfresonator using embedded inductor and via capacitor. In *IECON 2010 - 36th Annual Conference on IEEE Industrial Electronics Society*, pages 2283–2287, Nov 2010. doi: 10.1109/IECON.2010.5675102.
- [181] R. Kamali-Sarvestani and J.D. Williams. Fabrication of high quality factor rf-solenoids using via structures. In Wireless and Microwave Technology Conference (WAMICON), 2011 IEEE 12th Annual, pages 1–4, April 2011. doi: 10.1109/WAMICON.2011.5872901.
- [182] P. Kamby, A. Knott, and M.A.E. Andersen. Printed circuit board integrated toroidal radio frequency inductors. In *IECON 2012 - 38th Annual Conference* on *IEEE Industrial Electronics Society*, pages 680–684, Oct 2012. doi: 10. 1109/IECON.2012.6388746.

- [183] C.R. Sullivan, Weidong Li, S. Prabhakaran, and Shanshan Lu. Design and fabrication of low-loss toroidal air-core inductors. In *Power Electronics Specialists Conference*, 2007. PESC 2007. IEEE, pages 1754–1759, June 2007. doi: 10.1109/PESC.2007.4342265.
- [184] C.P. Yue and S.S. Wong. On-chip spiral inductors with patterned ground shields for si-based rf ics. *Solid-State Circuits, IEEE Journal of*, 33(5):743– 752, May 1998. ISSN 0018-9200. doi: 10.1109/4.668989.
- [185] A.M. Niknejad and R.G. Meyer. Analysis, design, and optimization of spiral inductors and transformers for si rf ics. *Solid-State Circuits*, *IEEE Journal* of, 33(10):1470–1481, Oct 1998. ISSN 0018-9200. doi: 10.1109/4.720393.
- [186] Jizheng Qiu and C.R. Sullivan. Inductor design for vhf tapped-inductor dc-dc power converters. In Applied Power Electronics Conference and Exposition (APEC), 2011 Twenty-Sixth Annual IEEE, pages 142–149, March 2011. doi: 10.1109/APEC.2011.5744588.
- [187] L. Daniel, C.R. Sullivan, and S.R. Sanders. Design of microfabricated inductors. *Power Electronics, IEEE Transactions on*, 14(4):709–723, Jul 1999. ISSN 0885-8993. doi: 10.1109/63.774209.
- [188] Thomas H. Lee. The Design of CMOS Radio-Frequency Integrated Circuits. Cambridge University Press, 2003.
- [189] S.S. Mohan, M. del Mar Hershenson, S.P. Boyd, and T.H. Lee. Simple accurate expressions for planar spiral inductances. *Solid-State Circuits, IEEE Journal of*, 34(10):1419–1424, Oct 1999. ISSN 0018-9200. doi: 10.1109/4.792620.
- [190] Frederick W. Grover. Inductance Calculations: Working Formulas and Tables. Courier Dover Publications, 2004.
- [191] J. Pejtersen, J. Dollner Monster, and A. Knott. Development and verification of printed circuit board toroidal transformer model. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 1654–1659, March 2013. doi: 10.1109/APEC.2013.6520519.
- [192] A. Bouabana, C. Sourkounis, and M. Mallach. Implementation of different layouts of a coreless planar transformer for a flyback converter. In *IECON* 2012 - 38th Annual Conference on *IEEE Industrial Electronics Society*, pages 483–487, Oct 2012. doi: 10.1109/IECON.2012.6388776.
- [193] S.C. Tang, S.Y.R. Hui, and H.S.-H. Chung. Characterization of coreless printed circuit board (pcb) transformers. *Power Electronics, IEEE Transactions on*, 15(6):1275–1282, Nov 2000. ISSN 0885-8993. doi: 10.1109/63. 892842.
- [194] S.C. Tang, S.Y.R. Hui, and H.S.-H. Chung. Coreless planar printed-circuitboard (pcb) transformers-a fundamental concept for signal and energy transfer. *Power Electronics, IEEE Transactions on*, 15(5):931–941, Sep 2000. ISSN 0885-8993. doi: 10.1109/63.867683.

- [195] A.D. Sagneri, D.I. Anderson, and D.J. Perreault. Transformer synthesis for vhf converters. In *Power Electronics Conference (IPEC)*, 2010 International, pages 2347–2353, June 2010. doi: 10.1109/IPEC.2010.5543674.
- [196] A. Majid, H.B. Kotte, J. Saleem, R. Ambatipudi, S. Haller, and K. Bertilsson. High frequency half-bridge converter using multilayered coreless printed circuit board step-down power transformer. In *Power Electronics and ECCE Asia (ICPE ECCE), 2011 IEEE 8th International Conference on*, pages 1177– 1181, May 2011. doi: 10.1109/ICPE.2011.5944712.
- [197] H.B. Kotte, R. Ambatipudi, and K. Bertilsson. High-speed (mhz) series resonant converter (src) using multilayered coreless printed circuit board (pcb) step-down power transformer. *Power Electronics, IEEE Transactions on*, 28(3):1253–1264, March 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2012. 2208123.
- [198] Dianbo Fu, F.C. Lee, and Shuo Wang. Investigation on transformer design of high frequency high efficiency dc-dc converters. In Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE, pages 940–947, Feb 2010. doi: 10.1109/APEC.2010.5433391.
- [199] E. Waffenschmidt and T. Staring. Limitation of inductive power transfer for consumer applications. In *Power Electronics and Applications*, 2009. EPE '09. 13th European Conference on, pages 1–10, Sept 2009.
- [200] S.Y. Hui. Planar wireless charging technology for portable electronic products and qi. *Proceedings of the IEEE*, 101(6):1290–1301, June 2013. ISSN 0018-9219. doi: 10.1109/JPROC.2013.2246531.
- [201] D. Bowen, A. Lee, C. Krafft, and I.D. Mayergoyz. Design control of performance in nested and interleaved winding printed circuit board transformers for ethernet applications. *Magnetics, IEEE Transactions on*, 49(7):4013– 4016, July 2013. ISSN 0018-9464. doi: 10.1109/TMAG.2013.2238513.
- [202] Jim Quilici. Embedded magnetic power transformer. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 1650–1653, March 2013. doi: 10.1109/APEC.2013.6520518.
- [203] J. D. Mø nster. Galvanic isolation in very high frequency offline power converters. Master's thesis, Technical University of Denmark, 2013.
- [204] D.A. Frickey. Conversions between s, z, y, h, abcd, and t parameters which are valid for complex source and load impedances. *Microwave Theory and Techniques, IEEE Transactions on*, 42(2):205–211, Feb 1994. ISSN 0018-9480. doi: 10.1109/22.275248.
- [205] O. Garcia, M. Vasic, P. Alou, J. Oliver, and J.A. Cobos. An overview of fast dc-dc converters for envelope amplifier in rf transmitters. *Power Electronics*, *IEEE Transactions on*, 28(10):4712–4722, Oct 2013. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2232941.
- [206] J.M. Alonso, M.S. Perdigao, D.G. Vaquero, A.J. Calleja, and E.S. Saraiva. Analysis, design, and experimentation on constant-frequency dc-dc resonant

converters with magnetic control. *Power Electronics, IEEE Transactions on*, 27(3):1369–1382, March 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2011. 2165083.

- [207] K. Fukui and H. Koizumi. Class e rectifier with controlled shunt capacitor. *Power Electronics, IEEE Transactions on*, 27(8):3704–3713, Aug 2012. ISSN 0885-8993. doi: 10.1109/TPEL.2012.2184560.
- [208] J.R. Warren, K.A. Rosowski, and D.J. Perreault. Transistor selection and design of a vhf dc-dc power converter. *Power Electronics, IEEE Transactions* on, 23(1):27–37, Jan 2008. ISSN 0885-8993. doi: 10.1109/TPEL.2007.911773.
- [209] J.M. Rivas, R.S. Wahby, J.S. Shafran, and D.J. Perreault. New architectures for radio-frequency dc-dc power conversion. *Power Electronics*, *IEEE Transactions on*, 21(2):380–393, March 2006. ISSN 0885-8993. doi: 10.1109/TPEL.2005.869740.
- [210] S. Hamamura, T. Ninomiya, M. Yamamoto, and M. Katsuno. Combined pwm and pfm control for universal line voltage of a piezoelectric transformer off-line converter. *Power Electronics, IEEE Transactions on*, 18(1):270–277, Jan 2003. ISSN 0885-8993. doi: 10.1109/TPEL.2002.807177.
- [211] A.S. Jurkov, L. Roslaniec, and D.J. Perreault. Lossless multiway power combining and outphasing for high-frequency resonant inverters. *Power Electronics, IEEE Transactions on*, 29(4):1894–1908, April 2014. ISSN 0885-8993. doi: 10.1109/TPEL.2013.2264773.
- [212] P.A. Godoy, D.J. Perreault, and J.L. Dawson. Outphasing energy recovery amplifier with resistance compression for improved efficiency. *Microwave Theory and Techniques, IEEE Transactions on*, 57(12):2895–2906, Dec 2009. ISSN 0018-9480. doi: 10.1109/TMTT.2009.2033976.
- [213] M. Kovacevic, A. Knott, and M.A.E. Andersen. A vhf interleaved selfoscillating resonant sepic converter with phase-shift burst-mode control. In *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, pages 1402–1408, March 2014. doi: 10.1109/APEC.2014. 6803490.
- [214] H. Chireix. High power outphasing modulation. Radio Engineers, Proceedings of the Institute of, 23(11):1370–1392, Nov 1935. ISSN 0731-5996. doi: 10. 1109/JRPROC.1935.227299.
- [215] D.J. Perreault. A new power combining and outphasing modulation system for high-efficiency power amplification. *Circuits and Systems I: Regular Papers, IEEE Transactions on*, 58(8):1713–1726, Aug 2011. ISSN 1549-8328. doi: 10.1109/TCSI.2011.2106230.
- [216] F.H. Raab. Efficiency of outphasing rf power-amplifier systems. Communications, IEEE Transactions on, 33(10):1094–1099, Oct 1985. ISSN 0090-6778. doi: 10.1109/TCOM.1985.1096219.
- [217] A. Birafane and A.B. Kouki. On the linearity and efficiency of outphasing microwave amplifiers. *Microwave Theory and Techniques, IEEE Transactions*

on, 52(7):1702–1708, July 2004. ISSN 0018-9480. doi: 10.1109/TMTT.2004. 830485.

- [218] Jingshi Yao and S.I. Long. Power amplifier selection for linc applications. Circuits and Systems II: Express Briefs, IEEE Transactions on, 53(8):763– 767, Aug 2006. ISSN 1549-7747. doi: 10.1109/TCSII.2006.876384.
- [219] F. H. Raab, P. Asbeck, S. Cripps, P. B. Kennington, Z. B. Popovich, N. Pothecary, J. F. Sevic, and N. O. Sokal. Rf and microwave power amplifier and transmitter technologies part 3. *High Frequency Electron*, pages 34–48, 2003.
- [220] A. Majid, J. Saleem, H.B. Kotte, R. Ambatipudi, and K. Bertilsson. Design and implementation of emi filter for high frequency (mhz) power converters. In *Electromagnetic Compatibility (EMC EUROPE)*, 2012 International Symposium on, pages 1–4, Sept 2012. doi: 10.1109/EMCEurope.2012.6396738.
- [221] A. Majid, J. Saleem, and K. Bertilsson. Emi filter design for high frequency power converters. In *Environment and Electrical Engineering (EEEIC)*, 2012 11th International Conference on, pages 586–589, May 2012. doi: 10.1109/ EEEIC.2012.6221444.
- [222] D.H. Liu and J.G. Jiang. High frequency characteristic analysis of emi filter in switch mode power supply (smps). In *Power Electronics Specialists Conference*, 2002. pesc 02. 2002 IEEE 33rd Annual, volume 4, pages 2039–2043, 2002. doi: 10.1109/PSEC.2002.1023114.
- [223] Liyu Yang, Bing Lu, Wei Dong, Zhiguo Lu, Ming Xu, F.C. Lee, and W.G. Odendaal. Modeling and characterization of a 1 kw ccm pfc converter for conducted emi prediction. In Applied Power Electronics Conference and Exposition, 2004. APEC '04. Nineteenth Annual IEEE, volume 2, pages 763–769 vol.2, 2004. doi: 10.1109/APEC.2004.1295909.
- [224] Peipei Meng, Junming Zhang, Henglin Chen, Zhaoming Qian, and Yuwen Shen. Characterizing noise source and coupling path in flyback converter for common-mode noise prediction. In Applied Power Electronics Conference and Exposition (APEC), 2011 Twenty-Sixth Annual IEEE, pages 1704–1709, March 2011. doi: 10.1109/APEC.2011.5744825.
- [225] Dansk Standard. Information technology equipment radio disturbance characteristics - limits and methods of measurement. DS/EN 55022, January 2011.
- [226] Tim Williams. EMC for Product Designers. Elsevier, 2006.
- [227] A. Hatanaka and T. Kawashima. A recovery-diode model for analyzing emc of an on-board power supply. In *Power Electronics and Applications (EPE)*, 2013 15th European Conference on, pages 1–5, Sept 2013. doi: 10.1109/EPE. 2013.6631818.
- [228] M.H. Nagrial and A. Hellany. Radiated and conducted emi emissions in switch mode power supplies (smps): sources, causes and predictions. In *Multi Topic Conference, 2001. IEEE INMIC 2001. Technology for the 21st Century. Proceedings. IEEE International*, pages 54–61, 2001. doi: 10.1109/INMIC. 2001.995314.

- [229] The world's smallest, lightest, smartest laptop charger, and more. Company webpage, May 2015. URL https://www.gozolt.com.
- [230] Dart -the world's smallest laptop charger. Company webpage, May 2015. URL http://finsix.com/dart/.
- [231] J.A. Garcia, R. Marante, and M. de las Nieves Ruiz Lavin. Gan hemt class e² resonant topologies for uhf dc/dc power conversion. *Microwave Theory* and *Techniques*, *IEEE Transactions on*, 60(12):4220–4229, Dec 2012. ISSN 0018-9480. doi: 10.1109/TMTT.2012.2222043.
- [232] F.H. Raab. Maximum efficiency and output of class-f power amplifiers. Microwave Theory and Techniques, IEEE Transactions on, 49(6):1162–1166, Jun 2001. ISSN 0018-9480. doi: 10.1109/22.925511.
- [233] H. Zirath and D. Rutledge. An ldmos vhf class e power amplifier using a high q novel variable inductor. In *Microwave Symposium Digest, 1999 IEEE MTT-S International*, volume 1, pages 367–370 vol.1, June 1999. doi: 10. 1109/MWSYM.1999.779495.
- [234] G. Spiazzi, P. Mattavelli, and L. Rossetto. Effects of parasitic components in high-frequency resonant drivers for synchronous rectification mosfets. *Power Electronics, IEEE Transactions on*, 23(4):2082–2092, July 2008. ISSN 0885-8993. doi: 10.1109/TPEL.2008.925200.
- [235] Kuang Sheng, Yongxi Zhang, Liangchun Yu, Ming Su, and J.H. Zhao. High-frequency switching of sic high-voltage lifet. *Power Electronics, IEEE Transactions on*, 24(1):271–277, Jan 2009. ISSN 0885-8993. doi: 10.1109/TPEL. 2008.2005984.
- [236] C.P. Wang, D.H. Liu, and Jiang Jianguo. Study of coupling effects among passive components used in power electronic devices. In *Power Electronics* and Motion Control Conference, 2004. IPEMC 2004. The 4th International, volume 3, pages 1500–1504 Vol.3, Aug 2004.
- [237] A.P. Mynster and M. Sorensen. Validation of emc near-field scanning amplitude and phase measurement data. In *Electromagnetic Compatibility (EMC EUROPE)*, 2012 International Symposium on, pages 1–6, Sept 2012. doi: 10.1109/EMCEurope.2012.6396836.
- [238] C.R. Sullivan. Aluminum windings and other strategies for high-frequency magnetics design in an era of high copper and energy costs. In Applied Power Electronics Conference, APEC 2007 - Twenty Second Annual IEEE, pages 78–84, Feb 2007. doi: 10.1109/APEX.2007.357498.
- [239] Seungbum Lim, John Ranson, David M. Otten, and David J. Perreault. Twostage power conversion architecture for an led driver circuit. In Applied Power Electronics Conference and Exposition (APEC), 2013 Twenty-Eighth Annual IEEE, pages 854–861, March 2013. doi: 10.1109/APEC.2013.6520311.
- [240] Seungbum Lim, D.M. Otten, and D.J. Perreault. Power conversion architecture for grid interface at high switching frequency. In Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, pages 1838–1845, March 2014. doi: 10.1109/APEC.2014.6803556.

- [241] P.J. Baxandall. Transistor sine-wave lc oscillators. some general considerations and new developments. Proceedings of the IEE - Part B: Electronic and Communication Engineering, 106(16):748–758, May 1959. ISSN 0369-8890. doi: 10.1049/pi-b-2.1959.0141.
- [242] J. Ebert and M. Kazimierczuk. Class e high-efficiency tuned power oscillator. Solid-State Circuits, IEEE Journal of, 16(2):62–66, Apr 1981. ISSN 0018-9200. doi: 10.1109/JSSC.1981.1051542.
- [243] V.G. Krizhanovski, D.V. Chernov, and M.K. Kazimierczuk. Low-voltage electronic ballast based on class e oscillator. *Power Electronics, IEEE Transactions on*, 22(3):863–870, May 2007. ISSN 0885-8993. doi: 10.1109/TPEL. 2007.896512.
- [244] J.J. Jozwik and M.K. Kazimierczuk. Analysis and design of class-e2 dc/dc converter. *Industrial Electronics, IEEE Transactions on*, 37(2):173–183, Apr 1990. ISSN 0278-0046. doi: 10.1109/41.52968.
- [245] M.K. Kazimierczuk and J. Jozwik. Class e2 narrow-band resonant dc/dc converters. Instrumentation and Measurement, IEEE Transactions on, 38 (6):1064–1068, Dec 1989. ISSN 0018-9456. doi: 10.1109/19.46401.
- [246] M. Mikolajewski and M.K. Kazimierczuk. Zero-voltage-ripple rectifiers and dc/dc resonant converters. *Power Electronics, IEEE Transactions on*, 8(1): 12–17, Jan 1993. ISSN 0885-8993. doi: 10.1109/63.208494.
- [247] J.G. Kassakian and T.M. Jahns. Evolving and emerging applications of power electronics in systems. *Emerging and Selected Topics in Power Electronics*, *IEEE Journal of*, 1(2):47–58, June 2013. ISSN 2168-6777. doi: 10.1109/ JESTPE.2013.2271111.
- [248] J. A. Pedersen. Bidirectional very high frequency converter. Master's thesis, Technical University of Denmark, 2013.
- [249] S. Wang. Characterization and Cancellation of High-Frequency Parasitics for EMI Filters and Noise Separators in Power Electronics Applications. PhD thesis, Virginia Polytechnic Institute and State University, 2005.

Appendix A

Publications and Patents

List of patent applications and publications at conference and in journal:

- A.1 A. Knott, T. M. Andersen, P. Kamby, M. Madsen, M. Kovacevic, M. A.E. Andersen: "On the Ongoing Evolution of Very High Frequency Power Supplies", IEEE Applied Power Electronics Conference and Exposition, Long Beach, CA, March 2013. Conference Proceedings p2514-2519.
- A.2 A. Knott, T. M. Andersen, P. Kamby, J. A. Pedersen, M. Madsen, M. Kovacevic, M. A.E. Andersen: "Evolution of Very High Frequency Power Supplies", IEEE Journal of Emerging and Selected Topics in Power Electronics, September 2014, vol.2, no.3, pp.386-394.
- A.3 M. Madsen, A. Knott, M. A.E. Andersen: "Low Power Very High Frequency Resonant Converter with High Step Down Ratio", IEEE AFRICON, Mauritius, September 2013. Conference Proceedings p1-6.
- A.4 M. Madsen, A. Knott, M. A.E. Andersen: "Low Power Very High Frequency Switch-Mode Power Supply with 50 V Input and 5 V Output", IEEE Transactions on Power Electronics, December 2014, vol.29, no.12, pp.6569-6580.
- A.5 M. Madsen, A. Knott, M. A.E. Andersen: "Very High Frequency Resonant DC/DC Converters for LED Lighting", IEEE Applied Power Electronics Conference and Exposition, Long Beach, CA, March 2013. Conference Proceedings p835-839.
- A.6 M. Madsen, A. Knott, M. A.E. Andersen: "Very High Frequency Half Bridge DC/DC Converter", IEEE Applied Power Electronics Conference and Exposition, Fort Worth, TX, March 2014. Conference Proceedings p1409-1414.
- A.7 M. Madsen, M. Kovacevic, J. D. Mønster, J. A. Pedersen, A. Knott and M. A.E. Andersen: "Input-Output Rearrangement of Isolated Converters", IEEE Power and Energy Conference at Illinois, Champaign, IL, February 2015. Conference Proceedings p1-6.

- A.8 M. Kovacevik and M. Madsen: "Step-up dc-dc power converter", WO2015091590 A2, June 25th 2015
- A.9 M. Kovacevik and M. Madsen: "Step-down dc-dc power converter", WO2015110427 A1, July 30th 2015
- A.10 M. Madsen, J. A. Pedersen, A. Knott, M. A.E. Andersen: "Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components", IEEE Applied Power Electronics Conference and Exposition, Fort Worth, TX, March 2014. Conference Proceedings p2029-2035.
- A.11 J. A. Pedersen, M. Madsen, A. Knott and M. A.E. Andersen: "Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter", IEEE Applied Power Electronics Conference and Exposition, Charlotte, NC, March 2015. Conference Proceedings.
- A.12 M. Madsen and J. A. Pedersen: "Self-oscillating resonant power converter", WO2014067915 A2, November 2nd 2012
- A.13 M. Madsen, A. Knott, M. A.E. Andersen, A. P. Mynster: "Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies", IEEE Energy Conversion Congress and Exhibition Asia DownUnder, Melbourne, Australia, June 2013. Conference Proceedings p1071-1078.
- A.14 M. Madsen, J. D. Mønster, A. Knott, M. A.E. Andersen: "Design Optimization of Printed Circuit Board Embedded Inductors through Genetic Algorithms with verification by COMSOL", COMSOL conference 2013, Boston, MA, October 2013. Conference Proceedings.
- A.15 H. Schneider, T. Andersen, J. D. Mønster, M. Madsen, A. Knott, M. A.E. Andersen: "Investigation of a Hybrid Winding Concept for Toroidal Inductors using 3D Finite Element Modeling", COMSOL conference 2013, Boston, MA, October 2013. Conference Proceedings.
- A.16 H. Schneider, T. Andersen, J. D. Mønster, M. Madsen, T. Andersen, A. Knott, M. A.E. Andersen: "Optimizing Inductor Winding Geometry for Lowest DC-Resistance using LiveLink between COMSOL and MATLAB", COM-SOL conference 2013, Boston, MA, October 2013. Conference Proceedings.
- A.17 J. D. Mønster, M. Madsen, J. A. Pedersen and A. Knott: "Investigation, development and verification of printed circuit board embedded air-core solenoid transformers", IEEE Applied Power Electronics Conference and Exposition, Charlotte, NC, March 2015. Conference Proceedings.
- A.18 M. Madsen and J. Mønster: "Embedded solenoid transformer for power conversion", WO2015092070 A1, June 25th 2015
- A.19 M. Madsen, D. J. Perreault, A. Knott and M. A.E. Andersen: "Outphasing Control of Gallium Nitride based Very High Frequency Resonant Converters", Accepted for presentation at IEEE COMPEL 2015 with subsequent publishing in the conference proceedings.

- A.20 M. Kovacevik and M. Madsen: "Burst Mode Control", WO2015128398 A1, September 3rd 2015
- A.21 M. Madsen and M. Kovacevik: "On and Off Controlled Resonant dc-dc Power Converter", WO2015128397 A1, September 3rd 2015
- A.22 M. Madsen: Confidential, DTU ref 95513

The full length versions of the papers and the four of the patent applications are included on the following pages.

A.1 A. Knott, T. M. Andersen, P. Kamby, M. Madsen, M. Kovacevic, M. A.E. Andersen:
"On the Ongoing Evolution of Very High Frequency Power Supplies",
IEEE Applied Power Electronics Conference and Exposition, Long Beach, CA, March 2013. Conference Proceedings p2514-2519.

On the Ongoing Evolution of Very High Frequency Power Supplies

Arnold Knott*, Toke M. Andersen*, Peter Kamby*, Mickey P. Madsen*, Milovan Kovacevic*, Michael A.E. Andersen*

*Technical University of Denmark Ørsteds Plads, bygning 349 2800 Kongens Lyngby Denmark Email: akn@elektro.dtu.dk

Abstract—The ongoing demand for smaller and lighter power supplies is driving the motivation to increase the switching frequencies of power converters. Drastic increases however come along with new challenges, namely the increase of switching losses in all components. The application of power circuits used in radio frequency transmission equipment helps to overcome those. However those circuits were not designed to meet the same requirements as power converters. This paper summarizes the contributions in recent years in application of very high frequency (VHF) technologies in power electronics, describes the remaining challenges and shows results of the recent advances, among others a 120MHz, 9 W LED driver with 89 % efficiency.

I. INTRODUCTION

The continuing trend of miniaturization in industrial and consumer electronics is continuously driving a demand for smaller power supplies. Weight and cost reduction demands accompany this trend. Within power supplies the major size, weight and cost drivers are typically the passive components. Increasing the switching frequency of power converters can reduce the size, weight and therefore the cost of those. For substantial size and weight reduction, the switching frequencies are increased up to the very high frequency (VHF) band (30 MHz to 300 MHz), which leads to a merge in circuit technologies used in radio frequency transmitters [1]–[6] and the classical power electronics circuits.

The VHF amplifiers are designed for DC-AC conversion, where the AC simultaneously is the switching frequency. Generally those circuits [1], [2] drive a known load impedance, typically a 50 Ω antenna. The most efficient standard representatives of radio frequency amplifiers are class-E [3], [4] and class-F [5], [6], where Class-E applies zero-voltageswitching (ZVS) and Class-F applies zero-current-switching (ZCS) techniques. Similarly to switch-mode power supplies, those VHF amplifiers convert the constant supply voltages into a high-frequent voltage by operating power semiconductors in the cut-off or saturation region only. The major difference is, that VHF amplifiers do not convert the energy back into a constant voltage or current level.

Numerous research works have been published [7]–[18], filling this gap and making VHF technologies available for power electronics. This paper describes the individual contributions of those in greater detail. However there are still some challenges left, before VHF switch-mode power supplies can relieve their advantages for products in industrial and consumer electronics.

This paper elaborates on the remaining challenges based on previous work and characterizes them in Section II. Section III describes the most recent advances, showing prototypes and measurement results. Section IV concludes the paper.

II. CHALLENGES OF VHF CONVERTERS

VHF operation of power supplies differs from submegahertz operated power supplies (here called traditional power converters) mainly by the following subjects:

- Electronic components, both active and passive,
- Circuit architectures for power stages and control parts,
 Adjacent behavior, such as electromagnetic compatibility
- (EMC) and mechanics.

A. Components

Especially inductive components are size, weight and cost optimization limitations in nowadays power circuits. Simultaneously VHF converters provide a major opportunity to overcome those.

Among the challenges are core losses, skin and proximity effect [19]–[25]. Another challenge within passive components for VHF is the creation of a galvanic isolation barrier [26], [27].

Despite passive components also active components, i.e. the power semiconductors, need to fulfill other requirements than in usual power supplies [28]–[30]. The parasitic components have a big influence on the design of the overall converter, as they are part of the design parameters. Unlike traditional power stages, the parasitic elements are therefore not considered undesired, but form an integral part of the stage. An example is the output capacitance C_{oss} of the power semiconductor in a Class-E based power supply. According to [17] it is dependent on output power P_{out} , input voltage V_{in} and switching frequency f_{sw} as shown in equation 1.

$$P_{out} = 2\pi^2 f_{sw} C_{oss} V_{in}^2 \tag{1}$$

B. Architectures

Where traditional power electronics circuits use square wave gate drive signals, the presented VHF converters so far utilized sinusoidal gate drive [16], [31]–[33]. This is mainly due to the input capacitance C_{iss} of VHF power semiconductors, which require a high peak current at extremely high speed. To consider the drive trapezoidal, the rise and fall times have to be less than 1 ns [33]. A trapezoidal or square wave drive would minimize the time of the power switch in linear operation and therefore decreases the losses.

The degrees of freedom in terms of modulation principles are less for VHF converters. Whereas power electronics circuits usually use pulse width modulation or phase modulation, the VHF converters efficiency is dependent on those parameters. Therefore they need to be adjusted statically to avoid losses by leaving the ZVS (or ZCS) range. A way to get around this is to apply burst mode control [15], [31], [34]. This method however introduces another low frequency component in the spectrum, which has to be buffered or filtered at both the inand output of the converter. A requirement that enforces the use of bulky components and therefore is counterproductive to the intended advantages of VHF converters in the first place. While the VHF converters offer good possibilities for fast transient regulations, their low frequency control performance is limited by intrinsic bandpass behaviors through serial capacitors. Even though some rectifiers are available with parallel capacitances and impedance transformation [17], [35], more suitable architectures are missing. Thereby it needs to be taken into account, that the original VHF power circuits are designed to match a defined load (typically the impedance of the antenna) and therefore impedance transformation circuits can be realized in a passive way. Power converters however are connected to highly varying loads, i.e. load circuit in idle - drawing no energy from the supply - and full load demanding the maximum output from the supply. Therefore active and lossless impedance matching circuits are required. Having such circuits at hand opens for utilization of the high gain bandwidth in VHF converters for line and load regulation.

C. Adjacencies

Lastly the interaction of VHF converters with its physical environment is different than the one of traditional power converters.

On the one hand, the electromagnetic interaction between circuits increases, the higher the relevant frequencies are [36]–[39]. Fields are distributed easier both inside the converter and to its surroundings. The electrical behavior also becomes highly dependent on electromechanical interfaces, such as cooling and housing. However the harmonics of the resonant waveforms are falling faster, than the harmonics in hard switched traditional power converters [18]. Also the harmonics of the fundamental switching frequency are spaced wider. That means the distance can be used to place strategically important EMC bands, dependent on the application.

On the other hand, the carefully adjusted operating points of VHF converters (for efficiency purposes) are highly dependent

on temperature [17], [18]. Adaptive mechanisms for ensuring optimal operation over industry standard temperature ranges are yet to come.

III. RECENT ADVANCES

Despite those challenges recent research results enhanced the state-of-the art in VHF converters and gives hope to overcome the remaining challenges.

The Class-E based power circuits allow for a second degree of soft switching. Despite turning the power switches on, when the voltage across them is zero (ZVS) or off, when the current through them is zero (ZCS), also the derivatives of these signals are taken into account. This is called ZdVS and ZdCS respectively. The technique has been applied to power converters in [17]. Figure 1 shows the full schematic of the power part of the self-oscillating VHF converter (DC-DC) from [17], [32].

Figure 2 shows the simulated waveforms of this converter, where v_s and i_s are the voltage and the current across and through the switch and v_D and i_D are voltage and current across and through the rectifier diode. v_G is the control signal of the power switch and V_o and V_i are input and output voltages of the converter. The top graph v_s visualizes the optimization of the converter for both ZVS and ZdVS.

Figure 3 is a photograph of the implementation of this converter. The overall efficiency of the 97 MHz converter is 55 %.

Due to the tight adjustment of the turn on instance of the power switch for achieving ZVS and ZdVS the degrees of freedom in this converter are low. That limits the input and output voltage ranges. Furthermore the efficiency is not acceptable. In this case, the majority of the losses are due to conduction losses in the power semiconductors.

Suboptimal operation of Class-E converters as described in [4] opened for higher degrees of freedom in the design of Class-E based DC-DC converters. This means, that the ZdVS condition is only fulfilled under nominal load conditions and only ZVS is fulfilled otherwise. The effects of these operation mode as described in [40] has been extended in [18] to LED lighting applications.

Furthermore [18] provides a detailed analysis of the power components parasitics and the effect of their nonlinearities. The most relevant parasitics of the power switch are the input and output capacitances. They are typically highly nonlinear. Figure 4 shows the relative voltage stress of the power switch



Fig. 1: Complete schematic of a self-oscillating VHF converter [32] with LED load.



Fig. 2: Simulation of waveforms for a ZVS and ZdVS Class-E based converter from [17].



Fig. 3: Photograph of the self-oscillating VHF converter from [32].

as a function of time and junction potential.

The other components of the power stage have been investigated in [18] as well. Thereby most focus is on the inductors, as these are the most volume consuming parts, have the biggest weight and typically a big impact on the overall price of the converter. Therefore the inductors have been integrated as toroids into the printed circuit board (PCB). This process



Fig. 4: Voltage stress of the power switch in relation to DC input voltage for a nonlinear output capacitance from [18]. V_{bi} is the junction potential of the process.

is described in [41] and Figure 5 shows the principle.

The resulting converter waveform in the optimal and suboptimal operating regions are shown in Figure 6. The converters efficiency is in the same area as the previous presented. For dealing with the efficiency challenge, [42] compared a number of power switches both in simulation and experiment. Figure 7 shows photographs of the implementations. On top of that an effective line- and load regulation scheme was implemented in those.

Figure 8 shows the implementation of the final prototype with 70 MHz switching frequency. The voltage step-down ratio of the converters is 10 and the output power range is between 1



Fig. 5: PCB integrated inductor from [41]. The copper contained in the PCB is shown in red. The blue arrows mark the magnetic field.



Fig. 6: Measurements of gate-source and drain-source voltages V_{gs} and V_{ds} of the power switch and the turn-on instances. Note that the drain-source voltage has an offset of -0.5 V, due to the oscilloscopes offset.



Fig. 7: Photograph of numerous prototypes for comparing measured efficiency with simulations [42].

and 4W at an efficiency within this range beyond 70 %. Additionally the self-oscillating principle from [17], [32] was applied to an interleaved Class-E converter in [43], resulting into a significant efficiency improvement. The complete schematic is shown in Figure 9. The realized converter is switching at 120 MHz, i.e. beyond the FM band, converts



Fig. 8: Photograph of a closed loop low-power VHF converter with an efficiency beyond 70 % from [42]. The TO220 components on the upper left is the dummy load resistance.



Fig. 9: Full schematic of interleaved Class-E converter from [43].

an input voltage between 6 and 9 V into an output current between 0.4 and 0.5 A and has an efficiency between 80 and 89 % within this operation range. The output power range is 3 to 9 W and the converter is built for LED drive. Figure 10 shows both a SPICE based simulation and a the measurement of the power switches voltage waveforms.

IV. CONCLUSION

The merge of techniques used in radio communication electronics and power electronics was pointed out. The development through the previous decades has been revisited and recent developments were summarized. Remaining challenges and the latest advances were described. The implementations of numerous VHF converters were presented. Among them are low-power, high-step-down converters with a switching frequency of 70 MHz and an efficiency beyond 70 % as well as a 120 MHz, 9 W LED driver with an efficiency up to 89 %. Both converters maintain high efficiencies over a wide load range.

REFERENCES

 A. Grebennikov and N.O. Sokal. Switchmode RF power amplifiers. Communications engineering series. Elsevier/Newnes, 2007, ISBN: 9780750679626.



Fig. 10: Drain-source waveforms of the two power switches in the interleaved converter from [43].

- [2] M. Kazimierczuk. *RF power amplifiers*. Wiley, 2008, ISBN: 9780470779460.
- [3] N. O. Sokal, A. D. Sokal. Class E-A new class of high-efficiency tuned single-ended switching power amplifiers. *IEEE Journal of Solid-State Circuits*, SC-10(3):168–176, June 1975.
- [4] N. O. Sokal. Class-E High-Efficiency RF/Microwave Power Amplifiers: Principles of Operation, Design Procedures, and Experimental Verification. Analog Circuit Design, Springer, 2003.
- [5] Raab, F.H. Maximum efficiency and output of class-F power amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 49(6):1162–1166, June 2001.
- [6] Raab, F.H. Class-F power amplifiers with maximally flat waveforms. *IEEE Transactions on Microwave Theory and Techniques*, 45(11):2007–2012, November 1997.
- [7] David A. Jackson. Design and Characterization of a Radio-Frequency dc/dc PowerConverter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [8] James Raymond Warren III. Cell Modulated dc/dc Converter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [9] Robert C. N. Pilawa-Podgurski. Design and Evaluation of a Very High Frequency dc/dc Converter. Master's thesis, Massachusetts Institute of Technology, 2007.
- [10] Anthony Sagneri. Design of a Very High Frequency dc-dc Boost Converter. Master's thesis, Massachusetts Institute of Technology, 2007.
- [11] Olivia Leitermann. Radio Frequency dc-dc Converters: Device Characterization, Topology Evaluation, and Design. Master's thesis, Massachusetts Institute of Technology, 2008.
- [12] Joshua W. Phinney. Multi-resonant Passive Components for Power Conversion. PhD thesis, Massachusetts Institue of Technology, May 2005.
- [13] Juan Rivas. Radio Frequency dc-dc Power Conversion. PhD thesis, Massachusetts Institue of Technology, September 2006.
- [14] Yehui Han. Circuits and Passive Components for Radio-Frequency Power Conversion. PhD thesis, Massachusetts Institue of Technology, February 2010.

- [15] Perreault, D.J.; Jingying Hu; Rivas, J.M.; Yehui Han; Leitermann, O.; Pilawa-Podgurski, R.C.N.; Sagneri, A.; Sullivan, C.R. Opportunities and Challenges in Very High Frequency Power Conversion. *Annual IEEE Applied Power Electronics Conference and Exposition*, (24):1–14, 2009.
- [16] Rivas, J.M. Wahby, R.S. Shafran, J.S. Perreault, D.J. New Architectures for Radio-Frequency DC-DC Power Conversion. *IEEE Transactions on Power Electronics*, Vol. 21(2):380–393, March 2006.
- [17] Andersen Toke. Master Thesis: Radio Frequency Switch-Mode Power Supply. Master's thesis, Technical University of Denmark, 2010.
- [18] Peter Kamby. Master Thesis: High Efficiency Radio Frequency Switchmode Power Supply for LED Applications. Master's thesis, Technical University of Denmark, 2011.
- [19] Shanshan Lu; Yuqin Sun; Goldbeck, M.; Zimmanck, D.R.; Sullivan, C.R. 30-MHz Power Inductor Using Nano-Granular Magnetic Material. *IEEE Power Electronics Specialists Conference*, pages 1773–1776, June 2007.
- [20] Sullivan, C.R.; Weidong Li; Prabhakaran, S.; Shanshan Lu. Design and fabrication of low-loss toroidal air-core inductors. *IEEE Annual Power Electronics Specialists Conference*, (38th):1754–1759, June 2007.
- [21] Jizheng Qiu, Sullivan, C.R. Inductor design for VHF tapped-inductor dc-dc power converters. Applied Power Electronics Conference and Exposition, (26th):142 – 149, March 2011.
- [22] Phinney, J.W.; Perreault, D.J.; Lang, J.H. Radio-Frequency Inverters With Transmission-Line Input Networks. *IEEE Transactions on Power Electronics*, 22(4):1154–1161, July 2007.
- [23] Yehui Han; Cheung, G.; An Li; Sullivan, C.R.; Perreault, D.J. Evaluation of magnetic materials for very high frequency power applications. *IEEE Power Electronics Specialists Conference*, pages 4270–4276, 2008.
 [24] Zirath, H.; Rutledge, D.; An LDMOS VHF class E power amplifier
- [24] Zirath, H.; Rutledge, D.; An LDMOS VHF class E power amplifier using a high Q novel variable inductor. *IEEE MTT-S International Microwave Symposium*, Vol. 1:367–370, June 1999.
- [25] M.K. Kazimierczuk. High-Frequency Magnetic Components. John Wiley & Sons, 2009, ISBN: 9780470714539.
- [26] Bowman, W.C.; Balicki, F.T.; Dickens, F.T.; Honeycutt, R.M.; Nitz, W.A.; Strauss, W.; Suiter, W.B.; Ziesse, N.G. A resonant DC-to-DC converter operating at 22 Megahertz. *Applied Power Electronics Conference and Exposition*, (3):3–11, 1988.
- [27] Sagneri, A.D.; Anderson, D.I.; Perreault, D.J. Transformer synthesis for VHF converters. *International Power Electronics Conference (IPEC)*, pages 2347–2353, June 2010.
- [28] Warren, J.R.; Rosowski, K.A.; Perreault, D.J. Transistor Selection and Design of a VHF DC-DC Power Converter. *IEEE Transactions on Power Electronics*, Vol. 23(1):27–37, January 2008.
- [29] Spiazzi, G.; Mattavelli, P.; Rossetto, L. Effects of Parasitic Components in High-Frequency Resonant Drivers for Synchronous Rectification MOSFETs. *IEEE Transactions on Power Electronics*, Vol. 23(4):2082– 2092, July 2008.
- [30] Kuang Sheng ; Yongxi Zhang ; Liangchun Yu ; Ming Su ; Zhao, J.H. . High-Frequency Switching of SiC High-Voltage LJFET. *IEEE Transactions on Power Electronics*, Vol. 24(1):271–277, January 2009.
- [31] Pilawa-Podgurski, R.; Sagneri, A.D.; Rivas, J.M.; Anderson, D.I.; Perreault, D.J. Very-High-Frequency Resonant Boost Converters. *IEEE Transactions on Power Electronics*, 24(6):1654–1665, June 2009.
- [32] Andersen, T.M.; Christensen, S.K.; Knott, A.; Andersen, M.A.E. A VHF class E DC-DC converter with self-oscillating gate driver. *Applied Power Electronics Conference and Exposition (APEC)*, pages 885–891, March 2011.
- [33] Tatsuta, T. ; Ishitani, Y. ; Suetsugu, T. Gate power loss of class E amplifier with rectangular wave gate drive. *TENCON 2010 IEEE Region* 10 Conference, pages 1784–1787, November 2010.
- [34] Rivas, J.M.; Jackson, D.; Leitermann, O.; Sagneri, A.D.; Yehui Han; Perreault, D.J. Design Considerations for Very High Frequency dc-dc Converters. *Power Electronics Specialists Conference*, pages 1–11, June 2006.
- [35] M.K. Kazimierczuk and D. Czarkowski. *Resonant Power Converters*. John Wiley & Sons, 2011, ISBN: 9780470905388.
- [36] Liu, D.H.; Jiang, J.G. High frequency characteristic analysis of EMI filter in switch mode power supply (SMPS). *Power Electronics Specialists Conference*, Vol. 4:2039–2043, June 2002.
- [37] Wang, C.P.; Liu, D.H.; Jiang Jianguo. Study of coupling effects among passive components used in power electronic devices. *Power Electronics* and Motion Control Conference, Vol. 3:1500–1504, August 2004.
- [38] Liyu Yang; Bing Lu; Wei Dong; Zhiguo Lu; Ming Xu; Lee, F.C.; Odendaal, W.G. Modeling and characterization of a 1 KW CCM PFC

converter for conducted EMI prediction. Applied Power Electronics Conference and Exposition, Vol. 2:763–769, February 2004.
[39] Shuo Wang. Characterization and Cancellation of High-Frequency

- [39] Shuo Wang. Characterization and Cancellation of High-Frequency Parasitics for EMI Filters and Noise Separators in Power Electronics Applications. PhD thesis, Virginia Polytechnic Institute and State University, Blacksburg, Virgina, May 2005.
- [40] Redl, Richard; Molnar, Bela; Sokal, Nathan O. Class E Resonant Regulated DC/DC Power Converters: Analysis of Operations, and Experimental Results at 1.5 MHz. *IEEE Transactions on Power Electronics*, PE-1(2):111–120, 1986.
- [41] P. Kamby; A. Knott, M. A.E. Andersen. Printed Circuit Board Integrated Toroidal Radio Frequency Inductors. *37th Annual Conference on IEEE Industrial Electronics Society - IECON 2011*, October, 2012.
 [42] Mickey Madsen. Master Thesis: Offline Very High Frequency Power
- [42] Mickey Madsen. Master Thesis: Offline Very High Frequency Power Converters. Master's thesis, Technical University of Denmark, 2012.
 [43] Kovacevic, Milovan; Knott, Arnold; Andersen, Michael A. E. Intertional Control Contr
- [43] Kovacevic, Milovan; Knott, Arnold; Andersen, Michael A. E. Interleaved Self-Oscillating Class E Derived Resonant DC/DC Converters. *International Conference on Electrical and Computer Systems*, August 2012.

A.2 A. Knott, T. M. Andersen, P. Kamby, J. A. Pedersen, M. Madsen, M. Kovacevic, M. A.E. Andersen:

"Evolution of Very High Frequency Power Supplies",

IEEE Journal of Emerging and Selected Topics in Power Electronics, September 2014, vol.2, no.3, pp.386-394. This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/JESTPE.2013.2294798, IEEE Journal of Emerging and Selected Topics in Power Electronics

IEEE JOURNAL OF EMERGING AND SELECTED TOPICS IN POWER ELECTRONICS, VOL . ?, NO. ?, NOVEMBER 2013

Evolution of Very High Frequency Power Supplies

Arnold Knott^{*} Member, IEEE, Toke M. Andersen^{*} Student Member, IEEE, Peter Kamby^{*}, Jeppe A. Pedersen^{*}, Mickey P. Madsen^{*} Student Member, IEEE, Milovan Kovacevic^{*} Student Member, IEEE, Michael A.E. Andersen^{*}

Member, IEEE *Technical University of Denmark

Ørsteds Plads, bygning 349, 2800 Kongens Lyngby, Denmark

Telefon: +45 45 25 34 90, Email: akn@elektro.dtu.dk

(Submission to "Special Issue on Miniaturized Power Electronics Systems, 2013")

Abstract—The ongoing demand for smaller and lighter power supplies is driving the motivation to increase the switching frequencies of power converters. Drastic increases however come along with new challenges, namely the increase of switching losses in all components. The application of power circuits used in radio frequency transmission equipment helps to overcome those. However those circuits were not designed to meet the same requirements as power converters. This paper summarizes the contributions in recent years in application of very high frequency (VHF) technologies in power electronics, shows results of the recent advances and describes the remaining challenges. The presented results include a self-oscillating gate-drive, air core inductor optimizations, an offline LED driver with a power density of 8.9 W/cm³ and a 120 MHz, 9 W DC powered LED driver with 89 % efficiency as well as a bidirectional VHF converter. The challenges to be solved before VHF converters can be used effectively in industrial products are within those three categories: components, circuit architectures and reliability testing.

Index Terms—VHF circuits, power conversion, DC-DC power converters, resonant inverters, zero voltage switching

I. INTRODUCTION

The continuing trend of miniaturization in industrial and consumer electronics is continuously driving a demand for smaller power supplies. Weight and cost reduction demands accompany this trend. Within power supplies the major size, weight and cost drivers are typically the passive components. Increasing the switching frequency of power converters can reduce the size, weight and therefore the cost of those. For substantial size and weight reduction, the switching frequency (VHF) band (30 MHz to 300 MHz), which leads to a merge in circuit technologies used in radio frequency transmitters [1]–[6] and the classical power electronics circuits.

The VHF amplifiers are designed for DC-AC conversion, where the AC simultaneously is the switching frequency. Generally those circuits [1], [2] drive a known load impedance, typically a 50 Ω antenna. Traditionally the topologies used for those circuits have been characterized as classes with running labels following the alphabet. Class-A, class-B and class-C are described in [2], [7]. These classes are characterized through the relative amount of time, the power transistor is conducting the load current with respect to the period of the VHF signal. For class-A the transistor conducts the load current 50 % of the time. Class-B operates between 25 % and 50 % and class-C between 0 % and 25 %. This leads to theoretical maximum achievable efficiencies of 50 %, up to 78.5 % and up to 100 % for class-A, B and C respectively. Their power electronics counter parts are linear regulators. Class-D is described in [8] and the first power circuit topology, that allows for theoretical 100 % efficiency under all operating conditions. The equivalent are strictly all hard-switched power converters. Class-E as described in [3], [4] and class-F as demonstrated in [5], [6] correspond to all power converters, that apply zero voltage switching (ZVS) and zero-current switching (ZCS) techniques respectively.

1

Similarly to switch-mode power supplies, those VHF amplifiers convert the constant supply voltages into a high-frequent voltage by operating power semiconductors in the triode region only. The major difference is that VHF amplifiers do not convert the energy back into a constant voltage or current level. Numerous research works have been published [9]–[20], filling this gap and making VHF technologies available for power electronics. This paper describes the individual contributions of those in greater detail. However there are still some challenges left, before VHF switch-mode power supplies can relieve their advantages for products in industrial and consumer electronics.

This paper elaborates on the most recent advances, showing prototypes and measurement results in section II. Section III describes the remaining challenges based on previous work and characterizes them. Section IV concludes the paper.

II. RECENT ADVANCES

Recent research results enhanced the state-of-the art in VHF converters. Most of the work in recent years has focused on class-E derived topologies.

A. Optimal operation

The class-E based power circuits allow for a second degree of soft switching. Despite turning the power switches on, when the voltage across them is zero (ZVS), also the derivatives of these signals are taken into account. This is called ZdVS and ZdCS respectively. The technique has been applied to power converters in [19]. The schematic in Fig. 1 shows the adoption of the principles of a class-E oscillator, e.g. shown in [21]–[23], to a class-E based power self-oscillating VHF converter (DC-DC) [19], [24]. A converter achieving both ZVS and ZdVS at all times operates in optimal mode.

Other implementation replaced either the resonant tank [25],

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/JESTPE.2013.2294798, IEEE Journal of Emerging and Selected Topics in Power Electronics

2

IEEE JOURNAL OF EMERGING AND SELECTED TOPICS IN POWER ELECTRONICS, VOL . ?, NO. ?, NOVEMBER 2013



Fig. 1: Schematic of a self-oscillating VHF converter [24] with LED load.

[26] or the input inductor [11], [27] with a transmission line. The resulting waveforms of this circuit have been reported in e.g. [28]–[33] and Fig. from 2 repeats the simulated waveforms of this converter, where v_s and i_s are the voltage and the current across and through the switch and v_D and i_D are voltage and current across and through the rectifier diode. v_G is the control signal of the power switch and V_o and V_i are input and output voltages of the converter. The top graph v_s visualizes the optimization of the converter for both ZVS and ZdVS.

Fig. 3 is a photograph of the implementation of this converter. The overall efficiency of the 97 MHz converter is 55 %.

The advantage of this converter is, that it is based on a widely documented circuit topology from the communication electronics applications. As implemented here, it also provides means of output regulation. The downside is the voltage stress across the power switch, 3.6 times higher as in hard-switched converters.

B. Suboptimal operation

Due to the tight adjustment of the turn on instance of the power switch for achieving ZVS and ZdVS the degrees of freedom in this converter are low. That limits the input and output voltage ranges. Furthermore the efficiency is not acceptable. In this case, the majority of the losses are due to conduction losses in the power semiconductors, which are due to the on-resistance of the power switch. As the gate voltage is not significantly higher than the threshold voltage, the devices minimum on-resistance could not be achieved.

Suboptimal operation of class-E converters as described in [4] opened for higher degrees of freedom in the design of class-E based DC-DC converters. This means that the ZdVS condition is only fulfilled under nominal load conditions and only ZVS is fulfilled otherwise. The resulting converter waveform in the optimal and suboptimal operating regions are shown in Fig. 4. The effects of these operation mode as described in [34] have been extended in [20] to LED lighting applications.

Note that the body diode of the MOSFET is conducting in the beginning of the MOSFETs conduction period. This is due to wrong timing in the turn-on of the power device. The energy lost in the body diode ruins the efficiency of this particular converter.

Furthermore [20] provides a detailed analysis of the power components parasitics and the effect of their nonlinearities. The basis for this analysis has been, among others, laid in



Fig. 2: Simulated waveforms for a ZVS and ZdVS class-E based converter from [19].



Fig. 3: Photograph of the self-oscillating VHF converter from [24].

[35], [36] for the analysis of class-E amplifiers, which is fully applicable to class-E based power converters when tuning the rectifier to act as a an ohmic load. The most relevant parasitics of the power switch are the input and output capacitances. The later is the most critical for the design of the converter.

Copyright (c) 2013 IEEE. Personal use is permitted. For any other purposes, permission must be obtained from the IEEE by emailing pubs-permissions@ieee.org.


Fig. 4: Measurements of gate-source and drain-source voltages V_{gs} and V_{ds} of the power switch and the turn-on instances. Note that the drain-source voltage has an offset of -0.5 V, due to the oscilloscopes offset.

Simultaneously the output capacitance is highly nonlinear, which was taken into account in the analysis in [20]. There the nonlinearity of the output capacitance C_{ds} is modeled with (1)

$$C_{ds}(V_c) = \frac{C_{j0}}{(1 + \frac{V_c}{V_{bi}})^{\gamma}},$$
(1)

where

 C_{j0} is the junction capacitance at 0 V,

- V_{bi} is the built-in junction potential, typically 0.5 0.9 V [29],
- γ is the junction sensitivity or gradual coefficient. Typically $\gamma = 1/3$ for gradient junctions, while $\gamma = 0.5$ for abrupt junctions [1] hence junction diodes [29], and
- v is the junction voltage.

This results in a voltage waveform V_c of the power switch as a function of the converters input current I_{in} and the above



3

Fig. 5: Voltage waveform of the power switch in relation to DC input voltage for a nonlinear output capacitance from [20]. V_{bi} is the junction potential of the process.

output capacitances parameters as given in (2).

$$V_{c} = V_{bi} \left(\left[\frac{I_{in}(1-\gamma)}{\omega C_{j0} V_{bi}} (\omega t - \frac{3\pi}{2} - \frac{\pi}{2} \cos \omega t - \sin \omega t) + 1 \right]^{\frac{1}{1-\gamma}} - 1 \right)$$
(2)

Fig. 5 shows the relative voltage waveform of the power switch as a function of time and junction potential V_{bi} for a junction sensitivity of $\gamma = 0.5$.

The remaining components of the power stage have been investigated in [20] as well. Thereby most focus is on the inductors, as these are the most volume consuming parts, have the biggest weight and typically a big impact on the overall price of the converter. Therefore the inductors have been integrated as toroids into the printed circuit board (PCB). This process is described in [37] and Fig. 6 shows the principle.

A power stage has been designed to operate in suboptimal mode under consideration of the power switches nonlinear output capacitance. The converters efficiency is in the same area as the one presented in II-A and again limited by a high on-resistance, which is due to a low gate drive voltage.

While giving up on the single operating point operation in optimal operation mode, the suboptimal operating converters theoretically allows for different conduction angle operation on the cost of tighter timing to operate in ZVS.

C. Class-E based SEPIC converter

For dealing with the efficiency challenge, [38] compared a number of power switches both in simulation and experiment. Furthermore multiple air-core inductors where calculated, designed and implemented. An extraction is shown in Fig. 8. The prototypes reach Q-values beyond 100 and resonance frequencies up to 340 MHz. Fig. 9 shows a photograph of the implemented converters. On top of that an effective lineand load regulation scheme was realized in those. The designs where verified in a SEPIC converter (Fig. 7) [39], based on the topologies presented in [40], achieving a power density of

4

IEEE JOURNAL OF EMERGING AND SELECTED TOPICS IN POWER ELECTRONICS, VOL . ?, NO. ?, NOVEMBER 2013



Fig. 6: PCB integrated inductor from [37]. The cross-section of the PCB toroid and the resulting flux arrows are shown.



Fig. 7: Schematic of a class-E based SEPIC VHF converter [39].



Fig. 8: Photograph of various air core inductors [38].

 $8.9~{\rm W/cm^3}$ (146 ${\rm W/in^3}$) by switching at 51 MHz for offline LED applications.

Fig. 10 shows the implementation of the final prototype with 70 MHz switching frequency. The voltage step-down ratio of the converters is 10 and the output power range is between 1 and 4W at an efficiency within this range beyond 70 %.

Compared to the above reported converters, the SEPIC converter is not based on an inverter that delivers a sinusoidal output. The later is crucial in telecommunication applications, when using the class-E inverter as a transmitter, but completely unnecessary demand as an intermediate VHF link within a DC/DC power converter. Relaxing this requirement removes the resonant tank inductor, and therefore the resonant tanks bandpass behavior. On the other hand the rectifier can no



Fig. 9: Photograph of numerous prototypes for comparing measured efficiency with simulations [38].



Fig. 10: Photograph of a closed loop low-power VHF converter with an efficiency beyond 70 % from [38]. The TO220 components in the upper left corner are the dummy load resistance.

longer freely be chosen between several topologies, but has to be implemented with a diode, not referenced to ground, which is a disadvantage in some implementation technologies, such as integrated circuits.

D. Interleaved VHF converters

Additionally the self-oscillating principle from [19], [24] was combined with the interleave principle from [41], [42] in [43], resulting into a significant efficiency improvement. Interleaving two converter legs allows furthermore to use the ripple cancelation as described in [44] and applied in [41]. The complete schematic of the open loop implementation is shown in Fig. 11. The realized converter is switching at 120 MHz, i.e. beyond the FM band, converts an input voltage between 6 and 9 V into an output current between 0.4 and 0.5 A and has an efficiency between 80 and 89 % within this operation range. The output power range is 3 to 9 W, corresponding to an output voltage range between 7 V and 20 V. The converter is designed to drive LEDs. Fig. 12 shows both a SPICE based simulation and a the measurement of the power switches voltage waveforms. Fig. 13 shows the efficiency graph of this converter.

KNOTT itet al.: EVOLUTION OF VHF-SMPS



Fig. 11: Full schematic of the open-loop interleaved class-E converter from [43].



Fig. 12: Drain-source waveforms of the two power switches in the interleaved converter from [43].

Interleaved converters allow for input and/or output ripple cancellation, segmented power stages, which enables higher power levels [45]. But those converters suffer from different optimal frequencies due to tolerances for each leg, which either might result in beat tones, when operating each of them at its own optimal resonant frequencies, or a non-optimal operation point with respect to efficiency for all legs, when operating all legs at the same frequency.

E. Bidirectional VHF converter

Replacing the diode in Fig. 1 with a transistor, the class-E amplifier and the class-E synchronous rectifier form a symmetric schematic as shown in Fig. 14. This was realized in [46] and resulted in a bidirectional converter with the same



Fig. 13: Efficiency of a battery driven LED driver switching at 120 MHz [43].



Fig. 14: Schematic of a VHF converter with class-E inverter and synchronous class-E rectifer [46].



Fig. 15: Photograph of a bidirectional VHF converter [46].

conversion ration from both sides. Operating in the forward mode, the transistor M_1 is the power switch, operating in class-E mode, and M_2 is used as synchronous rectifier in class-E operation. In the reverse operating mode, the voltage designated V_{out} is acting as the input voltage and M_2 becomes the inverter switch, while M_1 turns into the synchronous rectifier. The maximum achieved efficiency with this topology was 70 % switching at 30 MHz. A photograph of the prototype and thermal pictures of the converter are shown in Fig. 15 and Fig. 16 respectively.

The bidirectional converter allows for lower conduction losses in the rectifier and allows for two-quadrant operation at the cost of an extra gate, which needs a control signal.

III. CHALLENGES OF VHF CONVERTERS

Lately remaining research challenges have been described in [47], [48] This section is summarizing the remaining challenges common in all above described converters with repect to implementation in products. It is dividing the major

6

IEEE JOURNAL OF EMERGING AND SELECTED TOPICS IN POWER ELECTRONICS, VOL . ?, NO. ?, NOVEMBER 2013



(a) class-E inverter



(b) class-E synchronous rectifier

Fig. 16: Thermal photographs of bidirectional VHF converter in thermal equilibrium [46].

remaining show stoppers into three categories and describes those afterwards with respect to existing products on the power supply market, with switching frequencies below the VHF range.

VHF operation of power supplies differs from sub-megahertz operated power supplies (here called traditional power converters) mainly by the following subjects:

- Electronic components, both active and passive,
- Circuit architectures for power stages and control parts,
- Adjacent behavior, such as electromagnetic compatibility (EMC), mechanics and other reliability tests.

A. Components

Especially inductive components are size, weight and cost optimization limitations in nowadays power circuits. Simultaneously VHF converters provide a major opportunity to overcome those.

Among the challenges are core losses, skin and proximity effect [27], [49]–[54]. For driving further miniaturization of VHF power supplies an obvious next step is to integrate the whole converter in a package (Power Supply in Package (PSiP)) or even on a single chip (Power Supply on Chip (PwrSoC)). The most challenging part for this goal is the integration of the inductors. Great progress has been made and summarized lately in [55], [56]. However realizations of integrated inductors with Q-values beyond 100 in the relevant frequency ranges remain to be seen. Hybrid concepts as shown in [57] might be applicable. Another challenge within passive components for VHF is the creation of a galvanic isolation barrier [58]–[60].

Despite passive components also active components, i.e. the power semiconductors, need to fulfill other requirements than in usual power supplies [61]–[63]. The parasitic components have a big influence on the design of the overall converter, as they are part of the design parameters. Unlike traditional power stages, the parasitic elements are therefore not considered undesired, but form an integral part of the stage. An example is the output capacitance C_{oss} of the power semiconductor in a class-E based power supply. According to [19] it is dependent on output power P_{out} , input voltage V_{in} and switching frequency f_{sw} as shown in (3).

$$P_{out} = 2\pi^2 f_{sw} C_{oss} V_{in}^2 \tag{3}$$

This means that the output capacitance C_{oss} is limiting the maximum switching frequency for a given application, which specifies P_{out} and V_{in}

B. Architectures

Where traditional power electronics circuits use square wave gate drive signals, the presented VHF converters so far utilized sinusoidal gate drive [18], [24], [64], [65]. This is mainly due to the input capacitance C_{iss} of VHF power semiconductors, which require a high peak current at extremely high speed. To consider the drive voltage trapezoidal its rise and fall times have to be less than 1 ns [65]. A trapezoidal or square wave drive would minimize the time of the power switch in linear operation and therefore decreases the losses.

The degrees of freedom in terms of modulation principles are less for VHF converters. Whereas power electronics circuits usually use pulse width modulation or phase modulation, the VHF converters efficiency is dependent on those parameters. Therefore they need to be adjusted statically to avoid losses by leaving the ZVS (or ZCS) range. A way to get around this is to apply burst mode control [17], [64], [66]. This method however introduces another low frequency component in the spectrum, which has to be buffered or filtered at both the inand output of the converter. A requirement that enforces the use of bulky components and therefore is counterproductive to the intended advantages of VHF converters in the first place. While the VHF converters offer good possibilities for fast transient regulations, their low frequency control performance is limited by intrinsic bandpass behaviors through serial capacitors. Even though some rectifiers are available with parallel capacitances and impedance transformation [19], [67], more suitable architectures are missing. Thereby it needs to be taken into account, that the original VHF power circuits are designed to match a defined load (typically the impedance of

the antenna) and therefore impedance transformation circuits can be realized in a passive way. Power converters however are connected to highly varying loads, i.e. load circuit in idle - drawing no energy from the supply - and full load demanding the maximum output from the supply. Therefore active and lossless impedance matching circuits are required. Having such circuits at hand opens for utilization of the high gain bandwidth in VHF converters for line and load regulation.

C. Adjacencies

Lastly the interaction of VHF converters with its physical environment is different than the one of traditional power converters.

On one hand, the electromagnetic interaction between circuits increases, the higher the relevant frequencies are [68]–[71]. Fields are distributed easier both inside the converter and to its surroundings. The electrical behavior also becomes highly dependent on electromechanical interfaces, such as cooling and housing. However the harmonics of the resonant waveforms are falling faster, than the harmonics in hard switched traditional power converters [20]. Also the harmonics of the fundamental switching frequency are spaced wider. That means the distance can be used to place strategically important EMC bands, dependent on the application.

On the other hand, the carefully adjusted operating points of VHF converters (for efficiency purposes) are highly dependent on temperature [19], [20]. Adaptive mechanisms for ensuring optimal operation over industry standard temperature ranges are yet to come.

IV. CONCLUSION

The merge of techniques used in radio communication electronics and power electronics was pointed out. The development through the previous decades has been revisited and recent developments were summarized. Remaining challenges and the latest advances were described. The implementations of numerous VHF converters were presented. Among them are low-power, high-step-down converters with a switching frequency of 70 MHz and an efficiency beyond 70 % as well as a 120 MHz, 9 W LED driver with an efficiency up to 89 %. Both converters maintain high efficiencies over a wide load range.

The remaining challenges, that require solutions before VHF converters can be implemented in numerous industrial applications were found to be within the categorizes components, circuit architectures and reliability testing.

REFERENCES

- A. Grebennikov and N.O. Sokal. Switchmode RF power amplifiers. Communications engineering series. Elsevier/Newnes, 2007, ISBN: 9780750679626.
- [2] M. K. Kazimierczuk. RF Power Amplifiers. Wiley, 2008, ISBN: 9780470779460.
- [3] N. O. Sokal, A. D. Sokal. Class E-A new class of high-efficiency tuned single-ended switching power amplifiers. *IEEE Journal of Solid-State Circuits*, SC-10(3):168–176, June 1975.
- [4] N. O. Sokal. Class-E High-Efficiency RF/Microwave Power Amplifiers: Principles of Operation, Design Procedures, and Experimental Verification. Analog Circuit Design, Springer, 2003.

- [5] Raab, F.H. Maximum efficiency and output of class-F power amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 49(6):1162– 1166, June 2001.
- [6] Raab, F.H. Class-F power amplifiers with maximally flat waveforms. *IEEE Transactions on Microwave Theory and Techniques*, 45(11):2007–2012, November 1997.
- [7] H.L. Krauss, C.W. Bostian, and F.H. Raab. Solid state radio engineering. Wiley, 1980, ISBN: 9780471030188.
- [8] Baxandall, P.J. Transistor sine-wave LC oscillators. Some general considerations and new developments. *Proceedings of the IEE - Part B: Electronic and Communication Engineering*, Vol. 106(16):748–758, May 1959.
- [9] David A. Jackson. Design and Characterization of a Radio-Frequency dc/dc PowerConverter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [10] James Raymond Warren III. Cell Modulated dc/dc Converter. Master's thesis, Massachusetts Institute of Technology, 2005.
- [11] Robert C. N. Pilawa-Podgurski. Design and Evaluation of a Very High Frequency dc/dc Converter. Master's thesis, Massachusetts Institute of Technology, 2007.
- [12] Anthony Sagneri. Design of a Very High Frequency dc-dc Boost Converter. Master's thesis, Massachusetts Institute of Technology, 2007.
- [13] Olivia Leitermann. Radio Frequency dc-dc Converters: Device Characterization, Topology Evaluation, and Design. Master's thesis, Massachusetts Institute of Technology, 2008.
- [14] Joshua W. Phinney. Multi-resonant Passive Components for Power Conversion. PhD thesis, Massachusetts Institue of Technology, May 2005.
- [15] Juan Rivas. Radio Frequency dc-dc Power Conversion. PhD thesis, Massachusetts Institue of Technology, September 2006.
- [16] Yehui Han. Circuits and Passive Components for Radio-Frequency Power Conversion. PhD thesis, Massachusetts Institue of Technology, February 2010.
- [17] Perreault, D.J.; Jingying Hu; Rivas, J.M.; Yehui Han; Leitermann, O.; Pilawa-Podgurski, R.C.N.; Sagneri, A.; Sullivan, C.R. Opportunities and Challenges in Very High Frequency Power Conversion. *Annual IEEE Applied Power Electronics Conference and Exposition*, (24):1–14, 2009.
- [18] Rivas, J.M. Wahby, R.S. Shafran, J.S. Perreault, D.J. New Architectures for Radio-Frequency DC-DC Power Conversion. *IEEE Transactions on Power Electronics*, Vol. 21(2):380–393, March 2006.
- [19] Andersen Toke. Master Thesis: Radio Frequency Switch-Mode Power Supply. Master's thesis, Technical University of Denmark, 2010.
- [20] Peter Kamby. Master Thesis: High Efficiency Radio Frequency Switchmode Power Supply for LED Applications. Master's thesis, Technical University of Denmark, 2011.
- [21] Ebert, Jan; Kazimierczuk, Marian. Class E high-efficiency tuned power oscillator. *Solid-State Circuits, IEEE Journal of*, Vol. 16(2):62–66, April 1981.
- [22] Kazimierczuk, Marian K; Krizhanovski, Vladimir G; Rassokhina, Julia V; Chernov, Dmitrii V. Class-E MOSFET tuned power oscillator design procedure. *Circuits and Systems I: Regular Papers, IEEE Transactions on*, Vol. 52(6):1138–1147, June 2005.
- [23] Krizhanovski, Vladimir G; Chernov, Dmitrii V; Kazimierczuk, Marian K. Low-voltage electronic ballast based on class E oscillator. *Power Electronics, IEEE Transactions on*, Vol. 22(3):863–870, May 2007.
- [24] Andersen, T.M.; Christensen, S.K.; Knott, A.; Andersen, M.A.E. A VHF class E DC-DC converter with self-oscillating gate driver. *Applied Power Electronics Conference and Exposition (APEC)*, pages 885–891, March 2011.
- [25] Smit, MC and Ferreira, JA and Van Wyk, JD. Application of transmission line principles to high frequency power converters. *Power Electronics Specialists Conference PESC'92 Record*, (23rd):1423–1430, 1992.
- [26] Sander, Sverker. Buck and Boost Converters With Transmission Lines. Power Electronics, IEEE Transactions on, 27(9):4013–4020, September 2012.
- [27] Phinney, J.W.; Perreault, D.J.; Lang, J.H. Radio-Frequency Inverters With Transmission-Line Input Networks. *IEEE Transactions on Power Electronics*, 22(4):1154–1161, July 2007.
- [28] Jozwik, Jacek J.; Kazimierczuk, Marian K. Analysis and design of class-E2 DC/DC converter. *IEEE Transactions on Industrial Electronics*, 37(2):173–183, 1990.
- [29] T. Suetsugu, M.K. Kazimierczuk. Analysis and design of class E amplifier with shunt capacitance composed of nonlinear and linear capacitances. *IEEE Transactions on Circuits and Systems I: Regular Papers*, Vol. 51(7):1261–1268, July 2004.

8

IEEE JOURNAL OF EMERGING AND SELECTED TOPICS IN POWER ELECTRONICS, VOL . ?, NO. ?, NOVEMBER 2013

- [30] Kazimierczuk, Marian K.; Jozwik, Jacek. Class E² narrow-band resonant DC/DC converters. *Instrumentation and Measurement, IEEE Transactions on*, Vol. 38(6):1064–1068, December 1989.
- [31] Kazimierczuk, Marian K and Jozwik, Jacek. Resonant dc/dc converter with class-E inverter and class-E rectifier. *Industrial Electronics, IEEE Transactions on*, 36(4):468–478, November 1989.
- [32] García, José A and Marante, Reinel and de las Nieves Ruiz Lavin, María. GaN HEMT class E2 resonant topologies for UHF DC/DC power conversion. *IEEE Transactions on Microwave Theory and Techniques*, Vol. 60(12):4220–4229, December 2012.
- [33] Gutmann, Ronald J. Application of RF circuit design principles to distributed power converters. *Industrial Electronics and Control Instrumentation, IEEE Transactions on*, IECI-27(3):156–164, August 1980.
- [34] Redl, Richard; Molnar, Bela; Sokal, Nathan O. Class E Resonant Regulated DC/DC Power Converters: Analysis of Operations, and Experimental Results at 1.5 MHz. *IEEE Transactions on Power Electronics*, PE-1(2):111–120, 1986.
- [35] Chudobiak, Michael J. The use of parasitic nonlinear capacitors in class E amplifiers. *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on*, Vol. 41(12):941–944, December 1994.
- [36] Suetsugu, Tadashi; Kazimierczuk, Marian K. Comparison of Class-E Amplifier With Nonlinear and Linear Shunt Capacitance. *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on*, Vol. 50(8):1089–1097, August 2003.
- [37] P. Kamby, A. Knott, M. A.E. Andersen. Printed Circuit Board Integrated Toroidal Radio Frequency Inductors. 37th Annual Conference on IEEE Industrial Electronics Society - IECON 2011, October, 2012.
- [38] Mickey Madsen. Master Thesis: Offline Very High Frequency Power Converters. Master's thesis, Technical University of Denmark, 2012.
- [39] M. P. Madsen, A. Knott, M.A.E. Andersen. Very high frequency resonant DC/DC converters for LED lighting. *IEEE Proc. of Applied Power Electronics Conference and Exposition*, (28th):835–839, March 2013.
- [40] Hu, Jingying; Sagneri, Anthony D; Rivas, Juan M; Han, Yehui; Davis, Seth M; Perreault, David J. High-Frequency Resonant SEPIC Converter with Wide Input and Output Voltage Ranges. *Power Electronics, IEEE Transactions on*, Vol. 27(1):189–200, January 2012.
- [41] Mikolajewski, Miroslaw; Kazimierczuk, Marian K. Zero-voltage-ripple rectifiers and DC/DC resonant converters. *Power Electronics, IEEE Transactions on*, Vol. 8(1):12–17, January 1993.
- [42] Schrom, Gerhard and Hazucha, Peter and Hahn, Jaehong and Gardner, Donald S and Bloechel, Bradley A and Dermer, Greg and Narendra, Siva G and Karnik, Tanay and De, Vivek. A 480-MHz, multi-phase interleaved buck DC-DC converter with hysteretic control. *Power Electronics Specialists Conference, 2004. PESC 04. 2004 IEEE 35th Annual*, (6):4702–4707, 2004.
- [43] Kovacevic, Milovan; Knott, Arnold; Andersen, Michael A. E. Interleaved Self-Oscillating Class E Derived Resonant DC/DC Converters. *International Conference on Electrical and Computer Systems*, August 2012.
- [44] Cuk, S.; Erickson R.W. A conceptually new high-frequency switch-mode power amplifier technique eliminates current ripple. *Proc. of Powercon* 5, the Fifth National Solid-State Power Conversion Conference, May 1978.
- [45] Glaser, John S and Nasadoski, Jeffrey and Heinrich, Richard. A 900W, 300V to 50V Dc-dc Power Converter with a 30MHz Switching Frequency. Applied Power Electronics Conference and Exposition, 2009. APEC 2009. 24th Annual IEEE, pages 1121–1128, 2009.
- [46] Jeppe Arnsdorf Pedersen. Master Thesis: Bidirectional Very High Frequency Converter. Master's thesis, Technical University of Denmark, 2013.
- [47] Foley, Raymond and Waldron, Finbarr and Slowey, John and Alderman, Arnold and Narveson, Brian and O'Mathuna, Sean Cian. Technology roadmapping for power supply in package (PSiP) and power supply on chip (PwrSoC). Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE, pages 525–532, 2010.
- [48] Kassakian, John G and Jahns, Thomas M. Evolving and Emerging Applications of Power Electronics in Systems. *Emerging and Selected Topics in Power Electronics, IEEE Journal of*, 1(2):47–58, June 2013.
- [49] Shanshan Lu; Yuqin Sun; Goldbeck, M.; Zimmanck, D.R.; Sullivan, C.R. 30-MHz Power Inductor Using Nano-Granular Magnetic Material. *IEEE Power Electronics Specialists Conference*, pages 1773–1776, June 2007.
- [50] Sullivan, C.R.; Weidong Li; Prabhakaran, S.; Shanshan Lu. Design and fabrication of low-loss toroidal air-core inductors. *IEEE Annual Power Electronics Specialists Conference*, (38th):1754–1759, June 2007.

- [51] Jizheng Qiu, Sullivan, C.R. Inductor design for VHF tapped-inductor dc-dc power converters. *Applied Power Electronics Conference and Exposition*, (26th):142 – 149, March 2011.
- [52] Yehui Han; Cheung, G.; An Li; Sullivan, C.R.; Perreault, D.J. Evaluation of magnetic materials for very high frequency power applications. *IEEE Power Electronics Specialists Conference*, pages 4270–4276, 2008.
- [53] Zirath, H.; Rutledge, D.; An LDMOS VHF class E power amplifier using a high Q novel variable inductor. *IEEE MTT-S International Microwave Symposium*, Vol. 1:367–370, June 1999.
- [54] M.K. Kazimierczuk. High-Frequency Magnetic Components. John Wiley & Sons, 2009, ISBN: 9780470714539.
- [55] Cian Ó Mathúna, Ningning Wang, Santosh Kulkarni, Saibal Roy. Review of Integrated Magnetics for Power Supply on Chip (PwrSoC). *IEEE Transactions on Power Electronics*, Vol. 27(11):4799–4816, November 2012.
- [56] Jizheng Qiu, Sullivan, C.R. Design and Fabrication of VHF Tapped Power Inductors Using Nanogranular Magnetic Films. *IEEE Transactions on Power Electronics*, Vol. 27(12):4965–4975, December 2012.
- [57] H. Schneider, T. Andersen, A. Knott, M.A.E. Andersen. Hybrid winding concepts for toroids. *Energy Conversion Congress and Exibition*, June 2013.
- [58] Bowman, W.C.; Balicki, F.T.; Dickens, F.T.; Honeycutt, R.M.; Nitz, W.A.; Strauss, W.; Suiter, W.B.; Ziesse, N.G. A resonant DC-to-DC converter operating at 22 Megahertz. *Applied Power Electronics Conference and Exposition*, (3):3–11, 1988.
- [59] Sagneri, A.D.; Anderson, D.I.; Perreault, D.J. Transformer synthesis for VHF converters. *International Power Electronics Conference (IPEC)*, pages 2347–2353, June 2010.
- [60] J. Pejtersen, J.D. Mønster, A. Knott. Development and verification of printed circuit board toroidal transformer model. *IEEE Proc. of Applied Power Electronics Conference and Exposition*, (28th):1654–1659, March 2013.
- [61] Warren, J.R.; Rosowski, K.A.; Perreault, D.J. Transistor Selection and Design of a VHF DC-DC Power Converter. *IEEE Transactions on Power Electronics*, Vol. 23(1):27–37, January 2008.
- [62] Spiazzi, G.; Mattavelli, P.; Rossetto, L. Effects of Parasitic Components in High-Frequency Resonant Drivers for Synchronous Rectification MOSFETs. *IEEE Transactions on Power Electronics*, Vol. 23(4):2082– 2092, July 2008.
- [63] Kuang Sheng ; Yongxi Zhang ; Liangchun Yu ; Ming Su ; Zhao, J.H. . High-Frequency Switching of SiC High-Voltage LJFET. *IEEE Transactions on Power Electronics*, Vol. 24(1):271–277, January 2009.
- [64] Pilawa-Podgurski, R.; Sagneri, A.D.; Rivas, J.M.; Anderson, D.I.; Perreault, D.J. Very-High-Frequency Resonant Boost Converters. *IEEE Transactions on Power Electronics*, 24(6):1654–1665, June 2009.
- [65] Tatsuta, T.; Ishitani, Y.; Suetsugu, T. Gate power loss of class E amplifier with rectangular wave gate drive. *TENCON 2010 IEEE Region* 10 Conference, pages 1784–1787, November 2010.
- [66] Rivas, J.M.; Jackson, D.; Leitermann, O.; Sagneri, A.D.; Yehui Han; Perreault, D.J. Design Considerations for Very High Frequency dc-dc Converters. *Power Electronics Specialists Conference*, pages 1–11, June 2006.
- [67] M.K. Kazimierczuk and D. Czarkowski. Resonant Power Converters. John Wiley & Sons, 2011, ISBN: 9780470905388.
- [68] Liu, D.H.; Jiang, J.G. High frequency characteristic analysis of EMI filter in switch mode power supply (SMPS). *Power Electronics Specialists Conference*, Vol. 4:2039–2043, June 2002.
- [69] Wang, C.P.; Liu, D.H.; Jiang Jianguo. Study of coupling effects among passive components used in power electronic devices. *Power Electronics* and Motion Control Conference, Vol. 3:1500–1504, August 2004.
- [70] Liyu Yang; Bing Lu; Wei Dong; Zhiguo Lu; Ming Xu; Lee, F.C.; Odendaal, W.G. Modeling and characterization of a 1 KW CCM PFC converter for conducted EMI prediction. *Applied Power Electronics Conference and Exposition*, Vol. 2:763–769, February 2004.
- [71] Shuo Wang. Characterization and Cancellation of High-Frequency Parasitics for EMI Filters and Noise Separators in Power Electronics Applications. PhD thesis, Virginia Polytechnic Institute and State University, Blacksburg, Virgina, May 2005.

KNOTT itet al.: EVOLUTION OF VHF-SMPS



Arnold Knott (M'10) received the Diplom-Ingenieur (FH) degree from the University of Applied Sciences in Deggendorf, Germany, in 2004. From 2004 until 2009 he has been working with Harman/Becker Automotive Systems GmbH in Germany and USA, designing switch-mode audio power amplifiers and power supplies for automotive applications. In 2010 he earned the Ph.D. degree from the Technical University of Denmark, Kongens Lyngby, Denmark working on a research project under the title "'Improvement of out-of-band Behaviour in

Switch-Mode Amplifiers and Power Supplies by their Modulation Topology". From 2010 to 2013 he was Assistant Professor and since 2013 Associate Professor at the Technical University of Denmark. His interests include switch-mode audio power amplifiers, power supplies, active and passive components, integrated circuit design, acoustics, radio frequency electronics, electromagnetic compatibility and communication systems.



Mickey P. Madsen (S'12) received the B.Sc.E.E. and M.Sc.E.E. degrees from the Technical University of Denmark, Kongens Lyngby, Denmark, in 2009 and 2012, respectively.

9

He is currently working towards a Ph.D. degree in power electronics under the title "Very High Frequency Switch Mode Power Supplies". His research interests includes switch-mode power supplies, resonant inverters/converters, wide band gab semiconductors, solid state (LED) lighting and radio frequency electronics.



Milovan Kovacevic received the B.Sc. and M.Sc. degrees from the University of Belgrade, Serbia, in 2008 and 2010, respectively. He is currently working toward the Ph.D. degree at the Department of Electrical Engineering, Technical University of Denmark, Kgs. Lyngby, Denmark.

His research interests include high-frequency power electronics, resonant and soft-switching techniques, analog and mixed-signal circuit design, and control of power converters.



land

Denmark (DTU), Kgs. Lyngby, Denmark in 2008 and 2010, respectively. His research interests include analysis, design, implementation, and optimization of on-chip power converters in deep submicron CMOS technologies. He is currently pursuing the Ph.D. degree at the Power Electronic Systems Laboratory (PES), Swiss Federal Institute of Technology (ETH) Zürich, Zürich, Switzerland in collaboration with IBM Research Zurich, Rüschlikon, Switzer-

Toke M. Andersen (S'10) received the B.Sc. and

M.Sc. degrees from the Technical University of



Michael A. E. Andersen (M'88) received the M.Sc.E.E. and Ph.D. degrees in power electronics from the Technical University of Denmark, Kongens Lyngby, Denmark, in 1987 and 1990, respectively. He is currently a Professor of power electronics at the Technical University of Denmark. Since 2009, he has been Deputy Director at the Department of Electrical Engineering. He is the author or coauthor of more than 200 publications. His research interests include switch-mode power supplies, piezoelectric transformers, power factor correction, and switch-

mode audio power amplifiers.



Peter Kamby received his B.Sc. and M.Sc. degrees from the Technical University of Denmark (DTU), Kongens Lyngbt, Denmark in 2009 and 2012 respectively. His research interests are very high frequency (VHF) switch-mode power supplies, high current power conversion and pulsed power.



Jeppe A. Pedersen received his B.Sc. and M.Sc. degrees from the Technical University of Denmark (DTU), Kongens Lyngbt, Denmark in 2010 and 2013 respectively. His research interests are very high frequency (VHF) switch-mode power supplies, bidirectional power conversion and LED drivers. Currently Jeppe is research assistant the the Technical University of Denmark.

A.3 M. Madsen, A. Knott, M. A.E. Andersen: "Low Power Very High Frequency Resonant Converter with High Step Down Ratio", IEEE AFRICON, Mauritius, September 2013. Conference Proceedings p1-6.

Low Power Very High Frequency Resonant Converter with High Step Down Ratio

Mickey Madsen, Arnold Knott and Michael A.E. Andersen Department of Electrical Enginneering, Technical University of Denmark Ørsteds Plads, Building 349, 2800 Kgs Lyngby, Denmark Tel.: +45 45253490 Fax.: +45 45880117 Email: {mpma, akn and ma}@elektro.dtu.dk

Abstract—This paper presents the design of a resonant converter with a switching frequency in the very high frequency range (30-300MHz), a large step down ratio and low output power. This gives the designed converters specifications which are far from previous results. The class E inverter and rectifier have been selected for the prototype and the circuits are analyzed and simulated. Three different power stages are implemented based on different design parameters. The first prototype is with a switch with small capacitances, the second one is with a switch with low on resistance and the last one is with a large input inductor. The power stages are designed with the same specs and efficiencies from 60.7 - 82.9% are achieved.

I. INTRODUCTION

When designing power converters it is always a goal to reduce the price and the physical size, i.e. increase the power density. The development of Switch Mode Power Supplies (SMPS) has made it possible to increase the power density significantly, but it is limited by the size of the passive energy storing components (inductors and capacitors). The value and size of these is however dependent on the switching frequency. By increasing the switching frequency it will hence be possible to reduce the size of SMPSs further.

Traditional SMPS topologies like Buck and Boost are hard switching, this means the MOSFET is switching while energy is stored in the output capacitance. The result is that energy is dissipated in the MOSFET every time it turns on. Although this introduces losses in the converter, it is not critical for converters switching at 50 - 400kHz. But when the frequency is increased to the Very High Frequency (VHF) range (30 - 300MHz) the dissipated power get almost 1000 times larger. This amount of energy would ruin the efficiency and require extreme cooling of the MOSFET. This leads to the development of resonant converters.

In order to avoid switching losses and be able to increase the frequency while keeping the efficiency high, new topologies have to be used. For the last two decades (since 1988 [1]) research has been done in order to enable the use of resonant RF amplifiers (inverters) combined with a rectifier for DC/DC converters. With this type of converters it is possible to achieve Zero Voltage Switching (ZVS) and/or Zero Current Switching (ZCS). In this case the MOSFET turns on when the voltage and/or current across/through it is zero. Theoretically this should eliminate switching losses if the switching is done instantaneously and at exactly the right time. This is not

practically achievable, but even with slight deviations from the ideal case very high efficiencies can be achieved.

Increasing the switching frequency has several benefits beside reducing the size. As already mentioned the size of the passive components depends on the switching frequency. This gives a reduction in size, but also in cost as smaller components are generally cheaper. If the frequency is increased enough, some of the components can even be left out as they can be constituted by the parasitic parts of other components (this will be explained further in section II and III). An increase in switching frequency will also make it easier to comply with EMI requirements, as switching harmonics can easily be filtered out by small and cheap filters.

With a switching frequency in the VHF range, it will also be possible to achieve very fast responses. However in order to fully benefit from this, an efficient and fast control loop has to be implemented. This still seems to be a big challenge as some of the best results are still achieved using burst mode (or cell modulation) as in [2], [3]. Due to the high switching frequency the converter will reach steady state after just a few μs , this makes it possible to use an array of small converters and switch them on and off as needed. In this way each converter is designed to operate with a defined load/output. This makes the design much easier as resonant inverters are generally very load depended.

The fact that resonant inverters are load depended, makes it very hard to achieve good performance at varying loads. Furthermore resonant inverters need a large load impedance to operate in the ideal situation (having both ZVS and ZCS). This makes them well suited for boost type converters, but making a buck type is a bit more challenging. The most commonly used way to overcome this challenge is to add an autotransformer at the output, in that way the load impedance seen by the inverter is increased [2], [4], [5]. Another way to achieve low output voltage is to use an array of converters with the input in series and the output in parallel as in [6].

The most commonly used inverter is the class E, however several other topologies exist. Some of the research results are summed up in table I. From the table it is seen that very high efficiencies are achievable for the inverters, up to 97%. However for the complete converters the efficiency drops around 10%, with a maximum at 91%.

From table I it is also seen that the converters have limited

978-1-4673-5943-6/13/\$31.00 ©2013 IEEE

TABLE I Results from previous research

		11	iverters			
Topology	f_s	V _{IN}	VOUT	POUT	η	Year
	[MHz]	[V]	[V]	[W]	[%]	
Class DE	5.3	330	N/A	1154	89	1999 [7]
Class E	1	128	N/A	366	96.6	2006 [8]
Class ϕ_2	1	129	N/A	526	97.1	2006 [8]
Class ϕ_2	30	160	N/A	330	93.7	2007 [9]
Class E	1	129	N/A	322.7	97	2007 [10]
Class E	100	9	N/A	6.8	82	2011 [11]
		Co	nverters			
Topology	f_s	VIN	VOUT	POUT	n	Year
1 00	[MHz]		[V]	[W]	[%]	
Class E	1	20	25	8.9	89	1989 [12]
Class ϕ_2	30	165	33	265	87	2006 [2]
Class E	100	11	12	10	75	2006 [13]
Class ϕ_2	30	150	33	180	84	2008 [4]
Class ϕ_2	10	170	75	250	91	2009 [5]
Class ϕ_2	30	165	33	225	87	2009 [5]
Class ϕ_2	30	330	50	900	79	2009 [6]
Class E	100	12	23.7	1.7	55	2010 [14]
Push-Pull ϕ_2	30	140	65.4	471.9	83.4	2010 [15]
100						
					v	**×
90-					×	<u></u>
					*	
80				×	*/*	
2					/ ×	
F 70						
70-						
						× Inverters
60-			/	·		× Converters
				×		 Linear fit O Thesis aim
50 10 ⁻¹⁰	10 ⁻⁸	10 ⁻⁶	10-4	1(n ⁻²	100 10
		° P	_{DUT} / (V _{IN} · f _s) [As]	-	10
				? c		

Fig. 1. Relation between $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ and η .

gains, with a step down of 6.6 times and an step up of 2 being the largest. As already stated the large reductions are not produced solely by the converter, but with different ways to make the load impedance appear larger. By further inspection a connection between $\frac{V_{IN}^2/f_S}{P_{OUT}}$ and η can be seen, this relation is shown in figure 1. It seems that it is problematic to have a high input voltage and switching frequency while having a low output power and still keep the efficiency high.

This paper will cover the design of a VHF power converter with a low voltage and power output. This will require a huge reduction in voltage and a combination of output power, input voltage and switching frequency unlike any of the previous results. This will put the converter in the area marked in figure 1, as it is seen this is very far from the results achieved by previous researchers (the specs for the converter is given in table II).

TABLE II DESIGN SPECIFICATION FOR THE CONVERTER.

f_S	V _{IN}	VOUT	POUT	R_L
30 - 300 MHz	50V	5V	1W	25

As the load is given it is believed that the first step should be to design the rectifier and then design the inverter for the given input and load. It is also possible to design the inverter to a given load and then use different resistance compression networks to make the impedance of the rectifier match [16], [17], [18]. Though this is a solution often used, it is believed that it will increase the complexity of the converter unnecessarily and possibly reduce the achievable efficiency, size, and price.

Section II will cover the selection and design of a resonant rectifier for the given load and output power level. Then a resonant inverter will be designed for the input voltage and load impedance in section III. Experimental results from three different power stages will be shown in section IV. Finally section V summarises and concludes the paper.

II. RESONANT RECTIFIERS

The purpose of the rectifier is to convert the AC current from the inverter to a DC output. Just as the MOSFET has an output capacitance, the diode has a junction capacitance. In order not to burn this energy in the diode, it is important that the transition is made smoothly, so the capacitance is discharged before the diode turns on.

As stated in section I, it is difficult to achieve high efficiency if the input voltage and switching frequency are high and the output power low. The switching frequency is therefore set to 30MHz for the initial design. If good results are achieved with this frequency, it might be increased further in order to minimize the converter.

Though there are several ways to do this, only the most commonly used class E rectifier will be considered here. The class E rectifier is a rather simple circuit, consisting of a diode, two capacitors, and an inductor as shown in figure 2.



Fig. 2. Schematic of the class E rectifier.

For now it will be assumed that the output capacitance is infinite, so the output voltage is constant, and the diode is assumed to be ideal, i.e. no forward voltage drop, no junction capacitance and no reverse current.

In this case the rectifier will appear resistive at the switching frequency, if the resonance frequency of L_R and C_R are set to this frequency. This will make the design of the inverter a bit simpler, as most design formulas are for a resistive load.

The scaling of the two components will determine the duty cycle of the diode, D_D . As the forward voltage drop of a diode increases with the current running through it, it is desirable to keep D_D as high as possible. However as the diode is connected to the output through an inductor, the average

voltage across it has to be V_{OUT} . Hence a high D_D will lead to a high peak voltage across the diode.

In order to select D_D , and thereby the scaling of the resonating components, the values of real components has to be considered. All the considered inverters has a capacitor at the output insuring a pure AC path, without this there would be a direct DC path from input to output and it would be impossible to reduce the voltage. The average current through the diode, will therefore be the same as the output current.

With the forward voltage drop of a standard schottky diode being around 0.5V (10% of the output voltage), it is crucial to use a diode with low forward voltage drop. Fairchilds MBR0520L has one of the lowest forward voltages available, max 385mV, and can handle a reverse voltage of 20V. With this diode, the diode loss will be up to 77mW or 7.7% of the output power. This clearly limits the maximum achievable efficiency and fits very well with the $\approx 10\%$ efficiency drop, seen in section I when going from an inverter to a complete converter.

A way to reduce the losses could be to use a synchronous rectifier, this will eliminate the forward voltage loss. This will however require an additional MOSFET, with the following need for gate drive and control.

If $D_D = 0.5\%$ is chosen, the peak diode voltage will be $3.562 \cdot 5V = 17.81V$ leaving a little margin up to the maximum. With this D_D the value of C_R should be 67.5pF and the value of L_R should be 417nH [19].

The inductor and output capacitor composes a 2nd order lowpass filter. In order to get maximum 1% (or 50mV) ripple at the output, the 20V AC needs to be attenuated by 52dB which leads to a cut of frequency of 1.5MHz and a output capacitance of 27nF.

The inductor has, as expected, a DC current of 0.2A (the output current with 1W and 5V) and on top of that an AC current with an amplitude of 120mA. The DC resistance of the inductor is estimated to 25m and the AC resistance to 330m (these values are based on an air core inductor with a diameter of 6mm and 8 turns of 0.4mm wire). The loss caused by the inductor can then be calculated to 1mW due to DC losses and 2.4mW due to AC losses.

This is 0.34% of the output power and these resistances are based on a relatively large air core inductor. The Equivalent Series Resistance (ESR) of ceramic capacitor in the sizes used here, will be less than 200m [20] and the currents running through them are smaller than than the current in the inductor. The loss caused by them will thus not be significant. Furthermore the parasitic capacitance of the diode can account for C_R . Actually the parasitic capacitance of the chosen diode is 65pF at 5V reverse voltage fitting almost perfectly with the calculated value.

To sum up, the class E rectifier is a simple circuit which can be made of only 3 components. The selection of the diode is crucial in order to get a decent efficiency and even with the best diodes available, the diode losses will be significant. But as long as an inductor with low ESR is chosen, the total loss in the rectifier should be below 100mW.

III. RESONANT INVERTERS

As mentioned in section I, a resonant inverter is used in order to eliminate switching losses. Either ZVS or ZCS can be achieved and in some special cases both. Generally ZVS will eliminate losses due to parasitic capacitances and ZCS will eliminate losses due to parasitic inductance. For MOSFETs and diodes in power applications the capacitances causes the dominating loss, ZVS will therefore be the main criteria.

However in some cases it is, as mentioned, possible to achieve both ZVS and ZCS switching (also called ZVS and zero slope switching, ZDS). If this can be achieved the exact timing of the switching is less important, as the voltage across the MOSFET will be zero for a small amount of time.

If only ZVS can be achieved, the MOSFET needs to turn on, at exactly the point where the voltage across it hits zero. If it switches just a little to early, there will be energy stored in the capacitor causing switching losses. If it switches a little too late, the drain source voltage will go below zero and the body diode will start to conduct which also gives losses.

The most commonly used resonant inverter is the class E, a schematic of it is shown in figure 3. It consists of a single MOSFET, two inductors, and two capacitors. In optimum operation L_{IN} is an infinite choke providing a pure DC input current. The resonant circuit (L_{RC} and C_{RC}) are inductive at the switching frequency and the inverter is designed to have both ZVS and ZDS.



Fig. 3. Schematic of the class E inverter.

As already mentioned ZVS and ZDS switching can only be achieved in very specific situations. According to [22] and [23] this can only be achieved if:

$$R_L = \frac{8}{\pi^2 + 4} \cdot \frac{V_{IN}^2}{P_{OUT}}$$
$$f_{S,max} = \frac{P_{OUT}}{2 \cdot \pi \cdot C_S \cdot V_{IN}^2}$$
(1)

With 50V input, 1W output, and a switching frequency of 30MHz, this would require a load impedance of 1.44k and an output capacitance of 2.1pF. From equation 1 it is seen that there is a special combination of f_S , V_{IN} and P_{OUT} which makes it possible to operate in the optimum situation. Similar equations can be found for other topologies [22] and this is the reason for the dependence seen in figure 1.

A MOSFET with an output capacitance of 2.1pF for this voltage level and switching frequency is not available, they have minimum 10 times that. Furthermore the impedance is very far from the input impedance of the rectifier. It will therefore not be possible to achieve both ZVS and ZDS switching. However it is still possible to achieve ZVS and thereby

high efficiency as long as the transitions of the MOSFET is controlled well.

If the drain source voltage of the MOSFET is assumed to be a half sine wave when it is off and zero when it is on, the peak voltage across the MOSFET will be:

$$V_{IN} = \int V_{DS} = V_{DS,peak} \frac{2 \cdot (D-1)}{\pi}$$

$$(2)$$

$$V_{DS,peak} = V_{IN} \frac{\pi}{2 \cdot (D-1)}$$

The RMS value of a half wave rectified sine wave is:

$$V_{DS,rms} = V_{DS,peak} \sqrt{\frac{D}{2}}$$
(3)

And the RMS value of the output voltage is:

$$V_{OUT,rms} = \sqrt{P_{OUT} \cdot R_L} \tag{4}$$

According to [2] the reactance of the resonance circuit can now be determined by:

$$X_{RC} = R_L \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1}$$
(5)

By combining equation 2, 3, 4, and 5, an expression for the needed reactance as function of input voltage, duty cycle, output power, and load is obtained:

$$X_{RC} = R_L \cdot \sqrt{\frac{V_{IN}^2 \cdot \pi^2 \cdot D}{2 \cdot (2 \cdot D - 2)^2 \cdot P_{OUT} \cdot R_L} - 1}$$
(6)

It is desirable to keep the duty cycle low in order to reduce the peak voltage across the MOSFET. However due to turn on and off times and delays, it is decided to keep it close to 50%. From equation 2 it is found that a duty cycle of 45% will give a peak voltage of 142.8V, leaving a little headroom if a 150V MOSFET is used. Using this value along with the previous results the needed reactance is found to be 326 . If a capacitor of 680pF is used, the value of the inductor becomes $1.77\mu H$.

The next step is to determine the values of L_{IN} and C_S . In order to minimize losses it is preferable to keep L_{IN} large, thus large AC currents running in and out of the converter and thereby causing unnecessary losses are avoided. If the input choke is assumed infinite, the next step is to calculate the value of C_S . C_S and the resonance circuit should resonate at a frequency with a period of:

$$T_R = 2 \cdot (1 - D) \cdot T_S \Leftrightarrow f_R = \frac{f_S}{2 \cdot (1 - D)}$$

If the reactance of C_S is the same as the reactance of the resonant tank (with opposite operational sign) at f_R the circuit will resonate at this frequency. However as the capacitor is only used when the MOSFET is off, it has to be scaled by 1-D:

$$C_S = \frac{1 - D}{2 \cdot \pi \cdot f_R \cdot X_R} \tag{7}$$

With the specifications for this converter this would require a MOSFET with an output capacitance of maximum 10.9*pF*. At the moment the MOSFETs with lowest output capacitances, C_{OSS} , which are able to switch in the MHz range and handle 150V, has an output capacitance of $\approx 20pF$ at 50V. It is therefore necessary to reduce the input inductor in order to increase C_S . The effective capacitance of the output capacitor is $\frac{C_S}{1-D} = 36.4pF$, hence the total inductance of the resonance circuit and the input inductor should be 936*nH*. Knowing the values of L_{RC} and C_{RC} the input inductance can be calculated to 2.91 μ H.

From a simulation the RMS current through the inductors and the MOSFET is found to be; $I_{IN,rms} = 102mA$, $I_{FET,rms} = 165mA$ and $I_{OUT,rms} = 208mA$.

If the MOSFET has a $R_{ON,max} = 1.2$, this will give a conduction loss in the MOSFET of up to 33mW. The AC resistance of the inductors is estimated to 100m, thus they will have a combined loss of 5.37mW. As for the rectifiers the losses in the capacitors are assumed to be negligible. Thus the total loss in the class E inverter is estimated to $\approx 38mW$.

The complete circuit with the class E inverter and class E rectifier can be seen in figure 4. With the loss calculated for the two parts the combined efficiency will be 88% excluding gate drive and gateing losses.



Fig. 4. Schematic of the class E inverter and class E rectifer.

IV. EXPERIMENTAL RESULTS

In section III the IRF5802 MOSFET were introduced and used for calculations and loss estimates. However several MOSFETs which can be used for this application exist, two of the best suited are compared in table III.

TABLE IIICOMPARISON OF MOSFET CHARACTERISTICS $(R_{ds(on)} \text{ at } V_{GS} = 10V \text{ and capacitances at } V_{DS} = 50V).$

Component	V _{DSS}	ID	R _{DS(on)}	CISS	Coss
FDN86246	150V	0.9A	1.2	85 <i>pF</i>	18 <i>pF</i>
IRF5802	150V	1.6A	359m	180 pF	28 pF

In section III an output capacitance of 10.9pF was found ideal. As the voltage waveform across the MOSFET and C_S is given by V_{IN} , f_S and D, the currents (and thereby losses) in L_{IN} , C_S and the MOSFET scales with the value of C_S . It is therefore desirable to keep the value of C_S as close to this as possible in order to achieve high efficiency.

Comparing the two MOSFETs, the FDN86246 has much lower on resistance than the IRF5802. However the output

capacitance is higher and will as mentioned increase the currents and thus reduce the benefit of the low on resistance. Assuming the waveform across the MOSFET is the same using the two MOSFETs, the current using the FDN86246 will be $\frac{C_{FDN86246} - C_{IRF5802}}{C_{IRF5802}} = 142\%$ larger than using the IRF5802. The on resistance will be reduced by $\frac{R_{IRF5820}-R_{FDN86246}}{IRF5802} = 70.1\%$. Combining this gives a total loss reduction of 27.4%, using the estimated loss found in section III this correspond to 9mW. Furthermore the increased current will also give losses in the input inductor, again using the estimate from section III the increased loss is found to $1.42 \cdot 5.37mW = 7.63mW$. Hence the total loss difference using the two MOSFET is estimated to be less than 2mW. The increased capacitance will however also make the timing of the switching more important, as a larger amount of energy will be stored in the capacitor and burned in the MOSFET if the switching is just a little wrong.

Based on the analysis above, the MOSFET from IRF are found most suited. But as they are very close, prototypes using both will be made to compare them further.

This next subsections will cover the results obtained with three different power stages. The first power stage is the one which has been designed in the previous sections, the second power stage is with the MOSFET with lower $R_{DS(ON)}$ and the last power stage is with a large input inductor and higher output power, this should, as described in section III, give a higher efficiency.

A. MOSFET with low Coss

A power stage with the components selected in sections II and III was implemented. Just as the case were when going from calculated values to simulations, slight adjustments had to be made. The tuning procedure was, first to tune the inductor in the output filter to make it resistive at the switching frequency. Once that was done, the inverter was added to the design and a low voltage was applied. It was seen that the converter was not ZVS switching as the output capacitance of the MOSFET was not discharged when it switched on.

In order to get it to discharge faster the values of the input and resonant inductors had to be lowered. First the resonant inductor was lowered to give the desired output power and then the input inductor was adjusted to make the converter ZVS. As the output capacitance of the MOSFET is 20pF, measuring with a probe with 10pF capacitance changes the switching a lot. This ruins the tuning and ZVS is thus not achieved, some waveforms have been measured and they are shown in figure 5. As it is seen the drain voltage has a small break when the MOSFET is switched on, this is due to the 10pF added by the probe.

So for the final fine tuning a thermal camera was used to measure the temperature of the MOSFET, it was then assumed that the MOSFET was ZVS when the temperature rise was lowest. Even though this is not a solid proof of ZVS, a low temperature rise comes from low power loss and thus the best tuned circuit. The efficiency was measured to 71.5% (see table IV).



Fig. 5. Measurements on the prototype with low C_{OSS} .



Fig. 6. Measurements on the prototype with large input inductor.

B. MOSFET with low R_{ON}

When the MOSFET was selected, the FDN86246 was found to be equally good to the IRF5802. The biggest difference between the two was on resistances and parasitic capacitances. A prototype was implemented using the FDN86246 in a circuit almost identical to the one used for the previous converter. A few turns was removed from the input inductor in order to make the converter ZVS with the increased output capacitance.

Due to the higher output capacitance more energy is stored and if the switch is switched at a few volts instead of zero, much more energy will be burned in the on resistance. Furthermore the AC current in the input inductor is larger which also increases the losses. The total efficiency of the converter was measured to 60.7%.

C. With large input inductor

As explained in section III, the highest efficiency should be achieved with a large input inductor (DC input current). To test this a prototype was made with the IRF5802, but this time with a 6.5μ H input inductor. Then the resonance circuit and the load was adjusted in order to get ZVS and 5V output.

The increased output power makes the current through the MOSFET closer to that seen for the ideal class E inverter. Thereby the loss due to slight deviations in the timing of the switching becomes smaller.

The waveforms shown in figure 6 clearly shows that the converter is not ZVS when probes are placed at the gate and drain. Furthermore the voltage drops below 4V, however removing the probes makes the output voltage increase to 5.0V.



Fig. 7. The achieved $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ -factor and η next to previous results.

When tuned, the output power of the circuit became 1.53W and the efficiency was measured to 82.9%. This efficiency is without gate drive, but it is still among the best results achieved by previous researchers. Furthermore the $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ -factor explained in section I is much smaller than for any of the previous converters.

D. Summary of experimental results

The efficiency achieved for the three power stages is shown in table IV. From the three prototypes it is seen that good efficiencies can be achieved just by having ZVS. However the larger the current through the MOSFET is at the switching instant, the more important becomes the timing of the switching and losses increase.

TABLE IV MEASUREMENTS ON POWER STAGES.

Converter	f_S	IIN	V _{OUT}	R_L	η	T _{Mos}	T _{Dio}
	[MHz]	[mA]	[V]	[]	[%]	[°C]	[°C]
Low Coss	30	28	5.00	25	71.5	55.3	52.2
Low R_{ON}	29	32	4.93	25	60.7	65.1	53.2
Large LIN	30	37	5.00	16.3	82.9	46.2	50.5

It has been shown that VHF converters with a very low $\frac{V_{IN}^2 \cdot f_S}{P_{OUT}}$ -factor can be made with high efficiency, the best even had an efficiency of 82.9% which puts it among the best VHF converters. For comparison the results achieved for the power stages are shown i figure 7. The efficiency is not as high as wanted and the factor is a little lower than desired for one of the prototypes due to the higher output power. However seen next to previously achieved results they are very close.

V. CONCLUSION

The theoretical design of the resonant converter was considered in section II and III. Several different topologies were considered and based on complexity and efficiency estimates a class E inverter and rectifier was chosen.

Three different power stages were made; one with a MOS-FET with the lowest available output capacitance, one with a MOSFET with low on resistance, and one with increased output power allowing a large input inductor. All the converters had 50V input and 5V output and the achieved efficiencies were between 60.7% and 82.9%. This shows that it is possible to make low power very high frequency converters with high step down ratio.

REFERENCES

- [1] W. C. Bowman, J. F. Balicki, F. T. Dickens et al.: "A resonant DC-to-DC converter operating at 22 megahertz", Applied Power Electronics Conference and Exposition, 1988. APEC '88. Conference Proceedings 1988., Third Annual IEEE, vol., no., pp.3-11, 1-5 Feb 1988
- [2] J. Rivas: "Radio Frequency dc-dc Power Conversion", Doctoral Thesis, Massachusetts Institute of Technology, 2006
- [3] J. R. Warren, III, K. A. Rosowski and D. J. Perreault: "Transistor Selection and Design of a VHF DC-DC Power Converter", Power Electronics, IEEE Transactions on, vol.23, no.1, pp.27-37, Jan. 2008
- [4] J. M. Rivas, O. Leitermann, Y. Han and D. J. Perreault: "A Very High Frequency dc-dc Converter Based on a Class \$\overline{2}\$ Resonant Inverter", Power Electronics Specialists Conference, 2008. PESC 2008. IEEE, vol., no., pp.1657-1666, 15-19 June 2008
- [5] D. J. Perreault, J. Hu, J. M. Rivas et al.: "Opportunities and Challenges in Very High Frequency Power Conversion", Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE, vol., no., pp.1-14, 15-19 Feb. 2009
- [6] J. S. Glaser, J. Nasadoski and R. Heinrich: "A 900W, 300V to 50V Dc-dc Power Converter with a 30 MHz Switching Frequency", Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE, vol., no., pp.1121-1128, 15-19 Feb. 2009
- [7] I. Douglas de Vries: "High Power and High Frequency Class-DE Inverters", Doctoral Thesis, University of Cape Town, 1999
- [8] Z. Kaczmarczyk: "High-Efficiency Class E, EF2, and E/F3 Inverters", IEEE Trans. Industrial Electronics, vol. IE-53, pp. 15841593, Oct. 2006
- [9] J. M. Rivas, Y. Han, O. Leitermann et al.: "A High-Frequency Resonant Inverter Topology with Low Voltage Stress", Power Electronics, IEEE Transactions on, vol.23, no.4, pp.1759-1771, July 2008
 [10] Z. Kaczmarczyk: "A high-efficiency Class-E inverter - computer model,
- [10] Z. Kaczmarczyk: "A high-efficiency Class-E inverter computer model, laboratory measurements and SPICE simulation", BULLETIN OF THE POLISH ACADEMY OF SCIENCES, TECHNICAL SCIENCES, Vol. 55, No. 4, 2007
- [11] P. Kamby: "High Efficiency Radio Frequency Switch-mode Power Supply for LED Applications", Master Thesis, Technical University of Denmark, 2011
- [12] M. K. Kazimierczuk and J. Jozwik: "Resonant dc/dc Converter with Class-E Inverter and Class-E Rectifier", Industrial Electronics, IEEE Transactions on, vol.36, no.4, pp.468-478, Nov 1989
- [13] J. M. Rivas, D. Jackson, O. Leitermann et al.: "Design Considerations for Very High Frequency dc-dc Converters", Power Electronics Specialists Conference, 2006. PESC '06. 37th IEEE, vol., no., pp.1-11, 18-22 June 2006
- [14] T. M. Andersen: "Radio Frequency Switch-Mode Power Supplies", Master Thesis, Technical University of Denmark, 2010
- [15] J. S. Glaser and J. M. Rivas: "A 500 W Push-pull Dc-dc Power Converter with a 30 MHz Switching Frequency", Applied Power Electronics Conference and Exposition (APEC), 2010 Twenty-Fifth Annual IEEE , vol., no., pp.654-661, 21-25 Feb. 2010
- [16] Y. Han, O. Leitermann, D. A. Jackson et al.: "Resistance Compression Networks for Radio-Frequency Power Conversion", Power Electronics, IEEE Transactions on, vol.22, no.1, pp.41-53, Jan. 2007
- [17] R. S. Wahby: "Radio Frequency Rectifiers for DC-DC Power Conversion", Master Thesis, Massachusetts Institute of Technology, 2004
- [18] D. C. Hamill: "Impedance plane analysis of class DE amplifier", Electronics Letters, vol.30, no.23, pp.1905-1906, 10 Nov 1994
- [19] M. K. Kazimierczuk and D. Czarkowski: "Resonant Power Converters", John Wiley & Sons, 2nd Edition, 2011, ISBN: 978-0-470-90538-8
- [20] T. Roy and L. Smith: "ESR and ESL of Ceramic Capacitor Applied to Decoupling Applications", Electrical Performance of Electronic Packaging, 1998. IEEE 7th topical Meeting on , vol., no., pp.213-216, 26-28 Oct 1998
- [21] D. C. Hamill: "Class DE Inverters and Rectifiers for DC-DC Conversion", Power Electronics Specialists Conference, 1996. PESC '96 Record., 27th Annual IEEE, vol.1, no., pp.854-860 vol.1, 23-27 Jun 1996
- [22] M. K. Kazimierczuk: "RF Power Amplifiers", John Wiley & Sons, 1st Edition, 2008, ISBN: 978-0-470-77946-0
- [23] A. Grebennikov and N. O. Sokal: "Switchmode RF Power Amplifiers", Newnes, 1st Edition, 2007, ISBN: 978-0-7506-7962-6

A.4 M. Madsen, A. Knott, M. A.E. Andersen: "Low Power Very High Frequency Switch-Mode Power Supply with 50 V Input and 5 V Output", IEEE Transactions on Power Electronics, December 2014, vol.29, no.12, pp.6569-6580.

Low Power Very High Frequency Switch-Mode Power Supply With 50 V Input and 5 V Output

Mickey Madsen, Student Member, IEEE, Arnold Knott, Member, IEEE, and Michael A. E. Andersen, Member, IEEE

Abstract—This paper presents the design of a resonant converter with a switching frequency in the very high frequency range (30– 300 MHz), a large step down ratio (ten times), and low output power (1 W). Several different inverters and rectifiers are analyzed and compared. The class E inverter and rectifier are selected based on complexity and efficiency estimates. Three different power stages are implemented; one with a large input inductor, one with a switch with small capacitances, and one with a switch with low on-resistance. The power stages are designed with the same specifications and efficiencies from 60.7–82.9% are achieved.

Index Terms—DC-DC power conversion, inverters, rectifiers, resonant power conversion, switched mode power supplies, VHF circuits.

I. INTRODUCTION

HEN designing power converters it is always a goal to reduce the price and the physical size, i.e. increase the power density. The development of switch mode power supplies (SMPS) has made it possible to increase the power density significantly, but it is limited by the size of the passive energy storing components (inductors and capacitors). The value and size of these are however dependent on the switching frequency. By increasing the switching frequency it will hence be possible to reduce the size of SMPSs further.

Traditional SMPS topologies like Buck and Boost are hard switching, this means the MOSFET is switching while energy is stored in the output capacitance. The result is that energy is dissipated in the MOSFET every time it turns on. Although this introduces losses in the converter, it is not critical for converters switching at 50–400 kHz. But when the frequency is increased to the very high frequency (VHF) range (30–300 MHz) the dissipated power get almost 1000 times larger. This amount of energy would ruin the efficiency and require extreme cooling of the MOSFET. This leads to the development of resonant converters.

In order to avoid switching losses and be able to increase the frequency while keeping the efficiency high, new topologies have to be used. For the last two decades (since 1988 [1]), research has been done in order to enable the use of resonant RF

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org.

Digital Object Identifier 10.1109/TPEL.2014.2305738

 V_{PC}

Fig. 1. Block diagram of the converter.

amplifiers (inverters) combined with a rectifier for dc/dc converters, see Fig. 1. With these type of converters, it is possible to achieve zero voltage switching (ZVS) and/or zero current switching (ZCS). In this case, the MOSFET turns ON when the voltage and/or current across/through it is zero. Theoretically, this should eliminate switching losses if the switching is done instantaneously and at exactly the right time. This is not practically achievable, but even with slight deviations from the ideal case very high efficiencies can be achieved.

As already mentioned the value of the passive components depends on the switching frequency. Hence, an increase in frequency will lead to a reduction in size, as long as the size of the passive scales with the value. This assumption generally holds, but magnetic materials and packaging introduce some challenges. When the frequency is pushed far into the meghertz range, magnetic core losses increase rapidly and become unacceptably high for most core materials [2]. At this point, air core and PCB embedded inductors become a viable solutions, as the inductances needed at these frequencies can be made in a small physical size and the core losses avoided [3], [4].

Increasing the switching frequency also leads to capacitors with lower values. Electrolytic capacitors which often limit the overall lifetime [5], [6] can hence be avoided. The reduction in component values also leads to a cost reduction as smaller components are generally cheaper. If the frequency is increased enough, some of the components can even be left out as they can be constituted by the parasitic parts of other components (this will be explained further in Sections II and III). An increase in switching frequency will also make it easier to comply with EMI requirements, as switching harmonics can easily be filtered out by small and cheap filters.

With a switching frequency in the VHF range, it will also be possible to achieve very fast transient responses [7] which are highly demanded, e.g., for envelope tracking [8]. However, in order to fully benefit from this, an efficient and fast control loop has to be implemented. This is a big challenge and while some ways of achieving continuous regulation have been found [9], [10], the best results are still achieved using burst mode (or cell modulation) as in [11]–[13]. Due to the high switching

0885-8993 © 2014 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications_standards/publications/rights/index.html for more information.

Manuscript received August 1, 2013; revised December 29, 2013 and October 28, 2013; accepted January 31, 2014. Date of publication February 12, 2014; date of current version August 13, 2014. Recommended for publication by Associate Editor J. A. Cobos.

The authors are with the Department of Electrical Engineering, Technical University of Denmark, 2800 Kongens Lyngby, Denmark (e-mail: mpma@elektro.dtu.dk; akn@elektro.dtu.dk; ma@elektro.dtu.dk).

		It	iverters			
Topology	f_s	V _{IN}	VOUT	POUT	η	Year
	[MHz]	[V]	[V]	[W]	[%]	
Class DE	5.3	330	N/A	1154	89	1999 [17]
Class E	1	128	N/A	366	96.6	2006 [18]
Class ϕ_2	1	129	N/A	526	97.1	2006 [18]
Class ϕ_2	30	160	N/A	330	93.7	2007 [19]
Class E	1	129	N/A	322.7	97	2007 [20]
Class E	100	9	N/A	6.8	82	2011 [21]
		Co	nverters			
Topology	f	V	V	n		XZ
ropology	Js	VIN	VOUT	P_{OUT}	η	Year
Topology	[MHz]	[V]	VOUT [V]	P_{OUT} [W]	ון [%]	Year
Class E	[MHz] 1	$\frac{[V]}{20}$	VOUT [V] 25	P _{OUT} [W] 8.9	1] [%] 89	1989 [22]
Class E Class ϕ_2	[MHz] 1 30		VOUT [V] 25 33	P _{OUT} [W] 8.9 265	1] [%] 89 87	1989 [22] 2006 [11]
Class E Class ϕ_2 Class E	5s [MHz] 1 30 100	[V] [V] 20 165 11	VOUT [V] 25 33 12	P _{OUT} [W] 8.9 265 10	1] [%] 89 87 75	1989 [22] 2006 [11] 2006 [23]
Class E Class ϕ_2 Class E Class ϕ_2	^{Js} [MHz] 1 30 100 30	[V] 20 165 11 150	VOUT [V] 25 33 12 33	P _{OUT} [W] 8.9 265 10 180	1] [%] 89 87 75 84	Year 1989 [22] 2006 [11] 2006 [23] 2008 [14]
Class E Class ϕ_2 Class E Class ϕ_2 Class ϕ_2 Class ϕ_2	Js [MHz] 1 30 100 30 10	V _{IN} [V] 20 165 11 150 170	VOUT [V] 25 33 12 33 75	$\begin{array}{c} P_{OUT} \\ [W] \\ \hline 8.9 \\ 265 \\ 10 \\ 180 \\ 250 \end{array}$	1] [%] 89 87 75 84 91	Year 1989 [22] 2006 [11] 2006 [23] 2008 [14] 2009 [15]
Class E Class ϕ_2 Class E Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2	Js [MHz] 1 30 100 30 10 30	$\begin{array}{c} V_{IN} \\ [V] \\ \hline 20 \\ 165 \\ 11 \\ 150 \\ 170 \\ 165 \end{array}$	VOUT [V] 25 33 12 33 75 33	Pour [W] 8.9 265 10 180 250 225	1] [%] 89 87 75 84 91 87	Year 1989 [22] 2006 [11] 2006 [23] 2008 [14] 2009 [15] 2009 [15]
Class E Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2	Js [MHz] 1 30 100 30 10 30 30 30	VIN [V] 20 165 11 150 170 165 330	VOUT [V] 25 33 12 33 75 33 50	Pour [W] 8.9 265 10 180 250 225 900	1] [%] 89 87 75 84 91 87 79	Year 1989 [22] 2006 [11] 2006 [23] 2008 [14] 2009 [15] 2009 [15] 2009 [16]
Class E Class ϕ_2 Class E Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class ϕ_2 Class E	Js [MHz] 1 30 100 30 10 30 30 100	VIN [V] 20 165 11 150 170 165 330 12	vour [V] 25 33 12 33 75 33 50 23.7	Pour [W] 8.9 265 10 180 250 225 900 1.7	1] [%] 89 87 75 84 91 87 79 55	Year 1989 [22] 2006 [11] 2006 [23] 2008 [14] 2009 [15] 2009 [15] 2009 [16] 2010 [24]

TABLE I Results From Previous Research

frequency the converter will reach steady state after just a few microseconds, this makes it possible to use an array of small converters and switch them on and off as needed. In this way, each converter is designed to operate with a defined load/output. This makes the design much easier as resonant inverters are generally very load dependent.

The fact that resonant inverters are load dependent, makes it very hard to achieve good performance at varying loads. Furthermore, resonant inverters need a large load impedance to operate in the ideal situation (having both ZVS and ZCS). This makes them well suited for boost-type converters, but making a buck type is a bit more challenging. The most commonly used way to overcome this challenge is to add an autotransformer at the output, in that way the load impedance seen by the inverter is increased [11], [14], [15]. Another way to achieve low output voltage is to use an array of converters with the input in series and the output in parallel in [16].

The most commonly used inverter is the class E, however several other topologies exist. Some of the research results are summed up in Table I. From the table it is seen that very high efficiencies are achievable for the inverters up to 97%. However, the efficiency drops around 10% for the complete dc/dc converters, i.e., when a rectifier is added.

From Table I, it is also seen that the converters have limited gains, with a step down of 6.6 times and a step up of 2 being the largest. As already stated the large reductions are not produced solely by the converter, but with different ways to make the load impedance appear larger. By further inspection, the connection given by (1) can be seen, this relation is shown in Fig. 2

$$\frac{V_{\rm IN}^2 \cdot f_S}{P_{\rm OUT}} \propto \frac{1}{\eta}.$$
 (1)

Equation (1) shows that it is problematic to have a high input voltage and switching frequency while having a low output power and still keeping the efficiency high. The input voltage sets (together with C_{OSS}) the energy stored in the output capacitance of the MOSFET each switching period and f_S sets



Fig. 2. Relation between $\frac{V_{1N}^2 \cdot f_S}{P_{OUT}}$ and η for the converters in Table I.

TABLE II DESIGN SPECIFICATION FOR THE CONVERTER

fs	V_{IN}	V_{OUT}	P_{OUT}	R_L
30-300 MHz	50 V	5 V	1 W	25 Ω

how many times this has to be done each second. Combined these values are hence proportional to the circulating energy, that needs to run in the converter in order to ensure ZVS. The relation given in (1) hence states that it is difficult to achieve high efficiency, if the circulating energy that is needed for ZVS is high compared to the output power. The reason for this and the factor $V_{\rm IN}^2 \cdot f_S / P_{\rm OUT}$ will be described further in Section III.

This paper will cover the design of a VHF power converter with a low voltage and power output. This will require a large reduction in voltage and a combination of output power, input voltage, and switching frequency unlike any of the previous results. This will put the converter in the area marked in Fig. 2. As it is seen this is very far from the results achieved by previous researchers (the specifications for the converter is given in Table II).

As the load is given, the first step should be to design the rectifier and then design the inverter for the given input and load. It is also possible to design the inverter to a given load and then use different resistance compression networks to make the impedance of the rectifier match [26]–[28]. Though this is a solution often used, it will increase the complexity of the converter unnecessarily and possibly reduce the achievable efficiency, size, and price.

Section II covers the selection and design of a resonant rectifier for the given load and output power level. Then, a resonant inverter is designed for the input voltage and load impedance in Section III. Experimental results from three different power stages are shown in Section IV. Finally Section V summarizes and concludes the paper.

II. RESONANT RECTIFIERS

The purpose of the rectifier is to convert the ac current from the inverter to a dc output. Just as the MOSFET has an output capacitance, the diode has a junction capacitance. In order not to dissipate this energy in the diode, it is important that the transition is made smoothly, so the capacitance is discharged before the diode turns on. Though there are several ways to do this, only two will be considered here. The most commonly used class E rectifier and the equivalent to the class DE inverter, the class DE rectifier.

A. Class E

The class E rectifier is a rather simple circuit, consisting of a diode, two capacitors, and an inductor as shown in Fig. 3. Together these components constitutes a resonant rectifier capable of rectifying the ac input current to a dc output.

For now, it will be assumed that the output capacitance is infinite, so the output voltage is constant, and the diode is assumed to be ideal, i.e., no forward voltage drop, no junction capacitance, and no reverse current.

In this case, the rectifier will appear resistive at the switching frequency, if the resonance frequency of L_R and C_R are set to this frequency. This will simplify the design of the inverter as most design formulas are for a resistive load.

The scaling of the two components will determine the duty cycle of the diode, D_D . As the forward voltage drop of a diode increases with the current running through it, it is desirable to keep D_D as high as possible. However, as the diode is connected to the output through an inductor, the average voltage across it has to be V_{OUT} . Hence, a high D_D will lead to a high peak voltage across the diode.

In order to select D_D , and thereby the scaling of the resonant components, the values of real components have to be considered. All the considered inverters have a capacitor at the output ensuring a pure ac path, without this there would be a direct dc path from input to output and it would be impossible to reduce the voltage. The average current through the diode will therefore be the same as the output current.

With the forward voltage drop of a standard Schottky diode being around 0.5 V (10% of the output voltage), it is crucial to use a diode with low forward voltage drop. Fairchilds MBR0520L has one of the lowest forward voltages available, max 385 mV, and can handle a reverse voltage of 20 V. With this diode, the diode loss will be up to 77 mW or 7.7% of the output power. This clearly limits the maximum achievable efficiency and fits very well with the $\approx 10\%$ efficiency drop, seen in Section I when going from an inverter to a complete converter.

If $D_D = 50\%$ is chosen, the peak diode voltage will be 3.6 · V_{OUT} = 17.8 V leaving a little margin up to the maximum [29]. With this D_D the value of C_R should be [29]

$$C_R = \frac{1}{2 \cdot \pi^2 \cdot f_S \cdot R_L} = 67.5 \text{ pF.}$$
 (2)

With this capacitance, the value of L_R can be calculated according to [29]

$$L_R = \frac{1}{(2 \cdot \pi \cdot f_S)^2 \cdot C_R} = 417 \text{ nH.}$$
(3)



Fig. 3. Schematic of the class E rectifier.



Fig. 4. Class E rectifier waveforms simulated with PLECS.

The inductor has, as expected, a dc current of 0.2 A (the output current with 1 W and 5 V) and on top of that an ac current with an amplitude of 120 mA (see Fig. 4). The dc resistance of the inductor is estimated to 25 m Ω and the ac resistance to 330 m Ω (these values are based on an air core inductor with a diameter of 6 mm and 8 turns of 0.4 mm wire). The loss caused by the inductor can then be calculated to 1 mW due to dc losses and 2.4 mW due to ac losses.

This is 0.34% of the output power and these resistances are based on a relatively large air core inductor. The equivalent series resistance (ESR) of a ceramic capacitor in the sizes used here will be less than 200 m Ω [30] and the currents running through them are smaller than the current in the inductor. The loss caused by them will thus not be significant. Furthermore, the parasitic capacitance of the diode can account for C_R . Actually, the parasitic capacitance of the chosen diode is, according to the datasheet, 65 pF at 5 V reverse voltage fitting almost perfectly with the calculated value.

B. Class DE

The class DE rectifier has an extra diode compared to the class E, but it does not have any inductors and the physical size and prize are expected to be more or less the same. As seen in the schematic in Fig. 5, the output capacitance is split in two. For now, they will both be assumed to be infinite making the output voltage pure DC. As the diodes are connected directly to the output, the total voltage across them will always be $V_{\rm OUT}$. The diode duty cycle can therefore be chosen freely between 0 and



Fig. 5. Schematic of the class DE rectifier.



Fig. 6. Class DE rectifier waveforms simulated with PLECS.

0.5, a higher duty cycle would require both diodes to conduct at the same time.

From [31], (4)–(6) can be found. If $I_{\rm IN,peak}$ is isolated in (4) and ϕ is isolated in (6), the results can be substituted into (5). C_R can then be isolated in order to find the capacitances needed to get a desired diode duty cycle, (7). If a diode duty cycle of 25% is chosen, the needed capacitance can be calculated to 667 pF

$$V_{\rm OUT} = \frac{I_{\rm IN, peak} \cdot R_L}{\pi + \omega \cdot C_R \cdot R_L} \tag{4}$$

$$\cos(\phi) = 1 - \frac{2 \cdot \omega \cdot C_R \cdot V_{\text{OUT}}}{I_{\text{IN,peak}}}$$
(5)

$$D_D = \frac{\pi - \phi}{2 \cdot \pi} \tag{6}$$

$$C_R = \frac{\pi \cdot (1 - \cos(\pi - 2 \cdot D_D \cdot \pi))}{\omega \cdot R_L \cdot (1 + \cos(\pi - 2 \cdot D_D \cdot \pi))}.$$
 (7)

As mentioned the diodes are coupled directly to the output, furthermore they provide the only dc path for the output current. The average current through each of the diodes will therefore be I_{OUT} (see Fig. 6), resulting in twice the diode loss as for the class E rectifier. This results in a total loss of more than 150 mW. Though several diodes could be put in parallel in order to reduce the forward voltage drop a bit, the diode losses will still be well above 100 mW, i.e., 10% of the output power.

TABLE III PROS AND CONS OF THE INVESTIGATED RECTIFIER TOPOLOGIES

	Class E	Class DE
Pros	Low complexity	 No inductors
	Well documented and tested	
Cons	High semiconductor stress	High loss
		• 2 semiconductors

C. Selection of Rectifier

Based on the analysis of the two rectifiers, the class E rectifier is found to be the best choice. The size and prize of the two rectifiers will be similar, but the loss of the class DE will be significantly higher than for the class E. Some of the pros and cons are shown in Table III.

For a high voltage output the DE might be better though, as the voltage across the diodes is lower and smaller diodes might be used. But for the low output needed for this converter, a class E rectifier is found to be the best choice.

The losses of the class E rectifier might even be unacceptable. A way to reduce the losses could be to use a synchronous rectifier, this will eliminate the forward voltage loss. This will however require an additional MOSFET, with the following need for gate drive and control.

III. RESONANT INVERTERS

As mentioned in Section I, a resonant inverter is used in order to eliminate switching losses. Either ZVS or ZCS can be achieved and in some special cases both. Generally, ZVS will eliminate losses due to parasitic capacitances and ZCS will eliminate losses due to parasitic inductance. For MOSFETs and diodes in power applications the capacitances causes the dominating loss, ZVS will therefore be the main criteria.

However in some cases it is, as mentioned, possible to achieve both ZVS and ZCS switchings (also called ZVS and zero derivative switching, ZDS). If this can be achieved the exact timing of the switching is less important, as the voltage across the MOSFET will be zero for a small amount of time.

If only ZVS can be achieved, the MOSFET needs to turn ON at exactly the point where the voltage across it hits zero. If it switches just a little too early, there will be energy stored in the capacitor causing switching losses. If it switches a little too late, the drain source voltage will go below zero and the body diode will start to conduct which also gives losses.

A. Class E

The most commonly used resonant inverter is the class E, a schematic of it is shown in Fig. 7. It consists of a single MOSFET, two inductors, and two capacitors. In optimum operation $L_{\rm IN}$ is an infinite choke providing a pure dc input current. The resonant circuit (L_R and C_R) is inductive at the switching frequency and the inverter is designed to have both ZVS and ZDS.

As already mentioned ZVS and ZDS switching can only be achieved in very specific situations. According to [32], [33] this



Fig. 7. Schematic of the class E inverter.



Fig. 8. Class E inverter waveforms with realistic $C_{\rm OSS}$ simulated with PLECS.

can only be achieved if

$$R_L = \frac{8}{\pi^2 + 4} \cdot \frac{V_{\rm IN}^2}{P_{\rm OUT}}$$
(8)

$$f_{S,\max} = \frac{P_{\rm OUT}}{2 \cdot \pi \cdot C_S \cdot V_{\rm IN}^2}.$$
(9)

With 50 V input, 1 W output, and a switching frequency of 30 MHz, this would require a load impedance of 1.44 k Ω and an output capacitance of 2.1 pF. From (9), it is seen that there is a special combination of f_S , V_{IN} , and P_{OUT} which makes it possible to operate in the optimum situation. Similar equations can be found for other topologies [32] and this is the reason for the dependence seen in Fig. 2.

A MOSFET with an output capacitance of 2.1 pF for this voltage level and switching frequency is not available, they have minimum 10 times that. Furthermore, the impedance is very far from the input impedance of the rectifier. It will therefore not be possible to achieve both ZVS and ZDS switchings. However, it is still possible to achieve ZVS and thereby high efficiency as long as the transitions of the MOSFET are controlled well. In this case, the components have to be selected carefully in order to ensure ZVS and the inverter will be running in a subnominal condition as described further in [34].

If the drain source voltage of the MOSFET is assumed to be a half sine wave when it is off and zero when it is on, the peak voltage across the MOSFET will be

$$V_{\rm IN} = \int V_{\rm DS} = V_{\rm DS,peak} \frac{2 \cdot (1-D)}{\pi} \qquad (10)$$

$$V_{\rm DS,peak} = V_{\rm IN} \frac{\pi}{2 \cdot (1-D)}.$$
(11)

The rms value of a half wave rectified sine wave is

$$V_{\rm DS,rms} = V_{\rm DS,peak} \sqrt{\frac{D}{2}}$$
(12)

and the rms value of the output voltage is

≙

I

$$V_{\rm OUT,rms} = \sqrt{P_{\rm OUT} \cdot R_L}.$$
 (13)

According to [11], the reactance of the resonance circuit can now be determined by

$$X_{\rm RC} = R_L \cdot \sqrt{\left(\frac{V_{\rm DS,rms}}{V_{\rm OUT,rms}}\right)^2 - 1}.$$
 (14)

By combining (11)–(14), an expression for the needed reactance as function of input voltage, duty cycle, output power, and load is obtained

$$X_{\rm RC} = R_L \cdot \sqrt{\frac{V_{\rm IN}^2 \cdot \pi^2 \cdot D}{2 \cdot (2 \cdot D - 2)^2 \cdot P_{\rm OUT} \cdot R_L}} - 1.$$
(15)

It is desirable to keep the duty cycle low in order to reduce the peak voltage across the MOSFET. However, due to turn on and off times and delays, it is decided to keep it close to 50%. From (11), it is found that a duty cycle of 45% will give a peak voltage of 142.8 V, leaving a little headroom if a 150 V MOSFET is used. Using this value along with the previous results, the needed reactance is found to be 326 Ω . If a capacitor of 680 pF is used, the value of the inductor can be calculated according to

$$L_R = \frac{C_R \cdot X_{\rm RC} \cdot \omega_S + 1}{C_R \cdot \omega_S^2} = 1.77 \,\,\mu\text{H}.\tag{16}$$

The next step is to determine the values of $L_{\rm IN}$ and C_S . In order to minimize losses, it is preferable to keep $L_{\rm IN}$ large, thus large ac currents running in and out of the converter and thereby causing unnecessary losses are avoided. If the input choke is assumed infinite, the next step is to calculate the value of C_S . In order to ensure ZVS, the voltage across C_S needs to rise to the peak and fall back down to zero within the period where the switch is open. This requires C_S and the resonance circuit to resonate at a frequency with a period equal to two times the period where the switch is open, i.e.,

$$T_R = 2 \cdot (1 - D) \cdot T_S \Leftrightarrow f_R = \frac{f_S}{2 \cdot (1 - D)}.$$
 (17)

If the reactance of C_S is the same as the reactance of the resonant tank (with opposite operational sign) at f_R the circuit will resonate at this frequency. However, as the capacitor is only used when the MOSFET is off, it has to be scaled by 1-D

$$C_S = \frac{1 - D}{2 \cdot \pi \cdot f_R \cdot X_{\rm RC}}.$$
(18)

With the specifications for this converter it would require a MOSFET with an output capacitance of maximum 10.9 pF. At the moment the MOSFETs with lowest output capacitances,

TABLE IV CURRENTS (RMS) IN THE CLASS E INVERTER

IN	MOSFET	OUT
102 mA	165 mA	208 mA

 C_{OSS} , which are able to handle 150 V, have an output capacitance of ≈ 20 pF at 50 V. It is therefore necessary to reduce the input inductor, in order to increase C_S while keeping the resonance frequency at the switch node equal to f_R . The output capacitance of the MOSFET is only contributing to the resonance in the part of the period where the MOSFET is OFF, hence it has to be scaled by 1-D in order to find the effective capacitance. The effective capacitance of the output capacitor is $C_{S,\text{eff}} = \frac{C_S}{1-D} = 36.4$ pF, hence the total inductance of the resonance circuit and the input inductor should be

$$L_{\text{total}} = \frac{1}{\omega_R^2 \cdot C_{S,\text{eff}}} = 936 \text{ nH}.$$
 (19)

Knowing the values of $X_{\rm RC}$, the input inductance can be calculated according to

$$L_{\text{total}} = \frac{1}{\frac{1}{L_{\text{IN}}} + \frac{\omega_R}{X_{\text{RC}}}} \Leftrightarrow L_{\text{IN}} = \frac{1}{\frac{1}{\frac{1}{L_{\text{total}}} - \frac{\omega_R}{X_{\text{RC}}}}} = 2.91 \mu \text{H}.$$
(20)

From a simulation, the rms current through the inductors and the MOSFET is found (see Fig. 8 and Table IV). The IRF5802 MOSFET has the lowest available output capacitance for a 150 V power MOSFET capable of switching in the VHF range. It has 20 pF output capacitance at 50 V and an on-resistance of 1.2 Ω , this will give a conduction loss in the MOSFET of up to 33 mW. The ac resistance of the inductors is estimated to be 100 m Ω , thus they will have a combined loss of 5.37 mW. As for the rectifiers, the losses in the capacitors are assumed to be negligible. Thus, the total loss in the class E inverter is estimated to \approx 38 mW.

B. Class ϕ_2

As written in Section I, the large voltage peak across the MOSFET is a big problem when the input voltage is large. The class ϕ_2 (or EF_2) inverter, which is a hybrid between the class E and F_2 inverters, was developed in order to make the voltage across the MOSFET closer to a square wave. The voltage across the MOSFET should thereby become significantly smaller (ideally $2 \cdot V_{\rm IN}$ for D = 50%). This is done by inserting a *LC* circuit in parallel with the MOSFET as shown in Fig. 9. This circuit is designed to have a resonance frequency at the second harmonic, which causes the voltage across the MOSFET to become a trapezoidal wave consisting of the first and third harmonics. The same benefits can be achieved with the flat-top class-E amplifier described in [35].

According to [14], the rms voltage across the MOSFET can be estimated by $V_{\rm DS,rms} = V_{\rm IN} \frac{4}{\pi \cdot \sqrt{2}}$; thus, the needed reactance of the resonant circuit is different (see (14)). The new values can be calculated to $L_R = 1.2 \ \mu \text{H}$ and $C_R = 522 \ \text{pF}$. No exact equations for the calculations of the added *LC* circuit or the input inductance are given in the literature; however, the following



Fig. 9. Schematic of the class ϕ_2 inverter.



Fig. 10. Tuned class ϕ_2 inverter waveforms simulated with PLECS (the peak drain source voltage is reduced to 132 V).

TABLE V Currents (RMS) in the Class ϕ_2 Inverter

IN	MR	MOSFET	OUT
311 mA	301 mA	207 mA	208 mA

gives results which are close [19]:

$$L_{\rm IN} = \frac{1}{9 \cdot \pi^2 \cdot f_S^2 \cdot C_S} = 625 \text{ nH}$$
(21)

$$L_{\rm MR} = \frac{1}{15 \cdot \pi^2 \cdot f_S^2 \cdot C_S} = 375 \text{ nH}$$
(22)

$$C_{\rm MR} = \frac{15}{16} \cdot C_S = 18.8 \,\mathrm{pF.}$$
 (23)

A PLECS simulation was used to tune the component to get exact ZVS (see Fig. 10), the final values are shown in Table VII. As for the class E inverter, the rms current through the inductors and the MOSFET was extracted, see Table V.

With the MOSFET used for the class E the conduction loss will be up to 51.4 mW. The current through L_R will be the same as for the class E and though the inductance is a bit lower, the ESR will still be estimated to 100 m Ω giving a loss of 4.3 mW. The input inductor and $L_{\rm MR}$ are noticeably smaller and their ESR will therefore be estimated to 50 m Ω and 25 m Ω , respectively. With these resistances and the listed rms currents, their loss will be 4.8 and 2.3 mW, respectively. The total loss of the class ϕ_2 inverter (again ignoring losses in the capacitors) is estimated to be 62.8 mW.



Fig. 11. Schematic of the class DE inverter.



Fig. 12. Class DE inverter waveforms simulated with PLECS.

TABLE VI					
CURRENTS (RMS) IN THE CLASS DE INVERTER					
	FET1	FET2	OUT		
	67.8 mA	67.8 mA	208 mA		

C. Class DE

The class DE inverter has the same ZVS properties as the class E inverter and the low voltage stresses of the class D inverter. It is the counterpart of the class DE rectifier considered in Section II and, as seen in Fig. 11, the two circuits are very alike. As the DE rectifier, the DE inverter has two switches connected directly to the dc voltage, in this case MOSFETs connected to the input voltage. Both MOSFETs have capacitors across them which can be tuned to achieve ZVS. The only additional components are a resonant circuit at the output, just as seen for the previous inverters.

As for the class E inverter, ZVS and ZDS switching can be achieved in very specific situations. However, the values needed are different [32]

$$R_L = \frac{V_{\rm IN}^2}{2 \cdot \pi^2 \cdot P_{\rm OUT}} \tag{24}$$

$$f_{S,\max} = \frac{P_{\rm OUT}}{2 \cdot C_S \cdot V_{\rm IN}^2}.$$
(25)

With these equations, the demands for load impedance and output capacitance become $R_L = 126.7 \Omega$ and $C_S = 6.67 \text{ pF}$. Though this cannot be achieved either, these values are much closer to the design values. Increasing P_{OUT} to ≈ 5 W would actually make it possible to use this topology in the ideal situation. The design criteria is however 1 W and the converter will thus be designed to have ZVS.

If the voltage across C_{S1} and C_{S2} are assumed to rise linear when they are charged, the rms value of the voltage at the node between the MOSFETs will be trapezoidal. If the duty cycle of each MOSFET is set to 25%, the rms value can be calculated as $V_{\rm IN}\sqrt{\frac{5}{12}}$. As with the two other inverters, this value can be used to find the needed reactance of the resonant circuit. Using (14) and choosing $C_R = 680$ pF, the value of L_R is calculated to 859 nH. As for the class E inverter, the value of each of C_{S1} and C_{S2} can be found using the reactance of the resonance circuit and scaling them according to the duty cycle. This gives

$$C_S = \frac{1}{2 \cdot (1 - D) \cdot 2 \cdot \pi \cdot f_S \cdot X_R} = 21.9 \text{ pF.}$$
 (26)

As the total voltage across the two MOSFETs always will be $V_{\rm IN}$, the average voltage across each of them is 25 V. The output capacitance of the MOSFET, used for the E and ϕ_2 inverters, almost fit the capacitance needed at this voltage and it will thus be used for the efficiency estimates. However, as the peak voltage across the MOSFETs is limited to the input voltage, several other MOSFETs could be used (or the input voltage could be increased).

Just as the case were for the two other inverters, it was necessary to adjust L_R a bit to give the desired output power. Adjusting the L_R to 550 nH and thus recalculating C_S to 21.4 pF gave the desired output power and the rms currents were extracted (see Fig. 12 and Table VI).

With these currents, the losses in the MOSFETs and the inductor are estimated to be $P_{L_R} = 2.2 \text{ mW}$ (using an ESR of 50 m Ω due to the small inductance) and $P_{\text{FET1}} = P_{\text{FET2}} = 5.5 \text{ mW}$. If the losses in the capacitors are assumed negligible, the total loss will be 13.2 mW.

D. Selection of Inverter

During the analysis of the inverters, the values for all the passive components where found and they are summarized in Table VII.

All the inverters had some pros and cons, thus the same inverter will not be best for all applications. Some of the pros and cons are listed in Table VIII.

The class E inverter consists of only one MOSFET, two inductors, and a capacitor (if C_S is composed by the output capacitance of the MOSFET). It has however the largest voltage peak across the MOSFET which will limit the input voltage for a given MOSFET. Furthermore, the two inductors are both larger than any of the inductors used for the two other converters. This might limit the minimum size of the inverter as inductors are assumed to be the largest components. The total loss was estimated to be 38 mW or $\approx 4\%$ of the output power.

TABLE VII COMPONENT VALUES FOR THE CLASS E, $\phi_2\,$ and DE Inverters

Component	Class E	Class ϕ_2	Class DE
L _{IN}	2.91 μH	794 nH	
L_{MR}		375 nH	
C_{MR}		18.8 pF	
C_S	20 pF	20 pF	2 · 21.4 pF
C_R	680 pF	680 pF	680 pF
L_R	1.43 μH	1.23 μH	550 nH

TABLE VIII PROS AND CONS OF THE INVESTIGATED INVERTER TOPOLOGIES

	Class E	Class ϕ_2	Class DE
Pros	• Low side switch	 Low side switch 	• One inductor
	 Easy tuning 	 Reduced stress 	• Low loss
	• Well documented		• Low stress
Cons	• Large stress	 Largest loss 	• High side switch
	 Large inductors 	• Complex	

TABLE IX COMPARISON OF MOSFET CHARACTERISTICS

Component	V _{DSS}	I_D	$R_{DS(on)}$	CISS	Coss
IRF5802	150 <i>V</i>	0.9A	1.2 Ω	85 pF	18 pF
FDN86246	150 <i>V</i>	1.6A	$359m\Omega$	180 pF	28 pF
$(R_{ds(on)} \text{ at } V_{GS} = 10 \text{ V} \text{ and capacitances at } V_{DS} = 50 \text{ V}).$					

The class ϕ_2 inverter was a lot like the class E, the only difference being the added *LC* circuit put in to reduce the voltage across the MOSFET. While this is a good way of keeping the voltage down, the steep voltage curves require larger currents making the loss larger than seen for the class E inverter. Although it has two extra components, compared to the class E inverter, the physical size is expected to be more or less the same as the values (and thereby the size) of the inductors are smaller. The total loss was estimated to 62.8 mW which is a 65% increase compared to the class E inverter.

The class DE inverter was the only inverter with two switches. However, as C_{S1} and C_{S2} are composed by the parasitic capacitance of the MOSFETs, the total component count of the class DE inverter are the same as for the class E inverter. The peak voltage across the MOSFETs were by far the lowest seen in any of the inverters and the currents were also the lowest. These things combined gave the smallest output inductor (which also is the only inductor) and the lowest losses (13.2 mW).

From this analysis, the class DE inverter seems to be the best solution and the class E inverter comes in second. However, during this analysis the gate drive has not been considered. A good high side gate drive which is capable of operating in the VHF range has yet to be developed whereas a low side gate drive can be made with few components [24]. The complexity, price, and losses associated with the added high side gate drive will, at least, reduce the benefits of the DE inverter.

With the above considerations in mind, the class E inverter was chosen for the final design.

IV. EXPERIMENTAL RESULTS

In Section III, the IRF5802 MOSFET was introduced and used for calculations and loss estimates. Although commercial available MOSFETs are generally designed for either hard switching at a few megahertz or less or for use in RF applications [36], several MOSFETs which can be used for this application exist. Two of the best suited are compared in Table IX.

In Section III, an output capacitance of 10.9 pF was found ideal, but 18 pF was the closest achievable with commercially available MOSFETs. As the voltage waveform across the MOS-FET and C_S is given by V_{IN} , f_S , and D, the currents (and thereby losses) in L_{IN} , C_S , and the MOSFET scale with the value of C_S . It is therefore desirable to keep the value of C_S as close to this as possible in order to achieve high efficiency.

Comparing the two MOSFETs, the IRF5802 (M_1) has much higher on-resistance than the FDN86246 (M_2) . However, the output capacitance is lower and will as mentioned decrease the currents and thus reduce the drawback of the high on-resistance. Assuming the waveform of the voltage across the MOSFET is the same using the two MOSFETs, the current using M_2 will be $\frac{C_{M2}-C_{M1}}{C_{M1}} = 142\%$ larger than using M_1 . The on resistance will be reduced by $\frac{R_{M1}-R_{M2}}{R_{M1}} = 70.1\%$. Combining this gives a total loss reduction of 27.4%, using the estimated loss found in Section III this correspond to 9 mW. Furthermore, the increased current will also give losses in the input inductor, again using the estimate from Section III the increased loss is found to $142\% \cdot 5.37 \text{ mW} = 7.63 \text{ mW}$. Hence, the total loss difference using the two MOSFETs is estimated to be less than 2 mW. The increased capacitance will however also make the timing of the switching more important, as a larger amount of energy will be stored in the capacitor and dissipated in the MOSFET if the switching is just a little wrong.

Based on the analysis above, the MOSFET from IRF are found most suited. But as they are very close, prototypes using both will be made to compare them further.

The next sections will cover the results obtained with three different power stages. The first power stage is the one which has been designed in the previous sections, the second power stage is with the MOSFET with lower $R_{DS(ON)}$ and the last power stage is with a large input inductor and higher output power, this should, as described in Section III, give a higher efficiency.

A signal generator has been used to drive the MOSFETs and hence the design and efficiency of this is not included. The gate signal is however a sinewave which several researchers have shown how to generate efficiently using various types of resonant gate drives, e.g. [37]–[39]. The duty cycle is controlled by adjusting the dc offset of the sinewave, hence a dc offset equal to the threshold of the MOSFET will lead to a duty cycle of 50% and a lower offset lead to a lower duty cycle.

A. MOSFET With Low C_{OSS}

A power stage with the components selected in Sections II and III was implemented. Ceramic COG capacitors were used for the resonating capacitors in order to ensure stable capacitance with varying voltages and ceramic X7R capacitors were used



Fig. 13. Measurements on the prototype with low C_{OSS} .



Fig. 14. Temperature measurements of the MOSFET and the diodes in the prototype with low C_{OSS} ($W_{board} = 40$ mm). (a) Placement of the components. (b) The MOSFET is 55.3 °C. (c) The diodes are 52.2 °C.

TABLE X MEASUREMENTS ON POWER STAGES

Converter	f_S	I_{IN}	VOUT	R_L	η	T _{Mos}	T_{Dio}
	[MHz]	[mA]	[V]	[Ω]	[%]	[°C]	[°C]
Low Coss	30	28	5.00	25	71.5	55.3	52.2
Low R_{DS}	29	32	4.93	25	60.7	65.1	53.2
Large L _{IN}	30	37	5.00	16.3	82.9	46.2	50.5

as input and output capacitors. Custom made air core solenoids were used in order to enable exact tuning of the inductances and thereby achieve ZVS.

Just as the case were when going from calculated values to simulations, slight adjustments had to be made. The tuning procedure was first to tune the inductor in the output filter to make it resistive at the switching frequency. Once that was done, the inverter was added to the design and a low voltage was applied. It was seen that the converter was not ZVS switching as the output capacitance of the MOSFET was not discharged when switched ON.

In order to get it to discharge faster, the values of the input and resonant inductors had to be lowered. First, the resonant inductor was lowered to give the desired output power and then the input inductor was adjusted to make the converter ZVS. As the output capacitance of the MOSFET is 20 pF, measuring with a probe with 10 pF capacitance changes the circuit a lot. This ruins the tuning and ZVS is thus not achieved, some waveforms have been measured and they are shown in Fig. 13. As it is seen the drain voltage has a small break when the MOSFET is switched ON, this is due to the 10 pF added by the probe. The fact that the converter is not ZVS also introduces the miller plateau in the gate charge and causes the gate signal to deviate from a sinewave when the voltage reaches the miller voltage.

So for the final fine tuning a thermal camera was used to measure the temperature of the MOSFET, it was then assumed that the MOSFET was ZVS when the temperature rise was lowest. Even though this is not a solid proof of ZVS, a low temperature rise comes from low power loss and thus the best tuned circuit.

Fig. 14 shows thermal pictures of the converter in a steady state. From the pictures, it is seen that the diodes get almost as hot as the MOSFET. As the size of the components is almost equal this indicates that the total diode loss is almost five times the total MOSFET loss (there are five diodes in parallel). The efficiency was measured to be 71.5% (see Table X).

B. MOSFET With Low R_{ON}

When the MOSFET was selected, the FDN86246 was found to be equally good to the IRF5802. The biggest difference between the two was on-resistances and parasitic capacitances. A prototype was implemented using the FDN86246 in a circuit almost identical to the one used for the previous converter. A few turns were removed from the input inductor in order to make the converter ZVS with the increased output capacitance in the new MOSFET.

The MOSFET gets almost $10 \,^{\circ}$ C warmer (see Fig. 15) clearly indicating a higher loss. Due to the higher output capacitance, more energy is stored and if the switch is switched at a few volts instead of zero, much more energy will be dissipated in the onresistance. Furthermore, the ac current in the input inductor is larger which also increases the losses. The total efficiency of the converter was measured to 60.7%.

C. With Large Input Inductor

As explained in Section III, the highest efficiency should be achieved with a large input inductor (dc input current). To test this, a prototype was made with the IRF5802, but this time with a 6.5 μ H input inductor. Then, the resonance circuit and the load were adjusted in order to get ZVS and 5 V output.

The increased output power makes the current through the MOSFET closer to that seen for the ideal class E inverter. Thereby, the loss due to slight deviations in the timing of the switching becomes smaller.

The waveforms shown in Fig. 16 clearly show that the converter is not ZVS when probes are placed at the gate and drain. Furthermore, the voltage drops below 4 V, however removing the probes makes the output voltage increase to 5.0 V.

When tuned, the output power of the circuit became 1.53 W and the efficiency was measured to 82.9%. This efficiency is without gate drive, but it is still among the best results achieved



Fig. 15. Temperature measurements of the MOSFET and the diodes in the prototype with low $R_{\rm DS(ON)}$ ($W_{\rm board}$ = 32 mm). (a) Placement of the components. (b) The MOSFET is 65.1 °C. The diodes are 53.2 °C.



Fig. 16. Measurements on the 1.53W prototype with IRF5802.



Fig. 17. Temperature measurements of the MOSFET and the diodes in the prototype with large input inductor. (a) Placement of the components. (b) The MOSFET is $46.2 \degree$ C. (c) The diodes are $50.5 \degree$ C.



Fig. 18. The achieved $\frac{P_{O UT}}{V_{IN} \cdot f_S}$ -factor and η next to previous results.

by previous researchers. Furthermore, the $\frac{V_{1N}^2 \cdot f_S}{P_{OUT}}$ -factor explained in Section I is much smaller than for any of the converters shown in Table I.

D. Summary of Experimental Results

The efficiency achieved for the three power stages is shown in Table X. From the three prototypes, it is seen that good efficiencies can be achieved just by having ZVS. However, the larger the current through the MOSFET is at the switching instant, the more important becomes the timing of the switching and losses increase.

It has been shown that VHF converters with a very low $\frac{V_{\text{IN}}^{*} \cdot f_{S}}{P_{\text{OUT}}}$ -factor can be made with high efficiency, the best even had an efficiency of 82.9% which puts it among the best VHF converters. For comparison, the results achieved for the power stages are shown in Fig. 18. The efficiency is not as high as wanted and the factor is a little higher than desired for one of the prototypes due to the higher output power. However, seen next to previously achieved results they are very close.

V. CONCLUSION

The theoretical design of the resonant converter was considered in Sections II and III. Several different topologies were considered and based on complexity and efficiency estimates a class E inverter and rectifier were chosen.

The class E inverter was chosen based on complexity, efficiency, and the fact that it did not require a high side switch. With a simple and efficient high side gate drive the DE inverter is theoretically better, especially for converters with even higher input voltages. Such a gate drive was however yet to be invented and this topology was therefore not used for the practical implementation.

For the rectifier part it was again the class E topology that were chosen, this time due to the forward voltage drop of the diodes. With a low-voltage output, the forward voltage drop of the diode becomes a significant percentage of the output voltage and a single diode rectifier was found to be the best choice. For higher output voltages, the DE rectifier might be better as the loss due to forward voltage drop in the diodes becomes insignificant and the voltages stress of the devices the major concern.

Three different power stages were made; one with a MOS-FET with the lowest available output capacitance, one with a MOSFET with low on-resistance, and one with increased output power allowing a large input inductor. All the converters had 50 V input and 5 V output and the achieved efficiencies were between 60.7% and 82.9%. This shows that it is possible to make low power very high frequency converters with high step down ratio running at subnominal condition as long as the components are chosen carefully.

REFERENCES

- [1] W. C. Bowman, J. F. Balicki, F. T. Dickens, R. M. Honeycutt, W. A. Nitz, W. Strauss, W. B. Suiter, and N. G. Ziesse, "A resonant DC-to-DC converter operating at 22 megahertz," in *Proc. IEEE Third Annu. Appl. Power Electron. Conf. Expo.*, Feb. 1–5, 1988, pp. 3–11.
- [2] Y. Han, G. Cheung, A. Li, C. R. Sullivan, and D. J. Perreault, "Evaluation of magnetic materials for very high frequency power applications," *IEEE Trans. Power Electron.*, vol. 27, no. 1, pp. 425–435, Jan. 2012.
- [3] J. Qiu and C. R. Sullivan, "Design and fabrication of VHF tapped power inductors using nanogranular magnetic films," *IEEE Trans. Power Electron.*, vol. 27, no. 12, pp. 4965–4975, Dec. 2012.
- [4] H. B. Kotte, R. Ambatipudi, and K. Bertilsson, "High-speed (MHz) series resonant converter (SRC) using multilayered coreless printed circuit board (PCB) step-down power transformer," *IEEE Trans. Power Electron.*, vol. 28, no. 3, pp. 1253–1264, Mar. 2013.
 [5] W. Chen and S. Y. Ron Hui, "Elimination of an electrolytic capacitor in
- [5] W. Chen and S. Y. Ron Hui, "Elimination of an electrolytic capacitor in AC/DC light-emitting diode (LED) driver with high input power factor and constant output current," *IEEE Trans. Power Electron.*, vol. 27, no. 3, pp. 1598–1607, Mar. 2012.
- [6] H. Ma, J.-S. Lai, Q. Feng, W. Yu, C. Zheng, and Z. Zhao, "A novel valley-fill SEPIC-derived power supply without electrolytic capacitor for LED lighting application," *IEEE Trans. Power Electron.*, vol. 27, no. 6, pp. 3057–3071, Jun. 2012.
- [7] P. Shamsi and B. Fahimi, "Design and development of very high frequency resonant DC-DC boost converters," *IEEE Trans. Power Electron.*, vol. 27, no. 8, pp. 3725–3733, Aug. 2012.
- [8] O. Garcia, M. Vasic, P. Alou, J. Oliver, and J. A. Cobos, "An overview of fast DC-DC converters for envelope," *IEEE Trans. Power Electron.*, vol. 28, no. 10, pp. 4712–4722, Oct. 2013.
- [9] J. J. Alonso, M. S. Perdigão, D. Vaquero, A. J. Calleja, and E. Saraiva, "Analysis, design, and experimentation on constant-frequency DC-DC resonant converters with magnetic control," *IEEE Trans. Power Electron.*, vol. 27, no. 3, pp. 1369–1382, Mar. 2012.
- [10] K. Fukui and H. Koizumi, "Class E rectifier with controlled shunt capacitor," *IEEE Trans. Power Electron.*, vol. 27, no. 8, pp. 3704–3713, Aug. 2012.
- [11] J. Rivas, "Radio Frequency DC-DC Power Conversion," Ph.D. dissertation, Massachusetts Inst. of Technol., Cambridge, MA, USA, 2006.
- [12] J. R. Warren, III, K. A. Rosowski, and D. J. Perreault, "Transistor selection and design of a VHF DC-DC power converter," *IEEE Trans. Power Electron.*, vol. 23, no. 1, pp. 27–37, Jan. 2008.
- [13] J. Hu, A. D. Sagneri, J. M. Rivas, Y. Han, S. M. Davis, and D. J. Perreault, "High-frequency resonant SEPIC converter with wide input and output voltage ranges," *IEEE Trans. Power Electron.*, vol. 27, no. 1, pp. 189– 200, Jan. 2012.
- [14] J. M. Rivas, O. Leitermann, Y. Han, and D. J. Perreault, "A very high frequency DC-DC converter based on a class φ₂ resonant inverter," *IEEE Trans. Power Electron.*, vol. 26, no. 10, pp. 2980–2992, Oct. 2011.
- [15] D. J. Perreault, J. Hu, J. M. Rivas, Y. Han, O. Leitermann, R. C. N. Pilawa-Podgurski, A. Sagneri, and C. R. Sullivan, "Opportunities and challenges in very high frequency power conversion," in *Proc. IEEE 24th Annu. Appl. Power Electron. Conf. Expo.*, Feb. 15–19, 2009, pp. 1–14.
- [16] J. S. Glaser, J. Nasadoski, and R. Heinrich, "A 900W, 300V to 50V DC-DC power converter with a 30 MHz switching frequency," in *Proc. IEEE 24th Annu. Appl. Power Electron. Conf. Expo.*, Feb. 15–19, 2009, pp. 1121– 1128.
- [17] I. Douglas de Vries, "High power and high frequency class-DE inverters," Ph.D. dissertation, Univ. Cape Town, Cape Town, South Africa, 1999.
- [18] Z. Kaczmarczyk, "High-Efficiency Class E, EF2, and E/F3 Inverters," IEEE Trans. Ind. Electron., vol. IE-53, no. 5, pp. 1584–1593, Oct. 2006.
- [19] J. M. Rivas, Y. Han, O. Leitermann, A. D. Sagneri, and D. J. Perreault, "A high-frequency resonant inverter topology with low voltage stress," *IEEE Trans. Power Electron.*, vol. 23, no. 4, pp. 1759–1771, Jul. 2008.

- [20] Z. Kaczmarczyk, "A high-efficiency class-E inverter—Computer model, laboratory measurements and SPICE simulation," *Bull. Polish Acad. Sci.*, *Tech. Sci.*, vol. 55, no. 4, pp. 411–417, 2007.
- [21] P. Kamby, "High efficiency radio frequency switch-mode power supply for LED applications," Master Thesis, Technical Univ. of Denmark, Kongens Lyngby, Denmark, 2011.
- [22] M. K. Kazimierczuk and J. Jozwik, "Resonant DC/DC converter with class-E inverter and class-E rectifier," *IEEE Trans. Ind. Electron.*, vol. 36, no. 4, pp. 468–478, Nov. 1989.
- [23] J. M. Rivas, D. Jackson, O. Leitermann, A. D. Sagneri, Y. Han, and D. J. Perreault, "Design considerations for very high frequency DC-DC converters," in *Proc. IEEE 37th Power Electron. Spec. Conf.*, Jun. 18–22, 2006, pp. 1–11.
- [24] T. M. Andersen, "Radio frequency switch-mode power supplies," Master Thesis, Technical Univ. of Denmark, Kongens Lyngby, Denmark, 2010.
- [25] J. S. Glaser and J. M. Rivas, "A 500 W push-pull DC-DC power converter with a 30 MHz switching frequency," in *Proc. IEEE 25th Annu. Appl. Power Electron. Conf. Expo.*, Feb. 21–25, 2010, pp. 654–661.
- [26] Y. Han, O. Leitermann, D. A. Jackson, J. M. Rivas, and D. J. Perreault, "Resistance compression networks for radio-frequency power conversion," *IEEE Trans. Power Electron.*, vol. 22, no. 1, pp. 41–53, Jan. 2007.
- [27] R. S. Wahby, "Radio frequency rectifiers for DC-DC power conversion," Master Thesis, Massachusetts Inst. of Technol., Cambridge, MA, USA, 2004.
- [28] D. C. Hamill, "Impedance plane analysis of class DE amplifier," *Electron. Lett.*, vol. 30, no. 23, pp. 1905–1906, Nov. 10, 1994.
- [29] M. K. Kazimierczuk and D. Czarkowski, *Resonant Power Converters*, 2nd ed. New York, NY, USA: Wiley, 2011.
- [30] T. Roy and L. Smith, "ESR and ESL of ceramic capacitor applied to decoupling applications," in *Proc. IEEE Seventh Top. Meet. Electr. Perform. Electron. Packag.*, Oct. 26–28, 1998, pp. 213–216.
- [31] D. C. Hamill, "Class DE inverters and rectifiers for DC-DC conversion," in Proc. IEEE 27th Annu. Power Electron. Spec. Conf., Jun. 23–27, 1996, vol. 1, pp. 854–860.
- [32] M. K. Kazimierczuk, *RF Power Amplifiers*, 1st ed. New York, NY, USA: Wiley, 2008.
- [33] A. Grebennikov and N. O. Sokal, Switchmode RF Power Amplifiers, 1st ed. Burlington, MA, USA: Newnes, 2007.
- [34] M. Hayati, A. Lotfi, M. K. Kazimierczuk, and H. Sekiya, "Performance study of class-E power amplifier with a shunt inductor at subnominal condition," *IEEE Trans. Power Electron.*, vol. 28, no. 8, pp. 3834–3844, Aug. 2013.
- [35] A. Mediano and N. O. Sokal, "A class-E RF power amplifier with a flattop transistor-voltage waveform," *IEEE Trans. Power Electron.*, vol. 28, no. 11, pp. 5215–5221, Nov. 2013.
- [36] A. D. Sagneri, D. I. Anderson, and D. J. Perreault, "Optimization of integrated transistors for very high frequency DC-DC converters," *IEEE Trans. Power Electron.*, vol. 28, no. 7, pp. 3614–3626, Jul. 2013.
- [37] M. Bathily, B. Allard, and F. Hasbani, "A 200-MHz integrated buck converter with resonant gate drivers for an RF power amplifier," *IEEE Trans. Power Electron.*, vol. 27, no. 2, pp. 610–613, Feb. 2012.
- [38] J. M. Burkhart, R. Korsunsky, and D. J. Perreault, "Design methodology for a very high frequency resonant boost converter," *IEEE Trans. Power Electron.*, vol. 28, no. 4, pp. 1929–1937, Apr. 2013.
- [39] H. Fujita, "A resonant gate-drive circuit with optically isolated control signal and power supply for fast-switching and high-voltage power semiconductor devices," *IEEE Trans. Power Electron.*, vol. 28, no. 11, pp. 5423– 5430, Nov. 2013.



Mickey Madsen (S'12) received the B.Sc.E.E. and M.Sc.E.E. degrees from the Technical University of Denmark, Kongens Lyngby, Denmark, in 2009 and 2012, respectively. He is currently working toward the Ph.D. degree in power electronics under the title "Very High Frequency Switch Mode Power Supplies."

His research interests include switch-mode power supplies, resonant inverters/converters, wide band gab semiconductors, solid state (LED) lighting, and radio frequency electronics.



Arnold Knott (M'10) received the Diplom-Ingenieur (FH) degree from the University of Applied Sciences, Deggendorf, Germany, in 2004. He received the Ph.D. degree from the Technical University of Denmark, Kongens Lyngby, Denmark, in 2010, working on a research project under the title "Improvement of outof-band Behavior in Switch-Mode Amplifiers and Power Supplies by their Modulation Topology."

From 2004 to 2009, he worked with Harman/Becker Automotive Systems GmbH in Germany and USA, designing switch-mode audio power am-

plifiers and power supplies for automotive applications. From 2010 to 2013, he was a Assistant Professor and since 2013 an Associate Professor at the Technical University of Denmark. His interests include switch-mode audio power amplifiers, power supplies, active and passive components, integrated circuit design, acoustics, radio frequency electronics, electromagnetic compatibility, and communication systems.



Michael A. E. Andersen (M'88) received the M.Sc.E.E. and Ph.D. degrees in power electronics from the Technical University of Denmark, Kongens Lyngby, Denmark, in 1987 and 1990, respectively.

He is currently a Professor of power electronics at the Technical University of Denmark. Since 2009, he has been the Deputy Director at the Department of Electrical Engineering. He is the author or coauthor of more than 200 publications. His research interests include switch-mode power supplies, piezoelectric transformers, power factor correction, and switch-

mode audio power amplifiers.

A.5 M. Madsen, A. Knott, M. A.E. Andersen: "Very High Frequency Resonant DC/DC Converters for LED Lighting", IEEE Applied Power Electronics Conference and Exposition, Long Beach, CA, March 2013. Conference Proceedings p835-839.

Very High Frequency Resonant DC/DC Converters for LED Lighting

Mickey P. Madsen*, Arnold Knott*, Michael A. E. Andersen* *Technical University of Denmark Ørsteds Plads, bygning 349 2800 Kongens Lyngby Denmark Email: {mpma, akn, ma}@elektro.dtu.dk

Abstract—This paper presents a very high frequency DC/DC converter for LED lighting. Several resonant topologies are compared and their usability discussed. At the end the resonant SEPIC converter is chosen based on the achievable power density and total bill of material. Simulations of a 51 MHz converter with 40 V input and 15 V output are made. The simulation shows possibility of achieving efficiency up to 87 % even with a HEXFET Power MOSFET. Three prototypes of the simulated converter are implemented showing good correlation with simulations. The prototypes have efficiencies up to 84 % and power densities up to 8.9 W/cm³ (146 W/in³).

I. INTRODUCTION

During the last decade the focus on green and environment friendly energy usage has been constantly increasing. This has lead to a large increase in the use of Light-Emitting Diodes (LEDs) for lightning. These bulbs are still quite expensive due to both expensive LEDs and the power converter needed to supply these. Hence there is a strong demand for small, cheap and efficient power converters.

The power density of Switch-Mode Power Supplies (SMPS) is limited by their passive energy storing elements. As both the physical size and price of these scale with the switching frequency (f_S), increasing f_S into the Very High Frequency band (VHF, 30-300 MHz) will make it possible to achieve higher power density and lower cost. Increasing the switching frequency has several other advantages, which has been discussed in [4], [6], [15] and [16].

The high increase in f_S causes several new problems to arise. Some of these have been solved [4], [11] and [18], but many still need to be investigated. One of the main problems is the switching loss, which increases linearly with f_S . As f_S increases into the VHF band the switching losses becomes so severe that it will be impossible to cool the switching device and keep the efficiency high. Several researchers [2]- [5] and [7] have tried to use different types of resonant converters with Zero Voltage Switching (ZVS) and/or Zero Current Switching (ZCS) in order to reduce or ideally eliminate these losses.

Due to the resonating behaviour of these converters it is however very difficult to control these converters for varying loads efficiently. For now the most efficient way is to use burst mode control to simply Pulse Width Modulate (PWM) the converter in order to achieve the desired output [4], [9] and [21]. When working continuously in open loop these converters will have an almost constant current output for a given input voltage. This makes them very well suited for LED applications where it is the current that needs to be controlled. As the output current is constant for a given input voltage the current through the LED will be constant even as the forward voltage changes due to changes in temperature. The life time of LED bulbs are limited by the electrolytic capacitors needed, increasing the switching frequency will eliminated this need and hence increase the life time of the bulb.

This paper will give an example of the design of a VHF resonant DC/DC converter for LED lighting. First an appropriate topology is selected in section II, then the a simulation is made and component selected in section III. Section IV covers the implementation and measurements and finally section V concludes the paper.

II. SELECTION OF TOPOLOGY

When designing resonant DC/DC converters it is very common to split the converter in to two parts; 1) a resonant inverter converting the DC input voltage to a sinusoidal output current 2) a resonant rectifier rectifying the AC current to a DC output [1], [8], and [14].

The most commonly used rectifier is the class E rectifier [8], [17] and [19]. Other alternatives exist [1], [7] and [10], but due to the very simple schematic with a single diode the class E rectifier is chosen for this converter.

For the inverter part several topologies has been used [2], [3] and [5], each with their own pros and cons.

The class DE inverter (used in [5], [10] and [11] and shown together with the selected rectifier in Figure 1) is the topology with the lowest stress on the switches due to the direct connection to the input, but it requires a high side driver. This is complex to design for these frequencies and it will increase both the complexity, size and price of the converter.

The class E inverter (used in [6] and [8]) shown in Figure 2 has only a low side switch and is therefore much simpler to drive, however it imposes a huge voltage stress on the switch. As the drain of the switch is connected to the input through an inductor, the average drain-source voltage (V_{DS}) of the switch has to be equal to the input voltage. Even if V_{DS} is assumed constant when the switch is closed, the peak voltage will be



Fig. 1. Schematic of the class DE inverter and class E rectifier.

two times V_{IN} for a duty cycle of 50 %. In reality the voltage across the switch is more like a half wave rectified sine wave in order to achieve ZVS, which result in a peak voltage of 3.56 times V_{IN} [12].



Fig. 2. Schematic of the class E inverter and rectifier.

In order to reduce this huge peak voltage the class EF_2 (or φ_2), which is a hybrid between the class E and class F_2 , has been developed ([13] and [14]). It introduces an extra resonant circuit (C_{MR} and L_{MR} in Figure 3) across the drain and source of the switch with a zero at the second harmonic of f_S . If tuned correctly this adds the third harmonic of f_S on top of the sine wave seen with the class E. This results in a trapezoidal waveform across the switch. This reduces the peak voltage a bit, but increases complexity and results in additional losses due to large AC current at three times f_S .



Fig. 3. Schematic of the class EF₂ inverter and class E rectifier.

The SEPIC converter [20] shown in Figure 4 is similar to the circuit in Figure 2 with L_T removed. However the waveforms are different, as this converter cannot be split into an inverter and a rectifier. The changed waveforms results in a much smaller input inductor than needed for the class E inverter, thus the achievable power density is higher due to fewer and smaller inductors.

Based on the analysis of the four converter topologies the SEPIC converter was chosen as it gives the highest power density and lowest cost.

III. SIMULATION AND COMPONENT SELECTION

A model of a resonant SEPIC converter has been set up in spice based on the tuning procedure explained in [20]. The converter is designed to have the specifications given in table



Fig. 4. Schematic of the SEPIC converter.

I. The converter will be used to supply a string of LEDs with a combined forward voltage drop of 12-15 V, depending on temperature and power level.

TABLE I DESIGN SPECIFICATIONS

Specification	Symbol	Value
Input voltage	V_{IN}	40 V
Output power	P_{OUT}	5 W
Output Voltage	V_{OUT}	15 V
Switching frequency	f_S	51 MHz

From the simulations it is seen that the MOSFET should have a break down voltage of at least 100V (see Figure 6), however if the duty cycle is adjusted closer to 50 % the peak voltage will get close to 143 V (3.56 times the input voltage as for the class E). For this reason it was decided to build the prototype around an IRF5802 MOSFET from International Rectifier. The MOSFET has a break down voltage of 150 V and small parasitic capacitances compared to its competitors.

The peak voltage is slightly above 40 V (see Figure 6 where V_{AC} is anode-cathode voltage) and a MBR0540 40 V Schottky diode has therefore been selected. C_R needs to be 105 pF which is more than the parasitic capacitance of a single diode (35 pF), thus it is necessary either to add a 70 pF capacitor or use three diodes in parallel. The last solution have the benefit of sharing the current between the three devices. As the forward voltage drop of the diodes increases with the current running through them, this will lead to reduced losses and this solution was therefore selected.

The inductors are all square air core inductors (1515SQ-68N, 1515SQ-82N and 2222SQ-161) as they have a fairly high Q factor and are available off the shelf which ease implementation. The C_T capacitor is implemented with 4 parallel capacitors as it was found that this increased the efficiency of the converter with 1-2 % compared to using a single capacitor of the same value. For the input and output capacitors standard 1 μ F X7R capacitors was selected.

IV. IMPLEMENTATION AND MEASUREMENTS

A prototype of the simulated converter has been implemented and is shown in Figure 7. The prototype had an efficiency of 84 %, this is slightly lower than the simulated efficiency but still on level with other state of the art VHF converters. The output power was 5.7 W at 40 V input, which is significantly higher than the expected. The increase in output



Fig. 5. Simulation results

TABLE II BILL OF MATERIALS

Component	Simulated	Prototype	Туре
C_{IN}	$1 \ \mu F$	$1 \ \mu F$	Capacitor (X7R)
L_{IN}	160 nH	160 nH	Inductor
S		IRF5802	n-channel MOSFET
C_S	$22 \ pF$	Parasitic	Capacitor
C_T	40 pF	$4 \cdot 10 \ pF$	Capacitor (NPO)
L_R	82 nH	82 nH	Inductor
D		3 · MBR0540	Schottky diode
C_R	105 pF	Parasitic	Capacitor
C_{OUT}	$1 \ \mu F$	$1 \ \mu F$	Capacitor (X7R)

power was achieved by reducing the switching frequency to 46 MHz and increasing the duty cycle.

The gate and output voltages were measured with the standard probes for a Rohde & Schwartz RTO1024 digital oscilloscope. The capacitance of the probes are 10 pF, adding this between the drain and source of the MOSFET would change the resonance of the converter making it hard switch and thereby destroy the switch. To avoid this a voltage divider was made by adding a 1 pF capacitor in series with the probe, this reduces the effectively added capacitance to 0.9 pF. This



Fig. 6. Measured waveforms



Fig. 7. Picture of the implemented prototype on a 15x22 mm PCB

still affect the behaviour of the converter, however as seen in Figure 6 the converter is still operating close to ZVS. In order to get the correct DC level a resistive voltage divider was added in parallel with the probe and the 1 pF capacitor. The same principle was used to measure the voltage at the anode of the diode, at this node 10pF would not be as crucial but it is still necessary to keep the converter close to ZVS.

From the measured waves it seems like the gate drive is a few nano second to slow when turning on the MOSFET. Whether this is actually happening or if it is due to a small delay caused by the resistive/capacitive voltage division made to measure the drain voltage is not certain. If the gate drive is actually turning on the MOSFET to late, it is partly saved



Fig. 8. Comparison of first VHF based prototype and commercial available 230 V power supply



Fig. 9. Top side of prototype with PCB inductors (10x16mm PCB)

by the body diode of the MOSFET and the efficiency can therefore still be acceptable. However in this situation fine tuning of the gate drive could increase efficiency a few percent as the body diode is a quite lossy device.

The peak voltage across the diode is far above the rated limit. This is mainly caused by the reduced switching frequency which makes the impedance of the inductor decrease while the impedance of the parasitic capacitance increase, this increases the duty cycle of the diode and there by increases the peak voltage. The increased power is also contributing to an increased peak voltage, but this will only cause a slight increase.

The ripple seen at the output is clearly some digital noise cause by the oscilloscope and not something that can be used to determine the output ripple of the converter.

A. Alternative implementations

The main reason for the increasing the output power was to get the same amount of power as the commercial supply shown in Figure 8(b), 8(d) and 8(f). This supply is made for the european mains (230 V) and has an efficiency of 77 %, i.e. 7 % below the 40 V VHF converter with the same output in a much smaller footprint. In order to make a fair comparison a prototype was made with a mains rectifier and a capacitive voltage divider to reduce the input voltage of the VHF converter (the offline prototype is shown in Figure 8(a), 8(c) and 8(e)). The offline prototype has an efficiency of 78 % and a volume which is reduced by approximately 80 %compared to the commercial supply. Hence it offers slightly increased efficiency in a dramatically reduced volume.

It should however be noted that the power factor of the offline prototype is very poor (below 0.5) whereas the commercial has a power factor of 0.82. Furthermore the commercial supply is galvanic isolated which the offline prototype is not.

In order to push the power density even further, a prototype has been made were the inductors are implemented as PCB solenoids. The complete converter has a footprint of 16x10x4 mm (LxWxH) and can handle an output power of up to 5 W. This gives a power density of 7.8 W/cm³ and if the connectors (2 times 2x5 mm on the PCB) are disregarded the power density is 8.9 W/cm³ (146 W/inch³). All the components used for this prototype are the same as for the one shown in Figure 7 (except for the inductors), but both sides of the PCB has been used. The series resistances of the PCB inductors are a

bit higher than their discrete counter parts and this causes the efficiency to drop to 80 %.

V. CONCLUSION

A very compact and cheap converter with state of the art efficiency for VHF converters has been implemented. It is shown that resonant VHF converters using standard MOSFETs can be implemented reducing price and making it possible to design for higher input voltages.

The paper has covered three different prototypes; 1) a basic design of a resonant SEPIC converter with a standard nchannel MOSFET and 84 % efficiency, 2) a offline (230 V_{AC}) prototype with a volume reduced by 80 % compared to a commercial supply and 3) a prototype with PCB inductors and a power density of 8.9 W/cm³.

REFERENCES

- Marian K. Kazimierczuk & Dariusz Czarkowski, Resonant Power Converters, John Wiley & Sons, New. York, 2011.
- [2] Marian K. Kazimierczuk, *RF Power Amplifiers*, John Wiley & Sons, New. York, 2008.
- [3] Andrei Grebennikov & Nathan O. Sokal, Switchmode RF Power Amplifiers, Newnes, Burlington, 2007
- [4] Juan Rivas, Radio Frequency dc-dc Power Conversion, Doctoral Thesis, Massachusetts Institute of Technology, 2006.
- [5] Ian Douglas de Vries, High Power and Frequency Class-DE Inverters, Doctoral Thesis, University of Cape Town, 1999.
- [6] Toke M. Andersen, A VHF Class E DC-DC Converter with Self-Oscillating Gate Driver, in IEEE 2011 Applied Power Electronics Conference, 2011, pp. 885-891.
- [7] W. C. Bowman, J. F. Balicki et al., A resonant DC-to-DC converter operating at 22 megahertz, in IEEE 1988 Applied Power Electronics Conference, 1988, pp. 3-11.
- [8] Marian K. Kazimierczuk and Jacek Jozwik, Resonant dc/dc Converter with Class-E Inverter and Class-E Rectifier, in IEEE Transactions on Industrial Electronics, vol. 36, no. 4, pp. 468-478, Nov 1989.
- [9] James R. Warren, III, Kathryn Anne Rosowski and David J. Perreault, Transistor Selection and Design of a VHF DC-DC Power Converter, in IEEE Transactions on Power Electronics, vol. 23, no. 1, pp. 27-37, Jan. 2008.
- [10] David C. Hamill, Class DE Inverters and Rectifiers for DC-DC Conversion, in IEEE 1996 Power Electronics Specialists Conference, 1996, pp. 854-860.
- [11] David C. Hamill, Impedance plane analysis of class DE amplifier, in IEEE Electronics Letters, vol. 30, no. 23, pp. 1905-1906, 10 Nov 1994.
- [12] Z. Kaczmarczyk, High-Efficiency Class E, EF2, and E/F3 Inverters, in IEEE Transactions on Industrial Electronics, vol. 53, pp. 15841593, Oct. 2006.
- [13] Juan M. Rivas, Yehui Han et al., A High-Frequency Resonant Inverter Topology with Low Voltage Stress, in IEEE Transactions on Power Electronics, vol. 23, no. 4, pp. 1759- 1771, July 2008.
- [14] Juan M. Rivas, Olivia Leitermann et al., A Very High Frequency dc-dc Converter Based on a Class φ₂ Resonant Inverter, in IEEE 2008 Power Electronics Specialists Conference, 2008, pp.1657-1666.
- [15] John S. Glaser and Juan M. Rivas, A 500 W Push-pull Dc-dc Power Converter with a 30 MHz Switching Frequency, in IEEE 2010 Applied Power Electronics, 2010, pp. 654- 661.
- [16] David J. Perreault, Jingying Hu, Juan M. Rivas et al., Opportunities and Challenges in Very High Frequency Power Conversion, in IEEE 2009 Applied Power Electronics Conference, 2009, pp.1-14.
- [17] Juan M. Rivas, David Jackson, Olivia Leitermann et al., Design Considerations for Very High Frequency dc-dc Converters, in IEEE 2006 Power Electronics Specialists Conference, 2006, pp.1-11.
- [18] Yehui Han, Olivia Leitermann, David A. Jackson et al., Resistance Compression Networks for Radio-Frequency Power Conversion, in IEEE Transactions on Power Electronics, vol. 22, no. 1, pp. 41-53, Jan. 2007.
- [19] John S. Glaser, Jeffrey Nasadoski and Richard Heinrich, A 900W, 300V to 50V Dc-dc Power Converter with a 30 MHz Switching Frequency, in IEEE 2009 Applied Power Electronics Conference, 2009, pp. 1121-1128.

- [20] Jingying Hu, Anthony D. Sagneri, Juan M. Rivas et al., *High Frequecny Resonant SEPIC Converter with Wide Input and Output Voltage Ranges*, in *IEEE 2008 Power Electronics Specialists Conference*, 2008, pp.1397-1406.
- [21] Robert C. N. Pilawa-Podgurski, Anthony D. Sagneri, Juan M. Rivas et al., Very High Frequency Resonant Boost Converters, in IEEE 2007 Power Electronics Specialists Conference, 2007, pp. 2718-2724.

A.6 M. Madsen, A. Knott, M. A.E. Andersen: "Very High Frequency Half Bridge DC/DC Converter", IEEE Applied Power Electronics Conference and Exposition, Fort Worth, TX, March 2014. Conference Proceedings p1409-1414.

Very High Frequency Half Bridge DC/DC Converter

Mickey P. Madsen*, Arnold Knott*, Michael A. E. Andersen*

*Technical University of Denmark

Ørsteds Plads, building 349

2800 Kongens Lyngby

Denmark

Email: {mpma, akn, ma}@elektro.dtu.dk

Abstract—This paper presents the first, off chip, class DE (resonant half bridge) converter working in the Very High Frequency (VHF) range. The benefits of using half bridge circuits both in the inverter and rectifier part of a VHF resonant dc/dc converter are analyzed and design equations for all components in the power stage are given. The circuit has been simulated to verify the accuracy of the presented equations and an efficiency of 89% has been shown.

A prototype has been implemented with self-oscillating resonant gate drives driving the switches. The prototype has been used to drive an LED string and shows an efficiency of 85% at 29 MHz with 130 V input and 13.4 W output. The efficiency was above 82% in the range 110-150 V input with output power between 10.3 W and 16.5 W.

I. INTRODUCTION

In the early 70s the constant strive for small, cheap and efficient power supplies lead to the development of Switch-Mode Power Supplies (SMPS) [1]. As the size of modern power supplies are mainly governed by the passive energy storing elements, which scales inversely with the switching frequency, this strive has lead to constantly increasing switching frequencies ever since. Commercially available converters today switch at frequencies up to several megahertz and can have efficiencies of more than 95% (e.g. [2]).

The reason not to increase the switching frequency further and thereby reaching even higher power densities are the switching losses. In order to avoid switching losses and be able to increase the frequency while keeping the efficiency high, new topologies have to be used. For more than two decades (since 1988 [3]) research has been done in order to enable the use of resonant RF amplifiers (inverters) combined with a rectifier for dc/dc converters. With this type of converters SMPS with switching frequencies in the Very High Frequency range (VHF, 30-300 MHz) have been designed with efficiencies up to approx. 90%, [4], [5].

Several of the benefits and challenges of the increased switching frequency are described in [6], [7]. One of the great benefits in terms of size and price is, that if the frequency is increased enough, some of the components can be left out as they can be constituted by the parasitic parts of other components or embedded in the PCB [8], [9]. One of the remaining challenges is the huge peak voltage across the MOSFET seen in most resonant circuits [10], [11]. The class DE inverter does not have this stress as the two MOSFETs act as each others clamps, but requires a high side gate drive which is another challenge.

In section II of this paper different resonant topologies suited for switching frequencies in the VHF range are compared. A new type of self-oscillating resonant gate drive suitable for high side gate drive at VHF is introduced in III. The design equations are verified in section IV and the theoretical efficiency is calculated. In section V experimental result are shown and finally section VI concludes the paper.

II. COMPARISON OF SINGLE AND TWO SWITCH TOPOLOGIES

The converter analyzed and implemented in this paper is designed to meet the specifications listed in table I. The load could either be resistive or appear as a voltage source, e.g. a battery or an array of LEDs.

TABLE I DESIGN SPECIFICATION FOR THE CONVERTER.

f_S	V_{IN}	V_{OUT}	P_{OUT}	R_L
30 MHz	100 V	40 V	10 W	160 Ω

Resonant converters are often designed in two parts, an inverter converting the dc input voltage to an ac current and a rectifier converting the ac current to a dc output voltage [12], [13]. The two parts are designed individually, but the design of the inverter is dependent on the input impedance of the rectifier.

It is possible to design the inverter to a load, which give optimal operating conditions for the inverter, and then use different resistance compression networks to make the impedance of the rectifier match [14]–[16]. Though this is a solution often used, it will increase the complexity of the converter unnecessarily and possibly reduce the achievable efficiency, size, and price. Therefore the design procedure will be firstly to design a rectifier for the given load and then design the inverter for the given input and rectifier impedance.

The rectifiers most commonly used in resonant converters switching in the VHF range are class E derived topologies [17], [18], consisting of a low pass filter with a cut of frequency at the switching frequency, a diode in parallel with the capacitor and an output capacitor to insure a constant output voltage. These all have a peak voltage across the diode of approx. 3.6 times the output voltage for a duty cycle of 50% [19], [20]. With an output voltage of 40 V this gives a peak diode voltage stress of almost 150 V, greatly limiting the number of suitable schottky diodes.

For this output voltage a half bridge solution (class DE [21]) is more suitable as the diodes are connected directly to the output and the voltage across them hence is limited to the output voltage (see Fig. 1). Furthermore the class DE rectifier is inductor less and hence offers a much higher power density, as the inductor is generally the biggest component.



Fig. 1. Schematic of the class DE rectifier.

For the inverter part it is again the class E derived topologies that are most commonly used, this could be either a class E [22], [23], a class EF₂ (ϕ_2) [24], [25], a resonant SEPIC [26], [27] or a resonant boost converter [10], [28]. This is generally chosen due to complexity and losses connected with a high side gate drive for operation in the VHF range.

As for the rectifier, a class E derived inverter imposes huge voltage stress across the MOSFET. For the class E, the resonant SEPIC and resonant boost it is 3.6 the input voltage for a duty cycle of 50% [10], [11], for the class EF₂ this is reduced to approximately 2.3-3 times [13], [25]. As for the rectifier, the semiconductors in the class DE inverter are directly connected to the input and the voltage across them is limited to the input voltage (see Fig. 2). The class DE inverter has two other great advantages over the other topologies, it has only a single inductor and due to the lower peak voltage across the MOSFET the stored energy is approximately ten times lower (E = $\frac{1}{2} \cdot C_{OSS} \cdot V^2$), leading to much smaller resonating currents in a buck type converter [29].



Fig. 2. Schematic of the class DE inverter.

If the maximum output voltage ripple is set to 1% (0.4 V) the values for C_{OUT1} and C_{OUT2} is calculated from eq. 1.

$$C_{OUT} = \frac{I_{OUT}}{V_{OUT,PP}} \cdot \frac{1 - D_D}{f_S} \tag{1}$$

The minimum diode duty cycle D_D is set by the parasitic capacitance of the diodes. Additional capacitance can be added to decrease the duty cycle, this will however increase the amount of resonating currents running in the converter and is hence not desired. The duty cycle for a given capacitance can be calculated according to eq. 2 [30].

$$C_{R} = \frac{\pi \cdot (1 - \cos(\pi - 2 \cdot D_{D} \cdot \pi))}{\omega \cdot R_{L} \cdot (1 + \cos(\pi - 2 \cdot D_{D} \cdot \pi))}$$

$$\updownarrow$$

$$D_{D} = \frac{1}{2 \cdot \pi} \cdot \arccos\left(\frac{C_{R} \cdot R_{L} \cdot \omega - \pi}{C_{R} \cdot R_{L} \cdot \omega + \pi}\right)$$
(2)

The input resistance of the rectifier is calculated through equations from [30] combined to eq. 3.

$$R_{REC} = \frac{2 \cdot R_L}{\pi \cdot (C_R \cdot R_L \cdot \omega + \pi)} \tag{3}$$

If the voltages on both sides of the resonant circuit (C_T and L_T) are assumed to be trapezoidal, the rms voltages in those nodes can be calculated according to eq. 4 and 5.

$$V_{DS,rms} = V_{IN} \cdot \sqrt{\frac{D_M + 1}{3}} \tag{4}$$

$$V_{REC,rms} = V_{OUT} \cdot \sqrt{\frac{D_D + 1}{3}} \tag{5}$$

The reactance of the resonance circuit is calculated according to eq. 6 [31].

$$X_{RC} = R_{REC} \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1} \tag{6}$$

Once the reactance of the resonant circuit is calculated, a suitable capacitance can be choosen and the inductance calculated. The capacitors C_{S1} and C_{S2} need to resonate with the resonant circuit at a frequency given by:

$$\omega_R = \frac{\omega}{2 \cdot (1/2 - D_M)} \tag{7}$$

Hence:

$$\omega_R^2 = \frac{1}{2 \cdot C_S \cdot \frac{X_{RC}}{\omega_R}} \tag{8}$$

Combining eq. 7 and 8 leads to the following expression for the value of C_{S1} and C_{S2} :
$$C_S = \frac{1/2 - D_M}{\omega \cdot X_{RC}} \tag{9}$$

Finally the input capacitance is calculated in the same way as the output capacitance.

III. SELF-OSCILLATING RESONANT GATE DRIVE

The class DE inverter requires a high side gate drive and the design of this for VHF have not been presented in any previous publications. The circuits shown at the gates of the MOSFET in Fig. 3 constitutes self-oscillating resonant gate drives capable of making this high side gate signal.

The inductors L_{G1} and L_{G2} combined with the parasitic capacitances of the switches constitutes high pass filters from the drains to the gates of the MOSFETs [32]. If the high pass filters are designed to have the right gain and phase at the resonance frequency of the power stage, a sine wave will apear on the gates of the switches and the inverter will be self oscillating.

The sine wave will have an amplitude given by the gain of the high pass filter at the switching frequency and the peak voltage across the MOSFET. Hence the right gain with 100 V input and a desired gate signal of 12 V_{PP} is -18.4 dB. The phase shift should be as close to 180° in order to avoid conduction through the body diode.

In order to design the gate drive a MOSFET has to be chosen so that its parasitics can be taken into account. International Rectifiers IRF5802 HEXFET is chosen to be used due to its 150 V break down and low parasitic capacitances. The 150 V leaves plenty of headroom with 100 V input and the small capacitances leads small resonating currents in the converters. The most important specifications have been extracted from the datasheet and shown in table II.

TABLE IIPARASITIC COMPONENTS OF INTERNATIONAL RECTIFIERS MOSFETIRF5802. PARASITIC CAPACITANCES TAKEN FOR $V_{DS} = V_{IN}/2$.

$$C_{DS}$$
 C_{GS}
 C_{GD}
 R_{ON}

 13 pF
 82 pF
 6 pF
 1.2 Ω

 L_G should now be selected so that the resonance frequency of L_G and the parasitics of the MOSFET is well above the switching frequency, in order to insure close to 180° phase shift, but still close enough to ensure the desired gain from drain to source. This can be calculated analytically, but a simple simulation is a faster and more practical approach. Fig. 4 shows the gain and phase with to values of L_G and with and without 39 pF added between the gate and source of the MOSFET.

From the plot it can be seen that the desired response can be achieved either with a 200 nH inductor or with a combination of a 150 nH inductor and a 39 pF capacitor. The phase and gain of the two designs are more or less exactly the same at the switching frequency and the solution with the capacitor is chosen as it will consume least volume.



Fig. 3. Schematic of the class DE inverter with self oscillating gate drive.



Fig. 4. Gain and phase of the filter from drain to source for different values of C_{GSext} and L_G .

The bias voltages V_{Bias1} and V_{Bias2} can be used to regulate the average of the sine wave and hence shift it up and down in order to control the duty cycle. The low side switch is connected to ground and the V_{Bias2} can therefore be connected directly to the inductor. For the high side switch an additional inductor and capacitor is required. C_{G1} insures that the node between L_H and L_{G1} has a fixed dc offset from source of the high side switch, just like V_{Bias2} for the low side gate drive. The value of C_{G1} should therefore be large enough to keep the voltage across is almost constant for a switching period. L_H is functioning as a choke only allowing dc current to run through it to set the dc voltage across C_{G1} . Setting this 20 times larger than L_G and C_{G1} to 100 nF works well in the VHF range.

IV. SIMULATION AND FINE TUNNING

The complete circuit with rectifier, inverter and selfoscillating gate drives are shown in Fig. 5. The MOSFETs for the inverter have already been selected in order to design the gate drive in section III. In order to design and simulate the rest of the converter the diodes have to be chosen as well. The MBR0540 schottky diodes are chosen as they have low forward voltage drop. The breakdown voltage of these diodes are 40 V, hence there is no headroom to the 40 V output voltage. Two will be used in parallel to reduce the current through a single device. Due to the parasitic capacitance of these diodes C_R will be 40 pF resulting in a diode duty cycle of 32%.

Based on the equations from section II the values of all components have been calculated (see table III). The equations for resonant converters are all based on several assumptions, e.g. pure sine or trapezoidal waves, and the calculated values therefore often have to be fine-tuned [24], [33]. A Spice simulation has been used for this purpose and the simulated waveforms are shown in Fig. 6.

TABLE III CALCULATED AND TUNED COMPONENT VALUES.

Component	Calculation	Simulation	
C_{IN}	4.7 nF	4.7 nF	
C_S	14 pF	14 pF	
C_{GD}	6 pF	6 pF	
C_{GS}	82 pF	82 pF	
C_{GSext}	39 pF	39 pF	
L_G	150 nH	150 nH	
C_{G1}	100 nF	100 nF	
L_H	3 uH	3.3 uH	
C_T	500 pF	500 pF	
L_T	338 nH	390 nH	
C_R	40 pF	40 pF	
C_{OUT}	14 nF	14 nF	

As ceramic capacitors have very high Q factors (more than 1000 [33], [35]), the components causing the major parts of the losses are the MOSFETs, the inductor and the diodes. The losses in the MOSFETs and the inductor are ohmic, the RMS value of the current through these components are therefore measured and shown in table IV. In the diodes the losses are caused by a forward voltage drop and it is hence the average current which is important. The individual component losses and the total loss are listed in table IV and it is seen that an efficiency of 89% (excluding gating losses and drive) is achievable.

The converter is designed with two 150 V HEXFETs [36]. The MOSFET has an on resistance of up to 1.2 Ω at V_{GS} = 10 V and more than 2 Ω when the gate voltage drops below 6.5 V. This causes a significant amount of conduction loss in the MOSFETs, but the low parasitic capacitances loweres the resonating current.

V. IMPLEMENTATION AND VALIDATION

The inverter was designed for 100V input, 30 MHz switching frequency and 10W output power. The converter was implemented according to the simulated values and with square air core inductors from CoilCraft for the resonant tank and gate drive. The output power at 100 V input was 10.14 W,



Fig. 5. Schematic of the class DE inverter with DE rectifier and self oscillating gate drive.



Fig. 6. Simulated waveforms of the class DE inverter and rectifier (solid = calculated values, dashed = tuned values).

TABLE IV CURRENTS AND ASSOCIATED LOSSES IN THE COMPONENTS CAUSING THE MAIN LOSS.

Component	Current	R or V_F	Loss
S_1	430 mA	2 Ω	370 mW
L_T	777 mA	$408 \text{ m}\Omega$	246 mW
D_1	0.25 A	460 mV	115 mW
Total			1.22 W

hence very close to the design specifications. The switching frequency was however 29 MHz and the efficiency 77.1% both a bit lower than simulated. A picture of the prototype is shown in Fig. 7 and the thermal picture in Fig. 8 clearly shows the heat caused by the conduction loss in the MOSFETs.

By reducing the inductance in the inductor in the resonant tank to 300 nH, it was possible to increase the switching frequency to 30.5 MHz. This did however also lead to increased output power and changed the efficiency. The output power and efficiency for different values of L_{RC} can be seen in Fig. 9. As it can be seen the peak efficiency is almost 80%, but generally drops below 75% when the output power is increased above 12 W.

Another way to change the switching frequency is to adjust the gate drive. The output power and efficiency for three values of C_{GSext} can be seen in Fig. 10. The plot shows that it is possible to adjust the output power a little bit in this way, but the efficiency remains almost constant.

The implemented converter is technically galvanic isolated by C_T and C_{OUT} . It is therefore possible to stak the input of the converter on top of the output and increase the output power and input voltage in that way. This leads to the input voltage, output power and efficiency shown in Fig. 11. In this way the efficiency is increased to more than 85% and the output power above 15 W.



Fig. 7. Picture of the implemented converter on a 50x17 mm PCB.



Fig. 8. Measured temperatures of the MOSFETs in the class DE inverter at 130 V input and 13.4 W output.



Fig. 9. Output power and efficiency for different values of L_T (C_{GSext} = 39 pF).



Fig. 10. Output power and efficiency for different values of C_{GSext} (L_T = 300 nH).



Fig. 11. Output power and efficiency when the input and output of the converter is connected in series.

VI. CONCLUSION

This paper has shown the design and implementation of the first (discrete) class DE converter working in the VHF range [37], [38]. Accurate equations, which almost eliminate the need for tuning of the resonant circuit, have been presented and demonstrated working.

A new type of self-oscillating gate drive which can be used to drive the high side switch has been presented and the design procedure explained.

Simulations have been presented which shows the accuracy of the formulas and the function of the gate drive. The simulation shows an efficiency of 89%, without gate drive and gating losses, when a HEXFET with high on resistance is used.

The implemented converter has a peak efficiency of 85% and above 82% for a wide input range when the input and output is connected in series across the supply. When used as an isolated converter the peak efficiency is almost 80%.

References

- R.B. Peterson, Switch Mode Power Supplies A New Approach for Consumer Audio Systems, IEEE Transactions on Broadcast and Television Receivers, vol.BTR-19, no.4, pp.263-268, Nov. 1973
- [2] ON Semiconductor, 1.2A, 2MHz Automotive Buck Switching Regulator, NCV890100 - Datasheet, Rev. 2, February 2013
- [3] W. C. Bowman, J. F. Balicki, F. T. Dickens et al., A resonant DC-to-DC converter operating at 22 megahertz, IEEE Applied Power Electronics Conference and Exposition, 1988, pp.3-11, 1-5 Feb 1988
- [4] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, *High-Frequency Resonant SEPIC Converter With Wide Input* and Output Voltage Ranges, IEEE Transactions on Power Electronics, vol.27, no.1, pp.189-200, Jan. 2012
- [5] J.M. Rivas, O. Leitermann, Yehui Han, D.J. Perreault, A very high frequency dc-dc converter based on a class 2 resonant inverter, IEEE Power Electronics Specialists Conference, 2008, vol., no., pp.1657-1666, 15-19 June 2008
- [6] D.J. Perreault, Jingying Hu, J.M. Rivas, Yehui Han, O. Leitermann, R.C.N. Pilawa-Podgurski, A. Sagneri, C.R. Sullivan, Opportunities and Challenges in Very High Frequency Power Conversion, IEEE Applied Power Electronics Conference and Exposition, 2009, vol., no., pp.1-14, 15-19 Feb. 2009
- [7] A. Knott, T.M. Andersen, P. Kamby, M.P. Madsen, M. Kovacevic, M.A.E. Andersen, On the ongoing evolution of very high frequency power supplies, IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.2514,2519, 17-21 March 2013
- [8] M.P. Madsen, A. Knott and M.A.E. Andersen, Very high frequency resonant DC/DC converters for LED lighting, IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.835,839, 17-21 March 2013
- [9] M.P. Madsen, A.P. Mynster, A. Knott and M.A.E. Andersen, Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies, IEEE International Energy Conversion Congress and Exhibition Asia 2013
- [10] J.M. Burkhart, R. Korsunsky, D.J. Perreault, Design Methodology for a Very High Frequency Resonant Boost Converter, IEEE Transactions on Power Electronics, vol.28, no.4, pp.1929,1937, April 2013
- [11] M.K. Kazimierczuk, D. Czarkowski, Resonant Power Converters, John Wiley & Sons, 2nd Edition, 2011, ISBN: 978-0-470-90538-8
- [12] T.M. Andersen, S.K. Christensen, A. Knott, M.A.E Andersen, A VHF class E DC-DC converter with self-oscillating gate driver, IEEE Applied Power Electronics Conference and Exposition, 2011, vol., no., pp.885,891, 6-11 March 2011
- [13] R.C.N. Pilawa-Podgurski, A.D. Sagneri, J.M. Rivas, D.I. Anderson, D.J. Perreault, Very-High-Frequency Resonant Boost Converters, IEEE Transactions on Power Electronics, vol.24, no.6, pp.1654,1665, June 2009
- [14] Yehui Han, O. Leitermann, D. A. Jackson et al. Resistance Compression Networks for Radio-Frequency Power Conversion, IEEE Transactions on Power Electronics, vol.22, no.1, pp.41-53, Jan. 2007

- [15] R. S. Wahby, Radio Frequency Rectifiers for DC-DC Power Conversion, Master Thesis, Massachusetts Institute of Technology, 2004
- [16] D. C. Hamill, Impedance plane analysis of class DE amplifier, Electronics Letters, vol.30, no.23, pp.1905-1906, 10 Nov 1994
- [17] M. K. Kazimierczuk, J. Jozwik, Resonant dc/dc Converter with Class-E Inverter and Class-E Rectifier, IEEE Transactions on Industrial Electronics, vol.36, no.4, pp.468-478, Nov 1989
- [18] J. M. Rivas, D. Jackson, O. Leitermann et al., Design Considerations for Very High Frequency dc-dc Converters, IEEE Power Electronics Specialists Conference, 2006, vol., no., pp.1-11, 18-22 June 2006
- [19] A. Grebennikov, N. O. Sokal, Switchmode RF Power Amplifiers, Newnes, 1st Edition, 2007, ISBN: 978-0-7506-7962-6
- [20] M. K. Kazimierczuk, RF Power Amplifiers, John Wiley & Sons, 1st Edition, 2008, ISBN: 978-0-470-77946-0
- [21] D.C. Hamill, Half bridge class DE rectifier, Electronics Letters, vol.31, no.22, pp.1885,1886, 26 Oct 1995
- [22] Z. Kaczmarczyk, High-Efficiency Class E, EF2, and E/F3 Inverters, IEEE Transactions on Industrial Electronics, vol. IE-53, pp. 15841593, Oct. 2006
- [23] Z. Kaczmarczyk, A high-efficiency Class-E inverter computer model, laboratory measurements and SPICE simulation, Bulletin Of The Polish Academy Of Sciences, Technical Sciences, Vol. 55, No. 4, 2007
- [24] A.D. Sagneri, D.I. Anderson, D.J. Perreault, Optimization of Integrated Transistors for Very High Frequency DCDC Converters, IEEE Transactions on Power Electronics, vol.28, no.7, pp.3614,3626, July 2013
- [25] J. M. Rivas, O. Leitermann, Y. Han and D. J. Perreault, A Very High Frequency dc-dc Converter Based on a Class φ₂ Resonant Inverter, IEEE Transactions on Power Electronics, vol.26, no.10, pp.2980,2992, Oct. 2011
- [26] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, *High-Frequency Resonant SEPIC Converter With Wide Input* and Output Voltage Ranges, IEEE Transactions on Power Electronics, vol.27, no.1, pp.189,200, Jan. 2012
- [27] M. Kovacević, A. Knott and M.A.E. Andersen, VHF Series-Input Parallel-Output Interleaved Self-Oscillating Resonant SEPIC Converter, IEEE Energy Conversion Congress and Exhibition US 2013
- [28] P. Shamsi, B. Fahimi, Design and Development of Very High Frequency Resonant DC-DC Boost Converters, IEEE Transactions on Power Electronics, vol.27, no.8, pp.3725,3733, Aug. 2012
- [29] M.P. Madsen, A. Knott and M.A.E. Andersen, Low Power Very High Frequency Resonant Converter with High Step Down Ratio, IEEE Africon 2013
- [30] D.C. Hamill, Class DE inverters and rectifiers for DC-DC conversion, IEEE Power Electronics Specialists Conference, 1996, vol.1, no., pp.854,860 vol.1, 23-27 Jun 1996
- [31] Juan Rivas, Radio Frequency dc-dc Power Conversion, Doctoral Thesis, Massachusetts Institute of Technology, 2006
- [32] M.P. Madsen, J. A. Pedersen, A. Knott and M.A.E. Andersen, Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components IEEE Applied Power Electronics Conference and Exposition 2014, 16-20 March 2013
- [33] M. Kovacevic, A. Knott, M.A.E. Andersen, Interleaved Self-Oscillating Class E Derived Resonant DC/DC Converters, Proceedings of the International Conference on Electrical and Computer Systems 2012
- [34] T. Roy, L. Smith, J. Prymak, ESR and ESL of ceramic capacitor applied to decoupling applications, IEEE Topical Meeting on Electrical Performance of Electronic Packaging, 1998, vol., no., pp.213,216, 26-28 Oct 1998
- [35] M. Ingalls, G. Kent, Measurement of the Characteristics of High-Q Ceramic Capacitors, IEEE Transactions on Components, Hybrids, and Manufacturing Technology, vol.10, no.4, pp.487,495, Dec 1987
- [36] International Rectifier, *PD-94086 HEXFET Power MOSFET*, IRF5802 Datasheet, January 2001
- [37] I.D. de Vries, J.H. Van Nierop, J.R. Greene, Solid state class DE RF power source, IEEE International Symposium on Industrial Electronics, 1998, vol.2, no., pp.524,529, 7-10 Jul 1998
- [38] T. Suetsugu, M.K. Kazimierczuk, Integrated Class DE Synchronized DC-DC Converter for On-Chip Power Supplies, IEEE Power Electronics Specialists Conference, 2006, vol., no., pp.1,5, 18-22 June 2006

A.7 M. Madsen, M. Kovacevic, J. D. Mønster, J. A. Pedersen, A. Knott and M. A.E. Andersen: "Input-Output Rearrangement of Isolated Converters", IEEE Power and Energy Conference at Illinois, Champaign, IL, February 2015. Conference Proceedings p1-6.

Input-Output Rearrangement of Isolated Converters

Mickey P. Madsen*, Milovan Kovacevic*, Jakob D. Mønster*,

Jeppe A. Pedersen*, Arnold Knott*, Michael A. E. Andersen*

*Technical University of Denmark

Ørsteds Plads, building 349

2800 Kongens Lyngby

Denmark

Email: {mpma, mikov, jdomo, jarpe, akn, ma}@elektro.dtu.dk

Abstract— This paper presents a new way of rearranging the input and output of isolated converters. The new arrangement posses several advantages, as increased voltage range, higher power handling capabilities, reduced voltage stress and improved efficiency, for applications where galvanic isolation is not a requirement. The proposed technique is particularly valuable in power conversion at very high frequencies, and may be combined with other stress reduction methods. Finally, the new arrangements are experimentally verified both on off the shelf converters and on a VHF resonant SEPIC converter. All results are in good agreement with the theory and twice the power handling capabilities and 5-10% higher efficiencies are shown.

I. INTRODUCTION

Constant strive for cost and size reduction of power converters, as well as better performance, results in higher and higher switching frequencies. From the size and cost perspective, higher frequencies are desirable since energy storage requirements drop linearly with frequency, even though the components size and price do not typically scale at the same rate. An added benefit is cost and size reduction of filtering components. Recent advances in switching devices technology [1] and magnetic materials [2], [3] allowed converter operation in low-MHz range with high efficiencies [4]–[7]. If the switching frequency is pushed beyond 30 MHz, class E derived topologies - class E, resonant boost, resonant SEPIC, etc. - made it possible to use air-core magnetics, which eliminates core loss mechanism entirely [8], [9].

However, loss mechanisms in the switches start to dominate the total loss budget at higher frequencies. In low-MHz range, switching loss already gives incentive for use of resonant transition or quasi square-wave converter topologies (boost [10], inverted buck [6], flyback [11]). For operation above 30 MHz, switching losses impact converter efficiency so severely that only ZVS (zero voltage switching) topologies become practical. Even then, gating loss needs to be accounted for, regardless of the gating technique: hard-gating, singleresonant, or multi-resonant [12]. Needless to say, proper utilization of the switching components, especially in pricesensitive applications, is a must.

Class E derived converters exhibit voltage stress on the inverter switch of typically between 2.5 and 3.6 times input voltage, depending on the switch duty cycle [13]. As a consequence, devices with high voltage ratings are required.

These devices typically perform worse compared to their lower voltage counterparts, as voltage rating comes at the expense of increase in $R_{DS,on}$ with all other parameters held the same. It is therefore of interest to reduce switch voltage blocking requirement in order to improve performance.

Voltage stress in these converters can be reduced by employing multi-resonant networks, either transmission line [14] or discrete with limited number of harmonics [15]-[17]. By implementing these techniques, switch voltage stress in VHF converters can be reduced to 2-2.5 times input voltage for the same duty cycle, but this technique requires extra resonant elements. Another approach infers single switch quasi squarewave converter topologies. The downside of quasi square-wave converters is limited converter transformation ratio (inverted buck, boost), or excessive ringing on the primary switch due to leakage inductance (flyback). In [18] a self-oscillating drive method was used to obtain a VHF self-oscillating DE inverter, in which the switch voltage stress is limited to input voltage. Finally, converter cell stacking [19], [20] reduces the switch voltage stress by N compared to a single cell case, where N is the number of converter cells. However, control becomes more complicated since it is required to balance voltages across the cells, and certain start-up issues may occur.

This paper presents a voltage stress reduction method, which may be used in either step-up or step-down applications where galvanic isolation is not a requirement. In addition to voltage stress reduction, it results in higher converter efficiency with minimum adjustments to the original circuit. Section II describes the method in detail. Section III shows experimental results when the principle is applied to conventional offthe-shelf design and a VHF prototype, respectively. Finally, Section IV concludes the paper.

II. THE BASIC PRINCIPLE

Fig. 1 shows the conventional input-output configuration of an isolated dc/dc converter. If two isolated converters have their inputs connected in parallel and outputs connected in series, we obtain the structure in Fig. 2 (top). Similarly, if the inputs are connected in series and outputs in parallel, the structure in Fig. 3 (top) is obtained. Let us assume that one of the converters in each configuration is with 1:1 voltage transformation ratio. In such case, it is possible to connect its input to its output directly and remove 1:1 converter from the circuit, without effecting the rest of the system. In doing so, part of the delivered power flows directly from input to the output, and is not processed by the converter. As a consequence, higher converter efficiency is achieved. Moreover, compared to the single cell design, voltage and/or current stresses on the switches are reduced. The only difference is that the input is not galvanic isolated from the load. Fig. 2 and 3 (bottom) demonstrate the final connection rearrangement to obtain step-up and step-down configuration, respectively.

To quantify the benefits from the proposed configurations, we analyze the case where input voltage V_{in} and output voltage V_{out} are held constant across all configurations. Moreover, output current I_{out} and output power P_{out} are held constant as well. A distinction is made between the power processed by the converter P_C , and P_{out} .

Voltage transformation ratio of the conventional converter is

$$M = \frac{V_{out}}{V_{in}} \tag{1}$$

In step-up and step-down configurations, transformation ratio of the converter cell only becomes

$$M_{up} = \frac{V_{out} - V_{in}}{V_{in}} \tag{2}$$

$$M_{down} = \frac{V_{out}}{V_{in} - V_{out}} \tag{3}$$

In general, switch voltage stress is a function of both input and output voltages. In step-up configuration, output voltage stress contribution is reduced by a factor of $(1 - V_{in}/V_{out})$. In step-down configuration, input voltage stress contribution is reduced by $(1 - V_{out}/V_{in})$.

Output power of all three configurations is the same:

$$P_{out} = V_{out} I_{out} \tag{4}$$

The power processed by the converter P_C in the conventional structure is equal to P_{out} . In the step-up and step-down cases, this power is given as

$$P_{C,up} = P_{out} \left(1 - \frac{V_{in}}{V_{out}} \right) \tag{5}$$

$$P_{C,down} = P_{out} \left(1 - \frac{V_{out}}{V_{in}} \right) \tag{6}$$



Fig. 1. Conventional isolated dc/dc converter topology. Dashed line represents galvanic isolation.



Fig. 2. Step-up configuration: two converter cells with inputs in parallel and outputs in series (top) and a single cell equivalent (bottom).



Fig. 3. Step-down configuration: two converter cells with inputs in series and outputs in parallel (top) and a single cell equivalent (bottom).



Fig. 4. Power processed by the converter and efficiency improvement by rearranging.

 $P_{C,up}$ and $P_{C,down}$ are smaller than P_{out} , and the remaining power is delivered through the DC path. Efficiency of that power transfer is very close to 100%. Assuming that P_C , $P_{C,up}$, and $P_{C,down}$ are transfered with the same efficiency η , effective efficiencies of the step-up and step-down configurations can be expressed in terms of η , P_{out} , and $P_{C,up/down}$:

$$\eta_{up/down} = \eta \; \frac{P_{C,up/down}}{P_{out}} + \frac{P_{out} - P_{C,up/down}}{P_{out}} \quad (7)$$

Equations 5-7 are illustrated in Fig. 4. The plot clearly shows that for limited step-up or step-down ratios, the percentage of the output power processed by the converter is small and hence the efficiency improvement becomes significant. For a step-up converter with 330V input and 400V output only 17.5% of the output power would be processed by the converter and if the converter efficiency were 80%, the rearranged efficiency would be 96.5%. The converter would in that case have to be designed as an galvanic isolated converter with 330V input, 70V output and a peak output power equal to 17.5% of the required power. Hence a high power boost converter with high efficiency could be replaced by a low power flyback or SEPIC converter with moderate efficiency while achieving the same overall system performance.

Adaptations shown in Fig. 2 and 3 can be used for any type of isolated converter, regardless of the isolation type (inductive or capacitive). In Fig. 5, a flyback converter is used in a step-up configuration, along with simulated waveforms. Fig. 6 presents



Fig. 5. Flyback converter in step-up configuration (top) and simulated waveforms of a quasi square-wave 1:2 flyback converter in conventional (solid black) and step-up (dashed red) configuration. Diode voltage and switch current stresses are reduced by a factor of 2. Output capacitance is formed by series connected C_{out} and C_{in} .

a class DE converter (DE inverter and DE rectifier) in stepdown configuration.

The proposed technique may be used together with multicell stacked designs, as shown in Fig. 7. This also adds complexity to control circuitry since it needs to monitor and balance voltages across the cells.



Fig. 6. Class DE converter in step-down configuration with output referenced to the inverter input. Parasitic capacitances of the switching devices are included implicitly.



Fig. 7. Two stacked converters in step-down configuration.

III. EXPERIMENTAL RESULTS

A. Off the shelf converter

In order to experimentally verify equation 5-7 a test setup with two isolated 12-12V DC/DC converters (Murata NTE1212MC and CUI Inc PDS1-S12-S12-S) has been made. The setup is made with five jumpers allowing the setup to be changed between normal configuration (12-12V), step down (24-12V) and step up (12-24V).

The measured efficiency as function of output power is shown in Fig. 9. The increased efficiency and power handling capability in the new configurations are clearly seen. With a 1-2 or 2-1 step ratio the converter only handles 50% of the output power (see equation 5-6). The power handling capability is hence doubled while the efficiency is increased across the entire load range with half the loss at full load.

B. Resonant SEPIC converter

As mentioned in Section I, the input and output voltage and the subsequent voltage stress on the semiconductors is of high importance in resonant converters. The availability of





(b) PCB

Fig. 8. Test board for evaluation of the principle.



Fig. 9. Measured performance improvements of two rearranged converters.



Fig. 10. Resonant SEPIC converter with capacitive galvanic isolation.

suitable semiconductors is reduced significantly as the break down voltage requirement gets above 100 V. Furthermore the amount of resonating current needed in order to achieve zero voltage switching scales with the voltage squared, hence a reduction in input and output voltage will lead to significantly lower currents and thereby also reduced losses due to ESR in the components.

As the switching loss in resonant converters with ZVS is reduced to a minimum, these converters are operable at several tens of MHz [12], [15]–[19], [21]–[32]. At these frequencies, galvanic isolation can be achieved simply by adding a small capacitor in the return path between the inverter and rectifier.

The proposed rearrangement is hence extremely well suited for non-isolated converters where very high frequency resonant converters can be used to achieve high power density, low cost and weight and still keep the efficiency high. In order to show the benefit in these type of converters a resonant SEPIC converter (see Fig. 10) switching at 50 MHz has been implemented and tested on the test board in Fig. 8.

Figure 11 shows the performance improvements measured with this setup. From the plot it is clearly seen that both the efficiency and the voltage and power handling capability are increased. The measurements and theoretical values for the step down version fits well as the loading conditions as well as the input voltage are the same. For the step-up version the output voltage from the converter changes quite dramatically as the input voltage is increased and becomes close to zero towards the end.

In [18] the same principle is shown for a capacitively isolated VHF class DE converter. Here the rearangement increases the efficiency by 5-10% across a wide input voltage range. At the same time the input voltage and output power is increased by approx 50%.

IV. CONCLUSION

The paper presents a voltage and/or current stress reduction technique for isolated dc-dc converters in applications where isolation is not a requirement. The technique provides higher efficiency compared to a conventional single cell design, and it can be combined with other methods of voltage stress reduction, as long as the initial converter provides capacitive or inductive isolation. Obtained benefits are considered very appealing for certain non-isolated applications, such as LED



Fig. 11. Measured performance improvement of a VHF resonant SEPIC.

lighting. Theoretical analysis and experimental results are provided and are in good agreement. Twice the power handling capabilities and 5-10% higher efficiencies are shown for the tested converters.

REFERENCES

- D. Reuch and J. Strydom, "Evaluation of Gallium Nitride Transistors in High Frequency Resonant and Soft-Switching DC-DC Converters," in *Proc. Applied Power Electronics Conference and Exposition (APEC)*, Fort Worth, TX, USA, Mar. 2014, pp. 464–470.
- [2] L. Daniel, C. R. Sullivan, and S. R. Sanders, "Design of Microfabricated Inductors," *IEEE Trans. Power Electron.*, vol. 14, pp. 709–723, July 1999.
- [3] C. R. Sullivan, "Aluminum Windings and Other Strategies for High-Frequency Magnetics Design in an Era of High Copper and Energy Costs," in *Proc. Applied Power Electronics Conference and Exposition* (APEC), Anaheim, CA, USA, Feb. 2007, pp. 78–84.
- [4] Y. Han and D. J. Perreault, "Inductor Design Methods With Low-Permeability RF Core Materials," *IEEE Trans. Ind. Appl.*, vol. 48, pp. 1616 – 1627, Sept. 2012.
- [5] A. D. Sagneri, D. I. Anderson, and D. J. Perreault, "Transformer synthesis for VHF converters," in *Proc. International Power Electronics Conference (IPEC)*, Sapporo, Japan, June 2010, pp. 2347–2353.
- [6] S. Lim, J. Ranson, D. M. Otten, and D. J. Perreault, "Two-Stage Power Conversion Architecture for an LED Driver Circuit," in *Proc. The Applied Power Electronics Conference and Exposition (APEC)*, Long Beach, California, USA, Mar. 2013, pp. 854–861.
- [7] S. Lim, D. M. Otten, and D. J. Perreault, "Power Conversion Architecture for Grid Interface at High Switching Frequency," in *Proc. The Applied Power Electronics Conference and Exposition (APEC)*, Fort Worth, Texas, USA, Mar. 2014, pp. 1838–1845.
- [8] D. J. Perreault, J. Hu, J. M. Rivas, Y. Han, O. Leitermann, R. C. N. Pilawa-Podgurski, A. Sagneri, and C. R. Sullivan, "Opportunities and Challenges in Very High Frequency Power Conversion," in *Proc. The*

Applied Power Electronics Conference and Exposition (APEC), Washington, DC, USA, Feb. 2009, pp. 1838–1845.

- [9] A. Knott, T. M. Andersen, P. Kamby, M. P. Madsen, M. Kovacevic, and M. A. E. Andersen, "On the Ongoing Evolution of Very High Frequency Power Supplies," in *Proc. The Applied Power Electronics Conference and Exposition (APEC)*, Long Beach, California, USA, Mar. 2013, pp. 2514–2519.
- [10] P. Shamsi and B. Fahimi, "Design and Development of VHF Resonant DC-DC Boost Converters," *IEEE Trans. Power Electron.*, vol. 27, pp. 3725–3733, June 2012.
- [11] Z. Zhang, K. D. T. Ngo, and J. L. Nilles, "A 30-W flyback converter operating at 5 MHz," in *Proc. Applied Power Electronics Conference and Exposition (APEC)*, Fort Worth, TX, USA, Mar. 2014, pp. 1415–1421.
- [12] M. P. Madsen, J. A. Pedersen, A. Knott, and M. A. E. Andersen, "Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components," in *Proc. IEEE Advanced Power Electronics Conference and Exposition (APEC)*, Fort Worth, Texas, USA, Mar. 2014, pp. 2029 – 2035.
- [13] M. K. Kazimierczuk and K. Puczko, "Exact Analysis of class E tuned power amplifier at any Q and duty cycle," vol. 34, pp. 149–159, Feb. 1987.
- [14] J. W. Phinney and D. J. Perreault, "Radio-Frequency Inverters With Transmission-Line Input Networks," *IEEE Trans. Power Electron.*, vol. 22, pp. 1154–1161, July 2007.
- [15] J. M. Rivas, Y. Han, O. Leitermann, A. D. Sagneri, and D. J. Perreault, "A High-Frequency Resonant Inverter Topology With Low-Voltage Stress," *IEEE Trans. Power Electron.*, vol. 23, pp. 1759–1771, July 2008.
- [16] R. C. N. Pilawa-Podgurski, A. D. Sagneri, J. M. Rivas, D. I. Anderson, and D. J. Perreault, "Very-High-Frequency Resonant Boost Converters," *IEEE Trans. Power Electron.*, vol. 24, pp. 1654–1665, June 2009.
- [17] J. M. Rivas, O. Leitermann, Y. Han, and D. J. Perreault, "A Very High Frequency DC-DC Converter Based on a Class Φ₂ Resonant Inverter," *IEEE Trans. Power Electron.*, vol. 26, pp. 2980–2992, Oct. 2011.
- [18] M. P. Madsen, A. Knott, and M. A. E. Andersen, "Very High Frequency Half Bridge DC/DC Converter," in *Proc. IEEE Advanced Power Electronics Conference and Exposition (APEC)*, Fort Worth, Texas, USA, Mar. 2014, pp. 1409–1414.
- [19] M. Kovacevic, A. Knott, and M. A. E. Andersen, "A VHF Series-Input Parallel-Output Interleaved Self-Oscillating Resonant SEPIC Converter," in *Proc. IEEE Energy Conversion Congress and Exposition (ECCE US)*, Denver, Colorado, USA, Sept. 2013, pp. 2052–2056.
- [20] N. Golbon, F. Ghodousipour, and G. Moschopoulos, "A DC-DC converter with stacked flyback converters," in *Proc. Energy Conversion Congress and Exposition (ECCE)*, Denver, CO, USA, Sept. 2013, pp. 4894–4899.
- [21] J. M. Rivas, R. S. Wahby, J. S. Shafran, and D. J. Perreault, "New Architectures for Radio-Frequency DC-DC Power Conversion," *IEEE Trans. Power Electron.*, vol. 21, pp. 380–393, Mar. 2006.
- [22] Y. Han, O. Leitermann, D. A. Jackson, J. M. Rivas, and D. J. Perreault, "Resistance Compression Networks for Radio-Frequency Power Conversion," *IEEE Trans. Power Electron.*, vol. 22, pp. 41–53, Jan. 2007.
- [23] J. Hu, A. D. Sagneri, J. M. Rivas, Y. Han, S. M. Davis, and D. J. Perreault, "High Frequency Resonant SEPIC Converter with Wide Input and Output Voltage Ranges," *IEEE Trans. Power Electron.*, vol. 27, pp. 189–200, Jan. 2012.
- [24] A. D. Sagneri, D. I. Anderson, and D. J. Perreault, "High Frequency Resonant SEPIC Converter with Wide Input and Output Voltage Ranges," *IEEE Trans. Power Electron.*, vol. 28, pp. 3614–3626, July 2013.
- [25] I. J. R. Warren, K. A. Rosowski, and D. J. Perreault, "Transistor Selection and Design of a VHF DC-DC Power Converter," *IEEE Trans. Power Electron.*, vol. 23, pp. 27–37, Jan. 2008.
- [26] W. Inam, K. K. Afridi, and D. J. Perreault, "High Efficiency Resonant dc/dc Converter Utilizing a Resistance Compression Network," in *Proc. The Applied Power Electronics Conference and Exposition (APEC)*, Long Beach, California, USA, Mar. 2013, pp. 1399–1405.
- [27] M. P. Madsen, A. Knott, M. A. E. Andersen, and A. P. Mynster, "Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies," in *Proc. International Energy Conver*sion Congress and Exhibition (ECCE Asia), Melbourne, Australia, June 2013, pp. 1071–1078.
- [28] M. P. Madsen, A. Knott, and M. A. E. Andersen, "Low Power Very High Frequency Converter with High Step Down Ratio," in *Proc. AFRICON2013*, Mauritius, Sept. 2013.

- [29] M. P. Madsen, A. Knott, and M. A. E. Andersen, "Low Power Very High Frequency Switch-Mode Power Supply with 50 V Input and 5 V Output," *IEEE Trans. Power Electron.*, vol. PP, pp. 1–23, Dec. 2014.
- [30] M. Kovacevic, A. Knott, and M. A. E. Andersen, "Interleaved Self-Oscillating Class E Derived Resonant DC/DC Converters," in *Proc. International Conference on Electrical and Computer Systems (ICECS)*, Ottawa, Canada, Aug. 2012, pp. 1–8.
- [31] M. Kovacevic, A. Knott, and M. A. E. Andersen, "A Very High Frequency Interleaved Self-Oscillating Resonant SEPIC Converter," in *Proc. European Conference on Power Electronics and Applications* (*EPE*), Lille, France, Sept. 2013, pp. 1–9.
- [32] M. Kovacevic, A. Knott, and M. A. E. Andersen, "A VHF Self-Oscillating Resonant SEPIC Converter with Phase-Shift Burst-Mode Control," in *Proc. IEEE Advanced Power Electronics Conference and Exposition (APEC)*, Fort Worth, Texas, USA, Mar. 2014, pp. 2052–2056.

A.8 M. Kovacevik and M. Madsen: "Step-up dc-dc power converter", WO2015091590 A2, June 25th 2015 (12) INTERNATIONAL APPLICATION PUBLISHED UNDER THE PATENT COOPERATION TREATY (PCT)

(19) World Intellectual Property

Organization International Bureau



(43) International Publication Date 25 June 2015 (25.06.2015)

- (51) International Patent Classification: Not classified
- (21) International Application Number:
- PCT/EP2014/078116
- (22) International Filing Date: 17 December 2014 (17.12.2014)
- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 13198052.6 18 December 2013 (18.12.2013) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: KOVACEVIC, Milovan; Aldershvilevej 8, 1., DK-2800 Kgs. Lyngby (DK). MADSEN, Mickey P; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK).
- (74) Agent: OLSEN, Henrik Bagger; Guardian IP Consulting I/S, Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).
- (81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM,

(10) International Publication Number WO 2015/091590 A2

AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN. TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.

(84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, ST, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

without international search report and to be republished upon receipt of that report (Rule 48.2(g))

(54) Title: STEP-UP DC-DC POWER CONVERTER

201



FIG 2

(57) Abstract: The present invention relates to a step-up DC-DC power converter which comprises a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier. The primary side circuit comprises a positive and a negative input terminal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals and the secondary side circuit comprises an output capacitor chargeable to a converter output voltage between a first positive electrode and a second negative electrode. A switched energy storage network is configured for alternatingly being charged from the input voltage and discharged to the output capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce the converter output voltage. The step-up DC-DC power converter comprises an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive input terminal of the primary side circuit or, in a second case, connecting the second positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection is established, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or, in the second case, between the second negative electrode of the output capacitor and the positive input terminal.

STEP-UP DC-DC POWER CONVERTER

The present invention relates to a step-up DC-DC power converter which comprises a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier. The primary side circuit comprises a positive and a negative input termi-

- 5 nal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals and the secondary side circuit comprises an output capacitor chargeable to a converter output voltage between a first positive electrode and a second negative electrode. A switched energy storage network is configured for alternatingly being charged from the input voltage and discharged to the output
- 10 capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce the converter output voltage. The step-up DC-DC power converter comprises an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive input terminal of the primary side circuit or, in a second case, connecting
- 15 the second positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection is established, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or, in the second case, between the second
- 20 negative electrode of the output capacitor and the positive input terminal.

BACKGROUND OF THE INVENTION

Power density and component costs are key performance metrics of both isolated and non-isolated DC-DC power converters to provide the smallest possible physical size and/or lowest costs for a given output power requirement or specification. Res-

- 25 size and/or lowest costs for a given output power requirement or specification. Resonant power converters are particularly useful for high switching frequencies such as frequencies above 1 MHz where switching losses of standard SMPS topologies (Buck, Boost etc.) tend to be unacceptable for conversion efficiency reasons. High switching frequencies are generally desirable because of the resulting decrease of
- 30 the electrical and physical size of circuit components of the power converter like inductors and capacitors. The smaller components allow increase of the power density of the DC-DC power converter. In a resonant power converter an input "chopper" semiconductor switch (often MOSFET or IGBT) of the standard SMPS is replaced with a "resonant" semiconductor switch. The resonant semiconductor switch

relies on resonances of circuit capacitances and inductances to shape the waveform of either the current or the voltage across the semiconductor switch such that, when state switching takes place, there is no current through or no voltage across the semiconductor switch. Hence power dissipation is largely eliminated in at least some

5 of the intrinsic capacitances or inductances of the input semiconductor switch such that a dramatic increase of the switching frequency becomes feasible for example to values above 10 MHz. This concept is known in the art under designations like zero voltage and/or zero current switching (ZVS and/or ZCS) operation. Commonly used switched mode power converters operating under ZVS and/or ZCS are often de-

10 scribed as class E, class F or class DE inverters or power converters.

In view of the above, it remains a challenge to reduce the size and lower the component costs of both isolated and non-isolated DC-DC power converters. Hence, novel step-up DC-DC power converter topologies which reduce the required maxi-

15 mum voltage rating of active and passive components of the DC-DC converter are highly desirable. Likewise, novel step-up DC-DC power converter topologies which reduce the physical size or cost of active and passive components for example inductors, capacitors, transistors and diodes are highly desirable.

20 SUMMARY OF THE INVENTION

A first aspect of the invention relates to a step-up DC-DC power converter which comprises a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier. The primary side circuit comprises a positive and a negative input terminal for receipt of an input voltage and an input capacitor coupled between

- 25 the positive and negative input terminals and the secondary side circuit comprises an output capacitor chargeable to a converter output voltage between a first positive electrode and a second negative electrode. A switched energy storage network is configured for alternatingly being charged from the input voltage and discharged to the output capacitor through the galvanic isolation barrier in accordance with a
- 30 switch control signal to produce the converter output voltage. The step-up DC-DC power converter comprises an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive input terminal of the primary side circuit or, in a second case, connecting the second positive electrode of the output capacitor to the nega-

tive input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection is established, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or, in the second case, be-

5 tween the second negative electrode of the output capacitor and the positive input terminal.

The present invention is described in detail in the following with reference to specific implementations in isolated resonant DC-DC power converters of Class E, DE and SEPIC topologies and a non-resonant flyback DC-DC converter topology. The skilled person will understand that the invention is equally applicable to other types of isolated resonant and non-resonant DC-DC power converter such as class π₂ inverters and rectifiers and resonant boost, buck, LCC converters etc.

- 15 The skilled person will understand that the electrical short-circuit connection across the galvanic isolation barrier eliminates the galvanic isolation between the primary and secondary side circuits of the step-up DC-DC converter by interconnecting the second electrode of the output capacitor and the negative input terminal. However, the electrical short-circuit connection provides numerous new benefits to the DC-DC
- 20 converter as a whole and the lack of galvanic isolation is acceptable in numerous applications where the converter circuit is isolated from users such as retrofit LED bulbs and tubes. The series connection of the output and input capacitors established by the electrical short-circuit connection has the effect that the secondary side circuit only needs to supply the output voltage minus the input voltage of the present
- 25 step-up DC-DC converter, instead of the entire converter output voltage as in ordinary isolated DC-DC power converters, to a converter load. The converter load is coupled between either the first positive electrode of the output capacitor and the negative input terminal or between the second negative electrode of the output capacitor and the positive input terminal depending on the connection points of the
- 30 electrical short-circuit connection as explained in further detail below with reference to FIGS. 1A), 1B) and 1C). Consequently, since, the switched energy storage network only supplies a fraction of the converter output voltage it also supplies only a corresponding fraction of the total power to the converter load. The reduced voltage in the secondary side circuit of the step-up DC-DC power converter reduces the

required maximum voltage rating of active and passive components therein such as semiconductor switch or switches, inductor(s), capacitors, diode(s) etc. The reduced maximum voltage rating of the active and passive components leads to physically smaller and/or less costly active and passive components. In addition, the life span

5 of the latter components may increase by the smaller voltage stress. Likewise, in the primary side circuit the smaller amount of power to be transferred through the stepup DC-DC converter for a given amount of output power delivered to the converter load leads to reduced power requirements for active semiconductors switches allowing less costly and physically smaller semiconductors to be applied.

10

15

The beneficial reduction of the amount of power to be transferred through the switched energy storage network is achieved because the residual fraction of the output power is transferred directly from the input voltage source and input capacitor to the output capacitor due to their series connection as explained in further detail below with reference to FIGS. 1A), 1B) and 1C).

The skilled person will appreciate that the switched energy storage network can comprise numerous types of ordinary switch topologies such as a single switch topology, a half-bridge switch topology or full-bridge switch topologies. The switched

- 20 energy storage network preferably comprises at least one semiconductor switch such as a MOSFET or IGBT such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) transistor. A control terminal, e.g. a gate or base, of the at least one semiconductor switch may be coupled to, and driven by, the switch control signal to alternatingly force the least one semiconductor switch between on-states and off-states. In
- 25 the on-state an inductor of the switched energy storage network may be charged with energy from the input voltage source and in the following off-state release stored energy to the output capacitor to charge the latter. The secondary side circuit of the step-up DC-DC converter may comprise a rectifying element such as a diode or transistor inserted in front of the converter load.

30

The galvanic isolation barrier may comprise a transformer which comprises a pair of magnetically coupled inductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the secondary side circuit. The first and second inductors could be discrete windings both

wound around a common magnetic permeable structure to form an isolation transformer. In an alternative embodiment, the first and second inductors are integrated in a printed circuit board without intervening magnetic material. The printed circuit board could have the entire step-up DC-DC power converter mounted thereon.

5

In yet another embodiment, the galvanic isolation barrier comprises a first capacitor coupled in series with the positive input terminal of the primary side circuit and the first positive electrode of the output capacitor and a second capacitor coupled in series with the negative input terminal of the primary side circuit and the second

- 10 negative electrode of the output capacitor. Each of the first and second capacitors may possess particularly small physically dimensions in step-up resonant DC-DC power converters with a switching frequency, or frequency of the switch control signal, at or above 10 MHz. In the latter embodiments each of the first and second capacitors may comprise a ceramic capacitor and may possess a capacitance smaller
- 15 than 10 nF such as smaller than 1 nF such as smaller than 100pF. Isolation capacitors with these capacitances may be SMD mounted ceramic capacitors with a very small footprint as discussed below.

The skilled person will appreciate that a practical electrical short circuit connection

- 20 will possess a finite DC resistance and an upper limit of this finite DC resistance will vary depending on input/output voltage and/or current requirements of the step-up DC-DC power converter. The electrical short-circuit connection may possess a DC resistance of less than 1 k Ω , even more preferably less than 100 Ω , such as less than 10 Ω . In other embodiments, the electrical short circuit connection may have a
- 25 unidirectional resistance such that the DC resistance only falls below the abovementioned upper limits in one direction and exhibits a much larger DC resistance in the opposite direction, i.e. a diode characteristic
- One embodiment of the step-up DC-DC power converter is based on a Class E converter and the switched energy storage network comprises first and second series connected inductors which are connected in series with the positive input terminal. A semiconductor switch is arranged with a first switch node connected between a midpoint node between the first and second series connected inductors and a second switch node connected to the negative input terminal of the primary side circuit. A

6

control terminal of the semiconductor switch is connected to the switch control terminal; and a third inductor has a first end connected to a second end of the second inductor through the first capacitor of a galvanic isolation barrier and a second node connected to the converter output voltage at the positive electrode of the output ca-

5 pacitor. A rectifier is connected between the first end of the third inductor and the negative electrode of the output capacitor.

Another embodiment of the step-up DC-DC power converter is based on a flyback converter topology wherein the first and second inductors of the isolation transform-

- 10 er are integrated in the switched energy storage network. The first inductor is arranged with a first inductor end connected to the positive input voltage terminal and a second inductor end connected to a first node of a semiconductor switch such as a drain terminal of a MOSFET switch. A second node of the semiconductor switch is connected to the negative input terminal of the primary side circuit. The second in-
- 15 ductor of the isolation comprising a first inductor end connected to the first positive electrode of output capacitor and a second inductor end connected to the second negative electrode, respectively, of the output capacitor through a rectifier.

The step-up DC-DC power converter may comprise a resonant DC-DC power converter to facilitate zero voltage and/or zero current switching of the semiconductor switch or switches of the switched energy storage network as discussed in additional detail below. The resonant DC-DC power converter is particularly advantageous at high switching frequencies of the switch control signal such as above 10 MHz or above 20 MHz such as at or above 30 MHz as discussed below.

25

The step-up DC-DC power converter may comprise a mode selecting semiconductor switch which is configured to switch the step-up DC-DC power converter between two distinct modes of operation. According to this embodiment, the step-up DC-DC power converter comprises a rectifying element, such as a diode, coupled

30 between the positive input terminal and second negative electrode of the output capacitor. The mode selecting semiconductor switch which is configured to selectively break and close the electrical short-circuit connection such that: in a first mode of the step-up DC-DC power converter, establishing the series connection of the output capacitor and the input capacitor; and

in a second mode of the step-up DC-DC power converter, break the series coupling of the output capacitor and the input capacitor.

The mode selecting semiconductor switch may be switched between a conducting

- 5 state and non-conducting state by a suitable control voltage applied on a control terminal of the mode selecting semiconductor switch such as a gate terminal of a MOSFET or FET semiconductor switch or base terminal of a BJT or IGBT semiconductor switch. A mode controlling circuit connected to, or integrated with, the step-up DC-DC power converter may be configured to supply this control voltage to the
- 10 mode selecting semiconductor switch. The first mode of the step-up DC-DC power converter is selected in the conducting or ON state of the mode selecting semiconductor switch and the second mode of the step-up DC-DC power converter is selected in the non-conducting or OFF state of the mode selecting semiconductor switch. The rectifying element may comprise an ordinary diode or an active diode for
- 15 example a semiconductor switch configured for diode operation by a suitable control signal applied to a control terminal of the semiconductor switch,

The mode switching feature of this embodiment of the step-up DC-DC power converter provides several advantages such as increasing the dynamic voltage operat-

20 ing range of the converter as discussed in additional detail below with reference to the appended drawings.

In a range of particularly advantageous embodiments of the present step-up DC-DC power converters the switch control signal of the switched energy storage network is placed in the so-called VHF range with a switching frequency at or above 10 MHz, or more preferably at or above 20 MHz such as at or above 30 MHz. These step-up DC-DC power converters preferably comprises resonant topologies as mentioned above to facilitate zero voltage and/or zero current switching of the semiconductor switch or switches of the switched energy storage network. The VHF operation of

30 these step-up DC-DC power converters provides considerable decrease of the electrical and physical size of active and passive components such as the previously discussed inductors and capacitors. Hence the previously mentioned transformer or capacitors of the galvanic isolation barrier of the present step-up DC-DC power converter can be physically small and inexpensive. The capacitor based galvanic isola-

tion becomes particularly advantageous in the VHF frequency range as the capacitance of each of the isolation capacitors can be small, such as 10 nF or even smaller in some cases for example smaller than 1 nF such as about 100pF. Isolation capacitors with these capacitances may comprise SMD mounted ceramic capacitors

- 5 with a very small footprint e.g. a footprint less than 1 cm² for example a footprint down to about 4 mm². In VHF frequency range operating embodiments of the step-up DC-DC power converter, such resonant step-up DC-DC power converters, each of the input capacitor and the output capacitor may have a capacitance smaller than 100 nF. The skilled person will understand that the input and output capacitors in
- 10 certain embodiments of the invention may be formed exclusively by a parasitic capacitance associated with the primary side circuit and the secondary side circuit, respectively.

The skilled person will furthermore understand that each of the present step-up DC-

- DC power converters may be constructed by conversion of an isolated DC-DC power converter with a corresponding topology as described in additional detail below with reference to FIGS. 2A) 2B) FIGS. 3A) 3B), FIGS. 4A) 4B) and FIGS. 5A) 5B). Hence, a second aspect of the invention relates to a method of converting an isolated DC-DC power converter to a step-up DC-DC power converter with higher
- 20 power conversion efficiency, said method comprising steps of:
 providing a primary side circuit and a secondary side circuit of the isolated DC-DC
 power converter,
 coupling an input capacitor between a positive and a negative input terminal of the
 primary side circuit,
- 25 coupling an output capacitor between a positive and a negative terminal of the secondary side circuit, providing electrical coupling of the primary side circuit and the secondary side circuit through a galvanic isolation barrier, providing a switched energy storage network configured for alternatingly being
- 30 charged from an input voltage of the converter and discharged to the output capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce a converter output voltage,

connecting, in a first case, an electrical short-circuit across the galvanic isolation barrier from the negative output terminal of the secondary side circuit to the positive input terminal of the primary side circuit or connecting, in a second case, the positive output terminal of the secondary side circuit to the negative input terminal of the primary side circuit thereby establishing in both the first case and the second case a series coupling of the output capacitor and the input capacitor,

- 5 coupling, in a first case, a power converter load between the positive terminal of the secondary side circuit and the negative input terminal or coupling, in the second case, the power converter load between the negative terminal of the secondary side circuit and the positive input terminal of the primary side circuit.
- 10 A preferred embodiment of the above conversion methodology generates the previously discussed step-up DC-DC power converter with the mode switching feature. This is achieved by adding further method steps of: connecting a rectifying element, such as a diode, between the positive input terminal and second negative electrode of the output capacitor; and
- 15 inserting a mode selecting semiconductor switch into the electrical short-circuit connection for selectively breaking and closing/making the short circuit connection such that:

establishing the series connection of the output capacitor and the input capacitor in a first mode of the step-up DC-DC power converter; and

20 breaking or disconnecting the series coupling of the output capacitor and the input capacitor in a second mode of the step-up DC-DC power converter.

The higher power conversion efficiency of the present step-up DC-DC power converter embodiments is achieved because a considerable amount of the power delivered to the converter load may be transferred directly from the input voltage source and input capacitor of the input side circuit to the output capacitor of the output side circuit due to the series connection of the input and output capacitors provided by the electrical short circuit connection as explained above. Hence, a smaller amount

30 of power has to be transferred through the switched energy storage network and isolation barrier leading to lower power losses in the active and/or passive components thereof. The isolated DC-DC power converter may comprise a resonant DC-DC power converter, preferably a resonant DC-DC power converter where the frequency of the switch control signal of the switched energy storage network has a frequency at or above 10 MHz such as at or above 20 MHz, more preferably at or above 30 MHz.

BRIEF DESCRIPTION OF THE DRAWINGS

- 5 Preferred embodiments of the invention will be described in more detail in connection with the appended drawings, in which:
 FIGS. 1A) and 1B) are simplified electrical circuit diagrams illustrating a step-up DC-DC power converter in accordance with a first embodiment of the present invention,
 FIG. 1C) is a simplified electrical circuit diagram of a step-up DC-DC power convert-
- er in accordance a second embodiment of the invention,
 FIG. 1D) is a simplified electrical circuit diagram of a step-up DC-DC power converter in accordance a third embodiment of the invention,
 FIG. 1E) is simplified electrical circuit diagram of a step-up DC-DC power converter in accordance a fourth embodiment of the invention,
- FIG. 1F) is simplified electrical circuit diagram of a step-up DC-DC power converter in accordance a fifth embodiment of the invention,
 FIG. 2A) is an electrical circuit diagram of a prior art isolated class E resonant DC-DC converter comprising a series resonant circuit,
 FIG. 2B) is an electrical circuit diagram of a class E resonant step-up DC-DC power
- 20 converter comprising a series resonant circuit in accordance with a sixth embodiment of the invention,
 - FIG. 3A) is an electrical circuit diagram of a prior art flyback DC-DC converter,

FIG. 3B) is an electrical circuit diagram of a flyback step-up DC-DC power converter in accordance with a 7th embodiment of the invention,

- FIG. 4A) is an electrical circuit diagram of a prior art isolated SEPIC converter,
 FIG. 4B) is an electrical circuit diagram of a step-up SEPIC DC-DC converter in accordance with an 8th embodiment of the invention,
 FIG. 5A) is an electrical circuit diagram of a prior art isolated class DE resonant DC-DC converter comprising a series resonant circuit; and
- **30** FIG. 5B) is an electrical circuit diagram of a class DE resonant step-up DC-DC power converter in accordance with a 9th embodiment of the invention.

11

DETAILED DESCRIPTION OF PREFERRED EMBODIMENTS

FIGS. 1A) and 1B) are simplified electrical circuit diagrams 100b illustrating basic operational principles of step-up DC-DC power converters in accordance with a first embodiment of the present invention. Two different variants of a generic converter

- 5 circuit topology are illustrated on FIGS. 1A, 1B) and FIG. 1C), respectively. FIG. 1A) shows a step-up DC-DC power converter 100b comprising a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 107b. The primary side circuit comprises a positive input terminal 102b and a negative input terminal 101b for receipt of a DC or AC input voltage V_{in} from a voltage or power
- 10 source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 102b and a negative input terminal 101b to form an energy reservoir for the voltage source. The primary side circuit additionally comprises an input side 106b of a switched energy storage network arranged in front of the isolation barrier 107. The secondary side circuit comprises an output capacitor C_{out} having a
- 15 first electrode electrically connected to the converter output voltage V_{out} at output terminal 104b. A second electrode of the output capacitor C_{out}, situated at a lower voltage potential than the first electrode, is connected to the positive input terminal 102b on the input side circuit via an electrical short-circuit connection or wire 109b extending across the isolation barrier 107b. The electrical short-circuit connection or
- 20 wire 109b effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade between the output voltage V_{out} at output terminal 104b and the negative input terminal 101b. An electrical load R_{load} of the step-up DC-DC converter 100b is coupled between the output terminal 104b and the negative input terminal 101b such that in effect the output and input capacitors C_{out}, C_{in} are coupled series to
- 25 supply power or current to the electrical load. The primary side circuit comprises the previously discussed input side 106b of the switched energy storage network of the step-up DC-DC converter 100b and the secondary side circuit comprises an output side 108b of the switched energy storage network. The skilled person will appreciate that the switched energy storage network may include numerous circuit topologies
- 30 depending on the particular type of DC-DC converter in question. The switched energy storage network preferably comprises at least one inductor for energy storage and release, but may alternatively exclusively comprise capacitors for energy storage. Generally, the switched energy storage network is configured for alternatingly being charged from the input voltage V_{in} and discharged to the output capacitor C_{out}

PCT/EP2014/078116

12

through the isolation barrier 107b in accordance with a switch control signal to produce the converter output voltage $V_{out.}$. The primary side circuit preferably comprises at least one semiconductor switch, for example a MOSFET, which is switched between on-states and off-states by the switch control signal such that the input volt-

- 5 age is modulated in accordance with a switch control signal. The frequency of the switch control signal of the switched energy storage network may be at or above 30 MHz to form a so-called VHF type of DC-DC power converter. The switch control signal may comprise a PWM modulated control signal. The primary side circuit may comprise an inductor that is charged with energy during an on-state of the least one
- 10 semiconductor switch from the input capacitor C_{in} and/or the DC or AC input voltage V_{in}. The inductor of the primary side circuit may subsequently be discharged through the output side 108b of the switched energy storage network and the output capacitor C_{out} in an off-state of the least one semiconductor switch. The secondary side circuit may comprise a diode based rectifier or a synchronous rectifier in front of the

15 output capacitor to produce the converter output voltage V_{out}.as a DC output voltage.

While the electrical short-circuit connection or wire 109b eliminates the galvanic isolation between the input and output side circuits of the step-up DC-DC converter 100b by interconnecting the second electrode of the output capacitor C_{out} and the

- 20 negative input terminal 101b, it provides numerous new benefits to the DC-DC converter as a whole as illustrated with reference to FIG 1B). The series connection of the output and input capacitors C_{out}, C_{in} means that the secondary side circuit only needs to supply the converter output voltage minus the input voltage (i.e. V_{out} minus V_{in}) to the electrical load R_{load} instead of the entire output voltage which is the situa-
- 25 tion in prior art isolated DC-DC converter topologies. Since, the switched energy storage network, including the input and output sides 106b, 108b, only supplies a fraction of the converter output voltage V_{out} it also supplies a corresponding fraction of the total power only to the electrical load R_{load}. The reduced voltage across the output section 108b reduces the required maximum voltage rating of active and
- 30 passive components therein leading to physically smaller and/or less costly active and passive components for example inductors, capacitors (including C_{out}), transistors and diodes etc. In addition, the life span of the latter components may increase by the smaller voltage stress. In the input section 106b, the smaller amount of power to be transferred through the DC-DC converter 100b for supplying a given converter

PCT/EP2014/078116

13

output power to the electrical load, leads to reduced power requirements for active semiconductors switches allowing less costly and physically smaller semiconductors to be applied.

- 5 These beneficial reductions of the amount of power to be transferred through the switched energy storage network 106b, 107b, 108b are achieved because the residual fraction of the output power supplied to the electrical load is transferred directly from the input voltage source V_{in} and input capacitor C_{in} to the output capacitor C_{out}. This power transfer mechanism is illustrated by the first output current path
- 10 I_{convert} which shows how secondary side current charges the output capacitor C_{out} when the current is drawn by the load and thereby delivers power that has passed through the switched energy storage network in a conventional manner. However, the present DC-DC converter also comprises a second output current path I_{direct} which illustrates how the output capacitor C_{out} is charged directly from the input volt-
- 15 age source V_{in} and input capacitor C_{in} when the current is drawn by the load without passing through input and output sides 106b, 108b and isolation barrier 107b of the switched energy storage network. The skilled person will appreciate that a practical electrical short circuit connection 109b will possess a certain DC resistance and an upper limit for this DC resistance will vary depending on input/output voltage and/or
- 20 current requirements of the converter 100b. The electrical short-circuit connection may possess a DC resistance of less than 1 k Ω , even more preferably less than 100 Ω , such as less than 10 Ω . In other embodiments, the electrical short circuit connection 109b may have a unidirectional resistance such that the DC resistance only falls below the above-mentioned upper limits in one direction and exhibits a much larger
- 25 DC resistance in the opposite direction, i.e. a diode characteristic.

FIG. 1C) is a simplified electrical circuit diagram 100c illustrating basic operational principles of step-up DC-DC power converters in accordance with a second embod-iment of the present invention. The step-up DC-DC power converter 100c may be

30 viewed as an alternative variant of the step-up DC-DC converter topology 100b in accordance with the first embodiment of the invention where the electrical short-circuit connection or wire 109c extending across the isolation barrier 107c is connecting the second positive electrode of the output capacitor C_{out} to the negative input terminal 102c of the primary side circuit. Thereby, a series coupling of the out-

PCT/EP2014/078116

14

put capacitor C_{out} and the input capacitor C_{in} from the converter output voltage V_{out} at the positive input terminal 104c to the negative electrode 101c of the output capacitor C_{out} is established. The negative electrode 101c of the output capacitor C_{out} is at a lower electric potential than the negative input terminal 102c. In this manner,

5 the input voltage V_{in} is stacked on top of the voltage across the first and second electrodes of the output capacitor C_{out}. Otherwise, circuit functions, electrical component characteristics and component values of the second embodiment of the step-up DC-DC power converter 100c may be identical to those discussed above in connection with the first embodiment of the step-up DC-DC power converter 100b.

10

FIG. 1D) shows a step-up DC-DC power converter 100d in accordance with third embodiment of the invention. The converter core 105 of the step-up DC-DC power converter 100d may be identical to the core 105 of the step-up DC-DC power converter 100b discussed above in connection with FIGS. 1A) and 1B). Hence, corre-

- 15 sponding features of these different step-up DC-DC power converter embodiments 100b, 100d have been provided with corresponding reference symbols to assist comparison. The third embodiment of the step-up DC-DC power converter 100d comprises a mode selecting controllable semiconductor switch SW1 inserted in a short-circuit connection or wire 109d. This short-circuit connection 109d effectively
- 20 places the output capacitor C_{out} and input capacitor C_{in} is series between the output voltage V_{out} at output terminal 104d and the negative input terminal 101d as discussed above.
- The mode selecting controllable semiconductor switch SW1 is configured to switch
 the step-up DC-DC power converter 110d between two distinct modes of operation as discussed below. The controllable semiconductor switch SW1 may comprise one or more BJT(s), FET(s) MOSFET(s) or IGBT(s) such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) transistor. SW1 may be switched between conducting/ON state and non-conducting/OFF state by a suitable control voltage applied on a gate
- 30 or base terminal of the switch SW1. A mode controlling circuit of, or associated with, the step-up DC-DC power converter 100d may supply this control voltage to SW1.

SW1 is configured to break/disconnect or close/connect the short-circuit connection 109d depending on a state of SW1. The short-circuit connection 109d is established

PCT/EP2014/078116

15

in a conducting/ON state of SW1 and the short-circuit connection 109d is broken/disconnected in a non-conducting/OFF state of SW1. In the conducting state of SW1, the primary side circuit and the secondary side circuit of the converter core 105 are connected by the short-circuit connection 109d. The step-up DC-DC power

- 5 converter 100d additionally comprises a diode 111d connected between the negative input terminal 101d and a negative electrode 115d of the output capacitor C_{out}. This diode 111d is reverse biased and hence non-conducting when SW1 is conducting/ON because the negative output electrode 115d is at a higher potential than the negative input terminal 101d. Consequently, when SW1 is ON or conducting the
- 10 step-up DC-DC power converter 100d operates in a first distinct mode where the functionality of the power converter 100d is similar to the functionality of the previously discussed step-up DC-DC power converter 100b with the accompanying advantages.

A second distinct mode of the step-up DC-DC power converter 100d is reached or

- 15 provided in the non-conducting/OFF state of SW1 where the short-circuit connection 109d is broken or opened. In this second distinct mode, the diode 111d will be forward biased and conducting such that the primary side circuit and the secondary side circuit are electrically connected both through galvanic isolation barrier 107d and through the diode 111d. Hence, the conducting diode 111d bypasses the gal-
- 20 vanic isolation barrier 107d in the second mode of the step-up DC-DC power converter 100d. However, the overall functionality of the step-up DC-DC power converter 100d in the second mode of operation remains similar to the functionality of a corresponding ordinary (i.e. lacking the first mode of operation) step-up DC-DC power converter.

25

The mode switching feature of the present step-up DC-DC power converter 100d is accompanied with several advantages. The mode switching feature increases the dynamic voltage operating range of the power converter 100d. To illustrate these advantages consider an ordinary DC-DC power converter designed for a DC input

30 voltage of 10 V and a DC output voltage range from 5 – 15 V. If this ordinary DC-DC power converter is converted or configured as the present step-up DC-DC power converter 100d, the DC output voltage range may be increased to 5 - 25 V by switching the re-configured power converter between the first and second modes of operation. This increase of DC output voltage range provided by the mode switching

PCT/EP2014/078116

16

feature of the present step-up DC-DC power converter 100d is particular advantageous for resonant power converters which generally suffer from a restricted or narrow DC output voltage range compared to non-resonant DC-DC power converters. However, exploiting the mode switching feature of the present step-up DC-DC pow-

5 er converter 100d requires that the intended application does not require galvanic isolation between the primary and secondary side circuits due to the electrical path through the diode 111d.

FIG. 1E) is simplified electrical circuit diagram of a first converter core 105e that may

- 10 be utilized as converter core 105 in each of the step-up DC-DC power converter embodiments 100b, 100c, 100d, illustrated on FIGS. 1A), 1B), 1C) and 1D), respectively. The first converter core 105e comprises a plurality of separate resonant DC-DC power converter cores 110e. Each of the separate resonant DC-DC power converter cores 110e comprises an input side 111 of a switched energy storage network
- 15 coupled to an output side 113 of the switched energy storage network through a galvanic isolation barrier 107e. The input sides 111 of the resonant DC-DC power converter cores 110e may be connected in parallel or series. The output sides 113 of the resonant DC-DC power converter cores 110e may likewise be connected in parallel or series. The parallelization of the plurality of input sides 111 and/or the
- 20 parallelization of the one or more output sides 113 increases the power rating of a step-up DC-DC power converter utilizing the first converter core 105e. The skilled person will understand that each of the separate resonant DC-DC power converter cores 110e may comprise one of the prior art resonant DC-DC power converter cores discussed below with reference to FIGS. 2, 3, 4, 5 and 6.

25

FIG. 1F) is simplified electrical circuit diagram of a second converter core 105f of each of the step-up DC-DC power converter embodiments 100b, 100c, 100d, illus-trated on FIGS. 1A), 1B), 1C) and 1D), respectively. The second converter core 105f comprises a plurality of separate resonant power inverters 114e. Each of the sepa-

30 rate resonant power inverter cores 114e comprises an input side 115 of a switched energy storage network coupled to one or more rectifier(s) 117 of the resonant DC-DC power converter core 105f through a galvanic isolation barrier 107f. The separate resonant power inverter cores 114e may be connected in parallel or series. Likewise, the respective output side of the one or more rectifier(s) 117 may also be

connected in series or parallel. However, galvanic isolation may be inserted between the one or more rectifier(s) 117 if these are coupled in series.

FIG. 2A) shows an electrical circuit diagram of a prior art isolated class E resonant

- 5 DC-DC converter 200 comprising a series resonant circuit including inductor L₂ and capacitor C₁. The prior art class E resonant converter comprises a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 207. The primary side circuit comprises a positive input terminal 202 and a negative input terminal 201 for receipt of a DC or AC input voltage V_{in} from a voltage or power
- 10 source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 202b and a negative input terminal 201 to form an energy reservoir for the voltage source. The primary side circuit additionally comprises a switched energy storage network which includes first and second series connected inductors L₁ and L₂ and a MOSFET switch S with a drain terminal connected to a
- 15 midpoint node between the L₁ and L₂. The primary side circuit is arranged in front of an isolation barrier 207 formed by coupling capacitors C₁ and C₂. The secondary side circuit comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage V_{out} at output terminal 204. A second negative electrode of the output capacitor C_{out} is coupled to a negative terminal 203
- 20 of the converter output voltage. A load of the isolated class E resonant DC-DC converter 200 is schematically illustrated by load resistor R_L and coupled between the positive and negative output terminals 204, 203.
- FIG. 2B) is an electrical circuit diagram of a class E resonant step-up DC-DC power
 converter 200b comprising a series resonant circuit in accordance with a sixth embodiment of the invention. The class E resonant step-up DC-DC power converter
 200b may be obtained by conversion of the above-mentioned prior art isolated class
 E resonant DC-DC converter 200 by inserting or adding an electrical short circuit
 connection 209b extending across a galvanic isolation barrier 207b of the converter
- 30 200b in accordance with the principles discussed above in connection with the first embodiment of the invention discussed above in connection with FIGS. 1A) and 1B). The galvanic isolation barrier 207b comprises series capacitors C₁ and C₂. The electrical short circuit connection 209b connects the positive input terminal 202b and the second negative electrode 203b of the output capacitor C_{out}. As discussed in con-

PCT/EP2014/078116

18

nection with FIGS. 1A) and 1B), the electrical short-circuit connection or wire 209b effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade between the output voltage V_{out} and the negative input terminal 201b. Hence, the electrical or power converter load, schematically illustrated by the load resistor

- 5 R_L, is coupled between the converter output voltage at the output terminal 204b and the negative input terminal 201b. The skilled person will understand that the series capacitor C₂ of the galvanic isolation barrier 207b prevents DC current from flowing from the second negative electrode 203b of the output capacitor C_{out} and back to the negative input terminal 201b electrode of the input voltage source. In this manner,
- 10 the DC current is directed or forced through the electrical short circuit connection 209b and back through the input capacitor C_{in}. In this manner, despite being electrically by-passed by the conversion, the isolation barrier 207 is important for the operation of the present class E resonant step-up DC-DC power converter 200b as node 201b, 203b and 202b would be directly electrically connected causing a short
- 15 circuit at the converter input.

The class E resonant step-up DC-DC power converter 200b may comprise a capacitor (not shown) arranged across drain and source terminals of the MOSFET switch S to increase a resonant current and/or adjust/fine-tune a resonance frequency of

20 the power converter 200b. Likewise, a yet further capacitor (not shown) may be arranged across the rectifying diode D to adjust a duty cycle of the secondary part of the power converter 200b, i.e. the class E rectifier.

FIG. 3A) is an electrical circuit diagram of a prior art flyback DC-DC converter 300.

- 25 The prior art DC-DC converter 300 comprises a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 307. The primary side circuit comprises a positive input terminal 302 and a negative input terminal 301 for receipt of a DC or AC input voltage V_{in} from a voltage or power source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal
- 30 302 and a negative input terminal 301 to form an energy reservoir for the input voltage source. The primary side circuit additionally comprises a switched energy storage network which comprises a first inductor L_P having a first end coupled to the positive input terminal 302 and a second end to a drain terminal of a MOSFET switch S. A source terminal of the MOSFET switch S is coupled to the negative input

PCT/EP2014/078116

19

terminal 301. The first inductor L_P is a primary transformer winding of a transformer which provides a galvanic isolation barrier 307 of this prior art DC-DC converter 300. A secondary side circuit of the power converter 300 comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage

- 5 V_{out} at output terminal 304. A second negative electrode of the output capacitor C_{out} is coupled to a negative terminal 303 of the converter output voltage. An electrical or power converter load is schematically illustrated by load resistor R_L and coupled between the positive and negative output terminals 304, 303 of the prior art DC-DC converter 300. The secondary side circuit furthermore comprises a second inductor
- 10 L_s which is a secondary transformer winding of the above-mentioned transformer. The secondary transformer winding L_s has a first end coupled to a rectifying diode D and a second end coupled to the negative electrode of the output capacitor C_{out}. The rectifying diode D rectifies AC current generated by the secondary transformer winding L_s and generates a DC voltage as the converter output voltage between the pos-
- 15 itive and negative output terminals 304, 303. An electrical or power converter load is schematically illustrated by load resistor R_L coupled between the positive and negative output terminals 304, 303.

FIG. 3B) is an electrical circuit diagram of a flyback step-up DC-DC power converter 300b in accordance with a 7th embodiment of the invention. The flyback power converter 300b may be obtained by conversion of the above-mentioned prior art isolated flyback DC-DC converter 300 by inserting or adding an electrical short circuit connection 309b extending across a galvanic isolation barrier formed by the transformer comprising the magnetically coupled primary and secondary transformer

- 25 windings L_P and L_S. The electrical short circuit connection 309b connects the positive input terminal 302b and the second negative electrode 303b of the output capacitor C_{out}. As discussed in connection with FIGS. 1A) and 1B), the electrical short-circuit connection or wire 309b effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade between the output voltage V_{out} and the negative
- 30 input terminal 301b. Hence, the electrical or power converter load, schematically illustrated by the load resistor R_L, is coupled between the converter output voltage at the output terminal 304b and the negative input terminal 301b. The skilled person will understand that the transformer coupling prevents DC current from flowing from the second negative electrode 303b of the output capacitor C_{out} and back to the

PCT/EP2014/078116

20

negative input terminal 301b electrode of the input voltage source. In this manner, the DC current is directed or forced through the electrical short circuit connection 309b and back through the input capacitor C_{in} .

- 5 FIG. 4A) is an electrical circuit diagram of a prior art isolated single-ended primaryinductor converter (SEPIC) 400. The prior art SEPIC 400 comprises a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 407. The primary side circuit comprises a positive input terminal 402 and a negative input terminal 401 for receipt of a DC or AC input voltage V_{in} from a voltage or power
- 10 source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 402 and a negative input terminal 401 to form an energy reservoir for the input voltage source. The primary side circuit additionally comprises a switched energy storage network which includes a first inductor L₁ having first node coupled to the DC or AC input voltage V_{in} and a second node coupled to a drain
- 15 terminal of a MOSFET switch S. A source terminal of the MOSFET switch S is coupled to the negative input terminal 401. The primary side circuit is arranged in front of an isolation barrier 407 formed by coupling capacitors C₁ and C₂. The secondary side circuit comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage V_{out} at output terminal 404. A second
- 20 negative electrode of the output capacitor C_{out} is coupled to a negative terminal 403 of the converter output voltage. A rectifying diode D rectifies AC current generated by a second inductor L₂ and generates a DC voltage as the converter output voltage V_{out} between the positive and negative output terminals 404, 403. A load of the SEPIC 400b, illustrated by load resistor R_L, is coupled between the positive and
- 25 negative output terminals 404, 403.

FIG. 4B) is an electrical circuit diagram of a SEPIC 400b in accordance with an 8th embodiment of the invention. The SEPIC 400b may be obtained by conversion of the above-mentioned prior art SEPIC 400 by inserting or adding an electrical short circuit connection 409b extending across a galvanic isolation barrier 407b of the SEPIC 400b. The galvanic isolation barrier 407b comprises series capacitors C₁ and C₂. The electrical short circuit connection 409b connects the positive input terminal 402b and the second negative electrode 403b of the output capacitor C_{out}. As discussed in connection with FIGS. 1A) and 1B), the electrical short-circuit connection

PCT/EP2014/078116

21

or wire 409b effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade between the output voltage V_{out} and the negative input terminal 401b. Hence, the electrical or power converter load, schematically illustrated by the load resistor R_L , is coupled between the converter output voltage at the output ter-

5 minal 404b and the negative input terminal 401b. The skilled person will understand that the series capacitor C₂ of the galvanic isolation barrier 407b prevents DC current from flowing from the second negative electrode 403b of the output capacitor C_{out} and back to the negative input terminal 401b electrode of the input voltage source as discussed previously.

10

15

The SEPIC 400b may comprise a capacitor (not shown) connected or arranged across drain and source terminals of the MOSFET switch S to increase a resonant current and/or adjust/fine-tune a resonance frequency of the SEPIC 400b. Likewise, a yet further capacitor (not shown) may be arranged across the rectifying diode D to adjust a duty cycle of the power converter 400b.

FIG. 5A) shows an electrical circuit diagram of a prior art isolated class DE resonant DC-DC converter 500 comprising a series resonant circuit including L_0 and C_0 . The prior art class DE converter 500 comprises a primary side circuit and a secondary

- 20 side circuit connected through a galvanic isolation barrier 507. The primary side circuit comprises a positive input terminal 502 and a negative input terminal 501 for receipt of a DC or AC input voltage V_{in} from a voltage or power source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 502 and a negative input terminal 501 to form an energy reservoir for the input volt-
- 25 age source. The primary side circuit additionally comprises a switched energy storage network comprising a half-bridge circuit comprising cascaded MOSFET switches S_1 and S_2 arranged across the positive and negative input terminals 502, 501, respectively. An output 506 of the half-bridge circuit is coupled to a first inductor L_0 of the series resonant circuit and the latter is coupled in series with the capacitor C_0 .
- 30 This primary side circuit is arranged in front of the isolation barrier 507 formed by the coupling capacitor C_0 of the series resonant circuit and a second capacitor C_B inserted between the negative input terminal 501 and a negative output voltage terminal 503 to provide DC isolation between these in this prior art class DE converter 500. The secondary side circuit comprises an output capacitor C_{out} having a first

PCT/EP2014/078116

22

electrode electrically connected to the converter output voltage V_{out} at output terminal 504. A second negative electrode of the output capacitor C_{out} is coupled to the negative terminal 503 of the converter output voltage. A pair of rectifying diodes D_1 and D_2 rectifies AC current generated by excitation of the series resonant circuit and

- 5 generates a DC voltage as the converter output voltage V_{out} between the positive and negative output terminals 504, 503. A load of the class DE converter 500, illustrated by load resistor R_{L_i} is coupled between the positive and negative output terminals 504, 503.
- FIG. 5B) is an electrical circuit diagram of a class DE resonant DC-DC converter 500b in accordance with a 9th embodiment of the invention. The class DE converter 500b may be obtained by conversion of the above-mentioned prior art isolated class DE resonant DC-DC converter 500 by inserting or adding an electrical short circuit connection 509b extending across a galvanic isolation barrier 507b of the class DE
- 15 converter 500b. The galvanic isolation barrier 507b comprises series capacitors C₁ and C₂. The electrical short circuit connection 509b connects the positive input terminal 502b and a second negative electrode 503b of the output capacitor C_{out}. As discussed in connection with FIGS. 1A) and 1B), the electrical short-circuit connection or wire 509b effectively places the output capacitor C_{out} and input capacitor C_{in}
- 20 is series or cascade between the output voltage V_{out} and the negative input terminal 501b. Hence, the electrical or power converter load, schematically illustrated by the load resistor R_L , is coupled between the converter output voltage at the output terminal 504b and the negative input terminal 501b. The skilled person will understand that the series capacitor C_2 of the galvanic isolation barrier 507b prevents DC cur-
- 25 rent from flowing from the second negative electrode 503b of the output capacitor C_{out} and back to the negative input terminal 501b electrode of the input voltage source as discussed previously. The series capacitor C₁ serves two purposes both forming part of the isolation barrier 507b and forming part of the series resonant circuit also including inductor L.

30

The class DE converter 500b may comprise a pair of capacitors (not shown) connected or arranged across the drain and source terminals of each of the MOSFET switches S_1 and S_2 to increase a resonant current and/or adjust/fine-tune a resonance frequency of the DE converter 500b. Likewise, a yet further pair of capacitors (not shown) may be arranged across the rectifying diodes D_1 and D_2 to adjust a duty cycle of the secondary part of the power converter 500b, i.e. the class DE rectifier.
CLAIMS

5

1. A step-up DC-DC power converter comprising:

a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier,

the primary side circuit comprising a positive and a negative input terminal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals,

the secondary side circuit comprising an output capacitor chargeable to a converter
 output voltage between a first positive electrode and a second negative electrode,
 a switched energy storage network configured for alternatingly being charged from
 the input voltage and discharged to the output capacitor through the galvanic isola tion barrier in accordance with a switch control signal to produce the converter output voltage,

15 an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive input terminal of the primary side circuit or, in a second case, connecting the second positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series cou-

20 pling of the output capacitor and the input capacitor, a load connection, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or, in the second case, between the second negative electrode of the output capacitor and the positive input terminal.

25 2. A step-up DC-DC power converter according to claim 1, wherein the galvanic isolation barrier comprises:

a pair of magnetically coupled inductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the secondary side circuit.

30

3. A step-up DC-DC power converter according to claim 2, wherein the first and second inductors are wound around a common magnetic permeable structure to form an isolation transformer.

4. A step-up DC-DC power converter according to claim 1, wherein the galvanic isolation barrier comprises:

a first capacitor coupled in series with the positive input terminal of the primary side circuit and the first positive electrode of the output capacitor; and

5 a second capacitor coupled in series with the negative input terminal of the primary side circuit and the second negative electrode of the output capacitor;

5. A step-up DC-DC power converter according to claim 1, wherein the electrical short-circuit connection has a DC resistance of less than 1 k Ω , even more preferably

10 less than 100 Ω , such as less than 10 Ω .

6. A step-up DC-DC power converter according to claim 4 or 5, wherein the switched energy storage network comprises:

first and second series connected inductors and connected in series with the posi-

- 15 tive input voltage terminal, a semiconductor switch having a first switch node connected between a mid-point node between the first and second series connected inductors, a second switch node connected to the negative input terminal of the primary side circuit and a control terminal connected to the switch control terminal; and
- 20 a third inductor having a first end connected to a second end of the second inductor through the first capacitor of the galvanic isolation barrier and a second end connected to the converter output voltage at the positive electrode of the output capacitor,

a rectifier connected between the first end of the third inductor and the negative

electrode of the output capacitor.

7. A step-up DC-DC power converter according to claim 3, wherein the first and second inductors are integrated in the switched energy storage network; the first inductor being arranged with a first inductor end connected to the positive

30 input voltage terminal and a second inductor end connected to a first node of a semiconductor switch,

a second node of the semiconductor switch being connected to the negative input terminal of the primary side circuit; and the second inductor comprising a first inductor end connected to the first positive electrode of output capacitor and a second inductor end connected to the second negative electrode, respectively, of the output capacitor through a rectifier.

- 5 8. A step-up DC-DC power converter according to any of the preceding claims, wherein the switched energy storage network comprises at least one semiconductor switch such as a MOSFET or IGBT such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET.
- 10 9. A step-up DC-DC power converter according to any of the preceding claims, wherein a frequency of the switch control signal of the switched energy storage network has a frequency at or above 10 MHz, more preferably at or above 30 MHz.

10. A step-up DC-DC power converter according to any of the preceding claims,comprising a resonant DC-DC power converter.

11. A step-up DC-DC power converter according to any of the preceding claims, wherein each of the input capacitor and the output capacitor has a capacitance smaller than 100 nF.

20

12. A step-up DC-DC power converter according to any of the preceding claims, further comprising:

a rectifying element, such as a diode, coupled between the positive input terminal and second negative electrode of the output capacitor; and

- 25 a mode selecting semiconductor switch configured to selectively break and close the electrical short-circuit connection such that: in a first mode of the step-up DC-DC power converter, establishing the series connection of the output capacitor and the input capacitor; and in a second mode of the step-up DC-DC power converter, break the series coupling
- 30 of the output capacitor and the input capacitor.

13. A method of converting an isolated DC-DC power converter to a step-up DC-DC power converter with higher power conversion efficiency, said method comprising steps of:

providing a primary side circuit and a secondary side circuit of the isolated DC-DC power converter,

coupling an input capacitor between a positive input terminal and a negative input terminal of the primary side circuit,

5 coupling an output capacitor between a positive and a negative terminal of the secondary side circuit,

providing electrical coupling of the primary side circuit and the secondary side circuit through a galvanic isolation barrier,

providing a switched energy storage network configured for alternatingly being

10 charged from an input voltage of the converter and discharged to the output capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce a converter output voltage,

connecting, in a first case, an electrical short-circuit across the galvanic isolation barrier from the negative output terminal of the secondary side circuit to the positive

15 input terminal of the primary side circuit or connecting, in a second case, the positive output terminal of the secondary side circuit to the negative input terminal of the primary side circuit thereby establishing in both the first case and the second case a series coupling of the output capacitor and the input capacitor,

coupling, in a first case, a power converter load between the positive terminal of the secondary side circuit and the negative input terminal or coupling, in the second

20 secondary side circuit and the negative input terminal or coupling, in the second case, the power converter load between the negative terminal of the secondary side circuit and the positive input terminal of the primary side circuit.

14. A method of converting an isolated DC-DC power converter to a step-up DC-DC

25 power converter according to claim 13, wherein the isolated DC-DC power converter comprises a resonant DC-DC power converter.

15. A method of converting an isolated DC-DC power converter to a step-up DC-DC power converter according to claim 13 or 14, comprising further steps of:

30 connecting a rectifying element, such as a diode, between the positive input terminal and second negative electrode of the output capacitor; and inserting a mode selecting semiconductor switch into the electrical short-circuit connection for selectively breaking and closing the short circuit connection such that: establishing the series connection of the output capacitor and the input capacitor in a first mode of the step-up DC-DC power converter; and breaking or disconnecting the series coupling of the output capacitor and the input capacitor in a second mode of the step-up DC-DC power converter.

5

1/8





FIGS. 1A) – 1B)



2/8











200b –

B)

FIGS. 2A, B







FIGS. 3A, B





400b —

B)







FIGS. 5A), B)

A.9 M. Kovacevik and M. Madsen: "Step-down dc-dc power converter", WO2015110427 A1, July 30th 2015 (12) INTERNATIONAL APPLICATION PUBLISHED UNDER THE PATENT COOPERATION TREATY (PCT)

(19) World Intellectual Property Organization

International Bureau



(43) International Publication Date 30 July 2015 (30.07.2015)

- (51) International Patent Classification: H02M 3/335 (2006.01)
- (21) International Application Number: PCT/EP2015/051007
- (22) International Filing Date:
 - 20 January 2015 (20.01.2015)
- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 14152070.0 22 January 2014 (22.01.2014) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: KOVACEVIC, Milovan; Aldershvilevej 8, 1., DK-2800 Kgs. Lyngby (DK). MADSEN, Mickey P; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK).
- (74) Agent: GUARDIAN IP CONSULTING I/S; Henrik Bagger Olsen, Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).
- (81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM,

(10) International Publication Number WO 2015/110427 A1

AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN, TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.

(84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, ST, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

- with international search report (Art. 21(3))
- before the expiration of the time limit for amending the claims and to be republished in the event of receipt of amendments (Rule 48.2(h))

(54) Title: RESONANT STEP-DOWN DC-DC POWER CONVERTERS



(57) Abstract: The present invention relates to a resonant step-down DC-DC power converter which comprises a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier. The primary side circuit comprises a positive and a negative input terminal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals and the secondary side circuit comprises an output capacitor chargeable to a converter output voltage between a first positive electrode and a second negative electrode. A resonant network is con-figured for alternatingly being charged from the input voltage and discharged to the output capacitor through the galvanic isolation barrier by a semiconductor switch arrangement in accordance with a switch control signal to produce the converter output voltage. The resonant step-down DC-DC power converter comprises an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the negative input terminal of the primary side circuit threeby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection is established, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or, in the second case, between the second negative electrode of the output capacitor and the negative input terminal.

PCT/EP2015/051007

1

RESONANT STEP-DOWN DC-DC POWER CONVERTERS

The present invention relates to a resonant step-down DC-DC power converter which comprises a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier. The primary side circuit comprises a positive and a neg-

- 5 ative input terminal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals and the secondary side circuit comprises an output capacitor chargeable to a converter output voltage between a first positive electrode and a second negative electrode. A resonant network is configured for alternatingly being charged from the input voltage and discharged to the
- 10 output capacitor through the galvanic isolation barrier by a semiconductor switch arrangement in accordance with a switch control signal to produce the converter output voltage. The resonant step-down DC-DC power converter comprises an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive
- 15 input terminal of the primary side circuit or, in a second case, connecting the second positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection is established, in the first case, between the first positive electrode of the output capacitor
- 20 and the positive input terminal or, in the second case, between the second negative electrode of the output capacitor and the negative input terminal.

BACKGROUND OF THE INVENTION

Power density and component costs are key performance metrics of both isolated

- 25 and non-isolated DC-DC power converters to provide the smallest possible physical size and/or lowest costs for a given output power requirement or specification. Resonant power converters are particularly useful for high switching frequencies such as frequencies above 1 MHz where switching losses of standard SMPS topologies (Buck, Boost etc.) tend to be unacceptable for conversion efficiency reasons. High
- 30 switching frequencies are generally desirable because of the resulting decrease of the electrical and physical size of circuit components of the power converter like inductors and capacitors. The smaller components allow increase of the power density of the DC-DC power converter. In a resonant power converter an input "chopper" semiconductor switch (often MOSFET or IGBT) of the standard SMPS is re-

PCT/EP2015/051007

2

placed with a "resonant" semiconductor switch. The resonant semiconductor switch relies on resonances of circuit capacitances and inductances to shape the waveform of either the current or the voltage across the semiconductor switch such that, when state switching occurs in the semiconductor switch, there is essentially no current

- 5 through or essentially no voltage across the semiconductor switch. Hence power dissipation is largely eliminated in at least some of the intrinsic capacitances or inductances of the input semiconductor switch such that a marked increase of the switching frequency becomes feasible for example to values above 10 MHz. This concept is known in the art under designations like zero voltage and/or zero current
- 10 switching (ZVS and/or ZCS) operation. Commonly used switched mode power converters operating under ZVS and/or ZCS are often described as class E, class F or class DE inverters or power converters.
- In view of the above, it remains a challenge to reduce the size and lower the com ponent costs of both isolated and non-isolated DC-DC power converters. Hence,
 novel resonant step-down DC-DC power converter topologies which reduce the re quired maximum voltage or power rating of active and passive components of the
 resonant DC-DC converter are highly desirable. Likewise, novel resonant step-down
 DC-DC power converter topologies which reduce the physical size or cost of active
 and passive components for example inductors, capacitors, transistors and diodes
- are highly desirable.

SUMMARY OF THE INVENTION

- A first aspect of the invention relates to a resonant step-down DC-DC power converter comprising a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier; the primary side circuit comprises a positive input terminal and a negative input terminal for receipt of an input voltage. An input capacitor is coupled or connected between the positive and negative input terminals on the primary side circuit. The secondary side circuit comprising an output capacitor charge-
- 30 able to a converter output voltage between a first positive electrode and a second negative electrode of the output capacitor. The resonant step-down DC-DC power converter comprises a resonant network configured for alternatingly being charged from the input voltage and discharged to the output capacitor through the galvanic isolation barrier by a semiconductor switch arrangement in accordance with a switch

PCT/EP2015/051007

3

control signal to produce the converter output voltage, wherein a frequency of the switch control signal has a frequency at or above 20 MHz, more preferably at or above 30 MHz. An electrical short-circuit connection across the galvanic isolation barrier connects, in a first case, the second negative electrode of the output capaci-

- 5 tor to the positive input terminal of the primary side circuit or, in a second case, connects the second positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor. A load connection of the resonant step-down DC-DC power converter exists or is established,
- 10 in the first case, between the first positive electrode of the output capacitor and the positive input terminal or, in the second case, between the second negative electrode of the output capacitor and the negative input terminal.
- The present invention is described in detail in the following with reference to specific 15 implementations derived from isolated resonant DC-DC power converters of class E, class DE and SEPIC topologies. The skilled person will understand that the invention is equally applicable to other types of isolated resonant DC-DC power converters such as class φ_2 (EF₂) inverters and rectifiers and resonant boost, buck, LCC converters etc. In this context the term DC can refer to a slowly varying input voltage 20 where the term "DC" applies to variations of the input voltage level which are slow compared to a switching frequency, i.e. the above-mentioned frequency of the switch control signal, of the resonant step-down DC-DC power converter. The input voltage may accordingly comprise a substantially constant DC input voltage or a rectified AC voltage comprising a DC voltage component and an AC voltage com-25 ponent at a frequency significantly lower, for example more than 100 times lower, than the switching frequency of the resonant step-down DC-DC power converter. In the latter case, the input voltage may be supplied by an output of a 50/60 Hz mains voltage rectifier connected to the positive and negative input terminals of the resonant step-down DC-DC power converter.

30

The secondary side circuit of the resonant step-down DC-DC power converter may comprise a rectification circuit coupled between the resonant network and the output capacitor to convert a resonant AC voltage waveform of the resonant circuit into a DC converter output voltage.

PCT/EP2015/051007

4

The semiconductor switch arrangement preferably comprises one or more individual semiconductor switches configured for zero-voltage-switching and/or zero-current-switching. The present resonant step-down DC-DC power converters can be oper-ated at very high switching frequencies, i.e. at or above the 20 MHz or 30 MHz, be-

5 cause the zero-voltage-switching and/or zero-current-switching of the one or more individual controllable semiconductor switches facilitated by the resonant network effectively reduces switching power losses of the semiconductor switch arrangement. A switching frequency at or above 30 MHz is normally considered as VHF operation of the resonant step-down DC-DC power converter.

10

The semiconductor switch arrangement may comprise various ordinary switch topologies such as single switch topologies, half-bridge switch topologies or full-bridge switch topologies. The one or more individual controllable semiconductor switches of the semiconductor switch arrangement may comprise a MOSFET or IGBT such

- 15 as a Gallium Nitride (GaN) or Silicon Carbide (SiC) transistor. A control terminal, e.g. a gate or base, of each of the controllable semiconductor switches may be coupled to, and driven by, the switch control signal to alternatingly force each of the controllable semiconductor switches between on-states and off-states. In the onstate an inductor of the resonant network may be charged with energy from the input
- 20 voltage source and in the following off-state release stored energy to the output capacitor to charge the latter. The resonant network preferably comprises at least one inductor and at least one capacitor where one or both of these components may comprise parasitic inductances or capacitances of active components or passive components of the resonant step-down DC-DC power converter. The secondary
- 25 side circuit of the resonant step-down DC-DC converter may comprise one or more passive and/or active rectifying element(s) such as a diode or transistor inserted in front of the converter load.
- The electrical connection of an input voltage or power source, e.g. a DC voltage generator, to the present resonant step-down DC-DC power converter is in the first case made between the negative input terminal of the primary side circuit and the positive electrode of the output capacitor such that the input and output capacitors of the converter are connected in series between the positive and negative terminals of the input voltage or power source. Likewise, in the second case the input voltage or

5

power source is connected between the negative electrode of the output capacitor and the positive input terminal of the primary side circuit such that the input and output capacitors of the converter once again are connected in series between the positive and negative terminals of the input voltage or power source.

- 5 The skilled person will understand that the electrical short-circuit connection across the galvanic isolation barrier eliminates the galvanic isolation between the primary and secondary side circuits of the resonant step-down DC-DC converter by interconnecting the second, negative, electrode of the output capacitor and the positive input terminal in the first case or the first, positive, electrode of the output capacitor
- 10 and the negative input terminal in the second case. However, the electrical shortcircuit connection provides numerous new benefits to the resonant step-down DC-DC converter as a whole and the lack of galvanic isolation is acceptable in numerous areas of applications where the converter circuit remains isolated from users such as retrofit LED bulbs and tubes. The series connection of the output and input
- 15 capacitors between the input voltage/power source terminals established by the electrical short-circuit connection has several advantageous effects. Input current flowing through the input capacitor to charge this to the input voltage is also forced to flow through the output capacitor and charges the output capacitor such that energy or power is transferred directly from the input/primary side to the out-
- 20 put/secondary side of the resonant step-down DC-DC converter. Hence, less power or energy needs to be transferred through the resonant network and the isolation barrier for a given output power of the converter as explained in further detail below with reference to the resonant step-down DC-DC converter embodiments of FIGS. 1A), 1B) and 1C). The reduced amount of power transferred to the secondary side
- 25 circuitry through the resonant network and isolation barrier reduces power losses of or in the resonant step-down DC-DC converter. This advantage is obtained because the direct energy or power transfer between the input and output capacitors of the converter leads to a lower power loss than the ordinary transfer of energy or power through the resonant network.

30

Furthermore, the reduced amount of power which needs to be transferred through the resonant step-down DC-DC converter leads to reduced power handling requirements for both passive and active components, e.g. semiconductors switches, capacitors, diodes etc., allowing less costly and physically smaller components to be

PCT/EP2015/051007

6

applied. A yet further significant advantage of the present resonant step-down DC-DC converter is that the primary side circuit is only subjected to the input voltage between the positive and negative input terminals of the primary side circuit instead of the entire converter input voltage supplied by the external input voltage or power

- 5 source as in ordinary isolated DC-DC power converters. As described above, the input voltage or power source is connected either between the negative input terminal and the positive electrode of the output capacitor or, in the second case, between the negative electrode of the output capacitor and the positive input terminal of the primary side circuit in the present resonant step-down DC-DC converter. The
- 10 reduced voltage levels in the primary side circuit of the resonant step-down DC-DC power converter reduce the required maximum voltage ratings of active and passive components therein such as semiconductor switch or switches, inductor(s), capacitors, diode(s) etc. The reduced maximum voltage rating leads to physically smaller and/or less costly active and passive components with increased life span.

15

The galvanic isolation barrier may comprise a transformer which comprises a pair of electro-magnetically or magnetically coupled inductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the secondary side circuit. The first and second inductors could be

- 20 discrete windings both wound around a common magnetic permeable structure to form an isolation transformer. In one alternative embodiment, the first and second inductors are arranged to be electro-magnetically coupled without a common magnetically permeable structure to form a coreless isolation transformer. In such a coreless isolation transformer, the first and second inductors may be integrated in a
- 25 printed circuit board without intervening magnetic material. The printed circuit board could have the entire resonant step-down DC-DC power converter mounted there-on. The first and second inductors are preferably arranged such that a magnetic coupling factor, k, between the first and second inductors is larger than 0.25. The first and second inductors of the coreless isolation transformer may for example
- 30 comprise first and second embedded coils, respectively, formed in one or more conductive layers of the printed circuit board as disclosed in the applicant's co-pending application No. PCT/EP2014/079037.

PCT/EP2015/051007

7

In yet another embodiment, the galvanic isolation barrier comprises a first capacitor coupled in series with the positive input terminal of the primary side circuit and the first positive electrode of the output capacitor and a second capacitor coupled in series with the negative input terminal of the primary side circuit and the second

- 5 negative electrode of the output capacitor. This embodiment is particularly advantageous in connection with the high switching frequency of the present resonant stepdown DC-DC power converter where the power loss in the above-discussed common magnetic permeable structure of the transformer often will be unacceptable because of the accompanying decrease of power conversion efficiency of the reso-
- 10 nant power converter. The capacitor based galvanic isolation barrier becomes particularly advantageous as the capacitance of each of the first and second (isolation) capacitors can be small, such as less than 100 nF for example smaller than 1 nF such as about 100pF. Such isolation capacitors may be formed by SMD mounted ceramic capacitors with a very small footprint e.g. a footprint less than 2 cm² for ex-
- 15 ample a footprint down to or less than about 5 mm^2 .

The high switching frequency of the present resonant step-down DC-DC power converter, i.e. at or above 20 MHz, also makes the capacitance of each of the input and output capacitors to be small compared to conventional non-resonant DC-DC power

- 20 converters operating below 1 MHz. Hence, neither the input capacitor nor the output capacitor needs to be an electrolytic capacitor which generally is plagued by relatively low reliability and short-life span instead the first and second isolation capacitors and/or the input and output capacitors of the present resonant step-down DC-DC power converter can be reliable, physically small and inexpensive. The skilled
- 25 person will understand that the input and output capacitors may be formed exclusively by parasitic capacitances associated with the primary side circuit and the secondary side circuit, respectively.
- Another advantage of the high switching frequency of the present resonant stepdown DC-DC power converter, as set by the switch control signal, is a marked size decrease of an EMI filter that may be placed in front of the positive and negative input terminals of present converter. The size decrease of the EMI filter is possible because ripple voltage components on the input voltage induced by the switching activity of the present resonant power converter are located around the switching

PCT/EP2015/051007

8

frequency of converter, i.e. around or above 20 MHz, where the necessary filter components of the EMI filter can be physically small.

The skilled person will appreciate that a practical embodiment of the electrical short 5 circuit connection will possess a finite DC resistance. An upper limit of this finite DC resistance will vary depending on input/output voltage and/or current requirements of the resonant step-down DC-DC power converter. The electrical short-circuit connection may possess a DC resistance of less than 1 k Ω , even more preferably less than 100 Ω , such as less than 10 Ω . In other embodiments, the electrical short cir-

10 cuit connection may have a unidirectional resistance such that the DC resistance only falls below the above-mentioned upper limits in one direction and exhibits a much larger DC resistance in the opposite direction, i.e. a diode characteristic for example provided by a diode element or a controlled semiconductor switch such as a MOSFET.

15

One embodiment of the resonant step-down DC-DC power converter is based on a Class E converter and the resonant network comprises first and second series connected inductors which are connected in series with the positive input terminal. A semiconductor switch is arranged with a first switch node connected between a mid-

- 20 point node between the first and second series connected inductors and a second switch node connected to the negative input terminal of the primary side circuit. A control terminal of the semiconductor switch is connected to the switch control terminal. A rectifying or rectification circuit of the resonant step-down DC-DC power is connected between the first and second capacitors of the galvanic isolation barrier
- 25 and the first positive electrode and the second negative electrode of the output capacitor. The rectifying circuit may comprise a semiconductor diode or a synchronous semiconductor switch coupled, i.e. electrically connected, to a third inductor of the resonant step-down DC-DC power converter.
- 30 Another embodiment of the resonant step-down DC-DC power converter is based on a converter topology wherein the first and second inductors of the isolation transformer are integrated in the resonant network. The first inductor is arranged with a first inductor end connected to the positive input voltage terminal and a second inductor end connected to a first node of a semiconductor switch such as a drain ter-

PCT/EP2015/051007

9

minal of a MOSFET switch. A second node of the semiconductor switch is connected to the negative input terminal of the primary side circuit. The second inductor comprises a first inductor end connected to the first capacitor of the isolation barrier and a second inductor end connected to the second capacitor of the isolation barri-

5 er. A rectifying circuit is connected across the second inductor between the first and second capacitors of the galvanic isolation barrier and the first positive electrode and the second negative electrode of the output capacitor.

One embodiment of the resonant step-down DC-DC power converter comprises a

- 10 mode switching feature that is accompanied with several advantages such as increases a dynamic voltage operating range and/or improving a power factor (PF) of the resonant power converter as discussed in additional detail below with reference to the appended drawings. This embodiment of the resonant step-down DC-DC power converter further comprises:
- 15 a rectifying element, such as a diode or or a controlled semiconductor switch such as a MOSFET, configured to: in the first case conduct current from the positive input terminal to the second negative electrode of the output capacitor in the second case, conduct current from the negative input terminal to the first positive electrode of the output capacitor; and
- 20 a mode selecting semiconductor switch configured to selectively break and close an electrical connection between the negative input terminal and the second negative electrode of the output capacitor, such that: in a first mode of the resonant step-down DC-DC power converter, establishing the
- series connection of the output capacitor and the input capacitor through the rectify ing element and in a second mode of the resonant step-down DC-DC power converter, opening or breaking the series coupling of the output capacitor and the input capacitor.

The skilled person will appreciate that each of the present resonant step-down DC-

30 DC power converters may be constructed by conversion of an isolated resonant DC-DC power converter with a corresponding topology as described in additional detail below with reference to FIGS. 2A) - 2B) FIGS. 3A) - 3B) and FIGS. 4A) - 4B). Hence, a second aspect of the invention relates to a method of converting a resonant DC-DC power converter to a non-isolated/uninsulated resonant step-down DC-

10

DC power converter possessing higher power conversion efficiency or smaller power loss. The method comprising steps of:

a) providing a primary side circuit and a secondary side circuit of the isolated DC-DC power converter,

5 b) optionally, coupling an input capacitor between a positive and a negative input terminal of the primary side circuit,

c) optionally, coupling a positive electrode of an output capacitor to a positive output terminal of the secondary side circuit and coupling a negative electrode of the output capacitor to a negative output terminal of the secondary side circuit,

- d) providing electrical, e.g. electromagnetic, coupling of the primary side circuit and the secondary side circuit through a galvanic isolation barrier,
 e) providing a resonant network configured for alternatingly being charged from an input voltage of the converter and discharged to the output capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce a
- converter output voltage,
 f) connecting, in a first case, an electrical short-circuit across the galvanic isolation barrier from the negative output terminal of the secondary side circuit to the positive input terminal of the primary side circuit or connecting, in a second case, the positive output terminal of the secondary side circuit to the negative input terminal of the
- 20 primary side circuit thereby establishing in both the first case and the second case a series coupling of the output capacitor and the input capacitor, g) coupling, in a first case, a power converter load between the positive terminal of the secondary side circuit and the positive input terminal or coupling, in the second case, the power converter load between the negative terminal of the secondary side
- 25 circuit and the negative input terminal of the primary side circuit.

As mentioned above, the coupling of the input capacitor under step b) and the coupling of the output capacitor under step c) are both optional because one or both of the input and output capacitors may be formed exclusively by parasitic capacitances

30 associated with the primary side circuit and the secondary side circuit, respectively.

The improved power conversion efficiency of the present resonant step-down DC-DC power converters is achieved because a smaller or larger amount of the output power delivered to the converter load may be transferred directly from the input volt-

PCT/EP2015/051007

11

age or power source and input capacitor of the input side circuit to the output capacitor of the secondary side circuit due to the series connection of the input and output capacitors between the input voltage or power source as explained above.
Hence, a smaller amount of the output power has to be transferred through the resonant network and isolation barrier leading to lower power losses in the active and/or passive components thereof.

The method may comprise a further step of:

h) in the first case, electrically connecting an input voltage source between the

10 negative input terminal of the primary side circuit and the positive output terminal of the secondary side circuit, or

i) in the second case, electrically connecting an input voltage source between the positive input terminal of the primary side circuit and the negative output terminal of the secondary side circuit.

15

5

The conversion of the resonant DC-DC power converter to the non-isolated resonant step-down DC-DC power converter may comprise certain further steps to add the previously discussed advantageous mode switching feature of the converted DC-DC power converter, i.e. the non-isolated resonant step-down DC-DC power

20 converter. According to this embodiment of the conversion methodology the latter comprises further steps of:

j) inserting a rectifying element in the electrical short-circuit connection,k) inserting a mode selecting semiconductor switch, in the first case, between the positive input terminal and the first positive electrode of the output capacitor and in

25 second case between the negative input terminal and the second negative electrode of the output capacitor.

The skilled person will appreciate that each of the present resonant step-down DC-DC power converters may be constructed by conversion of a conventional or prior
art isolated DC-DC power converter possessing a corresponding topology. The skilled person will appreciate that the conventional or prior art isolated DC-DC power converter may be step-up or boost converter delivering a higher output voltage than input voltage despite that the converted resonant step-down DC-DC power converterer in accordance with the present invention delivers a lower output voltage than in-

12

put voltage due to the series connection of the input capacitor and output capacitor across the input voltage source.

A third aspect of the invention relates to a resonant step-down DC-DC power con-

- 5 verter assembly comprising:
 a resonant step-down DC-DC power converter according any of the abovedescribed embodiments thereof,
 a printed circuit board having at least the resonant network integrated thereon,
 wherein the galvanic isolation barrier comprises a pair of magnetically coupled in-
- 10 ductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the secondary side circuit; wherein the first and second inductors are formed by first and second electrical trace patterns, respectively, of the printed circuit board. The pair of magnetically coupled
- 15 inductors may be coupled to each other without any magnetic permeable core material since the latter may be difficult to integrate on a printed circuit board in an efficient manner. The resonant step-down DC-DC power converter according to the latter embodiment of the invention preferably comprises a resonant converter such as class E, class DE or SEPIC converter in accordance with the present invention. The high frequency operation at or above 20 MHz makes the inductances of the first and second inductors sufficiently small to allow these to be integrally formed in the
 - conductor pattern of the printed circuit board. Furthermore, the high switching frequency of the resonant VHF step-down DC-DC power converter as set by the frequency of the switch control signal provides a high magnetic coupling between the first and second inductors despite lack of the magnetic permeable core material.
- The high switching frequency of the resonant power converter provides a high magnetic coupling, as discussed above, between the pair of magnetically coupled inductors despite the lack of the magnetic permeable core material. The high magnetic coupling reduces the otherwise significant energy losses that would be incurred in
- the pair of magnetically coupled inductors of the galvanic isolation barrier if the resonant DC-DC power converter was operating at ordinary switching frequencies.
 A fourth aspect of the invention relates to a LED light assembly, a charger assembly or a flat-screen display assembly, comprising:

a resonant step-down DC-DC power converter according any of the above-

PCT/EP2015/051007

13

described embodiments thereof mounted on a printed circuit board of the assembly, an AC mains voltage input connected to an input of a mains rectifier, an output of the mains rectifier connected, in the first case, between the first positive electrode of the output capacitor and the negative input terminal of the primary side

- 5 circuit or connected, in the second case, between the negative input terminal of the primary side circuit and the negative electrode of the output capacitor to directly supply a rectified mains voltage to the resonant step-down DC-DC power converter in both cases.
- 10 The LED light assembly may be mounted in housing of a LED lamp or tube. The converter load may be formed by a plurality of LEDs coupled to the converter output voltage. The AC mains voltage may lie between 110 V and 240 V depending on the electricity system. The LED lamp application is helpful to illustrate the advantages of the present resonant step-down DC-DC power converter. In one exemplary embod-
- 15 iment, the plurality of LEDs may need a DC voltage of about 60 V and consume 10 W. A conventional isolated step-down DC-DC power converter in the LED light assembly connected directly to U.S. mains of 110 V would be required to handle a peak rectified input voltage of about 170 V on a primary side circuit. However, the primary side circuit of the present resonant step-down DC-DC power converter is
- 20 only required to handle 170 V minus 60 V (i.e. the rectified input voltage minus the output voltage of converter) which is about 110 V. This reduction of DC voltage across the primary side circuit means that smaller and cheaper components such as semiconductor switches can be used. Furthermore, the conventional isolated step-down DC-DC power converter is required to transmit the required 10 W of power to
- 25 the LEDs of the lamp while the present resonant step-down DC-DC power converter only needs to transmit or supply 110/200 * 10 W = 5.5 W. The residual 4.5 W of output power to the LED load is supplied directly from the 110 V AC mains source to the output of the converter through the mains rectifier and input capacitor.
- 30 The skilled person will appreciate that the electrical short-circuit connection across the galvanic isolation barrier provides noteworthy benefits to the present resonant step-down DC-DC power converter for example a reduction of the input voltage across the primary side circuit. These benefits are due to a marked reduction of peak AC voltage across the semiconductor switch or switches of the resonant power

converters. This peak AC voltage is often about 3 times larger than the input voltage of the primary side circuit due to the resonant waveforms inside the resonant network of the resonant power converter. Furthermore, since resonant power converters rely on a resonant current, securing ZVS/ZCS operation of the semiconductor

- 5 switch or switches, which scales linearly with the input voltage, the resistive power losses in components of the resonant network incurred by this resonating current scale quadratically with the resonant current. Hence, the total power loss in the resonant network scales quadratically with the input voltage. The advantages, in particular the increase of power conversion efficiency, derived from the reduction of the
- 10 input voltage to the primary side circuit provided by resonant step-down DC-DC power converters in accordance with the present invention are therefore particularly pronounced for resonant converter topologies.

BRIEF DESCRIPTION OF THE DRAWINGS

- Preferred embodiments of the invention will be described in more detail in connection with the appended drawings, in which:
 FIGS. 1A) and 1B) are simplified electrical circuit diagrams illustrating a resonant step-down DC-DC power converter in accordance with a first embodiment of the present the invention,
 EV. 1C) is a simplified electrical circuit diagram of a resonant step. DC DC
- 20 FIG. 1C) is a simplified electrical circuit diagram of a resonant step-down DC-DC power converter in accordance with a second embodiment of the present the invention,

FIG. 1D) is a simplified electrical circuit diagram of a resonant step-down DC-DC power converter in accordance with a 3rd embodiment of the present the invention,

- FIG. 1E) is a simplified electrical circuit diagram of a resonant step-down DC-DC power converter in accordance with a 4th embodiment of the present the invention, FIG. 1F) is simplified electrical circuit diagram of a first converter core that may be utilized in resonant step-down DC-DC power converter in accordance the present the invention,
- 30 FIG. 1G) is simplified electrical circuit diagram of a second converter core that may be utilized in resonant step-down DC-DC power converter in accordance the present the invention,

FIG. 2A) is an electrical circuit diagram of a prior art isolated class E resonant DC-DC converter comprising a series resonant circuit, FIG. 2B) is an electrical circuit diagram of a class E resonant step-down resonant DC-DC power converter comprising a series resonant circuit in accordance with a 7th embodiment of the invention,

FIG. 3A) is an electrical circuit diagram of a prior art transformer coupled isolated

5 class E resonant DC-DC converter comprising a series resonant circuit, FIG. 3B) is an electrical circuit diagram of a transformer coupled class E resonant step-down DC-DC power converter comprising a series resonant circuit in accordance with an 8th embodiment of the invention,

FIG. 4A) is an electrical circuit diagram of a prior art isolated SEPIC converter; and

10 FIG. 4B) is an electrical circuit diagram of a step-down SEPIC resonant DC-DC converter in accordance with a 9th embodiment of the invention.

DETAILED DESCRIPTION OF PREFERRED EMBODIMENTS

FIGS. 1A), 1B) and 1C) are simplified electrical circuit diagrams 100a, 100b, 100c

- 15 illustrating basic operational features of two different embodiments of resonant stepdown DC-DC power converters in accordance with the present invention. The first embodiment is illustrated on FIGS. 1A) and 1B) while the second embodiment is illustrated on FIG. 1C).
- FIG. 1A) shows a resonant step-down DC-DC power converter 100b comprising a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 107b. The primary side circuit comprises a positive input terminal 102b and a negative input terminal 101b for receipt of a DC or AC input voltage V_{in} from an input voltage or power source (not shown). An input capacitor C_{in} is electrically
- 25 connected between the positive input terminal 102b and a negative input terminal 101b to form an energy reservoir for the input power or input voltage source. The primary side circuit additionally comprises an input side 106b of a resonant network arranged in front of a galvanic isolation barrier 107. The secondary side circuit comprises an output capacitor C_{out} having a first electrode electrically connected to the
- 30 converter output voltage V_{out} at output terminal 104b. A second electrode of the output capacitor C_{out}, situated at a lower voltage potential than the first electrode, is connected to the positive input terminal 102b on the input side circuit via an electrical short-circuit connection or wire 109b extending across the galvanic isolation barrier 107b. The output capacitor C_{out} and input capacitor C_{in} are connected in series

PCT/EP2015/051007

16

or cascade between the DC or AC input voltage V_{in} , i.e. between positive and negative terminals of the input voltage or power source coupled to the output terminal 104b and the negative input terminal 101b, respectively. The skilled person will understand that the presence of the electrical short-circuit connection 109b provides

- 5 the conversion from an ordinary resonant isolated DC-DC power converter to the present resonant step-down DC-DC power converter which lacks galvanic isolation between the input side circuit and the output side circuit. In this conversion process, the input voltage port of the original resonant isolated DC-DC power converter between the positive and negative terminals 102b, 101b (i.e. across C_{in}) is altered such
- 10 that the input voltage port, for receipt of the DC or AC input voltage V_{in}, of the present, converted, resonant step-down DC-DC power converter 100b is arranged either between the output terminal 104b and the negative input terminal 101b in the first case or in the second case between the positive input terminal 102c and the negative electrode 103c of the output capacitor C_{out} as illustrated on FIG. 1A) and
- 15 FIG. 1C), respectively. The resonant step-down DC-DC power converter embodiments 100b, 100c may comprise permanently connected electrical short-circuit connections or wires 109b, 109c to provide a single mode of operation of the resonant DC-DC power converter in question with fixed voltage step-down functionality. In alternative embodiments of the present DC-DC power converters, the coupling or
- 20 interconnection of the primary and secondary circuits of the power converters 100b, 100c through the electrical short-circuit connections 109b, 109c may be selectable or programmable such that each of the power converters 100b, 100c may possess two distinct and selectable modes of operation as discussed below in further detail with reference to FIGS. 1D) and 1E). The skilled person will understand that the
- 25 input and output capacitors C_{in} and C_{out} of the resonant step-down DC-DC power converter embodiments 100b, 100c may be formed exclusively by parasitic capacitances associated with the primary side circuit and the secondary side circuit, respectively.
- 30 An electrical load R_L of the step-down DC-DC converter 100b is coupled between the output terminal 104b and the positive input terminal 102b, which is across the output capacitor, such that these terminals form an output port of the DC-DC converter 100b. The primary side circuit comprises the previously discussed input side 106b of the resonant network of the step-down DC-DC converter 100b and the sec-

PCT/EP2015/051007

17

ondary side circuit comprises an output side 108b of the resonant network. The skilled person will appreciate that the resonant network may include numerous circuit topologies depending on the particular type of DC-DC converter in question. The resonant network preferably comprises at least one inductor for energy storage

- 5 and release, but may alternatively exclusively comprise capacitors for energy storage. Generally, the resonant network is configured for alternatingly being charged from the input voltage V_{in} and discharged to the output capacitor C_{out} through the isolation barrier 107b in accordance with a switch control signal to produce the converter output voltage V_{out}. The primary side circuit preferably comprises at least one
- 10 semiconductor switch, for example a MOSFET, which is switched between on-states and off-states by the switch control signal such that the input voltage is modulated in accordance with a switch control signal. The frequency of the switch control signal of the semiconductor switch arrangement which excites the resonant network may be at or above 30 MHz to form a so-called VHF type of step-down DC-DC power con-
- 15 verter. The switch control signal may comprise a PWM modulated control signal. The primary side circuit may comprise an inductor that is charged with energy during an on-state of the least one semiconductor switch from the input capacitor C_{in} and/or the DC or AC input voltage V_{in}. The inductor of the primary side circuit may subsequently be discharged through the output side 108b of the resonant network and the
- 20 output capacitor C_{out} in an off-state of the least one semiconductor switch. The secondary side circuit may comprise a diode based rectifier or a synchronous rectifier in front of the output capacitor to produce the converter output voltage V_{out} as a DC output voltage.
- 25 While the electrical short-circuit connection or wire 109b eliminates the galvanic isolation between the primary and secondary side circuits of the resonant step-down DC-DC converter 100b by interconnecting the second electrode of the output capacitor C_{out} and the positive input terminal 102b, it provides numerous new benefits to the DC-DC converter as a whole as illustrated with reference to FIG 1B). The series
- 30 connection of the output and input capacitors C_{out}, C_{in} means that the primary side circuit only needs to withstand the converter output voltage V_{out} minus the input voltage across C_{in} instead of the entire DC or AC input voltage V_{in} delivered by the input voltage or power source which is the situation in the prior art isolated DC-DC converter topology. The reduced voltage across the primary side circuit reduces the

PCT/EP2015/051007

18

required maximum voltage rating of active and passive components therein leading to physically smaller and/or less costly active and passive components for example inductors, capacitors (including C_{in}), transistors and diodes etc. In addition, the life span of the latter components may increase by the smaller voltage stress. In the

5 input section 106b and output section 108b, the smaller amount of power to be transferred through the DC-DC converter 100b for supplying a given amount of power to the load R_L of the converter leads to reduced power requirements for active semiconductors switches allowing less costly and physically smaller semiconductors to be applied.

10

These beneficial reductions of the amount of power transferred through the resonant network 106b, 107b, 108b are achieved because the residual fraction of the output power supplied to the load R_L is transferred directly from the DC or AC input voltage source V_{in} to the output capacitor C_{out} during charging of the input capacitor C_{in} This

- 15 power transfer mechanism is illustrated on FIG. 1B) by the first output current path $I_{convert}$ which shows how secondary side current charges the output capacitor C_{out} and delivers output power to the load R_L of the converter. The secondary side current comprises an AC component 113a and a DC component 113b such that the latter is dissipated in the load R_L . Hence, the secondary side current delivers output
- 20 power passing from the source of the DC or AC input voltage V_{in} and through the resonant network to the output side circuit in a conventional manner. However, the present DC-DC converter 100b also comprises a second output current path delivering current or power directly from the DC or AC input voltage source to the output capacitor C_{out} and the load R_L without passing through the power converter 100b.
- 25 This direct current comprises an AC current component delivered to the output capacitor C_{out} as illustrated by AC input current path 111a and a DC current component delivered to the load R_L as illustrated by DC input current path 111b. The AC current component 111a and the DC current component 111b of the input current path passes through the short-circuit connection 109b and through the positive and nega-
- 30 tive inputs of the primary side circuit 106b such that the DC current component 111b of the direct current remains unprocessed by the converter. Hence, this DC current component is supplied directly to the load R_L without any noticeable power loss.

The skilled person will appreciate that a practical electrical short circuit connection

PCT/EP2015/051007

19

109b will possess a certain DC resistance and an upper limit for this DC resistance will vary depending on input/output voltage and/or current requirements of the converter 100b. The electrical short-circuit connection may possess a DC resistance of less than 1 k Ω , even more preferably less than 100 Ω , such as less than 10 Ω . In

- 5 other embodiments, the electrical short circuit connection 109b may have a unidirectional resistance such that the DC resistance only falls below the above-mentioned upper limits in one direction and exhibits a much larger DC resistance in the opposite direction, i.e. a diode characteristic.
- 10 In the alternative embodiment of the resonant step-down DC-DC converter topology illustrated on FIG. 1), the electrical short-circuit connection or wire 109c extending across the isolation barrier 107c is connecting the first positive electrode of the output capacitor C_{out} to the negative input terminal 101c of the primary side circuit. Thereby, a series coupling of the output capacitor C_{out} and the input capacitor C_{in}
- 15 from the input voltage V_{in} at the positive input terminal 102c to the negative electrode 103c of the output capacitor C_{out} is established. The negative electrode 101c of the output capacitor C_{out} is at a lower electric potential than the negative input terminal 101c. In this manner, the DC or AC input voltage V_{in} to the power converter 100c is once again applied across the series connected input and output capacitors
- 20 C_{in} and C_{out}. The converter load R_L is coupled between the output terminal 104c and the terminal 103c which also is across the output capacitor. Otherwise, circuit functions, electrical component characteristics and component values of this second embodiment of resonant power converter 100c may be identical to those of the first embodiment of the resonant power converter 100b.

25

FIG. 1D) is a simplified electrical circuit diagram of a resonant step-down DC-DC power converter 100d in accordance with a third embodiment of the present the invention. A converter core 105d of the resonant power converter 100d may be identical to the core 105b of the resonant power converter 100c discussed above in con-

30 nection with FIG. 1A) and 1B). Hence, corresponding features of these different power converter embodiments 100b, 100d have been provided with corresponding reference numerals/symbols to assist comparison. The present resonant power converter embodiment 100d comprises a mode selecting controllable semiconductor switch SW1 operating in conjunction with a rectifying element 111e such as a diode

20

or an active semiconductor diode such as a controllable semiconductor switch. The operation of the resonant step-down DC-DC power converter 100d corresponds to the operation of the resonant step-down DC-DC power converter 100e discussed in detail below.

5

FIG. 1E) shows a simplified electrical circuit diagram of a resonant step-down DC-DC power converter 100e in accordance with a 4th embodiment of the present the invention. A converter core 105e of the resonant power converter 100e may be identical to the core 105 of the resonant power converter 100c discussed above in con-

- 10 nection with FIG. 1C). Hence, corresponding features of these different power converter embodiments 100c, 100e have been provided with corresponding reference numerals/symbols to assist comparison. The present resonant power converter embodiment 100e comprises a mode selecting controllable semiconductor switch SW1 operating in conjunction with a rectifying element 111e such as a diode or an active
- 15 semiconductor diode such as a controllable semiconductor switch. The rectifying element 111e is inserted in a short-circuit connection or wire 109e which connects a negative input terminal 109e of the power converter to a first positive electrode of the output capacitor C_{out}. The latter short-circuit connection corresponds to the short circuit connection 109c of the previously discussed resonant power converter 100c.
- 20 Hence, when the rectifying element 111e is forward biased the short-circuit wire 109e effectively places the output capacitor C_{out} and input capacitor C_{in} is series, albeit possibly with a minor diode voltage drop, between the input voltage V_{in} terminals 104e, 103e of the resonant power converter 100e. On the other hand, when the rectifying element 111e is reversely biased the series coupling of C_{out} and C_{in}
- 25 through short-circuit wire 109e is broken or disconnected to break the electrical connection between the primary side circuit and the secondary side circuit of the resonant power converter 100e.
- The mode selecting controllable semiconductor switch SW1 is coupled between the
 negative input terminal 101e and the second negative electrode of the output capacitor 103e and configured to selectively break and close the electrical connection between these terminals 101e, 103e. The switch SW1 may comprise one or more
 BJT(s), FET(s) MOSFET(s) or IGBT(s) such as a Gallium Nitride (GaN) or Silicon
 Carbide (SiC) transistor. SW1 may be switched between a conducting or ON state

PCT/EP2015/051007

21

and non-conducting or OFF state to connect and break, respectively, the electrical connection between terminals 101e, 103e by a suitable control voltage applied on a gate or base terminal of the switch SW1. A mode controlling circuit (not shown) integrated with or coupled to the present resonant power converter 100e may supply

5 this control voltage to SW1.

When the mode selecting controllable semiconductor switch SW1 is placed in the non-conducting or OFF state (as illustrated) the negative terminals of the C_{out} and C_{in} capacitors are disconnected and the rectifying element 111e forward biased.

- 10 Hence, the input and output capacitors C_{out} and C_{in} are series connected through the rectifying element 111e. The rectifying element 111e is forward biased because the DC or AC input voltage V_{in} is larger than the output voltage V_{out} allowing the previously discussed input current of the power converter to flow directly through C_{in} and into C_{out} to charge the output capacitor. Hence, the resonant power converter 100e
- 15 is placed in a first mode of operation where it largely functions as the resonant power converter 100c of FIG. 1C) discussed above, i.e. as a resonant step-down nonisolated DC-DC power converter. By placing the controllable semiconductor switch SW1 in the conducting or ON state, the present resonant power converter embodiment 100e is switched to the second mode of operation. In the second state of op-
- eration, the rectifying element 111e is reversely biased and hence non-conducting due to the electrical connection between the negative capacitor terminals 101e, 103e. Hence, the above-discussed series connection (through the rectifying element 111e) of the input and output capacitors C_{out} and C_{in} in the first mode is broken or opened. In this second mode of operation of the the resonant power converter 100e,
- 25 it largely functions as an ordinary resonant DC-DC power converter albeit still without any galvanic isolation between the primary and secondary side circuits because of the presence of the rectifying element 111e.
- The mode switching feature, i.e. from the first mode to the second mode or vice ver-30 sa, of the present resonant power converter 100e is accompanied with several advantages. The mode switching feature may be used to dynamically switch the present resonant power converter 100e between the first and second modes during operation of the converter 100e by appropriate control of the control terminal of SW1.The dynamic mode switching feature increases the input voltage range of the
PCT/EP2015/051007

22

resonant power converter 100e. To illustrate these advantages consider an ordinary resonant DC-DC power converter designed for a DC input voltage range of 10 - 20 V and a DC output voltage range of 10 V. If this ordinary DC-DC power converter is converted or configured as the present resonant step-down DC-DC power convert-

- 5 er, the DC input voltage range may be increased to 10 30 V by utilizing both the first mode and second mode of the resonant step-down DC-DC power converter during operation of the converter. Hence, the resonant step-down DC-DC power converter is switched dynamically between the first and second modes of operation in accordance with the input voltage waveform. The increase of DC input voltage
- 10 range achieved by the dynamic mode switching feature is particular advantageous because resonant power converters generally suffer from a restricted or narrow DC input voltage range compared to non-resonant DC-DC power converters. Another advantage of the dynamic mode switching feature is an improved power factor (PF) of the resonant power converter 100e. The power factor of a switched mode power
- 15 converter or supply is an important performance metric in numerous applications such as LED lamps where regulatory requirements for minimum power factor, such as larger than 0.9, may exist. The improved PF is achieved by the dynamic mode switching feature because this feature allows the converter to track AC variations of the input voltage to the converter such as the AC variations in the waveform of a
- 20 rectified 50/60 Hz mains voltage.

25

The flow of input current in the resonant step-down DC-DC power converter 100e in the first and second modes of operation is illustrated by a first DC input current path 111b1 and a second DC input current path 111b2. When the first mode of operation of the converter 100e is selected, i.e. the step-down functionality, the input current flows through the first DC input current path 111b1 where DC input current flows through the diode 111e and through the converter load R_L . When the second mode

- of operation of the converter 100e is selected, i.e. SW1 is conducting and ordinary conversion functionality, the input current flows through the second DC input current
- 30 path 111b2 where the diode 111e is blocking and the DC input current flows through SW1 to the negative rail or terminal 103d of the converter without passing through the converter load.

PCT/EP2015/051007

23

FIG. 1F) is a simplified electrical circuit diagram of a first converter core 105f that may be utilized as converter core 105b, 105c, 105d, 105e of each of the step-down DC-DC power converter embodiments 100b, 100c, 100d, 100e, illustrated on FIGS.
1A), 1B), 1C), 1D) and 1E), respectively. The first converter core 105f comprises a

- 5 plurality of separate resonant DC-DC power converter cores 110f. Each of the separate resonant DC-DC power converter cores 110f comprises an input side 111 of a resonant network coupled to an output side 113 of the resonant network through a galvanic isolation barrier 107f. The input sides 111 of the resonant DC-DC power converter cores 110e may be connected in parallel or series. The output sides 113
- 10 of the resonant DC-DC power converter cores 110f may likewise be connected in parallel or series. The parallelization of the plurality of input sides 111 and/or the parallelization of the one or more output sides 113 increases the power rating of a step-up DC-DC power converter utilizing the first converter core 105f. The skilled person will understand that each of the separate resonant DC-DC power converter
- 15 cores 110f may comprise one of the prior art resonant DC-DC power converter cores discussed below with reference to FIGS. 2, 3 and 4.

FIG. 1G) is simplified electrical circuit diagram of a second converter core 105g that may be utilized as converter core 105b, 105c, 105d, 105e of each of the step-down

- DC-DC power converter embodiments 100b, 100c, 100d, 100e illustrated on FIGS.
 1A), 1B), 1C), 1D) and 1E), respectively. The second converter core 105g comprises a plurality of separate resonant power inverters 114g. Each of the separate resonant power inverter cores 114g comprises an input side 115 of a resonant network coupled to one or more rectifier(s) 117 of the resonant DC-DC power converter core
- 25 105g through a galvanic isolation barrier 107g. The separate resonant power inverter cores 114g may be connected in parallel or series. Likewise, the respective output side of the one or more rectifier(s) 117 may also be connected in series or parallel. However, galvanic isolation may be inserted between the one or more rectifier(s) 117 if these are coupled in series.

30

FIG. 2A) shows an electrical circuit diagram of a prior art isolated class E resonant DC-DC converter 200 comprising a series resonant network or circuit including inductor L_2 and capacitor C_1 . The prior art class E resonant converter comprises a primary side circuit and a secondary side circuit connected through a galvanic isola-

PCT/EP2015/051007

24

tion barrier 207. The primary side circuit comprises a positive input terminal 202 and a negative input terminal 201 for receipt of a DC or AC input voltage V_{in} from a voltage or power source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 202b and a negative input terminal 201 to form

- 5 an energy reservoir for the voltage source. A converter core 205 comprises a resonant network which includes first and second series connected inductors L₁ and L₂ and a semiconductor switch arrangement comprising a MOSFET switch S (or another suitable type of semiconductor switch) with a drain terminal connected to a midpoint node between the L₁ and L₂. The primary side circuit is arranged in front of
- 10 an isolation barrier 207 formed by coupling capacitors C₁ and C₂. The secondary side circuit comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage V_{out} at output terminal 204. A second negative electrode of the output capacitor C_{out} is coupled to a negative terminal 203 of the converter output voltage. A load of the isolated class E resonant DC-DC con-
- 15 verter 200 is schematically illustrated by load resistor R_L and coupled between the positive and negative output terminals 204, 203.

FIG. 2B) is an electrical circuit diagram of a class E resonant step-down DC-DC power converter 200c comprising a series resonant circuit in accordance with a 7th

- 20 embodiment of the invention. The class E resonant step-down DC-DC power converter 200b may be obtained by conversion of the above-mentioned prior art isolated class E resonant DC-DC converter 200 by inserting or adding an electrical short circuit connection 209c extending across a galvanic isolation barrier 207c of the converter 200c. The galvanic isolation barrier 207c comprises series capacitors C1
- and C₂. The electrical short circuit connection 209c electrically connects the negative input terminal 201c and the first positive electrode 204c of the output capacitor C_{out}. The first positive electrode 204c is also connected to the converter output voltage V_{out}. As discussed in connection with FIG. 1C), the electrical short-circuit connection or wire 209c effectively places the output capacitor C_{out} and input capacitor C_{in} in
- 30 series or cascade across the DC or AC input voltage V_{in} . Hence, the converter load, schematically illustrated by the load resistor R_L , is coupled in parallel across the output capacitor C_{out} . The skilled person will understand that the series capacitor C_2 of the galvanic isolation barrier 207c prevents DC current from flowing from the second negative electrode 203b of the output capacitor C_{out} and back to the negative input

PCT/EP2015/051007

25

terminal 201c. In this manner, the DC current is directed or forced through the electrical short circuit connection 209b and back through the input capacitor C_{in} . In this manner, despite being electrically by-passed by the conversion, the isolation barrier 207c is important for the operation of the present class E resonant step-down DC-

5 DC power converter 200c as otherwise would nodes 201c and 203c and 204c be directly electrically connected causing a short circuit at the converter input.

The series resonant network of the converter core 205c of the converter 200c may, in addition to the above-discussed first and second series connected inductors L_1

- 10 and L₂, comprise a capacitor (C_s) arranged across drain and source terminals of the MOSFET switch S to increase a resonant current and/or adjust/fine-tune a resonance frequency of the power converter 200c. Likewise, a yet further capacitor C_D may be arranged across the rectifying diode D to adjust a duty cycle of the secondary part of the power converter 200c, i.e. the class E rectifier. During operation of
- 15 the converter 200c, the series resonant network is excited by the MOSFET switch S such that the series resonant network is alternatingly charged from the DC or AC input voltage V_{in} and discharged to the output capacitor C_{out} through the galvanic isolation barrier 207c and through a rectification circuit comprising inductance L₃ and diode D.

20

FIG. 3A) is an electrical circuit diagram of a prior art transformer isolated class E resonant DC-DC converter 300 comprising a series resonant circuit. A converter core 305 of the converter 300 comprises a series resonant circuit which at least inductors L_2 , L_3 and capacitors C_8 and C_1 . The prior art DC-DC converter 300 com-

- 25 prises a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier 307 provided by the transformer 308. The primary side circuit comprises a positive input terminal 302 and a negative input terminal 301 for receipt of a DC or AC input voltage V_{in} from a voltage or power source (not shown). An input capacitor C_{in} is electrically connected between the positive input terminal 302
- 30 and a negative input terminal 301 to form an energy reservoir for the input voltage source. The primary side circuit additionally comprises a portion of a series resonant network or circuit which at least comprises inductors L₂ and capacitors C_s and C₁. A first inductor L₁ has a first end coupled to the positive input terminal 302 and a second end to a drain terminal of a MOSFET switch S which forms a switch arrange-

PCT/EP2015/051007

26

ment of the present power converter 300. A secondary side circuit of the power converter 300 comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage V_{out} at output terminal 304. A second negative electrode of the output capacitor C_{out} is coupled to a negative terminal 303

- 5 of the converter output voltage. An electrical or power converter load is schematically illustrated by load resistor R_L and coupled between the positive and negative output terminals 304, 303 of the prior art DC-DC converter 300. The secondary side circuit furthermore comprises the third inductor L_3 connected across a secondary transformer winding of the above-mentioned transformer 308. The secondary trans-
- 10 former winding has a first end coupled to a cathode of rectifying diode D and a second end coupled to the positive electrode of the output capacitor C_{out}. The rectifying diode D rectifies AC current generated by the secondary transformer winding and generates a DC voltage as the converter output voltage between the positive and negative output terminals 304, 303. An electrical or power converter load is sche-
- 15 matically illustrated by load resistor R_L coupled between the positive and negative output terminals 304, 303.

FIG. 3B) is an electrical circuit diagram of a transformer coupled class E resonant step-down DC-DC power converter 300c in accordance with an 8th embodiment of

- 20 the invention. The step-down DC-DC power converter 300c may be obtained by conversion of the above-mentioned prior art resonant isolated DC-DC converter 300 by inserting or adding an electrical short circuit connection 309c extending across a galvanic isolation barrier formed by the transformer 308b. The transformer 308b comprises magnetically coupled primary and secondary transformer windings with
- 25 the polarity inversion indicated by the black dots at the top of the windings. A converter core 305c of the converter 300 comprises a series resonant network or circuit which at least comprises inductors L₂, L₃ and capacitors C_s and C₁. During operation of the converter 300c, the series resonant network is excited by the MOSFET switch S such that the series resonant network is alternatingly charged from the DC or AC
- 30 input voltage V_{in} and discharged to the output capacitor C_{out} through the galvanic isolation barrier 307c and through a rectification circuit comprising diode D.

PCT/EP2015/051007

27

The electrical short circuit connection 309c connects the negative input terminal 302b of the primary side circuit and the first positive electrode 304c of the output capacitor C_{out} where the latter electrode also supplies the converter output voltage. As discussed in connection with FIGS. 1C), the electrical short-circuit connection or

- 5 wire 309c effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade across the positive and negative connections to the DC or AC input voltage V_{in} . Hence, the converter load, schematically illustrated by the load resistor R_L , is coupled between the converter output voltage across the positive and negative electrodes of the output capacitor C_{out} . The skilled person will understand that the
- 10 transformer coupling prevents DC current from flowing from the second negative electrode 303c of the output capacitor C_{out} and back to the negative input terminal 301c of the primary side circuit.
- FIG. 4A) is an electrical circuit diagram of a prior art isolated single-ended primaryinductor converter (SEPIC) 400. The prior art SEPIC 400 comprises a primary side circuit and a secondary side circuit connected through a galvanic isolation barrier
 407. The primary side circuit comprises a positive input terminal 402 and a negative input terminal 401 for receipt of a DC or AC input voltage V_{in} from a voltage or power source (not shown). An input capacitor C_{in} is electrically connected between the pos-
- 20 itive input terminal 402 and a negative input terminal 401 to form an energy reservoir for the input voltage source. A converter core 405 of the converter 400 comprises a series resonant circuit which includes a first inductor L₁ having first node coupled to the DC or AC input voltage V_{in} and a second node coupled to a drain terminal of a MOSFET switch S. A source terminal of a MOSFET switch S of a switch arrange-
- 25 ment is coupled to the negative input terminal 401. The prior art SEPIC 400 is a resonant type of power converter wherein a resonance frequency of the converter is determined by the resonant network of the converter core 405 comprising the first and second inductors L₁, L₂ and the capacitors C_s and C_D. The primary side circuit is arranged in front of a galvanic isolation barrier 407 within the converter core 405 and
- 30 formed by coupling capacitors C₁ and C₂. The secondary side circuit comprises an output capacitor C_{out} having a first electrode electrically connected to the converter output voltage V_{out} at a positive output terminal 404. A second negative electrode of the output capacitor C_{out} is coupled to a negative terminal 403 of the converter output voltage. A rectifying diode D rectifies AC current generated by a second inductor

28

 L_2 and charges the output capacitor C_{out} such that the converter output voltage V_{out} between the positive and negative output terminals 404, 403 is a DC voltage. A converter load of the SEPIC 400, illustrated by load resistor R_{L_1} is coupled between the positive and negative output terminals 404, 403.

5

FIG. 4B) is an electrical circuit diagram of a SEPIC 400c in accordance with a 9th embodiment of the invention. The SEPIC 400c may be obtained by conversion of the above-mentioned prior art SEPIC 400 by inserting or adding an electrical short circuit connection 409c extending across a galvanic isolation barrier 407c of the

- SEPIC 400c. The galvanic isolation barrier 407c comprises series capacitors C₁ and C₂ which prevent the flow of DC current between the secondary side circuit and the primary side circuit in the isolated SEPIC 400 discussed above. In the present SEPIC 400c, the electrical short circuit connection 409c electrically connects the negative input terminal 401c and the positive electrode 404c of the output capacitor
- 15 C_{out}. As discussed in connection with FIGS. 1A) and 1B), the electrical short-circuit connection or wire 409c effectively places the output capacitor C_{out} and input capacitor C_{in} is series or cascade across the positive and negative connections to the DC or AC input voltage V_{in} such that the galvanic isolation between the primary and secondary side circuits is bypassed or eliminated. The electrical or power converter
- 20 load, schematically illustrated by the load resistor R_L, is coupled between the converter output voltage V_{out} at the output terminal 404c and the negative electrode 403c of the output capacitor. The skilled person will understand that the series capacitor C₂ of the galvanic isolation barrier 407b prevents DC current from flowing from the second negative electrode 403b of the output capacitor C_{out} and directly
- 25 back to the negative input terminal 401b of the primary side circuit. The SEPIC 400c comprises a converter core 405c comprises a resonant network as discussed above comprising first and second inductors L₁, L₂ and capacitors C_S and C_D. During operation of the converter 400c, the series resonant network is excited by the MOSFET switch S such that the series resonant network is alternatingly charged from the DC
- 30 or AC input voltage V_{in} and discharged to the output capacitor C_{out} through the galvanic isolation barrier 407c and through a rectification circuit comprising a rectifying diode D. The capacitor C_s connected or arranged across drain and source terminals of the MOSFET switch S is optional, but may be helpful to increase a resonant current and/or adjust/fine-tune a resonance frequency of the resonant network SEPIC

400c. In some embodiments of the SEPIC 400c the capacitor C_S may be formed exclusively by a parasitic capacitance of the MOSFET S. The optional capacitor C_D may be arranged across the rectifying diode D to adjust a duty cycle of the power converter 400c.

5

CLAIMS

the output capacitor,

5

1. A resonant step-down DC-DC power converter comprising:

a primary side circuit and a secondary side circuit coupled through a galvanic isolation barrier,

the primary side circuit comprising a positive input terminal and a negative input terminal for receipt of an input voltage and an input capacitor coupled between the positive and negative input terminals,

the secondary side circuit comprising an output capacitor chargeable to a converteroutput voltage between a first positive electrode and a second negative electrode of

a resonant network configured for alternatingly being charged from the input voltage and discharged to the output capacitor through the galvanic isolation barrier by a semiconductor switch arrangement in accordance with a switch control signal to

- 15 produce the converter output voltage, wherein a frequency of the switch control signal has a frequency at or above 20 MHz, an electrical short-circuit connection across the galvanic isolation barrier connecting, in a first case, the second negative electrode of the output capacitor to the positive input terminal of the primary side circuit or, in a second case, connecting the second
- 20 positive electrode of the output capacitor to the negative input terminal of the primary side circuit thereby establishing in both the first and second cases a series coupling of the output capacitor and the input capacitor,

a load connection, in the first case, between the first positive electrode of the output capacitor and the positive input terminal or, in the second case, between the second

25 negative electrode of the output capacitor and the negative input terminal.

2. A resonant step-down DC-DC power converter according to claim 1, wherein the galvanic isolation barrier comprises:

a pair of magnetically coupled inductors comprising a first inductor electrically con-

30 nected to the primary side circuit and a second inductor electrically connected to the secondary side circuit.

PCT/EP2015/051007

3. A resonant step-down DC-DC power converter according to claim 2, wherein the first and second inductors are wound around a common magnetically permeable structure to form an isolation transformer.

- 5 4. A resonant step-down DC-DC power converter according to claim 2, wherein the first and second inductors are arranged to be magnetically coupled without a common magnetically permeable structure to form a coreless isolation transformer.
- 5. A resonant step-down DC-DC power converter according to claim 4, wherein thefirst and second inductors comprises first and second embedded coils, respectively,
 - formed in one or more conductive layers of a printed circuit board. 6. A resonant step-down DC-DC power converter according to claim 4 or 5, wherein
- 15 0.25.

7. A resonant step-down DC-DC power converter according to claim 1, wherein the galvanic isolation barrier comprises:

a magnetic coupling factor, k, between the first and second inductors is larger than

a first capacitor coupled in series with the positive input terminal of the primary side

20 circuit and the first positive electrode of the output capacitor; and a second capacitor coupled in series with the negative input terminal of the primary side circuit and the second negative electrode of the output capacitor;

8. A resonant step-down DC-DC power converter according to claim 7, wherein acapacitance of each of the first and second capacitors is smaller than 100 nF.

9. A resonant step-down DC-DC power converter according to any of the preceding claims, wherein a DC resistance of the electrical short-circuit connection is less than 1 k Ω .

30

10. A resonant step-down DC-DC power converter according to any of the preceding claims, wherein the resonant network comprises: first and second series connected inductors connected in series with the positive input voltage terminal, a semiconductor switch having a first switch node connected between a mid-point node between the first and second series connected inductors, a second switch node connected to the negative input terminal of the primary side circuit and a control terminal connected to the switch control terminal; and

5 a rectifying circuit connected between the first and second capacitors of the galvanic isolation barrier and the first positive electrode and the second negative electrode of the output capacitor.

11. A resonant step-down DC-DC power converter according to any of the preced-

10 ing claims, wherein the semiconductor switch arrangement comprises one or more controllable semiconductor switches.

12. A resonant step-down DC-DC power converter according to claim 11, wherein the one or more controllable semiconductor switches are configured for zero-

15 voltage-switching and/or zero-current-switching.

13. A resonant step-down DC-DC power converter according to any of the preceding claims further comprising:

a rectifying element such as a diode configured to:

- 20 in the first case conduct current from the positive input terminal to the second negative electrode of the output capacitor in the second case, conduct current from the negative input terminal to the first positive electrode of the output capacitor; and a mode selecting semiconductor switch configured to, in the first case, selectively break and close an electrical connection between the positive input terminal and the
- 25 first positive electrode of the output capacitor and in second case selectively break and close an electrical connection between the negative input terminal and the second negative electrode of the output capacitor, such that: in a first mode of the resonant step-down DC-DC power converter, establishing the
- series connection of the output capacitor and the input capacitor through the rectify ing element; and
 in a second mode of the resonant step-down DC-DC power converter, opening or

breaking the series coupling of the output capacitor and the input capacitor.

14. A method of converting a resonant isolated DC-DC power converter having a switching frequency at or above 20 MHz to a resonant non-isolated step-down DC-DC power converter possessing higher power conversion efficiency, said method comprising steps of:

5 a) providing a primary side circuit and a secondary side circuit of the isolated DC-DC power converter,

b) optionally, coupling an input capacitor between a positive and a negative input terminal of the primary side circuit,

c) optionally, coupling a positive electrode of an output capacitor to a positive output

terminal of the secondary side circuit and coupling a negative electrode of the output capacitor to a negative output terminal of the secondary side circuit,
d) providing electrical coupling of the primary side circuit and the secondary side circuit through a galvanic isolation barrier,

e) providing a resonant network configured for alternatingly being charged from an

15 input voltage of the converter and discharged to the output capacitor through the galvanic isolation barrier in accordance with a switch control signal to produce a converter output voltage,

f) connecting, in a first case, an electrical short-circuit across the galvanic isolation barrier from the negative output terminal of the secondary side circuit to the positive

20 input terminal of the primary side circuit or connecting, in a second case, the positive output terminal of the secondary side circuit to the negative input terminal of the primary side circuit thereby establishing in both the first case and the second case a series coupling of the output capacitor and the input capacitor,

g) coupling, in a first case, a power converter load between the positive terminal of

25 the secondary side circuit and the positive input terminal or coupling, in the second case, the power converter load between the negative terminal of the secondary side circuit and the negative input terminal of the primary side circuit.

15. A method of converting an isolated DC-DC power converter according to claim

30 14, comprising a further step of:

h) in the first case, electrically connecting an input voltage source between the negative input terminal of the primary side circuit and the positive output terminal of the secondary side circuit, or

i) in the second case, electrically connecting an input voltage source between the positive input terminal of the primary side circuit and the negative output terminal of the secondary side circuit.

5 16. A method of converting an isolated DC-DC power converter according to claim14 or 15, comprising further steps of:

j) inserting a rectifying element in the electrical short-circuit connection,

k) inserting a mode selecting semiconductor switch, in the first case, between the positive input terminal and the first positive electrode of the output capacitor and in

10 second case between the negative input terminal and the second negative electrode of the output capacitor.

17. A step-down DC-DC power converter assembly comprising:

a resonant step-down DC-DC power converter according any of claims 1-13,

15 a printed circuit board having at least the resonant network mounted thereon, a pair of magnetically coupled inductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the secondary side circuit;

wherein the first and second inductors are formed by first and second electrical trace

20 patterns, respectively, of the printed circuit board.

18. A LED light assembly comprising:

a resonant step-down DC-DC power converter according any of claims 1-13 mounted on a printed circuit board of the assembly,

- 25 an AC mains voltage input connected to an input of a mains rectifier, an output of the mains rectifier connected, in the first case, between the first positive electrode of the output capacitor and the negative input terminal or connected, in the second case, between the negative input terminal and the negative electrode of the output capacitor to supply a rectified mains voltage to the resonant step-down DC-
- 30 DC power converter in both cases.

19. A LED lamp comprising a LED light assembly according to claim 18.





FIGS. 1A), 1B)



FIG. 1C)







FIGS. 1D)-1E)







200c —

B)

FIGS. 2A), 2B)



A)



B)

FIGS. 3A), 3B)



A)



B)

FIGS. 4A), 4B)

A.10 M. Madsen, J. A. Pedersen, A. Knott, M. A.E. Andersen: "Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components", IEEE Applied Power Electronics Conference and Exposition, Fort Worth, TX, March 2014. Conference Proceedings p2029-2035.

Self-Oscillating Resonant Gate Drive for Resonant Inverters and Rectifiers Composed Solely of Passive Components

Mickey P. Madsen*, Jeppe A. Pedersen*, Arnold Knott*, Michael A. E. Andersen* *Technical University of Denmark Ørsteds Plads, building 349 2800 Kongens Lyngby Denmark Email: {mpma, jarpe, akn, ma}@elektro.dtu.dk

Abstract— This paper presents a new self-oscillating resonant gate drive composed solely of passive components. The gate drive can be used in various resonant converters and inverters and can be used for both low and high side gate drive. The paper presents examples of how higher order harmonics can be used to improve the performance of the gate drive and how the gate drive can be implemented in a class E inverter, a class DE inverter and in class E inverter with a synchronous class E rectifier. The paper shows practical implementations of all the proposed inverters and converters operating in the Very High Frequency (VHF) range, all showing good results with peak efficiency up to 82% and output regulation from 70% to full load without bursting.

I. INTRODUCTION

In the early 70s the constant strive for small, cheap and efficient power supplies lead to the development of Switch-Mode Power Supplies (SMPS) [1]. As the size of modern power supplies are mainly governed by the passive energy storing elements, which scales inversely with the switching frequency, this strive has lead to constantly increasing switching frequencies ever since. Commercially available converters today switch at frequencies up to several megahertz and can have efficiencies of more than 95% (e.g. [2]).

The reason not to increase the switching frequency further and thereby reaching even higher power densities is the switching losses. For the last two decades (since 1988 [3]) research has been done in order to enable the use of resonant RF amplifiers (inverters) combined with a rectifier for dc/dc converters in order to avoid switching losses. With this type of converters SMPSs with switching frequencies in the Very High Frequency range (VHF, 30-300MHz) have been designed with efficiencies up to approx. 90%, [4], [5].

Several of the benefits and challenges of the increased switching frequency are described in [6], [7]. One of the big challenges is to drive the MOSFET without too high gating losses and so far resonant gate drives have shown to be the best solution for this [8], [9]. Control of these converters is also a challenge due to their resonant behavior and burst mode control has been used to overcome this challenge [10], [11], but it introduces spectral components at the bursting frequency. The compact, low cost converters that can be designed with these resonant converters are very well suited for LED lighting. Price, efficiency and reliability are key parameters in this market and with the presented gate drive the price can be reduced to a minimum without compromising the reliability or efficiency.

In section II of this paper a new type of resonant gate drive and several ways of designing it will be shown. Section III shows how the gate drive can be implemented in a complete circuit, both as low and high side gate drivers and used for synchronous rectification and bidirectional power conversion. Section IV shows results from various prototypes utilizing this gate drive and finally section V concludes the paper.

II. SELF-OSCILLATING GATE DRIVE

For a converter switching in the VHF range, hard gating leads to gating losses which are unacceptably high, at least for the semiconductors available today and for low to medium power levels [12], [13]. Several researchers have therefore used resonant gating to reduce the gating losses and drive the gate in an efficient way [14], [15]. All of these circuits need either a transformer, switching semiconductor, adjustable passives and/or feedback from other nodes in the circuit [16], [17].



Fig. 1. Schematic of the basic self-oscillating gate drive. The gate resistance and the body diode has been left out for simplicity.

All these things adds to both price, size and complexity of the complete design.

This paper presents a new resonant gate drive solely constituted by passives around the main semiconductor. This leads to a robust and very simple and low cost gate drive.

The simplest implementation of the gate drive is shown in Fig. 1, the only thing added is an inductor at the gate of the semiconductor and a dc bias voltage. If the V_D is seen as input and V_G as output, the inductor and the parasitic capacitances composes a high pass filter with a capacitive load.

If this high pass filter is designed to have a gain, $G = V_{GS,pp}/V_{DS,pp}$, and have a phase shift close to 180° at the resonance frequency of the power stage, it will create a sine wave at the gate with a peak to peak voltage swing of $V_{GS,pp}$ and a DC offset equal to V_{Bias} .

This dc offset can be used to control the switching frequency and the duty cycle of the power stage and thereby to regulate the output power. In some situations the parasitic capacitances of the MOSFET will lead to too high or too low gain at the desired frequency and additional capacitors, C_{GDext} and C_{GSext} , can be added to adjust this gain.

In order to improve the turn on speed of the MOSFET, and thereby lower the drain to source resistance, higher order harmonics can be added to the fundamental sine wave leading to a more trapezoidal gate signal. This can be achieved by adding small LC circuits between the gate and drain or source of the MOSFET (see Fig. 2). LC circuits connected to drain will cause the higher harmonics to be in phase with V_{DS} and LCs circuit connected to the source will cause the harmonics to be out of phase with V_{DS} (see Fig. 3).

The number of harmonics to include in a given design will depend on several parameters as price, complexity, efficiency etc. Adding higher order harmonics will in general increase the performance of the converter, but it is important to consider which harmonics to include and the magnitude of those harmonics compared to the fundamental. Fig. 4 shows the fundamental and the 3rd and 5th harmonics in and out of phase with the drain signal. It is clear that it is desirable to have the fundamental out of phase with the drain signal, but for the 3rd and 5th harmonic it depends on the duty cycle and the current waveform. From the figure it can be seen that it is desirable to have the 3rd harmonic out of phase for a duty cycle of 50%, but for a duty cycle of 25% it has to be in phase and will only add to the center part of the conducting period where the current usually is the smallest [18], [19]. The gate signals that can be achieved by adding harmonics are shown in Fig. 5.

III. USE IN RESONANT CIRCUITS

As the resonant gate drive is only relying on the resonance of the power stage and is floating between the drain and source of the MOSFET, it can be used in all resonant circuits. It can for instance be used in the class E inverter [20], [21], a class EF₂ (ϕ_2) inverter [22], [23], a resonant SEPIC [24], [25], a resonant boost converter [26], [27] or a class DE inverter (including high side gate drive) [28], [29], but it can also



Fig. 2. Schematic of the self-oscillating gate drive with the 2nd and 4th harmonic to source and drain.



Fig. 3. Examples of transfer functions from drain to source for different implementations of the resonant gate drive.



Fig. 4. The 1st, 3rd and 5th harmonics in and out of phase with the drain signal.



Fig. 5. Gate signals with higher order harmonics for 25% and 50% duty cycle (*=signal in phase with V_{DS}).

be used for synchronous rectification and bidirectional power flow.

A. In a class E inverter

The simple version of the gate drive implemented in a class E inverter is shown in Fig. 6. As the gate drive relies on the resonance of the inverter the switching frequency will automatically adjust for any tolerances in the passives, L_{IN} , L_R and C_R . Furthermore the dependency on the resonance of the power stage gives and inherent open load protection as an open load situation will remove the resonance of the circuit.

When designing a class E inverter with the proposed gate drive, the design procedure will be first to design the power stage according to the specifications for the inverter, then select a proper MOSFET and finally design the gate drive for the selected MOSFET. An inverter with the specifications in table I will be designed in the following to give an example.

TABLE I Design specification for the class E inverter

$$\begin{array}{c|cccc} f_S & V_{IN} & P_{OUT} & R_L \\ \hline 50 \text{ MHz} & 45 \text{ V} & 5 \text{ W} & 25 \Omega \end{array}$$

The first step is to design the power stage and this can be done as described in [8]. The drain source voltage of the MOSFET is assumed to be a half wave rectified sine wave, hence the peak voltage is:

The rms value of a half wave rectified sine wave is:

$$V_{DS,rms} = V_{DS,peak} \sqrt{\frac{D}{2}} \tag{2}$$

And the rms value of the output voltage is:

$$V_{OUT,rms} = \sqrt{P_{OUT} \cdot R_L} \tag{3}$$

According to [10] the reactance of the resonance circuit can now be determined by:

$$X_{RC} = R_L \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1} \tag{4}$$

By combining equation 1, 2, 3, and 4, an expression for the needed reactance as function of input voltage, duty cycle, output power, and load is obtained:

$$X_{RC} = R_L \cdot \sqrt{\frac{V_{IN}^2 \cdot \pi^2 \cdot D}{2 \cdot (2 \cdot D - 2)^2 \cdot P_{OUT} \cdot R_L} - 1} \quad (5)$$

From equation 1 it is found that a duty cycle of 45% will give a peak voltage of 128 V, leaving approximately 20% headroom if a 150 V MOSFET is used. Substituting this and the values from table I into equation 5 gives a reactance of 134



Fig. 6. Schematic of a class E inverter with self oscillating gate drive.



Fig. 7. Bodeplot of the transferfunction from drain to source for different gate inductors.



Fig. 8. Simulated class E inverter waveforms with the simple resonant gate drive.

 $\Omega.$ If a capacitor of 100 pF is used, the value of the inductor becomes 528 nH.

When designing low power inverters switching at VHF it is important to select a MOSFET with low C_{OSS} [8]. For this reason Fairchilds FDT86256 is selected and the values in table II are extracted from the datasheet.

TABLE IIParasitic components of Fairchilds MOSFET FDT86256.Parasitic capacitances taken for $V_{DS} = V_{IN}$.

The effective output capacitance is $C_{DS,eff} = \frac{C_{DS}}{1-D} = 30$ pF, hence the total inductance of the resonance circuit and the input inductor should be:

$$L_{total} = \frac{1}{\omega_R^2 \cdot C_{S,ef}}$$

Where ω_R is given by:

$$\omega_R = \frac{\omega_S}{2 \cdot (1 - D)}$$

Knowing the values of X_{RC} , the input inductance can be calculated according to:

The values of all the components in the power stages have now been calculated and the only thing remaining is the gate inductor. The gate signal should have an amplitude of 10 V_{PP} and a phase shift as close to 180° as possible. The transfer function from drain to source for different gate inductors is shown in Fig. 7.

From the plot it can be seen that the phase shift is around 175° for all the values of L_G at 40 MHz, but starts to decrease for the two largest inductors around the switching frequency. As the peak to peak voltage across the drain and source of the MOSFET is 128 V, a gain of $20 \cdot log(10V/128V) = -22$ dB is needed to get a gate signal with an amplitude of $10 V_{PP}$. This is almost exactly what is achieved with the 140 nH inductor and as the phase shift is still 170° at the switching frequency this value is chosen for this design. Simulated waveforms of the designed inverter can be seen in Fig. 8.

The class E inverter (as most resonant circuits) behaves as a voltage controlled current source in open-loop situations; hence the output power increases with the input voltage. As the amplitude of the gate signal is a fixed ratio of the input voltage, a change in input voltage will lead to a change in amplitude of the gate signal. Hence the gate signal will be small at low input voltages, giving low gating losses at low power levels, and be large at high power, levels leading to lower conduction losses in the MOSFET. B. Used for synchronous rectification in a class E rectifier



Fig. 9. Schematic of a class E dc/dc converter with synchronous rectification.

The self oscillating gate drive is very well suited for synchronous rectification as it doesn't require a control signal to control the phase between the two gate signals. The two gate signals will automaticly oscillate out of phase with the drainsource voltage of the MOSFET they are controling, hence the gate drive on the inverters side will act as a master drive and the rectifier drive act as a slave drive following the frequency set by the master drive. This principle automatically takes component tolerances and temperature variations of critical design parameters into account. If used in an isolated converter this will further more benefit from not having a control signal and hence no need for communication across the isolation barier. Fig. 8 shows how a dc/dc converter with synchronous rectification can be made with this gate drive. As it can be seen the converter is completely symmetric across the resonant tank, C_R and L_R , hence operation in both directions is possible allowing for bidirectional power flow.

C. Used for high side gate drive in a class DE inverter

An implementation of the self-oscillating resonant gate drive in a half bridge is shown in Fig. 7. Here the two MOSFETs have equally sized inductors at the gates, the only difference is that the other node of L_{G1} is connected to V_{bias} through L_{H} and to the switch node through C_{G1} . In this way the voltage at this node will follow the switch node but have a dc offset set by V_{Bias1} . As the phase of gate signals of the MOSFETs are directly coupled to drain source voltages, which are 180 °



Fig. 10. Schematic of the class DE inverter with self oscillating gate drive.



Fig. 11. Picture of a SEPIC converter with the simple version of the gate drive.



Fig. 12. Measured waveforms of the SEPIC converter with the simple gate drive.

shifted, the gate signals will automatically be phase shifted and shoot through is avoided.

The simple version of the gate drive has been used for simplicity, but all the implementations can be used.

IV. PRACTICAL IMPLEMENTATIONS

A resonant SEPIC converter switching at 51MHz showing efficiency of 84% with the simple version of the gate drive has been implemented. A picture of the prototype is shown in Fig. 11 and the measured waveforms are shown in Fig. 12. By varying the bias voltage it is possible to regulate the output power from 4.3 W to 6.2 W. The efficiency is above 80% in the range 5-6.2 W but drops to 77% at the lowest output power. A plot of the power and efficiency for varying bias voltages is shown in Fig. 13.



Fig. 13. Measured efficiency and output power for varying bias voltages.



(a) Top



(b) Bottom

Fig. 14. A SEPIC converter with the gate inductor embedded in the PCB.

The three air core inductors takes up around half of the PCB footprint and they are the tallest components. In order to decrease the size further it is hence necessary to use another type of inductors. As the skin depth is very small and the maximum inductance 160 nH embedding the inductors in the PCB becomes a viable solution. A prototype of the same converter with the inductors embedded in the PCB has therefore been made [30] (see Fig. 14). Doing so the efficiency drops 2% but the power density increases approximately 4 times and the price becomes significantly lower.

The converter shown in Fig. 9 has also been implemented showing that the synchronous rectification works even without synchronization of M_1 and M_2 and allows for bidirectional power flow as well. The bidirectional converter is implemented as a class E inverter and a synchronous class E rectifier both using the simple self-oscillating gate drive.



Fig. 15. Picture of a class E inverter and a synchronous class E rectifier both with the simple version of the gate drive.



Fig. 16. Measured waveforms of the Class ${\ensuremath{\mathsf{E}}}$ converter with synchronous rectifier.



Fig. 17. Measured gate voltage of the SEPIC converter implemented with the gate drive with a 2.harmonic LC circuit connected to ground.

The reason for using a synchronous rectification is to minimize losses in the diode when dealing with low voltages and high currents; it also enables the converter to be used as a bidirectional converter. The converter with the synchronous rectifier is shown in Fig. 15. The converter reaches a maximum efficiency of 67% with the synchronous rectifier compared to 66.5% with a standard rectifier. The waveforms measured on the class E converter are shown in Fig. 16. It is evident that the inverter dictates the switching frequency. The simple gate drive in the rectifier is designed with an 180° phase shift from drain to gate at the switching frequency. This ensures that the

rectifiers gate drive follows the same frequency as the inverter.

Since the converter is implemented with a synchronous rectifier it could be used as a bidirectional converter with a load connected to the input of the inverter and a power source connected to the output of the rectifier. The transfer ratio $\frac{V_{in}}{I_{out}}$ was roughly the same in both directions $\frac{14V}{2A} \approx \frac{5V}{730mA} \approx 7$. The efficiency was kept above 50% when driven in reverse direction. The switching frequency was in this case dictated by the rectifier and the inverter just followed. With this self-oscillating gate drive it is possible to implement a small and cheap bi-directional converter using only a few passive components for driving the MOSFETs.

One of the more advance self-oscillating gate dives described earlier with a 2. harmonic LC circuit connected to ground is used with the SEPIC converter shown in Fig. 11. The transfer function is showed in Fig 3 is designed to remove the 2. harmonic and adding a 3. harmonic in phase with the fundamental. To ensure that the two peaks of the gate drives transfer function is placed above the fundamental frequency and the 3. harmonic the value of L_G is reduced and a small value for L_{2HS} is used. The total value of the inductances is reduced compared to the simple gate drive, in this case the total inductance was reduced by 45% which will free up more space when using PCB inductors.

The waveform of the gate voltage are shown in Fig. 17, it is evident that the gate signal generated by the more advance self-oscillating gate drive is more like a square wave than in 12. The peak-peak voltage is however twice of the simple gate drive and these increases the gate losses and in this case resulting in the efficiency reduction of a few % compared to the simple gate drive.

A implementation of a class DE inverter showing the function of the high side gate drive is shown in [31]. This is the first implementation of an off chip high side gate drive for operation in the VHF range.

V. CONCLUSIONS

This paper has presented several ways of designing a new self-oscillating resonant gate drive along with several examples of implementations in complete circuits. The presented gate drive is a simple, compact, low cost and reliable solution composed only of passive components. The gate drive is floating between the drain and source of the MOSFET and is directly coupled to the resonance frequency of the power stage. Due to this the gate drive can be used in all resonant circuits, also for high side drive and synchronous rectification. The gate drive offers an inherent open load protection and allows for synchronous rectification in isolated converters without synchronization across the isolation barrier. Finally the gate drive allows for output control without the low frequency ripple seen when utilizing burst mode control.

The paper has shown practical implementations of a low side gate drive, a high side gate drive and synchronous rectification and efficiencies up to 84% has been achieved.

REFERENCES

- R.B. Peterson, Switch Mode Power Supplies A New Approach for Consumer Audio Systems, IEEE Transactions on Broadcast and Television Receivers, vol.BTR-19, no.4, pp.263-268, Nov. 1973
- [2] ON Semiconductor, 1.2A, 2MHz Automotive Buck Switching Regulator, NCV890100 - Datasheet, Rev. 2, February 2013
- [3] W. C. Bowman, J. F. Balicki, F. T. Dickens et al., A resonant DC-to-DC converter operating at 22 megahertz, IEEE Applied Power Electronics Conference and Exposition, 1988, pp.3-11, 1-5 Feb 1988
- [4] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, High-Frequency Resonant SEPIC Converter With Wide Input and Output Voltage Ranges, IEEE Transactions on Power Electronics, vol.27, no.1, pp.189-200, Jan. 2012
- [5] J.M. Rivas, O. Leitermann, Yehui Han, D.J. Perreault, A very high frequency dc-dc converter based on a class 2 resonant inverter, IEEE Power Electronics Specialists Conference, 2008, vol., no., pp.1657-1666, 15-19 June 2008
- [6] D.J. Perreault, Jingying Hu, J.M. Rivas, Yehui Han, O. Leitermann, R.C.N. Pilawa-Podgurski, A. Sagneri, C.R. Sullivan, *Opportunities and Challenges in Very High Frequency Power Conversion, IEEE Applied Power Electronics Conference and Exposition*, 2009, vol., no., pp.1-14, 15-19 Feb. 2009
- [7] A. Knott, T.M. Andersen, P. Kamby, M.P. Madsen, M. Kovacevic, M.A.E. Andersen, On the ongoing evolution of very high frequency power supplies, IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.2514,2519, 17-21 March 2013
- [8] M.P. Madsen, A. Knott and M.A.E. Andersen, Low Power Very High Frequency Resonant Converter with High Step Down Ratio, IEEE Africon 2013
- [9] Yehui Han, O. Leitermann, D. A. Jackson et al. Resistance Compression Networks for Radio-Frequency Power Conversion, IEEE Transactions on Power Electronics, vol.22, no.1, pp.41-53, Jan. 2007
- [10] J. M. Rivas, Radio Frequency dc-dc Power Conversion, Doctoral Thesis, Massachusetts Institute of Technology, 2006
- [11] J.M. Rivas, R.S. Wahby, J.S. Shafran, D.J. Perreault, New architectures for radio-frequency dcdc power conversion, IEEE Transactions on Power Electronics, vol. 21, no. 2, pp. 380393, Mar. 2006
- [12] J. M. Rivas, D. Jackson, O. Leitermann et al., Design Considerations for Very High Frequency dc-dc Converters, IEEE Power Electronics Specialists Conference, 2006, vol., no., pp.1-11, 18-22 June 2006
- [13] I.D. de Vries, J.H. Van Nierop, J.R. Greene, Solid state class DE RF power source, IEEE International Symposium on Industrial Electronics, 1998, vol.2, no., pp.524,529, 7-10 Jul 1998
- [14] T.M. Andersen, S.K. Christensen, A. Knott, M.A.E Andersen, A VHF class E DC-DC converter with self-oscillating gate driver, IEEE Applied Power Electronics Conference and Exposition, 2011, vol., no., pp.885,891, 6-11 March 2011
- [15] M. Kovacevic, A. Knott, M.A.E. Andersen, Interleaved Self-Oscillating Class E Derived Resonant DC/DC Converters, Proceedings of the International Conference on Electrical and Computer Systems 2012
- [16] R.C.N. Pilawa-Podgurski, A.D. Sagneri, J.M. Rivas, D.I. Anderson, D.J. Perreault, Very-High-Frequency Resonant Boost Converters, IEEE Transactions on Power Electronics, vol.24, no.6, pp.1654,1665, June 2009
- [17] H. Fujita, A Resonant Gate-Drive Circuit Capable of High-Frequency and High-Efficiency Operation, IEEE Transactions on Power Electronics, vol.25, no.4, pp.962,969, April 2010
- [18] M.K. Kazimierczuk, D. Czarkowski, Resonant Power Converters, John Wiley & Sons, 2nd Edition, 2011, ISBN: 978-0-470-90538-8
- [19] T. Suetsugu, M.K. Kazimierczuk, Integrated Class DE Synchronized DC-DC Converter for On-Chip Power Supplies, IEEE Power Electronics Specialists Conference, 2006, vol., no., pp.1,5, 18-22 June 2006
- [20] Z. Kaczmarczyk, High-Efficiency Class E, EF2, and E/F3 Inverters, IEEE Transactions on Industrial Electronics, vol. IE-53, pp. 15841593, Oct. 2006
- [21] Z. Kaczmarczyk, A high-efficiency Class-E inverter computer model, laboratory measurements and SPICE simulation, Bulletin Of The Polish Academy Of Sciences, Technical Sciences, Vol. 55, No. 4, 2007
- [22] A.D. Sagneri, D.I. Anderson, D.J. Perreault, Optimization of Integrated Transistors for Very High Frequency DCDC Converters, IEEE Transactions on Power Electronics, vol.28, no.7, pp.3614,3626, July 2013

- [23] J. M. Rivas, O. Leitermann, Y. Han and D. J. Perreault, A Very High Frequency dc-dc Converter Based on a Class φ₂ Resonant Inverter, IEEE Transactions on Power Electronics, vol.26, no.10, pp.2980,2992, Oct. 2011
- [24] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, *High-Frequency Resonant SEPIC Converter With Wide Input* and Output Voltage Ranges, IEEE Transactions on Power Electronics, vol.27, no.1, pp.189,200, Jan. 2012
- [25] M. Kovacevic, A. Knott and M.A.E. Andersen, VHF Series-Input Parallel-Output Interleaved Self-Oscillating Resonant SEPIC Converter, IEEE Energy Conversion Congress and Exhibition US 2013
- [26] P. Shamsi, B. Fahimi, Design and Development of Very High Frequency Resonant DC-DC Boost Converters, IEEE Transactions on Power Electronics, vol.27, no.8, pp.3725,3733, Aug. 2012
- [27] J.M. Burkhart, R. Korsunsky, D.J. Perreault, Design Methodology for a Very High Frequency Resonant Boost Converter, IEEE Transactions on Power Electronics, vol.28, no.4, pp.1929,1937, April 2013
- [28] D. C. Hamill, Impedance plane analysis of class DE amplifier, Electronics Letters, vol.30, no.23, pp.1905-1906, 10 Nov 1994
- [29] D.C. Hamill, Class DE inverters and rectifiers for DC-DC conversion, IEEE Power Electronics Specialists Conference, 1996, vol.1, no., pp.854,860 vol.1, 23-27 Jun 1996
- [30] M.P. Madsen, A. Knott and M.A.E. Andersen, Very high frequency resonant DC/DC converters for LED lighting, IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.835,839, 17-21 March 2013
- [31] M.P. Madsen, A. Knott and M.A.E. Andersen, Very High Frequency Half Bridge DC/DC Converter, IEEE Applied Power Electronics Conference and Exposition 2014

 A.11 J. A. Pedersen, M. Madsen, A. Knott and M.
 A.E. Andersen: "Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter", IEEE Applied Power Electronics Conference and Exposition, Charlotte, NC, March 2015. Conference Proceedings.

Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter

Jeppe A. Pedersen*, Mickey P. Madsen*, Arnold Knott*, Michael A. E. Andersen* *Technical University of Denmark

Ørsteds Plads, building 349

2800 Kongens Lyngby

Denmark

Email: {jarpe, mpma, akn, ma}@elektro.dtu.dk

Abstract—This paper describes a galvanic isolated bidirectional Very High Frequency (VHF = 30 MHz - 300MHz) Class-E converter. The reason for increasing the switching frequency is to minimize the passive components in the converter. To make the converter topology bidirectional the rectifier has to be synchronous. This increases the complexity of the gate drives, which in this paper is solved by using a self-oscillating gate drive. A bidirectional converter has been implemented and is described in this paper; the converter reaches efficiencies above 80% in forward conduction mode and 73.5% in reverse conduction mode. The designed converter operates at a switching frequency of 35.6MHz, which is well within the VHF range. The same converter is also implemented with PCB embedded inductors to minimize cost and the physical volume of the total converter.

I. INTRODUCTION

In traditional Switch Mode Power Supplies (SMPS) small physical size is often a priority. The passive components usually take up most of the space and scale inversely with frequency and therefore increasing the switching frequency reduces the physical size of the overall converter. The switching frequency is typically limited to a few megahertz, this is where the switching losses become dominant as they increase linearly with frequency. Using resonant converters the switching losses can be significantly reduced, these types of converters are used in [1]-[5]. Increasing the switching frequency into the Very High Frequency (VHF = 30 MHz - 300 MHz) reduces the value as well as the physical size of passive components significantly, not just the energy storing components inside the converter also the EMI filter can be greatly reduced [6], [7]. The Class-E DC-DC converter topology, presented in [8] inspired by [9], is made with an inverter and a rectifier that both uses a resonant circuit to enable soft switching. This paper describes how the Class-E topology can be used for bidirectional VHF converters; this topology has two switching devises placed on each side of a resonant tank made with a capacitor in series with an inductor, to make this converter galvanic isolated the resonant capacitor is split into two capacitors where one is placed in series with the resonant inductor and one in the ground return path from the rectifier. Then an input and output filter which also helps to achieve soft switching in the inverter and rectifier. The symmetry around the resonant tank make the topology ideal for bidirectional operation. The schematic of the converter is seen in figure 1. The synchronous Class-E rectifier has been described in [10] and [11], and a bidirectional full-bridge class-E converter is implemented in [12] running at several hundred kilohertz. Bidirectional LLC converter have been used in different setups and shown that resonant converters are very suitable as bidirectional converter [13]-[15]. The bidirectional Class-E converter topology described in this paper could be suitable for many applications were size and price is a key consideration. One application for this converter is a single converter to charge a battery and then deliver the power from the battery to a load when it is not charging. This could be charging a battery from a PV panel during the day and delivering the power to an LED during the night. Another use of such a converter could be voltage sharing of batteries were two batteries need to have the same voltage. In future prospects this type of converter could also be integrated inside a chip, which can have two different transfer ratios depending on which way the chip is placed. This would cut the production cost for such a converter. Since this converter is galvanic isolated it is suitable for auxiliary supplies on the secondary side of an isolated converter. The converter is designed with a self-oscillating gate drive described in [10] that drives the MOSFET both in the inverter and the rectifier. This gate drive has shown great potential for normal VHF converters and in this paper it is used for driving a bidirectional converter.

II. THEORY

The converter is made as a standard Class-E converter in both directions, this is done by calculating the component values of the converter in forward direction, then afterward the reverse direction from the given resonant tank. To enable bidirectional power flow in a Class-E converter the rectifier needs to be implemented with a MOSFET and thereby becoming a synchronous rectifier as shown in figure 1. The use of VHF converters with synchronous rectification is a challenge due to the gate signals timing. The gate signals amplitude and duty cycle needs to be tightly controlled to ensure ZVS on the MOSFET in the rectifier and to keep the gate losses to a minimum. The design for the inverter is inspired from the method used in [16], made for a Class- ϕ_2 converter which is described in [17]. Where the impeadance of the circuit is matched to create a half sine on the drain of the MOSFET at the frequency needed to ensure ZVS in the inverter. The design method is changed to fit a Class-E inverter as described in [10]. This method is not ensuring a perfect sinusoidal current through the resonant tank but only that the ZVS of the MOSFET is achieved.



Fig. 1. Schematic of a bidirectional Class-E converter

A. Forward conduction mode

The forward conduction mode is when power is drawn from V_{IN} and delivered to V_{OUT} and the reverse conduction mode is the other way. The first step when designing the power stage is to make the rectifier. For a rectifier with a duty cycle D = 50%, the inductor L_{OUT} and the capacitor C_{S2} is chosen so they resonate at the switching frequency, and the output capacitor C_{OUT} is chosen to be large enough to limit the output ripple. The input impedance of a Class-E rectifier at different duty cycles can be found in [18], for D = 50% the input impedance of the rectifier is:

$$Z_{IN} = 0.5760 \cdot R_L \tag{1}$$

When the rectifier and its input impedance is found the inverter can be calculated. First the drain source voltage of the MOSFET is assumed to be a half wave rectified sine, hence the peak voltage is:

The peak voltage is used to find the rms value of a half wave rectified sine:

$$V_{DS,rms} = V_{DS,peak} \sqrt{\frac{D}{2}} \tag{3}$$

The rms value of the output voltage is found from:

$$V_{OUT,rms} = \sqrt{P_{OUT} \cdot Z_{IN}} \tag{4}$$

The needed reactance of the resonance circuit can now be determined by [10]:

$$X_{RC} = Z_{IN} \cdot \sqrt{\left(\frac{V_{DS,rms}}{V_{OUT,rms}}\right)^2 - 1}$$
(5)

By combining equation 2, 3, 4, and 5, an expression for the needed reactance as function of input voltage, duty cycle, output power, and load is obtained:

$$X_{RC} = Z_{IN} \cdot \sqrt{\frac{V_{IN}^2 \cdot \pi^2 \cdot D}{2 \cdot (2 \cdot D - 2)^2 \cdot P_{OUT} \cdot Z_{IN}} - 1}$$
(6)

The reactance of the resonant tank has to match the calculated reactance for the given design, if the capacitor C_{RES2} is chosen to be much bigger than the C_{RES1} it can be neglected in the calculations, and hence the reactance of the resonant tank is:

$$X_{RC} = 2 \cdot \pi \cdot f_r \cdot L_{RES} - \frac{1}{2 \cdot \pi \cdot f_r \cdot C_{RES1}}$$
(7)

Where $f_r = \frac{f_s}{2 \cdot (1-D)}$ is the frequency of the half wave sine. Last the input inductor is found this helps ensuring soft switching in the inverter, this is found by:

$$L_{IN} = \frac{1}{4 \cdot C_S \cdot f_r^2 \cdot \pi^2 - \frac{2 \cdot \pi \cdot f_r}{X_{RC}}}$$
(8)

Where $C_S = \frac{C_{S1}}{1-D}$ is the effective capacitance of C_{S1} . When the converter is designed in the forward direction the component values is fixed this limits the possibilities of optimizing the circuit in reverse direction. However there is still some degree of freedom such as the choice of load impedance or change of switching frequency.

B. Reverse conduction mode

When designing the bidirectional converter in the reverse conduction mode the first thing is to calculate the resonance of L_{IN} and C_{S1} , this limits the frequencies of the converter in this direction, and if the same frequency is maintained as in forward operation the load options is limited. In this section the switching frequency is set to be the same as in forward direction to simplify the calculations. The values of L_{IN} and C_{S1} determines the duty cycle of the rectifier in reverse conduction mode and this affects the input impedance of the rectifier. When this is found the equations 6, 7 and 8 described in the forward conduction mode can be used to determining the duty cycle of the inverter. Most VHF converter have a resonant gate drive driving the MOSFETs with a sinusoidal current, when using such a gate drive, the duty cycle can easily be changed by a varying DC offset. By doing this the duty cycle can be changed and there by also changing the output power of the converter.

III. BIDIRECTIONAL CONVERTER

To test the circuit a bidirectional converter has been designed. The converter is operating at low voltages were the benefit of having a synchronous rectifier is highest. This circuit is intended to operate from 14 V to 7 V in the forward conduction mode and from 7 V to 3.5 V in the reverse conduction mode. All specifications are shown in table I. The switching frequency is set just inside the VHF range. The power level is intended for charging of a battery in the forward conduction mode and delivering power from the battery to an LED in the reverse conduction mode. The circuit can either be connected to the charger or the LED.

TABLE I. DESIGN SPECIFICATION FOR THE CLASS-E CONVERTER

Power flow	V_{IN}	V_{OUT}	f_s	P_{OUT}	R_L
Forward	14	7 V	37 MHz	5 W	10 Ω
Reverse	7	3.5 V	37 MHz	1.5 W	$10 \ \Omega$

A. Gate drive

One of the main reasons to keep the switching frequency constant in both directions is to make the gate drive implementation easier. For the prototypes in this paper the gate drive is implemented with the self-oscillating passive gate drive described in [10]. The benefit of this gate drive is that it is controlled by the drain voltage so when the inverter starts sending power through the resonant tank the voltage across the MOSFET in the rectifier will rise and start the oscillation on the gate. For simplicity the gate drives for the MOSFETs are the same in the inverter and rectifier. the schematic is shown in figure 2.



Fig. 2. Schematic of a basic self-oscillating gate drive were L_G and C_G is external components

To be able to simulate the converter the gate drive has to be found. Both MOSFETs (M_1 and M_2) are chosen to be the FDC8601 from Fairchild as they have relatively low parasitic capacitance. Because they are both the same MOSFETs the external components in the gate drive can be the same for simplicity. First the gate drive for the inverter in the forward direction are designed to have gain of -10 dB and a phase shift close to 180° from drain to gate. This ensures that the MOSFET is soft switching and that the gate signal is large



Fig. 3. Waveforms from simulation of the converter in forward conduction mode, V_{M1} and V_{M1} are the voltage across the MOSFETs

enough to fully turn on the MOSFET. The gate drive in the rectifier is implemented with the same external components but here the gain of the gate dive are -4 dB as the parasitic capacitances are higher at lower drain-source voltages. The higher gain in the gate drive is beneficial since the drain voltage in the rectifier is smaller.

B. Simulation

With the gate drive, the converter was simulated in LTspice. The drain voltages of the two MOSFETs in forward conduction mode is shown in figure 3 and for the reverse conduction mode in figure 4. It is evident that the peak voltages are much higher than the in/output voltages, at 50% duty cycle the peak voltage across the MOSFET in a Class-E converter are 3.6 times the in/output for the inverter and rectifier respectively. The body diode is conduction a small period of time before the MOSFET turns on, this can be avoided by increasing the bias voltage seen in figure 2, the simulation are made with the components values stated in table II.

TABLE II. COMPONENT VALUES USED IN THE CLASS-E CONVERTER

Component	Value	
C_{IN}	100nF	
LIN	350nH	
C_{S1}	165pF	
C_{RES1}	1nF	
C_{RES2}	100nF	
L_{RES}	110nH	
C_{S2}	160pF	
L_{OUT}	156nH	
C_{OUT}	100nF	
L_G	68µH	
C_G	40pF	

C. Implemented converter

The converter was implemented with discrete inductors from Coilcraft as shown in figure 5. The waveforms measured in the forward conduction mode resembles the simulations as shown in figure 6. The switching frequency of the converter is measured to be 35.6 MHz. This is close the desired 37 MHz, the shift in frequency is reasonable since the gate drive is selfoscillating and is not externally controlled. The converter has been measured in reverse conduction mode and the waveforms



Fig. 4. Waveforms from simulation of the converter in reverse conduction mode



Fig. 5. Prototype with discrete inductors and synchronous rectifier

is shown in figure 7. The voltage across M_1 in reverse direction does not reach as high a voltage level as in the simulation, this could be due to the parasitics of the MOSFET which are more dominant at lower voltages. The efficiency of the converter in both directions is shown in figure 8, the efficiency is 80% for output powers above 1.5 W. In the forward conduction mode and above 70 %. The output power is changing over input voltage as shown in figure 9.

Implementing the magnetics as part of the PCB is widely used in traditional converters as described in [19], [20]. For VHF converters there is limited selection of magnetic materials since most materials does not operate well at these frequencies and therefore air core inductors are used. One form of air core inductors used for VHF converters is the PCB embedded inductors described in [21], [22]. A second identical prototype is made with PCB embedded inductors and compared to the



Fig. 6. Measured waveforms of the converter in forward direction



Fig. 7. Measured waveforms of the converter in reverse direction



Fig. 8. Measured efficiency vs. output power

first converter. The converter with PCB embedded inductors, is seen in figure 10. It only reach 74% efficiency in the forward conduction mode and 65% in reverse conduction mode. The reason for the lower efficiency is the low Q values for the PCB embedded inductors. The Q values of the PCB embedded inductors varies from 80 to 110 which is lower than what can be achieved with larger air core inductors. The volume of this converter with PCB inductors is significantly reduced as the PCB area remains the same where as the hight of the converter is significantly reduced. This can be very useful in many applications were size is an issue. The PCB embedded inductors has the benefit of being cheap and has a low spread in production. Both the converter with discrete inductors and the one with PCB embedded inductors is implemented as unidirectional converter with a diode rectifier operating in the forward conduction mode see figure 11. In this configuration the efficiency is roughly 2.5% lower than with the synchronous



Fig. 9. Measured output power vs. input voltage



Fig. 10. Prototype with PCB inductors and synchronous rectifier



Fig. 11. Unidirectional converter prototypes with diode rectifier, (a) with discreate inductors and (b) with PCB enbedded inductors

rectifier due to higher conduction losses in the diodes. This proves that the use synchronous rectifiers are very suitable for low voltage high current applications also in the VHF range.

IV. CONCLUSIONS AND FUTURE WORK

In this paper a VHF bidirectional converter has been implemented and presented, it has been proven that the Class-E converter topology it is able to operate as a bidirectional converter in the VHF range. The implemented converter achieves high efficiency in both operating modes over varying input voltages and output powers. This type of converter can be used in various applications where a small bidirectional converter is needed. The price of this type of converter is relatively low since it only uses ceramic capacitors, air core inductors and two MOSFETs. By using PCB embedded inductors the price and the physical size of the converter can be reduced further however in this case it does decrease the efficiency.

REFERENCES

- T. Andersen, S. Christensen, A. Knott, and M. A. E. Andersen, "A vhf class e dc-dc converter with self-oscillating gate driver," in *Applied Power Electronics Conference and Exposition (APEC), 2011 Twenty-Sixth Annual IEEE*, 2011, pp. 885–891.
- [2] M. Kovacevic, A. Knott, and M. Andersen, "A vhf interleaved selfoscillating resonant sepic converter with phase-shift burst-mode control," in *Applied Power Electronics Conference and Exposition (APEC)*, 2014 Twenty-Ninth Annual IEEE, March 2014, pp. 1402–1408.
- [3] M. Madsen, A. Knott, and M. Andersen, "Very high frequency half bridge dc/dc converter," in *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, March 2014, pp. 1409–1414.
- [4] M. K. Song, M. Dehghanpour, J. Sankman, and D. Ma, "A vhflevel fully integrated multi-phase switching converter using bond-wire inductors, on-chip decoupling capacitors and dll phase synchronization," in Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE, March 2014, pp. 1422–1425.

- [5] W. Cai, Z. Zhang, X. Ren, and Y.-F. Liu, "A 30-mhz isolated push-pull vhf resonant converter," in *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, March 2014, pp. 1456–1460.
- [6] A. Majid, J. Saleem, H. Kotte, R. Ambatipudi, and K. Bertilsson, "Design and implementation of emi filter for high frequency (mhz) power converters," in *Electromagnetic Compatibility (EMC EUROPE)*, 2012 International Symposium on, Sept 2012, pp. 1–4.
- [7] A. Majid, J. Saleem, and K. Bertilsson, "Emi filter design for high frequency power converters," in *Environment and Electrical Engineering (EEEIC), 2012 11th International Conference on*, May 2012, pp. 586–589.
- [8] R. Redl, B. Molnar, and N. Sokal, "Class e resonant regulated dc/dc power converters: Analysis of operations, and experimental results at 1.5 mhz," *Power Electronics, IEEE Transactions on*, vol. PE-1, no. 2, pp. 111–120, April 1986.
- [9] R. Gutmann, "Application of rf circuit design principles to distributed power converters," *Industrial Electronics and Control Instrumentation*, *IEEE Transactions on*, vol. IECI-27, no. 3, pp. 156–164, Aug 1980.
- [10] M. Madsen, J. Pedersen, A. Knott, and M. Andersen, "Self-oscillating resonant gate drive for resonant inverters and rectifiers composed solely of passive components," in *Applied Power Electronics Conference and Exposition (APEC), 2014 Twenty-Ninth Annual IEEE*, March 2014, pp. 2029–2035.
- [11] A. Knott, T. Andersen, P. Kamby, J. Pedersen, M. Madsen, M. Kovacevic, and M. Andersen, "Evolution of very high frequency power supplies," *Emerging and Selected Topics in Power Electronics, IEEE Journal of*, vol. 2, no. 3, pp. 386–394, Sept 2014.
- [12] S. Jalbrzykowski, A. Bogdan, and T. Citko, "A dual full-bridge resonant class-e bidirectional dc-dc converter," *Industrial Electronics, IEEE Transactions on*, vol. 58, no. 9, pp. 3879–3883, Sept 2011.
- [13] E.-S. Kim, J.-H. Park, Y.-S. Jeon, Y.-S. Kong, S.-M. Lee, and K. Kim, "Bidirectional secondary llc resonant converter using auxiliary switches and inductor," in *Applied Power Electronics Conference and Exposition* (APEC), 2014 Twenty-Ninth Annual IEEE, March 2014, pp. 1941–1947.
- [14] T. Jiang, J. Zhang, X. Wu, K. Sheng, and Y. Wang, "A bidirectional llc resonant converter with automatic forward and backward mode transition," *Power Electronics, IEEE Transactions on*, vol. 30, no. 2, pp. 757–770, Feb 2015.
- [15] T. LaBella, W. Yu, J.-S. Lai, M. Senesky, and D. Anderson, "A bidirectional-switch-based wide-input range high-efficiency isolated resonant converter for photovoltaic applications," *Power Electronics, IEEE Transactions on*, vol. 29, no. 7, pp. 3473–3484, July 2014.
- [16] J. Rivas, Y. Han, O. Leitermann, A. Sagneri, and D. Perreault, "A high-frequency resonant inverter topology with low voltage stress," in *Power Electronics Specialists Conference*, 2007. *PESC* 2007. *IEEE*, June 2007, pp. 2705–2717.
- [17] J. Rivas, O. Leitermann, Y. Han, and D. Perreault, "A very high frequency dc-dc converter based on a class ϕ_2 resonant inverter," *Power Electronics, IEEE Transactions on*, vol. 26, no. 10, pp. 2980–2992, Oct 2011.
- [18] D. Kazimierczuk, Marian K. Czarkowski, *Resonant Power Converters*, 2nd ed. Wiley, 2011.
- [19] H. Schneider, T. Andersen, A. Knott, and M. Andersen, "Hybrid winding concept for toroids," in *ECCE Asia Downunder (ECCE Asia)*, 2013 IEEE, June 2013, pp. 936–940.
- [20] R. Pittini, Z. Zhang, and M. Andersen, "High current planar magnetics for high efficiency bidirectional dc-dc converters for fuel cell applications," in *Applied Power Electronics Conference and Exposition* (APEC), 2014 Twenty-Ninth Annual IEEE, March 2014, pp. 2641–2648.
- [21] M. Madsen, A. Knott, M. Andersen, and A. Mynster, "Printed circuit board embedded inductors for very high frequency switch-mode power supplies," in *ECCE Asia Downunder (ECCE Asia)*, 2013 IEEE, June 2013, pp. 1071–1078.
- [22] J. Pejtersen and A. Knott, "Design and measurement of planar toroidal transformers for very high frequency power applications," in *Power Electronics and Motion Control Conference (IPEMC)*, 2012 7th International, vol. 1, June 2012, pp. 688–692.

Publications and Patents

A.12 M. Madsen and J. A. Pedersen: "Self-oscillating resonant power converter", WO2014067915 A2, November 2nd 2012

(12) INTERNATIONAL APPLICATION PUBLISHED UNDER THE PATENT COOPERATION TREATY (PCT)

(19) World Intellectual Property Organization

International Bureau



(43) International Publication Date 8 May 2014 (08.05.2014)

- (51) International Patent Classification: H02M 3/338 (2006.01) H02M 7/538 (2007.01)
- (21) International Application Number:
 - PCT/EP2013/072548
- (22) International Filing Date:
 - 29 October 2013 (29.10.2013)
- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 12191129.1 2 November 2012 (02.11.2012) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: MADSEN, Mickey P; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK). PEDERSEN, Jeppe Arnsdorf; Jagtvej 109E, DK-2200 Copenhagen N (DK).
- Agent: OLSEN, Henrik Bagger; Guardian IP Consulting (74)I/S, Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).
- (81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM,

(10) International Publication Number WO 2014/067915 A2

- AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LT, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN, TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.
- (84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

without international search report and to be republished upon receipt of that report (Rule 48.2(g))

(54) Title: SELF-OSCILLATING RESONANT POWER CONVERTER





(57) Abstract: The present invention relates to resonant power converters and inverters comprising a self-oscillating feedback loop coupled from a switch output to a control input of a switching network comprising one or more semiconductor switches. The self-oscillating feedback loop sets a switching frequency of the power converter and comprises a first intrinsic switch capacitance coupled between a switch output and a control input of the switching network and a first inductor. The first inductor is coupled in-between a first bias voltage source and the control input of the switching network and has a substantially fixed inductance. The first bias voltage source is configured to generate an adjustable bias voltage applied to the first inductor. The output voltage of the power converter is controlled in a flexible and rapid manner by controlling the adjustable bias voltage.
SELF-OSCILLATING RESONANT POWER CONVERTER

The present invention relates to resonant power converters and inverters comprising a self-oscillating feedback loop coupled from a switch output to a control input of a switching network comprising one or more semiconductor switches. The self-

- 5 oscillating feedback loop sets a switching frequency of the power converter and comprises a first intrinsic switch capacitance coupled between a switch output and a control input of the switching network and a first inductor. The first inductor is coupled in-between a first bias voltage source and the control input of the switching network and has a substantially fixed inductance. The first bias voltage source is
- 10 configured to generate an adjustable bias voltage applied to the first inductor. The output voltage of the power converter is controlled in a flexible and rapid manner by controlling the adjustable bias voltage.

BACKGROUND OF THE INVENTION

- 15 Power density is always a key performance metric of a power supply circuit such as AC-DC, DC-AC and DC-DC power converters to provide the smallest possible physical size for a given output power specification. Resonant power converter topologies are well-known types of DC-DC/switched mode power supplies or converters (SMPS) in the art. Resonant power converters are particularly useful for high switch-
- 20 ing frequencies such as above 1 MHz where switching losses of standard SMPS topologies (Buck, Boost etc.) tend to be unacceptable for conversion efficiency reasons. High switching frequencies are generally desirable because of the resulting decrease of the electrical and physical size of circuit components of the power converter like inductors and capacitors. The smaller components allow increase of the
- 25 power density of the SMPS. In a resonant power converter an input "chopper" semiconductor switch (often MOSFET or IGBT) of the standard SMPS is replaced with a "resonant" semiconductor switch. The resonant semiconductor switch relies on the resonances of circuit capacitances and inductances to shape the waveform of either the current or the voltage across the switching element such that, when switching
- 30 takes place, there is no current through or voltage across the switching element. Hence power dissipation is largely eliminated in at least some of the intrinsic capacitances of the input switching element such that a dramatic increase of the switching frequency becomes feasible for example to values above 10 MHz. This concept is known in the art under designations like zero voltage and/or current switching (ZVS)

and/or ZCS) operation. Commonly used switched mode power converters operating under ZVS and/or ZCS are often described as class E, class F or class DE inverters or power converters.

- 5 However, fast and accurate control of the output voltage of the resonant power converter remains a challenge. Prior art power converters described in the references below propose to utilize a self-oscillating feedback loop around the input switching element and driven by the intrinsic or inherent drain-to-source capacitance of a MOSFET switch in combination with a variable series inductance coupled to the
- 10 gate terminal of the MOSFET switch.

U.S. 4,605,999 discloses a self-oscillating power converter comprising a self-oscillating inverter circuit build around a single MOSFET switch. The inherent drain-to-source capacitance of the MOSFET switch supplies a feedback path sufficient to sustain self-oscillation of the inverter circuit if the frequency of operation is sufficient-ly high. The power converter is voltage regulated by a feedback loop deriving the control signal from a DC output voltage of the converter and applying the control signal to a variable inductance network comprising an inductor and a pair of non-linear capacitances.

20

U.S. 5,430,632 discloses a self-oscillating power converter utilizing a pair of MOSFET transistor switches in a half bridge configuration wherein the junction of the two MOSFET transistors is coupled to a reactive network which in turn is connected to an output rectifier. Intrinsic gate-to-drain inter-electrode capacitances of

- 25 the switching transistors serve as the sole means of sustaining oscillations. Oscillations are initiated at the gate-to-source terminals of the MOSFET transistor switches by a start-up circuit. The frequency of oscillation is determined by the gate-to-source capacitance of the MOSFET transistor switches and the inductance of an isolated gate drive transformer. The frequency of oscillation is controlled by varying induct-
- 30 ance of the isolated gate drive transformer coupled to the gate terminals of the MOSFET transistor switches through a pair of control windings.

However, the possible regulation range of adjustable inductances and/or capacitances tend to be very narrow due to physical component limitations and the accu-

3

racy may also be limited. Furthermore, adjustable inductances and/or capacitances are difficult to integrate on semiconductor substrates or on ordinary circuit carriers like printed circuit boards. Finally, the maximum regulation speed of the inductance or capacitance may be limited due to the reactive nature of the component leading

5 to an undesirable limitation of the speed of the regulation of the converter output voltage. This is of course particularly undesirable in view of the advantages of moving to higher converter switching frequencies for the reasons discussed above.

Consequently, it would be advantageous to provide a control mechanism for the

10 oscillation frequency that eliminates the need of variable reactive components like inductors and capacitors such that the converter output voltage can be controlled by appropriately controlling a level of a circuit voltage or circuit current for example in the form of an adjustable bias voltage

15 SUMMARY OF THE INVENTION

A first aspect of the invention relates to a resonant power converter or inverter comprising an input terminal for receipt of an input voltage and a switching network comprising one or more semiconductor switches controlled by respective control inputs. The switching network comprises a switch input operatively coupled to the

- 20 input terminal for receipt of the input voltage and a switch output operatively coupled to an input of a resonant network of the resonant power converter. The resonant network comprises a predetermined resonance frequency (f_R) and an output operatively coupled to a converter output terminal. A self-oscillating feedback loop is coupled from the switch output to a control input of the switching network to set a
- 25 switching frequency of the power converter. The self-oscillating feedback loop comprises a first intrinsic switch capacitance coupled between the switch output and the control input of the switching network,

a first bias voltage source configured to generate a first adjustable bias voltage, a first inductor with substantially fixed inductance coupled in-between the first bias

30 voltage source and the control input of the switching network. A voltage regulation loop of the resonant power converter is configured to control an output voltage of the power converter by controlling the first adjustable bias voltage applied to the first inductor.

PCT/EP2013/072548

4

The present resonant power converter allows flexible, rapid and accurate control of the converter output voltage by controlling the adjustable bias voltage applied to the first inductor coupled to the control input of the switching network. By adjusting a level of the adjustable bias voltage, an oscillation frequency of the self-oscillating

- 5 feedback loop coupled around the switching network can be controlled so as to set a switching frequency of the resonant power converter. The adjustment of the oscillation frequency of the self-oscillating feedback loop is achieved without making any adjustment of the inductance of the first inductor which therefore has a substantially fixed inductance independent of the level of the adjustable bias voltage. The skilled
- 10 person will understand that the term "substantially fixed" characterizing the inductance of the first inductor includes an inductance that vary slightly over temperature depending on electrical characteristics of a particular material of the selected inductor type. Furthermore, the application of the first adjustable bias voltage to the first inductor is preferably carried out without any adjustment of an inductive or capaci-
- 15 tive reactance of a component coupled in series with the first inductor in the voltage regulation loop. Hence, the first adjustable bias voltage generated by the voltage regulation loop is preferably applied to the first inductor without any transformer, tuneable inductor or tuneable capacitor in series with the first inductor.
- 20 The ability of adjusting the switching frequency of the present resonant power converter by adjusting the level of the first adjustable bias voltage enables a wide and accurate control range of the switching frequency and eliminates or circumvents the previously discussed disadvantages of relying on adjustable inductances and/or capacitances to adjust the switching frequency of the resonant power converter.
- 25 Power losses in intrinsic or parasitic capacitances such as the first intrinsic switch capacitance of the one or more semiconductor switches are furthermore reduced to a low level by the presence of first inductor because energy stored in these parasitic capacitances during charging is discharged to, and temporarily stored in, the first inductor. The stored energy in the first inductor is subsequently returned to parasitic
- 30 or intrinsic capacitances of the one or more semiconductor switches. The parasitic or intrinsic capacitances may comprise gate-source, gate-drain and drain-source capacitances of a MOSFET switch.

5

PCT/EP2013/072548

5

While the present invention is described in detail in the following with reference to implementations in resonant power converters/inverters and corresponding DC-DC power converters of Class E or DE type or topology, the skilled person will understand that the invention is equally applicable to other types of resonant power inverters, rectifiers and converters such as class E, F, DE and π_2 inverters and rectifiers and rectifiers and resonant boost, buck, SEPIC, LCC, LLC converters etc.

The voltage regulation loop may comprise a reference voltage generator supplying a DC or AC reference voltage to a first input of a comparator or error amplifier. A second input of the comparator may be coupled to the converter output voltage and an output of the comparator may be operatively coupled to a control input of the first bias voltage source. In this manner, the comparator or error amplifier may be configured to generate a suitable error signal as control signal for the first bias voltage source by a comparison of the output voltage of the converter with the DC or AC reference voltage. The error signal or signals applied to the first bias voltage source increases or decreases the first adjustable bias voltage in an appropriate direction to adjust the converter output voltage to the target output voltage indicated by the DC or AC reference voltage as explained in additional detail below in connection with the accompanying drawings.

The skilled person will appreciate that the switching network can comprise numerous types of switch topologies such as single switch topology, half-bridge or fullbridge switch topologies. According to a preferred embodiment, the switching network comprises a first semiconductor switch with a control terminal coupled to the

- 25 control input of the switching network and an output terminal coupled to the switch input and to the switch output. An input inductor is coupled between the input voltage and the switch input. This embodiment may comprise a basic class E power inverter or converter wherein the switching network comprises a single semiconductor switch with its output terminal, e.g. a drain terminal of a MOSFET, coupled both
- 30 to the input and output of the switching network. The input inductor forms part of the resonant network to control the setting of the predetermined resonance frequency (f_R). The control terminal, e.g. a gate or base terminal, of the single semiconductor switch is coupled to the control input of the switching network.

PCT/EP2013/072548

6

The input inductor and the first inductor may be magnetically coupled with a predetermined magnetic coupling coefficient, preferably a magnetic coupling coefficient larger than 0.1 or even more preferably larger than 0.4. The magnetic coupling provides a number of advantages relative to the case of uncoupled input and first induc-

- 5 tors such as improved phase response between the signal at the control input of the switching network and the switch output and larger and more constant gain. The magnetic coupling ensures that the inductor currents of the input inductor and first inductor are out of phase. Consequently, a phase shift between the control input signal, e.g. a gate voltage of the MOSFET switch, of the switching network and the
- 10 switch output is very close to 180 degrees. Furthermore, the magnetic coupling is preferably substantially constant across a wide frequency range to provide a more constant level of the first adjustable bias voltage when the output voltage V_{OUT} of the power converter is regulated.

15

Another preferred embodiment of the present resonant power converter comprises a half-bridge based switching network. The switching network comprises a first semiconductor switch coupled between the switch output and a voltage supply rail of the resonant power converter and having a control terminal coupled to the control input

- 20 of the switching network. The switching network additionally comprises a second semiconductor switch coupled between the switch output and the input terminal. A control terminal of the second semiconductor switch is coupled to a second bias voltage source through a cascade of a second inductor with substantially fixed inductance and a third inductor with substantially fixed inductance. A feedback
- 25 capacitor of the switching network is coupled between the switch output and an intermediate node between the second and third inductors. This embodiment of the present resonant power converter may comprise a class DE power converter, invertor or form part of a class DE based DC-DC power converter.
- 30 The feedback capacitor serves as a bootstrap device which raises a voltage level supplied to the control terminal of the second semiconductor switch and thereby facilitates use of a N-channel MOSFET transistor as semiconductor switch device. The second inductor serves as a high impedance signal path at the oscillation frequency allowing passage of a relatively slowly varying bias voltage component gen-

7

erated by the second bias voltage source, but blocking passage of a relatively high frequency voltage component supplied through the feedback capacitor. Consequently, by combining bias voltage components supplied through the second inductor and the feedback capacitor, the control voltage at the second switch is level

- 5 shifted and referred to the switch output instead of the voltage supply rail of the first semiconductor switch such as ground or a negative power supply voltage if the input voltage is a positive DC voltage. The self-oscillation loop may be configured to ensure that each of the semiconductor switches S₁ and S₂ is alternately switched between conducting and non-conducting states. The semiconductor switches S₁ and
- 10 S_2 are also switched in opposite phase according to a non-overlapping scheme.

The first inductor and the third inductor may be magnetically coupled with a predetermined magnetic coupling coefficient, preferably a magnetic coupling coefficient larger than 0.1 or even more preferably larger than 0.4. The magnetic coupling will

15 force a phase shift that is substantially 180 degrees between the control input signals, e.g. gate signals or voltages, of the first and second semiconductor switches. To provide a large magnetic coupling coefficient between the input inductor and the first inductor these may be wound around a common magnetically permeable member or core. For the same reason, the first inductor and the third inductor may be

20 wound around a common magnetically permeable member or core.

The first bias voltage source may be configured in various ways. In one embodiment, the first bias voltage source may be coupled between a suitable DC bias or reference voltage of the resonant power converter and a ground potential or nega-

- 25 tive supply rail thereof. The first adjustable bias voltage may be derived from the DC bias or reference voltage by suitable voltage division or regulation circuitry. In one embodiment, the first bias voltage source comprises a capacitor coupled from the first adjustable bias voltage to a fixed electric potential of the resonant power converter such as ground. A first adjustable resistor is coupled between the first adjust-
- 30 able bias voltage and a first DC reference voltage and a second adjustable resistor is coupled between the first adjustable bias voltage and a second DC reference voltage. The first DC reference voltage may possess a DC voltage higher than a maximum peak voltage of the first adjustable bias voltage. The second DC reference voltage may possess a DC voltage lower than an expected minimum voltage

of the first adjustable bias voltage such that the first adjustable bias voltage can be varied through a suitable voltage regulation range by adjusting a resistance ratio between the first and second adjustable resistances. Each of the first and second adjustable resistors preferably comprises a MOS transistor allowing the respective

5 resistances to be controlled from a high impedance gate terminal of the MOS transistor.

The first inductor may have an inductance between 1 nH and 10 µH such as between 1 nH and 50 nH. The latter inductance range makes it possible to form the
first inductor as an electrical trace pattern of a printed circuit board or as an integrated passive semiconductor component leading to considerable size reduction and reliability advantages of the resonant power converter.

- The substantially fixed inductance of the first inductor is preferably determined experimentally for example by adjusting its value until a suitable voltage swing is obtained at the control input of the switching network as explained below in additional detail. Preferably, the substantially fixed inductance is set such that a peak voltage at the control input of the switching network exceeds a threshold voltage of at least one of the semiconductor switches of the switching network. This threshold voltage
- 20 may for example lie between 5 and 10 V for an N-channel power MOSFET, but the skilled person will appreciate that other types of semiconductor switches may have different threshold voltages depending on characteristics of the semiconductor technology in question.
- In one embodiment, the substantially fixed inductance of the first inductor is selected such that a peak-peak voltage swing at the control input of the switching network is approximately equal to a numerical value of the threshold voltage of the at least one of the semiconductor switches of the switching network. In the above-mentioned example in respect of the N-channel power MOSFET, the peak-peak voltage swing would accordingly be adjusted to a value between 5 and 10 V in accordance with
 - the threshold voltage.

In another embodiment, the self-oscillating feedback loop comprises a series resonant circuit coupled in-between the control input of the first semiconductor switch

PCT/EP2013/072548

9

and a fixed electric potential of the converter. The series resonant circuit preferably comprises a cascade of capacitor and an inductor connected between the control input of the semiconductor switch and a negative power supply rail e.g. ground. The series resonant circuit functions to introduce additional uneven frequency compo-

5 nents, by attenuating one or more even harmonic frequency components, to a fundamental frequency component of the oscillating voltage waveform at the control input of the switching network, e.g. the gate of the first semiconductor switch. This leads to a trapezoidal waveform shape of the oscillating voltage waveform and results in faster switch turn-on and turn-off times.

10

A useful embodiment of the present resonant power converter comprises a DC-DC power converter. The DC-DC power converter is preferably constructed or derived by coupling a rectifier between the output of the resonant network and the inverter or converter output terminal to generate a rectified DC output voltage. The rectifier may

- 15 comprise one or more diodes to provide passive rectification of the DC output voltage. The rectifier of an alternative embodiment of the resonant power converter comprises a synchronous rectifier which may comprise one or more semiconductor switches. According to one such embodiment the synchronous rectifier comprises: a rectification semiconductor switch configured to rectify an output voltage of the
- 20 resonant network in accordance with a rectifier control input of the rectification semiconductor switch. A first rectification inductor with a substantially fixed inductance is coupled in-between a fixed or adjustable rectifier bias voltage and the rectifier control input. It is a significant advantage of this embodiment that the fixed or adjustable rectifier bias voltage of the rectifier may be left decoupled or unconnected to the first
- 25 bias voltage source generating the first adjustable bias voltage for the switching network on the input side of the resonant power converter for the reasons discussed in detail below with reference to FIG. 8 of the appended drawings. The fixed or adjustable rectifier bias voltage may for example be coupled to a fixed DC bias voltage source of the resonance power converter or to the rectified DC output voltage
- 30 through a resistive or capacitive voltage divider.

The skilled person will appreciate that numerous types of semiconductor transistors may be used to implement each of the first and second semiconductor switches depending on requirements such as threshold voltage, gate source break-down

10

voltage, drain source break-down voltage etc., imposed by any particular resonant power converter. Each of the first and second semiconductor switches may for example comprise a MOSFET or IGBT such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET.

5

A second aspect of the invention relates to a resonant power converter assembly comprising a resonant power converter according to any of the above described embodiments thereof and a carrier substrate having at least the switching network and the resonant circuit integrated thereon wherein an electrical trace pattern of the

- 10 carrier substrate is forming the first inductor. The carrier substrate may comprise a single-layer or multi-layer printed circuit board with integrally formed electrical wiring patterns interconnecting various electronic components of the resonant power converter. The relative small inductance required for the first inductance for achieving VHF switching frequencies of the power converter, e.g. in the order of tens of nH,
- 15 facilitates an advantageous integration of the first inductor, and potentially other inductors of the power converter of suitable size, directly in the wiring pattern of carrier substrates like printed circuit boards. This type of integration leads to several advantages such as saving component costs, reducing assembly time and costs and possibly improving reliability of the power converter assembly.

20

A particularly advantageous embodiment of the carrier substrate comprises a semiconductor die, such as a CMOS based integrated circuit, integrating all active and passive components of the present resonant power converter thereon.

25 BRIEF DESCRIPTION OF THE DRAWINGS

A preferred embodiment of the invention will be described in more detail in connection with the appended drawings, in which:

FIG. 1A) is an electrical circuit diagram of a class E resonant power converter in accordance with a first embodiment of the invention,

30 FIG. 1B) is an electrical circuit diagram of a class E resonant power converter comprising a pair of magnetically inductors in accordance with a second embodiment of the invention, FIG. 2A) is an electrical circuit diagram of a class E resonant power converter comprising a series resonant circuit in accordance with a third embodiment of the invention,

FIG. 2B) is an electrical circuit diagram of a class E resonant power converter com-

5 prising a series resonant circuit in accordance with a fourth embodiment of the invention,

FIG. 2C) is an electrical circuit diagram of a gate drive circuit for class E and DE resonant power converters comprising a plurality of series resonant circuits,FIG. 2D) shows a plurality of magnitude and phase response curves of transfer

- 10 functions of a MOSFET switch of the class E resonant power converter in accordance with the third embodiment of the invention, FIG. 2E) shows a plurality of control input signal waveforms of the MOSFET switch of the class E resonant power converter in accordance with the third embodiment of the invention.
- 15 FIG. 3A) is an electrical circuit diagram of a class DE resonant power converter in accordance with a fifth embodiment of the invention, FIG. 3B) is an electrical circuit diagram of a class DE resonant power converter comprising a pair of magnetically coupled inductors in accordance with a sixth embodiment of the invention,
- 20 FIG. 4 is an electrical circuit diagram of an exemplary DC-DC power converter based on the class E resonant power converter in accordance with the first embodiment of the invention,

FIG. 5 shows a series of graphs illustrating voltage waveforms at the output of a switching network of the class E resonant power converter of the first embodiment

- 25 for different bias voltage levels applied to the control input of the switching network, FIG. 6 is a circuit simulation model of a second exemplary DC-DC power converter based on the first embodiment of the class E resonant power converter, FIG. 7 shows a series of graphs illustrating various simulated voltage waveforms of the second DC-DC power converter for four different DC bias voltage levels of an
- 30 adjustable bias voltage; and FIG. 8 is an electrical circuit diagram of a third DC-DC power converter with synchronous rectification on the output side based on the class E resonant power converter in accordance with the first embodiment of the invention

12

DETAILED DESCRIPTION OF PREFERRED EMBODIMENTS

FIG. 1A) is a simplified electrical circuit diagram of a class E resonant power converter 100 in accordance with a first embodiment of the invention. The present class E resonant power converter is particularly well-adapted for operation in the VHF

- 5 frequency range for example at switching frequencies above 10 MHz or even higher such as between 30 and 300 MHz due to, amongst other factors, low switching losses in connection with the operation of a self-oscillating feedback loop connected around a transistor switch element S₁ as explained in further detail below. The class E resonant power inverter or converter 100 comprises an input pad or
- 10 terminal 102 for receipt of a DC input voltage V_{IN} from a DC power supply 104. The DC voltage level may vary considerably according to requirements of any particular conversion application such as lying between 1 V and 500 V for example between 10 V and 230 V. A switching network comprises a single switch transistor S₁. The skilled person will understand that the switch transistor S₁ can comprise different
- 15 types of semiconductor transistors such as MOSFETs and IGBTs. The skilled person will likewise understand that the switch transistor S₁ in practice can be formed by a plurality of parallel separate transistors e.g. to distribute operational currents between multiple devices. In one embodiment of the invention, S₁ is formed by an IRF5802 power MOSFET available from the manufacturer International Rectifier. A
- 20 gate terminal V_{GS} of the switch transistor S_1 forms a control input of the switching network allowing S_1 to be switched between a conducting state or on-state with low resistance between the drain and source terminals and a non-conducting state or off-state with very large resistance between the drain and source terminals. A drain terminal V_{DS} of the switch transistor S_1 forms both a switch input and a switch output
- 25 of the switching network in the present embodiment based on a single switch transistor. The drain terminal V_{DS} is at one side coupled to the DC input voltage through an input inductor L_{IN} (108). The drain terminal V_{DS} is also coupled to a first side of a series resonant network comprising resonant capacitor C_R and resonant inductor L_R . The input inductor L_{IN} resonant capacitor C_R , an intrinsic drain-source capacitance
- 30 C_{DS} of the MOSFET S₁ and the resonant inductor L_R (112) form in conjunction a resonant network of the power converter 100. A second and opposite side of the series resonant network is operatively coupled to an output terminal 114 or node of the class E resonant power converter 100 either directly as illustrated or through a suitable rectification circuit as illustrated in detail below. An inverter load is schematical-

PCT/EP2013/072548

13

ly indicated by a load resistor R_{LOAD} connected to the converter at the output terminal 114 and may generally exhibit inductive, capacitive or resistive impedance. The resonant network is designed with a resonance frequency (f_R) of about 50 MHz in the present implementation, but the resonance frequency may vary depending on

5 requirements of the application in question. In practice, the respective values of the resonant capacitor C_R and resonant inductor L_R may be selected such that a target output power at the converter output is reached for a particular load impedance. Thereafter, the value of the input inductor L_{IN} is selected such that a desired or target value of the predetermined resonance frequency (f_R) is reached in view of the

10 intrinsic drain-source capacitance C_{DS} for the selected switch transistor.

The present class E resonant power converter 100 comprises a self-oscillating feedback loop arranged around the transistor switch S_1 such that the oscillation frequency of the loop sets the switching or operational frequency of the power convert-

- 15 er 100 as briefly mentioned above. The self-oscillating feedback loop comprises an intrinsic gate-drain capacitance C_{GD} of the transistor switch S_1 which transmits a 180 degree phase shifted portion of the switch output signal at the drain terminal V_{DS} back to the gate terminal of the transistor switch S_1 . Additional loop phase shift is introduced by the gate inductor L_G which preferably comprises a substantially fixed
- 20 inductance. The gate inductor L_G is coupled in-between a variable bias voltage V_{Bias} and the gate terminal of the transistor switch S_1 . The variable bias voltage V_{Bias} is generated by a bias voltage generator or source with a design explained in further detail below in connection with FIG. 4. However, the adjustable bias voltage V_{Bias} applied to the gate terminal of transistor switch S_1 through the gate inductor L_G pro-
- vides an advantageous mechanism for controlling the converter output voltage V_{OUT}. This mechanism exploits that the time period of the cycle time, the cycle time being the reciprocal of the oscillation frequency of the feedback loop, during which S₁ remains in a non-conducting state is controlled by the previously mentioned components of the resonant network defining the resonance frequency (f_B). The latter fre-
- 30 quency controls when the voltage at the switch output at V_{DS} reaches ground or zero volts, being the lower power supply rail of the converter in the present embodiment, and thereby allowing S₁ to be turned on again without introducing switching losses to discharge the intrinsic drain-source capacitance C_{DS}. This operation mechanism where the resonant circuit is used to discharge the intrinsic semiconductor switch

capacitance until the voltage across the semiconductor switch reaches approximately zero is normally denoted zero voltage switching (ZVS) operation.

Conversely, the time period of the cycle time during which S_1 remains conducting, or

- 5 in its on-state, can be controlled by the level of the adjustable bias voltage. This property allows a duty cycle, and hence the oscillation frequency of the self-oscillating loop, to be adjusted. This is explained in further detail in connection with FIG. 5 below. Since the switch output at V_{DS} is coupled directly to the DC input voltage through the input inductor L_{IN} the average voltage at the switch output V_{DS} is
- 10 forced to equal the DC input voltage. The integral of a half-period sine waveform of frequency (f_R) equals the sine amplitude divided by pi times the resonance frequency (f_R). Furthermore, when S₁ is conducting the voltage across S₁ is essentially zero such that the voltage at the switch output V_{DS} becomes substantially zero. These circumstances lead to the following equation for a peak voltage, V_{DS,PEAK}, across S₁:

$$V_{DS,PEAK} = \frac{V_{IN} * \pi * f_R}{f_S} \qquad (1);$$

wherein f_S = The oscillation frequency of the self-oscillation loop which equals the switching frequency of the power converter.

20

Equation (1) reveals that a decreasing oscillation frequency leads to increasing switch output voltage V_{DS} as illustrated below by switch output voltages V_{DS} of FIG. 5.

- 25 The voltage waveforms, duty cycle control and oscillation frequency control discussed above are illustrated on the graphs 500, 510 and 520 of FIG. 5 for three different levels of the adjustable bias voltage V_{Bias} applied to the substantially fixed inductance gate inductor L_G. The scale on the y-axis of all graphs indicates voltage in volts while the x-axis scale indicates time in steps of 10 ns such that the entire x-
- 30 axis spans over about 100 ns. As mentioned above, L_G is coupled to the control input or gate V_{GS} of the transistor switch S_1 . In graph 500, the adjustable bias voltage V_{Bias} has been adjusted to a level which results in a duty cycle of approximately 0.5 in the switch output voltage V_{DS} . Waveform 501 shows the switch output voltage V_{DS}

PCT/EP2013/072548

15

while waveform 503 shows the corresponding gate-source voltage applied to the gate V_{GS} of S_1 . It is evident that the cycle time of the switch output voltage V_{DS} is about 10 s corresponding to an oscillation frequency of about 100 MHz.

- 5 In practice, the substantially fixed inductance of the gate inductor L_G may be selected such that a desired voltage amplitude of the (oscillating) gate-source voltage waveform is achieved. The voltage amplitude is preferably adjusted such that a suitable peak voltage at the gate terminal of MOSFET switch S₁ is reached in view of its threshold voltage and its gate break-down voltage. This means that the peak voltage
- 10 at the gate terminal should be sufficiently large to exceed the threshold voltage of the chosen semiconductor switch, e.g. V_{TH} of MOSFET switch S_1 . The oscillation frequency f_S of the self-oscillation loop will inherently lie close to the resonance frequency (f_R) of the resonant network if the bias voltage is adjusted approximately to the threshold voltage of the MOSFET switch S_1 . If the adjustable bias voltage V_{Bias} is
- 15 increased above the threshold voltage, the on-period of the MOSFET switch S₁ increases and leads to increase of the duty cycle of the oscillating switch output voltage waveform. This leads to a decreasing oscillation frequency or switching frequency of the power converter. The decrease of the oscillation frequency leads to an increase of the peak voltage V_{DS,PEAK} at the switch output as explained above in
- 20 connection with equation (1), and a corresponding increase of the peak voltage across the series resonant network comprising resonant capacitor C_R and resonant inductor L_R due to its coupling to the switch output voltage V_{DS} . Furthermore, because the series resonant network exhibits inductive impedance, the decreasing oscillation frequency of the switch output voltage waveform leads to a decrease of
- 25 the impedance of the series resonant network. The decrease of impedance leads in turn to increasing current and power through the series resonant network and through the load resistor R_{LOAD} - in effect increasing the converter output voltage V_{OUT} .
- 30 Consequently, the converter output voltage V_{OUT} can be controlled by appropriately controlling the adjustable bias voltage V_{Bias} applied to the substantially fixed inductance gate inductor L_G. This feature provides a highly flexible and fast way of controlling the converter output voltage V_{OUT} compared to prior art mechanism based on adjustable inductances and/or capacitances. In particular, the range of

16

adjustment of the adjustable bias voltage V_{Bias} can be very wide compared to the possible regulation range of the adjustable inductances and/or capacitances.

In graph 510, the adjustable bias voltage $V_{\mbox{\tiny Bias}}$ has been increased to a level which

- 5 results in a duty cycle of approximately 0.7 in the switch output voltage V_{DS} . Waveform 511 shows the switch output voltage V_{DS} while waveform 513 shows the corresponding gate-source voltage applied to the gate V_{GS} of S₁. As illustrated, the switch output voltage V_{DS} has increased from a peak level of approximately 30 volt for the 0.5 duty cycle condition depicted above to approximately 50 volt. It is evident that
- 10 the cycle time of the switch output voltage V_{DS} has decreased to about 18 ns corresponding to an oscillation frequency of about 55 MHz. Finally, in graph 520, the adjustable bias voltage V_{Bias} has been further increased to a level which results in a duty cycle of approximately 0.9 in the switch output voltage V_{DS} . Waveform 521 shows the switch output voltage V_{DS} while waveform 523 shows the corresponding
- 15 gate-source voltage applied to the gate V_{GS} of S_1 . As illustrated, the switch output voltage V_{DS} has further increased from a peak level of approximately 50 volt for the 0.7 duty cycle condition depicted above to approximately 150 volt. It is evident that the cycle time of the switch output voltage V_{DS} has further decreased to about 50 ns corresponding to an oscillation frequency of about 20 MHz.

20

FIG. 1B) is an electrical circuit diagram of a class E resonant power converter 100b comprising a pair of magnetically coupled inductors in accordance with a second embodiment of the invention. The skilled person will appreciate that the above discussed features, functions and components of the first embodiment of the class E

- 25 resonant power converter 100 may apply to the present embodiment as well. Likewise, corresponding components in the first and second embodiments of the present class E resonant power converter have been provided with corresponding reference numerals to ease comparison. The main difference between the first and second embodiments is that the previously discussed separate and substantially uncoupled
- 30 input inductor L_{IN} and gate inductor L_G have been replaced by the pair of magnetically coupled inductors L_{In} and L_G where the respective functions in the present class E resonant power converter 100b are similar to those of the first embodiment. The skilled person will appreciate that magnetic coupling between the input inductor L_{in} and gate inductor L_G may be achieved in numerous ways for example by a close-

PCT/EP2013/072548

17

ly spaced arrangement of the inductors e.g. coaxially arranged. The magnetic coupling provides a number of advantages over the first embodiment such as improved phase response between the control input and switch output of the MOSFET switch S_1 and larger and more constant gain. The magnetic coupling ensures that the re-

- 5 spective inductor currents of the input inductor L_{in} and gate inductor L_G are out of phase. Consequently, the phase shift between control input of the switch S₁ and the switch output is very close to 180 degrees. Furthermore, the magnetically coupled input inductor L_{in} and gate inductor L_G may be configured such that the magnetic coupling is substantially constant across a wide frequency range to provide a more
- 10 constant level of the first adjustable bias voltage when the output voltage V_{OUT} of the power converter is regulated.

The magnetic coupling between the magnetically coupled input inductor L_{in} and gate inductor L_{G} may also be accomplished by a transformer structure as schematically

15 indicated on FIG. 1B). The input inductor L_{in} and gate inductor L_G may for example be wound around a common magnetically permeable member or core. The latter embodiment has the advantage of a stronger coupling of magnetic fields between the input inductor L_{in} and gate inductor L_G. This forces a phase shift even closer to 180 degrees between the control input of the switch S₁ (i.e. gate voltage of switch

 S_1 and the switch output (i.e. drain voltage of the switch S_1).

The magnetically coupled input inductor L_{in} and gate inductor L_G may be configured to possess a magnetic coupling which is sufficient to ensure that inductor current forced in L_G by L_{IN} is sufficiently large to drive the control input of the switch S_1 . In this case the gate drive can also be used to drive cascode coupled transistors where

25 this case the gate drive can also be used to drive cascode coupled transistors where the intrinsic capacitance C_{GD} is small or non-existent.

FIG. 2A) is a simplified electrical circuit diagram of a class E resonant power converter 200 in accordance with a third embodiment of the invention. The present

30 power converter is of similar topology to the above discussed power converter based on a single switch transistor S₁. The skilled person will appreciate that the above discussed features, functions and components of the first embodiment may apply to the present embodiment as well. Likewise, corresponding components in the first and second embodiments of the present class E resonant power converter

18

have been provided with corresponding reference numerals to ease comparison. The main difference between the first and second embodiments lies in an addition of a series resonant circuit, comprising a cascade of capacitor C_{MR} and inductor L_{MR} , connected between the gate node or terminal V_{GS} of switch transistor S_1 and the

- 5 negative supply rail e.g. ground. The function of the series resonant circuit is to introduce additional uneven frequency components, by attenuating one or more even harmonic frequency components, to the fundamental frequency component of the oscillating gate voltage waveform of the switch transistor S₁. This leads to a trapezoidal waveform shape of the gate voltage of switch transistor S₁ leading to faster
- 10 switch turn-on and turn-off times. This is beneficial because it reduces the conduction losses, as the switch MOSFET S₁ will have relatively high resistance when the gate voltage is just above the threshold voltage. FIG. 2C) shows generally applicable embodiments of a series resonant network 201a coupled to the control input, e.g. a gate terminal, of a switch transistor or a switching network of a class E or DE
- 15 resonant power converter such as the class E and DE resonant power converters depicted on FIGS. 1A)-1B), FIG. 2A), FIGS. 3A)-3B), FIG. 4 and FIG. 8. The series resonant network 201 comprises a plurality of series resonant circuits of which one or more may be included in particular design of the class E or DE resonant power converter.

20

If a transistor switch like a MOSFET is driven by a sine wave the gate signal will be right above the threshold voltage of the MOSFET in a beginning and end of a conduction period of the MOSFET. This causes the on resistance to be very high in these periods as the MOSFET is only fully turned on when the gate signal is larger

- 25 than around twice the threshold voltage. In many resonant power converters these time periods are also where the largest currents are running through the MOSFET. Hence a lot of power is dissipated in these time periods. In order to improve the turn on speed of the MOSFET, higher order harmonics can be added to the fundamental sine wave leading to a more trapezoidal gate signal as mentioned above. This can
- 30 be achieved by adding one or more series resonant circuits, each preferably comprising an LC circuit, between the control input, i.e. the gate of the present MOSFET switch, and a drain or source of the MOSFET as illustrated on FIG. 2C). Here the capacitor, C_{GDext}, is optional and may be used to increase overall gain of the gate signal as shown in FIG. 2D). In the same way capacitor, C_{GSext}, can optionally be

PCT/EP2013/072548

19

used to lower the gain. The first and second LC based series resonant circuits C_{4HI} and L_{2HI} and L_{2HI} and L_{2HI} , respectively, are both connected to drain of MOSSFET switch S_1 and will cause higher harmonics to be in phase with the switch output voltage at the switch output, V_{DS} . The third and fourth LC based series resonant

- 5 circuits C_{4HO} and L_{4HO} and C_{2HO} and L_{2HO} , respectively, connected to the ground will cause the harmonics to be out of phase with V_{DS} as illustrated in FIG. 2D). The magnitude response curve 250 of graph 245 of FIG. 2D) illustrates how a LC circuit with a resonance at the second harmonic of the switching frequency of the power converter causes a peak in the gain at the third harmonic and that it is in phase with
- 10 the switch output V_{DS}. It can be shown that a 3rd harmonic in phase will be desirable for a duty cycle of 25%, but for a duty cycle of 50% it would be more desirable to have the signal out of phase as this would increase the signal right after and turn on of the MOSFET and just before turn off of the MOSFET. This feature can be achieved by setting a LC series resonant circuit with resonant frequency at the 2nd
- 15 harmonic to ground instead as indicated by the third and fourth series resonant circuits C_{4HO} and L_{4HO} and C_{2HO} and L_{2HO} , respectively, of FIG. 2C). By this connection, the magnitude response curve 252 of FIG. 2D) is achieved. Here a zero is seen at the 2nd harmonic of the switching frequency and again a peak at the 3rd harmonic, but this time with a phase shift of nearly 180 degrees (please refer to curve 252 of
- 20 the phase graph 246). The skilled person will understand that the number of harmonics to include in a given power converter design will depend on several parameters as price, complexity, efficiency etc. Adding higher order harmonics will in general increase the performance of the power converter, but it is important to consider which harmonics to include and the magnitude of those harmonics compared to the
- fundamental. Graphs 247 and 248 of FIG. 2E) show the fundamental and the 3rd and 5th harmonics of the switching frequency are in and out of phase with the switch output signal for the duty cycle D set to 25 % and 50 %. Note that the symbol * indicates that the depicted signal is in phase with the switch output signal V_{DS}. By comparing, the gate drive signal waveforms with the indicated ideal (rectangular) wave-
- 30 form shape of the same, it is clear that it is desirable to place the fundamental out of phase with the switch output signal, but for the 3rd and 5th harmonic it depends on the duty cycle and the current waveform. Exemplary gate drive waveforms that can be achieved by adding harmonics by the above-described series resonant networks are shown in the graphs 247 and 248 of FIG. 2E).

PCT/EP2013/072548

20

FIG. 2B) is an electrical circuit diagram of a class E resonant power converter 200b comprising a series resonant circuit in accordance with a fourth embodiment of the invention. The skilled person will appreciate that the above discussed features, func-

- 5 tions and components of the third embodiment of the class E resonant power converter 200 may apply to the present embodiment as well. Likewise, corresponding components in the third and fourth embodiments of the present class E resonant power converter have been provided with corresponding reference numerals to ease comparison. The main difference between the third and fourth embodiments is that
- 10 the previously discussed a series resonant circuit, comprising the cascade of capacitor C_{MR} and inductor L_{MR} , connected between the gate node or terminal V_{GS} of switch transistor S_1 and ground have been replaced by another type of resonant circuit comprising the parallelly coupled capacitor C_{MR} and inductor L_{MR} . The parallelly coupled capacitor C_{MR} and inductor L_{MR} are connected between the adjustable
- 15 bias voltage V_{Bias} and the gate inductor L_g . This connection with the parallelly coupled capacitor C_{MR} and inductor L_{MR} provides the same advantages as the series resonant circuit employed in the third embodiment, but with much smaller inductances of inductors L_g and L_{MR} leading to a significant reduction in costs and size.
- 20 FIG. 3A) is a simplified electrical circuit diagram of a class DE resonant power converter or inverter 300 in accordance with a fifth embodiment of the invention. The present resonant power inverter 300 is based on a switching network which comprises a half-bridge semiconductor topology. The present DE resonant power converter 300 provides several important advantages. One of the biggest challenges
- 25 when designing resonant power converters is a huge voltage stress imposed on the switch element in the single switch power converter topology described above in connection with the first, second, third and fourth embodiments of the invention. This voltage stress may reach 3-4 times the level of the DC input voltage. Using a half bridge switch topology instead limits a peak voltage across the each of the semi-
- 30 conductor switches S₁ and S₂ to a level of the input voltage. However, this requires a fast and efficient high side driver which can pose a significant advantage if an operating frequency or switching frequency above approximately 5 MHz is desired. The present generation of the first adjustable bias voltage solves this problem as it can also be used as a high side drive (V_{Bias1}) at several tens of megahertz. The half-

bridge comprises a cascade of the first semiconductor switch S_1 coupled between a switch output terminal 311 and ground and a second semiconductor switch S_2 coupled between the switch output terminal 311 and a DC input voltage rail supplied through power input terminal 302 from an external DC voltage source or generator

- 5 304. A coupling or mid-point node interconnecting the first and second semiconductor switches S_1 and S_2 form the switch output terminal 311. This switch output terminal 311 is the drain terminal of the first semiconductor switch S1. This switch output terminal or node 311 is coupled to a first side of a series resonant network comprising resonant capacitor C_B and resonant inductor L_B . A drain node of the transistor
- 10 switch S₂, coupled to the DC input voltage, comprises the switch input terminal of the present half-bridge switch. Each of semiconductor switches S₁ and S₂ may comprise a NMOS power transistor as illustrated by the switch symbol. Intrinsic drain-gate, gate-source and drain-source capacitances of the first NMOS transistor switch S₁ are depicted as C_{GD2}, C_{GS2} and C_{DS2} and likewise as C_{GD1}, C_{GS1} and C_{DS1} for NMOS transistor switch S
- 15 transistor switch S₂.

The resonant capacitor C_R , intrinsic drain-source capacitances of switches S_1 and $S_{2,}C_{DS1}$ and C_{DS2} , respectively, and the resonant inductor L_R in conjunction form a resonant network of the power converter 300. A second and opposite side of the

- 20 series resonant network is coupled to an output terminal 314 or node of the power converter 300. A converter load is schematically illustrated by a load resistor R_{LOAD} connected to the converter at the output terminal 314 and may generally exhibit inductive, capacitive or resistive impedance. The class DE resonant power inverter 300 furthermore includes a self-oscillating feedback loop arranged around the tran-
- 25 sistor switch S₁ such that an oscillation frequency of the loop sets the switching or operational frequency of the power converter in a manner similar to the one discussed in detail above in connection with the first embodiment of the invention. The self-oscillating feedback loop comprises an intrinsic gate-drain capacitance C_{GD2} of the transistor switch S₁ and a first gate inductor L_{G2} which preferably comprises a
- 30 substantially fixed inductance as discussed above. The gate inductor L_{G2} is coupled in-between a variable bias voltage V_{Bias2} and the gate terminal V_{GS2} of the transistor switch S₁. The variable bias voltage V_{Bias2} may be generated in numerous ways by a suitably configured bias voltage generator or source for example as explained in further detail below in connection with FIG. 4. In addition to the circuitry forming the

PCT/EP2013/072548

22

self-oscillating feedback loop arranged around transistor switch S₁, the current power inverter 300 comprises a second or high side adjustable bias voltage V_{Bias1} that is coupled to the gate terminal of the second semiconductor switch S₂ through a cascade of a second substantially fixed inductance L_H and a third substantially fixed

- 5 inductance L_{G1}. The inductances of the gate inductors L_{G2} and L_{G1} may be substantially identical. A feedback capacitor C_{G1} is coupled between the switch output node 311 and an intermediate node between the second and third substantially fixed inductances L_H and L_{G1}. The feedback capacitor C_{G1} serves as a bootstrap device which raises the voltage level supplied to the upper transistor switch S₂ and facili-
- 10 tates use of a N-channel MOSFET transistor as switch device. The inductor L_H serves as a high impedance signal path at the oscillation frequency allowing passage of a relatively slowly varying bias voltage component generated by the second adjustable bias voltage V_{Bias1}, but blocking passage of a relatively high frequency voltage component supplied through the bootstrap capacitor or feedback capacitor
- 15 C_{G1} . Consequently, combining the bias voltage components from L_H and C_{G1} , the gate control voltage at the gate terminal of the second switch S_2 is level shifted. In this manner, the gate control voltage is referred to the switch output node 311 instead of ground. The self-oscillation loop ensures that each of the semiconductor switches S_1 and S_2 is alternately switched between conducting and non-conducting
- 20 states in opposite phase in a non-overlapping manner. Thereby, the switch output node 311 becomes alternatingly clamped to the DC input voltage V_{IN} and ground through the semiconductor switches S_1 and S_2 at a frequency defined by the oscillation frequency of the self-oscillating loop.
- 25 The duty cycle of the switch output voltage waveforms and hence the converter output voltage at V_{out} can once again be controlled by synchronously controlling the respective bias voltages supplied by the first and second adjustable bias voltages V_{Bias2} and V_{Bias1} .
- 30 FIG. 3B) is an electrical circuit diagram of a class DE resonant power converter 300b comprising a pair of magnetically coupled inductors L_{G1} and L_{G2} in accordance with a sixth embodiment of the invention. The skilled person will appreciate that the above discussed features, functions and components of the first embodiment of the class DE resonant power converter 300 may apply to the present embodiment as

PCT/EP2013/072548

23

well. Likewise, corresponding components in the fifth and sixth embodiments of the present resonant power converters have been provided with corresponding reference numerals to ease comparison. The main difference between the fifth and sixth embodiments is that the previously discussed separate and substantially uncoupled

- 5 gate inductors L_{G1} and L_{G2} have been replaced by the pair of magnetically coupled inductors L_{G1} and L_{G2} where their respective functions in the present class E resonant power converter 300b are similar to those of the first embodiment. The skilled person will appreciate that magnetic coupling between the gate inductors L_{G1} and L_{G2} may be achieved in numerous ways for example by a closely spaced arrange-
- 10 ment of the inductors e.g. coaxially arranged. The magnetic coupling provides a number of advantages over the above-described first embodiment of the class DE resonant power converter 300 such as improved phase response between the respective gate signals at the gate terminals, or control inputs, of the inductors L_{G1} and L_{G2} and larger gain. The magnetic coupling ensures that the respective inductor cur-
- rents in the inductors L_{G1} and L_{G2} are out of phase. Hence, forcing a phase shift that is substantially 180 degrees between the gate signals of the inductors L_{G1} and L_{G2} .

The magnetic coupling between the inductors may also be accomplished by a transformer structure as schematically indicated on FIG. 3B) wherein the inductors L_{G1}

and L_{G2} are wound around a common magnetically permeable core. The latter embodiment has the advantage that a larger magnetic coupling between the inductors L_{G1} and L_{G2} can be achieved and the relative phase shift of substantially 180 degrees between the respective gate signals or voltages of the MOSFET switches S_1 and S_2 is enforced even stronger.

25

FIG. 4 is a schematic electrical circuit diagram of a DC-DC or switched mode power converter/supply (SMPS) 400 which is based on the class E resonant power converter or inverter 100 disclosed above in a first embodiment of the invention. The DC-DC power converter 400 comprises, in addition to the circuitry of the class E

30 resonant power converter 100, a voltage control loop controlling the level of a DC output voltage V_{OUT} of the DC-DC converter and a rectifier 413 schematically illustrated by a storage capacitor and a diode. The rectifier 413 preferably includes a series inductor coupled between the illustrated diode and the output voltage terminal V_{OUT} . The skilled person will appreciate that the illustrated diode(s) based rectifier

PCT/EP2013/072548

24

413 may be replaced by a synchronous rectifier based on one or more actively controlled semiconductor switches rather than diodes as described in additional detail below with reference to FIG. 8. The voltage control loop regulates respective resistances of a pair of pull-up and pull-down MOSFET resistors M₁ and M₂ forming

- 5 part of bias voltage source or generator supplying the adjustable bias voltage V_{Bias} . The adjustable bias voltage V_{Bias} is applied to the gate terminal of transistor switch S_1 through the gate inductor L_G as explained in connection with FIG. 1A) above. The voltage control loop comprises a comparator or error amplifier 414 which has a first input coupled to a DC or AC reference voltage V_{BEF} and a second input coupled to
- 10 the DC output voltage V_{OUT} of the converter. A resulting error signal V_{ERR} reflecting whether the output voltage is lower or higher than the reference voltage is fed to an optional level converter 414. The level converter 414 is configured to provide appropriate gate control signals V_{C1} and V_{C2} for the pair of pull-up and pull-down MOSFET resistors M_1 and M_2 to either increase or decrease the adjustable bias voltage V_{Bias} .
- 15 The bias voltage source or generator comprises the MOSFET resistors M_1 and M_2 coupled between the DC input voltage and ground. Hence, the adjustable bias voltage V_{Bias} can either be pulled towards the DC input voltage or ground depending on the adjustable on-resistances of the MOSFET resistors M_1 and M_2 . The skilled person will appreciate that the voltage control loop can be configured in numerous ways
- 20 to provide appropriate control signals to the MOSFET resistors M₁ and M₂ for example by proportional voltage control or by purely binary voltage control, i.e. up/down.

FIG. 6 is a circuit simulation model of a second DC-DC power converter based on the first embodiment of the class E resonant power converter. The DC-DC converter comprises a rectifier coupled between an output of the series resonant circuit, including C1 and L4, and a load resistance R6 coupled to an output voltage of the converter. The rectifier comprises components C3, D, L2 and C5. Inductor and capacitor component values of the second DC-DC power converter are listed on the figure in Henry and Farad, respectively. Accordingly, the inductance of the gate in-

30 ductor Lg is set to a substantially fixed value of 68 nH. The semiconductor switch is modelled by an ideal switch ISW with the listed parameters, i.e. an on-state resistance of 1.0 Ω off-state resistance of 1M Ω and threshold voltage of 4.5 V. FIG. 7 shows a series of graphs 600, 610, 620, 630 and 640 illustrating various simulated voltage waveforms of the simulation model of the second DC-DC power converter for four different fixed DC bias voltage levels of the adjustable bias voltage V_{bias} . V_{bias} is stepped through fixed DC voltage levels of -7.0, -2.0, 3.0 and 8.0 volt as

- 5 illustrated by waveforms 607, 605, 603, 601, respectively, of graph 600 showing the DC bias voltage level. The DC input voltage V2 (Vin) is kept constant at 50 volts for all simulations.
- The scale on the y-axis of all graphs indicates voltage in volts while the x-axis scale indicates time in steps of 0.01 μ s such that the entire x-axis spans over about 0.05 μ s.
- Graph 610 illustrates the corresponding oscillating control input voltage waveforms 617, 615, 613, 611 at the indicated gate node (refer to FIG. 6) for the four different levels of the DC bias voltage. The higher average level of the oscillating control input voltage waveforms for the highest DC bias voltage of 8.0 V is evident. Graph 620 illustrates the corresponding switch output voltage waveforms 627, 625, 623, 621 at the switch output node i.e. at the indicated drain node (refer to FIG. 6). The longer conducting states or on-states of the switch ISW for the highest DC bias voltage of 8.0 V is evident leading to a lower oscillation frequency or switching frequency of the converter.

Graph 640 illustrates the corresponding load power waveforms 627, 625, 623, 621
for the power delivered the load resistor R6 through the converter output. The gradually increasing load power from about 1.5 W at the lowest DC bias voltage of -7.0 V
to about 3.5 W at the highest DC bias voltage of 8.0 V is evident. Hence, converter
output power and therefore converter output voltage can be controlled by adjusting
the voltage supplied by the adjustable bias voltage V_{bias}.

30 FIG. 8 is a schematic electrical circuit diagram of a DC-DC or switched mode power converter/supply (SMPS) 800 based on the class E resonant power converter or inverter 100 according to the first embodiment of the invention discussed above. The DC-DC power converter 800 comprises, in addition to the circuitry of the class E resonant power converter 100, a synchronous rectifier building around transistor

PCT/EP2013/072548

26

switch S_{R1} and comprising additional passive components L_{G2} and L_{OUT} . The skilled person will understand that the DC-DC power converter 800 may comprise an output capacitor coupled from V_{OUT} to the negative supply rail (e.g. ground) and a voltage control loop similar to the one discussed above in connection with FIG. 4 in the

- fourth embodiment of the invention. The voltage control loop being configured to control the output voltage at V_{OUT} of the power converter 800 as defined by a DC or AC reference voltage. The transistor switch element S_{R1} and inductors L_{G2} and L_{OUT} provide a synchronous rectifier in the DC-DC power converter 800 and replaces the diode based asynchronous rectifier circuit 413 discussed above. Since the control
- 10 input, e.g. the gate drive signal, of the switching network of the present class E and DE resonant power converters does not need a traditional PWM or PDM type of control signal (but only the two adjustable bias voltage V_{Bias1} and V_{Bias2}), the resonant power converters in accordance with the present embodiments are generally very well suited for synchronous rectification as illustrated on FIG. 8 for this particu-
- 15 lar embodiment. The traditional PWM or PDM type of control signals are not required because is not necessary to control a phase between the respective control input signals of the first transistor switch S₁ and the rectification transistor switch S_{R1}. The rectification transistor switch S_{R1} may for example be coupled to a suitable fixed rectifier DC bias voltage V_{Bias2} applied to the inductor L_{G2} coupled to the gate
- (i.e. control input) of S_{R1}. For rectification purposes, the gate terminal of S_{R1} is driven by an oscillation output voltage, i.e. the drain voltage V_{DS}, of the first semiconductor switch S₁ to automatically maintain synchronous operation between S₁ and SR1. This absence of the traditional PWM or PDM type of control signals on the respective gate terminals of the first transistor switch S₁ and rectification transistor switch
- S_{R1} is a significant advantage leading to simplified power converter design and smaller component count. In isolated power converter applications, the present diode based asynchronous rectifier circuit 413 possess an additional advantage because it eliminates the need for transmitting or communicating the traditional PWM or PDM type control signal or signals across a voltage isolation barrier of the reso-
- 30 nant power converter. This type of voltage isolation barrier will typically require expensive and space consuming components like optocouplers or fast transformers in traditional power converter topologies. As illustrated by FIG. 8, the present DC-DC power converter with synchronous rectification may be completely symmetrical in terms of circuit topology across a series resonant network comprising resonant ca-

pacitor C_R and resonant inductor L_R allowing for bidirectional power flow between the DC input power source V_{IN} 804 and the output voltage at V_{OUT} . The skilled person will appreciate that the input transistor switch S_1 and rectifier transistor switch S_{R1} may be substantially identical or different components and the same applies to the

5 fixed inductance inductors L_{G2} and L_{G1} depending on factors such as the voltage conversion ratio of the resonant power converter.

The skilled person will appreciate that the above-described synchronous rectifier may be added to each of the above discussed class E and DE resonant power con-

10 verter embodiments depicted above on FIG. 1B), FIGS. 2A)-2B) and FIGS. 3A)-3B).

<u>CLAIMS</u>

1. A resonant power converter comprising:

an input terminal for receipt of an input voltage,

5 a switching network comprising one or more semiconductor switches controlled by respective control inputs,

the switching network comprising a switch input operatively coupled to the input terminal for receipt of the input voltage and a switch output operatively coupled to an input of a resonant network of the resonant power converter,

- 10 the resonant network comprising a predetermined resonance frequency (f_R) and an output operatively coupled to a converter output terminal, a self-oscillating feedback loop coupled from the switch output to a control input of the switching network to set a switching frequency of the power converter;
- the self-oscillating feedback loop comprising:
 a first intrinsic switch capacitance coupled between the switch output and the control input of the switching network.

a first bias voltage source configured to generate a first adjustable bias voltage, a first inductor with substantially fixed inductance coupled in-between the first bias voltage source and the control input of the switching network,

20 a voltage regulation loop configured to control an output voltage of the power converter by controlling the first adjustable bias voltage applied to the first inductor.

2. A resonant power converter according to claim 1, comprising: an input inductor coupled between the input terminal and the switch input,

- 25 the switching network comprising a first semiconductor switch with a control terminal coupled to the control input of the switching network and an output terminal coupled to the switch input and to the switch output.
- 3. A resonant power converter according to claim 2, wherein the input inductor and
 the first inductor are magnetically coupled with a predetermined magnetic coupling
 coefficient, preferably a magnetic coupling coefficient larger than 0.1 or even more
 preferably larger than 0.4.

4. A resonant power converter according to claim 1, wherein the switching network comprises:

a first semiconductor switch coupled between the switch output and a voltage supply rail of the resonant power converter and having a control terminal coupled to the

- 5 control input of the switching network, a second semiconductor switch coupled between the switch output and the input terminal; and wherein a control terminal of the second semiconductor switch is coupled to a second bias voltage source through a cascade of a second inductor with substantially
- 10 fixed inductance and a third inductor with substantially fixed inductance, and wherein a feedback capacitor is coupled between the switch output and an intermediate node between the second and third inductors.

5. A resonant power converter according to claim 4, wherein the first inductor and
the third inductor are magnetically coupled with a predetermined magnetic coupling
coefficient, preferably a magnetic coupling coefficient larger than 0.1 or even more
preferably larger than 0.4.

6. A resonant power converter according to claim 3 or 5, wherein the input inductor

20 and the first inductor are wound around a common magnetically permeable member of core; or

the first inductor and the third inductor are wound around a common magnetically permeable member or core.

- 7. A resonant power converter according to any of the preceding claims, wherein the first bias voltage source comprises:
 a capacitor coupled from the first adjustable bias voltage and a fixed electric potential of the resonant power converter such as ground,
 a first adjustable resistor coupled between the first adjustable bias voltage and a first
- DC reference voltage,
 a second adjustable resistor coupled between the first adjustable bias voltage and a second DC reference voltage.

5

8. A resonant power converter according to any of the preceding claims, wherein the voltage regulation loop comprises:

a reference voltage generator supplying a reference DC or AC voltage to a first input of a comparator,

a second input of the comparator being coupled to the converter output voltage, an output of the of the comparator operatively coupled to a control input of the first bias voltage source.

9. A resonant power converter according to any of the preceding claims, wherein the
first inductor has an inductance between 1 nH and 10 μH such as between 1 nH and
50 nH.

10. A resonant power converter according to any of the preceding claims, wherein the substantially fixed inductance of the first inductor is set such that a peak voltage
at the control input of the switching network exceeds a threshold voltage of at a semiconductor switch of the switching network.

11. A resonant power converter according to claim 10, wherein the substantially fixed inductance of the first inductor is selected such that a peak-peak voltage swing

20 at the control input of the switching network is approximately equal to a numerical value of the threshold voltage of the at least one of the semiconductor switches of the switching network.

12. A resonant power converter according to any of the preceding claims, wherein
the self-oscillating feedback loop further comprises:
a series resonant circuit coupled in-between the control input of the switching network and a fixed electric potential of the power converter.

13. A resonant power converter according to claim 12, wherein the self-oscillating
feedback loop further comprises:
a first series resonant circuit coupled in-between the control input of the first semi-

conductor switch and fixed electric potential of the converter such as a positive or negative DC supply voltage or a ground voltage,

5

15

a second series resonant circuit coupled in-between the control input of the first semiconductor switch and the switch output.

- 14. A resonant power converter according to any of the claims 1-11, wherein the self-oscillating feedback loop further comprises:
 - a parallel resonant circuit coupled in series with the first inductor in-between the first adjustable bias voltage and the first inductor.

15. A resonant power converter according to any of the preceding claims, furthercomprising:

a rectifier coupled between the output of the resonant network and the converter output terminal to provide a rectified DC output voltage.

16. A resonant power converter according to claim 15, wherein the rectifier comprises a synchronous rectifier.

17. A resonant power converter according to claim 16, wherein the synchronous rectifier comprises:

a rectification semiconductor switch configured to rectify an output voltage of the

20 resonant network in accordance with a rectifier control input of the rectification semiconductor switch,

a first rectification inductor with a substantially fixed inductance coupled in-between a fixed or adjustable rectifier bias voltage and the rectifier control input.

- 25 18. A resonant power converter according to claim 17, wherein the fixed or adjustable rectifier bias voltage is coupled to a fixed DC bias voltage source or to the rectified DC output voltage through a resistive or capacitive voltage divider.
- 19. A resonant power converter according to any of claims 2-18, wherein one of the
 first and second semiconductor switches comprises a MOSFET or IGBT such as a
 Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET.

20. A resonant power converter assembly comprising: a resonant power converter according to any of the preceding claims,

a carrier substrate having at least the switching network and the resonant circuit integrated thereon,

an electrical trace pattern of the carrier substrate forming the first inductor.

5 21. A resonant power converter assembly according to claim 16, wherein the carrier substrate comprises a semiconductor die.

1/10





FIGS. 1A, B

2/10







FIG. 2C



3/10

4/10



D)





E)

FIGS. 2D, E
300b -

5/10



FIGS. 3A, B

B)



FIG. 4

6/10

7/10



FIG. 5





.model ISW SW(Ron=1 Roff=1Meg Vt=4.5 Vh=-0.5)

.model D D

.param Vin=50 .param Lin 1u .param Cds 26p .param Cgd 10p .param Cgs 80p .param Rg 1.5 .param Ct 1n .param Lt 250n .param Lg 68n .param Lr 30n .param Cr 80p 9/10



FIG. 7

10/10



A.13 M. Madsen, A. Knott, M. A.E. Andersen, A. P. Mynster:

"Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies",

IEEE Energy Conversion Congress and Exhibition Asia DownUnder, Melbourne, Australia, June 2013. Conference Proceedings p1071-1078.

Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies

Mickey Madsen, Arnold Knott and Michael A.E. Andersen Department of Electrical Eng., Technical University of Denmark Ørsteds Plads, Building 349, 2800 Kgs Lyngby, Denmark Tel.: +45 45252525 Fax.: +45 45880117 Email: {mpma, akn and ma}@elektro.dtu.dk

Abstract—The paper describes the design of three different structures for printed circuit board embedded inductors. Direct comparison of spirals, solenoids and toroids are made with regard to inductance, dc and ac resistance, electromagnetic field and design flexibility.

First the equations for the impedances are given and an example of the achievable impedances are given. Prototypes are then made and measured. The differences between the three structures and the accuracy of the formulas are evaluated.

Finite element simulations are used to investigate the magnetic field around the structure, in order to take possible electromagnetic interference problems into account, when the structures are compared. The simulated fields are verified through near field measurements performed on the prototypes.

Finally deign flexibility are considered, both regarding scalability and design of the individual inductors and implementation in a complete design. At the end of the paper a summary of pros and cons of the three structures are listed.

I. INTRODUCTION

The constant strive for small, cheap and efficient power supplies firstly lead to the development of Switch-Mode Power Supplies (SMPS) in the early 70s [1]. Since this strive has lead to constantly increasing switching frequencies, as the size of modern power supplies are mainly governed by the passive energy storing elements, which scales inversely with the switching frequency.

Commercially available converters today switch at frequencies up to several mega hertz and can have efficiencies of more than 95% (e.g. [2]).

The reason not to increasing the switching frequency even higher and thereby reaching even higher power densities is the switching losses. However with the use a soft switching techniques and resonant converters, SMPSs with switching frequencies in the Very High Frequency range (VHF, 30-300MHz) have been designed with efficiencies up to approx. 90%, [3] and [4].

This highly increased switching frequency has several pros and cons which has been discussed in various publication, e.g. [5]. One of the challenges is to design high-Q inductors, as most magnetic materials are limited to operation up to a few megahertz. With the high frequency, the inductance needed is however also lowered significantly and it is hence possible to use relatively small air core inductors. So far the type of inductor most commonly used is an air core solenoid. Anders P. Mynster

DELTA - Danish Electronic, Light and Acoustics Venlighedsvej 4, 2970 Hørsholm, Denmark Tel.: +45 72194425 Fax +45 72194001 Email: apm@delta.dk

Given the small inductance, the inductors can however also be designed as Printed Circuit Board (PCB) embedded inductors. Several structures for the design of these inductors exist and many of them have been described several times in previous publication, e.g. [6], [7] and [8]. Spiral inductors are commonly used in Integrated Circuits (ICs) and hence several publications with design guides for these exists ([9], [10] and [11]). Though discrete air core solenoid are commonly used, solenoids are not as common for embedded inductors. Some publications are however available showing inductors with Q values of up to more than 300 ([12] and [13]).

Both spirals and solenoids have strong external magnetics fields. This make them likely to cause Electromagnetic Interference (EMI) problems, either by direct radiation or by coupling to nearby metallic structures such as cables and other circuit boards. The fields have been described for spirals in [6] and for solenoids and toroids in [7]. The external field from the toroid is much weaker than for the two other structures. For this reason the design of this structure has been investigated in [6], [14] and [15].

Though the three structures have been described in several publications, no direct comparison have been made. In this paper this comparison will be made, the paper hence describes the three different structures for PCB embedded inductors; solenoids, spirals and toroids (see Fig 1). In section two, formulas for calculating the inductance and Equivalent Series Resistance (ESR) are given. The formulas are then verified through impedance measurements in section four. Finite Element (FEM) simulations are described in section three and used to investigate the magnetic fields. These are verified via near field measurements in section five. Section six list pros and cons of the different structures and finally section seven concludes the paper.

II. PARAMETER CALCULATIONS

Equations for the inductance and ESR of the three types of inductors are given in the following subsections. Equations for the capacitance and resonance frequency are not given, as accurate formulas are not available and derivation of these is outside the scope of this paper.

The inductors are evaluated for use in VHF SMPS and for this purpose the inductance will usually not be more than maximum a few hundred mega hertz, the size small and the frequency up to around 100MHz. In this case the capacitance is relatively small, hence the self-resonance frequency becomes high and exact knowledge of these are hence not crucial.

A. Spirals

The spirals are designed as archimedean spirals, i.e. with constant trace width and a constant distance between the traces. The length of the trace in an arcimedean spiral can be calculated according to eq. 1 with the assumption that $1+2+3+\ldots+N=N\cdot\frac{N+1}{2}$

$$l_{TRACE} = \pi \cdot N_{TURNS} \frac{D+d}{2} = \pi \frac{D^2 - d^2}{4 \cdot T_{TRACE}} \qquad (1)$$

Here N_{TURNS} is the number of turns, T_{TRACE} is the thickness of the copper and D and d is the outer and inner diameter, respectively. MATLAB has been used to design the inductors, here the function pdist2 can be used to get a more accurate length when the coordinates for the trace are known.

Now that trace length is know the dc resistance is simply calculated from eq. 2. If a two layer spiral is made, the resistance of course has to be doubled and the resistance of the via has to be added. The ac resistance, caused by the skin effect, of a planar inductor can, according to [16], be approximated by eq. 3.

$$R_{DC} = \frac{\rho \cdot l_{TRACE}}{W_{TRACE} \cdot T_{TRACE}}$$
(2)

$$R_{AC} = \frac{\rho \cdot l_{TRACE}}{W_{TRACE} \cdot \delta (1 - e^{T_{TRACE}/\delta})}$$
(3)

Here ρ is the electrical resistivity of copper, W_{TRACE} is the width of a trace and δ is the skin depth.

The inductance of a spiral can be calculated by eq. 4 [17]. Here c_1 and c_2 are constants of 2.46 and 0.20, respectively, and p is a fill ratio defined as $p = \frac{D-d}{D+d}$.

$$L = \frac{\mu \cdot N_{TURNS}^2 \frac{D+d}{2}}{2} \left(ln \left(\frac{c_1}{p} \right) + c_2 \cdot p^2 \right)$$
(4)

B. Solenoids

In order to calculate the resistance of a solenoid, the structure is split in two; the vias and the traces. The dc resistance of a single via and a single trace, can now be calculated according to eq. 5 and 6, respectively.

$$R_{DC,VIA} = \frac{\rho \cdot T_{PCB}}{A_{VIA}} = \frac{\rho \cdot T_{PCB}}{\pi \cdot T_{VIA}(D_{VIA} - T_{VIA})}$$
(5)

$$R_{DC,TRACE} = \frac{\rho \cdot W_L}{W_{TRACE} \cdot T_{TRACE}} \tag{6}$$

Here T_{PCB} is the thickness of the PCB, A_{VIA} is the area of a cross section of a via, D_{VIA} is the via diameter, T_{VIA} is the thickness of the copper in a via and W_L is the width of the inductor.

The total dc resistance can then be calculated as the sum of all the vias and traces. Using multiple vias in parallel for each turn will lower the total resistance. If this is done the total resistance can be calculated according to eq. 7. Here N_{VIAS} is the number of parallel vias in a single turn.

$$R_{DC} = 2 \cdot N_{TURNS} \left(\frac{R_{DC,VIA}}{N_{VIAS}} + R_{DC,TRACE} \right)$$
(7)

The ac resistance is calculated as a first order approximation of the skin effect, hence neglecting the proximity effect. This will not give exact results, but it ease the comparison of the three structures and is sufficient for evaluation of pros and cons of each structure.

It is assumed that the ac current will run along the inner edge of the solenoid, hence along the inner edge of the traces and the inner edge of the inner half of the vias. This assumption will be verified through FEM simulations. Using this assumption and the skin depth, the ac resistance is calculated according to eq. 8.

$$R_{AC} = 2 \cdot N_{TURNS} \left(\frac{R_{AC,VIA}}{N_{VIAS}} + R_{AC,TRACE} \right)$$
(8)

Where:

$$R_{AC,VIA} = \frac{\rho \cdot T_{PCB}}{A_{VIA}} = \frac{2 \cdot \rho \cdot T_{PCB}}{\pi \cdot \delta(D_{VIA} - \delta)}$$
$$R_{AC,TRACE} = \frac{\rho \cdot W_L}{W_{TRACE} \cdot \delta}$$

The inductance of the solenoid can be calculated according to Niwas formula [18]. In eq. 9, L_L is the length of the inductor and g is the square of the hypotenuse of a cross section of the inductor given by; $g^2 = W_L^2 + T_{PCB}^2$

$$\begin{split} L &= 8e^{-9} \cdot N_{TURNS}^2 \frac{W_L \cdot T_{PCB}}{L_L} \left[\frac{1}{2} \frac{L_L}{T_{PCB}} \sinh^{-1} \frac{T_{PCB}}{L_L} \right. \\ &+ \frac{1}{2} \frac{L_L}{W_L} \sinh^{-1} \frac{T_{PCB}}{L_L} \\ &- \frac{1}{2} \left(1 - \frac{T_{PCB}^2}{L_L^2} \right) \frac{L_L}{T_{PCB}} \sinh^{-1} \frac{W_L}{L_L \sqrt{1 + T_{PCB}^2 / L_L^2}} \\ &- \frac{1}{2} \left(1 - \frac{W_L^2}{L_L^2} \right) \frac{L_L}{W_L} \sinh^{-1} \frac{T_{PCB}}{L_L \sqrt{1 + W_L^2 / L_L^2}} \\ &- \frac{1}{2} \frac{T_{PCB}}{L_L} \sinh^{-1} \frac{W_L}{T_{PCB}} - \frac{1}{2} \frac{W_L}{U_L} \sinh^{-1} \frac{T_{PCB}}{W_L} \\ &+ \frac{\pi}{2} - \tan^{-1} \frac{W_L \cdot T_{PCB}}{L_L^2 \sqrt{1 + g^2 / L_L^2}} \\ &+ \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{g^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{g^2}{L_L^2} \right) + \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \\ &- \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{W_L^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{W_L^2}{L_L^2} \right) \\ &- \frac{1}{3} \frac{L_L^2}{W_L \cdot T_{PCB}} \sqrt{1 + \frac{T_{PCB}^2}{L_L^2}} \left(1 - \frac{1}{2} \frac{T_{PCB}^2}{L_L^2} \right) \\ &+ \frac{1}{6} \frac{L_L^2}{W_L \cdot T_{PCB}} \left(\frac{g^2 - W_L^2 - T_{PCB}^2}{L_L^2} \right) \right] \end{split}$$

C. Toroids

A toroid is simply a solenoid bend in a circle, hence the same procedure and some of the assumption can be used again. It is thus assumed that the current run along the inner edges and the toroid is split into slabs and vias to get the resistance. The dc resistance of a slab is determined by integration from the inner radius to the outer radius [14].

$$R_{DC,SLAB} = \int_{r_{I}}^{r_{O}} \frac{\rho}{T_{TRACE} \left(\frac{2 \cdot \pi \cdot r}{N_{TURNS}} - C_{TRACE}\right)}$$
$$= \frac{\rho \cdot N_{TURNS}}{2 \cdot \pi \cdot T_{TRACE}} ln \left(\frac{2 \cdot \pi \cdot r_{O} - C_{TRACE} \cdot N_{TURNS}}{2 \cdot \pi \cdot r_{I} - C_{TRACE} \cdot N_{TURNS}}\right)$$
(10)

Here C_{TRACE} is the clearance between to slabs and r_I and r_O are the inner and outer radius, respectively. As for the solenoid a first order approximation of the skin effect is used for the ac resistance, hence this simply becomes:

$$R_{AC,SLAB} = \int_{r_{I}}^{r_{O}} \frac{\rho}{\delta\left(\frac{2\cdot\pi\cdot r}{N_{TURNS}} - C_{TRACE}\right)}$$
$$= \frac{\rho \cdot N_{TURNS}}{2\cdot\pi\cdot\delta} ln\left(\frac{2\cdot\pi\cdot r_{O} - C_{TRACE} \cdot N_{TURNS}}{2\cdot\pi\cdot r_{I} - C_{TRACE} \cdot N_{TURNS}}\right)$$
(11)

The resistance of the vias are the same as for the solenoid and the total resistance becomes:

$$R_{DC} = N_{TURNS} \left(2 \cdot R_{DC,SLAB} + \frac{R_{DC,VIA}}{N_{INNER}} + \frac{R_{DC,VIA}}{N_{OUTER}} \right)$$
(12)

$$R_{AC} = N_{TURNS} \left(2 \cdot R_{AC,SLAB} + \frac{R_{AC,VIA}}{N_{INNER}} + \frac{R_{AC,VIA}}{N_{OUTER}} \right)$$
(13)

Here N_{INNER} and N_{OUTER} are the number of inner and outer vias, respectively. The inductance of a toroid is derived in [15] and repeated in eq. 14.

$$L = \frac{\mu \cdot N_{TURNS}^2 \cdot T_{PCB}}{2 \cdot \pi} ln\left(\frac{r_O}{r_I}\right) \dots + \frac{r_O + r_I}{2} \mu \left[ln\left(8\frac{r_O + r_I}{r_O - r_I}\right) - 2\right]$$
(14)

An example of the calculated impedances of the three structures is given in table I. The table shows that the solenoid has almost twice the ac resistance of the spiral and 20% more than the toroid. It should however be noted that a quadratic solenoid is far from optimal and higher Q could be obtained if a more narrow solenoid was designed. A 6mm wide solenoid with 8 turns can have 86nH and a Q factor of 92 in $48mm^2$.

From the formulas for the inductance of each of the three structures, it can be seen that each of the three structures has a limited number of parameters which can be adjusted if the PCB parameters are given. The number of turns is the only common parameter for the three structures.

The only parameter left for the spiral, is the width of the trace (and the number of layers or inductors in series).

 TABLE I

 PARAMETERS FOR THREE INDUCTORS WITH APPROX. 82NH INDUCTANCE

 IN 8x8mm

Structure	Spiral	Solenoid	Toroid
N _{TURNS}	3.2	6	16
L	81 nH	81 nH	85 nH
R_{DC}	$65 m\Omega$	$72 m\Omega$	$100 \ m\Omega$
R_{AC} @ 50MHz	$253 m\Omega$	495 $m\Omega$	$392 m\Omega$
Q @ 50MHz	100	51	68

For the solenoid both the length and width can be adjusted and though it is generally best to avoid the extremes, i.e. long and narrow or short and wide, this gives a larger amount of design flexibility.

For the toroid it is possible to adjust both the inner and outer radius. For a small structure, the minimum inner radius will however be limited by the size of a via (incl. annular ring) and the number of turns. If the inner radius is increased, it is possible to ad more vias in parallel to reduce the resistance. The core of the inductor is then decreased and more turns are needed to keep the same inductance, thus the gain of doing so is limited.

III. FINITE ELEMENT SIMULATIONS

FEM simulations has been used for several reasons. First of all it is a good way to investigate, if the assumptions made about the current distribution in order to calculate the ac resistance holds. Secondly the inductance and resistance can be extracted and thereby be used to confirm the accuracy of the formulas. Lastly the simulations can be used to investigate the magnetic field from each of the three structures, so possible EMI problems can be included in the comparison.

When the formulas for the ac resistance of the spiral was derived, it was assumed that the current would run along the surface of the spiral and the resistance caused by proximity effect was neglected. A FEM simulation of a four turns spiral has been made and the current distribution is shown in figure 5 and the magnetic flux density is shown in figure 2(a). The simulation shows how the increasing strength of the magnetic



Fig. 5. Cross sections of a four turns spiral inductor showing how the current distribution (A/m^2) changes as the centre is approached







(c) Toroid

Fig. 1. Three types of PCB embedded inductors



Fig. 2. Finite element simulations of the magnetic flux density (B-field [T]) inside the three types of PCB embedded inductors



(a) Spiral



Fig. 3. Finite element simulations of the magnetic field (H-field [A/m]) 2mm above the three types of PCB embedded inductors



Fig. 4. Tangential near-field scans on three types of PCB embedded inductors. The color indicates the field strength of the combined x and y component, i.e. not including the z component as in the simulation



Fig. 6. The current distribution (A/m^2) of turn 1-3 of a six turns solenoid, the bottom of the traces and the top of the vias face the centre of the structure

field towards the centre causes the current to crowd along the edges, this leads to higher resistance than given by eq. 3.

From the simulation it is also seen that the magnetic flux is concentrated at the centre and not shielded by the structure in any way. The fact that the field is not shielded is clear in figure 3(a) where the magnetic field 2mm above the structure is plotted. It can be seen how there is a strong field just above the structure, but also that the field decreases fast away from the centre.

Similar simulations has been made for a six turns solenoid (figure 6, 2(b) and 3(b)) and a 16 turns toroid (figure 7, 2(c) and 3(c)). The three simulated inductors has been to designed to have equal inductances in equal areas, hence the varying number of turns.

From the simulations of the solenoid, it is seen that the assumption that the current would be running along the outer surface of the inner half of the vias holds. The angle between the centre of the current and the centre of the structure is changed a bit due to the magnetic flux around the ends of the solenoid, but the area conducting the current is still the same.

On the traces the simulation shows that current is running along the surface towards the centre as expected, but also along the edges due to the magnetic field.

As the solenoid and toroid are very similar, the current distribution found in the simulation of the toroid is very similar to those found for the solenoid. The major difference is that the toroid has a even magnetic field around each turn and hence there is no change in the distribution of current between the turns.

The simulated toroid had a single via on the inner side of a turn and two parallel vias on the outer side. Figure 7 shows that the assumption about the current distribution in the vias still holds, as the current is still running in the inner surface of the via.

Contrary to the spiral, both the solenoid and the toroid structure incapsulates the magnetic field. In the solenoid all the magnetic flux goes through the centre of the structure and returns along the outer edges of the structure. This results in a much weaker field outside the structure than seen for the spiral, however the return path outside the structure can be long and hence a weak but wide field is seen in figure 3(b).



Fig. 7. Current distribution (A/m^2) in a turn of a toroid inductor

The toroid has a much weaker external magnetic field as the field caused by the turns will be fully incapsulated by the structure. Only the small single turn field (which resembles the shape of the field of the spiral) is outside the structure. This leads to a very weak and narrow external field as seen in figure 3(c).

IV. MEASUREMENTS

More than 50 different inductors with each structure have been made and measured, in order to evaluate the accuracy of the formulas for resistance and inductance. As the impedance of these inductors are very small and the frequencies at which they need to be measured are very high, the measurements has to be done with very high accuracy [15].

Initial measurement of the inductance and ac resistance at 50MHz where made with a HP 4195A. The measured values fit for most of the inductors with around 10-20% deviation. However especially for the small inductors the measured values are up to 50% lower than the calculated.

An Agilent 4294A, which has an error of less than 3% for the desired measurements, has been used for the final measurements in order to get more accurate results. The measured values fit very well with the calculated values and the results for several of the prototypes is shown in figure 8.

The measured inductance and ac resistance of the largest spirals is higher than calculated. This is not surprising as the inductance of the spiral width a thin trace becomes very large in the last measurements, hence the resonance frequency starts to approach 50MHz even though the parasitic capacitance is small (see figure 9). For the spirals with thicker traces, the parasitic capacitance is larger and the resonance frequency is therefore lower.

The measurements of the solenoid impedance fits very well for both the inductance and ac resistance. The dc resistance however deviates a lot, surprisingly the inductors with a low number of turns has too high resistance whereas the inductors with high number of turns has too low resistance. The reason for the big jump between the inductor with less than 9 and more than 13 turns, can be due to the fact that they are from two different production panels. Both orders were placed for $35\mu m$ copper, but if $18\mu m$ is used for the calculation



Fig. 8. Measurements of the inductance and dc and ac resistance of some of the experimental prototypes

of the ones with low number of turns and $50\mu m$ is used for the rest the deviation from the measurements are less than 1%. This shows that the tolerances of the manufacturing process is crucial to get exact results for the dc resistance, but the inductance and ac resistance are much less sensitive. As the inductance and ac resistance will be the most important parameters for use in VHF SMPS, this should not cause problems in implementations.

The calculated values of the inductance and resistance of the toroids generally fits very well with measurements. There is a small deviation in the dc resistance for the toroid with a radius of 3mm and 19 turns, but the deviation is still within 10%. The measured ac resistance of the big toroids is up to 20% higher than the calculated values. The reason for this is not clear, but it could be due to an uneven distribution of the current in the vias when there are many in parallel.

Figure 9 shows a frequency sweep of the three structures, both of prototypes with a low number of turns and of ones

with a high number of turns. The inductance is very close to constant across the entire frequency range, except for the big spiral which is close to the resonance at the higher frequencies.

From the figure it can be seen that the resistance of the spiral starts to increase due to the skin effect much earlier than the two other structures. This is because the resistance of the spiral is almost entirely due to the resistance of the trace and here the skin effect starts at 1-5MHz ($\delta = 65 - 30\mu m$). The resistance of the solenoid and the toroid only start to increase at approx. 10MHz where the skin depth equals the thickness of the vias ($20\mu m$). At 50MHz, where the ac resistance used for figure 8 has been measured, the skin depth is approx. $10\mu m$ and all the structures are clearly affected by the skin depth.

V. NEAR FIELD

The importance of controlling the magnetic near-fields cannot be understated. Since one of the greatest advantages of switching in the VHF frequency range is the size of the



Fig. 9. Inductance and series resistance of prototypes with a low number of turns (solid) and high number of turns (dashed)

SMPS, it will in most cases be integrated in products with tight space requirements. Thus other metallic structures will be nearby. If the magnetic near-field is too strong or too widely spread it will couple to the structure and generate a current on this, which will then be re-radiated. This would make it almost impossible for the product to pass EMC requirements.

Therefore the magnetic near-fields have also been measured with a setup as described in [19]. The setup can measure both amplitude and phase of the magnetic near-field. It consist of an oscilloscope measuring the time domain response from a reference probe and a scan probe. The reference probe is placed below the circuit in a stationary position. The scan probe is positioned in each measurement point on a x-y grid surface over the structure. In the measurements the probe measures the field in 2 mm distance above the structure and with a grid step size in both the x and y direction of 2 mm.

When comparing the simulated and measured magnetic near-field, it is important to note that the z component of the fields has not been measured. From the simulations we expect that the strongest external field from both the spiral and the toroid, will be a vertical field through the centre of the structure (i.e. the z component). In the measurements it therefore seems like the field have a local minimum in the centre of the spiral and toroid and at the ends of the solenoid, but this is not the case.

The shape of the measured magnetic fields is the same as seen in the FEM simulations. The spiral has the strongest external field of the three structures and the toroid the weakest. Both the spiral and toroid has their external fields located right above the structure, whereas the solonoids field extends far outside the structure.

VI. COMPARISON

The paper has evaluated the impedances and the field of the structures. A very important parameter is however the



Fig. 10. The experimental setup for the near field scans

flexibility or design freedom, both when scaling the inductor and laying out the complete PCB.

The spiral is by far superior in this perspective. It can be made in a single layer, the width of the trace can be scaled to get the desired relation between size and Q factor and finally it does not need and integer number of turns, which makes it possible to get the terminals placed in a way that fits into the rest of the circuit.

The solenoid also has some benefits over the two other structures. It has a rectangular shape and it has both the terminals on the same layer. The rectangular shape is much easier to align together with other rectangular components on a board which most often also is rectangular. The ability to adjust the width and length and thereby design it to fit perfectly into the circuit is also very convenient.

The toroid is clearly the least flexible as the connectors are fixed to be right above each other. Furthermore it is the structure that can achieve the lowest inductance pr area and both of the two other structures can be design with lower dc and ac resistances for a given inductance in the same area.

The only real advantage of the toroid is the small external field. As the possible EMI problems of VHF SMPS are yet to be investigated in detail, it is not clear if the compensate for this larger area and/or lower Q factor.

Several aspects of the three PCB inductors have now been evaluated and the pros and cons has been summed up in table II.

The three converters in figure 11 has been designed to give and idea of how the complete circuit with PCB embedded inductors will behave. Embedding the inductors in the PCB clearly gives some benefits, i.e. a much lower profile and a higher power density [21]. However the design still has to be optimised in order to compete with discrete air core solenoids which can have Q factors of more than 200 [20].

TABLE II PROS AND CONS OF THE THREE STRUCTURES

Structure	Pros	Cons
Toroid	Smallest external field	Least flexible
		Lowest Q and L pr area
Solenoid	Rectangular shape	Wide external field
	High Q	
Spiral	Highest Q	Strong external field
	Most flexible	



Fig. 11. Resonant converters with the different structures for PCB embedded inductors

VII. CONCLUSION

This paper has given the first direct comparison between three different structures for PCB embedded inductors. The inductors have been compared with regard to resistance, inductance, magnetic field and design flexibility, both theoretically and experimentally.

The results show that there are huge differences in what can be achieved with the different structure and that a single structure cannot be selected as the superior for all applications. The spiral can achieve the highest inductance pr area and is the only structure that can be made in a single layer; the solenoid is easy to scale and can be designed to fit perfectly into a complete design and the toroid is by far superior with regard to the external magnetic field.

Future research within PCB embedded inductors should include derivation of formulas to calculate the parasitic capacitance and thereby also the resonance frequency, as these will become important for designs where a large structure is needed in order to get a higher Q factor.

Another way to increase the Q factor is to investigate how the design of each structure can be optimised to reduce the resistance without increasing the size.

Detailed investigation of the EMI from VHF SMPS is also needed. If it turns out that this is a problem, structures surrounding the inductor to encapsulate the field could be a solution.

References

 Peterson, R.B., Switch Mode Power Supplies - A New Approach for Consumer Audio Systems, Broadcast and Television Receivers, IEEE Transactions on , vol.BTR-19, no.4, pp.263-268, Nov. 1973

- [2] ON Semiconductor, 1.2A, 2MHz Automotive Buck Switching Regulator, NCV890100 - Datasheet, Rev. 2, February 2013
- [3] Jingying Hu; Sagneri, A.D.; Rivas, J.M.; Yehui Han; Davis, S.M.; Perreault, D.J., *High-Frequency Resonant SEPIC Converter With Wide Input and Output Voltage Ranges*, Power Electronics, IEEE Transactions on , vol.27, no.1, pp.189-200, Jan. 2012
- [4] Rivas, J.M.; Leitermann, O.; Yehui Han; Perreault, D.J., A very high frequency dc-dc converter based on a class 2 resonant inverter, Power Electronics Specialists Conference, 2008. PESC 2008. IEEE, vol., no., pp.1657-1666, 15-19 June 2008
- [5] Perreault, D.J.; Jingying Hu; Rivas, J.M.; Yehui Han; Leitermann, O.; Pilawa-Podgurski, R.C.N.; Sagneri, A.; Sullivan, C.R., *Opportunities and Challenges in Very High Frequency Power Conversion*, Applied Power Electronics Conference and Exposition, 2009. APEC 2009. Twenty-Fourth Annual IEEE, vol., no., pp.1-14, 15-19 Feb. 2009
- [6] Orlandi, S.; Allongue, B.A.; Blanchot, G.; Buso, S.; Faccio, F.; Fuentes, C.A.; Kayal, M.; Michelis, S.; Spiazzi, G., *Optimization of Shielded PCB Air-Core Toroids for High-Efficiency DCDC Converters*, Power Electronics, IEEE Transactions on , vol.26, no.7, pp.1837-1846, July 2011
- [7] Dong Hui; Zhu Yisheng; Zhao Baishan, *Research on the electromagnetic radiation of a PCB planar inductor*, Microwave Conference Proceedings, 2005. APMC 2005. Asia-Pacific Conference Proceedings, vol.1, no., pp. 3 pp., 4-7 Dec. 2005
- [8] Liangliang Li; Dok Won Lee; Kyu-Pyung Hwang; Yongki Min; Hizume, T.; Tanaka, M.; Ming Mao; Schneider, T.; Bubber, R.; Wang, S.X., *Small-Resistance and High-Quality-Factor Magnetic Integrated Inductors on PCB*, Advanced Packaging, IEEE Transactions on , vol.32, no.4, pp.780-787, Nov. 2009
- [9] Dong Hui; Zhu Yi-sheng; Zhao Bai-shan, Analysis of electromagnetic scattering from a PCB inductor based on wavelet-Mom, Microwave and Millimeter Wave Technology, 2008. ICMMT 2008. International Conference on , vol.2, no., pp.544-547, 21-24 April 2008
- [10] Zolog, M.; Pitica, D.; Pop, O., Characterization of Spiral Planar Inductors Built on Printed Circuit Boards, Electronics Technology, 30th International Spring Seminar on , vol., no., pp.308-313, 9-13 May 2007
- [11] Cheon, S.J.; Park, W.C.; Park, J.Y., Highly performed PCB embedded inductors using inverted-patterned ground structure, Microwave Conference (EuMC), 2010 European, vol., no., pp.312-315, 28-30 Sept. 2010
- [12] Kamali-Sarvestani, R.; Williams, J.D., Fabrication of high quality factor RF-resonator using embedded inductor and via capacitor, IECON 2010 - 36th Annual Conference on IEEE Industrial Electronics Society, vol., no., pp.2283-2287, 7-10 Nov. 2010
- [13] Kamali-Sarvestani, R.; Williams, J.D., Fabrication of high quality factor RF-solenoids using via structures, Wireless and Microwave Technology Conference (WAMICON), 2011 IEEE 12th Annual, vol., no., pp.1-4, 18-19 April 2011
- [14] Kamby, Peter; Knott, Arnold; Andersen, Michael A.E., Printed circuit board integrated toroidal radio frequency inductors, IECON 2012 - 38th Annual Conference on IEEE Industrial Electronics Society, vol., no., pp.680-684, 25-28 Oct. 2012
- [15] Sullivan, C.R.; Weidong Li; Prabhakaran, S.; Shanshan Lu, Design and Fabrication of Low-Loss Toroidal Air-Core Inductors, Power Electronics Specialists Conference, 2007. PESC 2007. IEEE, vol., no., pp.1754-1759, 17-21 June 2007
- [16] Thomas H. Lee, *The Design of CMOS Radio-Frequency Integrated Circuits*, Cambridge University Press, 2003, ISBN-13: 978-0-521-83539-8, pp. 50-51
- [17] Mohan, S.S.; del Mar Hershenson, M.; Boyd, S.P.; Lee, T.H., Simple accurate expressions for planar spiral inductances, Solid-State Circuits, IEEE Journal of, vol.34, no.10, pp.1419-1424, Oct 1999
- [18] Frederick W. Grover, Inductance Calculations: Working Formulas and Tables, Courier Dover Publications, 2004, ISBN-13: 978-0-486-47440-3, pp. 70-71
- [19] Mynster, A.P.; Sorensen, M., Validation of EMC near-field scanning amplitude and phase measurement data, Electromagnetic Compatibility (EMC EUROPE), 2012 International Symposium on , vol., no., pp.1,6, 17-21 Sept. 2012
- [20] Coilcraft, High Frequency, High Current Power Inductors, 2014VS -Datasheet, Revised 23th January 2013
- [21] Madsen, M.; Knott, A.; Andersen, M., Very High Frequency Resonant DC/DC Converters for LED Lighting, Applied Power Electronics Conference and Exposition, 2013. APEC 2013, unpublished.

A.14 M. Madsen, J. D. Mønster, A. Knott, M. A.E. Andersen:
"Design Optimization of Printed Circuit Board Embedded Inductors through Genetic Algorithms with verification by COMSOL", COMSOL conference 2013, Boston, MA, October 2013. Conference Proceedings.

Design Optimization of Printed Circuit Board Embedded Inductors through Genetic Algorithms with Verification by COMSOL

Mickey P. Madsen^{*}, Jakob D. Mønster, Arnold Knott and Michael A.E. Andersen Technical University of Denmark, Department of Electrical Engineering, Electronics *Corresponding author: Oersteds Plads 349, Kgs. Lyngby, 2800, Denmark, mpma@elektro.dtu.dk

Abstract: This paper describes the implementation of a complete design tool for design, analysis, optimization and production of PCB embedded inductors.

The papers shows how the LiveLink between MATLAB and COMSOL makes it possible to combine the scripting and calculation power of MATLAB with the simulation power of COMSOL in order to get an extremely efficient tool for inductor design.

The tool has been used to investigate PCB embedded spiral, solenoid and toroidal inductors. Due to the fact that the spirals are axisymmetric they can be simulated in 2D, which speeds up the simulation significantly. This is not possible for the solenoid and toroid, hence complete 3D structure has been made and simulated for these structures.

Keywords:	Inducto	or,	Resistance,
Optimization,	LiveLink	to	MATLAB,
axisvmmetric.			

1. Introduction

Passive energy storing elements constitute the major part of modern switch-mode power supplies (SMPS), both regarding size and price. The physical and numerical size of these, scales directly with the inverse of the switching frequency. The constant strive for smaller and cheaper power supplies has therefore let to constantly increasing switching frequencies ever since the development of the SMPS in the early 70's.

Recent research combines radio frequency (RF) circuits and power electronics to design SMPSs with switching frequencies in the Very High Frequency range (VHF, 30-300MHz) [1]-[2]. This is a dramatic increase from the .1-5 MHz which traditional SMPS topologies are limited to due to switching losses.

Several benefits arise from the increased switching frequency; higher power density, decreased cost, reduced weight and faster transient response [3]-[4]. The increased frequency however also leads to some new challenges [5]-[6]. One of these challenges is the magnetic components.

The core materials that are normally used to get high coupling in transformers and high inductance in inductors have high core losses at these frequencies. New core materials or ways of designing the magnetic components therefore has to be developed.

The inductance needed at these frequencies are very low (~10-200nH) and the skin depth is 4-12 μ m. Hence embedding these as air core inductors in the Printed Circuit Board (PCB), and thereby reduce the size and price of these even further, becomes a viable solution [7]-[8].

In order to keep the efficiency high at these highly elevated frequencies soft switching is essential, as hard switching would lead to breakdown of the semiconductors. A key parameter to achieve this is accurate inductance value. Formulas for these are all based on approximations leading to small deviations. As the inductance cannot be fine tuned once the PCB has been manufactured, Finite Element Analysis (FEA) becomes a very attractive way to insure that the circuit is correctly designed the first time.

2. System level

In order to design and produce the inductors a MATLAB program with a GUI has been designed, see figure 1. The program takes the production parameters and the desired inductance and size as input and uses a genetic algorithm and formulas to find parameters for the initial design.

Then COMSOL is used to investigate the current distribution and magnetic fields and to verify the inductance and resistance.

Once the design has been verified in COMSOL and any error has been corrected, a PCB file is generated directly from the same interface.

3. MATLAB optimization

The optimization routine is used to find the best inductor, within a given area. Best is defined



Figure 1. GUI of the MATLAB program, with the extracted simulation results from COMSOL via LiveLink for MATLAB.

as the required inductance the user needs, with the highest Q factor and with the lowest area, in that preferred order. The optimization routine is based on a genetic algorithm already implemented in MATLAB. The optimization parameters are the desired inductance, a maximum area and the Q factor. The Q factor is the relationship between the inductance and resistance, given by:

$$Q = \frac{\omega L}{R}$$

The inductor should have the highest possible Q factor and the desired inductance. This is achieved by having a cost function that the genetic algorithm minimizes. The genetic algorithm takes any number of inputs, with possibility to define the variable as an integer. Each variable is setup so a closer value to the desired will lower the cost function, and each variable is weighted according to the importance of that specific parameter. The cost function used is:

$$Cost = a \cdot \left| \frac{L}{L_{desired}} - 1 \right| + b \cdot \frac{1}{Q} + c \cdot A$$

where $L_{desired}$ is the desired inductance and A is the area of the inductor. The weight factors used are shown in Table 1.

Table 1. The weight factors used for the cost function.

а	b	С
100	10,000	1

The overall MATLAB program with the optimization routine is described by the flowchart shown in Figure 2.

4. COMSOL:

To verify that the calculations used to derive the optimum structure and evaluate that the formulas in overall are usable the MATLAB program is used to interface to COMSOL via COMSOL LiveLink for MATLAB [9].

The AC/DC module in COMSOL is used to simulate the three possible inductor structures. The AC/DC module gives possibility for extracting the inductance and resistance, which

are the desired parameters [10]. A precondition for using the AC/DC module is that the length of the structure l, is much less than the wavelength of the frequency at which the system is simulated. That is:

$$l < 1/10 \cdot \lambda$$

At VHF frequencies the wavelength is still in the range of a few meters, why the AC/DC module is a viable approximation to the system, as the structures are very small.



Figure 2. Flowchart of the MATLAB program, with the optimization routine and the COMSOL routine

4.1 Design steps

The COMSOL model is set up in five steps. Each are described here in detail. The program can simulate three different structures: A solenoid, a toroid and a spiral. The spiral is simulated in 2D to achieve faster simulation time. Examples of the three different structures are shown in Figure 3-5.

LiveLink for MATLAB gives access to all the normal features of COMSOL with the scripting capabilities of MATLAB. In general five steps are used in the interface.

4.2 Step 1 – Generating the structure

The structures are made by implementing the layers of the PCB in a z-shifted plane, and then extruding them. Cylinders are added as vias, and then subtracting smaller cylinders inside the other for the holes in the vias. This gives a very exact representation of the full 3D inductor structures [11]. There is a short loop for the 2 connectors in the inductors, and between is made a gap. This is used when setting up the physics. At last the entire inductor is made into a union. During the generation of the structure, several selections are made, both with boxes and explicit. These are automatically updated if more structures are added in the process.

The spiral is only modeled in an axisymmetric 2D model with infinite elements [12] to lower the simulation time. The two other structures are (unfortunately) not possible to split into an symmetry.

The PCB's dielectric material is not added to the model, as it does not influence the inductances and resistance of structure, and only will increase simulation time. If the parasitic capacitance or resonance frequency of the inductors is to be extracted from the simulation, this has to be added as it do influences the capacitance.



Figure 3. Example of a solenoid inductor in 3D.



Figure 4. Example of a toroid inductor in 3D.



Figure 5. Example of a spiral inductor in 2D.

4.3 Step 2 – Setting up the physics

The physics is based on the Magnetic Fields (MF) with a frequency domain study. The physic is set to work at the entire system expect the gap created between the terminals. This will introduce a small error but smaller than if the connectors were taken to the exterior of the bounding box. The surroundings are modeled as air, and the structure as copper. As the magnetic field is not guided by a core, there is added an infinite element to ensure the entire field is simulated within the bounding box's boundaries.

For exciting the inductor a "Single-turn coil" is used [13], with a voltage input. Furthermore the physics is set to linear, as this decreases the simulation time dramatically, and does not change the results.

4.4 Step 3 - Meshing

As there is great difference between the overall size of the structure and the holes in the vias and the height of the copper traces, which have an even smaller skin-depth, the mesh must be customized to the structure. The structure is split in three. The inductor structure uses "Free Tetrahedrals" with the size set to "Extra fine". Boundary layers is used to get the correct current density in the copper tracks,. The air is also meshed with "Free Tetrahedrals", but with the size set to "Fine". The infinity domain is swept with a simple distribution of 3 elements. This configuration gives a mesh quality of approximately 0.7.

4.5 Step 4 – Study setup

The study is set up to use a frequency domain. The standard stationary solver is changed to use a SOR solver instead of the Multigrid, as the system is set to linear.

4.6 Step 5 – Extracting parameters

The inductance and ac-resistance are added to a table from a global evaluation. The table is then imported into MATLAB together with a plot of the magnetic field with the LiveLink for MATLAB functions "mphtable" and "mphplot".

5. Verification

In order to verify the formulas used to calculate the inductance and resistance and to ensure that the implementation of the models in COMSOL is a good approximation a comparison of calculated, simulated and measured values are

performed. The extensive comparison is only made for spirals, as the simulation time if done for solenoids and toroids would accumulate to weeks.

The simulations are shown for two set of spiral structures: one with one layer and a 150um trace shown in Figure 6 and one set with a 2 layer structure and 300um trace in Figure 7. The number of turns is swept from 1 to 15 and 1 to 10 respectively.

The comparison shows that there in general is a good agreement between calculated, simulated



Figure 7. Inductance and resistance of two layer spirals with 300µm trace width.

and measured inductances, but with higher accuracy for the simulated values.

The same is the case for the ac-resistance. In both sets there is a large increase in the measured resistance when the structures become large. This is the start of the first resonance, and this isn't part of neither the simulation nor the calculation.

In Figure 8 and 9 is shown a cutout of the magnetic field of a solenoid and a toroid. The simulation shows that solenoid and toroid structures encapsulate the magnetic field, resulting in a much weaker external field than seen for spirals. In the solenoid all the magnetic flux goes through the center of the structure and returns along the outer edges of the structure. For the toroid the major part of the field is running inside the toroidal structure and only a small part of the field (caused by the single turn) runs outside.



Figure 8. Magnitude of the B-field in a solenoid inductor (Multislice).



Figure 9. Magnitude of the B-field in a toroid inductor (Multislice).

6. Conclusion

It has been demonstrated how the LiveLink between COMSOL and MATLAB can be used to design, analyze and optimize PCB embedded inductors. The scripting and calculation power of MATLAB combined with the simulation power of COMSOL can serve as an extremely efficient tool for inductor design.

A genetic algorithm has been used to design the optimal inductors based on the available equations. Then COMSOL has been used to verify the calculated inductance and resistance and investigate the current distribution and the magnetic fields. This analysis is very useful for finding new ways to design inductors in order to make new shapes with even better performance.

Ones the inductor has been investigated and verified to give the desired impedances the PCB files can be generated directly from the same MATLAB GUI giving a very simple and yet accurate design tool.

7. References

- Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, "High-Frequency Resonant SEPIC Converter With Wide Input and Output Voltage Ranges", IEEE Transactions on Power Electronics, vol.27, no.1, pp.189-200, Jan. 2012
- M.P. Madsen, A. Knott and M.A.E. Andersen, "Very high frequency resonant DC/DC converters for LED lighting," IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.835,839, 17-21 March 2013
- Juan M. Rivas, David Jackson, Olivia Leitermann et al., "Design Considerations for Very High Frequency dc-dc Converters", IEEE Power Electronics Specialists Conference, 2006, pp.1-11.
- 4. M.P. Madsen, A. Knott and M.A.E. Andersen, "Low Power Very High Frequency Resonant Converter with High Step Down Ratio", IEEE Africon, 2013

- David J. Perreault, Jingying Hu, Juan M. Rivas et al., "Opportunities and Challenges in Very High Frequency Power Conversion", IEEE Applied Power Electronics Conference, 2009, pp.1-14.
- A. Knott, T.M. Andersen, P. Kamby, M.P. Madsen, M. Kovacevic, M.A.E. Andersen, "On the ongoing evolution of very high frequency power supplies", IEEE Applied Power Electronics Conference and Exposition 2013, vol., no., pp.2514,2519, 17-21 March 2013
- Kamby, P.; Knott, A.; Andersen, M.A.E., "Printed circuit board inte- grated toroidal radio frequency inductors", IECON 2012 -38th Annual Conference on IEEE Industrial Electronics Society, vol., no., pp.680- 684, 25-28 Oct. 2012
- M.P. Madsen, A.P. Mynster, A. Knott and M.A.E. Andersen, "Printed Circuit Board Embedded Inductors for Very High Frequency Switch-Mode Power Supplies", IEEE International Energy Conversion Congress and Exhibition Asia 2013
- 9. "LiveLink[™] for MATLAB", COMSOL, Presentation, <u>www.comsol.com</u>.
- 10. Introduction to AC/DC Module, COMSOL, Application note, <u>www.comsol.com</u>.
- 11. COMSOL model library: "Modeling of a 3D Inductor", Model ID: 10299
- 12. D. Kováč, M. Ocilka, M. Vansáč, "Calculating of Inductance of Spiral Coil Using Comsol Multiphysics", Electromechanical and energy systems, modeling and optimization methods, March 28-29, 2012, Kremenchuk. – Kremenchuk, Mykhailo Ostrohradskyi National University, 2012 P. 89-90. - ISSN 2079-5106
- COMSOL model library: "Single-turn and Multi-turn Coil Domains in 3D", Model ID: 14067

A.15 H. Schneider, J. D. Mønster, M. Madsen, T. Andersen, A. Knott, M. A.E. Andersen:
"Investigation of a Hybrid Winding Concept for Toroidal Inductors using 3D Finite Element Modeling",
COMSOL conference 2013, Boston, MA, October 2013. Conference Proceedings.

Investigation of a Hybrid Winding Concept for Toroidal Inductors using 3D Finite Element Modeling

H. Schneider^{*}, T. Andersen, J. D. Mønster, M. P. Madsen, A. Knott and M. A. E. Andersen Department of Electrical Engineering, Technical University of Denmark - DTU *Corresponding author: Oersteds Plads 349, Kgs. Lyngby, 2800, Denmark, hensc@elektro.dtu.dk

Abstract: This paper investigates a hybrid winding concept for a toroidal inductor by simulating the winding resistance as a function of frequency. The problem of predicting the resistance of a non-uniform and complex winding shape is solved using 3D Finite Element Modeling. A prototype is built and tested experimentally to verify the simulation results. Finally COMSOL LiveLink to CAD is utilized to highlight a bottleneck for this kind of winding scheme.

Keywords: Inductor, Resistance, Mesh, Boundary layers, LiveLink.

1. Introduction

The conventional wire wound toroid shown in Figure 1 is used extensively in switch mode power supplies in EMC filters and as inductors and transformers. However the space between the windings increases gradually from the inner diameter towards the outer diameter of the core which limits the utilization of the available winding space. This effect can increases the resistance and thus the conduction loss of the component [1,2].



Figure 1. Conventional toroidal inductor

Furthermore, in the manufacturing process of a toroidal inductor with a large diameter wire and a small core a hook/pull type winding machine must be utilized. Under manufacturing a tool pass through the center of the toroidal core to grab and pull the wire which is manually feed to the hook for each turn.

To overcome these disadvantages a hybrid winding scheme may be used [3]. The basic idea of the hybrid winding scheme is to cut and bend copper foils into "U" shaped pieces that fit around the toroidal core as shown in Figure 2. To complete the winding the foil pieces are connected through the traces in a printed circuit board (PCB). In a final product the foil pieces would be pre-attached to a former for easy handling and alignment on the PCB.



Figure 2. Exploded view of the hybrid winding concept for a toroidal inductor.

The hybrid winding concept has the following benefits compared to the conventional single layer wire wound inductor:

- Large scale production and distributed stock of foil assemblies reduce cost and delivery time.
- Fully automated manufacturing process. Foil pieces are stamped, bended and attached to a plastic former. The core and the foil assembly can be placed by a pick and place machine.
- Better utilization of winding space. Constant resistance with increased number of turns or constant number of turns with decreased resistance for the same core size
- Low AC resistance due to increased surface area using foil.
- Configurable winding structure. The connections of the foil pieces through the PCB can be freely defined by the design engineer. The number of turns can be decreased by paralleling the foil pieces. It enables the same foil assembly to be used for inductors, common mode EMC chokes, transformers etc.
- Different core materials can be used with the same foil assembly.

The disadvantages may be as follows:

- Commercial available PCB layer thickness and interconnections between the foil cutouts and the PCB tracks is a bottleneck for high power applications.
- The assembly of the foil cutouts is limited to a specific core size.

2. Use of COMSOL Multiphysics

The challenge of predicting the resistance as a function of the frequency of such a complex 3D winding geometry is solved using the COMSOL AC/DC module[4]. The greatest challenge was to create the mesh since eddy currents need to be considered and the thickness of the foil to the PCB traces in this work vary from 0.5mm to 70 μ m. A boundary layer mesh with 4 layers and a stretch factor of 2.5 was used. The first layer fitness was set manually to 5 μ m. A free tetrahedral mesh was used for the DC resistance simulation.

3. Results

In Figure 3 the current density at 100kHz in the foil cutouts and the PCB traces are shown. The 3D model of the hybrid inductor is shown from beneath in order to inspect the PCB traces. Some of the windings and the core are hidden in order to look at the inner part of the winding. From the colors it is easy to see that the current density is highest closest to the core where the current path is shortest and in the places where the width of the foil cutout is lowest. This suggests that the foil cutouts should be angled instead of the PCB traces in order to reduce the resistance.

The implemented prototype is shown in figure 4. The PCB were milled with a router and the foil pieces were cut with a scissor and soldered manually. It is clear that the prototype has rounded corners and unequal spacing compared to the 3D drawing shown in figure 3.

In Figure 5 the simulated and measured resistance as a function of the frequency is shown. An offset is added to the simulation result to compensate for any static error sources such as imprecise cuts and clearance of the bent foils, all of which is not included in the 3D model. The simulation with offset fits the measured results from 10 Hz to 100 kHz which is the common operating frequency range. It is eddy currents in the core that causes the increase in the measured AC-resistance beyond the 100 kHz which is not accounted for in the simulation model.

A LiveLink to a computer aided design (CAD) program [5] was also implemented and several simulations were performed in order to investigate the DC-resistance versus the PCB Thickness. The result shown in figure 6 indicates that the DC-resistance could be halfed by using 220 um thick PCB traces compared to the common 70 um PCB trace thickness. The simulation prove that the PCB trace thickness is a bottleneck for high power applications.



Figure 3. Simulated current density and magnetic flux density



Figure 4. Prototype used for validation of the simulation results



Figure 5. Resistance as a function of the frequency



Figure 6. Simulated DC-resistance as a function of PCB copper thickness. Foil thickness is fixed at 500 μ m.

4. Conclusion

A hybrid winding concept for toroids using the traces in a printed circuit board to make connection to bent copper foil cutouts has been evaluated in terms of AC-resistance and the DC-resistance as a function of PCB trace thickness.

A FEM analyses have been carried out and the model have been experimentally validated.

It is found that the commercially available layer thickness in a PCB is a bottleneck for high power applications. Finally a plot of the simulated current density in the winding reveals that the winding configuration can be optimized which is crucial for performance.

5. References

1. M. Seitz and M. Roeber, Squeeze more performance out of toroidal inductors, *Power Electronics Technology*, vol. 31, p. 30 (2005)

2. M. Nigam and C. R. Sullivan, Multi-layer folded high-frequency toroidal inductor windings, *Applied Power Electronics Conference and Exposition – APEC*, Twenty-Third Annual IEEE,, pp. 682-688 (2008)

3. H. Schneider, T. Andersen, A. Knott and M. A. E. Andersen, "Hybrid winding concept for toroids", ECCE Asia (2013)

4. Introduction to AC/DC Module, COMSOL, Application note, <u>www.comsol.com</u>.

5. CAD Import Module and LiveLink[™] for CAD V4.3b, COMSOL, Presentation, www.comsol.com.

A.16 H. Schneider, J. D. Mønster, M. Madsen, T. Andersen, A. Knott, M. A.E. Andersen:
"Optimizing Inductor Winding Geometry for Lowest DC-Resistance using LiveLink between COMSOL and MATLAB",
COMSOL conference 2013, Boston, MA, October 2013. Conference Proceedings.

Optimizing Inductor Winding Geometry for Lowest DC-Resistance using LiveLink between COMSOL and MATLAB

H. Schneider^{*}, T. Andersen, J. D. Mønster, M. P. Madsen, A. Knott and M. A. E. Andersen Department of Electrical Engineering, Technical University of Denmark - DTU *Corresponding author: Oersteds Plads 349, Kgs. Lyngby, 2800, Denmark, hensc@elektro.dtu.dk

Abstract: An optimization routine is presented to optimize a hybrid winding geometry for a toroid inductor in terms of the DC resistance. The hybrid winding geometry consist of bended foil pieces connected through traces in a printed circuit board. MATLAB is used to create a graphical user interface that visually plots the winding using input parameters such as core dimensions, number of turns, clearance between windings, and the winding angle of each segment of the winding. COMSOL LiveLink is used to import the winding geometry from MATLAB and create a 2D finite element model to simulate the DC reisistance. Finally the winding configuration with the lowest DC resistance is found by sweeping the parameters of the winding geometry and simulate and save the result in each step. An improvement of more than 30% compared to previous work where achieved in this way

Keywords: Inductor, Resistance, Mesh, Boundary layers, LiveLink.

1. Introduction

The conventional wire wound toroid is used in many power electronic applications such as EMC filters, power inductors, transformers and so on. However it has the following disadvantages regarding the winding scheme and the manufacturing/production process [1, 2]:

- The spacing between the windings increases gradually from the inner diameter towards the outer diameter of the core limiting the utilization of the winding area.
- For large wire diameters a hook/pull type manufacturing machine is used. The space required for the hook and the wire being pulled through the center of the core further decrease the utilized winding area.
- For large wire diameters manual work is required for feeding the hook with the wire leading to increased cost.

- Complex manufacturing machines are needed for winding due to non-separable core.
- Manufacturer lead time and price strongly depends on the purchase history and number of ordered magnetic components.
- The leaded toroid may be placed and soldered manually in a production increasing time to market and cost.



Figure 1. Surface mounted hybrid toroidal inductor.

This work focus on a hybrid foil combined with printed circuit board (PCB) trace winding scheme in an attempt to improve both the utilized winding space and improve the time to market by creating more freedom in the manufacturing and production process.

The basic idea is to cut and bend copper foils into "U" shaped pieces that fit around the toroidal core. The number of bended foil pieces determines the number of turns and in a final product all the foil pieces would be pre-attached to a plastic former for easy handling and alignment on the PCB.

In previous work [4] a hybrid inductor with 15 turns, a foil thickness of 500um and a single layer PCB with a layer thickness of 70um was implemented as a prototype. The measurements were used to evaluate a 3D Finite element model (FEM) created in COMSOL with good agreement. The results showed that typical available PCB layer thicknesses (18um - 210um)

are a bottle neck for the DC resistance in a hybrid inductor.

In the implemented model the bended copper pieces went straight over the core and the traces in the PCB was angled to complete the winding. It is easy to imagine how the DC resistance would be improved if the thin traces in the PCB were straight and short and the thicker copper pieces were angled over the core connecting the winding. It is however hard to predict the optimum angles of each segment in a turn and to predict the impact on the DC resistance. It is necessary to find an answer to these questions in order to take full advantage of the hybrid inductor.

2. Use of COMSOL Multiphysics

COMSOL is used to simulate and find the DC resitance of the windings in the inductor. The system is set up as a 2D simulation to improve simulation time, which is needed for the optimization algorithm solution time not to get extremely long.

The 3D model is based on four segments per turn, labeled F1-F4 as shown in Figure 2. The optimization algorithms input parameters is the coil dimensions, number of turns, minimum clearance between turns, the starting position of the winding following the tangent of the core, the angle of each segment and the thickness of each segment.



Figure 2. Geometry of the winding and the related parameters.



Figure 3. The designed MATLAB GUI with LiveLink to COMSOL.

To control the optimization rutine a MATLAB program was developed, that can take all the inputs, and set the different constraints, and in general control the optimization. For setting up the 2D structure in COMSOL the LiveLink for MATLAB [5] was used. The following steps were implemented in the MATLAB code in order to automate the calculation of the DC resistance with changing winding geometry:

- Creating a MATLAB GUI See Figure 3
- Creating the winding geometry
 - The geometry of a single turn is created based on the parameters set by the GUI for the winding such as size of the core, clearance, thickness of the foil and the angle of each segment in a turn. The coordinates positions are translated from 3D to 2D by unfolding the each turn.
 - A for loop creates the desired turns by copying the coordinates of the single turn by rotating them in a polar coordinate system. Every section of a turn is created as a polygon and the coordinates is saved for later selection of domains and boundaries. Each segmet is shifted so they lay in layers. This is done to ensure that the windings will not

cross each other, as shown in Figure 4.

- Selecting the boundaries. As the specific boundaries are not number as the are generated, it is nessary to find them in order to set up the rest of the simulation. This is done by using the LiveLink method "mphselectbox". The saved positions used to generate the turns are used and the boundary are stored for later use.
 - The material setup is defined for each segment in the simulation, instead of using the standard Copper material, this is done since the different segments can (and usaully have) different thickness of the copper. The difference is used to define an electrical conductivity for each segment modelling the thickness.
 - The physics is set up as an "Electrical Currents" model, with a stationary study, and the discretixation of the electric potential to (the standard) quadratic. To



Figure 1 Current density plot of each of the turns structures.

enforce a voltage over the entire structure a Terminal is added on the first structure (bottom structure, red boundary). Here a voltage of 1V is set. On the last (top) structure a Ground node is added. To ensure the connection betweeen each turns structure the Periodic Condition is used, between the boundaries of the end of one structure and the input of the next, as indicated by the arrows in Figure 4.

- The geometry is meshed using the standard mesh Free Triangular, with the seize set to "Normal". This gives an simple, yet accurate enough approximation.
- The study is set as a stationary standard study, with direct solving.
- The results are taken in two parts
 - \circ A global variable is used to calculate the DC resistance seen from the terminals. The conductance *G* is available directly from the solution and the DC resistance is then calculated as The current density was evaluated in the



Figure 2 Ilustration of an unfold winding which is copied and connected in series via "Periodic Condition"

$$R_{DC} = \frac{1}{6}$$

same way but as a surface maximum and minimum, as well as plotted using in COMSO, see Figure and shown in a MATLAB figure using the function "mphplot".

3. Results

The result of the optimization routine is shown in Figure 5. The plot shows the simulated resistance for a given number of solutions where the previously mentioned parameters of the winding were altered. In this specific case the resistance don't change much as a function of the winding parameters as in other cases. In this work the following 3 cases were examined:

- Case 1: Few turns, PCB layer thickness << Foil thickness
- Case 2: Many turns, PCB layer thickness << Foil thickness
- Case 3: Few turns, PCB layer thickness = Foil thickness

In table 1-3 the parameters and result for each of the 3 cases are summarized. In each case 3 winding configurations are compared. One where only the PCB trace is angled called "Bottom", one where the lowest DC-resistance were found called "Opt." for optimized and one where only the top segment of the bended foil was angled called "Top".

In Case 1 the optimized solution resulted in an improvement of 32 % compared with only angling the PCB trace. However an 31 % improvement was achieved by only angling the top segment of the bonded foil piece. If few turns and a big difference in foil and PCB trace is utilized the "Top" solution may be sufficient since it will result in a low DC-resistance and may be easier to fabricate.

In Case 2 a very small improvement of 0.8 % for both the "Opt." and the "Top" compared to the "Bottom" is achieved. The limited space due to many turns reduce the influence of angling the segments in the winding and obviate the optimization. In Case 3 the "Opt." and "Top" configuration resulted in 11 % and 6 % improvement respectively. For few turns and equal winding thickness the optimized solution is therefore attractive. This could be an important conclusion since this configuration is highly suitable for high frequency operation which is in high demand. However this must be confirmed in a 3D simulation of the AC-resistance which is out of the scope in this work.



Figure 5. Illustration of optimizing routine for a single integration. The x-axis is the combination number for the given angle span and the y-axis is corresponded DC resistance.

Table 1: Case 1 (Few turns, PCB layer thickness << Foil thickness)

Number of turns: 10, Segment thickness: F1, F2, F3 = 500 μm, F4 = 70um, Clearance = 1mm				
Winding angle configuration	Bottom	Opt.	Тор	
Starting point SP [%]	0	20	0	
Outer Foil Segment F1 [%]	0	63	0	
Top Foil Segment F2 [%]	0	37	100	
Inner Segment F3 [%]	0	0	0	
Bottom Segment F4 [%]	100	0	0	
DC resistance [mΩ]	6.78	4.64	4.71	
Improvement [%]	Ref.	32	31	

 Table 2: Case 2 (Many turns, PCB layer thickness <</th>

 Foil thickness)

Number of turns: 100, Segment thickness: F1, F2, F3 = 500 μm, F4 = 70um, Clearance = 1mm				
Winding angle configuration	Bottom	Opt.	Тор	
Starting point SP [%]	0	0	0	
Outer Foil Segment F1 [%]	0	75	0	
Top Foil Segment F2 [%]	0	25	100	
Inner Segment F3 [%]	0	0	0	
Bottom Segment F4 [%]	100	0	0	
DC resistance [mΩ]	435	432	432	
Improvement [%]	Ref.	0.8	0.8	

Table 3: Case 3 (Few turns, PCB layer thickness = Foil thickness)

Number of turns: 10, Segment thickness: F1, F2, F3, F4 = 500 μm, Clearance = 1mm				
Winding angle configuration	Bottom	Opt.	Тор	
Starting point SP [%]	0	100	0	
Outer Foil Segment F1 [%]	0	38	0	
Top Foil Segment F2 [%]	0	21	100	
Inner Segment F3 [%]	0	0.5	0	
Bottom Segment F4 [%]	100	40.5	0	
DC resistance [mΩ]	2.6	2.3	2.5	
Improvement [%]	Ref.	11	6	

4. Conclusion

A MATLAB program for optimizing the structure of an inductor in order to minimize the DC resistance have been created. The program sets up the 2D simulation of the structure dividide into turns, again dividede into 4 segments, each with different thickness. The system is simlated and the numerical as well as grapical results are extracted. The program shows that it is capacble of finding the optimum winding geometry, leading to an improved DC-resistance. The findings are in general:

- Few turns and large difference in foil and PCB trace thickness
 - Angeling the top segment (F2) is a suitable solution
- Many turns

- Less degree of freedom to alter the shape of the winding
- Same thickness of all the segments in a turn • An optimized solution is preferred

5. References

1. M. Seitz and M. Roeber, Squeeze more performance out of toroidal inductors, *Power Electronics Technology*, vol. 31, p. 30 (2005)

2. M. Nigam and C. R. Sullivan, Multi-layer folded high-frequency toroidal inductor windings, *Applied Power Electronics Conference and Exposition – APEC*, Twenty-Third Annual IEEE,, pp. 682-688 (2008)

3. H. Schneider, T. Andersen, A. Knott and M. A. E. Andersen, "Hybrid winding concept for toroids", ECCE Asia (2013)

4. Introduction to AC/DC Module, COMSOL, Application note, <u>www.comsol.com</u>.

5. CAD Import Module and LiveLink[™] for CAD V4.3b, COMSOL, Presentation, www.comsol.com.

A.17 J. D. Mønster, M. Madsen, J. A. Pedersen and A. Knott: "Investigation, development and verification of printed circuit board embedded air-core solenoid transformers", IEEE Applied Power Electronics Conference and Exposition Charlotte NC March 2015 Confer-

Exposition, Charlotte, NC, March 2015. Conference Proceedings.

Investigation, development and verification of printed circuit board embedded air-core solenoid transformers

Jakob D. Mønster, Mickey P. Madsen, Jeppe A. Pedersen and Arnold Knott

DTU Elektro Technical University of Denmark Denmark, Kongens Lyngby 2800 Email: {jdomo,mpma,jarpe,akn}@elektro.dtu.dk

Abstract—A new printed circuit board embedded air-core transformer/coupled inductor is proposed and presented. The transformer is intended for use in power converter applications operating at very high frequency between 30 MHz to 300 MHz. The transformer is based on two or more solenoid structures in different configurations. The different configurations are compared for usefulness as a transformer solution, and an analytical model of the inductive parameters for the most suitable configuration is derived for design purpose. The analytical model is verified by comparing calculations and measurements of prototypes. The analytical model shows good agreement with the measured results. The model can predict the inductive parameters of the transformer with a deviation range of approximately 3% to 22%. Lastly a prototype is used in a VHF converter to achieve a rise of 2.2% points in efficiency.

I. INTRODUCTION

The increasing demand from consumers and industry for high power density converters is leading to continuous work on resonant switch-mode power supplies (SMPS) switching in the very high frequency (VHF) range between 30 MHz to 300 MHz [1], with designs achieving efficiencies up to approximately 90% [2], [3]. VHF power converters reduces the requirements for inductance and capacitance for a desired output power as the need for energy storing per switching cycle is inversely coupled to the frequency [4]. Galvanic isolation in traditional high-frequency SMPS is usually implemented with an isolation transformer, that utilize a magnetic core. This approach is not suitable for VHF-SMPS as the core-losses will be significant at VHF [5]. Air-core magnetics with the lack of magnetic core removes the core-losses and the possibility of saturation, making them a suitable solution for VHF operation with the converters low inductance requirements. Air-core magnetic design can be implemented into a printed circuit board (PCB) [6]–[14], and can be implemented with different structures. This paper propose an implementation and analytic inductance model of a PCB embedded rectangular solenoid air-core transformer. First the concept of PCB embedded solenoid transformers is introduced, followed by an evaluation of different winding configurations. An analytic model of the inductive parameters is developed for the most promising configuration and finally the model is verified by measurement of implemented prototypes, together with the use of a coupled inductor in a VHF resonant converter, showing improved efficiency.

II. TRANSFORMERS

Transformers in general exist in a vast number of designs, most focused on versions with a core to guide the flux to achieve a good coupling between the primary and secondary side. Air-core transformers can also be made in many designs, but are for practical purposes limited to structures that in it self will guide the flux through a common winding area, since there is no core to help guiding the flux. Spiral and toroid PCB embedded air-core transformer structures have previously been proposed [10], [13]. In this paper the solenoid PCB embedded air-core transformer is proposed.

A. Printed circuit board embedded air-core transformers

The idea behind PCB embedded air-core transformers is to utilize the easy, cheap and consistent production method of PCBs with air-core magnetics. In general aircore magnetics are considered a viable solution for magnetic components in converters operating in the VHF range since they do not have any core losses, which in a normal core based transformer would be very high [5]. The disadvantages are the low inductance, coupling and typically a larger external magnetic field, due to the missing magnetic material [8]. The 3D transformer structures are created in the PCB by using areas of copper in a two or more layer PCB as horizontal wires and by using vias as vertical wires.

1) Solenoid transformers: The PCB embedded rectangular air-core solenoid transformer can be designed in different configurations, where two main configurations are investigated here. The solenoid transformer is either built up by having a solenoid winding structure nested inside a slightly larger solenoid structure, embedded in a minimum four layer PCB, giving the transformer a rectangular cross section as seen in Figure 1a, or built up by having two or more windings in an interleaved pattern on the same layers. The interleaved configurations ranges from the bifilar configuration to the end-to-end configuration as seen in Figures 1b to 1d. The interleaved configurations can be implemented in a two layer PCB.



(d) End-to-end

Fig. 1. Different winding configurations of rectangular PCB embedded air-core solenoid transformers

III. COMPARISON

In this section different configurations similar to each of the configurations shown in Figure 1 is evaluated for their inductive parameters and coupling. A good transformer will have a high coupling to efficiently transfer energy from the primary side to the secondary.

A. Transformer parameters

The inductive model of a transformer can be defined as a two-port passive device and is described by the inductance matrix [15] in (1). The inductance matrix consists of the self-inductance for each winding L_{pp} and L_{ss} . L_{ps} and L_{sp} are the mutual inductance, and is determined by the flux that flows through the mutual area of the transformer. Note that L_{ps} and L_{sp} are equal since the transformer is a passive device.

$$\begin{bmatrix} v_p(t) \\ v_s(t) \end{bmatrix} = \begin{bmatrix} L_{pp} & L_{ps} \\ L_{sp} & L_{ss} \end{bmatrix} \frac{d}{dt} \begin{bmatrix} i_p(t) \\ i_s(t) \end{bmatrix}$$
(1)

The coupling ratio is defined as (2), and is a ratio of how much flux couples between the primary and secondary windings. In regular transformers wound around a core, the coupling is in the range of k = 0.99 for the best [15].

$$k = \frac{L_{ps}}{\sqrt{L_{pp}L_{ss}}} \tag{2}$$

The measurement of the inductance matrix defined in (1) was carried out by measuring the S-parameters [13] on a Agilent 4396B, set up as a vector network analyzer. The S-parameter measurements are converted to Z-parameters [16] and then inductances by (3).

$$L = \frac{\Im(Z)}{\omega} \tag{3}$$

B. Evaluation of configurations

Prototypes in each of the configurations similar to the ones shown in Figure 1 were produced and measured. The prototypes are shown in Figure 2. The resulting inductances, in Table I, used for comparison are mean values from 1 MHz to 10 MHz, to remove any small measurement noise. The impedances measurements of the prototype transformers, are shown in Figure 3.

TABLE I INDUCTANCES OF DIFFERENT CONFIGURATIONS OF PCB EMBEDDED SOLENOID TRANSFORMERS

Variable	L_{pp}	L_{ps}	L_{ss}	k	
Measured [nH]	67.40	17.62	9.58	0.70	
(a) Nested configuration					
Variable	L_{pp}	L_{ps}	L_{ss}	k	
Measured [nH]	42.25	18.95	41.87	0.45	
(b) Bifilar					
Variable	L_{pp}	L_{ps}	L_{ss}	k	
Measured [nH]	40.95	10.52	38.81	0.26	
(c) Interleaved in sections (2:2)					
Variable	L_{pp}	L_{ps}	L_{ss}	k	
Measured [nH]	41.00	3.22	39.45	0.08	



The interleaved configurations shows coupling in the range from k = 0.08 to k = 0.45. The more intertwined the primary and secondary side is, the higher the coupling. The coupling is in general limited by the clearance between the traces / sections on the PCB, leaving the end-to-end configuration with very bad coupling for a transformer. The sections configuration shows better properties with the bifilar configuration as the optimum of the interleaved with a coupling of k = 0.45. Compared to the nested configuration, the interleaved configurations are inferior, as the coupling is k = 0.7 for the nested configuration. This coupling coefficient is suitable for a VHF converter as the leakage inductance can be used in the design of the converter. As the nested configuration is the most promising, the inductance formulas for inductance matrix will be derived and evaluated.






Fig. 3. Impedances of different winding configurations of the PCB embedded air-core solenoid transformer

IV. ANALYTIC MODEL

In this section the nested configuration is analytically analyzed. The cylindrical solenoid transformer is a classic textbook example in the magnetic literature, with relatively simple equations [17]. The general equations for a solenoid inductor with a rectangular cross section was derived in [18], the short version given in [19]. The formula is valid as long as the width is greater than the height of the inductor, which, for all practical purposes, is the case for PCB embedded solenoids. There are no equations in the literature for solenoid transformers with rectangular cross sections, so the model will be deduced here. The following assumptions are made to simplify the model expression:

- The lengths of each solenoid structure are equal
- The flux is evenly distributed inside the solenoid.

The assumption that the flux is evenly distributed inside the the solenoid is an approximation, and will be a source of error. The approximation is made on basis of 3D finite element simulations made in COMSOL of a single inductor. An example of the strength of the magnetic field is shown in Figure 4. The flux is slightly concentrated where there is a wire or a via, but is on average close to an evenly distribution.

1) Inductance formula: For the self-inductance L_{pp} and L_{ss} the equation is given in (5) [19], with the respective



Fig. 4. Flux distribution in a PCB solenoid inductor

heights h, widths w and numbers of turns N for the primary and secondary side respectively. The dimensions can be seen in Figure 5. For reference the inner structure of windings is denoted the primary side and the outer the secondary. g is given by (4).

$$g^2 = w^2 + h^2 \tag{4}$$

The mutual inductance, L_{ps} or L_{ps} , is determined by the shared flux, which will pass through the windings mutual area inside the transformer. The mutual inductance must therefore be the same inductance formula (5) as the inner solenoid with respect to the width and height but with the number of turns as $N^2 = N_p N_s$.

$$\begin{split} L &= 8 \cdot 10^{-9} N^2 \frac{hw}{l} \\ &\left[\frac{1}{2} \frac{l}{h} \sinh^{-1} \left(\frac{w}{l} \right) + \frac{1}{2} \frac{l}{w} \sinh^{-1} \left(\frac{h}{l} \right) \\ &- \frac{1}{2} \left(1 - \frac{h^2}{l^2} \right) \frac{l}{h} \sinh^{-1} \left(\frac{w}{l\sqrt{1 + \frac{h^2}{l^2}}} \right) \\ &- \frac{1}{2} \left(1 - \frac{w^2}{l^2} \right) \frac{l}{w} \sinh^{-1} \left(\frac{h}{l\sqrt{1 + \frac{w^2}{l^2}}} \right) \\ &- \frac{1}{2} \frac{h}{l} \sinh \left(\frac{w}{h} \right) - \frac{1}{2} \frac{w}{l} \sinh^{-1} \left(\frac{h}{l} \right) \\ &+ \left(\frac{\pi}{2} - \tan^{-1} \left(\frac{wh}{l^2 \sqrt{1 + \frac{g^2}{l^2}}} \right) \right) \\ &+ \frac{1}{3} \frac{l^2}{wh} \sqrt{1 + \frac{g^2}{l^2}} \left(1 - \frac{1}{2} \frac{g^2}{l^2} \right) + \frac{1}{3} \frac{l^2}{wh} \\ &- \frac{1}{3} \frac{l^2}{wh} \sqrt{1 + \frac{h^2}{l^2}} \left(1 - \frac{1}{2} \frac{h^2}{l^2} \right) \\ &+ \frac{1}{6} \frac{l}{wh} \left(\frac{g^3 - w^3 - h^3}{l^2} \right) \bigg] \end{split}$$
(5)

A. Verification

The nested configuration is produced in different windings ratios, sizes, and coupling ratios to validate the formulas. The transformers are produced in a standard 4 layer PCB configuration. The full configuration of the prototypes is listed in Table II. A prototype transformer is shown in Figure 2a.

TABLE II PARAMETERS OF THE NESTED CONFIGURATION SOLENOIDS

		Dimensions [mm]					rns
Prototype	w_p	h_p	w_s	h_s	l	N_p	N_s
1	23.2	1.24	25	1.6	10.3	19	19
2	6.6	1.24	8.2	1.6	6.5	7	2
3	16.1	1.24	17.7	1.6	7.4	15	2
4	5.2	1.24	6.8	1.6	5.7	7	2

The measured inductances shown in Table III fit closely to the calculated with the highest deviation between calculated and measured being 18.7%. The coupling has a higher deviation than the inductances, since it is a combination of the deviations of the inductances, but still with a maximum deviation of 21.8%, which is comparable to formulas for other structures [20]. The inductance equations is in general good enough to use for design of transformers or coupled inductors.

TABLE III

COMPARISON OF ANALYTIC MODEL AND MEASUREMENTS

Variable	L_{pp}	L_{ps}	L_{ss}	k
Calculated [nH]	1119.17	1119.17	1499.16	0.86
Measured [nH]	1154.33	1104.82	1506.57	0.84
Δ [%]	3.05	1.30	0.49	2.89

	(-)	C - 1 1	1 C
(aı	Solenoid	transformer
. 1	(~)	001011010	or combror more

Variable	L_{pp}	L_{ps}	L_{ss}	k
Calculated [nH]	65.98	20.91	9.17	0.85
Measured [nH]	67.40	17.62	9.58	0.70
Δ [%]	2.11	18.71	4.21	21.78

(b) Solenoid transformer - 2

Variable	L_{pp}	L_{ps}	L_{ss}	k
Calculated [nH]	596.30	86.45	15.77	0.89
Measured [nH]	592.14	76.84	18.54	0.74
Δ [%]	0.70	12.50	14.95	20.54

(c) Solenoid transformer - 3

Variable	L_{pp}	L_{ps}	L_{ss}	k
Calculated [nH]	52.47	16.64	7.68	0.83
Measured [nH]	53.40	14.17	8.22	0.68
Δ [%]	1.75	17.42	6.63	21.83

(d) Solenoid transformer - 4



Fig. 5. Dimensions of the PCB embedded solenoid transformer (nested configuration)

V. Applied design

The nested configuration is used in an half-bridge VHF resonant power converter presented in [21]. The converter is a class-DE type, with a passive gate drive [22] for both the high-side and low-side. Each gate drive circuit has an inductor $L_{G(1,2)}$ in series with the MOSFETs gate. For the class-DE converter to work properly in the VHF area, the switch timing between the high-side and lowside is critical. The two separate inductors can be made by a coupled inductor instead, which will force the gate currents to be 180° phase-shifted, and ensure a better timing, resulting in higher efficiency. As the gate drive circuit should see the same inductance as when the gate inductors are not coupled, the primary L_{pp} and secondary inductances L_{ss} values should be chosen so the leakage inductance corresponds to the original L_G , leading to (6). If the coupling is close to one, the inductances would go to infinity, hence large couplings are for any applied purpose unpractical, as the coupled inductor will be very large. A coupling of k = 0.6 is chosen as a good compromise, leading to an inductance matrix of (7).

$$L_{pp,ss} = \frac{L_G}{1-k} \tag{6}$$

$$\begin{bmatrix} L_{pp} & L_{ps} \\ L_{sp} & L_{ss} \end{bmatrix} = \begin{bmatrix} 375\text{nH} & 225\text{nH} \\ 225\text{nH} & 375\text{nH} \end{bmatrix}$$
(7)

A. Coupled inductor design

The coupled inductor is designed with the formulas presented in Section IV, with the simplifications implemented. The length of the inductors are matched (as close as possible) and the widths w_p and w_h are set as close as possible. The PCB production sets certain limits on the inductors, e.g. as the width of each trace can be no smaller than the minimum via hole size, the clearance between the traces, as well as the width of the inner inductor must have a minimum clearance to the outer inductor, etc. All the restrictions does that not all combination of L_{pp} , L_{ss} and L_{ps} can be obtained. The specific design is found using a genetic algorithm to optimize the transformer to the given inductance matrix. The algorithm has been set to favorite the leakage inductance to ensure the high-side and low-side gate-drive are matched.

B. Experimental

The class-DE converter from [21] is used as base performance. The schematic is shown in Figure 6, with the corresponding component values in Table IV.



Fig. 6. Converter schematic of the class-DE inverter with coupled inductors in the self-oscillating gate drive

TABLE IV Component values for class-DE converter

Component	Value
C_{in}	4.7nF
C_s	14pF
C_{GD}	6pF
C_{GS}	82pF
C_{GDext}	39pF
L_G	150nH
C_{G1}	100nF
L_H	3.3µH
C_T	500pF
L_T	360nH
C_R	40pF
C_{OUT}	14nF

TABLE V Comparison of the class-DE converter with different gate drive inductors

Converter	L_{pp}	L_{ps}	L_{ss}	k	$L_{pp,leak}$	$L_{ss,leak}$	P_{in}	P_{out}	n
Base configuration	150 nH	-	150 nH	-	-	-	12.21W	9.57W	78.87%
Leakage based	120nH	-	100 nH	-	-	-	10.25W	6.44W	63.1%
Coupled inductor	238.5nH	165 nH	$333.5 \mathrm{nH}$	0.58	120.5	102.8	11.32W	9.12W	81.09%

The converter performance is tested at 80V input, and the efficiency is measured with a Yokogawa WT1600. The converter, shown in Figure 7, is tested with three different L_G configurations:

- 1) The original inductors
- 2) Inductors with the value of the leakage inductance
- 3) The coupled inductor

The leakage inductor value, are two separate inductors with the value of the leakage inductance of the coupled inductor. This is to check whether it is the coupling or the slight change in inductance that causes any effect in the circuit. The inductor measurement and efficiencies are shown in Table V. The base efficiency is 78.87%, and with the leakage-valued inductors the efficiency is at 63.1%. The coupled inductor increase the efficiency by 2.2% points from the base value to 81.09%.



(a) Non-coupled gate inductors



(b) Coupled PCB embedded gate driver inductors

Fig. 7. Class-DE converters used for efficiency comparison

VI. CONCLUSION

New structures for PCB embedded air-core transformers was proposed based on the rectangular solenoid structure, for use in VHF converters. The structures were evaluated for the use as a transformer, by comparing the coupling of the transformers. The structure of two solenoids nested inside each other, was found to be the most promising. An analytic two-port model for the structure was developed and verified by measurements on prototypes of the configuration. The formula shows a good consistency between the calculated and measured inductances, with a deviation range of approximately 3 % to 22 %. The proposed analytical model is suitable to predict the inductive parameters of the nested configuration of the PCB embedded air-core solenoid transformers.

The use of a coupled inductor in the self-oscillating gate drive in a class-DE converter shows that PCB embedded air-core solenoid inductor are suitable for using in VHF converters.

References

- A. Knott, T. Andersen, P. Kamby, J. Pedersen, M. Madsen, M. Kovacevic, and M. Andersen, "Evolution of very high frequency power supplies," *Emerging and Selected Topics in Power Electronics, IEEE Journal of*, vol. PP, no. 99, pp. 1–1, 2013.
- [2] J. Hu, A. Sagneri, J. Rivas, Y. Han, S. Davis, and D. Perreault, "High-frequency resonant sepic converter with wide input and output voltage ranges," *Power Electronics, IEEE Transactions* on, vol. 27, no. 1, pp. 189–200, Jan 2012.
- [3] J. Rivas, O. Leitermann, Y. Han, and D. Perreault, "A very high frequency dc-dc converter based on a class φ₂ resonant inverter," in *Power Electronics Specialists Conference*, 2008. *PESC* 2008. *IEEE*, June 2008, pp. 1657–1666.
- [4] D. Perreault, J. Hu, J. Rivas, Y. Han, O. Leitermann, R. Pilawa-Podgurski, A. Sagneri, and C. Sullivan, "Opportunities and challenges in very high frequency power conversion," in *Applied Power Electronics Conference and Exposition*, 2009. APEC 2009. Twenty-Fourth Annual IEEE, Feb 2009, pp. 1–14.
- [5] Y. Han, G. Cheung, A. Li, C. Sullivan, and D. Perreault, "Evaluation of magnetic materials for very high frequency power applications," in *Power Electronics Specialists Conference*, 2008. *PESC 2008. IEEE*, june 2008, pp. 4270–4276.
- [6] S. Orlandi, B. Allongue, G. Blanchot, S. Buso, F. Faccio, C. Fuentes, M. Kayal, S. Michelis, and G. Spiazzi, "Optimization of shielded pcb air-core toroids for high-efficiency dc-dc converters," *Power Electronics, IEEE Transactions on*, vol. 26, no. 7, pp. 1837–1846, July 2011.
- [7] A. Sagneri, D. Anderson, and D. Perreault, "Transformer synthesis for vhf converters," in *Power Electronics Conference* (*IPEC*), 2010 International, June 2010, pp. 2347–2353.
- [8] M. P. Madsen, A. Knott, M. A. E. Andersen, and A. P. Mynster, "Printed circuit board embedded inductors for very high frequency switch-mode power supplies," *ECCE Asia*, 2013.
- [9] P. Kamby, A. Knott, and M. Andersen, "Printed circuit board integrated toroidal radio frequency inductors," in *IECON 2012 -38th Annual Conference on IEEE Industrial Electronics Society*, 2012, pp. 680–684.
- [10] S. C. Tang, S. Hui, and H. Chung, "Characterization of coreless printed circuit board (pcb) transformers," in *Power Electronics* Specialists Conference, 1999. PESC 99. 30th Annual IEEE, vol. 2, 1999, pp. 746–752 vol.2.
- [11] S. Orlandi, B. Allongue, G. Blanchot, S. Buso, F. Faccio, C. Fuentes, M. Kayal, S. Michelis, and G. Spiazzi, "Optimization of shielded pcb air-core toroids for high-efficiency dc-dc converters," *Power Electronics, IEEE Transactions on*, vol. 26, no. 7, pp. 1837–1846, 2011.
- [12] R. Ambatipudi, H. babu Kotte, and D. K. Bertilsson, "Coreless printed circuit board (pcb) step-down transformers for dcdc applications," World Academy of Science, Engineering and Technology, vol. 46, pp. 379–388, 2010.
- [13] J. Pejtersen and A. Knott, "Design and measurement of planar toroidal transformers for very high frequency power applications," in *IEEE 7th International Power Electronics and Motion Control Conference - ECCE Asia*, 2012, pp. 688 – 692.

Inc., 2004.

- [14] S. C. Tang, S. Hui, and H.-H. Chung, "Coreless planar printedcircuit-board (pcb) transformers-a fundamental concept for signal and energy transfer," *Power Electronics, IEEE Transactions* on, vol. 15, no. 5, pp. 931–941, 2000.
- [15] R. W. Erickson and D. Maksimic, Fundamentals of Power Electronics, 2nd ed. Springer, 2001.
 [16] D. Frickey, "Conversions between s, z, y, h, abcd, and t param-
- [16] D. Frickey, "Conversions between s, z, y, h, abcd, and t parameters which are valid for complex source and load impedances," *Microwave Theory and Techniques, IEEE Transactions on*, vol. 42, no. 2, pp. 205–211, Feb 1994.
- [17] D. K. Cheng, Field and Wave Electromagnetics, 2nd ed., ser. World Student. Addison-Wesley Publishing Company, 1989.
- [18] Niwa, "Research paper 141," 1924.
- [19] F. W. Grover, Inductance Calculations. Dover Publications,
- [20] J. Pejtersen, J. D. Mønster, and A. Knott, "Development and verification of printed circuit board toroidal transformer model," *Twenty-Eighth Annual IEEE Applied Power Electronics Conference and Exposition*, 2013.
- [21] M. Madsen, A. Knott, and M. Andersen, "Very high frequency half bridge dc/dc converter," in *Applied Power Electronics Conference and Exposition (APEC)*, 2014 Twenty-Ninth Annual *IEEE*, March 2014, pp. 1409–1414.
- [22] M. Madsen, J. Pedersen, A. Knott, and M. Andersen, "Selfoscillating resonant gate drive for resonant inverters and rectifiers composed solely of passive components," in *Applied Power Electronics Conference and Exposition (APEC)*, 2014 Twenty-
- Ninth Annual IEEE, March 2014, pp. 2029–2035.

A.18 M. Madsen and J. Mønster: "Embedded solenoid transformer for power conversion", WO2015092070 A1, June 25th 2015



(19) World Intellectual Property

Organization International Bureau



(43) International Publication Date 25 June 2015 (25.06.2015)

- (51) International Patent Classification: H01F 27/28 (2006.01)
- (21) International Application Number: PCT/EP2014/079037
- (22) International Filing Date:
 - 22 December 2014 (22.12.2014)
- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 13198912.1 20 December 2013 (20.12.2013) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: MADSEN, Mickey P.; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK). MØNSTER, Jakob Døllner; Tranvej 3, tv., DK-2400 Copenhagen NV (DK).
- (74) Agent: KUHN, Oliver; Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).

(10) International Publication Number

WO 2015/092070 A1

- (81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM, AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN, TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.
- (84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, ST, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

— with international search report (Art. 21(3))

(54) Title: EMBEDDED SOLENOID TRANSFORMER FOR POWER CONVERSION



(57) Abstract: A resonant power converter for operation in the radio frequency range, preferably in the VHF, comprises at least one PCB-embedded transformer. The transformer is configured for radio frequency operation and comprises a printed circuit board defining a horizontal plane, the printed circuit board comprising at least two horizontal conductive layers separated by an isolating layer, a first embedded solenoid forming a primary winding of the transformer and a second embedded solenoid being arranged parallel to the first solenoid and forming a secondary winding of the transformer, wherein the first and second embedded solenoid sare formed in the conductive layers of the printed circuit board, wherein each full turn of an embedded solenoid has a horizontal top portion formed in an upper conductive layer, a horizontal bottom portion formed in a lower conductive layer, and two vertical side portions formed by vias extending between the upper and the lower conducting layers.

1

EMBEDDED SOLENOID TRANSFORMER FOR POWER CONVERSION

The present invention relates in one aspect to a transformer for radio frequency operation comprising a printed circuit board defining a horizontal plane, the printed

- 5 circuit board comprising at least two horizontal conductive layers separated by an isolating layer, a first solenoid forming a primary winding of the transformer and a second solenoid being arranged parallel to the first solenoid and forming a second-ary winding of the transformer.
- 10 According to a further aspect, the invention relates to a power converter operating at radio frequencies and comprising a transformer of the above-mentioned kind. More particularly, the invention relates to a resonant DC-DC converter operating at radio frequencies and comprising a transformer of the above-mentioned kind.

15 BACKGROUND OF THE INVENTION

Over many years there has been a constant need for small, cheap and efficient power supplies. This has led to the development of Switch-Mode Power Supplies (SMPS). The size of modern power supplies are mainly governed by the passive energy storing elements, which scales inversely with the switching frequency.

- 20 Therefore, much development has been devoted to increasing switching frequencies. Commercially available converters today switch at frequencies up to several megahertz and can have efficiencies of more than 95%. The reason for not increasing the switching frequency further and thereby reaching even higher power densities are switching losses. One way of addressing this issue has been the use of res-
- 25 onant Radio Frequency (RF) amplifiers (inverters) combined with a rectifier for DC-DC converters. With this type of converters, SMPSs with switching frequencies in the Very High Frequency range (VHF, 30- 300MHz) have been designed with efficiencies up to approx. 90%. By operating the DC-DC converters at radio frequencies, or even at VHF frequencies, it is possible to reduce the size, and thereby also
- 30 price, of the passive energy storing elements significantly. However, the inductive components, and in particular the power transformers, are still the largest components in the circuits. Therefore, there is a need for small and highly efficient transformers.

Object of the present invention is therefore to provide a transformer for radio frequency operation, with a small and efficient design that can be reliably produced at low cost.

5 DEFINITIONS

Throughout the present text, the term "conductive" refers to electrically conductive. The term "isolating" refers to electrically isolating. The term "coupled" is to be understood as a galvanically separated electromagnetic coupling, which is to be distinguished from electrically "connected" referring to a galvanic connection through a

10 conductive material.

Depending on the particular use, a transformer may sometimes also be referred to as coupled inductors. The term "transformer" usually refers to devices adapted for power transfer. Transformers require a high level of coupling to avoid inhibitive losses. The term "coupled inductors" is usually employed when referring to devices adapted for signal transfer, where a lower level of coupling may still be useful or maybe even desirable for a particular application. However, the topologies of a transformer and of coupled inductors cannot be distinguished in practice. Therefore, throughout the present application, the terms transformer and "coupled inductors"

20 are treated as interchangeable.

25

When arranged in a circuit, a transformer may be operated in different configurations depending on the way the transformer is connected into the circuit. When referring to an "isolating transformer", it is to be understood that a given transformer is connected in an isolating configuration. Accordingly, when referring to an "autotransformer" it is to be understood that the transformer is connected in an autotransformer configuration.

When referring to power converters, this includes any of a DC-DC converter, an AC-30 DC converter, an AC-AC converter, and a DC-AC converter.

The term "horizontal" refers to directions parallel to the plane of the printed circuit board (PCB). The term "vertical" refers to directions perpendicular to the horizontal plane defined by the printed circuit board, wherein a first vertical direction points in

an upward direction, and a second vertical direction opposite to the first vertical direction points in a downward direction, wherein a vector pointing from a "lower" element to an "upper" element has an upward pointing vertical component.

- 5 The term "parallel" covers geometries with a lateral offset as well as coinciding geometries without offset, i.e. when referring to parallel axes coinciding axes are also covered. In this context, the orientation of a solenoid is determined by the direction of its longitudinal axis. Any angle between solenoids including the angle zero, i.e. a configuration with parallel solenoids, is thus to be understood as the angle between
- 10 the respective longitudinal axes of the solenoids.

SUMMARY OF THE INVENTION

According to a first aspect, the object of the invention is achieved by a resonant power converter for operation in the radio frequency range, preferably in the VHF,

- 15 the power converter comprising at least one transformer, wherein the at least one transformer comprises
 - a printed circuit board defining a horizontal plane, the printed circuit board comprising at least two horizontal conductive layers separated by an isolating layer,
 - a first embedded solenoid forming a primary winding of the transformer and
- a second embedded solenoid being arranged parallel to the first solenoid and forming a secondary winding of the transformer,

wherein the first and second embedded solenoids are formed in the conductive layers of the printed circuit board, wherein each full turn of an embedded solenoid has a horizontal top portion formed in an upper conductive layer, a horizontal bottom

- 25 portion formed in a lower conductive layer, and two vertical side portions formed by vias extending between the upper and the lower conducting layers. In general, a single winding formed in this way may already be considered as a PCB-embedded solenoid.
- 30 Thereby an efficient power converter is achieved with a small size, which can be produced in an automated manner with high reproducibility and at low cost. The power converter may be any of a DC-DC converter, an AC-DC converter, an AC-AC converter, and a DC-AC converter.

PCT/EP2014/079037

4

Further according to one embodiment of the power converter, at least one of the transformers is arranged as an isolating transformer between an input side and an output side. In such a configuration, the power passed through the converter is transferred from an input side of the converter through the isolating transformer to the output side of the converter. Therefore, a good coupling between the first and

5 the output side of the converter. Therefore, a good coupling between the first and second solenoids is required to avoid excessive losses, such as a coupling factor k above 0,6 or preferably above 0,8.

Further according to one embodiment, the power converter is a resonant DC-DC converter. In this context the term DC can refer to a slowly varying voltage. In the context of a switch mode power converter, the term "DC" may thus be considered fulfilled as long as the variation of the DC-voltage level is slow as compared to the switching frequency of the power converter. When building a resonant DC-DC converter, a good coupling between the first and second solenoids is desired in order to

- 15 achieve an efficient resonant operation of the power converter. This is achieved by using a PCB-embedded transformer of the solenoid type as disclosed in the present application. Advantageously, this may be such a PCB-embedded transformer with bifilar interleaved first and second solenoids, wherein single turns of the first solenoid alternate with single turns of the second solenoid. Alternatively, this may ad-
- 20 vantageously also be a PCB-embedded transformer formed in a multilayer PCB, such as a four-layer PCB, wherein the second solenoid is arranged inside the first solenoid as seen in a cross-sectional plane perpendicular to a longitudinal axis of the solenoids (or vice versa).
- 25 Furthermore, by using a PCB-embedded transformer of the solenoid type as disclosed in the present application for coupling first and second solenoids in a resonant DC-DC power converter, a particularly small form factor, low profile (i.e. a low vertical dimension), low weight, and low production cost is achievable. Furthermore, using a PCB-embedded transformer in a power converter as described in the pre-
- 30 sent invention has the advantage that the foot-print of the transformer may be shaped as desired by the circuit designer. This freedom of design gives a considerable flexibility to the circuit designer and is particularly relevant when the available space is a critical parameter.

The above-mentioned advantages are particularly useful, for example when integrating a power converter into LED-lighting devices or flat-screen displays, due to the very stringent spatial limitations in such lighting applications. In particular the limitations on the vertical dimensions (low profile) of the converters are very demanding in

5 these applications. These challenges can be met by a power converter including PCB-embedded transformer as disclosed by the present invention. The abovementioned advantages are also useful, for example when building charging devices as e.g. used in many mobile consumer devices, due to the reduced weight and an enhanced mechanical stability that may be achieved thereby.

10

Further according to one embodiment of the power converter, a switching frequency on the input side is at least 10 MHz, preferably at least 20 MHz, more preferably at least 30 MHz. This operation range of the power converter is particularly advantageous for the resonant DC-DC converter.

15

20

Operating in this frequency range facilitates the use of embedded transformers in a particularly advantageous way for designing efficient power converters. The choice of operating at frequencies of at least 10MHz, at least 20Mhz, or even at least 30MHz synergistically adds to the above-mentioned advantages of a low overall form factor, low profile, low weight, freedom of design, etc. For example, the possibility of an efficient coreless transformer design of the embedded PCB-transformer configured for operation in the VHF as discussed in this application avoids losses in

- a core material; the low capacitance needed when operating at these frequencies makes it possible to use ceramic capacitors which have a much longer lifespan than
- 25 electrolytic capacitors; at these frequencies a rapid transient response and a small component form factor can be achieved; the smaller size of the components required leads to a reduced cost of components and a cheaper assembly, thereby reducing the overall production cost for the power converter. In particular, designing the power converter as a resonant power converter facilitates a particularly efficient
- 30 operation at the above recited switching frequencies.

Further according to some embodiments of the resonant power converter, the transformer is provided according to one of the advantageous embodiments as detailed in the following.

5

6

According to a broader aspect, the object of the invention is achieved by a transformer for radio frequency operation, the transformer comprising

- a printed circuit board defining a horizontal plane, the printed circuit board com-

- prising at least two horizontal conductive layers separated by an isolating layer,
- a first embedded solenoid forming a primary winding of the transformer and
- a second embedded solenoid being arranged parallel to the first solenoid and forming a secondary winding of the transformer,

wherein the embedded solenoids are formed in the conductive layers of the printed circuit board, wherein each full turn of an embedded solenoid has a horizontal top portion formed in an upper conductive layer, a horizontal bottom portion formed in a lower conductive layer, and two vertical side portions formed by conductive vias extending between the upper and the lower conducting layers.

15 By using solenoids embedded in a PCB in a transformer arrangement, an embedded solenoid transformer is obtained. An embedded solenoid transformer has several advantages over other configurations as further detailed in the following.

The printed circuit board is stratified with a vertical stacking direction, i.e. different

- layers of the printed circuit board are oriented horizontal and parallel to each other.
 At least one isolating layer usually also forms the carrier layer. In a multilayer PCB, the stacked layers may in combination form a carrier layer.
- The embedded solenoids are helical coils wound around a horizontal axis that defines a longitudinal direction of the solenoid. In a preferred embodiment, each coil is formed by a series of traces of conductive material patterned into the horizontal conductive layers of the PCB. The traces are electrically connected in series by means of vias penetrating the isolating layer in a vertical direction to form a horizontal coil embedded in the PCB. Top portions of the coil that are formed by traces in an upper conductive layer alternate with bottom portions that are formed by traces in a lower conductive layer. Starting with a top portion, a full turn of a coil is thus formed, by the top portion, followed by a downward via, a bottom portion, and an upward via. Each of the upward and downward vias may also be formed by multiple

vias that are coupled in parallel, thereby lowering the total resistance of the turn.

The embedded solenoid transformer is easy to scale by merely adjusting the layout of the traces forming the top and bottom portions and the positions of the vias.

The embedded solenoids have an essentially rectangular cross-section as seen in a cross-sectional plane perpendicular to the longitudinal axis. Typically, the rectangular cross-section has a slab-like shape with a flat aspect ratio where the vertical dimension is smaller than the horizontal dimension.

The first and second solenoids are placed next to each other, with partial overlap, or with full overlap, and are inductively coupled. The coupling may be external, i.e. through the external field of the solenoids, and/or internal, i.e. through the internal field of the solenoids.

Advantageously according to a preferred embodiment, the transformer is "coreless", 15 i.e. the solenoids are not coupled through a magnetically conducting material. The term "coreless" is thus to be understood as to also include embodiments where the transformer is coupled through a magnetically non-permeable material, such as a typical substrate material used for producing printed circuit board. For example, the embedded solenoids may be wound about, and thus filled with, such magnetically

20 non-permeable material. Such coreless embodiments are in the following also referred to as "air-core". Such coreless transformer arrangements have the advantage that they are easier to produce than an embedded transformer with core-based coupling. Furthermore, such coreless embodiments have the advantage that they are not subject to dissipation due to losses in a core material.

25

30

Advantageously according to a preferred embodiment, the first and second solenoids are embedded on the same PCB. Thereby, additional assembly steps can be avoided, which may carry the inherent risk of misalignment and reduced precision. As a consequence, reliable and reproducible high-volume production of the embedded transformer can be achieved.

In the most general configuration, the longitudinal axes of the first and second solenoids may have an arbitrary angle with respect to each other. However, in order to achieve an appropriate coupling, in particular in an air-core configuration, the longi-

PCT/EP2014/079037

8

tudinal axes of the first and second solenoids are arranged parallel to each other. Furthermore, the first and second solenoids may have an essentially rectangular footprint on the PCB as seen in projection in a horizontal plane. Owing to that rectangular footprint, the embedded transformer built from the first and second sole-

5 noids can be designed to fit easily into a complete circuit design which most frequently also has rectangular layout geometry.

According to one embodiment, an embedded transformer relies on external coupling through a horizontal side-by-side arrangement, or a vertical arrangement of two solenoids stacked on top of each other. This structure is easy to implement and it is very easy to make more than one primary or secondary winding. Each winding will give the best possible inductor in terms of inductance divided by the series resistance of the inductor, but at the price of a considerably lower coupling between the first and second solenoids than for other arrangements.

15

As mentioned above, the use of embedded solenoids for the embedded transformer has a number of advantages over other coil geometries. For example, as compared to spiral windings solenoids have the benefit that the current is more evenly distributed over the full width of the traces, whereas the current density of spirals tends to

- 20 concentrate at one edge of the traces. Therefore the solenoid geometry can achieve a lower series resistance of the coil, in particular for large structures. Furthermore, as mentioned above, solenoids have a rectangular footprint with connectors that are typically arranged at the corners of the structure. From a design flexibility point of view this is preferable over the round structure of a spiral with one connector in the
- 25 center of the structure. Furthermore the external magnetic field from the solenoid geometry is much weaker than for the spiral geometry, thereby reducing the risk of failing electromagnetic interference requirements. In a further example, as compared to toroidal coil geometries, solenoids have lower series resistance as the viases in the solenoid geometry are distributed equally on both sides of the traces,
- 30 whereas in the toroidal structure would result in a concentration of the current density in the vertical viases at the inner edge of the toroid. Furthermore, the rectangular footprint of the solenoid geometry is again to be considered an advantage from a circuit designer's point of view as compared to the round shape of toroids, amongst others due to a higher inductance per area, better usage of footprint, and better de-

sign compatibility with rectangular geometry circuit layouts. Furthermore, a surprisingly good coupling efficiency can be achieved as compared to e.g. toroidal structures.

- 5 Further according to one embodiment of the transformer, the first and second solenoids are wound around a common axis. Thereby an improved coupling between the first and second solenoids is achieved as compared to e.g. a horizontal side-byside arrangement, or a vertical arrangement of two solenoids stacked on top of each other.
- 10

15

Further according to one embodiment of the transformer, the first and second solenoids at least partially overlap in a longitudinal direction. In the region of the overlap, the coupling between the first and second solenoids is achieved via the internal magnetic field in the region of the overlap. Thereby an improved coupling is achieved as compared to geometries relying on an external coupling. The overlap may be partial or, preferably, over the full lengths of the first and/or second solenoids. The overlap may be achieved in different geometries as further detailed below.

- 20 Further according to one embodiment of the transformer, top portions of the first and second solenoids are formed in a common upper conductive layer, and/or bottom portions of the first and second solenoids are formed in a common lower conductive layer. This embodiment has the advantage that it only requires two conductive layers. This is advantageous for a cheap and reliable production of the embedded so-
- 25 lenoid transformer, for example in a dual layer PCB.

Advantageously according to one embodiment, the PCB of the transformer is a dual layer PCB.

30 Further according to one embodiment of the transformer, the first and second solenoids are section-wise interleaved. This embodiment may be considered a hybrid of external and internal coupling between the first and second solenoids, wherein each of the sections are tightly packed coils with an optimum inductance divided by the series resistance. The sections of the first and second solenoids are all aligned on a

10

common axis, thereby sharing a common magnetic path with a good coupling efficiency. This embodiment has the advantage that it allows for providing more than one primary and/or secondary winding in a simple layout.

5 Further according to one embodiment of the transformer, first sections of the first solenoid alternate with second sections of the second solenoid. The first and second sections share a common magnetic path with a good coupling efficiency.

Further according to one embodiment of the transformer, first sections are single turns and/or second sections are single turns. Thereby, in the region of the overlap, the solenoids are fully interleaved turn by turn. To make space for this interleaving, the axial pitch (distance between adjacent turns in an axial direction) of the respective solenoids needs to be increased. This has the disadvantage that the series resistance for a given inductance of the respective solenoids increases. However, the embodiment has the advantage that a highly efficient coupling between the interleaved solenoids is achieved for a two-layer embedded structure.

Further according to one embodiment of the transformer, the PCB has at least fourlayers. This embodiment allows for more complex transformer geometries.

20

Further according to one embodiment of the transformer, when seen in a crosssectional plane perpendicular to a longitudinal axis of the solenoids, the second solenoid is arranged inside the first solenoid or, alternatively the first solenoid is arranged inside the second solenoid. This embodiment needs to be implemented in at

25 least a four layer PCB. Despite the inherent reduction of the overlap in crosssectional area shared by the first and second solenoids (the inside solenoid has inherently a smaller cross-section than the outside/enclosing solenoid), a surprisingly high coupling efficiency is achieved in this embodiment as compared to embodiments implemented in a two-layer PCB configuration, such as even the fully inter-30 leaved configuration with alternating single turns.

Furthermore, for a given application this embodiment has a smaller footprint than the above-mentioned embodiments, and magnetic flux is encapsulated better than in e.g. the interleaved structure as the turns of each of the windings is placed as

11

close as possible. This structure can also be preferable for galvanic isolation applications as the isolation is achieved through the layers of the PCB, which typically requires much shorter separation distances than isolation along the surface within the same layer. Preferably, the first solenoid forming the primary side is the inner

5 solenoid, and the second solenoid forming the secondary winding is the outer solenoid enclosing the inner solenoid.

Most preferably, first and second solenoids are arranged concentrically. Thereby an optimum coupling efficiency is achieved.

10

Further according to some embodiments, the power converter comprises further solenoids coupled to the first solenoid and/or the second solenoid. By adding further windings/solenoids a transformer with multiple input and/or output windings can be formed.

15

Further according to one embodiment of the invention, the transformer is adapted for operation at a radio frequency in the range 1MHz - 1 GHz; alternatively 10 MHz - 500 MHz; further alternatively 30 MHz - 300 MHz. The typical dimensions of PCB structures are particularly suited for viable operation of power electronics in the radio

- 20 frequency range, in particular around and above 1 MHz, and more particular in the VHF-ranges given above. Since at these frequencies the skin depth is around 10μm (micrometer), the penetration of the current into the conductor is comparable to and even less than the thickness of the conductive layers of a typical PCB. Typical PCBs have conductive layer thicknesses in the range from about 0.1mm down to about
- 25 15μm. For example, a so-called "1oz Cu" PCB layer is specified to a copper amount of 1 ounce per sqft, resulting in a copper layer thickness between 30μm-40μm, typically about 35μm. The thickness scales linearly with the copper amount, wherein typical specifications copper layers in a PCB are 1/2oz, 0.75oz, 1oz, 2oz, and 3oz. Due to the small skin depth at high frequencies, the resistance of the solenoid induc-
- 30 tors is not significantly affected by the small thickness of typical PCB conductive layers.

Further according to one embodiment of the invention, the transformer has a coupling factor k of above 0,1, preferably above 0,2, preferably above 0,3, preferably above 0,4, preferably above 0,5, more preferably above 0,6, even more preferably above 0,7, and most preferably above 0,8. Transformers intended for transferring power, e.g. an isolating transformer arranged between an input side and an output side, should have a high coupling factor k, such as above 0,6 or preferably above

- 5 0,8. On the other hand, coupled inductors intended for synchronizing parallel stages in an AC-circuit may already benefit from much lower coupling factors. In fact, it may be beneficial to provide a circuit with coupled inductors having a specified coupling factor that may be chosen within a broad range of coupling factors. The coupling factor may be chosen e.g. by selecting a layout type and/or tailoring the parameters
- 10 of a given layout, such as illustrated by the examples in the detailed description below.

Further according to one embodiment, at least one transformer is arranged as an isolating transformer between an input side and an output side.

15

Further according to one embodiment, at least one transformer is arranged as an autotransformer between an input side and an output side.

Further according to one embodiment, at least one transformer is arranged such that the first solenoid is used as a power carrying inductor in the power converter and the second solenoid is used as a sensing inductor to measure the current through the first inductor. The measured current signal may be for display, presentation at a current monitoring output port, and/or for use as a control signal in the power converter itself, or as a control signal in a system including a power converter

25 according to the present invention. The PCB-embedded transformer configuration allows for a simple, space-saving and cost-effective integration of the current sensing arrangement in the power converter.

Further according to particularly preferred embodiments, the power converter is a resonant DC-DC converter.

Further according to preferred embodiments of the power converter, the switching frequency on the input side is at least 10MHz, preferably at least 20MHz, more preferably at least 30MHz.

According to a further aspect, an inverter is adapted for operation at radio frequencies, preferably in the VHF, and comprises a transformer according to any of the above cited embodiments. The transformer structure may be used as coupled induc-

5 tors for coupling two complementary gate drive signal of first and second semiconductor switches. Thereby an efficient power inverter is achieved with a small size, which can be produced in an automated manner with high reproducibility and at low cost.

10 BRIEF DESCRIPTION OF THE DRAWINGS

Preferred embodiments of the invention will be described in more detail in connection with the appended drawings, which show schematically in

FIG. 1a-c an embedded solenoid transformer according to a first embodiment,

15

FIG. 2a-c an embedded solenoid transformer according to a second embodiment,

- FIG. 3a-c an embedded solenoid transformer according to a third embodiment,
- 20 FIG. 4a-c an embedded solenoid transformer according to a fourth embodiment,
 - FIG. 5 a DC-DC converter circuit comprising an embedded transformer according to a first embodiment,
- 25 FIG. 6 a class-DE power inverter comprising coupled inductors according to another embodiment,
 - FIG. 7 a DC-DC converter circuit comprising an embedded transformer according to a second embodiment,

30

FIG. 8 a DC-DC converter circuit comprising an embedded transformer according to a third embodiment,

5

14

- FIG. 9 a DC-DC converter circuit comprising an embedded transformer according to a fourth embodiment,
- FIG. 10 a DC-DC converter circuit comprising an embedded transformer according to a fifth embodiment, and in
- FIG. 11 a DC-DC converter circuit comprising an embedded transformer according to a sixth embodiment.

10 DETAILED DESCRIPTION OF PREFERRED EMBODIMENTS

Figs. 1-4 show different layouts for the implementation of a transformer (x00) embedded in a printed circuit board (PCB), wherein in all the shown embodiments the first solenoids (x01) and the second solenoids (x02) are arranged on a common longitudinal axis Z. In the shown embodiments, the conductive paths of each of the

- 15 solenoids (x01, x02, 203, 204) is made up of slab-shaped horizontal traces (x10, x12, x20, x22, 230, 232, 240, 242) formed in the conductive layers of a printed circuit board having at least two conductive layers, and vertical vias (x11, x13, x21, x23, 231, 233, 241, 243) made of a conductive material connecting the horizontal traces (x10, x12, x20, x22, 230, 232, 240, 242) in series to form a coil. Figs.1 3
- 20 show embodiments of PCB embedded solenoid transformers that require a PCB with two conductive layers separated by an isolating layer, such as a dual layer PCB. Fig.4 shows an embodiment requiring a multilayer PCB with at least four vertically stacked conductive layers separated from each other by isolating layers. Only the conductive paths forming the solenoids are drawn, whereas isolating layers are
- 25 omitted in the drawings. Respective sub-figures (a) show perspective views of the conductive paths for both the first and second solenoids (x01, x02), sub-figures (b) show an end view as seen in the direction of the longitudinal axis Z, and sub-figures (c) show a top elevation view of the solenoid arrangements (x00).
- 30 Fig.1 shows a first embodiment of a PCB embedded transformer 100 in an end-toend configuration with a first solenoid 101 having four turns wound around a longitudinal axis Z, each turn having a top portion 110, followed by a downward via 111, a bottom portion 112, and an upward via 113. The first solenoid 101 has first connection terminals 114, 115 for connection to further circuitry. In direct extension to the

PCT/EP2014/079037

15

first solenoid 101, a second solenoid 102 is formed by four turns wound around the same longitudinal axis Z, each turn having a top portion 120, followed by a downward via 121, a bottom portion 122, and an upward via 123. The second solenoid 102 has connection terminals 124, 125. The end-to end configuration is essentially a solenoid, which has been broken up in two solenoids by omitting the vertical con-

5 solenoid, which has been broken up in two solenoids by omitting the vertical connection between the contact terminals 114 and 125.

Fig.2 shows a second embodiment of a PCB embedded transformer 200 in a section-wise interleaved configuration. The arrangement comprises four identical sec-

- tions 210, 220, 230, 240 of four turns each, wherein the four sections 210, 220, 230, 240 are aligned along a common longitudinal axis Z. The turns of the solenoid sections 210, 220, 230, 240 are formed in the same way by respective horizontal top and bottom portions connected in series through vertical vias as described above, and each of the sections 210, 220, 230, 240 has respective connection terminals 214, 215, 224, 225, 234, 235, 244, 245. The first and third sections 210, 230 form a primary side of the transformer 200. The first and third sections 210, 230 may be operated as separate first solenoids with four turns each or they may be connected in series for operation as one first solenoid with eight turns. Accordingly, the second and fourth sections 220, 240 form the secondary side of the transformer 200 and
- 20 may be operated as separate second solenoids or as one second solenoid.

Fig.3 shows a third embodiment of a PCB embedded transformer 300 with two solenoids in a 1:1 interleaved configuration. The transformer 300 has a first solenoid 301 with turns formed by top portions 310, bottom portions 312, and vertical viases 311,

- 313 connecting alternating top and bottom portions 310, 312 in series to provide a coil as described above. The first solenoid 301 has first connection terminals 314, 315 for connection to further circuitry. A second solenoid 302 is formed accordingly from top portions 320, bottom portions 322, and vertical viases 321, 323. The second solenoid 302 has second connection terminals 324, 325 for connection to further circuitry. Both solenoids 301, 302 are wound around a common longitudinal
- axis Z. However, in contrast to the above-described embodiments, the pitch of the first and second solenoids (i.e. the distance between adjacent turns within the same solenoid) is doubled to allow for an alternating arrangement of single turns from the first solenoid 301 and from the second solenoid 302. The increased pitch tends to

10

PCT/EP2014/079037

16

produce a more leaky solenoid, and somewhat increases the length and thus the resistance of the solenoids with the same number of turns. However, the 1:1 interleaved configuration has the benefit of a considerably increased coupling between the first and second solenoids as compared to other two-layer designs, such as the ambedimente mentioned above.

5 embodiments mentioned above.

Fig.4 shows a fourth embodiment of a PCB embedded transformer 400, wherein a first solenoid 401 encloses a second solenoid 402. In the shown embodiment, the first and second solenoids are wound around the same longitudinal axis Z in a pre-ferred concentric arrangement. The transformer 400 requires a multilayer PCB with four conductive layers as best seen in Fig.4b. The turns of the outer first solenoid

- 401 are formed by top portions 410 in an uppermost conductive layer, bottom portions 412 in a lowermost conductive layer and vertical vias 411, 413 connecting alternating top and bottom portions 410, 412 in series. The turns of the inner, second
- 15 solenoid are formed by top portions 420 in an upper inner conductive layer of the multilayer PCB, bottom portions 422 in a lower inner conductive layer of the multilayer PCB, and vertical vias 421, 423 connecting the top and bottom portions 420, 422 in series. The transformer has primary side connection terminals 414, 415 for connecting the outer, first solenoid 401 to further circuitry, and secondary side con-
- 20 nection terminals 424, 425 for connecting the inner, second solenoid 402 to further circuitry. Note, that the role of the inner and outer solenoids may be interchanged, such that the inner solenoid forms the primary side of the transformer 400, and the outer solenoid forms the secondary side of the transformer 400.
- 25 Figs.5 and 6 give examples of advantageous uses of PCB embedded solenoid transformers in power converters operating at radio frequencies, preferably in the VHF range.
- Fig. 5 shows a class-E isolated resonant DC-DC converter 50, which may be adapted for radio frequency operation, preferably in the VHF. The converter 50 receives a DC-input voltage Vin on an input side 51, and provides a DC-output voltage to a load on an output side 52. The input side 51 may comprise a semiconductor switch, such as a MOSFET, with a control terminal Q1 that may be driven by an external oscillator or e.g. a self-oscillating gate drive circuit as described in the co-

PCT/EP2014/079037

pending International Patent Application PCT/EP2013/072548, and a resonant tank formed by capacitance C2 and L2. The output side 52 may comprise a class-E rectifier, here represented schematically by diode D1, capacitance C3, and inductance L3. Input side 51 and output side are coupled by an isolating transformer T1. Advan-

- 5 tageously, the transformer T1 is a PCB embedded solenoid transformer 500 with a primary winding 501 and a secondary winding 502 inductively coupled to the primary winding 501, wherein the coupling designated by the double line may be a coreless coupling.
- 10 Fig.6 shows a self-oscillating class-DE power inverter 60 adapted for radio frequency operation, preferably in the VHF. The inverter 60 receives a DC-input voltage Vin at input terminals of an input stage 61, and provides an AC-output voltage in the radio frequency range, e.g. in the VHF, to a load at output terminals of an output stage 62. The input stage 61 may comprise semiconductor switches with control
- 15 terminals Q1, Q2 and respective drain-source capacitors C2, C3 in a half-bridge configuration, wherein the semiconductor switches are controlled to oscillate with opposite phase. As shown schematically in Fig.6, the oscillation may be driven by a self-oscillating circuit as described in the co-pending International Patent Application PCT/EP2013/072548, here illustrated by inductance L1 and C1, in combination with
- 20 coupled inductors M1 locking the control signals acting on the semiconductor switches Q1, Q2 to opposite phase. Advantageously, the coupled inductors M1 are a PCB embedded solenoid structure 600 with a first winding 601 and a second winding 602 inductively coupled to the first winding 601, wherein the coupling designated by the Z-shaped thick line may be a coreless coupling.

25

30

Figs.7-11 show further embodiments of resonant DC-DC converters comprising an embedded solenoid transformer coupling an input side of the converter to an output side. In all embodiments, the embedded transformers are made up of embedded PCB solenoid inductors that are coupled to form an embedded solenoid transformer as described above. In figs. 7–11 coupling is indicated by a double line, wherein the

coupling may be a coreless coupling.

Fig.7 shows a generic schematic of a resonant DC-DC converter 70 with an input side (inverter) 71 coupled to an output side (rectifier) 72 through an embedded iso-

lating transformer 700, wherein the embedded printed circuit board solenoid inductors 701, 702 are coupled to be used as the transformer 700 between the input side inverter 71 and the output side rectifier 72.

- 5 Fig.8 shows a generic schematic of a resonant DC-DC converter 80 with an input side inverter 81 coupled to an output side rectifier 82 through an embedded autotransformer 800. The autotransformer 800 is formed by coupled embedded printed circuit board solenoid inductors 801, 802. In applications where a high step ratio is required, but isolation is not, this configuration gives several advantages. The auto-
- 10 transformer 800 can be used to transform the impedance of the output side rectifier 82 to the desired output impedance for the input side inverter 81. This is particularly desirable in resonant converters as they are very dependent on the input and output impedances.
- 15 Fig.9 shows schematically an embodiment of a resonant DC-DC converter 90 with an input side inverter 91 coupled to an output side rectifier 92 through an embedded autotransformer 900. The resonant DC-DC converter 90 of Fig.9 differs from the resonant DC-DC converter 80 of Fig.8 in that the input side inverter 91 comprises a plurality of smaller inverters Inv1, Inv2. In the particular embodiment shown, the in-
- 20 put side inverter 91 has two smaller inverters Inv1, Inv2 coupled with serial input and parallel output, a configuration also referred to as "SIPO". The input side inverter 91 can also comprise an even higher number of smaller inverters. Furthermore, the plurality of smaller inverters may also be arranged with other combinations of parallel and serial input and output. The output side rectifier 92 may comprise multiple smaller rectifiers in the same way (not shown). Galvanic isolation in the sub-block (i.e. the smaller inverters and/or the smaller rectifiers) is a requirement if serial input or output is to be used. Generally, serial connection is desirable for high voltages
- 30 Fig.10 shows schematically an embodiment of a resonant DC-DC converter 100 with an input side inverter 101 and an output side rectifier 102 coupled together through an embedded isolating transformer 1000. The isolating transformer is formed by coupled embedded printed circuit board solenoid inductors 1001, 1002. In this particular embodiment of the resonant DC-DC converter 100, the input side in-

and parallel connection is desirable for high currents.

PCT/EP2014/079037

19

verter 101 comprises a class DE inverter and the output side comprises a class E rectifier 102. This topology is especially suitable for applications with high input voltage and low output voltages. This is due to the half bridge configuration on the input side 101, which limits the peak voltage across the two switches S1, S2 to the input

5 voltage, and the single diode rectifier on the output side 102, which minimizes the loss in the rectifier due to forward conduction loss in the diode. For example, this may be useful for a USB charging device, which needs to convert an input with a voltage of 110V/230V from a rectified mains input to an output of 5V, and which has to be galvanic isolated to properly protect a user of the charging device.

10

15

Fig.11 shows schematically an embodiment of a resonant DC-DC converter 110 with an input side inverter 111 and an output side rectifier 112. As in the embodiment of Fig.10, the input side inverter 111 comprises a class DE inverter and the output side rectifier comprises a class E rectifier. The embodiment 110 of Fig.11 differs from the embodiment 100 of Fig.10 in that the input side inverter 111 and the output side rectifier 112 are coupled together through an embedded autotransformer

- output side rectifier 112 are coupled together through an embedded autotransformer 1100. The autotransformer 1100 is formed by coupled embedded printed circuit board solenoid inductors 1101, 1102 as also shown in Fig.8 and Fig.9.
- 20 In applications where galvanic isolation is not a requirement it is generally more desirable to use the coupled inductors in an autotransformer configuration. Furthermore, for applications with low input voltages, inverters with just a low side switch, like switch Q1 in the input side inverter 51 shown in Fig.5, are typically more desirable, for example in battery driven applications like automotive, portable devices, etc.
- 25 Furthermore, for applications with high output voltage, an output side rectifier comprising a half bridge rectifier is preferable.

EXAMPLES

30 A transformer can be described by an inductance matrix.

 $\begin{bmatrix} v_p(t) \\ v_s(t) \end{bmatrix} = \begin{bmatrix} L_{11} & L_{12} \\ L_{21} & L_{22} \end{bmatrix} \frac{d}{dt} \begin{bmatrix} i_p(t) \\ i_s(t) \end{bmatrix}$

wherein v(t) and i(t) refer to time-dependent voltage and current, respectively, and indices p and s refer to primary-side and secondary-side, respectively.

5 An interpretation of the inductance matrix yields transformer parameters for characterizing a given transformer. The diagonal elements *L*11 and *L*22 of the inductance matrix are the self-inductances for each winding, and the off-diagonal elements *L*12 and *L*21 are the mutual inductances as determined by the mutual flux. Since the transformer is a passive device *L*12 and *L*21 are equal. The coupling factor k is giv-

10 en by

$$k = \frac{L_{12}}{\sqrt{L_{11}L_{22}}}$$

.

- In the following, examples are given for different types of layouts of PCB embedded transformers as described above with reference to Figs.1 – 4, wherein the number of turns of the transformers of the below examples may vary with respect to the transformers shown in the figures. The actual numbers of turns and corresponding geometry data of the measured transformers are given in each of the examples below. The geometry data comprise values for the width *w* of the transformer and/or
- 20 first and second solenoids. These widths w are given as the distance between the respective vertical centers of the vias in a horizontal direction transverse to the longitudinal direction. The geometry data further comprise the overall lengths l of the transformers and/or the first and second solenoids in the longitudinal direction. The respective heights h of the transformers and/or first and second solenoids are de-
- 25 termined by the PCB layer stack used for producing the respective embedded solenoid transformer. Depending on the transformer layout, the structures have been implemented in dual-layer PCB or in a four-layer PCB as also indicated in the examples below. The dual-layer PCB stack comprises an isolating core layer with a thickness of about 1600µm carrying an upper conductive layer made of copper with
- 30 a thickness of 35µm on one side, and a corresponding lower conductive layer made of copper with a thickness of 35µm on the opposite side. The four-layer PCB stack has a total thickness of about 1600µm and comprises an isolating inner layer with a

thickness of 710µm carrying inner upper and lower copper layers on either side. The inner copper layers have a thickness of 35µm each. Each of the inner copper layers is covered by a respective outer isolating layer of 7628 prepreg with a thickness of 360µm each. Finally, each of the outer isolating layers carries an outer conductive

5 layer made of copper foil with a thickness of 18µm.

The examples are transformers designed for operation in the VHF frequency range, and the different transformers are characterized by measured values for the abovementioned transformer parameters L11, L22, L12, L21, k. Measurements of the

- 10 transformer parameters L11, L22, L12, L21, k have been obtained using a network analyzer, wherein all measurements have been made in the frequency range 1MHz - 1.8GHz. Based on the measurement results a frequency range of operation is specified, for which the obtained transformer values are considered valid. Since the transformer is a passive device, L12 equals L21. Therefore only values for L12 are
- 15 given below.

Example 1

Example 1 refers to an end-to-end layout embedded in a dual layer PCB as illustrated above with respect to Figs.1a-c and in the following implementation:

20

	1 st solenoid	2 nd solenoid	transformer
Number of turns	5	5	
Width w / mm	10	10	10
Length <i>l</i> / mm	10	10	20
Height h / mm	1,6	1,6	1,6
Trace width t / mm	1,8	1,8	1,8

The following transformer parameters have been measured for this implementation:

L11 / nH	L12 / nH	L22 / nH	К
41,00	3,22	39,45	0,08

The values of the transformer parameters are valid for operation in the frequency range from 100MHz – 850MHz.

Example 2

- 5 Example 2 refers to section-wise interleaved layout embedded in a dual layer PCB as illustrated above with respect to Figs.2a-c and in the following implementation with one first solenoid and one second solenoid arranged on a common longitudinal axis. Both solenoids have five turns distributed over two sections, with two and three turns, respectively. When seen along the longitudinal direction, starting from one
- 10 end of the transformer, two turns of the first solenoid are followed by three turns of the second solenoid, which is followed by three turns of the first solenoid, which is again followed by two turns of the second solenoid.

	1 st solenoid	2 nd solenoid	transformer
Number of turns	5	5	
Number of sections	2	2	
Number of turns per section	2/3	2/3	
Width w / mm	10	10	10
Length <i>l</i> / mm			20
Height h / mm	1,6	1,6	1,6
Trace width t / mm	1,8	1,8	1,8

15 The following transformer parameters have been measured for this implementation:

L11 / nH	L12 / nH	L22 / nH	k
40,95	10,52	38,81	0,26

The values of the transformer parameters are valid for operation in the frequency range from 1MHz – 900MHz.

20

Example 3

Example 3 refers to an interleaved layout, wherein single turns of a first solenoid alternate with single turns of a second solenoid. The layout is embedded in a dual

layer PCB as illustrated above with respect to Figs.3a-c and in the following implementation:

	1 st solenoid	2 nd solenoid	transformer
Number of turns	10	10	
Width w / mm	10	10	10
Length 1 / mm			16,4
Height h / mm	1,6	1,6	1,6
Trace width t / mm	0,6	0,6	0,6

5 The following transformer parameters have been measured for this implementation:

L11 / nH	L12 / nH	L22 / nH	k
159,07	94,03	162,89	0,59

The values of the transformer parameters are valid for operation in the frequency range from 1MHz - 650MHz.

10

Example 4

Example 4 refers to a layout, wherein a first solenoid is enclosed inside a second solenoid. The layout is embedded in a four-layer PCB as illustrated above with respect to Figs.4a-c and in the following implementation:

15

	1 st solenoid	2 nd solenoid	transformer
Number of turns	19	19	
Width w / mm	23,4	25	25
Length <i>l</i> / mm	11	11	11
Height h / mm	0,71	1,6	1,6
Trace width t / mm	0,4	0,4	0,4

The following transformer parameters have been measured for this implementation:

L11 / nH	L12 / nH	L22 / nH	k
1154,33	1104,82	1506,57	0,84

The values of the transformer parameters are valid for operation in the frequency range from 1MHz - 25MHz.

CLAIMS

- 1. Resonant power converter for operation in the radio frequency range, preferably in the VHF, the power converter comprising at least one transformer, wherein
- 5 the at least one transformer comprises
 - a printed circuit board defining a horizontal plane, the printed circuit board comprising at least two horizontal conductive layers separated by an isolating layer,
 - a first embedded solenoid forming a primary winding of the transformer and
- 10
- a second embedded solenoid being arranged parallel to the first solenoid and forming a secondary winding of the transformer,

wherein the first and second embedded solenoids are formed in the conductive layers of the printed circuit board, wherein each full turn of an embedded solenoid has a horizontal top portion formed in an upper conductive layer, a horizon-

- 15 tal bottom portion formed in a lower conductive layer, and two vertical side portions formed by vias extending between the upper and the lower conducting layers.
- Power converter according to claim 1, wherein the transformer is a coreless
 transformer.
 - 3. Power converter according to any of the preceding claims, wherein the first and second solenoids are embedded on the same PCB.
- Power converter according to any of the preceding claims, wherein the first and second solenoids are wound around a common axis.
 - 5. Power converter according to claim 4, wherein the first and second solenoids at least partially overlap in a longitudinal direction.
- 30
- 6. Power converter according to any of the preceding claims, wherein top portions of the first and second solenoids are formed in a common upper conductive layer, and/or bottom portions of the first and second solenoids are formed in a common lower conductive layer.

- 7. Power converter according to claim 6, wherein the first and second solenoids are section-wise interleaved.
- 5 8. Power converter according to claim 7, wherein first sections of the first solenoid alternate with second sections of the second solenoid.
 - 9. Power converter according to claim 6 or claim 7, wherein first sections are single turns and/or second sections are single turns.
- 10

25

30

- 10. Power converter according to any of the preceding claims, wherein the PCB has at least four-layers.
- Power converter according to claim10, wherein when seen in a cross-sectional
 plane perpendicular to a longitudinal axis of the solenoids, the second solenoid
 is arranged inside the first solenoid or, alternatively the first solenoid is arranged
 inside the second solenoid.
 - 12. Power converter according to any of the preceding claims, wherein the trans-
- 20 former comprises further solenoids coupled to the first solenoid and/or the second solenoid.
 - Power converter according to any of the preceding claims, wherein the transformer is adapted for operation at a radio frequency in the range 1MHz – 1 GHz, alternatively 10 MHz – 500 MHz, further alternatively 30 MHz – 300 MHz.
 - 14. Power converter according to any of the preceding claims, wherein the transformer has a coupling factor k of above 0,1, preferably above 0,3, preferably above 0,4, preferably above 0,5, more preferably above 0,6, and most preferably above 0,8.
 - 15. Power converter according to any of the preceding claims, wherein at least one transformer is arranged as an isolating transformer between an input side and an output side.

- 16. Power converter according to any of the preceding claims, wherein at least one transformer is arranged as an autotransformer between an input side and an output side.
- 5
- 17. Power converter according to any of the preceding claims, wherein at least one transformer is arranged such that the the first solenoid is used as a power carry-ing inductor in the power converter and the second solenoid is used as a sensing inductor to measure the current through the first inductor.
- 10
- 18. Power converter according to any of the preceding claims, wherein the power converter is a resonant DC-DC converter.
- 19. Power converter according to any of the preceding claims, wherein a switching
 frequency on the input side is at least 10 MHz, preferably at least 20 MHz, more preferably at least 30 MHz.
 - 20. Power converter according to any of the preceding claims, wherein the power converter is an inverter.













Fig.2c

Fig.2b






















Fig.6







7/7







Fig.11

 A.19 M. Madsen, D. J. Perreault, A. Knott and M.
 A.E. Andersen: "Self-oscillating Galvanic Isolated Bidirectional Very High Frequency DC-DC Converter", Accepted for presentation at IEEE COMPEL 2015 with subsequent publishing in the conference proceedings.

Outphasing Control of Gallium Nitride based Very High Frequency Resonant Converters

Mickey P. Madsen, Arnold Knott, Michael A.E. Andersen Technical University of Denmark, Department of Electrical Engineering, Oersteds Plads 349, DK-2800 Kgs Lyngby, Denmark

Abstract — In this paper an outphasing modulation control method suitable for line regulation of very high frequency resonant converters is described.

The pros and cons of several control methods suitable for very high frequency resonant converters are described and compared to outphasing modulation. Then the modulation technique is described and the design equations given.

Finally a design example is given for a converter consisting of two class E inverters with a lossless combiner and a common half bridge rectifier. It is shown how outphasing modulation can be used for line regulation while insuring equal and purely resistive loading of the inverters. Combined with a proper design of the inverters that, insures they can achieve zero voltage switching across a wide load range, and gallium nitride FETs for the switching devices, this makes it possible to achieve more than 90% efficiency across most of the input voltage range with good line regulation.

Keywords — Gallium nitride, Phase control, Power control, VHF circuits, Zero voltage switching.

I. INTRODUCTION

The development of Switch-Mode Power Supplies (SMPS) has made it possible to increase the power density of power converters significantly. Modern power supplies are however limited by the passive energy storing elements, which is needed to store energy between each switching period. The size of these components scale inversely with the switching frequency, and there have therefore been a constant strive for higher switching frequencies ever since. Commercially available converters today switch at frequencies up to several megahertz and can have efficiencies of more than 95% (e.g. [1]).

The reason not to increase the switching frequency further, and thereby reaching even higher power densities, is the switching losses. For more than two decades (since 1988 [2]) research has been done, in order to enable the use of resonant RF amplifiers (inverters) combined with a rectifier for dc/dc converters, in order to avoid switching losses. With this type of converters, SMPSs with switching frequencies in the Very High Frequency range (VHF, 30-300MHz) have been designed with efficiencies up to approx. 90% [3], [4].

Several of the benefits and challenges of the increased switching frequency are described in [5], [6]. Among the benefits are higher power densities, lower weight and removed need for electrolytic capacitors and magnetic core materials

978-1-4673-6847-6/15/\$31.00 ©2015 IEEE

David J. Perreault

Massachusetts Institute of Technology, The Laboratory for Electromagnetic and Electronic Systems, 77 Massachusetts Avenue, MA-02139 Cambridge, USA

[7]. The increased frequency also enables the possibility of very fast transient responses, to achieve this a fast and efficient control scheme is needed. This together with the fact that resonant converters are very load dependent remains two of the major challenges for very high frequency resonant converters.

In section II of this paper different control schemes for resonant converters are investigated and compared. Section III describes a method for efficient outphasing control of resonant converters. Section IV covers the design and simulation of a converter with the described control and finally section V concludes the paper.

II. CONTROL OF RESONANT CONVERTERS

Resonant converters are dependent on precise timing between the resonances in the power stage and the switching of the power semiconductors. The resonances in the power stage are used to shape the voltages and currents, so that the voltage across or the current through the power semiconductors are zero at the switching instant, known as zero-voltage-switching (ZVS) and zero-current-switching (ZCS).

In power applications the energy stored in the parasitic capacitance of the switching device is generally the dominating contributor to switching losses, hence ZVS is crucial to insure high efficiency when the switching frequency is pushed into the VHF range.

Pulse width modulation (PWM) is the most commonly used control scheme for power converters, but if a resonant converter is controlled in this way, ZVS is lost as soon as the duty cycle moves away from one optimal value.

Pulse frequency modulation (PFM) is another option, but it requires a very wide frequency operation range to insure even a limited controllability of the converter. Combinations of PWM and PFM have shown good results [8], but the control circuit becomes quite complex and still requires a wide frequency operation range.

A. Burst mode control

So far the most commonly used control method for VHF converters has been burst mode control (also called cell modulation, on-off or bang-bang control) [4], [9]-[12]. Here the entire converter is switched on and off in order to control the output. The benefit of controlling the converter in this way is that the converter will either be on and working in its optimal operating point or off with only small standby losses. The result is a wide control range with an almost flat efficiency curve [4], [10]. Further the converter can be optimized for high

This research was funded by a grant from The Danish National Advanced Technology Foundation and performed in cooperation with The Laboratory for Electromagnetic and Electronic Systems (LEES) at Massachusetts Institute of Technology (MIT), funded under the MIT-SkTech Program.

peak efficiency in this specific operation point instead of compromising the peak efficiency for higher average efficiency.

One down side of burst mode control is that low frequency harmonics is introduced, which increase the size of the needed input and output filters. Another is added circuitry that adds both to complexity and to the losses, which reduces the efficiency at full load. Further burst mode requires the converter to be able to start and reach steady state quickly, most implementations use a burst frequency approximately two decades below the switching frequency in order to insure that the converter has time to reach steady state and operate efficiently for some time before it is shut down again [4], [9]-[12].

Several control schemes for burst mode control are available with hysteresis [4], [9]-[10] and PWM [11] being the most commonly used, phase shift [12] and constant on time are two other alternatives.

Hysteresis control has the best transient response, as the transitions between the on and off states are defined by the hysteresis window, instead of a fixed frequency or times as in the other schemes. Another advantages is very good efficiency at light loads due to a very low burst frequency. These advantages come at the expense of widely varying burst frequencies increases the complexity and potentially size of the needed input/EMI filter.

The design of the input filter is more straightforward with PWM where a fixed burst frequency is used, but the transient response and light load efficiency degrades. The performance of a VHF converter controlled by PWM burst mode, is in many aspects similar to a conventional hard switched converter operating at the burst frequency.

Phase shift control has the same pros and cons as hysteresis control, but the delay in the control circuitry is utilized in the design which makes it possible to use slower components than would have been needed with hysteresis control.

Constant on time control can be compared to PWM control, but with the on time fixed instead of the frequency. This gives a more stable frequency than seen for hysteresis and phase shift control, but still with good light load efficiency.

B. Outphasing

Outphasing, also referred to as Linear Amplification with Nonlinear Components (LINC), was introduced for RF amplifiers in the 1930's [13]. This control method has however only been used a few times in previous publication [14]-[15] and posses several advantages compared to burst mode.

Outphasing control utilizes a phase shift between two or more inverters to control the combined output through a common rectifier, as illustrated with two inverters in Fig. 1. In this way the individual inverters run continuously at a fixed frequency while the total output to the load is regulated.

With this control method all the benefits of the high switching frequency can be achieved, both fast transient response, wide control range and small input and output filters. The main drawback of outphasing is that the losses in the inverters are almost constant at light and full load, which decrease the light load efficiency significantly. Further the varying load and the inverters impact on each other is a challenge, this will be investigated further in section III.

III. PROPOSED OUTPHASING SYSTEM

Selecting the right combiner is crucial to achieve a good result with outphasing. Several approaches have been suggested [16]-[19], but the lossless combiner proposed in [20] (shown in Fig. 2) posses several benefits for use in resonant converters. The combiner only consists of an inductor and capacitor, if these are assumed ideal the combiner is lossless. Further the admittance seen by the two inverters can be made zero, insuring purely resistive loading of the inverters. As resonant converters are very load dependent and quickly looses ZVS when load reactively this is crucial in order to insure high efficiency.

The input impedance of the rectifier, R_{REC} , can be assumed purely resistive at the switching frequency. Further the output voltage of the two inverters, V_{INV1} and V_{INV2} , can be assumed equal to:

$$V_{INV1} = V_{IN} \cdot G_{INV} \cdot e^{-I \cdot \varphi} \tag{1}$$

$$V_{INV2} = V_{IN} \cdot G_{INV} \cdot e^{I \cdot \varphi}$$
(2)

Where V_{IN} is the input voltage, G_{INV} is a fixed gain between the input voltage to the converters and the peak output voltage.



Fig. 1. System view of a converter using outphasing modulation.



Fig. 2. Schematic of the lossless combiner for outphasing modulation.

The admittance seen by the two inverters is now written as:

$$Y_{1} = \frac{1}{X} - \frac{I \cdot e^{I \cdot \varphi} \cdot V_{OUT}}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(3)

$$Y_2 = \frac{1}{X} - \frac{I \cdot V_{OUT}}{X \cdot e^{I \cdot \varphi} \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(4)

Where G_{REC} is a fixed voltage gain from the peak input voltage of the rectifier to the output voltage, V_{OUT} . The imaginary part of the admittance is:

$$\Im(Y_1) = \frac{1}{X} - \frac{V_{OUT} \cdot \cos(\varphi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(5)

$$\Im(Y_2) = -\frac{1}{X} + \frac{V_{OUT} \cdot \cos(\varphi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(6)

Setting this to zero and solving for the phase, ϕ , gives:

$$\varphi = \arccos\left(\frac{V_{IN} \cdot G_{INV} \cdot G_{REC}}{V_{OUT}}\right)$$
(7)

If the phase is controlled according to this equation, the loading of the inverters will be purely resistive. The real part of the admittance is:

$$\Re(Y_1) = \Re(Y_2) = \frac{V_{OUT} \cdot \sin(\varphi)}{X \cdot V_{IN} \cdot G_{INV} \cdot G_{REC}}$$
(8)

From this the output power, P_{OUT} , can be calculated for a given value of the combiner reactance, X, and input and output voltage by:

$$P_{OUT} = \frac{G_{INV} \cdot V_{IN} \sqrt{V_{OUT}^{2} - V_{IN}^{2} \cdot G_{INV}^{2} \cdot G_{REC}^{2}}}{X \cdot G_{REC}}$$
(9)

From (8) it is seen that it is not possible to control the power, if the loading of the converters needs to be purely resistive. If the output voltage is kept fixed, the output power will however be a quadratic function of the input voltage that can be offset by the chosen reactance for the combiner.

IV. DESIGN AND SIMULATION

The outphasing modulation will be evaluated for use as line control of a resonant converter with the specifications given in table 1. The aim is to use outphasing to achieve max $\pm 20\%$ output power variation for a input variation of $\pm 33\%$.

TABLE I. SPECIFICATIONS FOR THE CONVERTER.			
Input voltage	Output voltage	Frequency	Output power
12-24 V	54 V	30 MHz	10-15 W

Equation (7) shows that it is only possible to adjust the phase to give a purely resistive loading of the inverters, if the input voltage multiplied by the combined gain of the inverter and rectifier is less than or equal to the output voltage. This is because outphasing modulation can only be used to reduce the output power, from the power that would be delivered if the two inverters where operating in phase.

Due to the high switching frequency class E inverters with only low side switches are chosen. Other suitable topologies, such as the class Φ_2 [4] or class DE [21], could have been chosen, but the class E inverter is chosen due to the low input voltage and its simplicity.

A half bridge rectifier with a gain of exactly $4/\pi$ is selected due to the high output voltage. The gain of a class E inverter is $\pi/2$. With these gains the phase, output power, load resistance and reactance can be calculated, results shown in Fig 3-6.

The value of the combiner reactances is set to 56 Ω , corresponding to a 300 nH inductor and 94 pF capacitor, in order to achieve a output power which is 14 W in average. The 14 W is based on the assumption that the total system efficiency will be around 90% and that 1.4 W hence will be lost in the converter. Fig. 5-6 clearly shows the purely resistive loading of the converter with a value between 28 and 110 Ω across the input voltage range. The complete circuit is shown in Fig. 7.



Fig. 3. The phase shift needed across the input voltage range for purely resistive loading of the inverters (in radians).



Fig. 4. The achieved output power across the input voltage range. The reactance of the combiner is set to 56 Ω to achieve the desired output power plus 10% that is expected to be losses.



Fig. 5. The load resistance seen by the inverters across the input voltage range.



Fig. 6. The load reactance seen by the inverters across the input voltage range, it is zero across the range as it should be.



Fig. 7. Schematic of the two inverters, the combiner and the half bridge rectifier.

The two inverters are designed to be identical and based on the methodology described in [22], but without the parallel resonant tank. The values of the input inductors, L_{I1} and L_{I2} , and the switch capacitances, C_{S1} and C_{S2} , are given by the equations:

$$L_{I1} = L_{I2} = \frac{Q_s \cdot R_{\min}}{k \cdot 2 \cdot \pi \cdot f_s}$$
(10)

$$C_{S1} = C_{S2} = \frac{1}{Q_S \cdot R_{\min} \cdot k \cdot 2 \cdot \pi \cdot f_s}$$
(11)

Where $R_{min} = 28 \Omega$, $f_s = 30$ MHz and Q_S and k are set to 0.7 and $\sqrt{2}$, respectively. This gives inductor values of 832 nH and capacitances of 192 pF. EPCs 8010 GaN FETs are selected for the switches and their parasitic output capacitance combined with additional 140 pF composes C_{S1} and C_{S2} . The values of the resonant tanks components, C_{T1} , C_{T2} , L_{T1} and L_{T2} , are also calculated based on (10) and (11), but here Q_S is set to 3.4 in order to insure that the currents running into the combiner is close to sine waves. MBR0560 schottky diodes are selected for the rectifier and their parasitic capacitance of 15 pF composes C_{R1} and C_{R2} . The inductor in the rectifier, L_R , is used to tune the rectifier to appear resistive at the switching frequency. Through a spice simulation the required value is found to be 350 nH. 2 μ F capacitors are selected for C_{O1} and C_{O2} . A complete bill-of-material is shown in table II.

TABLE II. BILL OF MATERIAL FOR THE CONVERTER.

Component	Value
L_{I1} and L_{I2}	832 nH
C_{S1} and C_{S2}	140 pF
S_1 and S_2	EPC 8010
C_{T1} and C_{T2}	40 pF
L_{T1} and L_{T2}	355 nH
L _C	300 nH
C _c	94 pF
D_1 and D_2	MBR0560
C_{R1} and C_{R2}	(parasitic)
L _R	350 nH
C_{01} and C_{02}	2 µF

A spice simulation of the deigned converter has been made and the waveforms are shown in Fig. 8. From the waveforms it can be seen that the two inverters is indeed loaded equally and that ZVS is achieved across the entire input voltage range. Further more the efficiency is high and only drops below 90% for 24 V input. The reason for the reduced efficiency at this point is that the resonating currents in the inverters increase with the input voltage and hence the losses increase, at the same time the output power drops as expected from Fig. 4. The output power is kept well within the desired range with a minimum at 10 W and a peak of 14 W.

V. CONCLUSION

This paper have covered a short description of the most commonly used control methods for resonant converters operating in the very high frequency range, followed by the design and evaluation of a converter where outphasing of two inverters is used to achieve line regulation.

The designed converter achieves very good performance with equally and purely resistive loading of the two inverters across the entire input voltage range. This combined with a design of the inverters that insures that they can operate with ZVS across a wide load range, insures efficient operation of the converter across the entire input voltage range. The aim was to achieve efficient line regulation with less than +/- 20% output power variation for an input variation of +/- 30%. This is achieved as the converters keeps within +/- 16.5% output power variation. Furthermore the design exhibits very high efficiency across the entire input voltage range and only drops below 90% at the maximum input voltage.

REFERENCES

- ON Semiconductor, "1.2A, 2MHz Automotive Buck Switching Regulator", NCV890100 - Datasheet, Rev. 2, February 2013
- [2] W. C. Bowman, J. F. Balicki, F. T. Dickens et al., "A resonant DC-to-DC converter operating at 22 megahertz", *IEEE Applied Power Electronics Conference and Exposition*, 1988, pp.3-11, 1-5 Feb 1988
- [3] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, "High-Frequency Resonant SEPIC Converter With Wide Input and Output Voltage Ranges", *IEEE Transactions on Power Electronics*, vol.27, no.1, pp.189-200, Jan. 2012
- [4] J.M. Rivas, O. Leitermann, Yehui Han, D.J. Perreault, "A very high frequency dc-dc converter based on a class phi2 resonant inverter", *IEEE Transactions on Power Electronics*, 2008, vol.26, no.10, pp.2980,2992, Oct. 2011
- [5] D.J. Perreault, Jingying Hu, J.M. Rivas, Yehui Han, O. Leitermann, R.C.N. Pilawa-Podgurski, A. Sagneri, C.R. Sullivan, "Opportunities and Challenges in Very High Frequency Power Conversion", *IEEE Applied Power Electronics Conference and Exposition*, 2009, vol., no., pp.1-14, 15-19 Feb. 2009
- [6] A. Knott, T.M. Andersen, P. Kamby, M.P. Madsen, M. Kovacevic, M.A.E. Andersen, "On the ongoing evolution of very high frequency power supplies", *IEEE Applied Power Electronics Conference and Exposition 2013*, vol., no., pp.2514,2519, 17-21 March 2013
 [7] M. Madsen, A. Knott, M. A.E. Andersen, "Low Power Very High
- [7] M. Madsen, A. Knott, M. A.E. Andersen, "Low Power Very High Frequency Switch-Mode Power Supply With 50 V Input and 5 V Output", *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp 6569-6080, December 2014
- [8] S. Hamamura, T. Ninomiya, M. Yamamoto, M. Katsuno, "Combined PWM and PFM control for universal line voltage of a piezoelectric transformer off-line converter," *IEEE Transactions on Power Electronics*, vol.18, no.1, pp.270,277, Jan 2003
- [9] J. M. Rivas, D. Jackson, O. Leitermann et al., "Design Considerations for Very High Frequency dc-dc Converters", *IEEE Power Electronics* Specialists Conference, 2006, vol., no., pp.1-11, 18-22 June 2006
- [10] R.C.N. Pilawa-Podgurski, A.D. Sagneri, J.M. Rivas, D.I. Anderson, D.J. Perreault, "Very-High-Frequency Resonant Boost Converters", *IEEE Transactions on Power Electronics*, vol.24, no.6, pp.1654,1665, June 2009
- [11] Jingying Hu, A.D. Sagneri, J.M. Rivas, Yehui Han, S.M. Davis, D.J. Perreault, "High-Frequency Resonant SEPIC Converter With Wide Input and Output Voltage Ranges", *IEEE Transactions on Power Electronics*, vol.27, no.1, pp.189,200, Jan. 2012



Fig. 8. Waveforms of the converter when controlled with outphasing.

- [12] M. Kovacevic, A. Knott, M. A.E. Andersen, "A VHF interleaved selfoscillating resonant SEPIC converter with phase-shift burst-mode control," *Applied Power Electronics Conference and Exposition (APEC)*, 2014 Twenty-Ninth Annual IEEE, vol., no., pp.1402,1408, 16-20 March 2014
- [13] H. Chireix, "High power outphasing modulation," *Proc. IRE*, vol. 23, no. 11, pp. 1370–1392, Nov. 1935.
- [14] A.S. Jurkov, L. Roslaniec, D.J. Perreault, "Lossless Multiway Power Combining and Outphasing for High-Frequency Resonant Inverters," *IEEE Transactions on Power Electronics*, vol.29, no.4, pp.1894,1908, April 2014
- [15] Godoy, P.A.; Perreault, D.J.; Dawson, J.L., "Outphasing Energy Recovery Amplifier With Resistance Compression for Improved Efficiency," *IEEE Transactions on Microwave Theory and Techniques*, vol.57, no.12, pp.2895,2906, Dec. 2009
- [16] F. H. Raab, "Efficiency of outphasing RF power-amplifier systems," *IEEE Trans. Commun.*, vol. 33, no. 10, pp. 1094–1099, Oct. 1985.
- [17] A. Birafane and A. Kouki, "On the linearity and efficiency of outphasing microwave amplifiers," *IEEE Trans. Microw. Theory Tech.*, vol. 52, no. 7, pp. 1702–1708, Jul. 2004.
- [18] J. Yao and S. I. Long, "Power amplifier selection for LINC applications," *IEEE Trans. Circuits Syst. II, Exp. Briefs*, vol. 53, no. 8, pp. 763– 767, Aug. 2006.
- [19] F. H. Raab, P. Asbeck, S. Cripps, P. B. Kennington, Z. B. Popovich, N. Pothecary, J. F. Sevic, and N. O. Sokal, "RF and microwave power amplifier and transmitter technologies—Part 3," *High-Frequency Electron.*, pp. 34–48, Sep. 2003.
- [20] D. J. Perreault, "A New Power Combining and Outphasing Modulation System for High-Efficiency Power Amplification," *IEEE Transactions* on Circuits and Systems I: Regular Papers, vol.58, no.8, pp.1713,1726, Aug. 2011
- [21] M.P. Madsen, A. Knott and M.A.E. Andersen, "Very High Frequency Half Bridge DC/DC Converter", *IEEE Applied Power Electronics* Conference and Exposition 2014
- [22] L. Roslaniec, D. J. Perreault, "Design of variable-resistance class E inverters for load modulation," 2012 IEEE Energy Conversion Congress and Exposition (ECCE), vol., no., pp.3226,3232, 15-20 Sept. 2012

Publications and Patents

A.20 M. Kovacevik and M. Madsen: "Burst Mode Control", WO2015128398 A1, September 3rd 2015 (12) INTERNATIONAL APPLICATION PUBLISHED UNDER THE PATENT COOPERATION TREATY (PCT)

(19) World Intellectual Property Organization

International Bureau



(43) International Publication Date 3 September 2015 (03.09.2015)

- (51) International Patent Classification: *H02M 3/155* (2006.01) *H02M 1/00* (2007.01)
- (21) International Application Number:
 - PCT/EP2015/053980
- (22) International Filing Date: 26 February 2015 (26.02.2015)
- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 14156917.8 27 February 2014 (27.02.2014) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: KOVACEVIC, Milovan; Aldershvilevej 8, 1., DK-2800 Kgs. Lyngby (DK). MADSEN, Mickey P; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK).
- (74) Agent: GUARDIAN IP CONSULTING I/S; Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).

(10) International Publication Number WO 2015/128398 A1

- (81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM, AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN, TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.
- (84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, ST, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

— with international search report (Art. 21(3))



WO 2015/128398 A1



Fig. 1

(57) Abstract: The present invention relates to a new method of power converter regulation, in particular regulation of very high frequency (VHF) power converters operating at frequencies in the MHz range, wherein accurate output regulation utilises inherent delays in the regulation loop, whereby, contrary to hysteresis on/off control, the new method does not require immediate responses to comparisons of a sense voltage to two reference voltages; rather, according to the new method, only one reference voltage is used, and delays in the feedback loop are allowed to cause some variation of an output of the power converter.

BURST MODE CONTROL

The present invention relates to a new method of power converter regulation, in particular regulation of very high frequency (VHF) power converters operating at frequencies in the MHz range.

5 BACKGROUND OF THE INVENTION

Reducing the physical size of electronic equipment in power applications is desired in order to add more features into existing products, integrate power converters in places normally unfit for such equipment, and reduce system cost. Increasing the operating frequency of the converter is a direct way of reducing the size of energy storage elements such as bulky

- 10 capacitors and inductors, which usually dominate the overall converter volume. Due to reduction in energy storage requirements the transient response is dramatically increased. LED lighting applications and point-of-load (PoL) converters particularly benefit from very high frequency (VHF) converters due to size, price, and weight reduction, and faster transient response.
- 15 Conventionally, burst mode control is used to control the output voltage or current of resonant power converters. Burst mode control allows the converter designer to optimize resonant power converters for operation in one operating point. The output voltage or current is controlled by turning the resonant power converters on or off as necessary to maintain constant output voltage or current. A disadvantage of burst mode control is that
- 20 the EMI performance is the same or worse compared to hard switched converters at the same modulation frequency. Typically, the modulation frequency ranges from 20 kHz to 1 Mhz.

Typically, prior art burst mode control is either hysteresis based, or pulse width modulation (PWM) based with constant switching frequency.

25 With hysteresis and PWM control, fast responses – ideally zero time delays - of the components of the regulation loop are required so that low cost components cannot be used.

The converter start-up circuit must also provide very little delay. As a consequence, passive start-up circuits (which have lower cost) are usually not an option.

30 Hysteresis based control results in tight output regulation, but requires a high cost, high performance comparator with very small propagation delays.

A conventional hysteretic burst mode control method of controlling a power converter, comprises the steps of

turning the power converter on when an absolute value of a sense voltage is less than or equal to an absolute value of a first reference voltage, and

turning the power converter off when the absolute value of the sense voltage is larger than or equal to an absolute value of a second reference voltage that is larger than the absolute value of the first reference voltage.

5

Typically, the difference between the first and second reference voltages is a predetermined fixed value.

SUMMARY OF THE INVENTION

The present invention provides a modulation method for obtaining accurate output

- 10 regulation of a power converter while exploiting inherent delays in the feedback loop. Contrary to hysteresis on/off control, i.e. hysteretic burst mode control, the new method does not require immediate responses to comparisons of a sense voltage to two reference voltages. Rather, according to the new method, only one reference voltage is used, and delays in the feedback loop are allowed to cause some variation of an output of the power
- 15 converter. The variation can be predicted and accounted for during the converter design process.

Thus, a new method of controlling a power converter is provided, comprising the steps of

turning the power converter on when an absolute value of a sense voltage is less than or equal to an absolute value of a reference voltage and a first time period has elapsed since a previous turn-off of the power converter, and

turning the power converter off when the absolute value of the sense voltage is larger than or equal to the absolute value of the reference voltage and a second time period has elapsed since a previous turn-on of the power converter.

In the following, the new method is termed phase-shift burst mode control due to the phase shift caused by the first and second time periods. According to the new method, hysteresis is provided in the time domain.

A power converter operating in accordance with the new method is also provided.

Thus, a new power converter is provided, comprising a control circuit coupled to compare a sense voltage with a reference voltage, and having a control output that is coupled to control turn-on and turn-off of the power converter in such a way that

the power converter is turned-on when the absolute value of the sense voltage is less than or equal to the absolute value of the reference voltage and a first time period has elapsed since a previous turn-off of the power converter, and

20

30

the power converter is turned-off when the absolute value of the sense voltage is larger than or equal to the absolute value of the reference voltage and a second time period has elapsed since a previous turn-on of the power converter.

The first time period may include a third time period of a predetermined duration that has
to elapse from the point in time when the absolute value of the sense voltage drops below the absolute value of the reference voltage until the power converter is turned-on.

The second time period may include a fourth time period of a predetermined duration that has to elapse from the point in time when the absolute value of the sense voltage raises above the absolute value of the reference voltage until the power converter is turned-off.

10 The predetermined durations of the third and fourth time periods may be different.

Throughout the following disclosure, the process of alternatingly turning a power converter on and off, is termed "modulation" of the power converter, and the time between two consecutive turn-on events is termed a "modulation period", and the rate of turn-on is termed the "modulation frequency".

15 The sense voltage may correspond to an output voltage of the power converter; or an output current of the power converter; or another desired characteristic of the power converter.

An output current of the power converter is a current consumed by a load connected to an output of the power converter.

- 20 The control circuit may perform turn-off of the power converter by turning at least one switch of the power converter off, e.g. by short-circuiting the gate of a Field-Effect-Transistor (FET) to its source; or, the base of a bipolar transistor to its emitter, and turn-on of the power converter by allowing the at least one switch of the power converter to turn-on again, e.g. by opening the short-circuit.
- 25 The control circuit may also perform turn-on and turn-off of a resonant power converter by changing the impedance or the loop-gain of the resonant part of the power circuit, whereby the resonant power converter is turned-off by changing the impedance or loop-gain to a first value at which the resonant part of the power circuit does not oscillate, and whereby the resonant power converter is turned-on by changing the impedance or loop-gain to a
- 30 second value at which the resonant part of the power circuit oscillates.

The control circuit may comprise one or more delays configured to provide at least part of one or more of the respective first and second time periods. Propagation delays in circuit components, such as propagation delays in a comparator, filtering elements, the power circuit, etc., may advantageously be incorporated into the appropriate time period.

The control circuit may comprise a comparator coupled to compare the sense voltage with the reference voltage, and an output of the comparator may constitute the control output that is coupled to control turn-on and turn-off of the power converter, and thus, the delays with which the comparator changes state of its output from high to low and vice versa, in

5 response to changed input(s), constitute part of the respective first and second time periods.

The delays with which the comparator changes state of its output from high to low and vice versa, in response to changed input(s), may provide at least part of the third and fourth time periods.

10 The output voltage of the power converter may be coupled to signal conditioning circuit, such as a low-pass filter, etc., configured to output the sense voltage.

The power converter may be of any known type of power converters, and in particular any type of VHF power converters, such as square wave power converters, quasi-square wave power converters, resonant transition power converters, resonant power converters, etc.

15 The power converter may have a plurality of power circuits operating in series and/or in parallel and/or interleaved, e.g. with inputs in series and/or parallel and/or with outputs in series and/or parallel. All or some of the plurality of power circuits may be controlled with a single control circuit.

A resonant power converter may be of any known type of resonant power converters, such
 as converters comprising: a class E inverter and a class E rectifier, a class DE inverter and a class DE rectifier, a class DE inverter and a class E rectifier, etc.; or, class EF2 (or class Φ) converters, resonant Single-Ended Primary-Inductor Converters (SEPIC), etc.

The power converter may be driven by an oscillator, or the converter may be selfoscillating. Further, the power converter may be interleaved.

- 25 Basing control of the output of the power converter on a single reference voltage and time periods during which the power converter does not change state from a turned-on state to a turned-off state, or vice-versa, leads to lowered performance requirements for control circuit components as compared to components of conventional control circuits, e.g. utilizing PWM-control or hysteresis control. E.g., low cost components and passive start-up aircuits may be used in the new central circuit.
- 30 circuits may be used in the new control circuit.

Furthermore, it is possible to include low-pass filtering, or other signal conditioning, of the output voltage for provision of the sense voltage, thus improving signal integrity.

Increasing at least one of the first and second time periods provided by the control circuit lowers modulation frequency of the control output signal controlling the on/off states of the

power converter. This in turn increases variation of the output voltage; however, output voltage variations may be lowered by provision of a suitable output filter at the output of the power converter.

Due to the time periods provided by the control circuit, the response time of the control circuit is longer than the response time of hysteresis control. Still, desired regulation is achieved within one modulation cycle.

Conventional control circuits, such as PWM-control or hysteresis control, rely on high-end components and on minimizing time delays in the control circuit, leading to higher cost and lower component availability than for the corresponding components of the new control

10 circuit.

5

For example, a comparator used in the new control circuit may be 9-10 times slower than a comparator used in a conventional control circuit, e.g. 4.5 ns vs 40 ns.

As further explained below, at least one of, or both of, the first and second time periods may be equal to, or substantially equal to, one fourth the modulation period, e.g. at 50 %

15 duty cycle, for example at least one of, or both of, the third and fourth time periods may be equal to, or substantially equal to, one fourth the modulation period, e.g. at 50 % duty cycle.

The sense voltage may be a function of the output voltage supplied to a load connected to an output of the power converter; or, the sense voltage may be a function of the output

20 current supplied to a load connected to an output of the power converter; or, the sense voltage may be a function of power, i.e. output voltage times output current, supplied to a load connected to an output of the power converter; etc.

An output capacitor connected to an output of the power converter stabilizes the output voltage supplied by the power converter. The output capacitor is charged during turn-on of

25 the power converter and discharged during turn-off of the power converter. During turn-on, the capacitor is charged with the current supplied by the power stage of the power converter minus the current supplied to the load connected to the output of the power converter. During turn-off, the capacitor is discharged with the load current. Thus, the slope of the voltage ripple ΔV_{out} developed across the output capacitor as a result of the charging

- 30 and discharging of the output capacitor depends on the output current. If the output current is low, the slope of the voltage ripple is steep during turn-on of the power converter and low during turn-off of the power converter, and vice versa if the output current is high. This together with the first and second time periods causes the output voltage ripple to vary with varying output current and also causes the mean value of the output voltage V_{out} to vary
- 35 with varying output current. Correspondingly, the ripple of the output current I_{out} and

25

output power P_{outr} and the mean value of output power P_{out} may also vary with varying output current.

This variation may be compensated in various ways in order to lower the amount of variation and making the desired output value more constant and independent of the output current at least within a predetermined output current range.

For example, the control circuit may be configured to compensate a dependence of the sense voltage, and thereby a dependence of the output voltage; or, the output current; or the output power; etc., on the output current, for example by varying the reference voltage in dependence on the output current in such a way that the dependence of the sense

- 10 voltage on the output current is compensated. For example, an increase in output voltage may be compensated by lowering the reference voltage so that the power converter is turned-on at a lower voltage thereby lowering the resulting output voltage. The change in reference voltage takes place over a plurality of modulation periods. When the load is constant, the reference voltage does not change unless an adjustment of the output, such as the output voltage or current, is performed.
 - Additionally; or, alternatively, the control circuit may be configured to vary at least one of the first and second time periods in dependence of the output current, whereby the dependence of the sense voltage on the output current is compensated. For example, an increase in output voltage may be compensated by lowering the second time period so that
- 20 the power converter is turned-on during a shorter time period thereby lowering the resulting output voltage. When the load is constant, the first and second time periods do not change.

BRIEF DESCRIPTION OF THE DRAWINGS

Below, the new method and the new power converter are explained in more detail with reference to the drawings in which various resonant examples of the new power converter are shown. In the drawings:

- Fig. 1 shows a schematic diagram of a resonant power converter with the new control circuit,
- Fig. 2 shows a schematic circuit diagram of an exemplary new resonant power converter,
- Fig. 3 shows a schematic circuit diagram of an exemplary new resonant power converter,
- 30 Fig. 4 shows a schematic circuit diagram of an exemplary new resonant power converter,
 - Fig. 5 shows a schematic circuit diagram of an exemplary new resonant power converter,
 - Fig. 6 shows exemplary signal conditioning circuits,
 - Fig. 7 shows exemplary switch drivers,

- Fig. 8 shows a schematic circuit diagram of an exemplary new resonant power converter with a stop circuit,
- Fig. 9 shows a switch driver with oscillator turn-off,
- Fig. 10 shows a power circuit diagram of a VHF interleaved self-oscillating resonant SEPIC converter,
- Fig. 11 is a plot of simulated waveforms of the power circuit shown in Fig. 10,
- Fig. 12 shows a schematic circuit diagram of a model of a new resonant power converter used for calculation of component values,
- Fig. 13 shows a circuit diagram of a control circuit used to control the power circuit shown in Fig. 10,
- Fig. 14 is a plot of characteristic waveforms of the model shown in Fig. 12,
- Fig. 15 is a plot of experimental waveforms of the control circuit shown in Fig. 13,
- Fig. 16 is a plot of experimental waveforms of the power circuit shown in Fig. 10,
- Fig. 17 (a) shows a plot of converter efficiency as a function of output power and (b) shows a plot of output voltage offset as a function of the load,
- 15

10

5

- Fig. 18 is a flowchart of the new method,
- Fig. 19 shows (a) a schematic circuit diagram of a model of the new resonant power converter together with (b) a conventional hysteretic control circuit and (c) a phase-shift burst mode control circuit,
- 20 Fig. 20 shows a plot of simulated waveforms of the model converter current and voltages controlled by the hysteretic control circuit,
 - Fig. 21 shows a plot of simulated waveforms of the model converter current and voltages controlled by the phase-shift burst mode control circuit, and
 - Fig. 22 shows schematic circuit diagrams of control circuits with varying sense voltage.
- 25 DETAILED DESCRIPTION OF EMBODIMENTS

The accompanying drawings are schematic and simplified for clarity, and they merely show details which are essential to the understanding of the new resonant power converter, while other details have been left out. The new resonant power converter according to the appended claims may be embodied in different forms not shown in the accompanying

30 drawings and should not be construed as limited to the examples set forth herein.

Like reference numerals refer to like elements throughout. Like elements may, thus, not be described in detail with respect to the description of each figure.

Fig. 1 shows a schematic block diagram of a resonant power converter 10 controlled in accordance with the new method. The illustrated resonant power converter 10 comprises a

5 VHF power circuit 12, a control circuit 14 with a control output 16, and a signal conditioning circuit 18 providing a sense voltage 20 based on and corresponding to the output voltage 22.

The control circuit 14 is coupled to compare the sense voltage 20 with a reference voltage 24.

10 Alternatively, the sense voltage 20 may be provided by a current sensor, such as a resistor, a hall element, etc., coupled so that the sense voltage corresponds to an output current of the resonant power converter 10.

The control circuit 14 has a control output 16 that is coupled to control turn-on and turn-off of the VHF power circuit 12 of the resonant power converter.

15 In the illustrated examples, the output voltage and the sense voltage have positive values, so that the absolute value of the sense voltage or the output voltage is equal to the value itself.

The VHF power circuit 12 of the resonant power converter 10 is turned-on when the sense voltage 20 is less than or equal to the reference voltage 24 and a first time period has

20 elapsed since a previous turn-off of the VHF power circuit 12. The VHF power circuit 12 of the resonant power converter 10 is turned-off when the sense voltage 20 is larger than or equal to the reference voltage 24 and a second time period has elapsed since a previous turn-on of the VHF power circuit 12.

The VHF power circuit 12 may be of any known resonant power converter topology with a

- 25 frequency of operation in the MHz range, such as at or above 20 MHz, such as at or above 30 MHz, such as in the 30 MHz – 300 MHz range, such as converters comprising: a class E inverter and a class E rectifier, a class DE inverter and a class DE rectifier, a class DE inverter and a class E rectifier, etc.; or, class EF2 (or class Φ) converters, resonant SEPIC converters, etc.
- 30 The resonant power converter may be driven by an oscillator, or the converter may be selfoscillating. Further, the resonant power converter may be interleaved.

Inherent signal propagation delays of the components of the control circuit 14 forms parts of the first and second time periods, and the inherent signal propagation delays may form the entire first and second time periods. Additionally, one or more delay circuits may

10

provide part of the first time period and/or part of the second time period, namely part of the third time period and/or part of the fourth time period.

The control circuit 14 may comprise a comparator that is coupled to compare the sense voltage 20 with the reference voltage 24, and having a comparator output that is the control output 16.

The delays with which the comparator changes state of its output from high to low and vice versa, in response to changed input(s), constitute part of the respective first and second time periods, namely part of the third time period and/or part of the fourth time period.

The signal conditioning circuit 18 may be a low-pass filter configured to output the sense voltage.

The control output 16 may be coupled to control turn-off of at least one power switch (not shown) of the resonant power converter 10 thereby turning the resonant power converter off.

This is illustrated in the class E inverter based resonant power converters shown in Figs. 2

and 3. The only difference between the resonant power converter of Fig. 2 and the resonant power converter of Fig. 3 is that inductor L_3 of Fig. 2 has been substituted by rectifier D_1 in Fig. 3.

Alternatively, or additionally, the control output 16 may be coupled to control other parts of the resonant power converter circuit than the power switches, e.g. by enabling and

20 disabling energy transfer from the input to the output of the resonant power converter 10, e.g. by turning the resonant power converter on and off by changing the impedance or the loop-gain of the resonant part of the power circuit, whereby the resonant power converter is turned-off by changing the impedance to a first value at which the power circuit does not oscillate, and whereby the resonant power converter is turned-on by changing the impedance to a second value at which the power circuit oscillates.

This is illustrated in Figs. 4 and 5 showing a class E inverter based resonant power converter similar to the class E inverter based resonant power converters shown in Fig. 3 except for the fact that the control output 16 turns S_2 on and off.

In Fig. 4, the resonant part of the resonant power converter 10 does not oscillate when
switch S₂ is turned-on, and the resonant part of the resonant power converter 10 oscillates and operates like the resonant power converter 10 of Fig. 3 when switch S₂ is turned-off.

In Fig. 5, the functions of rectifier D2 and switch S2 of Fig. 4 are combined in switch S2 functioning as a synchronous rectifier in Fig. 5 when the second self-oscillating gate driver is enabled so that the resonant power converter is turned on.

Fig. 6 shows circuit diagrams of two examples of low-pass filters that may constitute the signal conditioner 18.

The signal conditioners shown in Figs. 6 (a) and (b) may be substituted with any suitable signal conditioner chosen from other signal conditioner designs available to the person skilled in the art

5 skilled in the art.

Examples of self-oscillating gate drivers are shown in Fig. 7.

In Fig. 7(a), a low enable signal keeps the gate of the power switch at zero voltage, whereby oscillation of the power circuit is inhibited, while a high enable signal allows propagation of the VHF oscillator signal driving the gate of the power switch causing the power circuit to oscillate.

In Fig. 7(b), a low enable signal switches switch transistor S_{aux} off and keeps the gate of the power switch at a constant voltage, whereby oscillation of the power circuit is inhibited, while a high enable signal allows propagation of the VHF oscillator signal driving the switch transistor S_{aux} on and off thereby driving the power switch on and off causing the power

15 circuit to oscillate.

The gate drivers shown in Figs. 7 (a) and (b) may be substituted with any suitable gate driver chosen from other gate driver designs available to the person skilled in the art.

Fig. 8 shows a schematic circuit diagram of an exemplary new resonant power converter with a stop circuit,

20 Fig. 9 shows a switch driver with oscillator turn-off,

Fig. 10 is a circuit diagram of a power circuit 12 of an interleaved self-oscillating resonant SEPIC converter, wherein two power circuits 12a, 12b drive each other via capacitive coupling C_{χ_1} and C_{χ_2} between the switches S_1 and S_2 and operate in interleaved mode.

The power circuits are substantially identical, i.e.:

25

10

 $L_{I1} = L_{I2} = L_I$ $C_{I1} = C_{I2} = C_I$ $C_{X1} = C_{X2} = C_X$ $C_{S1} = C_{S2} = C_S$

The values of the most important parasitic components of the semiconductor devices,

30

namely diode junction capacitors and parasitic capacitors of the semiconductor switches, are included in determination of operating frequency of the resonant power converter. The

oscillation frequency f_S is determined mainly by the inductance L_I and the total capacitance seen from the drain when the rectifiers are shorted, $C_{DS,tot}$

$$f_s = \frac{1}{2\pi\sqrt{L_I C_{DS,tot}}}$$

where

5

 $C_{DS,tot} = C_I + C_{OSS} + C_S + C_X ||C_{ISS}$ $C_{OSS} = C_{DS} + C_{DG}$ $C_{ISS} = C_{GS} + C_{DG}$

Oscillations start once the gate voltages of the MOSFET switches S_1 and S_2 become slightly higher than the MOSFET threshold voltage. Simulated waveforms of the resonant power circuit of Fig. 10 are shown in Fig. 11. Ideally, respective waveforms of the two interleaved power converters are identical and shifted 180° with relation to each other.

If VHF ripple is neglected, the converter output can be modeled as a current source with the current value of I_0 . When an on/off modulation is applied on the converter, the current supplied by the converter i_{conv} to C_{out} and the load may be approximately modeled as a current square wave:

$$i_{conv} = \begin{cases} I_0, \text{ when the converter is ON} \\ 0, \text{ when the converter is OFF} \end{cases}$$

Output current I_{out} is equal to average value of i_{conv} over one modulation cycle. The resulting current going into C_{out} is i_{conv} - I_{out} , which has no DC component in steady state. If parasitics of C_{out} are negligible, the resulting V_{out} voltage waveform is a triangular wave. Assuming that turn-on and turn-off delays are independent of the output voltage rate of change, the output voltage ripple is

$$\Delta V_{out} = \frac{I_{out}}{C_{out}} t_{D,on} + \frac{I_0 - I_{out}}{C_{out}} t_{D,off}$$

where $t_{D,on}$ is the fourth time period and $t_{D,off}$ is the third time period, i.e. $t_{D,on}$ and $t_{D,off}$ are the control loop turn-on and turn-off delays, respectively, from the sense voltage crossing the reference voltage and to turn-on or turn-off, respectively, of the power converter. If the

control circuit's delays are constant, the equation shows that V_{out} is a linear function of I_{out} , and the longer delay defines $V_{out,max}$. In the special case of $t_{D,on}$ and $t_{D,off}$ being equal, V_{out} is independent of I_{out} . At any given load, the offset of V_{out} : $\Delta V_{out,off}$, and f_M are determined from

the values of C_{out} and the delays $t_{D,on}$ and $t_{D,off}$ by:

15

$$\Delta V_{out,off} = \frac{\Delta V_{out}}{2} \frac{t_{D,on} - t_{D,off}}{t_{D,off} + t_{D,on}}$$

and

5

$$f_M = \frac{I_0}{C_{out} \Delta V_{out}} \frac{I_{out}}{I_0} \left(1 - \frac{I_{out}}{I_0}\right)$$

Modulation frequency is highest at 50% duty cycle, i.e. the power converter is turned on half the time:

$$f_{M,max} = \frac{1}{2\left(t_{D,on} + t_{D,off}\right)}$$

 I_0 is not known from values of circuit components. An approximate value of I_0 can be determined from Spice simulations. Once I_0 is obtained, the output filter and the feedback circuit need to be designed to provide a desired modulation frequency f_M at a specified load.

10 Fig. 12 shows a block diagram of such a low frequency model of a resonant power. The VHF power circuit 12 is modelled as an on-off controllable DC current source.

The conditioning circuit 18 at the input of the control circuit 14, see Fig. 13, has a simple transfer function:

$$H(s) = \frac{A_{FB}}{1 + s\tau_{FB}}$$

Where

$$A_{FB} = \frac{R_{FB2}}{R_{FB1} + R_{FB2}}$$
$$\tau_{FB} = \frac{R_{FB1}R_{FB2}}{R_{FB1} + R_{FB2}} C_{FB}$$

15 The comparator 26 of the control circuit 14 model is ideal; the propagation delay of the real comparator is added into the delay block. The delay block is represented by two different time delays, since the shutdown of the power circuit is significantly faster than the start-up sequence. This is because shutdown is performed by the auxiliary switches S_{aux1} and S_{aux2} , while during start-up C_{ISS} is passively charged from the bias voltage V_B through the biasing resistors.

Fig. 13 shows the control circuit 14. The sense voltage 20 is low-pass filtered in H(s), a voltage divider/low pass filter 18 formed by R_{FB1} , R_{FB2} , and C_{FB} , and input to the comparator 26. The comparator output 16 turns the switches S_{aux1} and S_{aux2} on and off in response to

the voltage difference at the comparator inputs. When S_{aux1} and S_{aux2} are on, v_{GS1} and v_{GS2} are zero and the power circuit oscillations are inhibited. Once S_{aux1} and S_{aux2} are off, C_{ISS1} and C_{ISS2} are charged through R_{B1} and R_{B2} , whereby v_{GS1} and v_{GS2} start to increase from 0 to V_B . After the first time period, when v_{GS1} and v_{GS2} exceed the power MOSFET threshold

5 voltage V_{th} , switches S_1 and S_2 enter saturation and initiate oscillations in the power circuit. H(s) has two primary purposes: to filter high frequency noise and adjust the feedback voltage level. It also contributes to propagation delay in the feedback loop.

In Fig. 14, characteristic voltage and current levels from a numerical example of the model are shown, where

- $I_0 = 1.04 \text{ A}, I_{out} = 0.52 \text{ A}$
- $R_{FB2} = 2 \text{ k}\Omega, R_{FB1} = 8.2 \text{ k}\Omega$
- C_{FB} = 220 pF
- $t_{d,on} = 700 \text{ ns} + 170 \text{ ns} = 870 \text{ ns}$

15 •
$$t_{d,off} = 170 \text{ ns}$$

The parameters are chosen to approximate the experimental setup described below. $v_{gate}(t)$ represents the gate voltages of S₁ and S₂ with removed VHF component. $V_{out}(t)$ passes through the single pole transfer function H(s) and results in a distorted triangular waveform $v_{FB}(t)$. Average value of $v_{FB}(t)$ is slightly lower than the reference V_{ref} voltage, which is due

- 20 to $t_{d,on} > t_{d,off}$. This is also the cause of the duty cycle of the comparator output $v_{cmp}(t)$ to be lower than 50%. Since the referent output voltage is 10 V, a small offset can be observed in $V_{out}(t)$. This offset is dependent on the duty cycle of the power circuit, the time difference $t_{d,on} - t_{d,off}$, and C_{out} . $t_{d,on}$ depends on the voltage difference between V_B and V_{th} . Obtained modulation frequency is very close to 300 kHz.
- 25 In order to verify the analysis explained above, a 10.5 W prototype converter with regulation of the output voltage was produced and the measurements for the prototype converter is disclosed below.

Plots of drain, gate, and rectifier voltages in the power circuit are shown in Fig. 15. The drain and rectifier voltages were measured with capacitance of 2.2 pF in series with an oscilloscope probe, in order to reduce influence of the probe to the power circuit. This introduces attenuation of A = 0.19 in the measurement and removes the DC component of the measured voltages. Component values of the power circuit are listed in Table I. Switching frequency of the power circuit is $f_s = 49$ MHz. Open loop output voltage and output voltage and current are $V_{out} = 10.2$ V and $I_0 = 1.04$ A, respectively.

When the duty cycle of 50% is obtained, $I_{out} = 0.5 I_0 = 0.52$ A. Plots of the waveforms of the relevant voltages in the converter for this case are shown in Fig 9. Modulation frequency f_M is at its maximum value of 300 KHz at 50% duty cycle, and drops as the duty cycle moves away from 50%. In addition, under these conditions output voltage ripple ΔV_{outrpp} is at its maximum as well

5 maximum as well.

The comparator used in the circuit is AD8468 from Analog Devices. The component datasheet specifies 40 ns propagation delay. For comparison, a high speed TLV3501 comparator has a 4.5 ns propagation delay, which is a reduction by a factor of 9. This is by no means a limit since there are other significant contributors as well (conditioning and

10 power circuit on-off circuit). This delay may be increased even further at a cost of a lower

Figs. 15 and 16 illustrate relationships between the signals in time domain, v_{cmp+} , v_{cmp-} , v_{GS} , v_{DS} , and v_{out} (upper waveforms) with respect to the comparator output (lower waveforms). High output voltage ripple is caused by a small output filtering capacitance ($C_{out} = 3.3 \mu$ F).

15 Efficiency of the converter is shown in Fig. 17(a) as a function of output power. Since f_M is allowed to drop significantly under light load conditions, efficiency is maintained high over wide load range (η >75%above 20% load) with peak efficiency above 81%.

modulation frequency f_M and higher output voltage ripple for a given C_{out} .

Since the control is based on phase shift, a small DC error is introduced in the value of v_{out} , which varies with the load. If the output voltage ripple is assumed triangular (which is

20 reasonable since i_{conv} is a current square wave), the peak values of v_{out} are determined as:

$$V_{out+} = V_{out,ref} + \frac{I_0 - I_{out}}{C_{out}} \Delta t_+$$
$$V_{out-} = V_{out,ref} + \frac{-I_{out}}{C_{out}} \Delta t_-$$

So that the offset of V_{out} is determined by

$$V_{out,offset} = V_{out,ref} - \frac{\Delta V_{out+} + \Delta V_{out-}}{C_{out}} \Delta t_{-}$$

 $v_{out,ref}$ is a target value for the output voltage set by V_{ref} and $R_{FB1} - R_{FB2}$ voltage divider. Δt_+ and Δt_- are the time delays from a point when v_{out} crosses $v_{out,ref}$ to a point where v_{out} reaches its maximum and minimum value, respectively. Depending on the variables in these equations, $v_{out,offset}$ may be either positive or negative, and decreases with I_{out} . Measured dependence of Δv_{out} is shown in the lower plot of Fig. 17. Both the offset and the output ripple are reduced with increase in C_{out} , while the switching frequency will be reduced.

A comparison between the model disclosed above and experimental results show close, but not perfect matching. The reasons for this are subjects of further investigation; it is

assumed that imperfections in the active components and tolerances of the passive components are the main contributors. Still, the model gives significant insight into the system operation, and can be used as a good estimate during the converter design.

Compared to hysteresis based burst mode control, the new method of controlling the
resonant power converter allows use of a significantly slower and less expensive
components in the control circuit, which is of importance for cost sensitive applications such as LED lighting and PoL converters. The illustrated power circuits and control circuits were
implemented using only low cost commercially available components, with peak efficiency above 81% and high efficiency over wide load range.

10 Fig. 18 is a flowchart 100 of the new method of controlling a resonant power converter. The method starts with method step 110 wherein the resonant power converter is turned on.

When the resonant power converter is turned on, the output voltage and/or output current increases. A sense voltage is provided in the resonant power converter that corresponds to the output voltage or output current, and a reference voltage is provided in the resonant

15 power converter that corresponds to a desired resulting output voltage or output current of the resonant power converter.

According to method step 120, the output voltage and/or output current continue to increase until the corresponding sense voltage V_{sense} is equal to or larger than the reference voltage V_{ref} , and

20 according to method step 130, the output voltage and/or output current continue to increase until also a second time period t_2 has elapsed since a previous turn-on of the resonant power converter.

Thus, when the sense voltage V_{sense} is equal to or larger than the reference voltage V_{ref} , and a second time period t_2 has elapsed since a previous turn-on of the resonant power converter, the resonant power converter is turned-off in method step 140.

When the resonant power converter is turned on, the output voltage and/or output current decreases.

According to method step 150, the output voltage and/or output current continue to decrease until the corresponding sense voltage V_{sense} is equal to or less than the reference

30 voltage V_{ref}, and

25

according to method step 160, the output voltage and/or output current continue to decrease until also a first time period t_1 has elapsed since a previous turn-off of the resonant power converter.

Thus, when the sense voltage V_{sense} is equal to or less than the reference voltage V_{ref} , and a first time period t_1 has elapsed since a previous turn-off of the resonant power converter, the resonant power converter is turned-off in method step 110.

In the following, properties of the phase-shift burst mode control method for very high

- 5 frequency (VHF) DC-DC converters are compared with a conventional control method with hysteresis based on comparison of a sense voltage with two threshold values. Again, an onoff controllable current source is used to model the low-frequency behaviour of VHF converters. Large output capacitance is used for output voltage filtering. The model is shown in Fig. 19(a).
- 10 Fig. 19(b) shows the model of Fig. 19(a) controlled with a conventional hysteretic control circuit and Fig. 19(c) shows the model of Fig. 19(a) controlled with a phase-shift burst mode control circuit.

For both circuits, the VHF power converter is operating (turned-on) when V_{ctrl} is high, e.g. 2 Volt, delivering an average current of $I_0 = 2$ A. The VHF power converter is turned-off when

- 15 V_{ctrl} is low, e.g. 0 Volt. In the present example, capacitor C_{out} and load R_{load} are set to 10 µF and 10 Ω , respectively. Reference voltage V_{ref} equals 1 V, high impedance signal conditioning network H(s) has ratio of $V_{sense}/V_{out} = 1$: N, e.g. 1 : 10, independent of frequency. The circuit configurations and component values are selected so that the target output V_{out} is equal to 10 V and at nominal load, the VHF power converter operates at a 50
- 20 % duty cycle. It is assumed that gate drivers do not introduce any delay in any of the circuits of Figs. 19 (a) (c) so that $V_{ctrl} = V_{cmp}$.

In the control circuit with hysteresis shown in Fig. 19(b), instantaneous response of the comparator is assumed so that the comparator does not add a delay in the control loop. In the present example, the hysteresis window is $V_H = 20$ mV. The comparator turns the

converter on when V_{sense} < V_{ref} and turns it off otherwise. Simulated waveforms of the converter voltages of the converter with hysteresis control are shown in Fig. 20.
 Modulation frequency f_M, i.e. the frequency at which the converter turns on and off, is given

by:

$$f_M = \frac{I_0/2}{2 \,\Delta V_{out} \,C_{out}} = \frac{I_0/2}{2 \,N \,V_H \,C_{out}}$$

30 This equation is derived under the assumption that propagation delay t_D of the comparator, the gate driver, and the power stage of the converter is zero, and the equation is a good approximation when $t_D \ll 1/f_M$. Expensive components, e.g. comparators, gate drivers, etc., have to be used in VHF designs to realize small propagation delays in the feedback loop.

Time difference ΔT from $V_{sense} = V_{ref}$ until $V_{sense} = V_{ref} + V_H$ in Fig. 20 equals $1/(4 f_M)$. If $V_H \rightarrow 0$ and $t_D \rightarrow 1/(4 f_M)$, the waveforms shown in Fig. 20 turn into the waveforms shown in Fig. 21 showing simulated waveforms of the converter voltages of the converter with phase shift burst mode control. The waveforms in Figs. 20 and 21 are aligned so that V_{sense} voltages cross the respective V_{ref} voltages at the same time, e.g. 2 µs, 4 µs, 6 µs, etc. The

5 voltages cross the respective V_{ref} voltages at the same time, e.g. 2 µs, 4 µs, 6 µs, etc. The output voltage ripples are identical in Figs. 20 and 21, however the signal propagation delay through the feedback loop is $1/(4 f_M)$, see the time difference between converter currents $I_{conv,H/PS}$, comparator output voltages $V_{cmp,H/PS}$, etc.

With the component values mentioned above, the modulation frequency is 250 kHz.

- In Fig. 21, the power converter is turned-on at 3 μ s, 7 μ s, 11 μ s, and 15 μ s, and the power converter is turned-off at 1 μ s, 5 μ s, 9 μ s, and 13 μ s, and the first time period is equal to the second time period that is equal to 2 μ s and the fourth time period, i.e. the turn-on delay $t_{D,on}$, is equal to the third time period, i.e. the turn-off delay $t_{D,off}$, that is equal to 1 μ s.
- 15 The converter with phase shift burst mode control provides the same output voltage ripple (i.e. $\Delta V_{out,H} = \Delta V_{out,PS}$) as the converter with conventional hysteresis control, while using one or more components in the feedback loop with significantly larger respective delays. The resulting delay may be distributed arbitrarily between the power stage, the comparator, and the gate driver(s). This is very important for VHF converters, since numerous start-up
- 20 and shutdown techniques (self-oscillating gate drivers and converters) with small, but finite delays may be utilized in a VHF converter with phase shift burst mode control.

Turn-on and turn-off delay ($t_{D,on}$ and $t_{D,off}$) do not need to be equal, either. Any one of the components in the feedback loop in any combination may contribute to the values of the turn-on and turn-off delays, with the restraint that:

$$t_{D,on} + t_{D,off} = \frac{1}{2f_M}$$

25

30

Output voltage of a converter with hysteresis control resides within the range from $H(s)^{-1}(V_{ref,H} - V_H)$ to $H(s)^{-1}(V_{ref,H} + V_H)$, resulting in a voltage ripple of $\Delta V_{out,H}$.

This is different of a converter with phase shift burst mode control as illustrate in Fig, 19 (c) in which the output voltage V_o is load dependent. The output voltage V_o decreases with increased load. If output voltage ripple at $I_{out} = 1$ A is $\Delta V_{out,PS}$ and under assumptions of constant delay in the comparator (and otherwise ideal components), average value of $V_{out,PS}$

ranges from
$$\Delta V_{out,PS}/2$$
 at $I_{out} = 0$ to $-\Delta V_{out,PS}$ at $I_{out} = I_0$ linearly with I_{out} .

The variation of the output voltage V_o as a function of output current may be decreased in various ways. One way is to allow the reference voltage to change as a function of the output voltage V_o to compensate for the change in the output voltage V_o .

Fig. 22(a) shows a control circuit in which the sense voltage 20 is compared to a reference voltage 28 that is adjusted as a function of the mean output voltage V_o . $H_1(s)$ is an ordinary low-pass filter. In the illustrated circuit $V_{ref1} = V_{ref} - (V_{ref,mean} - V_{ref})$, where $V_{ref,mean}$ is the output of the low-pass filter $H_1(s)$. Thus, $V_{ref1} = V_{ref}$ when $V_{ref,mean} = V_{ref}$ and V_{ref1} decreases when the mean output voltage V_o increases and vice versa whereby the variation of the mean output voltage V_o as a function of output current is counteracted.

10 Fig. 22(b) shows another control circuit in which the sense voltage 20 is compared to a reference voltage 28 that is adjusted as a function of the mean of the comparator output voltage $V_{cmp,PS}$ 16. The operation of the illustrated control circuit is based on the fact that the average value of $V_{cmp,PS}$ varies linearly with the output current I_{out} .

Reference voltage V_{ref1} is formed by superposition of $V_{ref,PS}$ and low-pass filtered $V_{cmp,PS}$. The 15 resistor and capacitor values in the compensation network need to be chosen to provide sufficient attenuation of the ac component of $V_{cmp,PS}$.

In the illustrated control circuit, when the converter operates at 50% modulation, the average of $V_{cmp,PS}$ equals V_{ref} . If the output current I_{out} is reduced, the average value of $V_{cmp,PS}$ is reduced, thus decreasing reference voltage V_{ref1} 28 thereby counteracting the

20 increase of the output voltage V_o that would otherwise result for the reduced output current I_{out} . If the output current I_{out} is increased, the average value of $V_{cmp,PS}$ is increased, thus increasing reference voltage V_{ref1} 28 thereby counteracting the decrease of the output voltage V_o that would otherwise result from the increased output current I_{out} . V_{ref1} is calculated with the following formulae:

$$V_{ref1} = V_{ref,PS} \frac{R_1 || (R_2 + R_3)}{R_1} + V_{cmp,PS,high} \frac{I_{out}}{I_0} \frac{R_1 || (R_2 + R_3)}{R_1}$$

25

CLAIMS

25

- 1. A power converter comprising a control circuit coupled to compare a sense voltage with a reference voltage, and having a control output that is coupled to control turn-on and turn-off of the power converter in such a way that
- 5 the power converter is turned-on when an absolute value of the sense voltage is less than or equal to an absolute value of the reference voltage and a first time period has elapsed since a previous turn-off of the power converter, and the power converter is turned-off when the absolute value of the sense voltage is larger than or equal to the absolute value of the reference voltage and a 10 second time period has elapsed since a previous turn-on of the power converter.
 - 2. A power converter according to claim 1, wherein the control circuit comprises a comparator that is coupled to compare the sense voltage with the reference voltage, and having a comparator output that is the control output.
- 15 3. A power converter according to claim 1 or 2, wherein the power converter is configured for operation at a frequency above 1 MHz.
 - 4. A power converter according to any of the previous claims, comprising a delay circuit configured for providing at least part of at least one of the respective time periods.
- 5. A power converter according to any of the previous claims, wherein an output voltage
 of the power converter is coupled to a signal conditioning circuit configured to output the sense voltage.
 - 6. A power converter according to any of the previous claims, wherein the control output is coupled to control turn-off of at least one power switch of the power converter.
 - 7. A power converter according to any of claims 1 6, wherein the control output is
 - coupled to control disconnection of energy transfer from the input to the output of the power converter.
 - 8. A power converter according to any of claims 1 6, wherein the control output is coupled to control connection of energy transfer from the input to the output of the power converter.
- A power converter according to any of the previous claims, wherein the power converter is a SEPIC converter.
 - 10. A power converter according to claim 9, wherein the SEPIC converter is a self-oscillating SEPIC converter.

- 20
- 11. A power converter according to claim 9 or 10, wherein the power converter is an interleaved SEPIC converter.
- 12. A power converter according to any of the previous claims, wherein the control circuit is configured to compensate a dependence of the sense voltage on an output current.
- 5 13. A power converter according to claim 12, wherein the control circuit is configured to vary the reference voltage in dependence on the output current, whereby the dependence of the sense voltage on the output current is compensated.
 - 14. A power converter according to claim 12, wherein the control circuit is configured to vary at least one of the first and second time periods in dependence of the output
- 10 current, whereby the dependence of the sense voltage on the output current is compensated.
 - 15. A method of controlling a power converter, comprising the steps of

turning the power converter on when an absolute value of a sense voltage is less than or equal to an absolute value of a reference voltage and a first time period has elapsed since a previous turn-off of the power converter, and

turning the power converter off when the absolute value of the sense voltage is larger than or equal to the absolute value of the reference voltage and a second time period has elapsed since a previous turn-on of the power converter. 1/22





2/22














(a)



(b)

Fig. 6



(a)



(b)









Fig. 11







Fig. 14



Fig. 15







(b)

Fig. 17



Fig. 18



(a)



(b)



Fig. 19





Fig. 21









Fig. 22

A.21 M. Madsen and M. Kovacevik: "On and Off Controlled Resonant dc-dc Power Converter", WO2015128397 A1, September 3rd 2015 (12) INTERNATIONAL APPLICATION PUBLISHED UNDER THE PATENT COOPERATION TREATY (PCT)

(19) World Intellectual Property Organization

International Bureau



(10) International Publication Number WO 2015/128397 A1

(43) International Publication Date 3 September 2015 (03.09.2015)

- Ho2M 3/158 (2006.01)
 H03F 3/217 (2006.01)

 H02M 3/337 (2006.01)
 H02M 1/00 (2007.01)

 H02M 3/338 (2006.01)
 H02M 1/00 (2007.01)
- (21) International Application Number:
 - PCT/EP2015/053977

(22) International Filing Date: 26 February 2015 (26.02.2015)

- (25) Filing Language: English
- (26) Publication Language: English
- (30) Priority Data: 14156928.5 27 February 2014 (27.02.2014) EP
- (71) Applicant: DANMARKS TEKNISKE UNIVERSITET [DK/DK]; Anker Engelundsvej 1, Bygning 101A, 2. sal, DK-2800 Kgs. Lyngby (DK).
- (72) Inventors: MADSEN, Mickey P; Kollegiebakken 15A, st. 63, DK-2800 Kgs. Lyngby (DK). KOVACEVIC, Milovan; Aldershvilevej 8, 1., DK-2800 Kgs. Lyngby (DK).
- (74) Agent: GUARDIAN IP CONSULTING I/S; Diplomvej, Building 381, DK-2800 Kgs. Lyngby (DK).

201

(81) Designated States (unless otherwise indicated, for every kind of national protection available): AE, AG, AL, AM, AO, AT, AU, AZ, BA, BB, BG, BH, BN, BR, BW, BY, BZ, CA, CH, CL, CN, CO, CR, CU, CZ, DE, DK, DM, DO, DZ, EC, EE, EG, ES, FI, GB, GD, GE, GH, GM, GT, HN, HR, HU, ID, IL, IN, IR, IS, JP, KE, KG, KN, KP, KR, KZ, LA, LC, LK, LR, LS, LU, LY, MA, MD, ME, MG, MK, MN, MW, MX, MY, MZ, NA, NG, NI, NO, NZ, OM, PA, PE, PG, PH, PL, PT, QA, RO, RS, RU, RW, SA, SC, SD, SE, SG, SK, SL, SM, ST, SV, SY, TH, TJ, TM, TN, TR, TT, TZ, UA, UG, US, UZ, VC, VN, ZA, ZM, ZW.

(84) Designated States (unless otherwise indicated, for every kind of regional protection available): ARIPO (BW, GH, GM, KE, LR, LS, MW, MZ, NA, RW, SD, SL, ST, SZ, TZ, UG, ZM, ZW), Eurasian (AM, AZ, BY, KG, KZ, RU, TJ, TM), European (AL, AT, BE, BG, CH, CY, CZ, DE, DK, EE, ES, FI, FR, GB, GR, HR, HU, IE, IS, IT, LT, LU, LV, MC, MK, MT, NL, NO, PL, PT, RO, RS, SE, SI, SK, SM, TR), OAPI (BF, BJ, CF, CG, CI, CM, GA, GN, GQ, GW, KM, ML, MR, NE, SN, TD, TG).

Published:

- with international search report (Art. 21(3))

(54) Title: ON AND OFF CONTROLLED RESONANT DC-DC POWER CONVERTER



(57) Abstract: The present invention relates to a resonant DC-DC power converter comprising an input side circuit comprising a positive and a negative input terminal for receipt of an input voltage or current and an output side circuit comprising positive and negative output terminals for supply of a converter output voltage and connection to a converter load. The resonant DC-DC power converter further comprises a rectification circuit connected between an output of a resonant network and the output side circuit. The resonant network is configured for alternatingly being charged from the input voltage or current and discharged through the rectification circuit by a first controllable switch arrangement in accordance with a first switch control signal. A second controllable switch arrangement of the resonant network is configured to adjust the converter output voltage and/or current by activating and interrupting the first switch control signal in accordance with the switch state of the second controllable switch arrangement.

PCT/EP2015/053977

ON AND OFF CONTROLLED RESONANT DC-DC POWER CONVERTER

The present invention relates to a resonant DC-DC power converter comprising an input side circuit comprising a positive and a negative input terminal for receipt of an input voltage or current and an output side circuit comprising positive and negative

- 5 output terminals for supply of a converter output power, voltage or current and connection to a converter load. The resonant DC-DC power converter further comprises a rectification circuit connected between an output of a resonant network and the output side circuit. The resonant network is configured for alternatingly being charged from the input voltage or current and discharged through the rectification
- 10 circuit by a first controllable switch arrangement in accordance with a first switch control signal. A second controllable switch arrangement of the resonant DC-DC power converter is configured to select a first impedance characteristic of the resonant network in a first switch state and select a second impedance characteristic of the resonant network in a second switch state. An output voltage or current control
- 15 circuit is configured to adjust the converter output voltage and/or current by activating and interrupting the first switch control signal in accordance with the switch state of the second controllable switch arrangement.

BACKGROUND OF THE INVENTION

- 20 Power density and component costs are key performance metrics of both isolated and non-isolated DC-DC power converters to provide the smallest possible physical size and/or lowest costs for a given output power requirement or specification. Resonant power converters are particularly useful for high switching frequencies such as frequencies above 1 MHz where switching losses of standard SMPS topologies
- 25 (Buck, Boost etc.) tend to be unacceptable for conversion efficiency reasons. High switching frequencies are generally desirable because of the resulting decrease of the electrical and physical size of circuit components of the power converter like inductors and capacitors. The smaller components allow increase of the power density of the DC-DC power converter. In a resonant power converter an input "chop-
- 30 per" semiconductor switch (often MOSFET or IGBT) of the standard SMPS is replaced with a "resonant" semiconductor switch. The resonant semiconductor switch relies on resonances of a resonant network typically involving various circuit capacitances and inductances to shape the waveform of either the current or the voltage across the semiconductor switch such that, when state switching takes place, there

WO 2015/128397

PCT/EP2015/053977

2

is no current through or no voltage across the semiconductor switch. Hence power dissipation is largely eliminated in at least some of the intrinsic capacitances or inductances of the input semiconductor switch such that a dramatic increase of the switching frequency into the VHF range becomes feasible for example to values

- 5 above 30 MHz. This concept is known in the art under designations like zero voltage and/or zero current switching (ZVS and/or ZCS) operation. Commonly used switched mode power converters operating under ZVS and/or ZCS are often described as class E, class F or class DE inverters or power converters.
- 10 However, it remains a significant challenge to adjust or control the output power/voltage/current of resonant DC-DC power converters in an efficient way. If the resonant power converter is controlled by Pulse Width Modulation (PWM) of the "resonant" semiconductor switch, the ZVS ability is lost and power conversion efficiency will drop significantly. Varying the switching frequency of the resonant power
- 15 converter has also been applied in prior art power converters to control the output voltage/current of the resonant power converter, but this control methodology suffers from a limited range of output voltage regulation and increasing power conversion losses. Controlling the output voltage/current of the resonant power converter by a control scheme which is a combination of variable switching frequency and PWM
- 20 has also been applied in existing resonant power converters and generally proved to work well. This control methodology or scheme unfortunately leads to highly complex control circuitry.

Another more simple yet efficient way of controlling or adjusting the output pow-

- 25 er/voltage/current of resonant DC-DC power converters has been to turn on and off the entire resonant power converter in an intermittent manner. This control scheme is designated "burst mode control" or "on/off control". Burst mode control allows the resonant power converter to operate at a fixed switching frequency where the conversion efficiency is high or optimal during on or activate time periods. During time
- 30 periods where the power converter is off or deactivated, power losses are essentially eliminated because of the lack of switching activity of the resonant transistor which drives the resonant power converter. Ideally burst mode control of resonant power converter leads to full load regulation and constant efficiency from zero to full load on the converter.

WO 2015/128397

PCT/EP2015/053977

3

On/off control of prior art resonant power converters has been achieved by controlling the signal voltage on the control terminal of a "resonant" semiconductor switch, e.g. a MOSFET gate terminal. This scheme may work in a satisfactory manner in some applications, but in order to regulate or adjust the converter output voltage and

- 5 current a feedback control signal from the output/secondary side of the converter to the control terminal of the resonant" semiconductor switch is required. This presents a significant problem in isolated resonant power converters because the feedback control signal must cross a galvanic isolation barrier between the primary side circuitry and the secondary side circuitry. Traditionally, to maintain the galvanic isola-
- 10 tion between input side circuitry and output side circuitry of the resonant power converter, the control signal to the resonant semiconductor switch has been transmitted through a relatively slow and expensive optocoupler or through a bulky and slow transformer. The time delay through the optocoupler or transformer presents, however, a serious obstacle to on/off control of resonant power converters where a fast
- 15 transient response is highly desirable to provide adequate control of the converter output voltage and current. The time delay problem is particularly pronounced for high frequency resonant power converters operating with switching frequencies at or above 20 MHz.
- 20 TSO-SHENG CHAN ET AL: "A Primary Side Control Method for Wireless Energy Transmission System", IEEE Transactions on Circuits and Systems i: regular papers, IEEE, Vol. 59, No. 8 discloses a wireless energy transmission system (WETS) transferring power from a primary side circuit to a secondary side circuit through a skin barrier. The IEEE paper discloses a resonant class E based DC-DC power
- 25 converter with an inductive power transformer connecting the input side circuit and output side circuit through the skin barrier. A charging protection circuit comprise a controllable secondary side switch (Ms) which selectively connects and disconnects a battery (Vb) load from the output of the power converter. A primary side controller operates by detecting variations of the input current and phase of the input reac-
- 30 tance to determine the state of the secondary side switch (Ms). The proposed range of switching frequencies of the class E based DC-DC power converter is between 83-175 kHz.

4

In view of these problems and challenges associated with prior art resonant power converters, it would be advantageous to provide a novel control mechanism for on/off control of resonant power converter eliminating the need to transmit the feedback control signal from an output voltage control circuit across a galvanic isolation

- 5 barrier to the control terminal of the resonant semiconductor switch. The elimination of the feedback control signal would also be advantageous in non-isolated resonant power converters because of the time delay and occupation of board or carrier area associated with wiring of the feedback control signal to the resonant transistor.
- 10 In view of the above, it remains a challenge to reduce the size and lower component costs of both isolated and non-isolated resonant DC-DC power converters. It also remains a challenge to provide an output voltage control mechanism with fast transient response to provide good regulation of the converter output voltage even for high frequency resonant power converters. Hence, a novel control mechanism for
- 15 resonant power converters which simplifies the control of the converter output voltage and reduces the number of electronic components required to perform the output voltage regulation is highly desirable.

SUMMARY OF THE INVENTION

- 20 A first aspect of the invention relates to a resonant DC-DC power converter comprising an input side circuit comprising a positive and a negative input terminal for receipt of an input voltage or current and an output side circuit comprising positive and negative output terminals for supply of a converter output power, voltage or current and connection to a converter load. The resonant DC-DC power converter further
- 25 comprises a rectification circuit connected between an output of a resonant network and the output side circuit. The resonant network is configured for alternatingly being charged from the input voltage or current and discharged through the rectification circuit by a first controllable switch arrangement in accordance with a first switch control signal, wherein a frequency of the switch control signal lies at or above 20
- 30 MHz, more preferably at or above 30 MHz. A second controllable switch arrangement of the resonant DC-DC power converter is configured to select a first impedance characteristic of the resonant network in a first switch state and select a second impedance characteristic of the resonant network in a second switch state. An output voltage or current control circuit is configured to adjust or regulate the con-

verter output voltage and/or current by activating and interrupting the first switch control signal in accordance with a switch state of the second controllable switch arrangement.

- 5 The first and second impedance characteristics of the resonant network may exhibit different resonance frequencies and/or different Q values at the resonance frequency as explained in additional detail below. A switching frequency of the first switch control signal is preferably located approximately at the resonance frequency of the resonant network when the resonant DC-DC power converter is operational or
- 10 turned on to ensure high power conversion efficiency.

The output voltage and/or current control circuit is capable of regulating the converter output voltage or current by controlling the state, i.e. conducting state or nonconducting state, of the second controllable switch arrangement and thereby turn off

- 15 or turn on operation of the present resonant DC-DC power converter. Since the second controllable switch arrangement may conveniently be arranged in the output side circuit of the resonant DC-DC power converter, the output voltage of the converter may be adjusted by a voltage or current control loop arranged entirely on the output side of the converter. Hence, output voltage and/current regulation of isolated
- 20 variants of the resonant DC-DC power converter can be achieved via control of components placed in the secondary side circuit only. The need of feeding an output control signal, such as an output voltage signal or output current signal, back to the input side or primary side of such isolated resonant DC-DC power converters has been eliminated. As mentioned above, this elimination of the need to transmit a
- 25 feedback control signal back to the input side or primary side circuit, in particular to the control terminal of the first switch arrangement, is advantageous in both nonisolated and isolated variants of the present resonant DC-DC power converters. In connection with the isolated resonant DC-DC power converters, the elimination of the feedback control signal to the first switch arrangement of the input side circuit
- 30 removes the need for expensive, bulky and costly isolation devices such as optocouplers or transformers to transmit the feedback control signal across the galvanic isolation barrier.

WO 2015/128397

PCT/EP2015/053977

6

In connection with the non-isolated resonant DC-DC power converters, the ability to carry out the output voltage and/or current regulation in the output side circuit eliminates the time delay and board space occupation associated with the wiring of the feedback control signal to the first switch arrangement of the input side circuit. This

- 5 improves the transient response of the non-isolated resonant DC-DC power converter so as to provide better regulation of the converter output voltage and/or current to the load. This feature is particularly advantageous for resonant DC-DC power converter operating at VHF switching frequencies at or above 30 MHz where short delay times in turning on and turning off the power converter improves regulation
- 10 performance.

The second controllable switch arrangement may be inserted at various locations of the output side circuitry of the resonant DC-DC power converter to select between the first and second impedance characteristics of the resonant network. The second

- 15 controllable switch arrangement is preferably coupled to an output of the resonant network. According to one embodiment, the second controllable switch arrangement is coupled in series between an output of the rectification circuit and the positive or the negative output terminal to connect the converter load in a conducting switch state and disconnect the converter load in a non-conducting switch state. In this
- 20 manner the positive or negative output terminal and converter load may be electrically disconnected from the residual portion of the resonant DC-DC power converter in the off state of the converter.
- The rectification circuit is preferably designed to exhibit substantially resistive impedance at a resonance frequency of the resonant network with the connected converter load. In this situation, the first impedance characteristic, including the resonance frequency, of the resonant network may be determined by one or more interconnected inductors and capacitors of the resonant network with negligible influence from components of the rectification circuit. On the other hand, when the converter
- 30 load is disconnected from the output of the rectification circuit by the second controllable switch arrangement, the rectification circuit may exhibit a different and nonresistive input impedance which loads the resonant network. This non-resistive loading leads to a change of the impedance characteristics of the resonant network so as to select the second impedance characteristic of the resonant network. This sec-

7

ond impedance characteristic of the resonant network may exhibit a lower resonance frequency Q value than the Q value at the resonance frequency of the first impedance characteristic. The second impedance characteristic of the resonant network may possess a lower or higher, i.e. different, resonance frequency than the

- 5 resonance frequency of the first impedance characteristic of the resonant network for example because an inductor and/or a capacitor of the rectification circuit has/have influence on the resonance frequency of the resonant network. These types of differences between the first and second impedance characteristics of the resonant network may be exploited to enable and disable oscillation of a feedback
- 10 loop around the first controllable switch arrangement as described in further detail below. According to another embodiment, the second controllable switch arrangement is coupled across the output of the resonant network to select the first impedance characteristic of the resonant network in a non-conducing state and the second impedance characteristic of the resonant network in in a conducing state.

15

In another embodiment, the impedance characteristics of the resonant network are changed from the first impedance characteristic to the second impedance characteristic of the resonant network by connecting one or more auxiliary capacitances and/or connecting one or more auxiliary inductances to existing capacitances and inductances, respectively, of the resonant network by the second controllable switch

20 inductances, respectively, of the resonant network by the second controllable switcl arrangement.

In a preferred embodiment of the resonant DC-DC power converter, the rectification circuit comprises the second controllable switch arrangement; and a control circuit is
 configured for generating a control signal for the second controllable switch arrangement synchronously to the first switch control signal. In this embodiment, the respective functions of the rectification circuit and second controllable switch arrangement are integrated. In this manner, the second controllable switch arrangement may, in addition to connecting and disconnecting the converter load, operate

30 as a synchronous full-wave or half-wave rectifier and replace functions of one or more ordinary rectifier diodes of the rectification circuit.

The output voltage or current control circuit of the resonant DC-DC power converter may comprise a self-oscillation feedback loop coupled between an output terminal

8

and a control terminal of the first controllable switch arrangement. In this embodiment, the first impedance characteristic of the resonant network is configured to enable oscillation of the self-oscillation feedback loop and the second impedance characteristic of the resonant network is configured to disable oscillation of the self-

- 5 oscillation feedback loop. Hence, the state switching of the second controllable switch arrangement can be used to activate and interrupt the operation of the power converter by enabling and disabling, respectively, the oscillation or switching of the first controllable switch arrangement. The latter may for example comprise a transistor such as a MOSFET where the self-oscillation feedback loop is connected be-
- 10 tween a drain and gate terminal of the MOSFET. The skilled person will understand that the self-oscillation feedback loop preferably is arranged entirely in the input side circuit of the resonant DC-DC power converter to avoid transmission of feedback signals of the self-oscillation feedback loop from the output side circuit.
- 15 The self-oscillation feedback loop may comprise a first intrinsic switch capacitance coupled between the output and control terminals of the first controllable switch arrangement. The self-oscillation feedback loop further comprises a first bias voltage source configured to generate a first adjustable bias voltage and a first inductor, preferably with substantially fixed inductance, coupled in-between the first bias volt-
- 20 age source and the control terminal of the first controllable switch arrangement. The self-oscillation feedback loop may in addition to the first intrinsic switch capacitance include an external capacitor connected between the output and control terminals of the first controllable switch. The skilled person will appreciate that the first intrinsic switch capacitance may comprises a drain-gate capacitance of a MOSFET transis-
- 25 tor contained in the first controllable switch arrangement. Several resonant DC-DC power converter designs based on self-oscillation feedback loops around the input side switch arrangement are disclosed in the applicant's co-pending application PCT/EP2013/072548. The skilled person will understand that these self-oscillation feedback loops may be utilized in the present resonant DC-DC power converters.

30

The first and second impedance characteristics of the resonant network which enable and disable, respectively, the oscillation of the first controllable switch arrangement may differ in various ways. In one embodiment the Q value of the first impedance characteristic is larger than 5 at the resonance frequency of the first impedWO 2015/128397

PCT/EP2015/053977

9

ance characteristic; and the Q factor of the second impedance characteristic is smaller than 2 at the resonance frequency of the second impedance characteristic. In another embodiment the resonance frequency of the first impedance characteristic is at least 1.4 times larger than the resonance frequency of the second imped-

5 ance characteristic.

In a range of advantageous embodiments of the resonant DC-DC power converter, the frequency of the first switch control signal is placed at or above 20 MHz such as above at or above 30 MHz in the so-called VHF range. In these embodiments, the

- 10 above-discussed resonance frequency of the first impedance characteristic of the resonant network is situated at or above 20 MHz, or at or above 30 MHz. The resonance frequency of the first impedance characteristic of the resonant network is preferably situated approximately at the frequency of the first switch control signal. The resonant DC-DC power converters preferably facilitate zero voltage and/or zero
- 15 current switching of the semiconductor switch or switches driving or exciting the resonant network.

The output voltage or current control circuit may comprise a regulation loop, e.g. a voltage, current or power regulation loop, connected between the converter output

- 20 voltage/current and a control terminal of the second controllable switch arrangement to adjust the converter output voltage and/or current in accordance with one or more DC reference voltage(s) or current(s). The voltage or current regulation loop may for example comprise a DC reference voltage generator configured to supply the one or more DC reference voltage(s),
- 25 one or more comparator(s) configured for comparing the converter output voltage to at least a first DC reference voltage and select the conducting state or the nonconducting state of the second controllable switch arrangement depending on a result of the comparison.
- 30 In this embodiment, the voltage regulation loop may use a single DC reference voltage to adjust the converter output voltage. The resonant DC-DC power converter may be turned off when the converter output voltage exceeds the single DC reference voltage by selecting an appropriate state of the second controllable switch arrangement. Likewise, the DC-DC power converter may be turned on when the con-

verter output voltage is smaller than the single DC reference voltage by selecting the opposite state of the second controllable switch arrangement.

In an alternative embodiment of the resonant DC-DC power converter the voltage or

- 5 current regulation loop comprises at least two different DC reference voltages or currents which are utilized as references for controlling output regulation. In this embodiment, the DC reference voltage generator is configured to supply at first DC reference voltage and a second DC reference voltage where the first DC reference voltage is higher than the second DC reference voltage; The one or more compara-
- 10 tor(s) is configured to: comparing the converter output voltage to the first and second DC reference voltages; and selecting one of the conducting and non-conducting states of the second controllable switch arrangement in response to the converter output voltage exceeding the
- 15 first DC reference voltage, selecting the opposite state of the second controllable switch arrangement in response to the converter output voltage falls below the second DC reference voltage.

As previously mentioned, the resonant DC-DC power converter may comprise a

- 20 galvanic isolation barrier. The galvanic isolation barrier is preferably arranged between the input side circuit and the output side circuit to provide galvanic isolation between the converter output voltage and the input side circuit. This galvanic isolation barrier may comprise a pair of magnetically coupled inductors comprising a first inductor electrically connected to the input side circuit and a second inductor electri-
- 25 cally connected to the input of the rectification circuit. The pair of magnetically coupled inductors may comprise a transformer. In another embodiment, the galvanic isolation barrier comprises first and second coupling capacitors. The first coupling capacitor may be arranged in a signal carrying line of the resonant power converter for example in series between the output of the resonant network and the rectifica-
- 30 tion circuit. The second coupling capacitor may be arranged in a negative voltage wire or ground wire of the resonant power converter between the input and output side circuits. The coupling capacitor based galvanic isolation barrier is particularly useful in resonant DC-DC power converters operating at or above 20 MHz because of the relatively small capacitances required of the first and second capacitors at

WO 2015/128397

PCT/EP2015/053977

11

such high frequencies. The small capacitance allows each of the first and second capacitors to be physically small and be implemented as non-electrolytic SMD capacitors for example ceramic capacitors with a capacitance below 100 nF. The skilled person will understand that all resonant DC-DC power converter topologies may be used in the present invention for example SEPIC, class E, class F, class DE or converter topologies derived from these. Several exemplary isolated and non-isolated resonant DC-DC power converters of Class E topology are described in detail below in connection with the appended drawings.

10

5

The first controllable switch arrangement may comprise one or more semiconductor switches and the second controllable switch arrangement may comprise one or more semiconductor switches. Each of the semiconductor switches of the first and second controllable switch arrangements may comprise a semiconductor transistor such as a MOSFET or IGBT such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET. The control terminals or terminals of the first controllable switch arrangement may accordingly comprise gate terminal(s) or base terminal(s) of the one or more semiconductor switches. The control terminal of each of the semiconductor switches may be driven by the first switch control signal to alternatingly force the semiconductor switches between on-states and off-states. The control terminals or terminals of the semiconductor switches.

One embodiment of the resonant DC-DC power converter comprises a wireless data receiver for receipt of remote data commands to control the switch state of the second controllable switch arrangement. The remote data command may be used to interrupt or activate operation of the resonant DC-DC power converter by commands received via a wireless home automation network. The remote data command may be inputted to the output voltage or current control circuit that controls the

state switching of the second controllable switch arrangement. In this manner, the

30 resonant DC-DC power converter can be switched ON or OFF or regulated via remote control. The wireless data receiver may be compliant with various industry standard wireless data communication protocols such as the ZigBee communication protocols or wired data communication protocols such as the Digital Addressable Lighting Interface (DALI) and protocol. WO 2015/128397

PCT/EP2015/053977

12

The skilled person will appreciate that the first controllable switch arrangement may be formed by a single transistor, for example a NMOS device, or several interconnected transistors depending on the selected topology of the resonant DC-DC power converter. In some embodiments, the first controllable switch arrangement may

comprise a half-bridge switch topology or a full-bridge switch topology. The galvanic isolation barrier may comprise a transformer which comprises a pair of magnetically coupled inductors comprising a first inductor electrically connected to the primary side circuit and a second inductor electrically connected to the output

10 side circuit. The first and second inductors could be discrete windings both wound around a common magnetic permeable structure to form an isolation transformer. In an alternative embodiment, the first and second inductors are integrated in a printed circuit board without intervening magnetic material. The printed circuit board could have the entire DC-DC power converter mounted thereon.

15

5

One embodiment of the resonant DC-DC power converter is arranged on a single substantially flat carrier substrate to form a compact, low cost and single unit power converter assembly that is well-suited for integration in various kind of consumer equipment such as LED lamps and battery chargers. The latter embodiment of the

- 20 resonant DC-DC power converter comprises a single substantially flat carrier substrate comprising a first surface and a second, opposing, surface, wherein the input side circuit, the output side circuit, the rectification circuit, the resonant network, the first controllable switch arrangement, the second controllable switch arrangement and the output voltage or current control circuit are arranged on the first surface
- 25 and/or the second surface. The flat carrier substrate may comprise a single sided or double-sided printed circuit board, which may comprise additional layers between a top layer comprising the first surface and a bottom layer comprising the second surface. The skilled person will appreciate that respective passive and active electronic components of the input side circuit, the output side circuit, the rectification
- 30 circuit, the resonant network, the first controllable switch arrangement, the second controllable switch arrangement and the output voltage or current control circuit may be attached to the upper and/lower carrier surfaces by soldering or gluing.

A second aspect of the invention relates to a method of operating a resonant DC-DC power converter to generate a converter output voltage or current, said method comprising steps of:

a) alternatingly charging and discharging a resonant network of the resonant con-

5 verter from an input voltage or current source by a first controllable switch arrangement in accordance with a first switch control signal,

b) rectifying a resonant current of the resonant network,

c) discharging a rectified resonant current to a rectification circuit to produce a rectified output voltage,

10 d) switching between a first impedance characteristic or a second impedance characteristic of the resonant network by selecting a first or a second switch state of a second controllable switch arrangement,

e) adjusting a converter output voltage or current by alternatingly activating and interrupting the first switch control signal in accordance with first and second switch

15 states of the second controllable switch arrangement.

BRIEF DESCRIPTION OF THE DRAWINGS

Preferred embodiments of the invention will be described in more detail in connection with the appended drawings, in which:

FIG. 1 is a top-level schematic block diagram of resonant DC-DC power converters in accordance with preferred embodiments of the invention,
 FIG. 1A) is a schematic block diagram of a power converter assembly comprising

resonant DC-DC power converters in accordance with preferred embodiments of the invention,

- FIG. 2 is a simplified electrical circuit diagram of an isolated class E type of DC-DC power converter in accordance with a first embodiment of the invention,
 FIG. 3 is a simplified electrical circuit diagram of isolated class E type of DC-DC power converter in accordance with a second embodiment of the invention,
 FIG. 4 is a simplified electrical circuit diagram of a self-oscillating isolated class E
- 30 DC-DC power converter in accordance with a third embodiment of the invention; and FIG. 5 is a simplified electrical circuit diagram of a non-isolated class E DC-DC power converter in accordance with a fourth embodiment of the invention.

WO 2015/128397

14

DETAILED DESCRIPTION OF PREFERRED EMBODIMENTS

FIG. 1 shows a simplified schematic block diagram of a resonant DC-DC power converter 100 in accordance with preferred embodiments of the present invention. Detailed schematic circuit diagrams of exemplary embodiments of the present reso-

- 5 nant DC-DC power converters are described below with reference to FIGS. 2, 3, 4 and 5. The resonant DC-DC power converter 100 comprises an input block 120 and an output block 150 connected through an optional galvanic isolation barrier 140. The galvanic isolation barrier 140 may comprise various types of electrical insulation elements or components such as a pair coupling capacitors or a pair of magnetically
- 10 coupled inductors such as a transformer. The input block 120 comprises an input side circuit which comprises a positive input terminal 101 and a negative input terminal 102 for receipt of a DC or AC input voltage V_{in} from a voltage or power source. The input side circuit may comprise an input capacitor (not shown) electrically connected between the positive and negative input terminals 101, 102 to form an ener-
- 15 gy reservoir for the input voltage source and suppress ac ripple voltage components and/or noise components of the DC or AC input voltage V_{in}. The output block 150 comprises an output side circuit comprising positive and negative output terminals 110, 111, respectively, for supplying a converter DC output voltage V_{out} and connection to a converter load. The converter load is schematically represented by load
- 20 resistor R_L. The output block 150 further comprise a rectification circuit, schematically represented by the diode symbol, connected between an output of a resonant network (not shown) of the resonant DC-DC power converter 100 and the output side circuit. The input block 120 comprises a first controllable switch arrangement, comprising at least one controllable semiconductor switch, operable in accordance
- 25 with a switch control signal. The controllable semiconductor switch or switches may comprise a transistor such as a BJT, MOSFET or IGBT where the base or gate terminal(s) are coupled to the switch control signal. The controllable switch arrangement is coupled to the resonant network such that the latter is alternatingly charged from the DC or AC input voltage and discharged the rectification circuit to produce
- 30 the DC output voltage V_{out}. A frequency of the switch control signal of the first controllable switch arrangement accordingly sets the switching frequency of the resonant DC-DC power converter 100. The resonant network may be exclusively arranged in the input block 120 in some embodiments of the invention while the resonant network in other embodiments may include certain passive components of the
PCT/EP2015/053977

15

galvanic isolation barrier 140 and/or rectification circuit. The resonant network preferably comprises at least one capacitor and at least one inductance connected to each other. The resonant DC-DC power converter 100 further comprises an output voltage or current control circuit 160 which is configured to adjust the DC output

- 5 voltage V_{out} by activating and interrupting the previously discussed switch control signal of the first controllable switch arrangement in accordance with a switch state of a second controllable switch arrangement (not shown). The switch state of a second controllable switch arrangement is controlled via control line or wire 112 which may be connected to a suitable control terminal of the second controllable switch
- 10 arrangement as discussed in additional detail below. When the switch control signal of the first controllable switch arrangement is interrupted or stopped, i.e. not switching, the resonant network is no longer charged or excited and the DC output voltage V_{out} gradually decreases due to the current drawn by the converter load. When the switch control signal of the first controllable switch arrangement is activated the res-
- 15 onant network is again charged or excited by the first controllable switch arrangement and supplies resonant current to the rectification circuit and output side circuit such that the DC output voltage V_{out} gradually increases.

Different types of control mechanisms are utilized in different embodiments of the
 present resonant DC-DC power converter to control the activation and interruption of
 the switch control signal of the first controllable switch arrangement. In one embod iment, the second controllable switch arrangement is configured to connect and dis connect the converter load R_L such that impedance characteristics of the resonant
 network are altered. In another embodiment, the impedance characteristics of the

- 25 resonant network are changed by switching the second controllable switch arrangement between a conducting state and a non-conducting state for example adding one or more auxiliary capacitances or inductances to the resonant network. This change of the impedance characteristics of the resonant network may in certain embodiments be detected by a resonant phase detector 130. The resonant phase de-
- 30 tector 130 may be configured to monitor, via monitor line of signal 131, a resonant voltage and a resonant current of the resonant network. The resonant phase detector 130 may configured to measure a change of phase or amplitude between the resonant current and resonant voltage at a nominal resonance frequency of the network caused by the change of the impedance characteristics of the resonant

16

network caused by the state switching of the second controllable switch arrangement.

The resonant phase detector 130 may generate the switch control signal for the first switch arrangement and supply this control signal 133 to the control terminal or terminals of the resonant network such as a base or gate terminal as discussed above. In one advantageous embodiment of the present resonant DC-DC power converter 100, the resonant phase detector 130 is integrated into a self-oscillation feedback loop formed around the first controllable switch arrangement. In this manner, oscilla-

- 10 tion of the self-oscillation feedback loop is either enabled or disabled according to the switch state of the second controllable switch arrangement as discussed in additional detail below with reference to FIGS. 4 & 5.
- FIG. 1A) illustrates schematically how the resonant DC-DC power converters 100 in
 accordance with preferred embodiments of the present invention may be arranged on a single substantially flat carrier substrate 180 such as a printed circuit board (PCB). The resonant DC-DC power converter mounted on such a single carrier substrate 180 forms a compact, low cost and single unit power converter assembly that is well-suited for integration in various kind of consumer equipment such as LED
- 20 lamps and battery chargers. Furthermore, the mounting of the input block 120, the galvanic isolation barrier 140 and the output block 150 on a common carrier substrate allows high power conversion efficiency because the input and output side circuits of the resonant DC-DC power converter are placed proximate to each other. The single substantially flat carrier substrate 180 may comprise an upper or first
- 25 surface 182 and a second, opposing, surface (not shown). The skilled person will appreciate that passive and active electronic components of the resonant DC-DC power converter embodiment in question may be attached to the upper and/lower carrier surfaces for example by soldering or gluing. These passive and active electronic components may comprise the respective semiconductor switches, capaci-
- 30 tors, inductors, terminals, comparators etc. of the resonant DC-DC power converter embodiments 200, 300, 400 and 500 discussed in detail below and illustrated on FIGS. 2, 3, 4 and 5. Likewise, suitable electrical wire traces may be formed on the upper and/lower carrier surfaces of the single flat carrier substrate 180 to interconnect the circuit blocks 120, 130, 140, 150, 160 in a desired manner. The upper or

PCT/EP2015/053977

17

under surface 182 of the substantially flat carrier substrate 180 also comprises the positive input terminal 101 and the negative input terminal 102 for receipt of the DC or AC input voltage V_{in} from the voltage or power source of the converter. The upper or under surface 182 of the substantially flat carrier substrate 180 comprises the

5 positive and negative output terminals 110, 111, respectively, for supplying the DC output voltage V_{out}, together with the accompanying output power and current, generated by converter to the intended load for example LED devices of a LED lamp.

The resonant DC-DC power converters 100 in accordance with preferred embodi-

- 10 ments of the present invention may comprise an optional wireless or wired data receiver 190 for receipt of remote data commands to the output voltage or current control circuit 160. The remote data command may be used to interrupt, activate or adjust operation of the resonant DC-DC power converter via the output voltage or current control circuit 160. The remote data command may be subjected to a logical OR
- 15 operation with the switch control signal otherwise provided by the output voltage or current control circuit 160 via signal wire or line 112. In this manner, the resonant DC-DC power converter can be switched ON or OFF or regulated via remote control and be integrated with a wireless or wired home automation network. The optional data receiver 190 may be compliant with various industry standard wireless data
- 20 communication protocols such as the ZigBee communication protocols or wired data communication protocols such as the Digital Addressable Lighting Interface (DALI) interface and protocol.
- FIG. 2 shows an electrical circuit diagram of an isolated class E DC-DC power converter 200. The power converter 200 comprises an input block 220 and an output block 250 connected through an optional galvanic isolation barrier 240 formed by series or coupling capacitors C2 and C3. The input side circuit 220 comprises a positive input terminal 202 and a negative input terminal 201 for receipt of a DC or AC input voltage V_{in} from a voltage or power source. An input capacitor (not shown)
- 30 may in addition be electrically connected between the positive and negative input terminals 201, 202 to form an energy reservoir for the input voltage source and suppress ac ripple voltage components and/or noise components of the DC or AC input voltage V_{in}. The input side circuit 220 additionally comprises a resonant network as discussed below which is alternatingly charged from the DC or AC input voltage V_{in}

PCT/EP2015/053977

18

and discharged through a class E rectification circuit 250 by a first controllable switch arrangement S1. The latter comprises a single controllable semiconductor transistor or switch S1 in the present embodiment. The skilled person will understand that the first controllable switch arrangement S1 may comprise a plurality of

- 5 controllable semiconductor transistors or switches in other embodiments of the invention. The first controllable switch arrangement S1 may for example comprise a half-bridge arrangement with a pair of stacked semiconductor transistors or switches in a corresponding class DE embodiment of the resonant DC-DC power converter. The single controllable semiconductor switch S1 may comprise a transistor such as
- 10 a MOSFET or IGBT, for example a Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET.

The resonant network comprises at least a first inductor L1, a first capacitor C1, which may be a parasitic capacitance of part of S1, and second inductor L2 and the

- 15 coupling capacitors C2 and C3 of the galvanic isolation barrier 240. The capacitance of the coupling capacitor C3 may be much larger than the capacitance of C2, e.g. more than 10 times larger, such that the effect of the capacitance of C3 may be neglected in setting of the resonance frequency of the resonant network. The charging and discharging, or excitation, of the resonant network follows the first switch control
- 20 signal applied to a control or gate terminal 204 of semiconductor switch S1 such that the switch S1 alternates between an conducting/on-state and a non-conducting/offstate at the frequency of the first switch control signal when the resonant power converter 200 is active or ON. The power converter 200 is preferably designed or configured such that a frequency of the first switch control signal is situated in close
- 25 proximity to the resonance frequency of the resonant network when an on-state or conducting state of a second controllable semiconductor switch S2 is selected as explained in detail below. Hence, the switching frequency of the resonant power converter 200 corresponds to the frequency of the first switch control signal when the converter is ON or active. The state switching of semiconductor switch S1 gen-
- 30 erates a resonant current in the resonant network flowing from an output of the resonant network through a class E rectification circuit of the output block 250 to produce a rectified DC output voltage V_{out} across a rectification capacitance. The class E rectification circuit comprises inductor L3, rectifier diode D1 and capacitor C4. The skilled person will understand that the inductor L3 and capacitor C4 may influence

PCT/EP2015/053977

19

the setting of resonance frequency of the resonant network depending on how the converter is dimensioned, However, the influence of the rectification components L3 and C4 may be minimal in power converter embodiments where the rectification circuit is configured to exhibit an essentially resistive input impedance at the reso-

5 nance frequency of the resonant network when the switch S2 is in its conducting state or on-state. The capacitors C5 and C6 ensure a stable converter output voltage or current.

The output block furthermore comprises a positive and a negative output terminal

- 10 210, 211, respectively, which supply the converter DC output voltage V_{out} to a converter load R_L of the class E power converter 200. The converter load is schematically illustrated by load resistor R_L on the drawing, but may in practice include different types of electric loads for example a set of LED diodes or a rechargeable battery etc. The second controllable semiconductor switch S2 is placed in-between a posi-
- 15 tive output node 209 of the rectification circuit and the converter load R_L. Hence, S2 is placed in series with the converter load such that the latter is disconnected from the class E power converter 200 when switch S2 is switched to its off-state or non-conducting state. In this off-state of S2, the capacitor C6 is electrically isolated from the converter rectification circuit, but may supply power to the converter load mo-
- 20 mentarily while the DC output voltage of the power converter declines due to current drawn by the converter load R_L . In the opposite situation, where S2 is placed in its on-state or conducting state, the positive output node 209 of the rectification circuit is connected to the converter load R_L and the rectification capacitors C5 and C6 are placed in parallel so as to form a joint output capacitor of the power converter 200.
- 25 The skilled person will understand that the second controllable semiconductor switch S2 preferably is designed or selected such that its on-resistance is markedly smaller than an equivalent load resistance, e.g. at least 10 times smaller, to minimize the on-state power loss in S2 and a voltage drop across S2.
- 30 The second controllable semiconductor switch S2 preferably comprises at least one MOSFET transistor such as a NMOS transistor. The second controllable semiconductor switch S2 may of course be formed by a plurality of parallelly connected individual semiconductor switches such as a plurality of parallelly connected MOSFETs.

PCT/EP2015/053977

20

The skilled person will understand that the arrangement of the second controllable semiconductor switch S2 operates to connect the converter load R_L to the output of the rectifier in the conducting state of switch S2 and disconnect the converter load R_L from the output node 209 of the rectifier in the non-conducting state of the switch

- 5 S2. This in effect changes the loading at the output of the resonant network presented by the rectification circuit because the equivalent input impedance of the rectification circuit increases markedly when the converter load R_L is disconnected by the switch S2. This change of loading on the output of the resonant network caused by the state switching of switch S2 changes the impedance response char-
- 10 acteristics of the resonant network such that the latter exhibits a second and different impedance response characteristic in the non-conducting state of switch S2 (where the converter load is disconnected). This change of the impedance characteristics over frequency of the resonant network may for example comprise a change of Q value and/or a change of the resonance frequency. The impedance
- 15 response characteristics of the resonant network may be represented by its impedance characteristics as seen from the output of the switch transistor S1, i.e. the drain terminal of S1 in the present embodiment. The Q of the impedance characteristics of the resonant network may for example change from a value between 5 and 20 in the conducting state of switch S2 down to a value between 0.5 and 2 in the
- 20 non-conducting state of switch S2. The resonance frequency of the resonant network may for example decrease with a factor somewhere between 1.4 and 3 such as about 1.41 from the conducting state to the non-conducting state of switch S2 for example from about 30 MHz to about 21 MHz. The class E power converter 200 further comprises an output voltage control circuit 260 which is configured to adjust
- 25 the DC output voltage V_{out} by enabling/activating or disabling/interrupting the first switch control signal on the gate terminal 204 of the first switch S1 in accordance with the selection of the first impedance characteristic or the second impedance characteristic of the resonant network. The output voltage control circuit comprises a comparator 208 and DC reference voltage generator (not shown) supplying a DC
- 30 reference voltage V_{ref} to a first input of the comparator 208. A comparator output is connected to a gate terminal 212 of the switch S2 to such that the comparator output selects one of the conducting state and non-conducting state of the switch S2 depending on the logic level of the comparator output. The control mechanism of the output voltage control circuit, which enables or disables the gate control signal on

PCT/EP2015/053977

21

the switch S1, may respond to the selected impedance characteristics of the resonant network, as controlled by the state of switch S2, via different control and detection mechanisms. In one embodiment, a resonant phase detector 230 is configured to monitor, via monitor line of signal 231, a resonant voltage and a resonant current

- 5 of the resonant network as described above in connection with the resonant phase detector 130. The resonant phase detector 230 produces in response supply a control signal 233 to the gate terminal 204 of the first switch S1 to selectively activate or interrupt the state switching of S1.
- 10 Another embodiment of the control mechanism of the output voltage control circuit comprises a self-oscillation feedback loop coupled between a drain, i.e. output terminal, and the gate terminal 204 of the switch S1. The first impedance characteristic of the resonant network is configured to enable oscillation of the self-oscillation feedback loop by design of a suitable loop gain. Hence, the gate signal of switch S1
- 15 will be active or enabled and switching at the oscillation frequency defined by the magnitude and phase characteristic of the loop of gain of the self-oscillation feed-back loop. This oscillation frequency is the switching frequency of the class E power converter 200 during active or ON operation and will typically be situated close to an impedance maximum of the resonant network where the loop gain has the largest
- 20 magnitude. The second impedance characteristic of the resonant network is conversely configured to disable or interrupt the oscillation of the self-oscillation feedback loop by changing the loop gain of the self-oscillation feedback loop in a suitable manner. When the oscillation of the self-oscillation feedback loop is interrupted, the gate control signal on the gate 204 of switch S1 is disabled or interrupted for
- 25 example clamped to a constant voltage level below a threshold voltage of S1 if the latter is a MOSFET. As explained above the output voltage control circuit activates or interrupts/deactivates the class E converter 200 by controlling the state of the second switch S2 which switch state in turn enables or disables the self-oscillation of the feedback loop around switch S1. The latter determines the switching frequen-
- 30 cy of the class E power converter 200. Consequently, the on-state or conducting state of switch S2 where the converter load is connected enables normal or ON operation of the class E power converter 200. The class E power converter 200 is furthermore switched OFF or to a non-operational state by selecting the off-state of the second switch S2 where the converter load is disconnected via a suitable gate sig-

15

PCT/EP2015/053977

22

nal on the gate terminal 212 of S2. In this manner, the output voltage control circuit 260 provides on/off control of the class E power converter 200 to adjust the DC output voltage via state switching of switch S2 in a highly efficient and convenient manner. In particular, the on/off control is carried out by the changing the state of the

- 5 switch S2, placed on a secondary or output side of the class E power converter 200, from a control signal derived from the secondary side circuit. Therefore, the adjustment of the DC output voltage is achieved without transmitting any control signal across the galvanic isolation barrier, formed by series capacitors C2 and C3 in the present embodiment, to the switch S1 on the input side circuit as explained in further
- 10 detail below with reference to the self-oscillation based DC-DC power converter embodiments depicted on FIG. 4 and FIG. 5.

The frequency of the switch control signal on the gate 204 of switch S1 is preferably at or above 20 MHz or even above 30 MHz to provide the so-called VHF type of resonant DC-DC power converter 200. The switch control signal may comprise a PWM modulated control signal. The rectification circuit may comprise a diode based

- rectifier or a synchronous rectifier in front of the rectification capacitor to produce the converter output voltage V_{out.}as a DC output voltage. The class E power converter 200 may comprise a capacitor C1 connected or arranged across drain and source
- 20 terminals of the switch S1 to increase a resonant current and/or adjust/fine-tune a resonance frequency of the class E converter 200. Likewise, a yet further capacitor C4 may be arranged across the rectifying diode D1 to adjust a duty cycle of the resonant power converter 200.
- FIG. 3 is a simplified electrical circuit diagram of an isolated resonant class E type of DC-DC power converter 300 in accordance with a second embodiment of the invention. The main difference between the present class E converter 300 and the previous class E embodiment 200 lies in the integration of the functions of the separate rectifying diode D1 and the second controllable semiconductor switch S2 of the
- 30 class E converter 200 in a single component S2 of the present class E power converter 300. The skilled person will appreciate that the above discussed features, functions and components of the first embodiment of the class E power converter 200 may apply to the present embodiment of the class E power converter 300 as well. Likewise, corresponding features and components of the first and second em-

PCT/EP2015/053977

23

bodiments of the class E power converters 200, 300 have been provided with corresponding reference numerals to ease comparison.

The skilled person will understand that the arrangement and control of the second

- 5 controllable semiconductor switch S2 in the class E power converter 300 serve two different functions. The first function is similar to the functionality of the switch S2 of the class E converter 200 discussed above, i.e. to connect the converter load R_L to the output of the rectifier in the conducting state of switch S2 and disconnect the converter load R_L from the output of the rectifier in the non-conducting state of the
- 10 switch S2. The loading on the output of the resonant network applied by the rectification circuit is altered between the conducting and non-conducting states of switch S2 as discussed above because the equivalent input impedance of the rectification circuit increases markedly when the converter load R_L is disconnected by switch S2. This change of loading on the output of the resonant network caused by the state
- 15 switching of switch S2 changes the impedance characteristics of the resonant network such that it exhibits a second and different impedance characteristic in the non-conducting state of switch S2 (where the converter load is disconnected). This state switching is controlled by a voltage control loop comprising a comparator 308 and gate enable or drive circuit 305. An output of the gate drive circuit 305 is con-
- 20 nected to a gate terminal 312 of the switch S2 and operable to determine the state of switch S2, i.e. conducting or non-conducting. A control circuit (not shown) is connected to the gate enable circuit 305 such that when switch S2 is in a conducting state during normal or ON operation of the class E power converter 300, the gate control signal on gate 312 is switched synchronously to the gate control signal of
- 25 switch S1 on the input block 320. In this manner, the switch S2 operates as a synchronous half-wave rectifier and replaces the operation of rectifier diode D1 of the class E power converter 200. The rectification action of switch S2 generates the DC output voltage V_{out} of the converter across a rectification capacitor C5 coupled across the negative and positive output terminals, 310, 311 of the class E power
- 30 converter 300. The class E power converter 300 may comprise a capacitor C1 as illustrated connected or arranged across drain and source terminals of the switch S1 to increase a resonant current and/or adjust/fine-tune a resonance frequency of the class E power converter 300. Likewise, a yet further capacitor C4 may be arranged across the integrated rectifying switch and converter load switch S2 to adjust a duty

24

cycle of the power converter 300. Also a self-oscillating gate drive may be used to drive the controllable switch S2 for synchronous rectification, in this way communication across the isolation barrier can be avoided.

- 5 FIG. 4 is a simplified electrical circuit diagram of a self-oscillating isolated class E DC-DC power converter 400 in accordance with a third embodiment of the invention. The main difference between the present class E power converter 400 and the previous class E converter 200 discussed in connection with FIG. 2 lies in the arrangement of a self-oscillation feedback loop coupled between an output or drain terminal
- 10 of a first controllable semiconductor switch S1 and a control terminal 404 of switch S1. The self-oscillation feedback loop comprises a feedback capacitor C7 coupled between the drain and gate terminals of switch S1 and a series inductor L4 placed in-between the gate terminal 404 and a bias voltage source V_{bias}. The self-oscillation of the controllable semiconductor switch or transistor S1 is achieved by an appropri-
- 15 ate phase shift induced by the combination of the feedback capacitor C7 and the gate inductor L4 in combination with an appropriate voltage gain provided by the first impedance characteristic of the resonant network. Hence, when the second control-lable semiconductor switch S2 is placed in a conducting state, i.e. with the converter load R_L connected to the output of the converter, the self-oscillation of the switch or
- 20 transistor S1 is enabled. When the transistor S1 is self-oscillating the power converter 400 is ON or operational to supply a DC output voltage and current to the converter load R_L. In the opposite state, i.e. off-state, of switch S2, the resonant network, comprising at least L1, L2, C1 and C2, exhibits a markedly different, impedance characteristic from the first impedance characteristic due to the disconnec-
- 25 tion of the converter load R_L from the output of the rectification circuit. The second impedance characteristic of the resonant network is configured to disable oscillation of the self-oscillation feedback loop around transistor switch S1. This may be achieved by designing the second impedance characteristic of the resonant network with a reduced Q factor and/or a changed resonance frequency compared to the
- 30 first impedance characteristic. The Q factor at the resonance frequency of the first impedance characteristic may for example be larger than 5 or 10 while the Q factor at the resonance frequency of the second impedance characteristic may be smaller than 2 or 1. In addition, the resonance frequency of the first impedance characteristic may be arranged at a desired/target switching frequency of the DC-DC power

25

PCT/EP2015/053977

25

converter 400 for example at or above 20 MHz or at or above 30 MHz while the resonance frequency of the second impedance characteristic may example at least 1.4 times lower than the resonance frequency of the first impedance characteristic. The skilled person will understand that the functionality of the previously discussed reso-

5 nant phase detectors 130, 230, 330 is integrated into the self-oscillation feedback loop formed around the transistor S1. In this manner, oscillation of the selfoscillation feedback loop is either enabled or disabled according to the switch state of the second controllable switch S2.

The skilled person will understand that the each of the illustrated capacitors C7, C8

- 10 and C1 in practice may represent intrinsic device capacitances only of the transistor switch S1 for example drain-gate, gate-source and drain-source capacitances of a MOSFET transistor embodiment of switch S1. In the alternative, one or more of these intrinsic device capacitances may be supplemented by a parallelly connected external capacitor to provide a desired capacitance. The skilled person will appreci-
- 15 ate that the above discussed features, functions and components of the output voltage control loop, rectification circuit and galvanic isolation barrier of first embodiment of the class E power converter 200 may apply to the corresponding components and circuits of present embodiment of the power converter 400 as well. Hence, corresponding components of the first and third embodiments of the class E
- 20 power converters 200, 400 have been provided with corresponding reference numerals to ease comparison.

FIG. 5 is a simplified electrical circuit diagram of a non-isolated resonant class E DC-DC power converter 500 in accordance with a fourth embodiment of the invention. The main difference between the present class E converter 500 and the previ-

- ous class E converter 200, discussed in connection with FIG. 2, lies in the lack of a galvanic isolation barrier between the input block 520 and the output block 550 and the different coupling of the second controllable semiconductor switch S2. The converter output voltage V_{out} is regulated by alternatingly activating and interrupting the
- 30 first switch control signal on the gate terminal 512 of semiconductor switch S2 to turn ON and disable the resonant power converter 500 as discussed in connection with the description of the previous resonant power converter embodiments. The regulation of the DC output voltage V_{out} of the power converter 500 is carried out by an output voltage control circuit 560 comprising a conditioning circuit 515 coupled to

PCT/EP2015/053977

26

the converter output voltage V_{out} and comparator 508. A reference input of the comparator 508 is coupled to a DC reference voltage V_{ref} while another comparator input receives the DC output voltage after conditioning. As explained above, the comparator output signal provides a control signal to the gate terminal of the second control-

- 5 lable switch S2 to alternatingly switch the latter between its conducting and nonconducting states depending on the level of the DC output voltage relative to the DC reference voltage V_{ref}. The present DC-DC power converter 500 comprises a selfoscillation feedback loop, schematically illustrated by circuit box 530, to turn off and turn on the power converter 500 depending on the selected state of switch S2. The
- 10 self-oscillation feedback loop may be connected between an output or drain terminal of a first controllable semiconductor switch S1 and a control terminal 504 of switch S1 as discussed above in connection with the third embodiment.
- While the semiconductor switch S2 of each of the previously discussed power converter embodiments 200, 300, 400 is coupled in series between the resonant network output and the converter load, the corresponding switch S2 of the present power converter 500 is coupled across an output of the resonant network and a negative supply rail or ground rail 511 of the power converter 500. The resonant network of the power converter 500 comprises at least L1, L2 and C1. Hence, the
- 20 switch S2 functions as an open circuit when it is placed in non-conducting state or off- state by the appropriate comparator output signal on the gate terminal 512 of S2. Hence, the switch S2 presents substantially no loading of the output of the resonant network in its non-conducting state so as to select a first impedance characteristic of the resonant network. In the conducting state of switch S2 it effectively short
- 25 circuits the output of the resonant network to ground or the negative supply voltage line 511 via a low impedance path since the on-resistance of switch S2 may be significantly smaller than an impedance of the resonant network. Hence, the resonant network exhibits a second and preferably markedly different impedance characteristic in the conducting state of switch S2. In a similar manner as discussed above in
- 30 connection with the self-oscillation network of the third embodiment 400, the second impedance characteristic of the resonant network is configured to disable oscillation of the self-oscillation feedback loop around transistor switch S1. This may be achieved by configuring the resonant network with a second impedance characteristic which exhibits a reduced Q factor and/or a changed resonance frequency com-

pared to the same features of the first impedance characteristic. The Q factor at the resonance frequency of the first impedance characteristic may for example be larger than 5 or 10 while the Q factor at the resonance frequency of the second impedance characteristic may be smaller than 2 or 1. In addition, the resonance frequency of

5 the first impedance characteristic may lie at or above 20 MHz such as at or above 30 MHz while the resonance frequency of the second impedance characteristic may be lower for example at least 1.4 times lower.

<u>CLAIMS</u>

1. A resonant DC-DC power converter comprising:

an input side circuit comprising a positive and a negative input terminal for receipt of

- 5 an input voltage or current, an output side circuit comprising positive and negative output terminals for supply of a converter output power, voltage or current and connection to a converter load, a rectification circuit connected between an output of a resonant network and the output side circuit,
- 10 wherein the resonant network is configured for alternatingly being charged from the input voltage or current and discharged through the rectification circuit by a first controllable switch arrangement in accordance with a first switch control signal, wherein a frequency of the switch control signal is at or above 20 MHz, more preferably at or above 30 MHz,
- 15 a second controllable switch arrangement configured to select a first impedance characteristic of the resonant network in a first switch state and select a second impedance characteristic of the resonant network in a second switch state, an output voltage or current control circuit configured to adjust the converter output voltage and/or current by activating and interrupting the first switch control signal in
- 20 accordance with a switch state of the second controllable switch arrangement.

2. A resonant DC-DC power converter according to claim 1, wherein the second controllable switch arrangement is coupled in series between an output of the rectification circuit and the positive or the negative output terminal to connect the convert-

er load in a conducting switch state and disconnect the converter load in a nonconducting switch state.

3. A resonant DC-DC power converter according to claim 1, wherein the second controllable switch arrangement is coupled across the output of the resonant net30 work to select the first impedance characteristic of the resonant network in a non-conducing state and the second impedance characteristic of the resonant network in in a conducing state.

4. A resonant DC-DC power converter according to claim 2 or 3, wherein the rectifi-

cation circuit comprises the second controllable switch arrangement; and a control circuit configured for generating a control signal for the second controllable switch arrangement synchronously to the first switch control signal.

- 5 5. A resonant DC-DC power converter according to any of claims 1-4, wherein the output voltage or current control circuit comprises:
 a self-oscillation feedback loop coupled between an output terminal and a control terminal of the first controllable switch arrangement; wherein the first impedance characteristic of the resonant network is configured to enable oscillation of the self-
- 10 oscillation feedback loop; and the second impedance characteristic of the resonant network is configured to disable oscillation of the self-oscillation feedback loop

6. A resonant DC-DC power converter according to any of the preceding claims,
wherein a Q factor of the first impedance characteristic is larger than 5 at a resonance frequency of the first impedance characteristic; and
a Q factor of the second impedance characteristic is smaller than 2 at a resonance frequency of the second impedance characteristic

20 7. A resonant DC-DC power converter according to any of the preceding claims, wherein a resonance frequency of the first impedance characteristic is at least 1.4 times larger than a resonance frequency of the second impedance characteristic.

8. A resonant DC-DC power converter according to any of the preceding claims,

25 wherein the output voltage or current control circuit comprises: a voltage or current regulation loop connected between the converter output voltage and a control terminal of the second controllable switch arrangement to adjust the converter output voltage in accordance with one or more DC reference voltage(s) or current(s).

30

9. A resonant DC-DC power converter according to claim 8, wherein the voltage or current regulation loop comprises:

a DC reference voltage generator configured to supply the one or more DC reference voltage(s) or current(s),

one or more comparator(s) configured for comparing the converter output voltage to at least a first DC reference voltage and select the conducting state or the nonconducting state of the second controllable switch arrangement depending on a result of the comparison.

5

30

10. A resonant DC-DC power converter according to claim 9, wherein the DC reference voltage generator is configured to supply at first DC reference voltage and a second DC reference voltage where the first DC reference voltage is higher than the second DC reference voltage;

10 wherein the one or more comparator(s) is configured to: comparing the converter output voltage to the first and second DC reference voltages; and selecting one of the conducting and non-conducting states of the second controlla-

ble switch arrangement in response to the converter output voltage exceeding the
 first DC reference voltage,
 selecting the opposite state of the second controllable switch arrangement in re sponse to the converter output voltage falls below the second DC reference voltage.

11. A resonant DC-DC power converter according to any of claims 5-10, wherein the

- self-oscillation feedback loop comprises:
 a first intrinsic switch capacitance coupled between the output and control terminals of the first controllable switch arrangement,
 a first bias voltage source configured to generate a first adjustable bias voltage,
 a first inductor, preferably with substantially fixed inductance, coupled in-between
- 25 the first bias voltage source and the control terminal of the first controllable switch arrangement.

12. A resonant DC-DC power converter according to any of the preceding claims, comprising a galvanic isolation barrier arranged between the input side circuit and the output side circuit.

13. A resonant DC-DC power converter according to claim 12, wherein the galvanic isolation barrier comprises first and second coupling capacitors.

13. A resonant DC-DC power converter according to any of the preceding claims, comprising a converter topology selected from a group of {class E, class F, class DE}.

5 14. A resonant DC-DC power converter according to any of the preceding claims, wherein the first controllable switch arrangement comprises one or more semiconductor switches and the second controllable switch arrangement comprises one or more semiconductor switches;

wherein each of the semiconductor switches comprises a semiconductor transistor

10 such as a MOSFET or IGBT such as a Gallium Nitride (GaN) or Silicon Carbide (SiC) MOSFET.

15. A resonant DC-DC power converter according to any of the preceding claims, further comprising:

15 a wireless or wired data receiver for receipt of remote data commands to control the switch state of the second controllable switch arrangement.

16. A resonant DC-DC power converter according to any of the preceding claims, further comprising:

20 a single substantially flat carrier substrate comprising a first surface and a second, opposing, surface,

wherein the input side circuit, the output side circuit, the rectification circuit, the resonant network, the first controllable switch arrangement, the second controllable switch arrangement and the output voltage or current control circuit are attached on

the first surface and/or the second surface.

17. A resonant DC-DC power converter according to claim 16, wherein the single substantially flat carrier substrate comprises a single sided or double-sided printed circuit board.

30

18. A method of operating a resonant DC-DC power converter to generate a converter output voltage or current, said method comprising steps of:a) alternatingly charging and discharging a resonant network of the resonant converter from an input voltage source by a first controllable switch arrangement in

accordance with a first switch control signal,

b) rectifying a resonant current of the resonant network,

c) discharging a rectified resonant current to a rectification circuit to produce a rectified output voltage,

5 d) switching between a first impedance characteristic or a second impedance characteristic of the resonant network by selecting a first or a second switch state of a second controllable switch arrangement,

e) adjusting a converter output voltage or current by alternatingly activating and interrupting the first switch control signal in accordance with first and second switch

10 states of the second controllable switch arrangement.



2/6





PCT/EP2015/053977

3/6



FIG. 2



FIG. 3



PCT/EP2015/053977

5/6





6/6





A.22 M. Madsen et. al.: "Confidential", DTU ref 95513





www.elektro.dtu.dk

Department of Electrical Engineering Electronics Group (ELE) Technical University of Denmark Oersteds Plads bygning 349 DK-2800 Kgs. Lyngby Denmark Tel: (+45) 45 25 25 25 Fax: (+45) 45 88 01 17 Email: info@elektro.dtu.dk